UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

| | Wasiii | mgton, B.O. 20043 | |
|---|---|---|---|
| | FC | ORM 10-K | |
| ☑ ANNUAL REPORT PUR | | 15(d) OF THE SECURITIES EXCHANGE ear ended December 31, 2018 or | ACT OF 1934 |
| ☐ TRANSITION REPORT I | PURSUANT TO SECTION 13 | OR 15(d) OF THE SECURITIES EXCHAI | NGE ACT OF 1934 |
| _ | For the transition period fro Commiss | m to sion File No. 001-7784 | |
| | 44. | Century Link ® | |
| | | istrant as specified in its charter) | |
| | Louisiana r other jurisdiction of ation or organization) | 72-065116 (I.R.S. Emplo Identification | pyer |
| 100 CenturyLin | k Drive, Monroe, Louisiana rincipal executive offices) | 71203 (Zip Code | |
| | (Registrant's teleph | 318) 388-9000 none number, including area code) ursuant to Section 12(b) of the Act: | |
| | Title of Each Class | Name of Each Exchange on Whic | ch Registered |
| Common S | Stock, par value \$1.00 per shar | re New York Stock Excha | ange |
| S | ecurities registered pursuan | t to Section 12(g) of the Act: Stock Option | ons |
| Indicate by check mark if the registrant is | a well-known seasoned issuer, a | as defined in Rule 405 of the Securities Act. Y | ∕es ⊠ No □ |
| Indicate by check mark if the registrant is | not required to file reports pursu | uant to Section 13 or Section 15(d) of the Act. | Yes □ No 🗷 |
| | | uired to be filed by Section 13 or 15(d) of the s required to file such reports) and (2) has be | |
| • | - | y every Interactive Data File required to be su d that the registrant was required to submit ar | • |
| | | 105 of Regulation S-K is not contained herein porated by reference in Part III of this Form 10 | |
| | | an accelerated filer, a non-accelerated filer, s ted filer", "smaller reporting company," and "e | |
| Large accelerated filer ■ | Accelerated filer □ | Non-accelerated filer □ | Smaller reporting company ☐ Emerging growth company ☐ |
| If an emerging growth company, indicate | by check mark if the registrant b | as elected not to use the extended transition | period for complying with any pow or |

DOCUMENTS INCORPORATED BY REFERENCE:

On February 22, 2019, 1,079,783,226 shares of common stock were outstanding. The aggregate market value of the voting stock held by non-affiliates as of

revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. □

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes $\ \square$

June 30, 2018 was \$17.7 billion .

Portions of the Registrant's Proxy Statement to be furnished in connection with the 2019 annual meeting of shareholders are incorporated by reference in Part III of this report.

TABLE OF CONTENTS

PART I

| Item 1. Business | <u>6</u> |
|--|------------|
| Item 1A. Risk Factors | <u>20</u> |
| Item 1B. Unresolved Staff Comments | <u>48</u> |
| Item 2. Properties | <u>49</u> |
| Item 3. Legal Proceedings | <u>49</u> |
| Item 4. Mine Safety Disclosures | <u>49</u> |
| <u>PART II</u> | |
| Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities | <u>50</u> |
| Item 6. Selected Financial Data | <u>51</u> |
| Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations | <u>53</u> |
| Item 7A. Quantitative and Qualitative Disclosures About Market Risk | <u>81</u> |
| Item 8. Consolidated Financial Statements and Supplementary Data | <u>82</u> |
| Consolidated Statements of Operations | <u>85</u> |
| Consolidated Statements of Comprehensive (Loss) Income | <u>86</u> |
| Consolidated Balance Sheets | <u>87</u> |
| Consolidated Statements of Cash Flows | <u>88</u> |
| Consolidated Statements of Stockholders' Equity | <u>89</u> |
| Notes to Consolidated Financial Statements | <u>90</u> |
| Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure | <u>153</u> |
| Item 9A. Controls and Procedures | <u>153</u> |
| <u>Item 9B. Other Information</u> | <u>155</u> |
| <u>PART III</u> | |
| Item 10. Directors, Executive Officers and Corporate Governance | <u>155</u> |
| Item 11. Executive Compensation | <u>155</u> |
| Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters | <u>156</u> |
| Item 13. Certain Relationships and Related Transactions and Director Independence | <u>156</u> |
| Item 14. Principal Accountant Fees and Services | <u>156</u> |
| <u>PART IV</u> | |
| Item 15. Exhibits and Financial Statement Schedules | <u>157</u> |
| Item 16. Summary Business and Financial Information | <u>169</u> |
| <u>Signatures</u> | <u>170</u> |
| | |

3

Unless the context requires otherwise, (i) references in this report on Form 10-K, for all periods presented, to "CenturyLink," "we," "us", the "Company" and "our" refer to CenturyLink, Inc. and its consolidated subsidiaries and (ii) references in this report to "Level 3" refer to Level 3 Communications, Inc., which we acquired and renamed Level 3 Parent, LLC, on November 1, 2017.

PART I

Special Note Regarding Forward-Looking Statements

This report and other documents filed by us under the federal securities law include, and future oral or written statements or press releases by us and our management may include, forward-looking statements about our business, financial condition, operating results and prospects. These "forward-looking" statements are defined by, and are subject to the "safe harbor" protections under, the federal securities laws. These statements include, among others:

- forecasts of our anticipated future results of operations, cash flows or financial position:
- statements concerning the anticipated impact of our transactions, investments, product development and other initiatives, including synergies or costs associated with our November 2017 combination with Level 3, the impact of our other acquisitions or dispositions, and the impact of our participation in government programs;
- · statements concerning the anticipated impact of the Tax Cuts and Jobs Act enacted in late 2017;
- statements about our liquidity, profit margins, tax position, tax rates, asset values, contingent liabilities, growth opportunities and growth
 rates, acquisition and divestiture opportunities, business prospects, regulatory and competitive outlook, market share, product
 capabilities, investment and expenditure plans, business strategies, dividend and stock repurchase plans, capital allocation plans,
 financing alternatives and sources, and pricing plans; and
- other similar statements of our expectations, beliefs, future plans and strategies, anticipated developments and other matters that are not historical facts, many of which are highlighted by words such as "may," "will," "would," "could," "should," "plan," "believes," "expects," "anticipates," "estimates," "projects," "intends," "likely," "seeks," "hopes," or variations or similar expressions with respect to the future.

These forward-looking statements are based upon our judgment and assumptions as of the date such statements are made concerning future developments and events, many of which are beyond our control. These forward-looking statements, and the assumptions upon which they are based, (i) are not guarantees of future results, (ii) are inherently speculative and (iii) are subject to a number of risks and uncertainties. Actual events and results may differ materially from those anticipated, estimated, projected or implied by us in those statements if one or more of these risks or uncertainties materialize, or if our underlying assumptions prove incorrect. All of our forward-looking statements are qualified in their entirety by reference to our discussion of factors that could cause our actual results to differ materially from those anticipated, estimated, projected or implied by us in those forward looking statements. Factors that could affect actual results include but are not limited to:

- the effects of competition from a wide variety of competitive providers, including decreased demand for our traditional wireline service offerings and increased pricing pressures;
- the effects of new, emerging or competing technologies, including those that could make our products less desirable or obsolete;
- our ability to attain our key operating imperatives, including simplifying and consolidating our network, simplifying and automating our service support systems, strengthening our relationships with customers and attaining projected cost savings;

- our ability to safeguard our network, and to avoid the adverse impact on our business from possible security breaches, service outages, system failures, equipment breakage, or similar events impacting our network or the availability and quality of our services;
- the effects of ongoing changes in the regulation of the communications industry, including the outcome of regulatory or judicial
 proceedings relating to intercarrier compensation, interconnection obligations, special access, universal service, broadband
 deployment, data protection and net neutrality;
- our ability to timely realize the anticipated benefits of our November 2017 combination with Level 3, including our ability to use Level 3's net operating losses in the amounts projected;
- our ability to effectively adjust to changes in the communications industry, and changes in the composition of our markets and product mix:
- possible changes in the demand for our products and services, including our ability to effectively respond to increased demand for high-speed data transmission services;
- our ability to successfully maintain the quality and profitability of our existing product and service offerings and to introduce profitable new offerings on a timely and cost-effective basis;
- our ability to generate cash flows sufficient to fund our financial commitments and objectives, including our capital expenditures, operating costs, debt repayments, dividends, pension contributions and other benefits payments;
- changes in our operating plans, corporate strategies, dividend payment plans or other capital allocation plans, whether based upon changes in our cash flows, cash requirements, financial performance, financial position, market conditions or otherwise;
- our ability to effectively retain and hire key personnel and to successfully negotiate collective bargaining agreements on reasonable terms without work stoppages;
- the negative impact of increases in the costs of our pension, health, post-employment or other benefits, including those caused by changes in markets, interest rates, mortality rates, demographics or regulations;
- adverse changes in our access to credit markets on favorable terms, whether caused by changes in our financial position, lower debt credit ratings, unstable markets or otherwise;
- our ability to meet the terms and conditions of our debt obligations, including our ability to make transfers of cash in compliance therewith;
- our ability to maintain favorable relations with our key business partners, suppliers, vendors, landlords and financial institutions;
- · our ability to collect our receivables from financially troubled customers;
- · any adverse developments in legal or regulatory proceedings involving us;
- changes in tax, communications, pension, healthcare or other laws or regulations, in governmental support programs, or in general
 government funding levels;
- the effects of changes in accounting policies, practices or assumptions, including changes that could potentially require future impairment charges;
- the effects of adverse weather, terrorism or other natural or man-made disasters;
- the effects of more general factors such as changes in interest rates, in exchange rates, in operating costs, in public policy, in the views of financial analysts, or in general market, labor, economic or geo-political conditions; and

other risks referenced in "Risk Factors" in Item 1A or elsewhere in this report or other of our filings with the SEC.

Additional factors or risks that we currently deem immaterial, that are not presently known to us or that arise in the future could also cause our actual results to differ materially from our expected results. Given these uncertainties, investors are cautioned not to unduly rely upon our forward-looking statements, which speak only as of the date made. We undertake no obligation to publicly update or revise any forward-looking statements for any reason, whether as a result of new information, future events or developments, changed circumstances, or otherwise. Furthermore, any information about our intentions contained in any of our forward-looking statements reflects our intentions as of the date of such forward-looking statement, and is based upon, among other things, existing regulatory, technological, industry, competitive, economic and market conditions, and our assumptions as of such date. We may change our intentions, strategies or plans (including our dividend or other capital allocation plans) at any time and without notice, based upon any changes in such factors, in our assumptions or otherwise.

ITEM 1. BUSINESS

Overview

We are an international facilities-based communications company engaged primarily in providing a broad array of integrated services to our business and residential customers. Our specific products and services are detailed below under the heading "Operations - Products and Services."

With approximately 450,000 route miles of fiber optic cable globally, we believe we are among the largest providers of communications services to domestic and global enterprise customers. Our terrestrial and subsea fiber optic long-haul network throughout North America, Europe, Latin America and Asia Pacific connects to metropolitan fiber networks that we operate. We provide services in over 60 countries, with most of our revenue being derived in the United States. We believe we are the third largest wireline telecommunications company in the United States, based on the number of access lines we serve.

We were incorporated under the laws of the State of Louisiana in 1968. Our principal executive offices are located at 100 CenturyLink Drive, Monroe, Louisiana 71203 and our telephone number is (318) 388-9000.

For a discussion of certain risks applicable to our business, see "Risk Factors" in Item 1A of Part I of this report. The summary financial information in this Item 1 should be read in conjunction with, and is qualified by reference to, our consolidated financial statements and notes thereto in Item 8 of Part II of this report and "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of Part II of this report.

Acquisition of Level 3

On November 1, 2017, CenturyLink acquired Level 3 through successive merger transactions, including a merger of Level 3 with and into a merger subsidiary, which survived such merger as our indirect wholly-owned subsidiary under the name of Level 3 Parent, LLC. Upon closing, CenturyLink shareholders owned approximately 51% and former Level 3 shareholders owned approximately 49% of the combined company.

For additional information about our acquisition of Level 3, see (i) Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report, (ii) our current report on Form 8-K/A filed by us with the Securities and Exchange Commission (the "SEC") on January 16, 2018, (iii) our current report on Form 8-K filed by us with the SEC on November 1, 2017 and (iv) the definitive joint proxy statement/prospectus filed by us with the SEC on February 13, 2017.

Sale of Data Centers and Colocation Business

On May 1, 2017, we sold a portion of our data centers and colocation business to a consortium led by BC Partners, Inc. and Medina Capital ("the Purchaser") in exchange for pre-tax cash proceeds of \$1.8 billion and a minority stake in the limited partnership that owns the consortium's newly-formed global secure infrastructure company, Cyxtera Technologies ("Cyxtera"). As part of the transaction, the Purchaser acquired 57 of our data centers and assumed \$294 million (as of May 1, 2017) of our capital lease obligations related to the divested properties.

See Note 3—Sale of Data Centers and Colocation Business to our consolidated financial statements in Item 8 of Part II of this report for additional information.

Financial Highlights

Our consolidated operating results and financial position include the operating results and financial position of Level 3 beginning as of November 1, 2017. For additional information, see Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report.

The following table summarizes the results of our consolidated operations.

| | Years Ended December 31, | | |
|--------------------|--------------------------|-------------|---------------------|
| | 2018 (1)(2)(3) | 2017 (1)(2) | 2016 ⁽¹⁾ |
| | (Dollars in millions) | | |
| | | | |
| Operating revenue | \$ 23,443 | 17,656 | 17,470 |
| Operating expenses | 22,873 | 15,647 | 15,137 |
| Operating income | \$ 570 | 2,009 | 2,333 |
| Net (loss) income | \$ (1,733) | 1,389 | 626 |
| | <u> </u> | · | |

- (1) During 2018, 2017 and 2016, we incurred Level 3 acquisition-related expenses of \$393 million, \$271 million and \$52 million, respectively. For additional information, see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Acquisition of Level 3" and Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report.
- (2) The enactment of the Tax Cuts and Jobs Act in December 2017 resulted in a re-measurement of our deferred tax assets and liabilities at the new federal corporate tax rate of 21%. The re-measurement resulted in tax expense of \$92 million and a tax benefit of approximately \$1.1 billion for 2018 and 2017, respectively.
- (3) During 2018, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill attributed to our consumer segment.

We estimate that during 2018 approximately 7.9% of our consolidated revenue was derived from providing telecommunications, colocation and hosting services outside the United States.

The following table summarizes certain selected financial information from our consolidated balance sheets:

| | | As of December 31, | | |
|----------------------------|----|-----------------------|--------|--|
| | ; | 2018 | 2017 | |
| | | (Dollars in millions) | | |
| | | | | |
| Total assets | \$ | 70,256 | 75,611 | |
| Total long-term debt (1) | | 36,061 | 37,726 | |
| Total stockholders' equity | | 19,828 | 23,491 | |

⁽¹⁾ For additional information on our total long-term debt, see Note 6—Long-Term Debt and Credit Facilities to our consolidated financial statements in Item 8 of Part II of this report. For information on our total obligations, see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources—Future Contractual Obligations" in Item 7 of Part II of this report.

Operations

Segments

At December 31, 2018, we had the following two segments:

- Business Segment. Under our business segment, we provide our products and services to large domestic and global enterprises, small
 and medium businesses, federal, state and local governments and wholesale customers, including other communication providers. Our
 products and services offered to these customers include our IP and Data Services suite of products, which includes VPN and hybrid
 networking, Ethernet and IP services; Transport and Infrastructure, which includes wavelengths and private line, dark fiber, colocation
 and data center services, and professional services; Voice Services, which includes local, long-distance, toll-free and unified
 communications services; and IT and Managed services, all of which are described further under "Products and Services"; and
- Consumer Segment. Under our consumer segment, we provide our products and services to residential customers. Our products and services offered to these customers include our broadband, local and long-distance voice, video and other ancillary services.

The following table shows the composition of our operating revenue by segment under our current segment categorization for the years ended December 31, 2018, 2017 and 2016:

| | Years Ended December 31, | | Percent Change | | |
|-------------------------|--------------------------|------|----------------|--------------|--------------|
| | 2018 | 2017 | 2016 | 2018 vs 2017 | 2017 vs 2016 |
| Percentage of revenue: | | | | | |
| Business segment | 74% | 64% | 61% | 10 % | 3 % |
| Consumer segment | 23% | 32% | 35% | (9)% | (3)% |
| Non-segment revenue* | 3% | 4% | 4% | (1)% | — % |
| Total operating revenue | 100% | 100% | 100% | | |

^{*} Consists of all revenue not attributable to our segment revenue.

For additional information on our segment data, including information on certain centrally-managed assets and expenses not reflected in our segment results, see Note 15—Segment Information to our consolidated financial statements in Item 8 of Part II of this report and "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of Part II of this report.

Our Chief Executive Officer, who is also our Chief Operating Decision Maker ("CODM") continues to review the operational and internal reporting structure as a result of our acquisition of Level 3, see Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report.

Products and Services

Global enterprises, governmental entities and regional organizations depend on our wide variety of technologies and the services engineered to work in conjunction with them. These range from specific offerings such as networks or cloud-based application hosting to complex multi-layered engagements where we develop custom solutions involving numerous technologies and professional consulting services. In many cases, enterprises engage with us to outsource many of their IT functions so they can focus on their core business.

While most of our customized customer interactions involve multiple integrated technologies and services, we organize our products and services according to the core technologies that drive them. We report our related revenue under the following categories: IP and data services, transport and infrastructure services, voice and collaboration services, IT and managed services and regulatory revenue, each of which is described in further detail below.

IP and Data Services

- VPN Data Network. Built on our extensive optical transport network, we create private networks tailored to our customers' needs. These technologies enable service providers, enterprises and government entities to streamline multiple networks into a single, cost-effective solution that simplifies the transmission of voice, video, and data over a single secure network;
- Ethernet. We deliver a robust array of networking services built on Ethernet technology. Ethernet services include point-to-point and multi-point equipment configurations that facilitate data transmissions across metropolitan areas and larger enterprise-class wide area networks. Our Ethernet technology is also used by wireless service providers for data transmission via our fiber-optic cables connected to their towers;
- Internet Protocol ("IP"). Our Internet Protocol services provide global internet access over a high performance, diverse network with
 connectivity in more than 60 countries with over 72 Tbps of global throughput. Our network features over 48 Tbps of global peering
 capacity, and spans approximately 450,000 route miles globally with extensive off-net access solutions across North America, Europe,
 Latin America and Asia Pacific; and
- Video . To our residential customers, we make available our own branded facilities-based Prism TV service, plus satellite digital television under an arrangement with DIRECTV.

Transport and Infrastructure

- Broadband. Our broadband services deliver a cost-effective Internet connection through existing telephone lines or fiber-optic cables while companies enjoy high speed data transfer services. A substantial portion of our broadband subscribers are located within the local service area of our wireline telephone operations.
- Private Line. We deliver private line (including business data services) services, a direct circuit or channel specifically dedicated for connecting two or more organizational sites. Private line service offers a high-speed, secure solution for frequent transmission of large amounts of data between sites, including wireless backhaul transmissions;
- Wavelength. We deliver high bandwidth optical networks to firms requiring an end-to-end transport solution with Ethernet technology by contracting for a scalable amount of bandwidth connecting sites or providing high-speed access to cloud computing resources;
- Colocation and Data Center Services. We provide different options for organizations' data center needs. Our data center services
 range from dedicated hosting and cloud services to more complex managed solutions, including disaster recovery, business continuity,
 applications management support and security services to manage mission critical applications;
- Professional Services. Our experts deliver a robust array of consulting services to organizations either as part of a larger engagement or
 as stand-alone services. This category includes network management, installation and maintenance of data equipment and the building
 of proprietary fiber-optic broadband networks for government and business customers; and
- Dark Fiber. We possess an extensive array of unlit optical fiber, known as "dark fiber." Many large enterprises are interested in building
 their networks with this high-bandwidth, highly secure optical technology and dark fiber gives them access to the technology.
 CenturyLink provides professional services to engineer these networks, and in some cases, manage them for customers.

Voice and Collaboration

 Voice. We offer to our business and residential customers a complete portfolio of traditional Time Division Multiplexing voice services including Primary Rate Interface service, local inbound service, switched one-plus, toll free, long distance and international services; and Voice Over IP (VoIP). We deliver a broad range of local and enterprise voice and data services built on VoIP (Voice over Internet Protocol) technology. Our local and enterprise voice services include VoIP enhanced local service, national and multinational SIP Trunking, Hosted VoIP, support of Primary Rate Interface service, long distance service, and toll-free service.

IT and Managed Services

Managed Services. We craft technology solutions for our customers and often manage those solutions on an ongoing basis. Managed services represent a blend of network, hosting, cloud (public and private), and IT services that typically require ongoing support such as managing applications, operating systems and hardware. This product line includes intuitive management tools that optimize efficiencies in companies' technology infrastructure. These services frequently enhance equipment or networks owned, acquired or controlled by the customer and often include our consulting or software development.

Regulatory Revenue

- Universal Service Fund ("USF") support payments. We receive federal and state USF support payment subsidies designed to reimburse us for various costs related to certain telecommunications services, including the costs of deploying, maintaining and operating voice and broadband infrastructure in high-cost rural areas where we are not able to fully recover our costs from our customers;
- Connect America Fund ("CAF"). We receive federal support payments from both Phase I and Phase II of the CAF program. The funding from the CAF Phase II support program has substantially replaced the funding from the interstate USF program that we previously utilized to support voice services in high-cost rural markets in 33 states; and
- Other. We generate other operating revenue from the leasing and subleasing of space in our office buildings, warehouses and other properties and from rental income associated with our failed-sale-leaseback. For additional information on our failed-leaseback transaction, see "Sale of Data Centers and Colocation Business-Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of Part II of this report.

Throughout 2018, we centrally managed the activities that generate this regulatory and other revenue and as such this revenue was not included in our segment revenue. Beginning in 2019, we will report USF support payments and CAF support revenue as part of our Consumer segment. The adoption of the new lease accounting standard results in us no longer reporting rental income associated with the 2017 failed-sale-leaseback.

From time to time, we may continue to change the categorization of our products and services.

Additional Information

From time to time, we also make investments in other communications or technology companies.

For further information on regulatory, technological and competitive factors that could impact our revenue, see "Regulation" and "Competition" under this Item 1 below and "Risk Factors" under Item 1A below. For more information on the financial contributions of our various services, see "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of Part II of this report.

Our Network

Most of our products and services are provided using our telecommunications network, which consists of fiber-optic and copper cables, high-speed transport equipment, electronics, voice switches, data switches and routers, and various other equipment. Our local exchange carrier networks also include central offices and remote site assets, and form a portion of the public switched telephone network. A substantial portion of our equipment operates with licensed software.

At December 31, 2018, our network included:

- Approximately 450,000 route miles of fiber optic plant;
- · Approximately 910,000 miles of copper plant;
- · More than 360 colocation facilities and data centers globally;
- Approximately 37,500 route miles of subsea fiber optic cable systems:
- More than 150,000 Fiber On-net buildings;
- Multiple gateway and transmission facilities used in connection with operating our network throughout North America, Europe and Latin America; and
- Central office and other equipment that enables us to provide telephone service as an incumbent local telephone company ("ILEC") in 37 states.

We continue to enhance and expand our network by deploying various technologies to provide additional capacity to our customers. Rapid and significant changes in technology are expected to continue in the telecommunications industry. Our future success will depend, in part, on our ability to anticipate and adapt to changes in technology and customer demands, including demands for enhanced digitization, automation and customer self-service capabilities. In addition, we anticipate that continued increases in internet usage by our customers will require us to make significant capital expenditures to increase network capacity or to implement network management practices to alleviate network capacity shortages. The FCC's definition of broadband service could create additional requirements for higher capital spending to address marketing and competitive issues. Any such additional expenditure could adversely impact our results of operations and financial condition.

Similarly, we continue to take steps to simplify and modernize our network, much of which we have assembled through acquisitions. To attain our objectives, we plan to continue to pursue several complex projects that we expect will be costly and may take several years to complete. The costs of these projects could materially increase if we conclude that we need to replace any or all of our legacy systems.

Like other large communications companies, we are a constant target of cyber-attacks of varying degrees, which has caused us to spend increasingly more time and money to deal with increasingly sophisticated attacks. Some of the attacks result in security breaches, and we periodically notify our customers, our employees or the public of these breaches when necessary or appropriate. None of these resulting security breaches to date have materially adversely affected our business, results of operations or financial condition.

Similarly, like other large communication companies operating complex networks, from time to time in the ordinary course of our business we experience disruptions in our service. Although none of these outages have thus far materially adversely affected us, certain of these outages have resulted in regulatory fines, negative publicity, service credits and other adverse consequences.

We rely on several other communications companies to provide our offerings. We lease a portion of our core fiber network from our competitors and other third parties. Many of these leases will lapse in future years. A portion of our services are provided by other carriers under agency agreements or through reselling arrangements with other carriers. Our future ability to provide services on the terms of our current offerings will depend in part upon our ability to renew or replace these leases, agreements and arrangements on terms substantially similar to those currently in effect.

For additional information regarding our systems, network assets, network risks, capital expenditure requirements and reliance upon third parties, see "Risk Factors," generally, in Item 1A of Part I of this report, and, in particular, "Risk Factors—Risks Affecting Our Business" and "Risk Factors—Risks Affecting Our Liquidity and Capital Resources." For more information on our properties, see Item 2 of Part I of this report.

Patents, Trade Names, Trademarks and Copyrights

Either directly or through our subsidiaries, we have rights in various patents, trade names, trademarks, copyrights and other intellectual property necessary to conduct our business. Our services often use the intellectual property of others, including licensed software. We also occasionally license our intellectual property to others as we deem appropriate.

Through acquisitions or our own research and development, as of December 31, 2018, we had approximately 3,000 patents and patent applications in the United States and other countries. Our patents cover a range of technologies, including those relating to data and voice services, content distribution and transmission and networking equipment. We have also received licenses to use patents held by others, including through certain extensive cross-license arrangements. Patents give us the right to prevent others, particularly competitors, from using our proprietary technologies. Patent licenses give us the freedom to operate our business without the risk of interruption from the holder of the patented technology. We plan to continue to file new patent applications as we enhance and develop products and services, and we plan to continue to seek opportunities to expand our patent portfolio through strategic acquisitions and licensing.

We periodically receive offers from third parties to purchase or obtain licenses for patents and other intellectual property rights in exchange for royalties or other payments. We also periodically receive notices, or are named in lawsuits, alleging that our products or services infringe on patents or other intellectual property rights of third parties. In certain instances, these matters can potentially adversely impact our operations, operating results or financial position. For additional information, see "Risk Factors—Risks Affecting Our Business" in Item 1A of Part I of this report, and Note 17—Commitments, Contingencies and Other Items to our consolidated financial statements in Item 8 of Part II of this report.

Sales and Marketing

We maintain local offices in most major and secondary markets within the U.S. and in most of the larger population centers within our local service area and in many of the primary markets of the more than 60 countries in which we provide services. These offices provide sales and customer support services to the communities in our local markets. We also rely on our call center personnel and a variety of channel partners to promote sales of services that meet the needs of our customers. Our sales and marketing strategy is to enhance our sales by offering solutions tailored to the needs of our various customers and promoting our brands. To meet the needs of different customers, our offerings include both standalone services and bundled services designed to provide a complete offering of integrated services.

We conduct most of our operations under the brand name "CenturyLink." Our satellite television service is offered on a co-branded basis under the "DIRECTV" name.

Our sales and marketing approach to our business customers includes a commitment to provide comprehensive communications and IT solutions for business, wholesale and government customers of all sizes, ranging from small business offices to the world's largest global enterprise customers. We strive to offer our business customers stable, reliable, secure and trusted solutions. Our marketing plans include marketing our products and services primarily through direct sales representatives, inbound call centers, telemarketing and third parties, including telecommunications agents, system integrators, value-added resellers and other telecommunications firms. We support our distribution through digital advertising, events, television advertising, website promotions and public relations.

Similarly, our sales and marketing approach to our residential customers emphasizes customer-oriented sales, marketing and service with a local presence. Our marketing plans include marketing our products and services primarily through direct sales representatives, inbound call centers, telemarketing and third parties, including retailers, satellite television providers, door to door sales agents and digital marketing firms. We support our distribution with digital marketing, direct mail, bill inserts, newspaper and television advertising, website promotions, public relations activities and sponsorship of community events and sports venues.

Regulation

Overview

Our domestic operations are regulated by the Federal Communications Commission (the "FCC"), various state utility commissions and occasionally by local agencies. Our non-domestic operations are regulated by supranational groups (such as the European Union, or EU), national agencies and frequently state, provincial or local bodies. Generally, we must obtain and maintain operating licenses from these bodies in most areas where we offer regulated services.

The following description discusses some of the major industry regulations that affect our operations, but numerous other regulations not discussed below also have a substantial impact on us. For additional information, see "Risk Factors" in Item 1A of Part I of this report.

Federal Regulation of Domestic Operations

General

The FCC regulates the interstate services we provide, including the business data service charges we bill for wholesale network transmission and intercarrier compensation, including the interstate access charges that we bill to long-distance companies and other communications companies in connection with the origination and termination of interstate phone calls. Additionally, the FCC regulates a number of aspects of our business related to international communications services, privacy, public safety and network infrastructure, including our access to and use of local telephone numbers and our provision of emergency 911 services.

In April 2017, the FCC revised the regulation of the provision of data services to businesses providing greater freedom to respond to competition for these business data services. Several parties appealed the FCC's decision and the resulting remand proceedings remain pending. Their ultimate impact on us of the FCC's recent action is currently unknown.

Universal Service

In 2015, we accepted Connect America Fund or "CAF" funding from the FCC of approximately \$500 million per year for six years to fund the deployment of voice and broadband capable infrastructure for approximately 1.2 million rural households and businesses in 33 of the 37 states in which we are an ILEC under the CAF II high-cost support program. The funding from the CAF II support program in these 33 states has substantially replaced the funding from the interstate USF high-cost program that we previously utilized to support voice services in high-cost rural markets in these 33 states. As a result of accepting CAF II support payments for 33 states, as well as existing merger-related commitments, we are obligated to make substantial capital expenditures to build infrastructure by certain specified milestone deadlines. For information on the risks associated with participating in this program, see "Risk Factors—Risks Relating to Legal and Regulatory Matters" in Item 1A of Part I of this report.

Currently, it is unclear when the FCC will complete the development of the next phase of the CAF program expected to follow the CAF Phase II program and what that plan will entail. Although the FCC has taken certain initial steps, we cannot assure you when such replacement plan will be proposed or implemented.

For additional information about the potential financial impact of the CAF II program, see "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of Part II of this report.

Broadband Regulation

In February 2015, the FCC adopted an order classifying Broadband Internet Access Services ("BIAS") under Title II of the Communications Act of 1934 and applying new regulations. In December 2017, the FCC voted to repeal most of those regulations and the classification of BIAS as a Title II service and to preclude states from imposing substantial regulations of their own on broadband. Opponents of this change have appealed this action in federal court and have advocated in favor of re-instituting regulation of Internet services under Title II of the Communications Act. Several states have also opposed the change and have initiated state executive orders or introduced legislation focused on state-specific Internet service regulation. The result of these appeals is pending and the potential impact to CenturyLink is currently unknown.

State Regulation of Domestic Operations

In recent years, most states have reduced their regulation of ILECs. Nonetheless, state regulatory commissions generally continue to (i) set the rates that telecommunication companies charge each other for exchanging traffic, (ii) administer support programs designed to subsidize the provision of services to high-cost rural areas, (iii) regulate the purchase and sale of ILECs, (iv) require ILECs to provide service under publicly-filed tariffs setting forth the terms, conditions and prices of regulated services, (v) limit ILECs' ability to borrow and pledge their assets, (vi) regulate transactions between ILECs and their affiliates and (vii) impose various other service standards. In most states, switched and business data services and interconnection services are subject to price regulation, although the extent of regulation varies by type of service and geographic region.

We operate in states where traditional cost recovery mechanisms, including state USF, are under evaluation or have been modified. Recent changes to the federal tax code have prompted several states to review regulated rates. As laws and regulations change, there can be no assurance that these mechanisms will continue to provide us with the same level of cost recovery.

International Regulations

Our subsidiaries operating outside of the United States are subject to various regulations in the markets where service is provided. The scope of regulation varies from country to country. The telecommunications regulatory regimes in certain of our non-domestic markets are in the process of development. Many issues, including the pricing of services, have not been addressed fully, or even at all. We cannot accurately predict whether and how these issues will be resolved, or their effect on our operations. Further, some of the legal requirements governing our foreign operations are more restrictive than or conflict with those governing our domestic operations, which raises our compliance costs and regulatory risks.

In June 2016, following a national referendum, the United Kingdom (the "UK") elected to terminate its membership in the EU ("Brexit"). The UK will remain a member of the EU until at least March 29, 2019. Several factors which are currently unknown will influence Brexit's impact on our business, including the form Brexit will take. We operate a staging facility in the UK, where certain core network elements and customer premises equipment is configured before being shipped to both UK and EU locations. The UK is currently also the central location of our stores of spare replenishment in our European operations. In respect of our UK workforce, we do not anticipate any adverse impact from Brexit as only a small percentage of the workforce are EU nationals. The same is true of UK nationals working in our EU located workforce. We are currently monitoring Brexit developments, reviewing our supply chain alternatives, and assessing the short and long-term implications of Brexit on our operations. Nonetheless, based on current information, we do not anticipate Brexit will have a substantial impact on our business.

Other Regulations

Our networks are subject to numerous local regulations, including codes that regulate our trenching and construction operations or that require us to obtain permits, licenses or franchises to operate. Such regulations are enacted by municipalities, counties or other regional governmental bodies, and can vary widely from jurisdiction to jurisdiction as a result. Such regulations may also require us to pay substantial fees.

Various foreign, federal and state laws govern our storage, maintenance and use of customer data, including a wide range of consumer protection, data protection, privacy, intellectual property and similar laws. The application, interpretation and enforcement of these laws are often uncertain, and may be interpreted and applied inconsistently from jurisdiction to jurisdiction. Various foreign, federal and state legislative or regulatory bodies have recently adopted increasingly restrictive laws or regulations governing the protection or retention of data, and others are contemplating similar actions. In particular, regulatory bodies in Europe have aggressively enforced the stringent terms of the EU's General Data Protection Regulation.

For additional information about these matters, see "Risk Factors—Risks Affecting Our Business" and "Risk Factors—Risks Relating to Legal and Regulatory Matters" in item 1A of Part I of this report.

Competition

General

We compete in a rapidly evolving and highly competitive market, and we expect intense competition from a wide variety of sources under evolving market conditions to continue. In addition to competition from larger telecommunication service providers, we are facing increasing competition from several other sources, including cable and satellite companies, wireless providers, technology companies, cloud companies, broadband providers, device providers, resellers, sales agents facilities-based providers, and smaller more narrowly focused niche providers either using their own networks, leasing parts of our network or providing services such as VoIP, SD-WAN and security independent of network assets. The ability to provide such services independently of the network challenges more traditional solution selling. Further technological advances and regulatory and legislative changes have increased opportunities for a wide range of alternative communications service providers, which in turn have increased competitive pressures on our business. These alternate providers often face fewer regulations and have lower cost structures than we do. In addition, the communications industry has, in recent years, experienced substantial consolidation, and some of our competitors in one or more lines of our business are generally larger, have stronger brand names, have more financial and business resources and have broader service offerings than we currently do. In certain overseas markets, we compete against national incumbent telecommunications providers and other regional or international companies that may have a longer history of providing service in the market.

We compete to provide services to business customers based on a variety of factors, including the comprehensiveness and reliability of our network, our data transmission speeds, price, the latency of our available intercity and metro routes, the scope of our integrated offerings, the reach and peering capacity of our IP network, and customer service. Depending on the applicable market and requested services, competition can be intense, especially if one or more competitors in the market have network assets better suited to the customer's needs or are offering faster transmission speeds or lower prices.

Wireless voice services are a significant source of competition with our traditional ILEC services. It is increasingly common for customers to completely forego use of traditional wireline phone service and instead rely solely on wireless service for voice services. We anticipate this trend will continue, particularly with younger customers who are less accustomed to using traditional wireline voice services. Technological and regulatory developments in wireless services, Wi-Fi, and other wired and wireless technologies have contributed to the development of alternatives to traditional landline voice services. Moreover, the growing prevalence of electronic mail, text messaging, social networking and similar digital non-voice communications services continues to reduce the demand for traditional landline voice services. These factors have led to a long-term systemic decline in the number of our wireline voice service customers.

The Telecommunications Act of 1996, which obligates ILECs to permit competitors to interconnect their facilities to the ILEC's network and to take various other steps that are designed to promote competition, imposes several duties on an ILEC if it receives a specific request from another entity which seeks to connect with or provide services using the ILEC's network. In addition, each ILEC is obligated to (i) negotiate interconnection agreements in good faith, (ii) provide nondiscriminatory "unbundled" access to all aspects of the ILEC's network, (iii) offer resale of its telecommunications services at wholesale rates and (iv) permit competitors, on terms and conditions (including rates) that are just, reasonable and nondiscriminatory, to colocate their physical plant on the ILEC's property, or provide virtual colocation if physical colocation is not practicable. Current FCC rules require ILECs to lease a network element only in those situations where competing carriers genuinely would be impaired without access to such network elements, and where the unbundling would not interfere with the development of facilities-based competition.

As a result of these regulatory, consumer and technological developments, ILECs also face competition from competitive local exchange carriers, or CLECs, particularly in densely populated areas. CLECs provide competing services through (i) reselling an ILEC's local services, (ii) using an ILEC's unbundled network elements or (iii) operating their own facilities.

Technological developments have led to the development of new products and services that have reduced the demand for our traditional services, as noted above, or that compete with traditional ILEC services. Technological improvements have enabled cable television companies to provide traditional circuit-switched telephone service over their cable networks, and several national cable companies continue to aggressively market these services. Similarly, companies providing VoIP services provide voice communication services over the Internet which compete with our traditional telephone service and our own VoIP services. In addition, demand for our broadband services could be adversely affected by advanced wireless data transmission technologies being deployed by wireless providers and by certain technologies permitting cable companies and other competitors to deliver generally faster average broadband transmission speeds than ours.

Similar to us, many cable, technology or other communications companies that previously offered a limited range of services are now offering diversified bundles of services, either through their own networks, reselling arrangements or joint ventures. As such, a growing number of companies are competing to serve the communications needs of the same customer base. Such activities will continue to place downward pressure on the demand for and pricing of our services.

As both residential and business customers increasingly demand high-speed connections for entertainment, communications and productivity, we expect the demands on our network will continue to increase over the next several years. To succeed, we must continue to invest in our networks to ensure that they can deliver competitive services that meet these increasing bandwidth and speed requirements. In addition, network reliability and security are increasingly important competitive factors in our business.

In addition to facing direct competition from those providers described above, ILECs increasingly face competition from alternate communication systems constructed by long distance carriers, large customers, municipalities or alternative access vendors. These systems are capable of originating or terminating calls without use of an ILEC's networks or switching services. Other potential sources of competition include non-carrier systems that are capable of bypassing ILECs' local networks, either partially or completely, through various means, including the provision of business data services or independent switching services and the concentration of telecommunications traffic on a few of an ILEC's access lines. We anticipate that all these trends will continue and lead to decreased billable use of our networks.

Additional information about competitive pressures is located (i) under the heading "Risk Factors—Risks Affecting Our Business" in Item 1A of Part I of this report and (ii) in the discussion immediately below, which contains more specific information on how these trends in competition have impacted our segments.

Business Segment

In connection with providing services to our business customers, which includes our small, medium and enterprise business, wholesale and government customers, we compete against other telecommunication providers, as well as other regional, national and international carriers, other data transport providers, cable companies, CLECs and other enterprises, some of whom are substantially larger than us. Competition is based on price, bandwidth, quality and speed of service, promotions and bundled offerings. In providing broadband services, we compete primarily with cable companies, wireless providers, technology companies and other broadband service providers. We face competition in Ethernet based services in the wholesale market from cable companies and fiber based providers.

Our competitors for providing integrated data, broadband, voice services and other IT services to our business customers range from midsized businesses to large enterprises. Due to the size and capacity of some of these companies, our competitors may be able to offer more inexpensive solutions to our customers. To compete, we focus on providing sophisticated, secure and performance-driven services to our business customers through our global infrastructure.

The number of companies providing business services has grown and increased competition for these services, particularly with respect to smaller business customers. Many of our competitors for business services are not subject to the same regulatory requirements as we are and therefore they are able to avoid significant regulatory costs and obligations.

Our competitors for cloud, hosting, colocation and other IT services include telecommunications companies, technology companies, cloud companies, colocation companies, hardware manufacturers and system integrators that support the in-house IT operations for a business or offer outsourcing solutions. Due to the size, capacity and strategically low pricing tactics of some of these companies, our competitors may be able to offer more inexpensive solutions to our customers. The increase in recent years in the number of companies providing these services has placed substantial downward pressure on pricing for a wide range of cloud, hosting, colocation and other IT services. To address these competitive pressures, we have focused on offering end-to-end integrated customer solutions which we believe could help differentiate our products and services from those offered by competitors with a narrower range of products and services.

For our wholesale customers, we will continue to be adversely affected by product substitution, technological migration, industry consolidation and mandated rate reductions. Competition for private line services is based on price, network reach and reliability, service, promotions and bundled offerings. We face significant competition for access services from CLECs, cable companies, resellers and wireless service providers as well as some of our own wholesale markets customers, many of which are deploying their own networks to provide customers with local services. By doing so, these competitors reduce revenue producing traffic on our network.

In providing equipment sales and professional services to our business customers, we compete primarily with large integrators, equipment providers and national telecommunication providers. Competition is based on package offerings, and as such our strategy is to provide these customers individualized and customizable packages that include other services. As such, in providing data integration we often face many of the same competitive pressures as we face in providing other services, as discussed above.

We expect equipment sales and professional services revenue to continue to fluctuate from quarter to quarter as these offerings tend to be more sensitive than others to changes in the economy and in spending trends of our governmental customers. We further expect the profit margins on our equipment sales and professional services offerings to continue to be lower than those of our other services.

Consumer Segment

With respect to providing our services to residential customers, technology advancements have increased both the quantity and type of competitors that we compete with for our services. More specifically, voice services face significant product and technology substitution. Additionally, wireless carriers' latest generation technologies are allowing them to more directly compete with our Broadband services. The fragmentation of the video market with the proliferation of Over the Top providers has made it difficult for us to offer a cost-effective video product. Lastly, the regulatory environment in which we operate, while it provides us certain advantages, can make us less nimble than cable, wireless, and other technology companies.

As a result, our strategy for competing in the consumer space is to continue to invest in our network with fiber solutions to increase connection speeds and service quality, partner with video providers such as DIRECTV to provide video and content options to customers, and encourage customers to bundle voice services by providing a high quality voice connection with discounts for bundling. In addition, we believe initiatives to improve the customer experience and digital experience should increase customer loyalty over time.

The domestic consumer market for broadband services is mature, with a significant portion of households already receiving those services. We compete for customers on the basis of pricing, packaging of services and features and quality of service. In order to remain competitive, we believe continually increasing connection speeds is important. As a result, we continue to invest in our network, which allows for the delivery of higher speed broadband services.

Although our status as an ILEC in our local service areas continues to provide us advantages in providing local services in those territories, as noted above, we increasingly face significant competition as an increasing number of consumers are willing to substitute cable, wireless and electronic communications for traditional voice telecommunications services. This has led to an increase in the number and type of competitors within our industry, price compression and a decrease in our market share. As a result of this product substitution, we face greater competition in providing local and long-distance voice services from wireless providers, resellers and sales agents (including ourselves), social media hosts and broadband service providers, including cable companies. We also continue to compete with traditional telecommunications providers, such as national carriers, smaller regional providers, CLECs and independent telephone companies.

Acquisitions and Dispositions

Since being incorporated in 1968, we have grown principally through acquisitions. By 2008, we had become one of the largest providers of rural telephone services in the United States. Since then, we acquired Embarq Corporation in mid-2009, Qwest Communications International Inc. in early 2011 and Level 3 Communications, Inc. in late 2017. These acquisitions have substantially changed our customer base, geographic footprint and mix of products and services.

We regularly evaluate the possibility of acquiring additional assets or disposing of assets in exchange for cash, securities or other properties, and at any given time may be engaged in discussions or negotiations regarding additional acquisitions or dispositions. We generally do not announce our acquisitions or dispositions until we have entered into a preliminary or definitive agreement.

See above under "Acquisition of Level 3", for additional information about our November 1, 2017 acquisition of Level 3, and the "Sale of Data Centers and Colocation Business" for additional information about our May 1, 2017 disposition.

See Note 4—Goodwill, Customer Relationships and Other Intangible Assets for additional information on these acquisitions.

Environmental Compliance

From time to time we may incur environmental compliance and remediation expenses, mainly resulting from owning or operating prior industrial sites or operating vehicle fleets or power supplies for our communications equipment. Although we cannot assess with certainty the impact of any future compliance and remediation obligations or provide you with any assurances regarding the ultimate impact thereof, we do not currently believe that future environmental compliance and remediation expenditures will have a material adverse effect on our financial condition or results of operations. For additional information, see "Risk Factors—Risks Relating to Legal and Regulatory Matters—Risks posed by other regulation" in Item 1A of Part I of this report and Note 17—Commitments, Contingencies and Other Items included in Item 8 of Part II of this report.

Seasonality

Overall, our business is not materially impacted by seasonality. Our network-related operating expenses are, however, generally higher in the second and third quarters of the year. From time to time, weather related problems have resulted in increased costs to repair our network and respond to service calls in some of our markets. The amount and timing of these costs are subject to the weather patterns of any given year, but have generally been highest during the third quarter and have been related to damage from severe storms, including hurricanes, tropical storms and tornadoes in our markets along the Atlantic and Gulf of Mexico coastlines.

Employees

At December 31, 2018, we had approximately 45,000 employees, of which approximately 12,000 are members of either the Communications Workers of America ("CWA") or the International Brotherhood of Electrical Workers ("IBEW"). See the discussion of risks relating to our labor relations in "Risk Factors—Risks Affecting Our Business" in Item 1A of Part I of this report and see Note 19—Labor Union Contracts to our consolidated financial statements in Item 8 of Part II of this report for additional information on the timing of certain contract expirations.

Website Access and Important Investor Information

Our website is www.centurylink.com . We routinely post important investor information in the "Investor Relations" section of our website at ir.centurylink.com . The information contained on, or that may be accessed through, our website is not part of this report. You may obtain free electronic copies of our annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and amendments to those reports in the "Investor Relations" section of our website (ir.centurylink.com) under the heading "SEC Filings." These reports are available on our website and on the SEC's website at www.sec.gov . From time to time we also use our website to webcast our earnings calls and certain of our meetings with investors or other members of the investment community.

We have adopted a written code of conduct that serves as the code of ethics applicable to our directors, officers and employees, in accordance with applicable laws and rules promulgated by the SEC and the New York Stock Exchange. In the event that we make any changes (other than by a technical, administrative or non-substantive amendment) to, or provide any waivers from, the provisions of our code of conduct applicable to our directors or executive officers, we intend to disclose these events on our website or in a report on Form 8-K filed with the SEC. The code of conduct, as well as copies of our guidelines on significant governance issues and the charters of our key board committees, are also available in the "Corporate Governance" section of our website at www.centurylink.com/Pages/AboutUs/Governance or in print to any shareholder who requests them by sending a written request to our Corporate Secretary at CenturyLink, Inc., 100 CenturyLink Drive, Monroe, Louisiana, 71203.

In connection with filing this report, our chief executive officer and chief financial officer made the certifications regarding our financial disclosures required under the Sarbanes-Oxley Act of 2002, and its related regulations. In addition, during 2018, our chief executive officer certified to the New York Stock Exchange that he was unaware of any violations by us of the New York Stock Exchange's corporate governance listing standards

We typically disclose material non-public information by disseminating press releases, making public filings with the SEC, or disclosing information during publicly accessible meetings or conference calls. Nonetheless, from time to time we have used, and intend to continue to use, our website and social media accounts to augment our disclosures.

Investors should also be aware that while we do, at various times, answer questions raised by securities analysts, it is against our policy to disclose to them selectively any material non-public information or other confidential information. Accordingly, investors should not assume that we agree with any statement or report issued by an analyst with respect to our past or projected performance. To the extent that reports issued by securities analysts contain any projections, forecasts or opinions, such reports are not our responsibility.

Unless otherwise indicated, information contained in this report and other documents filed by us under the federal securities laws concerning our views and expectations regarding the communications industry are based on estimates made by us using data from industry sources, and on assumptions made by us based on our management's knowledge and experience in the markets in which we operate and the communications industry

generally. You should be aware that we have not independently verified data from industry or other third-party sources and cannot guarantee its accuracy or completeness.

ITEM 1A. RISK FACTORS

The following discussion identifies the most significant risks or uncertainties that could (i) materially and adversely affect our business, financial condition, results of operations, liquidity or prospects or (ii) cause our actual results to differ materially from our anticipated results or other expectations. The following information should be read in conjunction with the other portions of this report, including "Special Note Regarding Forward-Looking Statements", "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 and our consolidated financial statements and related notes in Item 8. Please note that the following discussion is not intended to comprehensively list all risks or uncertainties faced by us. Our operations or actual results could also be similarly impacted by additional risks and uncertainties that are not currently known to us, that we currently deem to be immaterial, that arise in the future or that are not specific to us, such as general economic conditions. In addition, certain of the risks described below apply only to a part or segment of our business.

Risks Affecting Our Business

Our failure to simplify our service support systems could adversely impact our competitive position.

For many of our services, we can effectively compete only if we can quickly and efficiently (i) quote and accept customer orders, (ii) provision and initiate ordered services, (iii) provide customers with adequate means to manage their services and (iv) accurately bill for our services. To attain these goals, we believe we must digitally transform our service support processes to permit greater automation and customer self-service options. This digital transformation is complex and will require a substantial amount of resources, especially in light of the multiplicity of our systems. Development of systems designed to support this transformation will continuously require our personnel and third-party vendors to, among other things, (i) adjust to changes in our offerings and customers' preferences, (ii) simplify our processes, (iii) improve our data management capabilities, (iv) eliminate inconsistencies between our legacy and acquired operations, (v) eliminate older support systems that are costly or obsolete, (vi) develop uniform practices and procedures, and (vii) automate them as much as possible. We cannot assure you that these undertakings will be successful. Our competitive position could be adversely impacted if we fail to continuously develop viable service support systems that are satisfactory to our current and potential customers.

We may not be able to compete successfully against current or future competitors.

Each of our offerings to our business and consumer customers face increasingly intense competition from a wide variety of sources under evolving market conditions. In particular, (i) aggressive competition from a wide range of communications and technology companies has limited the prospects for several of our offerings to business customers, (ii) intense competition from wireless and other communications providers has led to a long-term systemic decline in the number of our wireline voice customers and (iii) strong competition from cable companies has impacted our business and consumer segments. We expect these trends will continue. For more detailed information, see "Business—Competition" in Item 1 of this report.

In addition to competition from a wide range of technology companies and communications providers (including those described above), we are facing increasing competition from several other sources, including satellite companies, cloud companies, broadband providers, software developers, device providers, resellers, sales agents and facilities-based providers using their own networks as well as those leasing parts of our network. Further competition could arise through industry consolidation, technological innovation, or changes in regulation, including changes allowing foreign carriers to more extensively compete in the U.S. market.

Some of our current and potential competitors (i) offer products or services that are substitutes for our traditional wireline voice services, including wireless voice and non-voice communication services, (ii) offer a more comprehensive range of communications products and services, (iii) offer products or services with features that we cannot readily match in some or all of our markets, (iv) install their services more quickly than we do, (v) have greater marketing, engineering, research, development, technical, provisioning, customer relations, financial or other resources, (vi) have larger or more diverse networks with greater transmission capacity, (vii) conduct operations or raise capital at a lower cost than us, (viii) are subject to less regulation, which we believe enables such competitors to operate more flexibly than us with respect to certain offerings, (ix) offer services nationally or internationally to a larger geographic area or larger base of customers, (x) have substantially stronger brand names, which may provide them with greater pricing power than ours, or (xi) have larger operations than ours, which may enable them to compete more successfully in recruiting top talent, entering into operational or strategic partnerships or acquiring companies. Consequently, these competitors may be better equipped to provide more attractive offerings, to charge lower prices for their products and services, to develop and expand their communications and network infrastructure more quickly, to adapt more swiftly to changes in technologies or customer requirements, to devote greater resources to the marketing and sale of their products and services, to provide more comprehensive customer service, to provide greater resources to research and development initiatives and to take advantage of business or other opportunities more readily.

Competition could adversely impact us in several ways, including (i) the loss of customers, market share or traffic on our networks, (ii) our need to expend substantial time or money on new capital improvement projects, (iii) our need to lower prices or increase marketing expenses to remain competitive and (iv) our inability to diversify by successfully offering new products or services.

We are continually taking steps to respond to these competitive pressures, but these efforts may not be successful. Our operating results and financial condition would be adversely affected if these initiatives are unsuccessful or insufficient.

Rapid technological changes could significantly impact our competitive and financial position.

The communications industry has been and continues to be impacted by significant technological changes, which in general are enabling a broader array of companies to compete with us. Many of these technological changes are (i) enabling customers to reduce or bypass use of our networks, (ii) displacing or reducing demand for our services, or (iii) enabling the development of competitive products or services. For years, the development of wireless and Internet-based voice and non-voice communications technologies and social media platforms have reduced demand for our traditional voice services, and these trends continue. More recently, continuous improvements in wireless data technologies have enabled wireless carriers to offer competing products, and we expect this trend to continue as technological advances enable these carriers to carry greater amounts of data faster and with less latency. Technological advancements have also permitted cable companies and other of our competitors to deliver generally faster average broadband transmission speeds than ours. Developments in software have permitted new competitors to offer affordable products that historically required more expensive hardware investment. Rapid changes in technology have also placed competitive pressures on our video, cloud and hosting businesses, and enabled new competitors to enter our markets. To enhance the competitiveness of certain of our services, we will likely be required to spend additional capital to install more fiber optic cable or to augment the capabilities of our copper-based services.

We may not be able to accurately predict or respond to changes in technology or industry standards, or to the introduction of newly-offered services. Any of these developments could make some or all of our offerings less desirable or even obsolete, which would place downward pressure on our market share and revenue. These developments could also require us to (i) expend capital or other resources in excess of currently contemplated levels, (ii) forego the development or provision of products or services that others can provide more efficiently, or (iii) make other changes to our operating plans, corporate strategies or capital allocation plans, any of which could be contrary to the expectations of our security holders or could adversely impact our business operating results.

Even if we succeed in adapting to changes in technology or industry standards by developing new products or services, there is no assurance that the new products or services would have a positive impact on our profit margins or financial performance.

In addition to introducing new technologies and offerings, we may need, from time to time, to phase out outdated and unprofitable technologies and services. If we are unable to do so on a cost-effective basis, we could experience reduced profits. Similarly, if new market entrants are not burdened by an installed base of outdated equipment or obsolete technology, they may have a competitive advantage over us.

For additional information on the risks of increased expenditures, see "Risk Factors—Risks Affecting our Liquidity and Capital Resources—Our business requires us to incur substantial capital and operating expenses, which reduces our available free cash flow."

Several of our services continue to experience declining revenue, and our efforts to offset these declines may not be successful.

Primarily as a result of the competitive and technological changes discussed above, we have experienced a prolonged systemic decline in our local voice, long-distance voice, network access and private line revenue. Consequently, we have experienced declining consolidated revenue (excluding acquisitions) for a prolonged period. More recently, we have experienced declines in revenue derived from the sale of certain of our business products and services.

We have taken a variety of steps to counter these declines in revenue, including an increased focus on selling services in greater demand. However, for the reasons described elsewhere in this report, most of our more recent product and service offerings generate lower profit margins than our traditional services, and some can be expected to experience slowing or no growth in the future. Moreover, we cannot assure you that the revenue generated from our newer offerings will offset revenue losses associated with our traditional services.

Our failure to meet the evolving needs of our customers could adversely impact our competitive position.

In order to compete effectively and respond to changing market conditions, we must continuously offer products and services on terms and conditions that allow us to retain and attract customers and to meet their evolving needs. To do so, we must continuously (i) invest in our network (ii) develop, test and introduce new products and services and (iii) rationalize and simplify our offerings by eliminating older or overlapping products or services. Our ability to maintain attractive products and services and to successfully introduce new product or service offerings on a timely and cost-effective basis could be constrained by a range of factors, including network limitations, support system limitations, limited capital, an inability to attract key personnel with the necessary skills, intellectual property constraints, inadequate digitization or automation, testing delays, technological limits or an inability to act as quickly or efficiently as other competitors. In addition, new product or service offerings may not be widely accepted by our customers. Our business could be materially adversely affected if we are unable to maintain competitive products and services and to timely and successfully develop and introduce new products or services.

Our failure to strengthen our relations with our customers could harm our competitive position.

In mid-2017, our Board of Directors formed a special committee of independent directors to review our policies, procedures and practices relating to consumer sales, service and billing following a former employee's allegations of sales-related misconduct. In late 2017, the special committee concluded its review and issued its key findings. Among other things, the committee found that (i) our investment in consumer sales monitoring was insufficient, (ii) our ordering and billing software contributed to customer confusion and (iii) systems and human errors contributed to inaccurate billing. As noted in our disclosures above, we are committed to simplifying and improving our customers' experience and believe that we have already made significant progress in addressing these concerns. Nonetheless, given the complexity of the undertaking, we cannot assure you that we will timely and effectively implement changes that will adequately address all of our service support issues to the full satisfaction of our customers.

We could experience difficulties in consolidating, integrating, updating and simplifying our technical infrastructure.

Our ability to consolidate, integrate, update and simplify our systems and information technology infrastructure in response to our growth and changing business needs is important to our ability to develop and maintain attractive product and service offerings and to interface effectively with our customers. As discussed further under "Business—Network" in Item 1 of this report, we are currently undertaking several complex, costly and time-consuming projects to simplify, consolidate and modernize our network, which combines our legacy network and the networks of companies we have acquired in the past. Delays in the completion of these projects has hampered our progress, and any additional delays may lead to increased project costs or operational inefficiencies. In addition, there may be issues related to our expanded or updated infrastructure that are not identified by our testing processes, and which may only become evident after we have started to fully utilize the redesigned systems. Our failure to modernize, consolidate and upgrade our technology infrastructure could have adverse consequences, including the delayed implementation of new service offerings, decreased competitiveness of existing service offerings, network instabilities, increased operating or acquisition integration costs, service or billing interruptions or delays, service offering inconsistencies, customer dissatisfaction, and the diversion of development resources. Any or all of the foregoing developments could have a negative impact on our business, results of operations, financial condition and cash flows.

We may not be able to successfully adjust to changes in our industry, our markets and our product mix.

Ongoing changes in the communications industry have fundamentally changed consumers' communications expectations and requirements. In response to these changes, we have substantially altered our product and service offerings through acquisitions and internal product development. Many of these changes have placed a higher premium on sales, marketing and product development functions, and necessitated ongoing changes in our processes and operating protocols, as well as periodic reorganizations of our sales and leadership teams. In addition, we now offer a much more complex range of products and services, operate larger and more complex networks and serve a much larger and more diverse set of global customers. Consequently, we now face greater challenges in effectively managing and administering our operations and allocating capital and other resources to our various offerings. For all these reasons, we cannot assure you that our efforts to adjust to these changes will be timely or successful.

Our revenue and cash flows from operating activities may not be adequate to fund all of our cash requirements.

As noted in greater detail elsewhere herein, our business is capital intensive, including our need to continually invest to update, consolidate and improve our network, our product offerings and our customer support systems. We expect our business to continue to be capital intensive for the foreseeable future. We will also continue to need substantial amounts of cash to meet our fixed commitments and other business objectives, including without limitation funding our operating costs, debt repayments, periodic pension contributions and other benefits payments. We further expect to continue to require significant cash to fund our quarterly dividend payments, subject to the discretionary right of our Board of Directors to change or terminate our current dividend practices at any time. We rely upon our consolidated revenue and cash flows from operating activities to fund our cash needs.

As noted in the risk factor disclosures appearing above and below, changes in competition, technology, regulation and demand for our traditional wireline services continue to place downward pressure on our consolidated revenue and cash flows from operating activities. Over the next several years, we expect that our future cash flows from operating activities will remain under pressure due to the factors discussed herein.

For these reasons, we cannot assure you that our future cash flows from operating activities will be sufficient to fund all of our cash requirements in the manner currently contemplated. Our inability to fund certain of these payments could have an adverse impact on our business, operations, competitive position, prospects or on the value of our securities.

We could be harmed by security breaches or other significant disruptions or failures of networks, information technology infrastructure or related systems owned or operated by us.

We are materially reliant upon our networks, information technology infrastructure and related technology systems (including our billing and provisioning systems) to provide products and services to our customers and to manage our operations and affairs. We face the risk, as does any company, of a security breach or significant disruption of our information technology infrastructure and related systems. As a communications company that transmits large amounts of information over communications networks, we face an added risk that a security breach or other significant disruption of our network, infrastructure or systems, or those that we operate or maintain for certain of our business customers, could lead to material interruptions or curtailments of service. Moreover, in connection with processing and storing sensitive and confidential customer data, we face a heightened risk that a security breach or disruption could result in unauthorized access to our customers' proprietary information.

We strive to maintain the security and integrity of information and systems under our control, and maintain contingency plans in the event of security breaches or other system disruptions. Nonetheless, we cannot assure you that our security efforts and measures will prevent unauthorized access to our systems, loss or destruction of data (including confidential customer information), account takeovers, unavailability of service, computer viruses, malware, ransomware, distributed denial-of-service attacks, or other forms of cyber-attacks or similar events. These threats may derive from human error, hardware or software vulnerabilities, aging equipment or accidental technological failure. These threats may also stem from fraud, malice or sabotage on the part of employees, third parties or foreign nations, including attempts by outside parties to fraudulently induce our employees or customers to disclose or grant access to our data or our customers' data, potentially including information subject to stringent domestic and foreign data protection laws governing personally identifiable information, protected health information or other similar types of sensitive data. These threats may also arise from failure or breaches of systems owned, operated or controlled by other unaffiliated operators to the extent we rely on such other systems to deliver services to our customers. Each of these risks could further intensify to the extent we maintain information in digital form stored on servers connected to the Internet.

Similar to other large telecommunications companies, we are a constant target of cyber-attacks of varying degrees. Although some of these attacks have resulted in security breaches, thus far none of these breaches have resulted in a material adverse effect on our operating results or financial condition. You should be aware, however, that the risk of breaches is likely to increase due to several factors, including the increasing sophistication of cyber-attacks, our greater use of open and software-defined networks, our increased operation of offshore systems and our increased profile due to the growth of our organization and our customer base. You should be further aware that defenses against cyber-attacks currently available to U.S. companies are unlikely to prevent intrusions by a highly-determined, highly-sophisticated hacker. Consequently, you should assume that we will be unable to implement security barriers or other preventative measures that repel all future cyber-attacks. Any such future security breaches or disruptions could materially adversely affect our business, results of operations or financial condition, especially in light of the growing frequency, scope and well-documented sophistication of cyber-attacks and intrusions.

Although we maintain insurance coverage that may, subject to policy terms and conditions (including self-insured deductibles, coverage restrictions and monetary coverage caps), cover certain aspects of our cyber risks, such insurance coverage may be unavailable or insufficient to cover our losses.

Additional risks to our network, infrastructure and related systems include, among others:

- capacity or system configuration limitations, including those resulting from changes in our customer's usage patterns, the introduction of new technologies or products, or incompatibilities between our newer and older systems;
- · theft or failure of our equipment;
- software or hardware obsolescence, defects or malfunctions;
- · power losses or power surges;
- physical damage, whether caused by fire, flood, adverse weather conditions, terrorism, sabotage, vandalism or otherwise;

- deficiencies in our processes or controls:
- our inability to hire and retain personnel with the requisite skills to adequately maintain or improve our systems;
- · programming, processing and other human error; and
- · inadequate building maintenance by third-party landlords or other service failures of our third-party vendors.

Due to these factors, from time to time in the ordinary course of our business we experience disruptions in our service. We could experience more significant disruptions in the future, especially if network traffic continues to increase and we continue to assume greater responsibility for managing our customers' critical systems and networks.

Disruptions, security breaches and other significant failures of the above-described networks and systems could:

- disrupt the proper functioning of these networks and systems, which could in turn disrupt (i) our operational, billing or other administrative functions or (ii) the operations of certain of our customers who rely upon us to provide services critical to their operations;
- result in the unauthorized access to, and destruction, loss, theft, misappropriation or release of proprietary, confidential, sensitive, classified or otherwise valuable information of ours, our customers or our customers' end users, including trade secrets, which others could use for competitive, disruptive, destructive or otherwise harmful purposes and outcomes;
- require us to notify customers, regulatory agencies or the public of data breaches;
- require us to provide credits for future service under certain service level commitments we have provided contractually to our customers or to offer expensive incentives to retain customers;
- subject us to claims for damages, fines, penalties, termination or other remedies under our customer contracts or service standards set by regulators, which in certain cases could exceed our insurance coverage;
- result in a loss of business, damage our reputation among our customers and the public generally, subject us to additional regulatory scrutiny or expose us to prolonged litigation; or
- require significant management attention or financial resources to remedy the resulting damages or to change our systems, including expenses to repair systems, add new personnel or develop additional protective systems.

Any or all of the foregoing developments could have a negative impact on our business, results of operations, financial condition and cash flows.

Negative publicity may adversely impact us.

Our ability to attract and retain customers depends in part upon external perceptions of our products, services, management integrity and financial performance. Customer complaints, governmental investigations, outages, or other service failures of networks operated by us could cause substantial adverse publicity affecting us. Similar events impacting other operators could indirectly harm us by causing substantial adverse publicity affecting our industry in general. In either case, press coverage, social media messaging or other public statements that insinuate improper actions by us or other operators, regardless of their factual accuracy or truthfulness, may result in negative publicity, litigation, governmental investigations or additional regulations. Addressing negative publicity and any resulting litigation or investigations may distract management, increase costs and divert resources. Negative publicity may have an adverse impact on our reputation and the morale of our employees. We could suffer similar adverse effects if shareholders, financial analysts or other financial professionals issue public statements that cast us or our industry in a negative light. Any of these developments could adversely affect our business, results of operations, financial condition, cash flows, prospects and the value of our securities.

Market prices for many of our services have decreased in the past, and any similar price decreases in the future will adversely affect our revenue and margins.

Over the past several years, a range of competitive and technological factors, including robust network construction and intense competition, have lowered market prices for many of our products and services. If these market conditions persist, we may need to continue to reduce prices to retain customers and revenue. If future price reductions are necessary, our operating results will suffer unless we are able to offset these reductions by reducing our operating expenses or increasing our sales volumes.

Our future growth potential will depend in part on the continued development and expansion of the Internet.

Our future growth potential will depend in part upon the continued development and expansion of the Internet as a communication medium and marketplace for the distribution of data, video and other products by businesses, consumers, and governments. The use of the Internet for these purposes may not grow and expand at the rate anticipated by us or others, or may be restricted by factors outside of our control, including (i) actions by other carriers or governmental authorities that restrict us from delivering traffic over other parties' networks, (ii) changes in regulations, (iii) technological stagnation or (iv) increased concerns regarding cyber threats or (v) changes in consumers' preferences or data usage.

Our failure to hire and retain qualified personnel could harm our business.

Our future success depends on our ability to identify, hire, train and retain executives, managers and employees with technological, engineering, software, product development, operational, provisioning, marketing, sales, customer service, administrative, managerial and other key skills. There is a shortage of qualified personnel in several of these fields, particularly in certain growth markets, such as the areas adjoining our Denver and Seattle offices. We compete with several other companies for this limited pool of potential employees. As our industry increasingly becomes more competitive, it could become especially difficult to attract and retain top personnel with skills in high demand. Our workforce reduction and integration initiatives over the past couple of years have further increased the challenges of attracting and retaining talented individuals. In addition, subject to limited exceptions, none of our executives or domestic employees have long-term employment agreements. For all these reasons, there is no assurance that our efforts to recruit and retain qualified personnel will be successful.

Increases in broadband usage may cause network capacity limitations, resulting in service disruptions, reduced capacity or slower transmission speeds for our customers.

Video streaming services, gaming and peer-to-peer file sharing applications use significantly more bandwidth than other Internet activity such as web browsing and email. As use of these services continues to grow, our broadband customers will likely use much more bandwidth than in the past. If this occurs, we could be required to make significant capital expenditures to increase network capacity in order to avoid service disruptions, service degradation or slower transmission speeds for our customers. Alternatively, we could choose to implement network management practices to reduce the network capacity available to bandwidth-intensive activities during certain times in market areas experiencing congestion, which could negatively affect our ability to retain and attract customers in affected markets. Competitive or regulatory constraints may preclude us from recovering the costs of network investments designed to address these issues, which could adversely impact our operating margins, results of operations, financial condition and cash flows.

We have been accused of infringing the intellectual property rights of others and will likely face similar accusations in the future, which could subject us to costly and time-consuming litigation or require us to seek third-party licenses.

Like other communications companies, we have increasingly in recent years received a number of notices from third parties or have been named in lawsuits filed by third parties claiming we have infringed or are infringing upon their intellectual property rights. We are currently responding to several of these notices and claims and expect this industry-wide trend will continue. Responding to these claims may require us to expend significant time and money defending our use of the applicable technology, and divert management's time and resources away from other business. In certain instances, we may be required to enter into licensing agreements requiring royalty payments. In the case of litigation, we could be required to pay significant monetary damages or cease using the applicable technology. If we are required to take one or more of these actions, our profit margins may decline or our operations could be materially impaired. In addition, in responding to these claims, we may be required to stop selling or redesign one or more of our products or services, which could significantly and adversely affect our business, results of operations, financial condition and cash flows.

Similarly, from time to time, we may need to obtain the right to use certain patents or other intellectual property from third parties to be able to offer new products and services. If we cannot license or otherwise obtain rights to use any required technology from a third party on reasonable terms, our ability to offer new products and services may be prohibited, restricted, made more costly or delayed.

We may not be successful in protecting and enforcing our intellectual property rights.

We rely on various patents, copyrights, trade names, trademarks, service marks, trade secrets and other similar intellectual property rights, as well as confidentiality agreements and procedures, to establish and protect our proprietary rights. The steps we have taken, however, may not prevent unauthorized use or the reverse engineering of our technology. Others may independently develop technologies that are substantially equivalent, superior to, or otherwise competitive to the technologies we employ in our services or that infringe on our intellectual property. We may be unable to prevent competitors from acquiring proprietary rights that are similar to or infringe upon our proprietary rights, or to prevent our current or former employees from using or disclosing to others our proprietary information. Enforcement of our intellectual property rights may depend on initiating legal actions against parties who infringe or misappropriate our proprietary information, but these actions may not be successful, even when our rights have been infringed. If we are unsuccessful in protecting or enforcing our intellectual property rights, our business, competitive position, results of operations and financial condition could be adversely affected.

Our operations, financial performance and liquidity are materially reliant on various third parties.

Reliance on other communications providers. To offer certain services in certain of our markets, we must either purchase services or lease network capacity from, or interconnect our network with the infrastructure of, other communications carriers or cloud companies who typically compete against us in those markets. Our reliance on these supply or interconnection arrangements exposes us to multiple risks. Typically these arrangements limit our control over the quality of our services and expose us to the risk that our ability to market our services could be adversely impacted by changes in the plans or properties of the carriers upon which we are reliant. In addition, we are exposed to the risk that the other carriers may be unwilling or unable to continue or renew these arrangements in the future on terms favorable to us, or at all. This risk is heightened when the other carrier is a competitor who may benefit from terminating the agreement or imposing price increases, or a carrier who suffers financial distress or bankruptcy. If we lose these arrangements and cannot timely replace them, our ability to provide services to our customers and conduct our business could be materially adversely affected. Moreover, many of our arrangements with other carriers are regulated by domestic or foreign agencies, which subject us to the additional risk that changes in regulation could increase our costs or otherwise adversely affect our ability to provide services. Finally, even when another carrier agrees or is obligated to provide services to us to permit us to obtain new customers, it is frequently expensive, difficult and time-consuming to switch the new customers to our network, especially if the other carrier fails to provide timely and efficient cooperation.

Conversely, certain of our operations carry a significant amount of voice or data traffic for other communications providers. Their reliance on our services exposes us to the risk that they may transfer all or a portion of this traffic from our network to existing or newly built networks, owned or leased by them, thereby reducing our revenue. For additional information, see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Business Trends" included in Item 7 of this report.

We also rely on reseller and sales agency arrangements with other communications companies to provide some of the services that we offer to our customers, including video and wireless services. As a reseller or sales agent, we do not control the availability, retail price, design, function, quality, reliability, customer service, marketing or branding of these products and services.

Our operations and financial performance could be adversely affected if our relationships with any of these other communications companies are disrupted or terminated for any other reason, including if such other companies:

- · become bankrupt or experience substantial financial difficulties;
- suffer work stoppages or other labor strife;
- challenge our right to receive payments or services under applicable regulations or the terms of our existing contractual arrangements;
 or
- are otherwise unable or unwilling to make payments or provide services to us.

Reliance on other key suppliers and vendors. We depend on a limited number of suppliers and vendors for equipment and services relating to our network infrastructure, including fiber optic cable, software, optronics, transmission electronics, digital switches and related components. If any of these suppliers experience interruptions or other problems delivering or servicing these network components on a timely basis, our operations could suffer significantly. To the extent that proprietary technology of a supplier is an integral component of our network, we may have limited flexibility to purchase key network components from alternative suppliers and may be adversely affected if third parties assert patent infringement claims against our suppliers or us. We also rely on a limited number of (i) software vendors to support our business management systems, (ii) content suppliers to provide programming to our video operations, and (iii) contractors to assist us in connection with our network construction and maintenance activities. In the event it becomes necessary to seek alternative suppliers and vendors, we may be unable to obtain satisfactory replacement supplies, services, utilities or programming on economically attractive terms, on a timely basis, or at all, which could increase costs or cause disruptions in our services.

Reliance on utility providers and landlords. Our energy costs can fluctuate significantly or increase for a variety of reasons, including changes in legislation and regulation. Several pending proposals designed to reduce greenhouse emissions could substantially increase our energy costs, which we may not be able to pass on to our customers.

We lease many of our office facilities. Although the majority of these leases provide us with the opportunity to renew the lease, many of these renewal options provide that rent for the renewal period will be equal to the fair market rental rate at the time of renewal. Any resulting increases in our rent costs could have a negative impact on our financial results.

Reliance on governmental payments. We receive a material amount of revenue or government subsidies under various government programs, which are further described under the heading "Risk Factors—Risks Relating to Legal and Regulatory Matters." We also provide products or services to various federal, state and local agencies. Our failure to comply with complex governmental regulations and laws applicable to these programs, or the terms of our governmental contracts, could result in us being suspended or disbarred from future governmental programs or contracts for a significant period of time. Moreover, certain governmental agencies frequently reserve the right to terminate their contracts for convenience or if funding is unavailable. If our governmental contracts are terminated for any reason, or if we are suspended or debarred from governmental programs or contracts, our results of operations and financial condition could be materially adversely affected.

Violating our government contracts could have other serious consequences.

We provide services to various governmental agencies with responsibility for national security or law enforcement. These governmental contracts impose significant requirements on us relating to network security, information storage and other matters, and in certain instances impose on us additional heightened responsibilities, including requirements related to the composition of our Board of Directors. While we expect to continue to comply fully with all of our obligations under these contracts, we cannot assure you of this. The consequences of violating these contracts could be severe, potentially including the revocation of our Federal Communications Commission (the "FCC") licenses in the U.S. (in addition to being suspended or debarred from government contracting, as noted above.)

If we fail to extend or renegotiate our collective bargaining agreements with our labor unions as they expire from time to time, or if our unionized employees were to engage in a strike or other work stoppage, our business and operating results could be materially harmed.

As of December 31, 2018, approximately 26% of our employees were members of various bargaining units represented by the Communications Workers of America or the International Brotherhood of Electrical Workers. From time to time, our labor agreements with unions expire. Although we typically are able to negotiate new bargaining agreements, we cannot predict the outcome of our future negotiations of these agreements. We may be unable to reach new agreements, and union employees may engage in strikes, work slowdowns or other labor actions, which could materially disrupt our ability to provide services and result in increased cost to us. In addition, new labor agreements may impose significant new costs on us, which could impair our financial condition or results of operations in the future. To the extent they contain benefit provisions, these agreements may also limit our flexibility to change benefits. In particular, retirement benefits provided under these agreements could cause us to incur costs not faced by many of our competitors, which could ultimately hinder our competitive position.

Portions of our property, plant and equipment are located on property owned by third parties.

We rely on rights-of-way, colocation agreements, franchises and other authorizations granted by governmental bodies, railway companies, utilities, carriers and other third parties to locate our cable, conduit and other network equipment on or under their respective properties. A significant number of these authorizations are scheduled to lapse over the next five to ten years, unless we are able to extend or renew them. Our operations could be adversely affected if any of these authorizations terminate or lapse, or if the landowner requests price increases. Moreover, our ability to expand our network could depend in part on obtaining additional authorizations, the receipt of which is not assured.

Over the past few years, certain utilities, cooperatives and municipalities in certain of the states in which we operate have requested significant rate increases for attaching our plant to their facilities. To the extent that these entities are successful in increasing the amount we pay for these attachments, our future operating costs will increase.

Our subsidiaries currently are, and in the past have been, subject to lawsuits challenging the subsidiaries' use of rights-of-way. Similar suits are possible in the future. Plaintiffs in these suits typically seek to have them certified as class action suits. These suits are typically complex, lengthy and costly to defend, and expose us to each of the other general litigation risks described elsewhere herein.

Our business customers may seek to shift risk to us.

We furnish to and receive from our business customers indemnities relating to damages caused or sustained by us in connection with certain of our operations. Our customers' changing views on risk allocation could cause us to accept greater risk to win new business or could result in us losing business if we are not prepared to take such risks. To the extent that we accept such additional risk, and seek to insure against it, our insurance premiums could rise.

Our international operations expose us to various regulatory, currency, tax, legal and other risks.

Our international operations are subject to U.S. and non-U.S. laws and regulations regarding operations in international jurisdictions in which we provide services. These numerous and sometimes conflicting laws and regulations include anti-corruption laws, anti-competition laws, trade restrictions, tax laws, immigration laws, privacy laws and accounting requirements. Many of these laws are complex and change frequently. Regulations that require the awarding of contracts to local contractors or the employment of local citizens may adversely affect our flexibility or competitiveness in these jurisdictions. Local laws and regulations, and their interpretation and enforcement, differ significantly among those jurisdictions. There is a risk that these laws or regulations may materially restrict our ability to deliver services in various international jurisdictions or could be breached through inadvertence or mistake, fraudulent or negligent behavior of our employees or agents, failure to comply with certain formal documentation or technical requirements, or otherwise. Violations of these laws and regulations could result in fines and penalties, criminal sanctions against us or our personnel, or prohibitions on the conduct of our business or our ability to operate in one or more countries, any of which could have a material adverse effect on our business, reputation, results of operations, financial condition or prospects.

Many non-U.S. laws and regulations relating to communications services are more restrictive than U.S. laws and regulations, particularly those relating to privacy rights and data retention. For example, all 28 current member states of the European Union have adopted new European data protection laws that we believe could impact our operations in Europe and could potentially expose us to an increased risk of litigation or significant regulatory fines. Moreover, national regulatory frameworks that are consistent with the policies and requirements of the World Trade Organization have only recently been, or are still being, enacted in many countries. Accordingly, many countries are still in the early stages of providing for and adapting to a liberalized telecommunications market. As a result, in these markets we may encounter more protracted and difficult procedures to obtain licenses necessary to provide the full set of products and services we seek to offer.

In addition to these international regulatory risks, some of the other risks inherent in conducting business internationally include:

- tax, licensing, political or other business restrictions or requirements, which may render it more difficult to obtain licenses or interconnection agreements on acceptable terms, if at all;
- uncertainty concerning import and export restrictions, including the risk of fines or penalties assessed for violations;
- · longer payment cycles and problems collecting accounts receivable;
- U.S. and non-U.S. regulation of overseas operations, including regulation under the U.S. Foreign Corrupt Practices Act (the "FCPA") and other applicable anti-corruption laws, including the U.K. Bribery Act of 2010 and the Brazilian Anti-corruption Law, (collectively with the FCPA, the "Anti-Corruption Laws");

- economic, social and political instability, with the attendant risks of terrorism, kidnapping, extortion, civic unrest and potential seizure or nationalization of assets;
- currency and exchange controls, repatriation restrictions and fluctuations in currency exchange rates;
- · challenges in securing and maintaining the necessary physical and telecommunications infrastructure;
- the inability in certain jurisdictions to enforce contract rights either due to underdeveloped legal systems or government actions that result in a deprivation of contract rights;
- increased risk of cyber-attacks or similar events to our network as we expand our network or interconnect our network with other networks internationally;
- the inability in certain jurisdictions to adequately protect intellectual property rights;
- laws, policies or practices that restrict with whom we can contract or otherwise limit the scope of operations that can legally or practicably be conducted within any particular country;
- potential submission of disputes to the jurisdiction of a non-U.S. court or arbitration panel;
- · reliance on third parties, including those with which we have limited experience;
- limitations in the availability, amount or terms of insurance coverage;
- the imposition of unanticipated or increased taxes, increased communications or privacy regulations or other forms of public or governmental regulation that increase our operating expenses; and
- · challenges in staffing and managing overseas operations.

Changes in multilateral conventions, treaties, tariffs or other arrangements between or among sovereign nations could impact us. Specifically, the British government is currently negotiating the terms of the United Kingdom's exit from the European Union ("Brexit"). Brexit could potentially impact our supply chains, logistics, and

human resources. Additionally, Brexit and other changes in multilateral arrangements may more broadly adversely affect our operations and financial results.

Many of these risks are beyond our control, and we cannot predict the nature or the likelihood of the occurrence or corresponding effect of any such events, each of which could have an adverse effect on our financial condition and results of operations.

Certain of our international operations are conducted in countries or regions experiencing corruption or instability, which subjects us to heightened legal and economic risks.

We do business and may in the future do additional business in certain countries or regions in which corruption is a serious problem. Moreover, in order to effectively compete in certain non-U.S. jurisdictions, it is frequently necessary or required to establish joint ventures, strategic alliances or marketing arrangements with local operators, partners or agents. In certain instances, these local operators, partners or agents may have interests that are not always aligned with ours. Reliance on local operators, partners or agents could expose us to the risk of being unable to control the scope or quality of our overseas services or products, or being held liable under any Anti-Corruption Laws for actions taken by our strategic or local partners or agents. Any determination that we have violated any Anti-Corruption Laws could have a material adverse effect on our business, results of operations, reputation or prospects.

We conduct significant operations in regions that have historically experienced high levels of political, economic and social instability, including the Latin American region. Various events in recent years have placed pressures on the stability of the currencies of several Latin American countries in which we operate, including Argentina, Brazil and Colombia. Pressures or volatility in local or regional currencies may adversely affect our customers in this region, which could diminish their ability or willingness to order products or services from us. Several Latin American countries have historically experienced high rates of inflation. Governmental actions taken to curb inflation, coupled with speculation about possible future actions, have in the past contributed to periodic economic uncertainty in many Latin American countries. Similar actions in the future, together with abrupt shifts in governmental administrations, could impede our ability to develop or implement effective business plans in the region. In addition, if high rates of inflation persist, we may not be able to adjust the price of our services sufficiently to offset our higher costs. A high inflation environment would also have negative effects on the level of economic activity and employment and adversely affect our business.

We are exposed to currency exchange rate risks and currency transfer restrictions and our results may suffer due to currency translations and re-measurements.

Declines in the value of non-U.S. currencies relative to the U.S. dollar could adversely affect us in several respects, including hampering our ability to market our services to customers whose revenue is denominated in depreciated currencies. In addition, where we issue invoices for our services in currencies other than U.S. dollars, our results of operations may suffer due to currency translations if such currencies depreciate relative to the U.S. dollar and we cannot or do not elect to enter into currency hedging arrangements regarding those payment obligations. Similarly, the strengthening of the U.S. dollar and exchange control regulations could negatively impact the ability of overseas customers to pay for our services in U.S. dollars.

Certain Latin American economies have experienced shortages in non-U.S. currency reserves and have adopted restrictions on the use of certain mechanisms to expatriate local earnings and convert local currencies into U.S. dollars. Any of these shortages or restrictions may limit or impede our ability to transfer or to convert those currencies into U.S. dollars and to expatriate those funds.

We expect rising costs and other industry changes will continue to adversely impact our video business.

The costs of purchasing video programming have risen significantly in recent years and continue to rise. Moreover, an increasing number of consumers are receiving access to video content through video streaming or other services pursuant to new technologies for a nominal or no fee, which will likely continue to reduce demand for our more traditional video products. Various new technologies have increased the number of entertainment choices available to consumers and intensified the challenges posed by audience fragmentation. All of these changes have reduced demand for our video products and services, and we expect this trend will continue.

We may not be able in the future to acquire new businesses on attractive terms.

Historically, much of our growth has been attributable to acquisitions. Our future ability to grow through additional acquisitions could be limited by several factors, including our leverage, debt covenants and inability to identify attractively-priced target companies. Moreover, we generally must devote significant management attention and resources to evaluate acquisition opportunities, which could preclude us from evaluating acquisition opportunities during periods when management is committed to other opportunities, tasks or activities. Accordingly, we cannot assure you that we will be able to attain future growth through acquisitions. See "Risks Relating to Our November 2017 Combination with Level 3" for a discussion of certain specific risks raised by our combination with Level 3 on November 1, 2017 and see the next risk factor immediately below for a discussion of certain general risks raised by acquisitions.

Any additional future acquisitions or strategic investments by us would subject us to additional business, operating and financial risks, the impact of which cannot presently be evaluated, and could adversely impact our capital structure or financial position.

In an effort to implement our business strategies, we may from time to time in the future pursue other acquisition or expansion opportunities, including strategic investments. These transactions could involve acquisitions of entire businesses or investments in start-up or established companies, and could take several forms, including mergers, joint ventures, investments in new lines of business, or the purchase of equity interests or assets. These types of transactions may present significant risks and uncertainties, including the difficulty of identifying appropriate companies to acquire or invest in on acceptable terms, distraction of management from current operations, insufficient revenue acquired to offset liabilities assumed, unexpected expenses, inadequate return of capital, regulatory or compliance issues, potential infringements, potential violations of covenants in our debt instruments and other unidentified issues not discovered in due diligence. To the extent we acquire part or all of a business that is financially unstable or is otherwise subject to a high level of risk, we may be affected by currently unascertainable risks of that business. Accordingly, there is no current basis to evaluate the possible merits or risks of the particular business or assets that we may acquire. Moreover, we cannot guarantee that any such transaction will ultimately result in the realization of the benefits of the transaction originally anticipated by us or that any such transaction will not have a material adverse impact on our financial condition or results of operations. In particular, we can provide no assurances that we will be able to successfully integrate the technology systems, billing systems, accounting processes, sales force, cost structure, product development and service delivery processes, standards, controls, policies, strategies and culture of the acquired company with ours. In addition, the financing of any future acquisition completed by us could adversely impact our capital structure as any such financing would likely include the issuance of additional securities or the borrowing of additional funds. Except as required by law or applicable securities exchange listing standards, we do not expect to ask our shareholders to vote on any proposed acquisition. Moreover, we generally do not announce our material transactions until we have entered into a preliminary or definitive agreement.

We may not be able to dispose of assets or asset groups on terms that are attractive to us, or at all.

In the past, we have disposed of assets or asset groups for a variety of reasons, and we may consider disposing of other assets or asset groups from time to time in the future. We may not be able to divest any such assets on terms that are attractive to us, or at all. In addition, if we agree to proceed with any such divestitures of assets, we may experience operational difficulties segregating them from our retained assets and operations, which could impact the execution or timing for such dispositions and could result in disruptions to our operations or claims for damages, among other things.

Unfavorable general economic conditions could negatively impact our operating results and financial condition.

Unfavorable general economic conditions, including unstable economic and credit markets, could negatively affect our business. While it is difficult to predict the ultimate impact of these general economic conditions, they could adversely affect demand for some of our products and services and could cause customers to shift to lower priced products and services or to delay or forego purchases of our products and services. These conditions impact, in particular, our ability to sell discretionary products or services to business customers that are under pressure to reduce costs or to governmental customers operating under budgetary constraints. Any one or more of these circumstances could continue to depress our revenue. Also, our customers may encounter financial hardships or may not be able to obtain adequate access to credit, which could negatively impact their ability to make timely payments to us. In addition, as discussed further below, unstable economic and credit markets may preclude us from refinancing maturing debt at terms that are as favorable as those from which we previously benefited, at terms that are acceptable to us, or at all. For these reasons, among others, weak economic conditions could adversely affect our operating results, financial condition, and liquidity.

For additional information about our business and operations, see "Business" in Item 1 of this report.

Risks Relating to Our November 2017 Combination with Level 3

We expect to continue to incur substantial expenses related to the Level 3 combination.

As noted elsewhere herein, we have incurred substantial expenses in connection with completing our business combination with Level 3 on November 1, 2017, and we expect to continue to incur substantial expenses in connection with integrating our business, operations, networks, systems, technologies, policies and procedures with those of Level 3. As a result of these expenses, we have taken, and expect to continue to take, charges against our earnings. Our future charges are expected to continue to be significant, although the timing and magnitude of such charges remains uncertain.

We may be unable to integrate successfully our incumbent business and Level 3's business and realize the anticipated benefits of the combination.

The transaction combined two companies which previously operated as independent public companies. The combined company continues to devote significant management attention and resources to integrating the business practices and operations of CenturyLink and Level 3. Potential difficulties we may encounter in the remainder of the integration process include the following:

- the inability to successfully combine our incumbent business and Level 3's business in a manner that permits us to fully and timely attain the cost savings and operating synergies anticipated to result from the acquisition;
- the additional complexities of combining two companies with different histories, cultures, regulatory restrictions, operating structures and markets;
- the complexities associated with managing the combined businesses out of several different locations and integrating personnel from the two companies, while at the same time attempting to provide consistent, high quality products and services under a unified culture;
- lost sales and customers as a result of certain customers of either of the two companies deciding to terminate or reduce their business with the combined company;
- the failure to retain key employees of either of the two companies;
- unanticipated impediments in integrating departments, systems, technologies, procedures and policies, and in maintaining uniform standards and controls;
- · potential unknown liabilities and unforeseen increased expenses, delays or regulatory conditions associated with the acquisition; and
- performance shortfalls as a result of the diversion of management's attention caused by completing the acquisition and integrating the companies' operations.

For all these reasons, you should be aware that it is possible that the integration process could result in the distraction of management, the disruption of our ongoing business or inconsistencies in our products, services, standards, controls, procedures and policies, any of which could adversely affect the ability of the combined company to maintain relationships with customers, vendors and employees or to achieve the anticipated benefits of the acquisition, or could otherwise adversely affect our business and financial results.

We may be unable to retain key employees.

Our success following the combination will depend in part upon our ability to retain key Level 3 and CenturyLink employees. Key employees may depart because of issues relating to uncertainty, changes in workplace responsibilities or demands, or difficulties associated with integrating companies with different histories and cultures. Accordingly, we cannot assure you that we will be able to retain key employees to the same extent as in the past.

We may be unable to obtain security clearances necessary to perform certain Level 3 government contracts.

Certain Level 3 legal entities and officers have security clearances required for Level 3's performance of customer contracts with various government entities. We continue to discuss with appropriate governmental representatives arrangements that will permit us or our officers to continue to perform these Level 3 contracts. Although we expect these efforts will be successful, we cannot assure you of this.

Risks Relating to Legal and Regulatory Matters

We operate in a highly regulated industry and are therefore exposed to restrictions on our operations and a variety of risks relating to such regulation.

General . Our domestic operations are regulated by the FCC, various state utility commissions and occasionally by local agencies. Our domestic operations are also subject to potential investigation and legal action by the Federal Trade Commission ("FTC") and other federal and state regulatory authorities over issues such as consumer marketing, competitive practices, and privacy protections. Our non-domestic operations are regulated by supranational groups (such as the European Union), national agencies and frequently state, provincial or local bodies.

Generally, we must obtain and maintain operating licenses from these bodies in most territories where we offer regulated services. We cannot assure you that we will be successful in obtaining or retaining all licenses necessary to carry out our business plan. Even if we are, the prescribed service standards and conditions imposed on us under these licenses may increase our costs and limit our operational flexibility. We also operate in some areas of the world without licenses, as permitted through relationships with locally-licensed partners.

We are subject to numerous requirements and interpretations under various international, federal, state and local laws, rules and regulations, which are often quite detailed and occasionally in conflict with each other. The regulation of telecommunications networks and services around the world varies widely. In some countries, the range of services we are legally permitted to provide may be limited or may change. As noted above, in other countries existing telecommunications legislation is in development, is subject to currently ongoing proceedings, is unclear or inconsistent, or is applied in an unequal or unpredictable fashion, often in the absence of adjudicative forums that are adequate to address disputes. Accordingly, we cannot ensure that we will always be considered to be in compliance with all these requirements at any single point in time (as discussed further elsewhere herein). Our inability or failure to comply with the telecommunications and other laws of one or more countries in which we operate could prevent us from commencing or continuing to provide service therein.

The agencies responsible for the enforcement of these laws, rules and regulations may initiate inquiries or actions based on customer complaints or on their own initiative. Even if we are ultimately found to have complied with applicable regulations, such actions or inquiries could create adverse publicity that negatively impacts our business.

Domestic regulation of the telecommunications industry continues to change, and the regulatory environment varies substantially from jurisdiction. A substantial portion of our local voice services revenue remains subject to FCC and state utility commission pricing regulation, which periodically exposes us to pricing or earnings disputes and could expose us to unanticipated price declines. In addition, from time to time carriers or other third parties refuse to pay for certain of our services or challenge our rights to receive certain service payments. Our future revenue, costs, and capital investment could be adversely affected by material changes to or decisions regarding the applicability of government requirements, and we cannot assure you that future regulatory, judicial or legislative activities will not have a material adverse effect on our operations.

Changes in the composition and leadership of the FCC, state commissions and other agencies that regulate our business could have significant impacts on our revenue, expenses, competitive position and prospects. Changes in the composition and leadership of these agencies are often difficult to predict, and make future planning more difficult.

Risks associated with changes in regulation. Changes in regulation can have a material impact on our business, revenue or financial performance. Changes over the past couple of decades in federal regulations have substantially impacted our operations including recent orders or laws overhauling intercarrier compensation, revamping universal service funding and increasing our responsibilities to assist various governmental agencies and safeguard customer data. These changes, coupled with our participation in the new FCC support programs, have significantly impacted various aspects of our operations, financial results and capital expenditures, including the amount of revenue we collect from our wholesale customers and from federal support programs. We expect these impacts will continue in the future. For more information, see "Business—Regulation" in Item 1 of this report, and "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of this report.

Many of the FCC's regulations adopted in recent years remain subject to judicial review and additional rulemakings, thus increasing the difficulty of determining the ultimate impact of these changes on us and our competitors.

Federal and state agencies that dispense universal service or other support program payments can, and from time to time do, reduce the amount of those payments to us and other carriers.

Risks of higher costs . Regulations continue to create significant operating and capital costs for us. Regulatory challenges to our business practices or delays in obtaining certifications and regulatory approvals could cause us to incur substantial legal and administrative expenses, and, if successful, such challenges could adversely affect our operations.

Our business also may be impacted by legislation and regulation imposing new or greater obligations related to regulations or laws related to regulating broadband services, storing records, fighting crime, bolstering homeland security or cyber security, increasing disaster recovery requirements, minimizing environmental impacts, enhancing privacy, restricting data collection, protecting intellectual property rights of third parties, or addressing other issues that impact our business. We expect our compliance costs to increase if future laws or regulations continue to increase our obligations.

Increased risks of fines. We have recently paid certain regulatory fines associated with network or service outages, particularly with respect to outages impacting the availability of emergency - 911 services. Over the past couple of years, we believe that regulators have assessed substantially higher fines than in the past for these types of incidents, and it is possible this trend will continue.

Risks of reduced flexibility. As a diversified full service incumbent local exchange carrier in many of our operating markets, we have traditionally been subject to significant regulation that does not apply to many of our competitors. This regulation in many instances restricts our ability to change rates, to compete and to respond rapidly to changing industry conditions. In particular, cable television companies in recent years have been able to exploit differences in regulatory oversight, which we believe has helped them to develop service offerings competitive with ours. As our business becomes increasingly competitive, regulatory disparities could continue to favor our competitors.

Risks posed by other regulations. All of our operations are also subject to a variety of environmental, safety, health and other governmental regulations. In connection with our current operations, we use, handle and dispose of various hazardous and non-hazardous substances and wastes. In prior decades, certain of our current or former subsidiaries owned or operated, or are alleged to have owned or operated, former manufacturing businesses, for which we have been notified of certain potential environmental liabilities. We monitor our compliance with applicable regulations or commitments governing these current and past activities. Although we believe that we are in compliance with these regulations in all material respects, our use, handling and disposal of environmentally sensitive materials, or the prior operations of our predecessors, could expose us to claims or actions that could potentially have a material adverse effect on our business, financial condition and operating results.

For a discussion of regulatory risks associated with our international operations, see "Risk Factors—Risks Affecting Our Business—Our international operations expose us to various regulatory, currency, tax, legal and other risks."

Our participation in the FCC's Connect America Fund ("CAF") Phase II support program poses certain risks.

Our participation in the FCC's CAF Phase II support program subjects us to certain financial risks. If we fail to attain certain specified infrastructure buildout requirements, the FCC could withhold future CAF support payments until these shortcomings are rectified. In addition, if we are not in compliance with FCC measures by the end of the CAF Phase II program, we would incur substantial penalties. To comply with the FCC's buildout requirements, we believe we will need to dedicate a substantial portion of our future capital expenditure budget to the construction of new infrastructure. The CAF-related expenditures could reduce the amount of funds we are willing or able to allocate to other initiatives or projects. In addition, if the FCC decides not to renew or extend the current CAF Phase II program, or initiate a replacement program, these support payments could end at the termination of the program.

Regulation of the Internet could limit our ability to operate our broadband business profitably and to manage our broadband facilities efficiently.

Since the creation of the Internet, there has been extensive debate about whether and how to regulate Internet service providers. A significant number of congressional leaders, state elected officials and various consumer interest groups have long advocated in favor of extensive regulation. In 2015, the FCC adopted new regulations that regulated broadband services as a public utility under Title II of the Communications Act of 1934. Although the FCC voted to repeal most of those regulations in December, 2017, opponents of the rescission have judicially challenged this action and continue to advocate in favor of re-instituting extensive regulation. Depending on the scope of any such future federal or state regulations, the imposition of heightened regulation of our Internet operations could hamper our ability to operate our data networks efficiently, restrict our ability to implement network management practices necessary to ensure quality service, increase the cost of network extensions and upgrades, and otherwise negatively impact our current operations. As the significance of the Internet expands, state, local or foreign governments may adopt new laws or regulations, or apply existing laws and regulations to the Internet. We cannot predict the outcome of any such changes.

We may be liable for the material that content providers or distributors distribute over our network.

The liability of private network operators for information stored or transmitted on their networks is impacted both by changing technology and evolving legal principles that remain unsettled in many jurisdictions. While we disclaim any liability for third-party content in our service contracts, as a private network provider we could be exposed to legal claims relating to third party content stored or transmitted on our networks. Such claims could involve, among others, allegations of defamation, invasion of privacy, copyright infringement, or aiding and abetting restricted activities such as online gambling or pornography. Although we believe our liability for these types of claims is limited, suits against other carriers have been successful and we cannot assure you that our defenses will prevail. If we decide to implement additional measures to reduce our exposure to these risks, or if we are required to defend ourselves against these kinds of claims, our operations and financial results could be negatively affected.

Our pending legal proceedings could have a material adverse impact on our financial condition and operating results, the trading price of our securities and our ability to access the capital markets.

There are several material proceedings pending against us, as described in Note 17—Commitments, Contingencies and Other Items to our consolidated financial statements included in Item 8 of this report. Results of these legal proceedings cannot be predicted with certainty. Irrespective of its merits, litigation may be both lengthy and disruptive to our operations and could cause significant expenditure and diversion of management attention. We review our litigation accrual liabilities on a quarterly basis, but in accordance with applicable accounting guidelines only establish accrual liabilities when losses are deemed probable and reasonably estimable and only revise previously-established accrual liabilities when warranted by changes in circumstances, in each case based on then-available information. As such, as of any given date we could have exposure to losses under proceedings as to which no liability has been accrued or as to which the accrued liability is inadequate. For each of these reasons, any of the proceedings described in Note 17—Commitments, Contingencies and Other Items, as well as current litigation not described therein or future litigation, could have a material adverse effect on our business, reputation, financial position, operating results, the trading price of our securities and our ability to access the capital markets. We can give you no assurances as to the ultimate impact of these matters on us.

We are subject to franchising requirements that could impede our expansion opportunities or result in potential fines or penalties.

We may be required to obtain from municipal authorities operating franchises to install or expand certain facilities related to our fiber transport operations and certain of our other services. Some of these franchises may require us to pay franchise fees, and may require us to pay fines or penalties if we violate or terminate our related contractual commitments. In some cases, certain franchise requirements could delay us in expanding our operations or increase the costs of providing these services.

We are exposed to risks arising out of recent legislation affecting U.S. public companies.

Changing laws, regulations and standards relating to corporate governance and public disclosure, including the Sarbanes-Oxley Act and the Dodd-Frank Wall Street Reform and Consumer Protection Act, and related regulations implemented thereunder, have increased our legal and financial compliance costs and made some activities more time consuming. Any failure to successfully or timely complete annual assessments of our internal controls required by Section 404 of the Sarbanes-Oxley Act could subject us to sanctions or investigation by regulatory authorities. Any such action could adversely affect our financial results or our reputation with investors, lenders or others.

Changes in any of the above-described laws or regulations may limit our ability to plan, and could subject us to further costs or constraints.

From time to time, the laws or regulations governing us or our customers, or the government's policy of enforcing those laws or regulations, have changed frequently and materially. The variability of these laws could hamper the ability of us and our customers to plan for the future or establish long-term strategies. Moreover, future changes in these laws or regulations could further increase our operating or compliance costs, or further restrict our operational flexibility, any of which could have a material adverse effect on our results of operations, competitive position, financial condition or prospects.

For a more thorough discussion of the regulatory issues that may affect our business, see "Business—Regulation" in Item 1 of this report.

Risks Affecting Our Liquidity and Capital Resources

Our high debt levels expose us to a broad range of risks.

We continue to carry significant debt. As of December 31, 2018, the aggregate principal amount of our consolidated long-term debt was \$35.6 billion, excluding unamortized discounts, net, unamortized debt issuance costs and capital lease and other obligations. As of such date, \$4.3 billion aggregate principal amount of this long-term debt was scheduled to mature prior to December 31, 2021. In addition, our term loans require mandatory principal payments of approximately \$164 million per year until maturity. While we currently believe we will have the financial resources to meet or refinance our obligations when they come due, we cannot fully anticipate our future performance or financial condition, the future condition of the credit markets or the economy generally.

Our significant levels of debt can adversely affect us in several other respects, including:

- limiting our ability to obtain additional financing for working capital, capital expenditures, acquisitions, refinancings or other general corporate purposes, particularly if, as discussed further in the risk factor disclosure below, (i) the ratings assigned to our debt securities by nationally recognized credit rating organizations are revised downward or (ii) we seek capital during periods of turbulent or unsettled market conditions;
- requiring us to dedicate a substantial portion of our cash flow from operations to the payment of interest and principal on our debt, thereby reducing the funds available to us for other purposes, including acquisitions, capital expenditures, strategic initiatives, dividends, stock repurchases, marketing and other potential growth initiatives;
- hindering our ability to capitalize on business opportunities and to plan for or react to changing market, industry, competitive or economic conditions;

- increasing our future borrowing costs;
- limiting or precluding us from entering into commercial, hedging or other financial arrangements with vendors, customers or other business partners;
- making us more vulnerable to economic or industry downturns, including interest rate increases;
- placing us at a competitive disadvantage compared to less leveraged competitors;
- increasing the risk that we will need to sell securities or assets, possibly on unfavorable terms, or take other unfavorable actions to meet payment obligations; or
- increasing the risk that we may not meet the financial covenants contained in our debt agreements or timely make all required debt payments, either of which could result in the acceleration of some or all of our outstanding indebtedness.

The effects of each of these factors could be intensified if we increase our borrowings.

A substantial portion of our indebtedness bears interest at variable rates. If market interest rates increase, our variable-rate debt will have higher debt service requirements, which could adversely impact our cash flows and financial condition.

Any failure to make required debt payments could, among other things, adversely affect our ability to conduct operations or raise capital.

Subject to certain limitations, our current debt agreements and the debt agreements of our subsidiaries allow us to incur additional debt, which could exacerbate the other risks described in this report.

Subject to certain limitations and restrictions, the current terms of our debt instruments and the debt instruments of our subsidiaries permit us or them to incur additional indebtedness. Incremental borrowings that impose additional financial risks could exacerbate the other risks described in this report.

We expect to periodically require financing, and we cannot assure you that we will be able to obtain such financing on terms that are acceptable to us, or at all.

We have a significant amount of indebtedness that we intend to refinance over the next several years, principally through the issuance of debt by CenturyLink, Inc., Qwest Corporation or Level 3 Financing, Inc. We may also need to obtain additional financing under a variety of other circumstances, including if:

- we engage in additional acquisitions or undertake substantial capital projects or other initiatives that increase our cash requirements;
- we are required to make pension or other benefits payments earlier or in greater amounts than currently anticipated;
- we become subject to significant judgments or settlements, including in connection with one or more of the matters discussed elsewhere herein; or
- we otherwise require additional cash to fund our cash requirements described elsewhere herein.

Our ability to arrange additional financing will depend on, among other factors, our financial position, performance, and credit ratings, as well as prevailing market conditions and other factors beyond our control. Global financial markets continue to be unpredictable and volatile. Prevailing market conditions could be adversely affected by (i) general market conditions, such as disruptions in domestic or overseas sovereign or corporate debt markets, geo-political instabilities, contractions or limited growth in the economy or other similar adverse economic developments in the U.S. or abroad and (ii) specific conditions in the communications industry. Volatility in the global markets could limit our access to the credit markets, leading to higher borrowing costs or, in some cases, the inability to obtain financing on terms that are as favorable as those from which we previously benefited, on terms that are acceptable to us, or at all.

In addition, our ability to borrow funds in the future will depend in part on the satisfaction of the covenants in our credit facilities and other debt instruments, which are discussed further below.

Our access to funds under our revolving credit facility is further dependent upon the ability of the facility's lenders to meet their funding commitments. If one or more of the lenders fails to fund, the remaining lenders will not be legally obligated to rectify the funding shortfall.

For all the reasons mentioned above, we can give no assurance that additional financing for any of these purposes will be available on terms that are acceptable to us, or at all.

If we are unable to make required debt payments or refinance our debt, we would likely have to consider other options, such as selling assets, issuing additional securities, reducing or terminating our dividend payments, cutting or delaying costs or otherwise reducing our cash requirements, or negotiating with our lenders to restructure our applicable debt. Our current and future debt instruments may restrict, or market or business conditions may limit, our ability to do some of these things on favorable terms, or at all. For these and other reasons, we cannot assure you that we could implement these steps in a sufficient or timely manner, or at all. Moreover, any steps taken to strengthen our liquidity, such as cutting costs, could adversely impact our business or operations.

We have a complex debt structure, and our various debt agreements include restrictions and covenants that could (i) limit our ability to conduct operations or borrow additional funds, (ii) restrict our ability to engage in inter-company transactions and (iii) lead to the acceleration of our repayment obligations in certain instances.

CenturyLink, Inc. and various of its subsidiaries (including Embarq Corporation, Qwest Corporation, Qwest Capital Funding, Inc., Level 3 Financing, Inc. and Level 3 Parent, LLC) have borrowed substantial amounts of money from financial institutions or investors. Under the associated debt and financing arrangements, CenturyLink, Inc. and these subsidiaries are subject to various covenants and restrictions, the most restrictive of which pertain to the debt of CenturyLink, Inc. and the Level 3 entities.

CenturyLink, Inc.'s revolving and term loan debt arrangements contain several significant limitations restricting CenturyLink, Inc.'s ability to, among other things:

- · borrow additional money or issue guarantees;
- pay dividends or other distributions to shareholders;
- make loans, advances or other investments;
- · create liens on assets;
- · sell assets;
- enter into sale-leaseback transactions;
- · enter into transactions with affiliates; and
- · engage in mergers or consolidations.

These above-listed restrictive covenants could materially adversely affect our ability to operate or expand our business, to pursue strategic transactions, or to otherwise pursue our plans and strategies.

The debt and financing arrangements of Level 3 Parent, LLC and its subsidiary Level 3 Financing, Inc. contain substantially similar limitations that restrict their operations on a standalone basis as a separate restricted group. Consequently, certain of these covenants may significantly restrict our ability to receive cash from Level 3, to distribute cash from Level 3 to other of our affiliated entities, or to enter into other transactions among our wholly-owned entities.

CenturyLink, Inc.'s above-referenced debt arrangements also contain financial covenants that require us to maintain certain financial ratios, and the term loan debt of Qwest Corporation includes a similar financial covenant. The ability of CenturyLink, Inc. and Qwest Corporation to comply with these provisions may be affected by events beyond their control.

Increasingly in recent years, certain debt investors have sought to financially benefit themselves by identifying and seeking to enforce defaults under borrowers' debt agreements. This development could increase the risk of claims made under our debt agreements.

The failure of CenturyLink, Inc. or any of its subsidiaries to comply with the above-described restrictive or financial covenants could result in an event of default, which, if not cured or waived, could accelerate our debt repayment obligations. Certain of our debt instruments have cross-default or cross-acceleration provisions. When present, these provisions could have a wider impact on liquidity than might otherwise arise from a default or acceleration of a single debt instrument. As noted elsewhere herein, we cannot assure you that we could adequately address any such defaults, cross-defaults or acceleration of our debt payment obligations in a sufficient or timely manner, or at all. For additional information, see "Risks Affecting Our Liquidity and Capital Resources—We expect to periodically require financing, and we cannot assure you that we will be able to obtain such financing on terms that are acceptable to us, or at all" and Note 6—Long-Term Debt and Credit Facilities.

Any downgrade in the credit ratings of us or our affiliates could limit our ability to obtain future financing, increase our borrowing costs and adversely affect the market price of our existing debt securities or otherwise impair our business, financial condition and results of operations.

Nationally recognized credit rating organizations have issued credit ratings relating to CenturyLink, Inc.'s long-term debt and the long-term debt of several of its subsidiaries. Many of these ratings are below "investment grade", which results in higher borrowing costs than "investment grade" debt as well as reduced marketability of our debt securities. There can be no assurance that any rating assigned to any of these debt securities will remain in effect for any given period of time or that any such ratings will not be lowered, suspended or withdrawn entirely by a rating agency if, in that rating agency's judgment, circumstances so warrant.

A downgrade of any of these credit ratings could:

- adversely affect the market price of some or all of our outstanding debt or equity securities;
- limit our access to the capital markets or otherwise adversely affect the availability of other new financing on favorable terms, if at all;
- trigger the application of restrictive covenants or adverse conditions in our current or future debt agreements;
- increase our cost of borrowing; and
- · impair our business, financial condition and results of operations.

For more information on the credit ratings of our secured and unsecured debt, see "Management's Discussion and Analysis of Financial Condition and Results of Operations - Liquidity and Capital Resources - Debt and Other Financing Arrangements" in Item 7 of this report.

Under our debt agreements, a change of control of us or certain of our affiliates could have certain adverse ramifications.

Under our June 19, 2017 credit agreement, a "change of control" of CenturyLink, Inc. constitutes an event of default. Moreover, if the credit ratings relating to certain of our currently outstanding long-term debt securities are downgraded in the manner specified thereunder in connection with a "change of control" of CenturyLink, Inc., then we will be required to offer to repurchase such debt securities. The long-term debt securities of several of our subsidiaries include similar covenants that could, under similar circumstances in connection with a "change of control" of one of the subsidiaries, require us to offer to repurchase such securities. If, due to lack of cash, legal or contractual impediments, or otherwise, we fail to offer to repurchase such debt securities, such failure could constitute an event of default under such debt securities. Any default under our credit facility or these debt securities could in turn constitute a default under other of our agreements relating to our indebtedness outstanding at that time.

Moreover, the existence of these default or repurchase provisions may in certain circumstances render it more difficult or discourage a sale or takeover of us, or the removal of our incumbent directors.

Our business requires us to incur substantial capital and operating expenses, which reduces our available free cash flow.

Our business is capital intensive. We expect to continue to require significant cash to maintain, upgrade and expand our network infrastructure as a result of several factors, including:

- changes in customers' service requirements, including increased demands by customers to transmit larger amounts of data at faster speeds;
- our above-described need to (i) consolidate and simplify our various legacy systems, (ii) strengthen our customer support systems and (iii) support our development and launch of new products and services;
- · technological advances of our competitors; and
- our regulatory commitments, including infrastructure construction requirements arising out of our participation in the FCC's CAF Phase II program, which are discussed further herein.

We may be unable to expand or adapt our network infrastructure to respond to these developments in a timely manner, at a commercially reasonable cost or on terms producing satisfactory returns on our investment.

In addition to investing in expanded networks, new products or new technologies, we must from time to time invest capital to (i) replace some of our aging equipment that supports many of our traditional services that are experiencing revenue declines or (ii) convert older systems to simplify and modernize our network. While we believe that our currently planned level of capital expenditures will meet both our maintenance and core growth requirements, this may not be the case if demands on our network continue to accelerate or other circumstances underlying our expectations change. Increased spending could, among other things, adversely affect our operating margins, cash flows, results of operations and financial position.

Similarly, we continue to anticipate incurring substantial operating expenses to support our incumbent services and growth initiatives. We may be unable to sufficiently manage or reduce these costs, even if revenue in some of our lines of business are decreasing. If so, our operating margins will be adversely impacted.

As a holding company, we rely on payments from our operating companies to meet our obligations.

As a holding company, substantially all of our income and operating cash flow is dependent upon the earnings of our subsidiaries and their distribution of those earnings to us in the form of dividends, loans or other payments. As a result, we rely upon our subsidiaries to generate cash flows in amounts sufficient to fund our obligations, including the payment of our long-term debt. Our subsidiaries are separate and distinct legal entities and have no obligation to pay any amounts owed by us, except to the extent they have guaranteed such payments. Similarly, subject to limited exceptions for tax-sharing or cash management purposes, our subsidiaries have no obligation to make any funds available to us to repay our obligations, whether by dividends, loans or other payments. As discussed in greater detail elsewhere herein, restrictions imposed by credit instruments or other agreements applicable to certain of our subsidiaries limit the amount of funds that our subsidiaries are permitted to transfer to us, including the amount of dividends that may be paid to us. Moreover, our rights to receive assets of any subsidiary upon its liquidation or reorganization will be effectively subordinated to the claims of creditors of that subsidiary, including trade creditors. In addition, the laws under which our subsidiaries were organized typically restrict the amount of dividends that they may pay. The ability of our subsidiaries to transfer funds could be further restricted under applicable tax laws or orders imposed by state regulators (either in connection with obtaining necessary approvals for our acquisitions or in connection with our regulated operations). For all these reasons, you should not assume that our subsidiaries will be able in the future to generate and distribute to us cash in amounts sufficient to fund our cash requirements. See "Management's Discussion and Analysis of Financial Condition and Results of Operations — Liquidity and Capital Resources" included elsewhere in this report for further discussion of

We cannot assure you that we will continue paying dividends at the current rates, or at all.

For the reasons noted below, we cannot assure you that we will continue periodic dividends on our capital stock at the current rates, or at all. From time to time, our board has reduced our dividend rate, including reductions in early 2019 and early 2013.

As noted in the immediately preceding risk factor, because we are a holding company with no material assets other than the stock of our subsidiaries, our ability to pay dividends will depend on our subsidiaries generating a sufficient amount of earnings and cash flow and their ability to furnish funds to us in the form of dividends, loans or other payments.

Any quarterly dividends on our common stock and our outstanding shares of preferred stock will be paid from funds legally available for such purpose when, as and if declared by our Board of Directors. Decisions on whether, when and in which amounts to continue making any future dividend distributions will remain at all times entirely at the discretion of our Board of Directors, which reserves the right to change or terminate our dividend practices at any time and for any reason without prior notice, including without limitation any of the following:

- · our supply of cash or other liquid assets is anticipated to remain under pressure for the various reasons described in this report;
- our cash requirements or plans might change for a wide variety of reasons, including changes in our financial position, capital allocation
 plans (including a desire to retain or accumulate cash), capital spending plans, stock purchase plans, acquisition strategies, strategic
 initiatives, debt payment plans (including a desire to maintain or improve credit ratings on our debt securities), pension funding or other
 benefits payments:
- our ability to service and refinance our current and future indebtedness and our ability to borrow or raise additional capital to satisfy our capital needs;
- the amount of dividends that we may distribute to our shareholders is subject to restrictions under Louisiana law and restrictions imposed by our existing or future credit facilities, debt securities, outstanding preferred stock securities, leases and other agreements, including restricted payment and leverage covenants; and
- the amount of cash that our subsidiaries may make available to us, whether by dividends, loans or other payments, may be subject to the legal, regulatory and contractual restrictions described in the immediately preceding risk factor.

Based on its evaluation of these and other relevant factors, our Board of Directors may, in its sole discretion, decide not to declare a dividend on our common stock or our outstanding shares of preferred stock for any period for any reason without prior notice, regardless of whether we have funds legally available for such purposes. Holders of our equity securities should be aware that they have no contractual or other legal right to receive dividends.

Similarly, holders of our common stock should be aware that repurchases of our common stock under any repurchase plan then in effect are completely discretionary, and may be suspended or discontinued at any time for any reason regardless of our financial position.

Our current dividend practices could limit our ability to deploy cash for other beneficial purposes.

The current practice of our Board of Directors to pay common share dividends reflects a current intention to distribute to our shareholders a substantial portion of our cash flow. As a result, we may not retain a sufficient amount of cash to apply to other transactions that could be beneficial to our shareholders or debtholders, including stock buybacks, debt prepayments or capital expenditures that strengthen our business. In addition, our ability to pursue any material expansion of our business through acquisitions or increased capital spending may depend more than it otherwise would on our ability to obtain third party financing.

We cannot assure you whether, when or in what amounts we will be able to use our net operating loss carryforwards, or when they will be depleted.

As of December 31, 2018, CenturyLink had approximately \$7.3 billion of federal net operating loss carryforwards, ("NOLs"), which for U.S. federal income tax purposes can be used to offset future taxable income. A significant portion of our federal NOLs were acquired through the Level 3 acquisition and are subject to limitations under Section 382 of the Internal Revenue Code ("Code") and related Treasury regulations. It should be noted that issuances or sales of our stock (including certain transactions outside of our control) could result in an ownership change of CenturyLink under Section 382, which may further limit our use of the NOLs. For these and other reasons, you should be aware that these limitations could restrict our ability to use these NOLs in the amounts we project or could limit our flexibility to pursue otherwise favorable transactions.

At December 31, 2018, we had state NOL carryforwards of approximately \$19 billion. A significant portion of the state NOL carryforwards are generated in states where separate company income tax returns are filed and our subsidiaries that generated the losses may not have the ability to generate income in sufficient amounts to realize these losses. In addition, certain of these state NOL carryforwards will be limited by state laws related to ownership changes. As a result, we expect to utilize only a small portion of the state NOL carryforwards, and consequently have determined that as of December 31, 2018, these state NOL carryforwards, net of federal benefit, had a net tax benefit (after giving effect to our valuation allowance) of \$438 million.

Additionally, we have foreign NOL carryforwards of \$6 billion. A significant portion of the foreign NOL carryforwards are generated in subsidiaries that do not have a history of earnings and may not have the ability to generate income in sufficient amounts to realize the losses. As of December 31, 2018, we have determined that these foreign NOL carryforwards had a net benefit of \$290 million (after giving effect to our valuation allowances).

Increases in costs for pension and healthcare benefits for our active and retired employees may reduce our profitability and increase our funding commitments.

As of December 31, 2018, we had approximately 35,000 active employees participating in our company sponsored benefit plans, approximately 68,000 active and retired employees and surviving spouses eligible for post-retirement benefits, approximately 66,000 pension retirees and approximately 13,000 former employees with vested pension benefits participating in our benefit plans. The cost to fund the pension and healthcare benefit plans for our active and retired employees has a significant impact on our profitability. Our costs of maintaining our pension and healthcare plans, and the future funding requirements for these plans, are affected by several factors, most of which are outside our control, including:

· decreases in investment returns on funds held by our pension and other benefit plan trusts;

- changes in prevailing interest rates and discount rates or other factors used to calculate the funding status of our pension and other post-retirement plans;
- · increases in healthcare costs generally or claims submitted under our healthcare plans specifically;
- increasing longevity of our employees and retirees;
- the impact of the continuing implementation, modification or potential repeal of current federal healthcare legislation and regulations promulgated thereunder;
- increases in the number of retirees who elect to receive lump sum benefit payments;
- increases in insurance premiums we are required to pay to the Pension Benefit Guaranty Corporation, an independent agency of the United States government that must cover its own underfunded status by collecting premiums from a declining population of pension plans that are qualified under the U.S. tax code:
- · changes in plan benefits; and
- · changes in funding laws or regulations.

Increased costs under these plans could reduce our profitability and increase our funding commitments to our pension plans. Any future material cash contributions could have a negative impact on our liquidity by reducing our cash flows available for other purposes. Similarly, depletion of assets placed in trust by us to fund these benefits, such as those discussed elsewhere herein, will similarly reduce our liquidity by requiring us to deploy a portion of our cash flows to fund such benefit payments.

As of December 31, 2018, our pension plans and our other post-retirement benefit plans were substantially underfunded from an accounting standpoint. See Note 10—Employee Benefits to our consolidated financial statements included in Item 8 of this report. For more information on our obligations under our defined benefit pension plans and other post-retirement benefit plans, please see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources—Pension and Post-retirement Benefit Obligations" included in Item 7 of this report.

For additional information concerning our liquidity and capital resources, see Item 7 of this report. For a discussion of certain currency and liquidity risks associated with our international operations, see "Risk Factors—Risks Affecting Our Business—Our international operations expose us to various regulatory, currency, tax, legal and other risks."

Other Risks

We face risks from natural disasters, which can disrupt our operations and cause us to incur substantial additional capital and operating costs.

A substantial number of our facilities are located in Florida, Alabama, Louisiana, Texas, North Carolina, South Carolina and other coastal states, which subjects them to the risks associated with severe tropical storms, hurricanes and tornadoes, including downed telephone lines, flooded facilities, power outages, fuel shortages, damaged or destroyed property and equipment, and work interruptions. Although we maintain property and casualty insurance on our property (excluding our above ground outside plant) and may, under certain circumstances, be able to seek recovery of some additional costs through increased rates, only a portion of our additional costs directly related to such natural disasters have historically been recoverable. We cannot predict whether we will continue to be able to obtain insurance for catastrophic hazard-related damages or, if obtainable and carried, whether this insurance will be adequate to cover our losses. In addition, we expect any insurance of this nature to be subject to substantial deductibles, retentions and coverage exclusions, and the premiums to be based on our loss experience. For all these reasons, any future hazard-related costs and work interruptions could adversely affect our operations and our financial condition.

Terrorist attacks and other acts of violence or war may adversely affect the financial markets and our business.

Future terrorist attacks or armed conflicts may directly affect our physical facilities or those of our customers. These events could cause consumer confidence and spending to decrease or result in increased volatility in the U.S. and world financial markets and economy. Any of these occurrences could materially adversely affect our business.

If conditions or assumptions differ from the judgments, assumptions or estimates used in our critical accounting policies or forward-looking statements, our consolidated financial statements and related disclosures could be materially affected.

The preparation of financial statements and related disclosures in conformity with U.S. generally accepted accounting principles requires management to make judgments, assumptions and estimates that affect the amounts reported in our consolidated financial statements and accompanying notes including the judgments, assumptions and estimates applied pursuant to our critical accounting policies, which are described in "Management's Discussion and Analysis of Financial Condition and Results of Operations—Critical Accounting Policies and Estimates" in Item 7 of this report. If future events or assumptions differ significantly from the judgments, assumptions and estimates applied in connection with preparing our historical financial statements, our future financial statements could be materially impacted.

While frequently presented with numeric specificity, the guidance and other forward-looking statements that we disseminate from time to time is based on numerous variables and assumptions (including, but not limited to, those related to industry performance and competition and general business, economic, market and financial conditions and additional matters specific to our business, as applicable) that are inherently subjective and speculative and are largely beyond our control. As a result, actual results may differ materially from our guidance or other forward-looking statements. Similarly, for a variety of reasons, we may change our intentions, strategies or plans at any time, which could materially alter our actual results from those previously anticipated. For additional information, see "Special Note Regarding Forward-Looking Statements" in Item 1 of this report.

We identified material weaknesses in our internal control over financial reporting as of December 31, 2018, and the occurrence of this or any other future material weakness or significant deficiencies could have a material adverse effect on us.

Our management recently concluded that, as described under the heading Item 9A. Controls and Procedures, we had material weaknesses as of December 31, 2018 and therefore as of that date did not maintain effective internal control over financial reporting, which is a requirement of the Securities Exchange Act of 1934. As a result of that evaluation, management concluded that two material weaknesses existed as described below.

- Ineffective design and operation of process level internal controls over the fair value measurement of certain assets acquired and liabilities assumed from Level 3.
 - These deficiencies arose because (i) we did not conduct an effective risk assessment to identify and assess changes we needed to make to our financial reporting and process level controls, related to fair value measurement of assets acquired and liabilities assumed in the transaction with Level 3, (ii) we did not clearly assign responsibility for the design, implementation, and operation of controls over the fair value measurements and (iii) we did not maintain effective information and communication processes to ensure the right information was available to personnel on a timely basis so they could fulfill their control responsibilities related to the fair value measurements.
- Ineffective design and operation of certain process level internal controls over the existence and accuracy of revenue transactions.
 - These deficiencies arose because we did not conduct an effective risk assessment to identify risks of material misstatement related to the existence and accuracy of revenue transactions.

We plan to execute our plans to remediate the material weaknesses identified above as soon as feasible. However, the remedial measures we take may not be adequate to avoid other control deficiencies in the future. There can be no assurance that any system of internal control over financial reporting will be successful in

preventing all errors or fraud or in making all material information known in a timely manner to the appropriate levels of management. As a result, it is possible that our current or future financial statements may not comply with generally accepted accounting principles, will contain a material misstatement or will not be available on a timely basis, any of which could cause investors to lose confidence in us and lead to, among other things, unanticipated legal, accounting and other expenses, delays in filing required financial disclosures, enforcement actions by regulatory authorities, fines, penalties, the delisting of our securities and liabilities arising from shareholder litigation.

Lapses in disclosure controls and procedures or internal control over financial reporting could materially and adversely affect our operations, profitability or reputation.

There can be no assurance that our disclosure controls and procedures will be effective in the future. Any such lapses or deficiencies may materially and adversely affect our business, operating results or financial condition, restrict our ability to access the capital markets, require us to expend significant resources to correct the lapses or deficiencies, expose us to regulatory or legal proceedings, including litigation brought by private individuals, subject us to fines, penalties or judgments, harm our reputation, or otherwise cause a decline in investor confidence and our stock price.

If our goodwill or other intangible assets become impaired, we may be required to record a significant charge to earnings and reduce our stockholders' equity.

As of December 31, 2018, approximately 55% of our total consolidated assets reflected on the consolidated balance sheet included in this report consisted of goodwill, customer relationships and other intangible assets. Under U.S. generally accepted accounting principles, most of these intangible assets must be tested for impairment on an annual basis or more frequently whenever events or circumstances indicate that their carrying value may not be recoverable. From time to time, including in the fourth quarter of 2018, we have recorded large non-cash charges to earnings in connection with required reductions of the value of our intangible assets. If our intangible assets are determined to be impaired in the future, we may be required to record additional significant, non-cash charges to earnings during the period in which the impairment is determined to have occurred. Any such charges could, in turn, have a material adverse effect on our results of operation, financial condition or ability to comply with financial covenants in our debt instruments.

Shareholder Activism Efforts Could Cause a Material Disruption to Our Business

While we always welcome constructive input from our shareholders and regularly engage in dialogue with our shareholders to that end, activist shareholders may from time to time engage in proxy solicitations, advance shareholder proposals or otherwise attempt to affect changes or acquire control over us. Responding to these actions can be costly and time-consuming, may disrupt our operations and divert the attention of the Board and management from the management of our operations and the pursuit of our business strategies, particularly if shareholders advocate actions that are not supported by other shareholders, our board or management. Further, the perceived uncertainty as to our future direction as a result of such activities could affect our market price and volatility in our common stock.

The Tax Cuts and Jobs Act will have a substantial impact on us.

The Tax Cuts and Jobs Act (the "Act") enacted in December 2017 significantly changed U.S. tax law by reducing the U.S. corporate income tax rate and making certain changes to U.S. taxation of income earned by foreign subsidiaries, capital expenditures, interest expense and various other items. The net impact of this Act, as applied to date, has been favorable to us. However, the Act is quite complex and the impacts could potentially change as additional regulatory guidance is received from the Internal Revenue Service. As a result, our views on the Act's ultimate impact on us could change.

Additional changes in tax laws or tax audits could adversely affect us.

Like all large multinational businesses, we are subject to multiple sets of complex and varying tax laws and rules. Legislators and regulators at various levels of government may from time to time change existing tax laws or regulations or enact new laws or regulations. In many cases, the application of existing, newly enacted or amended tax laws may be uncertain and subject to differing interpretations that could negatively impact our operating results or financial condition. We are also subject to frequent and regular audits by a broad range of foreign, federal, state and local tax authorities. These audits could subject us to tax liabilities if adverse positions are taken by these tax authorities.

We believe that we have adequately provided for tax contingencies. However, our tax audits and examinations may result in tax liabilities that differ materially from those that we have recognized in our consolidated financial statements. Because the ultimate outcomes of all of these matters are uncertain, we can give no assurance as to whether an adverse result from one or more of them will have a material effect on our financial results.

The trading price of our common stock could be reduced if a large number of shares of our common stock are sold in the public market.

Our articles of incorporation currently authorize us to issue additional shares of our common stock, frequently without shareholder approval. Such additional issuances may dilute the beneficial ownership and voting power of our shareholders, and could reduce the trading price of our common stock. Similarly, the market price of our common stock could drop significantly if certain large holders of our common stock sell all or a substantial portion of their holdings in the public markets, or indicate their intent to do so. Similarly, the market price of our stock could be adversely affected if analysts or other market participants issue reports or make other statements that recommend the sale of our shares.

The rights agreement that we entered into to protect our ability to use our accumulated NOLs could discourage third parties from seeking strategic transactions with us that could be beneficial to our shareholders.

On February 13, 2019, we entered into the rights agreement in an effort to deter acquisitions of our common stock that might reduce our ability to use our NOL carryforwards. Under the rights agreement, from and after the record date of February 25, 2019, each share of our common stock carries with it one preferred share purchase right until the earlier of the date when the preferred share purchase rights become exercisable or expire. The rights agreement and the preferred share purchase rights issuable thereunder could discourage a third party from proposing a change of control or other strategic transaction concerning CenturyLink or otherwise have the effect of delaying or preventing a change of control of CenturyLink that other shareholders may view as beneficial.

Our other agreements and organizational documents and applicable law could similarly limit another party's ability to acquire us.

In addition to other restrictions mentioned above, a number of provisions in our organizational documents and various provisions of applicable law may delay, defer or prevent a future takeover of CenturyLink unless the takeover is approved by our Board of Directors. These provisions could deprive our shareholders of any related takeover premium. For additional information, please see our Registration Statement on Form 8-A/A filed with the SEC on March 2, 2015.

ITEM 1B. UNRESOLVED STAFF COMMENTS

None.

ITEM 2. PROPERTIES

Our property, plant and equipment consists principally of fiber-optic and metallic cables, high-speed transport equipment, electronics, switches, routers, cable landing stations, central office equipment, land and buildings related to our operations. Our gross property, plant and equipment consisted of the following components:

| | As of Decen | nber 31, |
|--|-------------|----------|
| | 2018 | 2017 |
| Land | 2% | 2% |
| Fiber, conduit and other outside plant (1) | 45% | 45% |
| Central office and other network electronics (2) | 35% | 36% |
| Support assets (3) | 15% | 15% |
| Construction in progress (4) | 3% | 2% |
| Gross property, plant and equipment | 100% | 100% |

- (1) Fiber, conduit and other outside plant consists of fiber and metallic cable, conduit, poles and other supporting structures.
- (2) Central office and other network electronics consists of circuit and packet switches, routers, transmission electronics and electronics providing service to customers.
- (3) Support assets consist of buildings, cable landing stations, data centers, computers and other administrative and support equipment.
- (4) Construction in progress includes inventory held for construction and property of the aforementioned categories that has not been placed in service as it is still under construction.

We own substantially all of our telecommunications equipment required for our business. However, we lease from third parties certain facilities, plant, equipment and software under various capital and operating lease arrangements when the leasing arrangements are more favorable to us than purchasing the assets. We also own and lease administrative offices in major metropolitan locations both in the United States and internationally. Substantially all of our network electronics equipment is located in buildings or on land that we own or lease, typically within our local service area. Outside of our local service area, our assets are generally located on real property pursuant to an agreement with the property owner or another person with rights to the property. It is possible that we may lose our rights under one or more of these agreements, due to their termination or expiration or in connection with legal challenges to our rights under such agreements. With the acquisition of Level 3 on November 1, 2017, we acquired, among other things, title or leasehold rights to various cable landing stations and data centers throughout the world related to undersea and terrestrial cable systems.

Our net property, plant and equipment was approximately \$26.4 billion and \$26.9 billion at December 31, 2018 and 2017, respectively. Substantial portions of our property, plant and equipment is pledged to secure the long-term debt of our subsidiaries or the guarantee obligations of our subsidiary guarantors. For additional information, see Note 8—Property, Plant and Equipment to our consolidated financial statements in Item 8 of Part II of this report.

ITEM 3. LEGAL PROCEEDINGS

The information contained under the subheadings "Pending Matters" and "Other Proceedings and Disputes" in Note 17—Commitments, Contingencies and Other Items to our consolidated financial statements included in Item 8 of Part II of this report is incorporated herein by reference.

ITEM 4. MINE SAFETY DISCLOSURES

Not applicable.

PART II

ITEM 5. MARKET FOR REGISTRANT'S COMMON EQUITY, RELATED STOCKHOLDER MATTERS AND ISSUER PURCHASES OF EQUITY SECURITIES

Our common stock is listed on the New York Stock Exchange ("NYSE") and the Berlin Stock Exchange and is traded under the symbol CTL and CYT, respectively.

At February 22, 2019, there were approximately 99,000 stockholders of record, although there were significantly more beneficial holders of our common stock.

As described in greater detail in "Risk Factors" in Item 1A of Part I of this report, the declaration and payment of dividends is at the discretion of our Board of Directors, and will depend upon our financial results, cash requirements, future prospects and other factors deemed relevant by our Board of Directors.

Issuer Purchases of Equity Securities

The following table contains information about shares of our previously-issued common stock that we withheld from employees upon vesting of their stock-based awards during the fourth quarter of 2018 to satisfy the related tax withholding obligations:

| | Total Number of Shares Withheld for Taxes | Average Price Paid Per Share |
|---------------|---|---------------------------------|
| Period | | |
| October 2018 | 19,346 | \$ 21.08 |
| November 2018 | 185,616 | 20.88 |
| December 2018 | 105,114 | 17.57 |
| Total | 310,076 | |

Equity Compensation Plan Information

See Item 12 of this report.

ITEM 6. SELECTED FINANCIAL DATA

The following tables of selected consolidated financial data should be read in conjunction with, and are qualified by reference to, our consolidated financial statements and notes thereto in Item 8 of Part II and "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Item 7 of Part II of this report.

The tables of selected financial data shown below are derived from our audited consolidated financial statements, which include the operating results, cash flows and financial condition of Level 3 beginning November 1, 2017. These historical results are not necessarily indicative of results that you can expect for any future period.

The following table summarizes selected financial information from our consolidated statements of operations.

| | | | Years Er | nded December 31 | , (1) | |
|--|----|------------------|----------------|---|---------------------|---------------------|
| | 2 | 018 (2)(3)(4)(5) | 2017 (3)(4)(5) | 2016 (4)(5) | 2015 ⁽⁵⁾ | 2014 ⁽⁶⁾ |
| | | | • | ns, except per sha ares in thousands | | |
| Operating revenue | \$ | 23,443 | 17,656 | 17,470 | 17,900 | 18,031 |
| Operating expenses | | 22,873 | 15,647 | 15,137 | 15,321 | 15,674 |
| Operating income | \$ | 570 | 2,009 | 2,333 | 2,579 | 2,357 |
| (Loss) income before income tax expense | \$ | (1,563) | 540 | 1,020 | 1,316 | 1,110 |
| Net (loss) income | \$ | (1,733) | 1,389 | 626 | 878 | 772 |
| Basic (loss) earnings per common share | \$ | (1.63) | 2.21 | 1.16 | 1.58 | 1.36 |
| Diluted (loss) earnings per common share | \$ | (1.63) | 2.21 | 1.16 | 1.58 | 1.36 |
| Dividends declared per common share | \$ | 2.16 | 2.16 | 2.16 | 2.16 | 2.16 |
| Weighted average basic common shares outstanding | | 1,065,866 | 627,808 | 539,549 | 554,278 | 568,435 |
| Weighted average diluted common shares outstanding | | 1,065,866 | 628,693 | 540,679 | 555,093 | 569,739 |

- (1) See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Results of Operations" in Item 7 of Part II of this report and in our preceding annual reports on Form 10-K for a discussion of unusual items affecting the results for each of the years presented.
- (2) During 2018, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill attributed to our consumer segment.
- (3) The enactment of the Tax Cuts and Jobs Act in December 2017 resulted in a re-measurement of our deferred tax assets and liabilities at the new federal corporate tax rate of 21%. The re-measurement resulted in tax expense of \$92 million and a tax benefit of approximately \$1.1 billion for 2018 and 2017, respectively.
- (4) During 2018, 2017 and 2016, we incurred Level 3 acquisition-related expenses of \$393 million, \$271 million and \$52 million, respectively. For additional information, see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Acquisition of Level 3" and Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report.
- (5) During 2018, 2017, 2016 and 2015, we recognized an incremental \$171 million, \$186 million, \$201 million and \$215 million, respectively, of revenue associated with the Federal Communications Commission ("FCC") Connect America Fund Phase II support program, as compared to revenue received under the previous interstate USF program.
- (6) During 2014, we recognized a \$60 million tax benefit associated with a deduction for the tax basis for worthless stock in a wholly-owned foreign subsidiary and a \$63 million pension settlement charge.

Selected financial information from our consolidated balance sheets is as follows:

| | | AS 0 | T December 31 | , | |
|---------------------------------------|--------------|--------|------------------|--------|--------|
| | 2018 | 2017 | 2016 | 2015 | 2014 |
| | | (Dol | lars in millions |) | |
| Net property, plant and equipment (1) | \$ 26,408 | 26,852 | 17,039 | 18,069 | 18,433 |
| Goodwill (1)(2) | 28,031 | 30,475 | 19,650 | 20,742 | 20,755 |
| Total assets (3) | 70,256 | 75,611 | 47,017 | 47,604 | 49,103 |
| Total long-term debt (3)(4) | 36,061 | 37,726 | 19,993 | 20,225 | 20,503 |
| Total stockholders' equity | 19,828 | 23,491 | 13,399 | 14,060 | 15,023 |

As of December 24

- (1) During 2016, as a result of our then pending sale of our colocation business and data centers, we reclassified \$1.1 billion in net property, plant and equipment and \$1.1 billion of goodwill to assets held for sale which is included in other current assets on our consolidated balance sheet. See Note 3—Sale of Data Centers and Colocation Business to our consolidated financial statements in Item 8 of Part II of this report, for additional information.
- (2) During 2018, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill attributed to our consumer segment.
- (3) In 2015, we adopted both ASU 2015-03 "Simplifying the Presentation of Debt Issuance Costs" and ASU 2015-17 "Balance Sheet Classification of Deferred Taxes" by retrospectively applying the requirements of the ASUs to our previously issued consolidated financial statements. The adoption of both ASU 2015-03 and ASU 2015-17 reduced total assets by \$1.0 billion and \$1.3 billion in each year for the two years ended December 31, 2014, respectively, and ASU 2015-03 reduced total long-term debt by \$168 million and \$157 million in each year for the two years ended December 31, 2014, respectively.
- (4) Total long-term debt includes current maturities of long-term debt and capital lease obligations of \$305 million for the year ended December 31, 2016 associated with assets held for sale. For additional information on our total long-term debt, see Note 6—Long-Term Debt and Credit Facilities to our consolidated financial statements in Item 8 of Part II of this report. For total contractual obligations, see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Future Contractual Obligations" in Item 7 of Part II of this report.

Selected financial information from our consolidated statements of cash flows is as follows:

| | | Years E | nded December 31, | | |
|---|-------------|---------|-------------------|---------|---------|
| | 2018 | 2017 | 2016 | 2015 | 2014 |
| | | (Dol | lars in millions) | | |
| Net cash provided by operating activities | \$ 7,032 | 3,878 | 4,608 | 5,153 | 5,188 |
| Net cash used in investing activities | (3,078) | (8,871) | (2,994) | (2,853) | (3,077) |
| Net cash (used in) provided by financing activities | (4,023) | 5,356 | (1,518) | (2,301) | (2,151) |
| Payments for property, plant and equipment and capitalized software | (3,175) | (3,106) | (2,981) | (2,872) | (3,047) |

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

All references to "Notes" in this Item 7 of Part II refer to the Notes to Consolidated Financial Statements included in Item 8 of Part II of this report. Certain statements in this report constitute forward-looking statements. See "Special Note Regarding Forward-Looking Statements" in Item 1 of Part I of this report for factors relating to these statements and "Risk Factors" in Item 1A of Part I of this report for a discussion of certain risk factors applicable to our business, financial condition, results of operations, liquidity or prospects.

Overview

We are an international facilities-based communications company engaged primarily in providing a broad array of integrated services to our residential and business customers. With approximately 450,000 route miles of fiber optic cable globally, we believe we are among the largest providers of communications services to domestic and global enterprise customers and the third largest wireline telecommunications company in the United States. We provide services in over 60 countries, with most of our revenue being derived in the United States.

At December 31, 2018, we served 4.8 million consumer broadband subscribers. Our methodology for counting consumer broadband subscribers may not be comparable to those of other companies. We no longer report or discuss access lines as a key operating metric given the significant migration in our industry from legacy services to IP-enabled services.

Acquisition of Level 3

On November 1, 2017, CenturyLink, Inc. ("CenturyLink") acquired Level 3 Communications, Inc. ("Level 3") through successive merger transactions, including a merger of Level 3 with and into a merger subsidiary, which survived such merger as our indirect wholly-owned subsidiary under the name of Level 3 Parent, LLC.

During the year ended December 31, 2018, we recognized \$ 391 million of integration-related expenses associated with our activities related to the Level 3 acquisition. During 2018, we also recognized \$2 million in merger-related transaction costs, including investment banker and legal fees.

Our consolidated financial statements include the accounts of CenturyLink and its majority owned subsidiaries, including Level 3 beginning on November 1, 2017. Due to the significant size of the acquisition, direct comparison of our results of operations for the periods ending on or after December 31, 2017 to prior periods are less meaningful than usual.

As a result of the acquisition, Level 3's assets and liabilities have been revalued and recorded at their fair value. The assignment of estimated fair value requires a significant amount of judgment. The use of fair value measures affects the comparability of our post-acquisition financial information and may make it more difficult to predict earnings in future periods. We completed our final fair value determinations during the fourth quarter 2018. Our final fair value determinations were different than those preliminary values reflected in our consolidated financial statements at December 31, 2017 and resulted in an increase in goodwill of \$340 million and an increase to other noncurrent assets offset by a decrease in customer relationships during 2018.

In the discussion that follows, we refer to the business that we operated prior to the Level 3 acquisition as "Legacy CenturyLink", and we refer to the incremental business activities that we now operate as a result of the Level 3 acquisition as "Legacy Level 3."

For additional information about our acquisition of Level 3, see (i) Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report and (ii) the documents we filed with the SEC on February 13, 2017, November 1, 2017 and January 16, 2018.

Sale of Data Centers and Colocation Business

On May 1, 2017, we sold our data centers and colocation business to a consortium led by BC Partners, Inc. and Medina Capital ("the Purchaser") in exchange for pre-tax cash proceeds of \$1.8 billion and a minority stake in the limited partnership that owns the consortium's newly-formed global secure infrastructure company, Cyxtera Technologies. As part of the transaction, the Purchaser acquired 57 of our data centers and assumed our capital lease obligations, which amounted to \$294 million on May 1, 2017, related to the divested properties.

Our colocation business generated revenue (excluding revenue from affiliates) of \$210 million from January 1, 2017 through May 1, 2017, and \$622 million for the year ended December 31, 2016 (a small portion of which has been retained by us).

This transaction did not meet the accounting requirements for a sale-leaseback transaction as described in ASC 840-40, *Leases - Sale-Leaseback Transaction*. Under the failed-sale-leaseback accounting model, we are deemed under GAAP to still own certain real estate assets sold to the Purchaser.

After factoring in the costs to sell the data centers and colocation business, excluding the impacts from the failed-sale-leaseback accounting treatment, the sale resulted in a \$20 million gain as a result of the aggregate value of the consideration we received exceeding the carrying value of the assets sold and liabilities assumed. Based on the fair market values of the failed-sale-leaseback assets, the failed-sale-leaseback accounting treatment resulted in a loss of \$102 million as a result of the requirement to treat a certain amount of the pre-tax cash proceeds from the sale of the assets as though it were the result of a financing obligation. The combined net loss of \$82 million is included in selling, general and administrative expenses in our consolidated statement of operations for the year ended December 31, 2017.

Effective with the January 1, 2019 implementation date of the new accounting standard for Leases (ASU 2016-02), this particular accounting treatment will no longer be applicable to our May 1, 2017 divestiture transaction. Consequently, the above-described real estate assets and corresponding financing obligation will be derecognized as of January 1, 2019 from our future consolidated balance sheets resulting in an increase of \$115 million to stockholder's equity.

See Note 3—Sale of Data Centers and Colocation Business for additional information on the sale and Note 1— Background And Summary Of Significant Accounting Policies for discussion of the impact of implementing ASU 2016-02 to our consolidated financial statements in Item 8 of Part II of this report.

Segments

At December 31, 2018, we had the following two segments:

- Business Segment. Under our business segment, we provide our products and services to large domestic and global enterprises, small
 and medium businesses, federal, state and local governments and wholesale customers, including other communication providers. Our
 products and services offered to these customers include our IP and Data Services suite of products, which includes VPN and hybrid
 networking, Ethernet and IP services; Transport and Infrastructure, which includes wavelengths and private line, dark fiber, colocation
 and data center services, and professional services; Voice Services, which includes local, long-distance, toll-free and unified
 communications services; and IT and Managed services, all of which are described further under "Products and Services"; and
- Consumer Segment. Under our consumer segment, we provide our products and services to residential customers. Our products and services offered to these customers include our broadband, local and long-distance voice, video and other ancillary services.

See Note 15—Segment Information to our consolidated financial statements in Item 8 of Part II of this report for additional information.

Results of Operations

The following table summarizes the results of our consolidated operations for the years ended December 31, 2018, 2017 and 2016:

| | Years | Ended December 3 | 31, |
|--|--------------------|---|---------------------|
| | 2018 (1)(2)(3) | 2017 (2)(3) | 2016 ⁽³⁾ |
| | ` | ers in millions exce er share amounts) | pt |
| Operating revenue | \$ 23,443 | 17,656 | 17,470 |
| Operating expenses | 22,873 | 15,647 | 15,137 |
| Operating income | 570 | 2,009 | 2,333 |
| Other expense, net | 2,133 | 1,469 | 1,313 |
| Income tax expense (benefit) | 170 | (849) | 394 |
| Net (loss) income | \$ (1,733) | 1,389 | 626 |
| Basic (loss) earnings per common share | \$ (1.63) | 2.21 | 1.16 |
| Diluted (loss) earnings per common share | \$ (1.63) | 2.21 | 1.16 |

- (1) During 2018, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill attributed to our consumer segment.
- (2) The enactment of the Tax Cuts and Jobs Act in December 2017 resulted in a re-measurement of our deferred tax assets and liabilities at the new federal corporate tax rate of 21%. The re-measurement resulted in tax expense of \$92 million and a tax benefit of approximately \$1.1 billion for 2018 and 2017, respectively.
- (3) During 2018, 2017 and 2016, we incurred acquisition-related expenses of \$393 million, \$271 million and \$52 million, respectively. For additional information, see "Acquisition of Level 3" above and Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report.

For over a decade, we have experienced revenue declines, excluding the impact of acquisitions during the period, primarily due to declines in private line customers, switched access rates and minutes of use. To partially mitigate these revenue declines, we remain focused on efforts to, among other things:

- promote long-term relationships with our customers through bundling of integrated services;
- increase the capacity, speed and usage of our networks;
- provide a wide array of diverse services, including enhanced or additional services that may become available in the future due to, among other things, advances in technology or improvements in our infrastructure;
- provide our premium services to a higher percentage of our customers;
- · pursue acquisitions of additional assets if available at attractive prices;
- · increase prices on our products and services if and when practicable; and
- market our products and services to new customers.

Operating Revenue

We categorize our products, services and revenue among the following five categories:

• IP and Data Services, which include primarily VPN data networks, Ethernet, IP, video (including our facilities-based video services and Vyvx broadcast services) and other ancillary services;

- Transport and Infrastructure, which include broadband, private line (including business data services), data center facilities and services, including cloud, hosting and application management solutions, wavelength, equipment sales and professional services, network security services and other ancillary services;
- Voice and Collaboration, which includes primarily local and long distance voice, including wholesale voice, and other ancillary services;
- IT and Managed Services, which include information technology services and managed services, which may be purchased in conjunction with our other network services; and
- Regulatory Revenue, which consist of (i) Universal Service Fund ("USF"), Connect America Fund ("CAF") and other support payments
 designed to reimburse us for various costs related to certain telecommunications services and (ii) other operating revenue from the
 leasing and subleasing of space, none of which is included in our segment revenue.

Years Ended December 31,

Increase /

For more detailed information, see "Products and Services" in Item I of this report.

The following tables summarize our consolidated operating revenue recorded under each of our five above described revenue categories:

| | | | • | IIICI Ease / | | |
|---|----|------------------------------------|---|-----------------------------|----------------------|--|
| | | 2018 | 2017 | (Decrease) | % Change | |
| | | 1) | Dollars in millions) | | | |
| IP and Data Services (1) | \$ | 7,279 | 4,083 | 3,196 | 78 % | |
| Transport and Infrastructure (2) | | 8,248 | 6,345 | 1,903 | 30 % | |
| Voice and Collaboration (3) | | 6,572 | 5,844 | 728 | 12 % | |
| IT and Managed Services (4) | | 621 | 652 | (31) | (5)% | |
| Regulatory Revenue (5) | | 723 | 732 | (9) | (1)% | |
| Total operating revenue | \$ | 23,443 | 17,656 | 5,787 | 33 % | |
| | | | | | | |
| | | | | | | |
| | • | Years Ended De | cember 31, | Increase / | | |
| | | ears Ended De 2017 | 2016 | Increase / (Decrease) | % Change | |
| | | 2017 | | | % Change | |
| IP and Data Services (1) | \$ | 2017 | 2016 | | % Change | |
| IP and Data Services ⁽¹⁾ Transport and Infrastructure ⁽²⁾ | | 2017 | 2016 Dollars in millions) | (Decrease) | _ | |
| | \$ | 2017 (E | 2016 Dollars in millions) 3,418 | (Decrease) | 19 % | |
| Transport and Infrastructure (2) | \$ | 2017 (E 4,083 6,345 | 2016 Dollars in millions) 3,418 6,583 | (Decrease) 665 (238) | 19 % (4)% | |
| Transport and Infrastructure (2) Voice and Collaboration (3) | \$ | 2017 (E 4,083 6,345 5,844 | 2016 Dollars in millions) 3,418 6,583 6,124 | (Decrease) 665 (238) (280) | 19 % (4)% (5)% | |

⁽¹⁾ Includes primarily VPN data network, Ethernet, IP, video and ancillary revenue.

⁽²⁾ Includes primarily broadband, private line (including business data services), colocation and data centers, wavelength and ancillary revenue.

⁽³⁾ Includes local, long-distance and other ancillary revenue.

⁽⁴⁾ Includes IT services and managed services revenue.

⁽⁵⁾ Includes CAF Phase I, CAF Phase II, federal and state USF support revenue, sublease rental income and failed-sale leaseback income.

Our total operating revenue increased by \$5.8 billion, or 33%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017 due to the inclusion of \$6.7 billion in Legacy Level 3 post-acquisition operating revenue in our consolidated operating revenue. Total operating revenue for Legacy CenturyLink decreased by \$935 million for the year ended December 31, 2018 as compared to the year ended December 31, 2017. The Legacy CenturyLink decline in total operating revenue is primarily due to lower voice and collaboration, transport and infrastructure and IP and data services revenue. The decrease in voice and collaboration was due to continued decreases in revenue from our traditional voice telecommunications services. The decrease in transport and infrastructure reflects the decrease in volumes from our colocation business sale, reduced sales volume of private line (including business data services) services, which were partially offset with an increase in broadband and managed security services. The decrease in IP and data services was primarily due to a decrease in retail revenue, partially offset by an increase in VPN data network services.

Total operating revenue increased by \$186 million, or 1% for the year ended December 31, 2017 as compared to the year ended December 31, 2016 due to the inclusion of \$1.4 billion in post-acquisition Legacy Level 3 operating revenue in our 2017 consolidated operating revenue. Total operating revenue for Legacy CenturyLink decreased by \$1.2 billion for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The decline in total operating revenue reflects the continuing loss of access lines, loss of long-distance revenue primarily due to the displacement of traditional wireline telephone services by other competitive products and services, including data and wireless communication services, and reductions in the volume of our private line (including business data services) services. Our total operating revenue for the year ended December 31, 2017 were also impacted by the May 1, 2017 sale of our data centers and colocation business, which resulted in a reduction of colocation revenue of \$396 million for the year ended December 31, 2017 as compared to the prior year period.

Further analysis of our segment operating revenue and trends impacting our performance are provided below in "Segment Results."

Operating Expenses

Our current definitions of operating expenses are as follows:

- Cost of services and products (exclusive of depreciation and amortization) are expenses incurred in providing products and services to our customers. These expenses include: employee-related expenses directly attributable to operating and maintaining our network (such as salaries, wages, benefits and professional fees); facilities expenses (which include third-party telecommunications expenses we incur for using other carriers' networks to provide services to our customers); rents and utilities expenses; equipment sales expenses (such as data integration and modem expenses); costs owed to universal service funds (which are federal and state funds that are established to promote the availability of telecommunications services to all consumers at reasonable and affordable rates, among other things, and to which we are often required to contribute); and other expenses directly related to our operations; and
- Selling, general and administrative expenses are corporate overhead and other operating expenses. These expenses include:
 employee-related expenses (such as salaries, wages, internal commissions, benefits and professional fees) directly attributable to
 selling products or services and employee-related expenses for administrative functions; marketing and advertising; property and other
 operating taxes and fees; external commissions; litigation expenses associated with general matters; bad debt expense; and other
 selling, general and administrative expenses.

These expense classifications may not be comparable to those of other companies.

The following tables summarize our operating expenses:

| | | Years Ended De | ecember 31, | Increase / | | |
|--|----|----------------|---------------------|------------|----------|--|
| | | 2018 | 2017 | (Decrease) | % Change | |
| | | (D | ollars in millions) | | | |
| Cost of services and products (exclusive of depreciation and amortization) | \$ | 10,862 | 8,203 | 2,659 | 32% | |
| Selling, general and administrative | | 4,165 | 3,508 | 657 | 19% | |
| Depreciation and amortization | | 5,120 | 3,936 | 1,184 | 30% | |
| Impairment of goodwill | | 2,726 | - | 2,726 | nm | |
| Total operating expenses | \$ | 22,873 | 15,647 | 7,226 | 46% | |
| | Y | ears Ended Dec | cember 31, | Increase / | | |

| | • | Years Ended De | ecember 31, | Increase / | | |
|--|----|----------------|---------------------|------------|----------|--|
| | | 2017 | 2016 | (Decrease) | % Change | |
| | | (D | ollars in millions) | | | |
| Cost of services and products (exclusive of depreciation and amortization) | \$ | 8,203 | 7,774 | 429 | 6% | |
| Selling, general and administrative | | 3,508 | 3,447 | 61 | 2% | |
| Depreciation and amortization | | 3,936 | 3,916 | 20 | 1% | |
| Total operating expenses | \$ | 15,647 | 15,137 | 510 | 3% | |

Percentages greater than 200% and comparisons between positive and negative values or to/from zero values are considered not meaningful.

Cost of Services and Products (exclusive of depreciation and amortization)

Cost of services and products (exclusive of depreciation and amortization) increased by \$2.7 billion, or 32%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017. The increase in costs of services and products (exclusive of depreciation and amortization) was attributable to the inclusion of \$3.2 billion Legacy Level 3 post-acquisition costs (net of intercompany eliminations) in our consolidated costs of services and products (exclusive of depreciation and amortization). Costs of services and products (exclusive of depreciation and amortization) for Legacy CenturyLink decreased \$588 million, or 8%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017. The decrease was primarily due to reductions in salaries and wages and employee related expenses from lower headcount, reduced overtime, lower real estate and power expenses and a decline in content costs for Prism TV.

Cost of services and products (exclusive of depreciation and amortization) increased by \$429 million, or 6%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The increase in costs of services and products (exclusive of depreciation and amortization) was attributable to the inclusion of \$690 million in post-acquisition Legacy Level 3 costs (net of intercompany eliminations) in our consolidated costs of services and products (exclusive of depreciation and amortization). Costs of services and products (exclusive of depreciation and amortization) for Legacy CenturyLink decreased by \$261 million, or 3%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The decrease in cost of services and products (exclusive of depreciation and amortization) was primarily due to reductions in salaries and wages and employee benefits from lower headcount and healthcare costs, lower real estate and power expenses from the sale of the data centers and colocation business, and reduced customer premises equipment costs due to a decrease in sales of customer premises equipment and USF rates, which were partially offset by increases in facility costs.

Selling, General and Administrative

Selling, general and administrative expenses increased by \$657 million, or 19%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017. The increase in selling, general and administrative expenses was attributable to the inclusion of \$1.1 billion Legacy Level 3 post-acquisition costs (net of intercompany eliminations) in our consolidated selling, general and administrative expenses. Selling, general and administrative expenses decreased by \$444 million, or 14%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017. The decrease was primarily due to (i) reductions in salaries and wages and employee related expenses from lower headcount, (ii) reduced overtime, professional fees, bad debt and marketing expenses and (iii) a loss on sale of data centers in 2017.

Selling, general and administrative expenses increased by \$61 million, or 2%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The increase in selling, general and administrative expenses was primarily attributable to the inclusion of \$253 million in post-acquisition Legacy Level 3 expenses (net of intercompany eliminations) in our consolidated selling, general and administrative expenses. Legacy CenturyLink's selling, general and administrative expenses decreased by \$192 million, or 6%, primarily due to (i) reductions in salaries and wages and employee benefits from lower headcount, (ii) reduced healthcare, external commissions and bad debt expenses and (iii) losses recognized from the sale of our data centers and colocation business and the related failed-sale-leaseback as further described in Note 3— Sale of Data Centers and Colocation Business to our consolidated financial statements in Item 8 of Part II of this report and increases in transaction and integration costs associated with the Level 3 acquisition.

Depreciation and Amortization

The following tables provide detail of our depreciation and amortization expense:

| | Y | Years Ended December 31, | | Increase / | | |
|-------------------------------------|----|--------------------------|---------------------|------------|----------|--|
| | | 2018 | 2017 | (Decrease) | % Change | |
| | | (D | ollars in millions) | | | |
| Depreciation | \$ | 3,339 | 2,710 | 629 | 23% | |
| Amortization | | 1,781 | 1,226 | 555 | 45% | |
| Total depreciation and amortization | \$ | 5,120 | 3,936 | 1,184 | 30% | |
| | | | | | | |

| | Υ | ears Ended De | ecember 31, | Increase / | |
|-------------------------------------|----|---------------|---------------------|------------|----------|
| | | 2017 | 2016 | (Decrease) | % Change |
| | | (D | ollars in millions) | | |
| Depreciation | \$ | 2,710 | 2,691 | 19 | 1% |
| Amortization | | 1,226 | 1,225 | 1 | —% |
| Total depreciation and amortization | \$ | 3,936 | 3,916 | 20 | 1% |

Annual depreciation expense is impacted by several factors, including changes in our depreciable cost basis, changes in our estimates of the remaining economic life of certain network assets, the addition of new plant (including from the acquisition of Level 3) and the divestiture of plant (including from sale of our data centers and colocation business). Depreciation expense increased by \$629 million, or 23%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017, primarily due to the inclusion of \$763 million Legacy Level 3 postacquisition depreciation expense in our consolidated depreciation expense, which was partially offset by lower Legacy CenturyLink depreciation expense. Depreciation expense increased by \$19 million, or 1%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The increase in depreciation expense for the year ended December 31, 2017 was primarily attributable to the inclusion of \$143 million in post-acquisition Legacy Level 3 depreciation expense in our consolidated depreciation expense. Legacy CenturyLink's depreciation expense was lower due to full depreciation and retirement of certain plant previously placed in service. Additionally, we ceased depreciating property, plant and equipment assets of our colocation business when we entered into the agreement to sell that business in late 2016. Absent the sale, we estimate that we would have recorded additional depreciation expense of \$54 million from January 1, 2017 through May 1, 2017 related to the conveyed property. These decreases were partially offset by an increase in depreciation expense attributable to new plant placed in service since January 1, 2016. As a result of not meeting sale-leaseback accounting requirements, prior to January 1, 2019 we are deemed under GAAP to still own certain real estate assets sold to Cyxtera; therefore, we are required to reflect a portion of the real estate assets on our consolidated balance sheet and depreciate these assets over their useful lives. As further described in Note 3—Sale of Data Centers and Colocation Business, of the \$91 million increase in depreciation expense on these real estate assets, \$44 million is not expected to recur in future periods.

Amortization expense increased by \$555 million, or 45%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017 and increased by \$1 million, or less than 1%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The increase in amortization expense for the year ended December 31, 2018 and December 31, 2017 was primarily attributable to the inclusion of \$659 million and \$139 million, respectively, in post-acquisition Legacy Level 3 amortization expense in our consolidated amortization expense. Legacy CenturyLink's amortization expense was lower for both periods primarily due to the use of accelerated amortization for a portion of our customer relationship assets and our entry into an agreement to sell our data centers and colocation business. The effect of using an accelerated amortization method results in an incremental decline in expense each period as the intangible assets amortize. In 2017, we ceased amortizing the intangible assets of our colocation business when we entered into the agreement to sell that business. Absent the sale, we estimate that we would have recorded additional amortization expense of \$13 million from January 1, 2017 through May 1, 2017, related to the conveyed intangible assets. In addition, amortization of capitalized software was lower in both periods due to software becoming fully amortized faster than new software was acquired or developed.

Impairment of Goodwill

At October 31, 2018, we estimated the fair value of our five reporting units, which we determined to be consumer, medium and small business, enterprise, international and global accounts and wholesale and indirect, by considering both a market approach and a discounted cash flow method. The market approach method includes the use of comparable multiples of publicly traded companies whose services are comparable to ours. The discounted cash flow method is based on the present value of projected cash flows and a terminal value, which represents the expected normalized cash flows of the reporting units beyond the cash flows from the discrete projection period. We reconciled the estimated fair values of the reporting units to our market capitalization as of October 31, 2018 and concluded that the indicated control premium of approximately 0.1% was reasonable based on recent transactions in the market place.

As of October 31, 2018, based on our assessment performed with respect to these reporting units as described above, we concluded that the estimated fair value of our consumer reporting unit was less than our carrying value of equity for such unit by approximately \$2.7 billion. As a result, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill assigned to our consumer segment during the fourth quarter of 2018. After the impairment charge described above, the estimated fair value of equity for our consumer segment equals the carrying value of equity for such segment.

Other Consolidated Results

The following tables summarize our total other expense, net and income tax expense (benefit):

| | ١ | ears Ended De | cember 31, | Increase / | |
|------------------------------------|----------|----------------|--|--------------------------|----------|
| | | 2018 | 2017 | (Decrease) | % Change |
| | | (Do | ollars in millions) | | |
| Interest expense | \$ | (2,177) | (1,481) | 696 | 47% |
| Other income, net | | 44 | 12 | 32 | nm |
| Total other expense, net | \$ | (2,133) | (1,469) | 664 | 45% |
| Income tax expense (benefit) | \$ | 170 | (849) | 1,019 | nm |
| | | | | | |
| | | | | | |
| | Y | ears Ended De | cember 31, | Increase / | |
| | <u> </u> | ears Ended Dec | cember 31, 2016 | Increase / (Decrease) | % Change |
| | Y | 2017 | | | % Change |
| Interest expense | Y \$ | 2017 | 2016 | | % Change |
| Interest expense Other income, net | | 2017 (Do | 2016 ollars in millions) | (Decrease) | |
| · | | 2017 (Do | 2016 Dilars in millions) (1,318) | (Decrease) | 12% |

nm Percentages greater than 200% and comparisons between positive and negative values or to/from zero values are considered not meaningful.

Interest Expense

Interest expense increased by \$696 million , or 47% , for the year ended December 31, 2018 as compared to the year ended December 31, 2017 . The increase in interest expense was primarily due to our issuance and assumption of debt in conjunction with the acquisition of Level 3. Interest expense increased by \$163 million , or 12% , for the year ended December 31, 2017 as compared to the year ended December 31, 2016 . The increase in interest expense was primarily due to (i) the issuance of \$7.9 billion of term loans in 2017 for the purpose of providing funding for the Level 3 acquisition, (ii) the assumption of Level 3's debt upon the consummation of the acquisition of Level 3, which accounted for \$80 million in post-acquisition interest expense and (iii) the recognition of imputed interest expense resulting from the failed-sale-leaseback as further described in Note 3—Sale of Data Centers and Colocation Business .

Other Income, Net

Other income, net reflects certain items not directly related to our core operations, including our share of income from partnerships we do not control, interest income, gains and losses from non-operating asset dispositions, foreign currency gains and losses and components of net periodic pension and postretirement benefit costs. Other income, net increased by \$32 million, for the year ended December 31, 2018 as compared to the year ended December 31, 2017. This increase in other income, net was primarily due to a decrease in pension service charges in 2018. Other income, net increased by \$7 million, or 140%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. This increase in other income, net was primarily due to a reduction in the loss on early retirement of debt, an increase in interest income from the \$6 billion Term Loan B funds held in escrow and income generated from our services agreements with Cyxtera, which was substantially offset by a lower expected return on assets in 2017 for our pension and post-retirement plans. The expected return on assets for our pension and post-retirement plans was lower in 2017 as compared to 2016, which resulted in us recording pension and post-retirement expense in 2017 as compared to recording pension and post-retirement income in 2016.

Income Tax Expense (Benefit)

The enactment of the Tax Cuts and Jobs Act in December 2017 resulted in a re-measurement of our deferred tax assets and liabilities at the new federal corporate tax rate of 21%. The re-measurement resulted in a tax benefit of approximately \$1.1 billion recorded in the fourth quarter of 2017, which was the predominant factor contributing to our recognition of an \$849 million income tax benefit for 2017, versus income tax expense of \$394 million in the prior year. For the years ended December 31, 2018, 2017 and 2016, our effective income tax rate was (10.9)%, (157.2)% and 38.6%, respectively. The effective tax rate for the year ended December 31, 2018 reflects a \$572 million unfavorable impact of the non-deductible goodwill impairment as well as the current year unfavorable impact of purchase price accounting adjustments resulting from the Level 3 acquisition and from the tax reform impact of those adjustments of \$92 million. The 2018 unfavorable impacts are partially offset by the tax benefit of a 2017 tax loss carryback to 2016 of \$142 million. The effective tax rate for the year ended December 31, 2017 reflects the benefit from the remeasurement of deferred taxes as noted above, a \$27 million tax expense related to the sale of our data centers and colocation business and a \$32 million tax impact of non-deductible transaction costs related to the Level 3 acquisition. The effective tax rate for the year ended December 31, 2016 reflects a tax impact of \$18 million from an intercompany dividend payment from one of our foreign subsidiaries to its domestic parent company that was made as part of our corporate restructuring in preparation for the sale of our colocation business. See Note 14—Income Taxes' below for additional information.

Segment Results

The results for our business and consumer segments are summarized below for the years ended December 31, 2018, 2017 and 2016:

| | Years Ended December 31, | | | | |
|-------------------------------|--------------------------|---------------------|--------|--|--|
| | 2018 | 2017 | 2016 | | |
| | (Do | ollars in millions) | | | |
| Total segment revenue | \$ 22,720 | 16,924 | 16,766 | | |
| Total segment expenses | 12,269 | 9,390 | 9,081 | | |
| Total segment adjusted EBITDA | \$ 10,451 | 7,534 | 7,685 | | |
| Total margin percentage | 46% | 45% | 46% | | |
| Business segment: | | | | | |
| Revenue | \$ 17,349 | 11,220 | 10,704 | | |
| Expenses | 10,076 | 6,847 | 6,391 | | |
| Adjusted EBITDA | \$ 7,273 | 4,373 | 4,313 | | |
| Margin percentage | 42% | 39% | 40% | | |
| Consumer segment: | | | | | |
| Revenue | \$ 5,371 | 5,704 | 6,062 | | |
| Expenses | 2,193 | 2,543 | 2,690 | | |
| Adjusted EBITDA | \$ 3,178 | 3,161 | 3,372 | | |
| Margin percentage | 59% | 55% | 56% | | |

Products and Services

In connection with our acquisition of Level 3 on November 1, 2017, we revised the way we categorize our products and services and now report our related revenue under the following categories: IP and data services, transport and infrastructure, voice and collaboration, IT and managed services and regulatory revenue. As a result of organization changes made in January 2019, we are making certain changes to revenue reporting categories in 2019 in our business and consumer segments to align with how the business is being managed as discussed further below.

We offer our customers the ability to bundle together several products and services. We believe our customers value the convenience and price discounts associated with receiving multiple services through a single company.

Business Segment

The operations of our business segment have been impacted by several significant trends, including those described below:

Revenue. Our mix of total business segment revenue continues to migrate from traditional wireline voice services to newer, lower cost and more technologically advanced products and services as our small, medium and enterprise business, wholesale and government customers increasingly demand integrated data, broadband, hosting and voice services. Our Ethernet-based services in the wholesale market face competition from cable companies and competitive fiber-based telecommunications providers. We anticipate continued pricing pressure for our colocation services as our competitors continue to expand their enterprise colocation operations. In recent years, our competitors, as well as several large, diversified technology companies, have made substantial investments in cloud computing. This expansion in competitive cloud computing offerings has led to increased pricing pressure, a migration towards lower-priced cloud-based services and enhanced competition for contracts, and we expect these trends to continue. Customers' demand for new technology has also increased the number of competitors offering services similar to ours. Price compression from each of these above-mentioned competitive pressures has negatively impacted the operating margins of certain business product and service offerings, and we expect this trend to continue. Our traditional wireline products and services revenue have been, and we expect they will continue to be, adversely affected by access line losses and price compression. In particular, our access, local services and long-distance revenue have been, and we expect will continue to be, adversely affected by customer migration to more technologically advanced services, a substantial increase in the use of non-voice communications, industry consolidation and price compression caused by regulation and rate reductions. For example, many of our business segment customers are substituting cable, wireless and Voice over Internet Protocol ("VoIP") services for traditional voice telecommunications services, resulting in continued access revenue loss. Demand for our private line services (including business data services) continues to decline due to our customers' optimization of their networks, industry consolidation and technological migration to higher-speed services. Although our traditional wireline services generally face fewer direct competitors than certain of our newer, lower cost more advanced products and services, customer migration and, to a lesser degree, price compression from competitive pressures have negatively impacted our traditional wireline revenue and the operating margins of these services. We expect this trend to continue. We expect both equipment sales and professional services revenue and the related costs will fluctuate from year to year as this offering tends to be more sensitive than others to changes in the economy and in spending trends of our federal, state and local government customers, many of whom have experienced substantial budget cuts over the past several years, with the possibility of additional future budget cuts.

Expenses. Our operating costs also impact the operating margins of all of our above-mentioned services, but to a lesser extent than price compression and customer disconnects. These operating costs include employee costs, sales commissions, software costs on selected services, installation costs and third-party facility costs. We believe increases in operating costs have generally had a greater impact on the operating margins of some of our newer, more technologically advanced services as compared to our traditional wireline services, principally because those newer services rely more heavily upon the above-listed support functions. Operating costs, such as installation costs and third-party facility costs, have also negatively impacted the operating margins of our traditional wireline products and services, but to a lesser extent than customer loss, customer migration and price compression.

Operating efficiencies. We continue to evaluate our segment operating structure and focus. This involves balancing our workforce in response to our workload requirements, productivity improvements and changes in industry, competitive, technological and regulatory conditions, while achieving operational efficiencies and improving our processes through automation. However, our ongoing efforts to increase revenue will continue to require that we incur higher costs in some areas. We also expect our business segment to benefit indirectly from enhanced efficiencies in our company-wide network operations.

The following tables summarize the results of operations from our business segment:

| | Business Segment | | | | |
|----------------------------------|------------------------------|---------------------|------------|----------|--|
| | Years Ended December 31, | | Increase / | | |
| | 2018 | 2017 | (Decrease) | % Change | |
| | (D | ollars in millions) | | | |
| Segment revenue: | | | | | |
| IP and Data Services (1) | \$ 6,971 | 3,682 | 3,289 | 89 % | |
| Transport and Infrastructure (2) | 5,356 | 3,569 | 1,787 | 50 % | |
| Voice and Collaboration (3) | 4,401 | 3,317 | 1,084 | 33 % | |
| IT and Managed Services (4) | 621 | 652 | (31) | (5)% | |
| Total segment revenue | 17,349 | 11,220 | 6,129 | 55 % | |
| Segment expenses: | | | | | |
| Total expenses | 10,076 | 6,847 | 3,229 | 47 % | |
| Segment adjusted EBITDA | \$ 7,273 | 4,373 | 2,900 | 66 % | |
| Segment margin percentage | 42% | 39% | | | |

| | Business Segment | | | | |
|----------------------------------|------------------------------|---------------------|------------|----------|--|
| | Years Ended December 31, | | Increase / | | |
| | 2017 | 2016 | (Decrease) | % Change | |
| | (D | ollars in millions) | | | |
| Segment revenue: | | | | | |
| IP and Data Services (1) | \$ 3,682 | 2,957 | 725 | 25 % | |
| Transport and Infrastructure (2) | 3,569 | 3,807 | (238) | (6)% | |
| Voice and Collaboration (3) | 3,317 | 3,299 | 18 | 1 % | |
| IT and Managed Services (4) | 652 | 641 | 11 | 2 % | |
| Total segment revenue | 11,220 | 10,704 | 516 | 5 % | |
| Segment expenses: | | | | | |
| Total expenses | 6,847 | 6,391 | 456 | 7 % | |
| Segment adjusted EBITDA | \$ 4,373 | 4,313 | 60 | 1 % | |
| Segment margin percentage | 39% | 40% | | | |

⁽¹⁾ Includes primarily VPN data network, Ethernet, IP and ancillary revenue.

⁽²⁾ Includes primarily broadband, private line (including business data services), colocation and data centers, wavelength and ancillary revenue.

⁽³⁾ Includes local, long-distance and other ancillary revenue.

⁽⁴⁾ Includes IT services and managed services revenue.

Segment Revenue

Business segment revenue increased by \$6.1 billion, or 55%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017 due to the inclusion of \$6.7 billion in Legacy Level 3 post-acquisition business segment revenue in our consolidated business segment revenue. Business segment revenue for Legacy CenturyLink decreased by \$571 million for the year ended December 31, 2018 compared to the year ended December 31, 2017 . The decline in Legacy CenturyLink business segment revenue for year ended December 31, 2018 is primarily due to lower transport and infrastructure and voice and collaboration revenue services. The transport and infrastructure decrease was primarily due to reductions in private line (including business data services) revenue while the decrease in voice and collaboration was due to continued decreases in revenue from our traditional voice telecommunications services.

Business segment revenue increased by \$516 million for the year ended December 31, 2017 as compared to the year ended December 31, 2016 due to the inclusion of \$1.4 billion in post-acquisition Legacy Level 3 business segment revenue in our consolidated business segment revenue. Business segment revenue for Legacy CenturyLink decreased by \$874 million for the year ended December 31, 2017 compared to the year ended December 31, 2016. The decline in business segment revenue for year ended December 31, 2017 is attributable to a reduction in access lines and lower volumes of long-distance and access services resulting from the competitive and technological factors noted above and to reductions in the volume of private line (including business data services) services. Our business segment revenue for the year ended December 31, 2017 was also impacted by the May 1, 2017 sale of our data centers and colocation business, which resulted in a reduction of colocation revenue of \$396 million for the year ended December 31, 2017 as compared to the prior year period.

The following tables summarize the results of operations from our business segment by customer sales channel:

| Years Ended December 31, | | Increase / | | |
|--------------------------|-----------------------|------------|---------|--|
| 2018 2017 | | (Decrease) | %Change | |
| | (Dollars in millions) | | | |
| | | | | |
| 3,429 | 3,114 | 315 | 10 % | |

| Segment revenue by customer sales channel: | | | | |
|--|--------------|--------|-------|--------|
| Medium and small business | \$ 3,429 | 3,114 | 315 | 10 % |
| Enterprise | 5,217 | 3,269 | 1,948 | 60 % |
| International and global accounts | 3,657 | 1,377 | 2,280 | 166 % |
| Wholesale and indirect | 5,046 | 3,289 | 1,757 | 53 % |
| Colocation | _ | 171 | (171) | (100)% |
| Total segment revenue by customer sales channel: | \$ 17 349 | 11 220 | 6 129 | 55 % |

| | | _ | |
|-------|------|-----|------|
| Busir | าess | Sea | ment |

Business Segment

| | · · · · · · · · · · · · · · · · · · · | | | | |
|--|---------------------------------------|-----------------------|------------|---------|--|
| | Years Ended D | ecember 31, | Increase / | | |
| · | 2017 | 2017 2016 | | %Change | |
| | | (Dollars in millions) | | | |
| Segment revenue by customer sales channel: | | | | | |
| Medium and small business | 3,114 | 3,127 | (13) | — % | |
| Enterprise | 3,269 | 2,841 | 428 | 15 % | |
| International and global accounts | 1,377 | 973 | 404 | 42 % | |
| Wholesale and indirect | 3,289 | 3,232 | 57 | 2 % | |
| Colocation | 171 | 531 | (360) | (68)% | |
| Total segment revenue by customer sales channel: | \$ 11,220 | 10,704 | 516 | 5 % | |
| | | | | | |

In 2019, we intend to make the following reporting changes for the business segment. The indirect channel, which primarily targets small to medium-sized enterprises, will move from the Wholesale and Indirect Business unit to the Small and Medium Business unit. The company is also moving State and Local Government customers from the Small and Medium Business unit to the Enterprise Business unit to gain efficiencies by managing all government customers in a single organization.

In addition, the new reporting structure reflects changes made to customer assignments between all five customer-facing business units.

Segment Expenses

Business segment expenses increased by \$3.2 billion, or 47%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017, primarily due to the inclusion of \$3.7 billion in Legacy Level 3 post-acquisition business segment expenses in our consolidated business segment expenses. Business segment expenses for Legacy CenturyLink decreased by \$504 million for the year ended December 31, 2018 compared to the year ended December 31, 2017. The decline in business segment expenses for Legacy CenturyLink was primarily due to decreased salaries and wages from lower headcount, a reduction of expenses generated by our colocation business and decreased Level 3 pre-acquisition expenses paid by CenturyLink during the first 10 months of 2017. Business segment expenses for Legacy CenturyLink increased by \$456 million, or 7%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016 primarily due to the inclusion of \$749 million in post-acquisition Legacy Level 3 business segment expenses in our consolidated business segment expenses. Business segment expenses for Legacy CenturyLink decreased by \$293 million for the year ended December 31, 2017 as compared to the year ended December 31, 2016 primarily due to decreases in salaries and wages and employee benefits from lower headcount, real estate and power costs due to the sale of the data centers and colocation business, marketing and advertising expenses and network expense. These decreases were partially offset by an increase in facility costs.

Segment Adjusted EBITDA

Business segment adjusted EBITDA increased by \$2.9 billion, or 66%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017 due to the inclusion of Legacy Level 3 post-acquisition business segment adjusted EBITDA of \$3.0 billion. Business segment adjusted EBITDA increased by \$60 million, or 1%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016 primarily due to the inclusion of \$641 million in post-acquisition Legacy Level 3 business segment adjusted EBITDA. The decrease of \$581 million in Legacy CenturyLink business segment adjusted EBITDA for the year ended December 31, 2017 was due to the loss of customers, lower service volumes and the loss of income generated by our colocation business.

Consumer Segment

The operations of our consumer segment have been impacted by several significant trends, including those described below:

Revenue. In order to remain competitive and attract additional residential broadband subscribers, we believe it is important to continually increase our broadband network's scope and connection speeds. As a result, we continue to invest in our broadband network, which allows for the delivery of higher-speed broadband services to a greater number of customers. We compete in a maturing broadband market in which most consumers already have broadband services and growth rates in new subscribers have slowed or declined. Moreover, as described further in Item 1A of Part I of this report, certain of our competitors continue to provide broadband services at generally higher average transmission speeds than ours or through advanced wireless data service offerings, both of which we believe have impacted the competitiveness of certain of our broadband offerings in certain of our markets. Our voice revenue has been, and we expect they will continue to be, adversely affected by access line losses and lower long-distance voice service volumes. Intense competition and product substitution continue to drive our access line losses. For example, many consumers are substituting cable and wireless voice services and electronic mail, texting and social networking non-voice services for traditional voice telecommunications services. We expect our video revenue to continue to decline, particularly due to our decision to discontinue active marketing of our facilities-based video services in light of competitive pressures and escalating content costs. The demand for new technology has increased the number of competitors offering services similar to ours. Price compression and new technology from our competitors have negatively impacted the operating margins of our newer, more technologically advanced products and services. We expect that these factors will continue to negatively impact our business. As a result of the expected loss of higher margin services associated with access lines, we continue to offer our customers service bundling and other product promotions to help mitigate this trend, as described below. Customer migration and price compression from competitive pressures have not only negatively impacted our traditional wireline services revenue, but they have also negatively impacted the operating margins of these services and we expect this trend to continue.

Additionally, we plan to make changes to the service type reporting to make it easier for investors to evaluate changes to consumer product revenue.

We plan to report consumer revenue in the following categories: broadband; voice; regulatory (includes CAF II and other support funds); and other, which includes retail video and other miscellaneous services.

Expenses. Operating costs also impact the operating margins of these services. These operating costs include employee costs, marketing and advertising expenses, sales commissions, TV content costs and installation costs. We believe increases in operating costs have generally had a greater impact on our operating margins of our newer, more technologically advanced products and services as compared to our traditional wireline services, principally because our newer, more technologically advanced products and services rely more heavily upon the above-listed operating expenses. Operating costs, such as installation costs and facility costs, have also negatively impacted the operating margins of our traditional wireline products and services, but to a lesser extent than customer migration and price compression. Operating costs also tend to impact our traditional wireline products and services margins to a lesser extent than our newer, more technologically advanced products and services as noted above.

Service bundling and product promotions. We offer our customers the ability to bundle multiple products and services. These customers can bundle broadband services with other services such as local voice, video and long-distance. While we believe our bundled service offerings can help retain customers, they also tend to lower our profit margins in the consumer segment due to the related discounts; and

Operating efficiencies. We continue to evaluate our segment operating structure and focus. This involves balancing our workforce in response to our workload requirements, productivity improvements and changes in industry, competitive, technological and regulatory conditions. We also expect our consumer segment to benefit indirectly from enhanced efficiencies in our company-wide network operations.

The following tables summarize the results of operations from our consumer segment:

| | | Consumer Segment | | | | |
|----------------------------------|-----------|--------------------------|-----------------------|------------|----------|--|
| | <u></u> - | Years Ended December 31, | | Increase / | | |
| | <u></u> - | 2018 | 2017 | (Decrease) | % Change | |
| | <u></u> - | | (Dollars in millions) | | | |
| Segment revenue: | | | | | | |
| IP and Data Services (1) | \$ | 308 | 401 | (93) | (23)% | |
| Transport and Infrastructure (2) | | 2,892 | 2,776 | 116 | 4 % | |
| Voice and Collaboration (3) | | 2,171 | 2,527 | (356) | (14)% | |
| Total segment revenue | _ | 5,371 | 5,704 | (333) | (6)% | |
| Segment expenses: | | | | | | |
| Total expenses | | 2,193 | 2,543 | (350) | (14)% | |
| Segment adjusted EBITDA | \$ | 3,178 | 3,161 | 17 | 1 % | |
| Segment margin percentage | | 59% | 55% | | | |

| | Consumer Segment | | | | |
|----------------------------------|--------------------------|-------|---------------------|------------|----------|
| | Years Ended December 31, | | Increase / | | |
| | | 2017 | 2016 | (Decrease) | % Change |
| | | (D | ollars in millions) | _ | _ |
| Segment revenue: | | | | | |
| IP and Data Services (1) | \$ | 401 | 461 | (60) | (13)% |
| Transport and Infrastructure (2) | | 2,776 | 2,776 | _ | — % |
| Voice and Collaboration (3) | | 2,527 | 2,825 | (298) | (11)% |
| Total segment revenue | | 5,704 | 6,062 | (358) | (6)% |
| Segment expenses: | | | | | |
| Total expenses | | 2,543 | 2,690 | (147) | (5)% |
| Segment adjusted EBITDA | \$ | 3,161 | 3,372 | (211) | (6)% |
| Segment margin percentage | | 55% | 56% | | |

⁽¹⁾ Includes retail video revenue (including our facilities-based video revenue).

Segment Revenue

Consumer segment revenue decreased by \$333 million, or 6%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017 and by \$358 million, or 6%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The decline in consumer segment services revenue during each year was primarily due to lower local and long-distance voice service volumes associated with access line losses resulting from the competitive and technological factors noted above and a decrease in the number of Prism TV customers.

⁽²⁾ Includes primarily broadband and equipment sales and professional services revenue.

⁽³⁾ Includes local, long-distance and other ancillary revenue.

Segment Expenses

Consumer segment expenses decreased by \$350 million, or 14%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017. The decrease in our consumer segment expenses was primarily due to lower salaries and wages from a reduction in headcount, decreases in marketing expenses and lower network expense in response to a smaller customer base. Consumer segment expenses decreased by \$147 million, or 5%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. This decrease in our consumer segment expenses was primarily due to decreases in salaries and wages and employee benefits from lower headcount, external commissions and USF surcharges from rate decreases, which were partially offset by increases in marketing and advertising expenses.

Segment Adjusted EBITDA

Consumer segment adjusted EBITDA increased by \$17 million, or 1%, for the year ended December 31, 2018 as compared to the year ended December 31, 2017. The increase in our consumer segment adjusted EBITDA was due predominantly to decreased segment expenses related to reduction in headcount, lower marketing and network expenses. Consumer segment adjusted EBITDA decreased \$211 million, or 6%, for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The decline in our consumer segment adjusted EBITDA was primarily due to loss of customers and increases in the costs associated with Prism TV.

Critical Accounting Policies and Estimates

Our consolidated financial statements are prepared in accordance with accounting principles that are generally accepted in the United States. The preparation of these consolidated financial statements requires management to make estimates and assumptions that affect the reported amounts of our assets, liabilities, revenue and expenses. We have identified certain policies and estimates as critical to our business operations and the understanding of our past or present results of operations related to (i) business combinations, (ii) goodwill, customer relationships and other intangible assets; (iii) property, plant and equipment; (iv) pension and post-retirement benefits; (v) loss contingencies and litigation reserves and (vi) income taxes. These policies and estimates are considered critical because they had a material impact, or they have the potential to have a material impact, on our consolidated financial statements and because they require us to make significant judgments, assumptions or estimates. We believe that the estimates, judgments and assumptions made when accounting for the items described below were reasonable, based on information available at the time they were made. However, there can be no assurance that actual results will not differ from those estimates.

Business Combination

We have accounted for our acquisition of Level 3 on November 1, 2017, under the acquisition method of accounting, whereby the tangible and separately identifiable intangible assets acquired and liabilities assumed are recognized at their fair values at the acquisition date. The portion of the purchase price in excess of the fair value of the net tangible and separately identifiable intangible assets acquired represents goodwill. The fair value and resulting assignment of the purchase price related to our acquisition of Level 3 involved significant estimates and judgments by our management. In arriving at the fair values of assets acquired and liabilities assumed, we considered the following generally accepted valuation approaches: the cost approach, income approach and market approach. Our estimates also included assumptions about projected growth rates, cost of capital, effective tax rates, tax amortization periods, technology life cycles, customer attrition rates, the regulatory and legal environment and industry and economic trends. For additional information about our acquisition of Level 3, see Note 2—Acquisition of Level 3 to our consolidated financial statements in Item 8 of Part II of this report.

Goodwill, Customer Relationships and Other Intangible Assets

We amortize customer relationships primarily over an estimated life of 7 to 15 years, using either the sum-of-years-digits or the straight-line methods, depending on the customer retention patterns for the type of customer at the companies we acquire. We amortize capitalized software using the straight-line method over estimated lives ranging up to 7 years, except for approximately \$237 million of our capitalized software costs, which represents costs to develop an integrated billing and customer care system which is amortized using the straight-line method over a 20-year period. We annually review the estimated lives and methods used to amortize our other intangible assets, primarily capitalized software. The amount of future amortization expense may differ materially from current amounts, depending on the results of our annual reviews.

Our goodwill was derived from numerous acquisitions where the purchase price exceeded the fair value of the net assets acquired.

We are required to reassign goodwill to reporting units each time we reorganize our internal reporting structure which causes a change in the composition of our reporting units. We assign goodwill to the reporting units using a relative fair value approach. We utilize the trailing twelve-months earnings before interest, taxes, depreciation and amortization as our allocation methodology as we believe that it represents a reasonable proxy for the fair value of the operations being reorganized. The use of other fair value assignment methods could result in materially different results. For additional information on our segments, see Note 15—Segment Information to our consolidated financial statements in Item 8 of Part II of this report.

We are required to assess goodwill for impairment at least annually, or more frequently, if an event occurs or circumstances change that would indicate an impairment may have occurred. We are required to write-down the value of goodwill in periods in which the carrying value of equity exceeds the fair value of equity. Our reporting units are not discrete legal entities with discrete full financial statements. Our assets and liabilities are employed in and relate to the operations of our reporting units. Therefore, the equity carrying value and future cash flows must be estimated each time a goodwill impairment analysis is performed on a reporting unit. As a result, our assets, liabilities and cash flows are assigned to reporting units using reasonable and consistent allocation methodologies. Certain estimates, judgments and assumptions are required to perform these assignments. We believe these estimates, judgments and assumptions to be reasonable, but changes in any of these can significantly affect each reporting unit's equity carrying value and future cash flows utilized for our goodwill impairment test. We consider both a market approach and a discounted cash flow method which are weighted equally. The market approach method includes the use of comparable multiples of publicly traded companies whose services are comparable to ours and the discounted cash flow method uses a market participant's weighted average cost of capital. Our annual assessment date for testing goodwill impairment is October 31.

As of October 31, 2018, we assessed goodwill for impairment for our five reporting units, which we determined to be consumer, medium and small business, enterprise, international and global accounts, and wholesale and indirect. We determined that the carrying value of our consumer reporting unit's equity was substantially above its estimated fair value of equity by approximately \$2.7 billion. As a result, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill assigned to our consumer reporting unit during the fourth quarter of 2018.

As of October 31, 2018, based on our assessment performed with respect to our four reporting units included in our business segment, the estimated fair value of our equity exceeded our carrying value of equity for our medium and small business, enterprise, international and global accounts and wholesale and indirect by 2%, 11%, 30% and 5%, respectively. After the impairment charge described above, the estimated fair value of equity for our consumer reporting unit equals the carrying value of equity for such unit.

For additional information on our goodwill balances by segment, see Note 4—Goodwill, Customer Relationships and Other Intangible Assets to our consolidated financial statements in Item 8 of Part II of this report.

As a result of the reporting unit changes we made in January 2019, we intend to perform a goodwill impairment analysis during the first quarter of 2019. We may also be required to assess our goodwill for impairment before our next required assessment date of October 31, 2019 under certain circumstances, including any failure to meet our forecasted future operating results or any significant increases in our weighted average cost of capital. In addition, we cannot assure you that adverse conditions will not trigger future goodwill impairment assessments or impairment charges. A number of factors, many of which we cannot control, could affect our financial condition, operating results and business prospects and could cause our actual results to differ from the estimates and assumptions we employed in our goodwill impairment assessment. These factors include, but are not limited to, (i) weakening in the overall economy or in any of the markets in which we operate; (ii) a significant decline in our stock price and resulting market capitalization as a result of an adverse change to our overall business operations; (iii) changes in the discount rate we use in our testing; (iv) successful efforts by our competitors to gain market share in our markets; (v) adverse changes as a result of regulatory or legislative actions; (vi) a significant adverse change in our legal affairs or in the overall business climate; and (vii) recognition of a goodwill impairment loss in the financial statements of one or more of our subsidiaries that are a component of our segments. For additional information, see "Risk Factors" in Item 1A of Part I of this report. We will continue to monitor certain events that impact our operations to determine if an interim assessment of goodwill impairment should be performed prior to the next required assessment date of October 31, 2019.

Property, Plant and Equipment

Property, plant and equipment acquired in connection with our acquisitions was recorded based on its estimated fair value as of its acquisition date, plus the estimated value of any associated legally or contractually required asset retirement obligation. Purchased and constructed property, plant and equipment is recorded at cost, plus the estimated value of any associated legally or contractually required asset retirement obligation. Renewals and betterments of plant and equipment are capitalized while repairs, as well as renewals of minor items, are charged to operating expense. Depreciation of property, plant and equipment is provided on the straight-line method specific unit or group method using class or overall group rates and specific asset life. The group method provides for the recognition of the remaining net investment, less anticipated net salvage value, over the remaining useful life of the assets. This method requires the periodic revision of depreciation rates.

Normal retirements of property, plant and equipment are charged against accumulated depreciation, with no gain or loss recognized. We depreciate such property on the straight-line method over estimated service lives ranging from 3 to 45 years.

We perform annual internal reviews to evaluate the reasonableness of the depreciable lives for our property, plant and equipment. Our reviews utilize models that take into account actual usage, physical wear and tear, replacement history, assumptions about technology evolution and, in certain instances, actuarially determined probabilities to estimate the remaining life of our asset base.

Due to rapid changes in technology and the competitive environment, determining the estimated economic life of telecommunications plant and equipment requires a significant amount of judgment. We regularly review data on utilization of equipment, asset retirements and salvage values to determine adjustments to our depreciation rates. The effect of a hypothetical one year increase or decrease in the estimated remaining useful lives of our property, plant and equipment would have decreased depreciation expense by approximately \$410 million annually or increased depreciation expense by approximately \$530 million annually, respectively.

Pension and Post-retirement Benefits

We sponsor a noncontributory qualified defined benefit pension plan (referred to as our qualified pension plan) for a substantial portion of our employees. In addition to this tax-qualified pension plan, we also maintain several non-qualified pension plans for certain eligible highly compensated employees. We also maintain post-retirement benefit plans that provide health care and life insurance benefits for certain eligible retirees. On November 1, 2017, we assumed Level 3's pension and post-retirement plans, and certain obligations associated with these plans. Due to the insignificant impact of these plans on our consolidated financial statements, we have excluded them from the following pension and post-retirement benefits disclosures for 2018 and 2017.

In 2018, approximately 55% of the qualified pension plan's January 1, 2018 net actuarial loss balance of \$2.9 billion was subject to amortization as a component of net periodic expense over the average remaining service period of participating employees expected to receive benefits, which ranges from 8 to 9 years for the plan. The other 45% of the qualified pension plan's beginning net actuarial loss balance was treated as indefinitely deferred during 2018. The entire beginning net actuarial loss of \$248 million for the post-retirement benefit plans was treated as indefinitely deferred during 2018.

In 2017, approximately 58% of the qualified pension plan's January 1, 2017 net actuarial loss balance of \$3.1 billion was subject to amortization as a component of net periodic expense over the average remaining service period of participating employees expected to receive benefits, which ranges from 9 to 10 years for the plan. The other 42% of the qualified pension plan's beginning net actuarial loss balance was treated as indefinitely deferred during 2017. The entire beginning net actuarial loss of \$137 million for the post-retirement benefit plans was treated as indefinitely deferred during 2017.

In 2016, approximately 53% of the qualified pension plan's January 1, 2016 net actuarial loss balance of \$2.8 billion was subject to amortization as a component of net periodic expense over the average remaining service period of participating employees expected to receive benefits, which ranges from 9 to 10 years for the plan. The other 47% of the qualified pension plan's beginning net actuarial loss balance was treated as indefinitely deferred during 2016. The entire beginning net actuarial loss of \$147 million for the post-retirement benefit plans was treated as indefinitely deferred during 2016.

In computing our pension and post-retirement health care and life insurance benefit obligations, our most significant assumptions are the discount rate and mortality rates. In computing our periodic pension and post-retirement benefit expense, our most significant assumptions are the discount rate and the expected rate of return on plan assets.

The discount rate for each plan is the rate at which we believe we could effectively settle the plan's benefit obligations as of the end of the year. We selected each plan's discount rate based on a cash flow matching analysis using hypothetical yield curves from U.S. corporate bonds rated high quality and projections of the future benefit payments that constitute the projected benefit obligation for the plans. This process establishes the uniform discount rate that produces the same present value of the estimated future benefit payments as is generated by discounting each year's benefit payments by a spot rate applicable to that year. The spot rates used in this process are derived from a yield curve created from yields on the 60th to 90th percentile of U.S. high quality bonds.

Mortality rates help predict the expected life of plan participants and are based on historical demographic studies by the Society of Actuaries ("SOA"). The SOA publishes new mortality rates (mortality tables and projection scales) on a regular basis which reflect updates to projected life expectancies in North America. Historically, we have adopted the new projection tables immediately after publication. In 2017, we adopted the revised morality tables and projection scale released by the SOA, which decreased the projected benefit obligation of our benefit plans by \$113 million . In 2016, we adopted the revised mortality table and projection scale released by the SOA, which decreased the projected benefit obligation of our benefit plans by \$268 million . The change in the projected benefit obligation of our benefit plans was recognized as part of the net actuarial loss and is included in accumulated other comprehensive loss, a portion of which is subject to amortization over the remaining estimated life of plan participants, which was approximately 17 years as of December 31, 2018 .

The expected rate of return on plan assets is the long-term rate of return we expect to earn on the plans' assets in the future, net of administrative expenses paid from plan assets. The rate of return is determined by the strategic allocation of plan assets and the long-term risk and return forecast for each asset class. The forecasts for each asset class are generated primarily from an analysis of the long-term expectations of various third-party investment management organizations to which we then add a factor of 50 basis points to reflect the benefit we expect to result from our active management of the assets. The expected rate of return on plan assets is reviewed annually and revised, as necessary, to reflect changes in the financial markets and our investment strategy.

To compute the expected return on pension and post-retirement benefit plan assets, we apply an expected rate of return to the fair value of the pension plan assets and to the fair value of the post-retirement benefit plan assets adjusted for contribution timing and for projected benefit payments to be made from the plan assets. Annual market volatility for these assets (higher or lower than expected return) is reflected in the net actuarial losses.

Changes in any of the above factors could significantly impact operating expenses in the consolidated statements of operations and other comprehensive income (loss) in the consolidated statements of comprehensive income as well as the value of the liability and accumulated other comprehensive loss of stockholders' equity on our consolidated balance sheets. The expected return on plan assets is reflected as a reduction to our pension and post-retirement benefit expense.

Loss Contingencies and Litigation Reserves

We are involved in several material legal proceedings, as described in more detail in Note 17—Commitments, Contingencies and Other Items to our consolidated financial statements in Item 8 of Part II of this report. We periodically assess potential losses in relation to these and other pending or threatened tax and legal matters. For matters not related to income taxes, if a loss is considered probable and the amount can be reasonably estimated, we recognize an expense for the estimated loss. To the extent these estimates are more or less than the actual liability resulting from the resolution of these matters, our earnings will be increased or decreased accordingly. If the differences are material, our consolidated financial statements could be materially impacted.

For matters related to income taxes, if we determine in our judgment that the impact of an uncertain tax position is more likely than not to be sustained upon audit by the relevant taxing authority, then we recognize in our financial statements a benefit for the largest amount that is more likely than not to be sustained. No portion of an uncertain tax position will be recognized if we determine in our judgment that the position has less than a 50% likelihood of being sustained. Though the validity of any tax position is a matter of tax law, the body of statutory, regulatory and interpretive guidance on the application of the law is complex and often ambiguous. Because of this, whether a tax position will ultimately be sustained may be uncertain.

Income Taxes

Our provision for income taxes includes amounts for tax consequences deferred to future periods. We record deferred income tax assets and liabilities reflecting future tax consequences attributable to tax credit carryforwards, differences between the financial statement carrying value of assets and liabilities and the tax basis of those assets and liabilities and tax net operating loss carryforwards, or NOLs. Deferred taxes are computed using enacted tax rates expected to apply in the year in which the differences are expected to affect taxable income. The effect on deferred income tax assets and liabilities of a change in tax rate is recognized in earnings in the period that includes the enactment date.

The measurement of deferred taxes often involves the exercise of considerable judgment related to the realization of tax basis. Our deferred tax assets and liabilities reflect our assessment that tax positions taken in filed tax returns and the resulting tax basis are more likely than not to be sustained if they are audited by taxing authorities. Assessing tax rates that we expect to apply and determining the years when the temporary differences are expected to affect taxable income requires judgment about the future apportionment of our income among the states in which we operate. Any changes in our practices or judgments involved in the measurement of deferred tax assets and liabilities could materially impact our financial condition or results of operations.

In connection with recording deferred income tax assets and liabilities, we establish valuation allowances when necessary to reduce deferred income tax assets to amounts that we believe are more likely than not to be realized. We evaluate our deferred tax assets quarterly to determine whether adjustments to our valuation allowance are appropriate in light of changes in facts or circumstances, such as changes in tax law, interactions with taxing authorities and developments in case law. In making this evaluation, we rely on our recent history of pre-tax earnings. We also rely on our forecasts of future earnings and the nature and timing of future deductions and benefits represented by the deferred tax assets, all which involve the exercise of significant judgment. At December 31, 2018, we established a valuation allowance of \$1.3 billion primarily related to foreign and state NOLs, as it is more likely than not that these NOLs will expire unused. If forecasts of future earnings and the nature and estimated timing of future deductions and benefits change in the future, we may determine that a valuation allowance for certain deferred tax assets is appropriate, which could materially impact our financial condition or results of operations. See Note 14—Income Taxes to our consolidated financial statements in Item 8 of Part II of this report for additional information.

Liquidity and Capital Resources

Overview of Sources and Uses of Cash

We are a holding company that is dependent on the capital resources of our subsidiaries to satisfy our parent company liquidity requirements. Several of our significant operating subsidiaries have borrowed funds either on a standalone basis or as part of a separate restricted group with certain of its subsidiaries or affiliates. The terms of the instruments governing the indebtedness of these borrowers or borrowing groups may restrict our ability to access their accumulated cash. In addition, our ability to access the liquidity of these and other subsidiaries may be limited by tax and legal considerations and other factors.

At December 31, 2018, we held cash and cash equivalents of \$488 million and we had approximately \$1.6 billion of borrowing capacity available under our revolving credit facility. We had approximately \$108 million of cash and cash equivalents outside the United States at December 31, 2018. We currently believe we have the ability to repatriate cash and cash equivalents into the United States without paying or accruing U.S. taxes, other than the possible payment of the Deemed Repatriation Transition Tax discussed elsewhere herein and other limited exceptions. We do not currently intend to repatriate to the United States any of our foreign cash and cash equivalents from operating entities outside of Latin America. We have no material restrictions on our ability to repatriate to the United States foreign cash and cash equivalents.

Our executive officers and our Board of Directors periodically review our sources and potential uses of cash in connection with our annual budgeting process. Generally speaking, our principal funding source is cash from operating activities, and our principal cash requirements include operating expenses, capital expenditures, income taxes, debt repayments, dividends, periodic stock repurchases, periodic pension contributions and other benefits payments. As discussed further below, the amount of cash we paid in 2018 for retiree healthcare benefits increased substantially compared to prior periods. We currently expect that our cash paid for retiree healthcare benefits in 2019 will remain flat.

Based on our current capital allocation objectives, during 2019 we project expending approximately \$3.5 billion to \$3.8 billion (excluding integration capital) of cash for capital investment in property, plant and equipment and approximately \$1.1 billion of cash for dividends on our common stock (based on the assumptions described below under "Dividends"). At December 31, 2018, we had debt maturities of \$400 million, scheduled debt principal payments of \$164 million and capital lease and other fixed payments of \$88 million, each due during 2019. Each of the expenditures is described further below.

We will continue to monitor our future sources and uses of cash, and anticipate that we will make adjustments to our capital allocation strategies when, as and if determined by our Board of Directors. We typically use our revolving credit facility as a source of liquidity for operating activities and our other cash requirements.

For additional information, see "Risk Factors—Risks Affecting Our Liquidity and Capital Resources".

Capital Expenditures

We incur capital expenditures on an ongoing basis in order to enhance and modernize our networks, compete effectively in our markets, expand and improve our service offerings. We evaluate capital expenditure projects based on a variety of factors, including expected strategic impacts (such as forecasted impact on revenue growth, productivity, expenses, service levels and customer retention) and our expected return on investment. The amount of capital investment is influenced by, among other things, demand for our services and products, cash flow generated by operating activities, cash required for other purposes and regulatory considerations (such as our CAF Phase II infrastructure buildout requirements). Based on current circumstances, we estimate that our total capital expenditures for 2019 will be approximately \$3.5 billion to \$3.8 billion, inclusive of CAF Phase II related capital expenditures, but excluding integration capital.

Our capital expenditures continue to be focused on keeping the network operating efficiently and supporting new service developments. For more information on our capital spending, see "Historical Information—Investing Activities" below and Item 1 of Part 1 of this report.

Debt and Other Financing Arrangements

Subject to market conditions, we expect to continue to issue debt securities from time to time in the future to refinance a substantial portion of our maturing debt, including issuing Qwest Corporation and Level 3 Financing, Inc. debt securities to refinance their maturing debt to the extent feasible. The availability, interest rate and other terms of any new borrowings will depend on the ratings assigned by credit rating agencies, among other factors.

Following the closing of our acquisition of Level 3, the rating agencies took action on the ratings of the debt in the table below. Generally, the agencies downgraded ratings of the CenturyLink, Inc. debt from previous levels as they indicated they intended to at the time of the announcement of the transaction. Additionally, immediately following the Level 3 acquisition, Moody's Investors Service, Inc. and Standard and Poor's placed such ratings on negative outlook while Fitch Ratings placed them on stable outlook. As for the Level 3 debt, Moody's Investors Service, Inc. upgraded the unsecured debt and affirmed the rating of the secured debt, with all ratings placed on negative outlook. Standard and Poor's and Fitch Ratings affirmed all previous Level 3 ratings with stable outlook.

On February 14, 2019, Standard and Poor's revised its outlook on all CenturyLink, Inc. credit ratings from negative to stable. As of the date of this report, the credit ratings for the senior unsecured debt of CenturyLink, Inc., Qwest Corporation, Level 3 Parent, LLC and Level 3 Financing, Inc. were as follows:

| Borrower | Moody's Investors Service, Inc. | Standard & Poor's | Fitch Ratings |
|-------------------------|------------------------------------|-------------------|---------------|
| CenturyLink, Inc.: | | | |
| Unsecured | B2 | B+ | BB |
| Secured | Ba3 | BBB- | BB+ |
| | | | |
| Qwest Corporation: | | | |
| Unsecured | Ba2 | BBB- | BB+ |
| | | | |
| Level 3 Parent, LLC: | | | |
| Unsecured | B1 | B+ | BB- |
| | | | |
| Level 3 Financing, Inc. | | | |
| Unsecured | Ba3 | BB | BB |
| Secured | Ba1 | BBB- | BBB- |

Our credit ratings are reviewed and adjusted from time to time by the rating agencies. Any future downgrades of the senior unsecured or secured debt ratings of us or our subsidiaries could impact our access to debt capital or further raise our borrowing costs. See "Risk Factors—Risks Affecting our Liquidity and Capital Resources" in Item 1A of Part I of this report.

Net Operating Loss Carryforwards

As of December 31, 2018, CenturyLink had approximately \$7.3 billion of net operating loss carryforwards. ("NOLs"), which for U.S. federal income tax purposes can be used to offset future taxable income. These NOLs are primarily related to federal NOLs we acquired through the Level 3 acquisition on November 1, 2017, and are subject to limitations under Section 382 of the Internal Revenue Code ("Code") and related U.S. Treasury Department regulations. On February 13, 2019, we entered into a Section 382 rights agreement designed to safeguard our ability to use those NOLs. Assuming that we can continue using these NOLs in the amounts projected, we expect to significantly reduce our federal cash taxes for the next several years. The amounts of our near-term future tax payments will depend upon many factors, including our future earnings and tax circumstances and results of any corporate tax reform. Based on current laws and our current estimates of 2019 earnings, we estimate our cash income tax liability related to 2019 will be approximately \$100 million.

We cannot assure you that we will be able to use these NOL carryforwards fully. See "Risk Factors—Risks Relating to Our November 2017 Acquisition of Level 3—We cannot assure you, whether, when or in what amounts we will be able to use Level 3's net operating loss carryforwards" in Item 1A of Part I of this report.

Dividends

We currently expect to continue our current practice of paying quarterly cash dividends in respect of our common stock subject to our Board of Directors' discretion to modify or terminate this practice at any time and for any reason without prior notice. Following a reduction announced on February 13, 2019, our current quarterly common stock dividend rate is \$0.25 per share, as approved by our Board of Directors, which we believe is a dividend rate per share which enables us to balance our multiple objectives of managing our business, investing in the business, de-leveraging our balance sheet and returning a substantial portion of our cash to our shareholders. Assuming continued payment during 2019 at this rate of \$0.25 per share, our average total dividend paid each quarter would be approximately \$274 million based on our current number of outstanding shares (assuming no increases or decreases in the number of shares, except in connection with the vesting of currently outstanding equity awards). See Risk Factors—Risks Affecting Our Business" in Item 1A of Part I of this report.

Revolving Facilities and Other Debt Instruments

To substantially fund our acquisition of Level 3, on June 19, 2017, one of our affiliates entered into a credit agreement (the "2017 CenturyLink Credit Agreement") providing for \$10.2 billion in senior secured credit facilities, consisting of a new \$2.0 billion revolving credit facility (which replaced our 2012 credit facility upon consummation of the Level 3 acquisition) and \$7.9 billion of term loan facilities, of which approximately \$6.0 billion were funded into escrow on such date, and \$1.945 billion of which were funded upon the closing of the acquisition on November 1, 2017. On November 1, 2017, CenturyLink, Inc., among other things, (i) assumed all rights and obligations under the 2017 CenturyLink Credit Agreement, (ii) borrowed \$400 million under the new \$2.0 billion revolving credit facility and (iii) received \$6.0 billion of Term Loan B loan proceeds from escrow. On January 29, 2018, the 2017 CenturyLink Credit Agreement was amended to increase the borrowing capacity of the new revolving credit facility from \$2.0 billion to \$2.2 billion, and to increase the borrowing capacity under one of the term loan tranches by \$132 million. For additional information, see (i) Note 6—Long-Term Debt and Credit Facilities to our consolidated financial statements in Item 8 of Part II of this report and (ii) our current reports on Form 8-K filed with the SEC on June 20, 2017 and November 1, 2017.

On November 1, 2017, we also amended our uncommitted revolving letter of credit facility to secure the facility and to permit us to draw up to \$225 million of letters of credit thereunder. At December 31, 2018, we had \$97 million of letters of credit outstanding under this facility.

For information on the terms and conditions of other debt instruments of ours and our subsidiaries, including financial and operating covenants, see Note 6—Long-Term Debt and Credit Facilities to our consolidated financial statements in Item 8 of Part II of this report.

Future Contractual Obligations

The following table summarizes our estimated future contractual obligations as of December 31, 2018:

| | | | | | | 2024 and | |
|---|-------------|-------|-------|-------------------|-------|------------|--------|
| | 2019 | 2020 | 2021 | 2022 | 2023 | thereafter | Total |
| | | | I) | Dollars in millio | ns) | | |
| Long-term debt (1)(2) | \$ 607 | 1,189 | 3,115 | 5,283 | 2,096 | 23,503 | 35,793 |
| Interest on long-term debt and capital leases (2) | 2,123 | 2,056 | 1,949 | 1,728 | 1,493 | 12,710 | 22,059 |
| Data centers obligation (3) | 86 | 29 | _ | _ | _ | _ | 115 |
| Operating leases | 675 | 443 | 355 | 279 | 241 | 969 | 2,962 |
| Right-of-way agreements | 157 | 134 | 112 | 120 | 115 | 755 | 1,393 |
| Purchase commitments (4) | 322 | 185 | 140 | 53 | 35 | 186 | 921 |
| Post-retirement benefit obligation ⁽⁵⁾ | 87 | 84 | 80 | 75 | 70 | 538 | 934 |
| Non-qualified pension obligations (5) | 5 | 5 | 5 | 4 | 5 | 18 | 42 |
| Asset retirement obligations | 23 | 29 | 16 | 10 | 14 | 98 | 190 |
| Total future contractual obligations ⁽⁶⁾ | \$ 4,085 | 4,154 | 5,772 | 7,552 | 4,069 | 38,777 | 64,409 |

- (1) Includes current maturities and capital lease obligations, but excludes unamortized discounts and premiums, net, unamortized debt issuance costs and data center obligation.
- (2) Actual principal and interest paid in all years may differ due to future refinancing of outstanding debt or issuance of new debt. Interest on our floating rate debt was calculated for all years using the rates effective at December 31, 2018. See Note 17—Commitments, Contingencies and Other Items to our consolidated financial statements in Item 8 of Part II of this report for additional information regarding the future commitments for capital leases related to our colocation operations.
- (3) Future minimum payments of principal, interest and executory costs less future imputed lease income on certain of the real estate assets associated with the data centers obligation. See Note 3—Sale of Data Centers and Colocation Business to our consolidated financial statements in Item 8 of Part II of this report.
- (4) We have various long-term, non-cancelable purchase commitments for advertising and promotion services, including advertising and marketing at sports arenas and other venues and events. We also have purchase commitments with third-party vendors for operating, installation and maintenance services for facilities. In addition, we have service-related commitments with various vendors for data processing, technical and software support services. Future payments under certain service contracts will vary depending on our actual usage. In the table above, we estimated payments for these service contracts based on estimates of the level of services we expect to receive.
- (5) Reflects only the portion of total obligation that is contractual in nature. See Note 6 below.
- (6) The table is limited solely to contractual payment obligations and does not include:
 - contingent liabilities;
 - our open purchase orders as of December 31, 2018. These purchase orders are generally issued at fair value, and are generally cancelable without penalty;
 - other long-term liabilities, such as accruals for legal matters and other taxes that are not contractual obligations by nature. We cannot determine with any degree of reliability the years in which these liabilities might ultimately settle;
 - cash funding requirements for qualified pension benefits payable to certain eligible current and future retirees. Benefits paid by our qualified pension plan are paid through a trust. Cash funding requirements for this trust are not included in this table as we are not able to reliably estimate required contributions to this trust. Our funding projections are discussed further below;
 - certain post-retirement benefits payable to certain eligible current and future retirees. Not all of our post-retirement benefit obligation amount is a contractual obligation and only the portion that we believe is a contractual obligation is reported in the table. See additional information on our benefits plans in Note 10

 —Employee Benefits to our consolidated financial statements in Item 8 of Part II of this report;
 - contract termination fees. These fees are non-recurring payments, the timing and payment of which, if any, is uncertain. In the ordinary course of business
 and to optimize our cost structure, we enter into contracts with terms greater than one year to use the network facilities of other carriers and to purchase
 other goods and services. Our contracts to use other carriers' network facilities generally have no minimum volume requirements and pricing is based upon
 volumes and usage. In the normal course of business, we do not believe payment of these fees is likely;
 - · service level commitments to our customers, the violation of which typically results in service credits rather than cash payments; and

• potential indemnification obligations to counterparties in certain agreements entered into in the normal course of business. The nature and terms of these arrangements vary.

For information on debt that we assumed or incurred in connection with consummating the Level 3 acquisition, see "Risk Factors" in Item 1A of Part I of this report and Note 6—Long-Term Debt and Credit Facilities to our consolidated financial statements in Item 8 of Part II of this report.

Pension and Post-retirement Benefit Obligations

We are subject to material obligations under our existing defined benefit pension plans and post-retirement benefit plans. At December 31, 2018, the accounting unfunded status of our qualified and non-qualified defined benefit pension plans and qualified post-retirement benefit plans was \$1.6 billion and \$3.0 billion, respectively. See Note 10—Employee Benefits to our consolidated financial statements in Item 8 of Part II of this report for additional information about our pension and post-retirement benefit arrangements.

Benefits paid by our qualified pension plan are paid through a trust that holds all of the plan's assets. Based on current laws and circumstances, we do not expect any contributions to be required for our qualified pension plan during 2019. The amount of required contributions to our qualified pension plan in 2020 and beyond will depend on a variety of factors, most of which are beyond our control, including earnings on plan investments, prevailing interest rates, demographic experience, changes in plan benefits and changes in funding laws and regulations. We occasionally make voluntary contributions in addition to required contributions. We made a voluntary contribution of \$500 million to the trust for our qualified pension plan during 2018. Based on current circumstances, we do not anticipate making a voluntary contribution to the trust for our qualified pension plan in 2019.

Substantially all of our post-retirement health care and life insurance benefits plans are unfunded. Several trusts hold assets that have been used to help cover the health care costs of certain retirees. As of December 31, 2018, assets in the post-retirement trusts had been substantially depleted and had a fair value of only \$18 million (a portion of which was comprised of investments with restricted liquidity), which has significantly limited our ability to continue paying benefits from the trusts. Benefits not paid from the trusts are expected to be paid directly by us with available cash. As described further in Note 10—Employee Benefits to our consolidated financial statements in Item 8 of Part II of this report, aggregate benefits paid by us under these plans (net of participant contributions and direct subsidy receipts) were \$249 million, \$237 million and \$129 million for the years ended December 31, 2018, 2017 and 2016, respectively, while the amounts paid from the trust were \$4 million, \$31 million and \$145 million, respectively. For additional information on our expected future benefits payments for our post-retirement benefit plans, please see Note 10—Employee Benefits to our consolidated financial statements in Item 8 of Part II in this report.

For 2018 and 2019, our expected annual long-term rates of return were 6.5% and 4.0% for the pension plan trust assets and post-retirement plans' trust assets, respectively, based on the assets held and net of expected fees and administrative costs. However, actual returns could be substantially different.

Connect America Fund

As a result of accepting CAF Phase II support payments, we must meet certain specified infrastructure buildout requirements in 33 states over the next several years. In order to meet these specified infrastructure buildout requirements, we may be obligated to make substantial capital expenditures. See "Capital Expenditures" above.

For additional information on the FCC's CAF order and the USF program, see "Business—Regulation" in Item 1 of Part I of this report and see "Risk Factors—Risks Affecting Our Liquidity and Capital Resources" in Item 1A of Part I of this report.

Historical Information

The following tables summarize our consolidated cash flow activities:

| | Years Ended December 31, | | |
|---|--------------------------|---------|------------|
| | 2018 | 2017 | (Decrease) |
| | (D | | |
| Net cash provided by operating activities | \$ 7,032 | 3,878 | 3,154 |
| Net cash used in investing activities | (3,078) | (8,871) | (5,793) |
| Net cash (used in) provided by financing activities | (4,023) | 5,356 | 9,379 |
| | | | |

| | Years Ended De | cember 31, | Increase / |
|---|----------------|---------------------|------------|
| | 2017 | 2016 | (Decrease) |
| | (D | ollars in millions) | |
| Net cash provided by operating activities | \$ 3,878 | 4,608 | (730) |
| Net cash used in investing activities | (8,871) | (2,994) | 5,877 |
| Net cash provided by (used in) financing activities | 5,356 | (1,518) | (6,874) |

Operating Activities

Net cash provided by operating activities increased by \$3.2 billion for the year ended December 31, 2018 as compared to the year ended December 31, 2017 primarily due to \$2.4 billion in cash generated by Level 3 in addition to a positive variance in net (loss) income after adjusting for non-cash items for impairment of goodwill and other assets and depreciation and deferred income taxes offset with a reduction in retirement benefits. Net cash provided by operating activities decreased by \$730 million for the year ended December 31, 2017 as compared to the year ended December 31, 2016 primarily due to a negative variance in net income adjusted for non-cash items and from negative variances in the changes in accounts payable, other current assets and liabilities, net and other noncurrent assets and liabilities, net, which were partially offset with a positive variance in the change in accounts receivable. For additional information about our operating results, see "Results of Operations" above.

Investing Activities

Net cash used in investing activities decreased by \$5.8 billion for the year ended December 31, 2018 as compared to the year ended December 31, 2017 and increased by \$5.9 billion for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The change in investing activities for both periods is primarily due to cash paid for the acquisition of Level 3 on November 1, 2017, which was partially offset with the cash proceeds from the May 2017 sale of our data centers and colocation business.

Financing Activities

Net cash used in financing activities increased by \$9.4 billion for the year ended December 31, 2018 as compared to the year ended December 31, 2017 and decreased by \$6.9 billion for the year ended December 31, 2017 as compared to the year ended December 31, 2016. The change in financing activities for both periods was primarily due cash received from net proceeds from issuance of new debt in 2017 relating to the acquisition of Level 3.

See Note 6—Long-Term Debt and Credit Facilities to our consolidated financial statements in Item 8 of Part II of this report, for information regarding indebtedness incurred or repaid by CenturyLink or its affiliates on our outstanding debt securities.

Other Matters

Recent Tax Changes

On December 22, 2017, the Tax Cuts and Jobs Act (the "Act") was signed into law. The Act reduced the U.S. corporate income tax rate from a maximum of 35% to 21% for all corporations, effective January 1, 2018, and made certain changes to U.S. taxation of income earned by foreign subsidiaries, capital expenditures and various other items.

As a result of the reduction in the U.S. corporate income tax rate from 35% to 21%, we provisionally re-measured our net deferred tax liabilities at December 31, 2017 and recognized a tax benefit of approximately \$1.1 billion in our consolidated statement of operations for the year ended December 31, 2017. As a result of finalizing our provisional amount recorded in 2017, we recorded a reduction to this amount of \$92 million in 2018. Based on current circumstances, we do not expect to experience a material near term reduction in the amount of cash income taxes paid by us from the Act due to utilization of net operating loss carryforwards. However, we anticipate that the provisions of the Act may reduce our cash income taxes in future years.

The Act imposed a one-time repatriation tax on certain earnings of foreign subsidiaries. We have completed our analysis of the impact of the one-time repatriation tax, and concluded that we do not have a tax liability under this provision.

During 2018, we accelerated a significant amount of tax deductions into 2017. The accelerated tax deductions resulted in a 2017 net operating loss for tax purposes, a portion of which was carried back to 2016 to generate a cash refund of \$392 million, which was received in the third quarter of 2018. Additionally, we received a \$314 million refund in the second quarter of 2018 related to 2017 federal income taxes.

For a more detailed description of the Act and its impact on us, please see Note 14—Income Taxes to the accompanying consolidated financial statements included in Item 8.

Other

We have cash management arrangements with certain of our principal subsidiaries, in which substantial portions of the subsidiaries' cash is regularly advanced to us. Although we periodically repay these advances to fund the subsidiaries' cash requirements throughout the year, at any given point in time we may owe a substantial sum to our subsidiaries under these advances, which, in accordance with generally accepted accounting principles, are eliminated in consolidation and therefore not recognized on our consolidated balance sheets.

We also are involved in various legal proceedings that could substantially impact our financial position. See Note 17—Commitments, Contingencies and Other Items to our consolidated financial statements in Item 8 of Part II of this report for the current status of such legal proceedings.

Market Risk

As of December 31, 2018, we are exposed to market risk from changes in interest rates on our variable rate long-term debt obligations and fluctuations in certain foreign currencies. We seek to maintain a favorable mix of fixed and variable rate debt in an effort to limit interest costs and cash flow volatility resulting from changes in rates.

Management periodically reviews our exposure to interest rate fluctuations and periodically implements strategies to manage the exposure. From time to time, we have used derivative instruments to (i) lock-in or swap our exposure to changing variable interest rates for fixed interest rates or (ii) to swap obligations to pay fixed interest rates for variable interest rates. As of December 31, 2018, we had no such instruments outstanding. We have established policies and procedures for risk assessment and the approval, reporting and monitoring of derivative instrument activities. As of December 31, 2018, we did not hold or issue derivative financial instruments for trading or speculative purposes.

As further discussed in Note 6—Long-Term Debt and Credit Facilities, on June 19, 2017, and on November 1, 2017, we borrowed substantial sums under a credit agreement dated June 19, 2017 with various lending institutions to provide a substantial amount of the funding for the Level 3 acquisition. As further noted in Note 6—Long-Term Debt and Credit Facilities, loans under the term loan facilities and new revolving credit facility under the June 19, 2017 credit agreement bear interest at floating rates. Based on debt outstanding at December 31, 2018, a hypothetical increase in 100 basis points in LIBOR relative to this debt would decrease our annual pre-tax earnings by \$132 million. On February 15, 2019, we executed a swap transaction that reduces our floating rate debt exposure by \$2.5 billion. For additional information see Note 6—Long-Term Debt and Credit Facilities to our consolidated financial statements in Item 8 of Part II of this report.

We conduct a portion of our business in currencies other than the U.S. dollar, the currency in which our consolidated financial statements are reported. Accordingly, our operating results could be adversely affected by foreign currency exchange rate volatility relative to the U.S. dollar. Our European subsidiaries and certain Latin American subsidiaries use the local currency as their functional currency, as the majority of their revenue and purchases are transacted in their local currencies. Certain Latin American countries previously designated as highly inflationary economies use the U.S. dollar as their functional currency. Although we continue to evaluate strategies to mitigate risks related to the effect of fluctuations in currency exchange rates, we will likely recognize gains or losses from international transactions. Changes in foreign currency rates could adversely affect our operating results.

Certain shortcomings are inherent in the method of analysis presented in the computation of exposures to market risks. Actual values may differ materially from those disclosed by us from time to time if market conditions vary from the assumptions used in the analyses performed. These analyses only incorporate the risk exposures that existed at December 31, 2018.

Off-Balance Sheet Arrangements

As of the date of this report, we have no special purpose or limited purpose entities that provide off-balance sheet financing, liquidity, or market or credit risk support and we do not engage in leasing, hedging or other similar activities that expose us to any significant liabilities that are not (i) reflected on the face of the consolidated financial statements, (ii) disclosed in Note 17—Commitments, Contingencies and Other Items to our consolidated financial statements in Item 8 of Part II of this report, or in the Future Contractual Obligations table included in this Item 7 of Part II above, or (iii) discussed under the heading "Market Risk" above.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

The information in "Management's Discussion and Analysis of Financial Condition and Results of Operations—Market Risk" in Item 7 of Part II of this report is incorporated herein by reference.

ITEM 8. CONSOLIDATED FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

Report of Independent Registered Public Accounting Firm

To the Stockholders and Board of Directors CenturyLink, Inc.:

Opinion on the Consolidated Financial Statements

We have audited the accompanying consolidated balance sheets of CenturyLink, Inc. and subsidiaries (the Company) as of December 31, 2018 and 2017, the related consolidated statements of operations, comprehensive (loss) income, cash flows, and stockholders' equity for each of the years in the three-year period ended December 31, 2018, and the related notes (collectively, the consolidated financial statements). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2018 and 2017, and the results of its operations and its cash flows for each of the years in the three-year period ended December 31, 2018, in conformity with U.S. generally accepted accounting principles.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the Company's internal control over financial reporting as of December 31, 2018, based on criteria established in *Internal Control - Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission, and our report dated March 11, 2019 expressed an adverse opinion on the effectiveness of the Company's internal control over financial reporting.

Basis for Opinion

These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement, whether due to error or fraud. Our audits included performing procedures to assess the risks of material misstatement of the consolidated financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the consolidated financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ KPMG LLP

We have served as the Company's auditor since 1977.

Shreveport, Louisiana March 11, 2019

Report of Independent Registered Public Accounting Firm

To the Stockholders and Board of Directors CenturyLink, Inc.:

Opinion on Internal Control Over Financial Reporting

We have audited CenturyLink, Inc. and subsidiaries' (the Company) internal control over financial reporting as of December 31, 2018, based on criteria established in *Internal Control - Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission. In our opinion, because of the effect of the material weaknesses, described below, on the achievement of the objectives of the control criteria, the Company has not maintained effective internal control over financial reporting as of December 31, 2018, based on criteria established in *Internal Control - Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the consolidated balance sheets of the Company as of December 31, 2018 and 2017, the related consolidated statements of income, comprehensive (loss) income, cash flows, and stockholders' equity for each of the years in the three-year period ended December 31, 2018, and the related notes (collectively, the consolidated financial statements), and our report dated March 11, 2019 expressed an unqualified opinion on those consolidated financial statements.

A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of the company's annual or interim financial statements will not be prevented or detected on a timely basis. Material weaknesses have been identified related to (i) ineffective design and operation of process level controls over the fair value measurement of certain assets acquired and liabilities assumed in a business combination, which arose because the Company did not conduct an effective risk assessment to identify and assess changes needed to process level controls resulting from the business combination, did not clearly assign responsibility for controls over the fair value measurements, and did not maintain effective information and communication processes to ensure the necessary information was available to personnel on a timely basis so they could fulfill their control responsibilities related to the fair value measurements; and (ii) ineffective design and operation of process level controls over the existence and accuracy of revenue transactions, which arose because the Company did not conduct an effective risk assessment to identify risks of material misstatement related to revenue transactions, and included in management's assessment.

The material weaknesses were considered in determining the nature, timing, and extent of audit tests applied in our audit of the 2018 consolidated financial statements, and this report does not affect our report on those consolidated financial statements.

Basis for Opinion

The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting, included in the accompanying Management's Report on Internal Control over Financial Reporting. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit of internal control over financial reporting included

obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. Our audit also included performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

Definition and Limitations of Internal Control Over Financial Reporting

A company's internal control over financial reporting is a process designed to provide reasonable assurance

regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ KPMG LLP

Shreveport, Louisiana March 11, 2019

CONSOLIDATED STATEMENTS OF OPERATIONS

| | Years Ended December 31, | | | | |
|--|--------------------------|-----------|--|---------|--|
| | 2018 | | 2017 | 2016 | |
| | | • | millions, except per share and shares in thousands) | | |
| OPERATING REVENUE | \$ | 23,443 | 17,656 | 17,470 | |
| OPERATING EXPENSES | | | | | |
| Cost of services and products (exclusive of depreciation and amortization) | | 10,862 | 8,203 | 7,774 | |
| Selling, general and administrative | | 4,165 | 3,508 | 3,447 | |
| Depreciation and amortization | | 5,120 | 3,936 | 3,916 | |
| Goodwill impairment | | 2,726 | - | _ | |
| Total operating expenses | | 22,873 | 15,647 | 15,137 | |
| OPERATING INCOME | | 570 | 2,009 | 2,333 | |
| OTHER (EXPENSE) INCOME | | | | | |
| Interest expense | | (2,177) | (1,481) | (1,318) | |
| Other income, net | | 44 | 12 | 5 | |
| Total other expense, net | | (2,133) | (1,469) | (1,313) | |
| INCOME (LOSS) BEFORE INCOME TAX EXPENSE | | (1,563) | 540 | 1,020 | |
| Income tax expense (benefit) | | 170 | (849) | 394 | |
| NET (LOSS) INCOME | \$ | (1,733) | 1,389 | 626 | |
| BASIC AND DILUTED (LOSS) EARNINGS PER COMMON SHARE | | | | | |
| BASIC | \$ | (1.63) | 2.21 | 1.16 | |
| DILUTED | \$ | (1.63) | 2.21 | 1.16 | |
| WEIGHTED AVERAGE COMMON SHARES OUTSTANDING | | | | | |
| BASIC | | 1,065,866 | 627,808 | 539,549 | |
| DILUTED | | 1,065,866 | 628,693 | 540,679 | |

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF COMPREHENSIVE (LOSS) INCOME

| | Years Ended December 31, | | | |
|--|--------------------------|---------|-----------------------|-------|
| | | 2018 | 2017 | 2016 |
| | | (| (Dollars in millions) | |
| NET (LOSS) INCOME | \$ | (1,733) | 1,389 | 626 |
| OTHER COMPREHENSIVE (LOSS) INCOME: | | | | |
| Items related to employee benefit plans: | | | | |
| Change in net actuarial gain (loss), net of (\$45), \$(60) and \$113 tax | | 133 | 83 | (168) |
| Change in net prior service credit, net of \$(3), \$(4) and \$(4) tax | | 9 | 8 | 6 |
| Foreign currency translation adjustment and other, net of \$50, \$(17) and \$— tax | | (201) | 31 | (21) |
| Other comprehensive (loss) income | | (59) | 122 | (183) |
| COMPREHENSIVE (LOSS) INCOME | \$ | (1,792) | 1,511 | 443 |

See accompanying notes to consolidated financial statements.

CONSOLIDATED BALANCE SHEETS

As of December 31,

| | As of December 31, | | | |
|--|--------------------|----------------|----------|--|
| | | 2018 | 18 2017 | |
| | | (Dollars in mi | | |
| <u>ASSETS</u> | | | | |
| CURRENT ASSETS | | | | |
| Cash and cash equivalents | \$ | 488 | 551 | |
| Restricted cash - current | | 4 | 5 | |
| Accounts receivable, less allowance of \$142 and \$164 | | 2,398 | 2,557 | |
| Assets held for sale | | 12 | 140 | |
| Other | | 918 | 941 | |
| Total current assets | | 3,820 | 4,194 | |
| NET PROPERTY, PLANT AND EQUIPMENT | | | | |
| Property, plant and equipment | | 53,267 | 51,204 | |
| Accumulated depreciation | | (26,859) | (24,352) | |
| Net property, plant and equipment | | 26,408 | 26,852 | |
| GOODWILL AND OTHER ASSETS | | | | |
| Goodwill | | 28,031 | 30,475 | |
| Restricted cash | | 26 | 31 | |
| Customer relationships, net | | 8,911 | 10,876 | |
| Other intangible assets, net | | 1,868 | 1,897 | |
| Other, net | | 1,192 | 1,286 | |
| Total goodwill and other assets | | 40,028 | 44,565 | |
| TOTAL ASSETS | \$ | 70,256 | 75,611 | |
| LIABILITIES AND STOCKHOLDERS' EQUITY | | | | |
| CURRENT LIABILITIES | | | | |
| Current maturities of long-term debt | \$ | 652 | 443 | |
| Accounts payable | | 1,933 | 1,555 | |
| Accrued expenses and other liabilities | | | | |
| Salaries and benefits | | 1,104 | 890 | |
| Income and other taxes | | 337 | 370 | |
| Interest | | 316 | 363 | |
| Other | | 357 | 344 | |
| Advance billings and customer deposits | | 832 | 892 | |
| Total current liabilities | | 5,531 | 4,857 | |
| LONG-TERM DEBT | | 35,409 | 37,283 | |
| DEFERRED CREDITS AND OTHER LIABILITIES | | | | |
| Deferred income taxes, net | | 2,527 | 2,413 | |
| Benefit plan obligations, net | | 4,319 | 5,178 | |
| Other | | 2,642 | 2,389 | |
| Total deferred credits and other liabilities | | 9,488 | 9,980 | |
| COMMITMENTS AND CONTINGENCIES (Note 17) | | | | |
| STOCKHOLDERS' EQUITY | | | | |
| Preferred stock — non-redeemable, \$25.00 par value, authorized 2,000 and 2,000 shares, issued and outstanding 7 and 7 shares | | _ | _ | |
| Common stock, \$1.00 par value, authorized 1,600,000 and 1,600,000 shares, issued and outstanding 1,080,167 and 1,069,169 shares | | 1,080 | 1,069 | |
| Additional paid-in capital | | 22,852 | 23,314 | |
| Accumulated other comprehensive loss | | (2,461) | (1,995) | |
| (Accumulated deficit) retained earnings | | (1,643) | 1,103 | |
| Total stockholders' equity | | 19,828 | 23,491 | |
| TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY | \$ | 70,256 | 75,611 | |

CONSOLIDATED STATEMENTS OF CASH FLOWS

| | _ | Years E | Ended December 31, | |
|--|----|---------|--------------------|--------|
| | | 2018 | 2017 | 2016 |
| | | (Do | llars in millions) | |
| OPERATING ACTIVITIES | | | | |
| Net (loss) income | \$ | (1,733) | 1,389 | 626 |
| Adjustments to reconcile net (loss) income to net cash provided by operating activities: | | | | |
| Depreciation and amortization | | 5,120 | 3,936 | 3,916 |
| Impairment of goodwill and other assets | | 2,746 | _ | 13 |
| Deferred income taxes | | 522 | (931) | 6 |
| Loss on the sale of data centers and colocation business | | _ | 82 | _ |
| Provision for uncollectible accounts | | 153 | 176 | 192 |
| Net long-term debt issuance costs and premium amortization | | 13 | 9 | 2 |
| Net loss on early retirement of debt | | 7 | 5 | 27 |
| Share-based compensation | | 186 | 111 | 80 |
| Changes in current assets and liabilities: | | | | |
| Accounts receivable | | 25 | 31 | (266 |
| Accounts payable | | 124 | (123) | 109 |
| Accrued income and other taxes | | 75 | 54 | (43 |
| Other current assets and liabilities, net | | 127 | (614) | 92 |
| Retirement benefits | | (667) | (202) | (152 |
| Changes in other noncurrent assets and liabilities, net | | 329 | (174) | (18 |
| Other, net | | 5 | 129 | 24 |
| Net cash provided by operating activities | | 7,032 | 3,878 | 4,608 |
| INVESTING ACTIVITIES | | | | |
| Payments for property, plant and equipment and capitalized software | | (3,175) | (3,106) | (2,981 |
| Cash paid for Level 3 acquisition, net of \$2.3 billion cash acquired | | _ | (7,289) | _ |
| Proceeds from sale of property and intangible assets | | 158 | 1,529 | 30 |
| Other, net | | (61) | (5) | (43 |
| Net cash used in investing activities | | (3,078) | (8,871) | (2,994 |
| FINANCING ACTIVITIES | | | | |
| Net proceeds from issuance of long-term debt | | 130 | 8,398 | 2,161 |
| Proceeds from financing obligation (Note 3) | | _ | 356 | _ |
| Payments of long-term debt | | (1,936) | (1,963) | (2,462 |
| Net proceeds (payments) on credit facility and revolving line of credit | | 145 | 35 | (40 |
| Dividends paid | | (2,312) | (1,453) | (1,167 |
| Other, net | | (50) | (17) | (10 |
| Net cash (used in) provided by financing activities | | (4,023) | 5,356 | (1,518 |
| Net (decrease) increase in cash, cash equivalents and restricted cash | | (69) | 363 | 96 |
| Cash, cash equivalents and restricted cash at beginning of period | | 587 | 224 | 128 |
| Cash, cash equivalents and restricted cash at end of period | \$ | 518 | 587 | 224 |
| Supplemental cash flow information: | | | | |
| Income taxes received (paid), net | \$ | 674 | (392) | (397 |
| Interest paid (net of capitalized interest of \$53, \$78 and \$54) | \$ | (2,138) | (1,401) | (1,301 |
| Therest paid (flet of capitalized interest of \$55, \$70 and \$54) | Ψ | (2,100) | (1,401) | (1,501 |
| Cash, cash equivalents and restricted cash: | | | | |
| Cash and cash equivalents | \$ | 488 | 551 | 222 |
| Restricted cash - current | | 4 | 5 | _ |
| Restricted cash - noncurrent | | 26 | 31 | 2 |
| Total | \$ | 518 | 587 | 224 |

CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

| | Years Ended December 31, | | | | |
|--|--------------------------|---------------------|---------------------|----------|--|
| | - | 2018 | 2017 | 2016 | |
| | | (Dollars in million | ns except per share | amounts) | |
| COMMON STOCK | | | | | |
| Balance at beginning of period | \$ | 1,069 | 547 | 544 | |
| Issuance of common stock to acquire Level 3, including replacement of Level 3's share-based compensation awards | | _ | 517 | _ | |
| Issuance of common stock through dividend reinvestment, incentive and benefit plans | | 11 | 5 | 3 | |
| Balance at end of period | ' <u> </u> | 1,080 | 1,069 | 547 | |
| ADDITIONAL PAID-IN CAPITAL | | | | | |
| Balance at beginning of period | | 23,314 | 14,970 | 15,178 | |
| Issuance of common stock to acquire Level 3, including replacement of Level 3's share-based compensation awards | | (2) | 9,462 | _ | |
| Issuance of common stock through dividend reinvestment, incentive and benefit plans | | _ | _ | 7 | |
| Shares withheld to satisfy tax withholdings | | (56) | (20) | (15) | |
| Share-based compensation and other, net | | 187 | 79 | 79 | |
| Dividends declared | | (586) | (1,177) | (279) | |
| Acquisition of additional minority interest in a subsidiary | | (5) | _ | _ | |
| Balance at end of period | | 22,852 | 23,314 | 14,970 | |
| ACCUMULATED OTHER COMPREHENSIVE LOSS | | | | | |
| Balance at beginning of period | | (1,995) | (2,117) | (1,934) | |
| Cumulative effect of adoption of ASU 2018-02, Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income | | (407) | _ | _ | |
| Other comprehensive (loss) income | | (59) | 122 | (183) | |
| Balance at end of period | | (2,461) | (1,995) | (2,117) | |
| RETAINED EARNINGS (ACCUMULATED DEFICIT) | | | | | |
| Balance at beginning of period | | 1,103 | (1) | 272 | |
| Net (loss) income | | (1,733) | 1,389 | 626 | |
| Cumulative effect of adoption of ASU 2018-02, Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income | | 407 | _ | _ | |
| Cumulative net effect of adoption of ASU 2014-09, Revenue from Contracts with Customers, net of \$(119), \$— and \$— tax | | 338 | _ | _ | |
| Cumulative effect of adoption of ASU 2016-09, Improvements to Employee Share- Based Payment Accounting | | _ | 3 | _ | |
| Dividends declared | | (1,758) | (288) | (899) | |
| Balance at end of period | | (1,643) | 1,103 | (1) | |
| TOTAL STOCKHOLDERS' EQUITY | \$ | 19,828 | 23,491 | 13,399 | |
| DIVIDENDS DECLARED PER COMMON SHARE | \$ | 2.16 | 2.16 | 2.16 | |

See accompanying notes to consolidated financial statements.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

References in the Notes to "CenturyLink," "we," "us", the "Company", and "our" refer to CenturyLink, Inc. and its consolidated subsidiaries, unless the content otherwise requires and except in Note 6, where such references refer solely to CenturyLink, Inc. References in the Notes to "Level 3" refer to Level 3 Communications, Inc. prior to our acquisition thereof and to its successor-in-interest Level 3 Parent, LLC after such acquisition, unless the context otherwise requires.

(1) Background and Summary of Significant Accounting Policies

General

We are an international facilities-based communications company engaged primarily in providing a broad array of integrated services to our residential and business customers.

On November 1, 2017, we acquired Level 3 Communications, Inc. ("Level 3") in a cash and stock transaction. See Note 2—Acquisition of Level 3 for additional information. On May 1, 2017, we sold our data centers and colocation business to a consortium of private equity purchasers for a combination of cash and equity. See Note 3—Sale of Data Centers and Colocation Business for additional information.

Basis of Presentation

The accompanying consolidated financial statements include our accounts and the accounts of our subsidiaries in which we have a controlling interest. These subsidiaries include Level 3 on and after November 1, 2017. Intercompany amounts and transactions with our consolidated subsidiaries have been eliminated. In connection with our acquisition of Level 3, we acquired its deconsolidated Venezuela subsidiary and due to exchange restrictions and other conditions we have assigned no value to this subsidiary's assets. Additionally, we have excluded this subsidiary from our consolidated financial statements.

To simplify the overall presentation of our consolidated financial statements, we report immaterial amounts attributable to noncontrolling interests in certain of our subsidiaries as follows: (i) income attributable to noncontrolling interests in other income, net, (ii) equity attributable to noncontrolling interests in additional paid-in capital and (iii) cash flows attributable to noncontrolling interests in other, net financing activities.

We reclassified certain prior period amounts to conform to the current period presentation, including the categorization of our revenue and our segment reporting for 2018, 2017 and 2016. See Note 15—Segment Information for additional information. These changes had no impact on total operating revenue, total operating expenses or net (loss) income for any period.

Summary of Significant Accounting Policies

Use of Estimates

Our consolidated financial statements are prepared in accordance with U.S. generally accepted accounting principles. These accounting principles require us to make certain estimates, judgments and assumptions. We believe that the estimates, judgments and assumptions we make when accounting for specific items and matters, including, but not limited to, revenue recognition, revenue reserves, network access costs, network access cost dispute reserves, investments, long-term contracts, customer retention patterns, allowance for doubtful accounts, depreciation, amortization, asset valuations, internal labor capitalization rates, recoverability of assets (including deferred tax assets), impairment assessments, pension, post-retirement and other post-employment benefits, taxes, certain liabilities and other provisions and contingencies, are reasonable, based on information available at the time they are made. These estimates, judgments and assumptions can materially affect the reported amounts of assets, liabilities and components of stockholders' equity as of the dates of the consolidated balance sheets, as well as the reported amounts of revenue, expenses and components of cash flows during the periods presented in our other consolidated financial statements. We also make estimates in our assessments of potential losses in relation to threatened or pending tax and legal matters. See Note 14—Income Taxes and Note 17—Commitments, Contingencies and Other Items for additional information.

For matters not related to income taxes, if a loss is considered probable and the amount can be reasonably estimated, we recognize an expense for the estimated loss. If we have the potential to recover a portion of the estimated loss from a third party, we make a separate assessment of recoverability and reduce the estimated loss if recovery is also deemed probable.

For matters related to income taxes, if we determine that the impact of an uncertain tax position is more likely than not to be sustained upon audit by the relevant taxing authority, then we recognize a benefit for the largest amount that is more likely than not to be sustained. No portion of an uncertain tax position will be recognized if the position has less than a 50% likelihood of being sustained. Interest is recognized on the amount of unrecognized benefit from uncertain tax positions.

For all of these and other matters, actual results could differ materially from our estimates.

Revenue Recognition

We earn most of our consolidated revenue from contracts with customers, primarily through the provision of telecommunications and other services. Revenue from contracts with customers is accounted for under Accounting Standards Codification ("ASC") 606. We also earn revenue from leasing arrangements (primarily fiber capacity agreements) and governmental subsidy payments, neither of which are accounted for under ASC 606.

Revenue is recognized when control of the promised goods or services is transferred to our customers, in an amount that reflects the consideration we expect to be entitled to receive in exchange for those goods or services. Revenue is recognized based on the following five-step model:

- · Identification of the contract with a customer;
- · Identification of the performance obligations in the contract;
- · Determination of the transaction price;
- · Allocation of the transaction price to the performance obligations in the contract; and
- · Recognition of revenue when, or as, we satisfy a performance obligation.

We provide an array of communications services to residential and business customers, including local voice, VPN, Ethernet, data, broadband, private line (including special access), network access, transport, voice, information technology, video and other ancillary services. We provide these services to a wide range of businesses, including global/international, enterprise, wholesale, government, small and medium business customers. Certain contracts also include the sale of equipment, which is not significant to our business.

We recognize revenue for services when we provide the applicable service or when control is transferred. Recognition of certain payments received in advance of services being provided is deferred. These advance payments include certain activation and certain installation charges. If the activation and installation charges are not separate performance obligations, we recognize them as revenue over the actual or expected contract term using historical experience, which ranges from one year to seven years depending on the service. In most cases, termination fees or other fees on existing contracts that are negotiated in conjunction with new contracts are deferred and recognized over the new contract term.

For access services, we generally bill fixed monthly charges one month in advance to customers and recognize revenue as service is provided over the contract term in alignment with the customer's receipt of service. For usage and other ancillary services, we generally bill in arrears and recognize revenue as usage or delivery occurs.

In certain cases, customers may be permitted to modify their contracts. We evaluate the change in scope or price to identify whether the modification should be treated as a separate contract, whether the modification is a termination of the existing contract and creation of a new contract, or if it is a change to the existing contract.

Customer contracts are evaluated to determine whether the performance obligations are separable. If the performance obligations are deemed separable and separate earnings processes exist, the total transaction price that we expect to receive with the customer is allocated to each performance obligation based on its relative standalone selling price. The revenue associated with each performance obligation is then recognized as earned.

We periodically sell optical capacity on our network. These transactions are structured as indefeasible rights of use, commonly referred to as IRUs, which are the exclusive right to use a specified amount of capacity or fiber for a specified term, typically 10 to 20 years. In most cases, we account for the cash consideration received on transfers of optical capacity as ASC 606 revenue which we recognize ratably over the term of the agreement. Cash consideration received on transfers of dark fiber is adjusted for the time value of money and is accounted for as non-ASC 606 lease revenue, which we also recognize ratably over the term of the agreement. We do not recognize revenue on any contemporaneous exchanges of our optical capacity assets for other non-owned optical capacity assets.

In connection with offering products and services provided to the end user by third-party vendors, we review the relationship between us, the vendor and the end user to assess whether revenue should be reported on a gross or net basis. In assessing whether revenue should be reported on a gross or net basis, we consider whether we act as a principal in the transaction and control the goods and services used to fulfill the performance obligations associated with the transaction.

We have service level commitments pursuant to contracts with certain of our customers. To the extent that such service levels are not achieved or are otherwise disputed due to performance or service issues or other service interruptions or conditions, we will estimate the amount of credits to be issued and record a corresponding reduction to revenue in the period that the service level commitment was not met.

Customer payments are made based on billing schedules included in our customer contracts, which is typically on a monthly basis.

We defer (i.e. capitalize) incremental contract acquisition and fulfillment costs and recognize (or amortize) such costs over the average customer life. Our deferred contract costs for our customers have average amortization periods of approximately 30 months for consumer and up to 49 months for business. These deferred costs are monitored every period to reflect any significant change in assumptions.

See Note 5—Revenue Recognition for additional information.

USF Surcharges, Gross Receipts Taxes and Other Surcharges

In determining whether to include in our revenue and expenses the taxes and surcharges collected from customers and remitted to government authorities, including USF surcharges, sales, use, value added and some excise taxes, we assess, among other things, whether we are the primary obligor or principal taxpayer for the taxes assessed in each jurisdiction where we do business. In jurisdictions where we determine that we are the principal taxpayer, we record the surcharges on a gross basis and include them in our revenue and costs of services and products. In jurisdictions where we determine that we are merely a collection agent for the government authority, we record the taxes on a net basis and do not include them in our revenue and costs of services and products.

Advertising Costs

Costs related to advertising are expensed as incurred and included in selling, general and administrative expenses in our consolidated statements of operations. Our advertising expense was \$98 million , \$218 million and \$216 million for the years ended December 31, 2018 , 2017 and 2016 , respectively.

Legal Costs

In the normal course of our business, we incur costs to hire and retain external legal counsel to advise us on regulatory, litigation and other matters. We expense these costs as the related services are received.

Income Taxes

We file a consolidated federal income tax return with our eligible subsidiaries. The provision for income taxes consists of an amount for taxes currently payable, an amount for tax consequences deferred to future periods and adjustments to our liabilities for uncertain tax positions. We record deferred income tax assets and liabilities reflecting future tax consequences attributable to tax net operating loss carryforwards ("NOLs"), tax credit carryforwards and differences between the financial statement carrying value of assets and liabilities and the tax basis of those assets and liabilities. Deferred taxes are computed using enacted tax rates expected to apply in the year in which the differences are expected to affect taxable income. The effect on deferred income tax assets and liabilities of a change in tax rate is recognized in earnings in the period that includes the enactment date.

We establish valuation allowances when necessary to reduce deferred income tax assets to the amounts that we believe are more likely than not to be recovered. Each quarter we evaluate the need to retain all or a portion of the valuation allowance on our deferred tax assets. See Note 14—Income Taxes for additional information.

Cash and Cash Equivalents

Cash and cash equivalents include highly liquid investments that are readily convertible into cash and are not subject to significant risk from fluctuations in interest rates. As a result, the value at which cash and cash equivalents are reported in our consolidated financial statements approximates their fair value. In evaluating investments for classification as cash equivalents, we require that individual securities have original maturities of ninety days or less and that individual investment funds have dollar-weighted average maturities of ninety days or less. To preserve capital and maintain liquidity, we invest with financial institutions we deem to be of sound financial condition and in high quality and relatively risk-free investment products. Our cash investment policy limits the concentration of investments with specific financial institutions or among certain products and includes criteria related to credit worthiness of any particular financial institution.

Book overdrafts occur when checks have been issued but have not been presented to our controlled disbursement bank accounts for payment. Disbursement bank accounts allow us to delay funding of issued checks until the checks are presented for payment. Until the issued checks are presented for payment, the book overdrafts are included in accounts payable on our consolidated balance sheet. This activity is included in the operating activities section in our consolidated statements of cash flows.

Restricted Cash and Securities

Restricted cash and securities consists primarily of cash and investments that serve to collateralize our outstanding letters of credit and certain performance and operating obligations. Restricted cash and securities are recorded as current or non-current assets in the consolidated balance sheets depending on the duration of the restriction and the purpose for which the restriction exists. Restricted securities are stated at cost which approximates fair value as of December 31, 2018 and 2017.

Accounts Receivable and Allowance for Doubtful Accounts

Accounts receivable are recognized based upon the amount due from customers for the services provided or at cost for purchased and other receivables less an allowance for doubtful accounts. The allowance for doubtful accounts receivable reflects our best estimate of probable losses inherent in our receivable portfolio determined on the basis of historical experience, specific allowances for known troubled accounts and other currently available evidence. We generally consider our accounts past due if they are outstanding over 30 days. Our collection process varies by the customer segment, amount of the receivable, and our evaluation of the customer's credit risk. Our past due accounts are written off against our allowance for doubtful accounts when collection is considered to be not probable. Any recoveries of accounts previously written off are generally recognized as a reduction in bad debt expense in the period received. The carrying value of accounts receivable net of the allowance for doubtful accounts approximates fair value. Accounts receivable balances acquired in a business combination are recorded at fair value for all balances receivable at the acquisition date and at the invoiced amount for those amounts invoiced after the acquisition date.

Property, Plant and Equipment

We record property, plant and equipment acquired in connection with our acquisitions based on its estimated fair value as of its acquisition date plus the estimated value of any associated legally or contractually required retirement obligations. We record purchased and constructed property, plant and equipment at cost, plus the estimated value of any associated legally or contractually required retirement obligations. The majority of our property, plant and equipment is depreciated using the straight-line group method, but certain of our assets are depreciated using the straight-line method over their estimated useful lives of the specific asset. Under the straight-line group method, assets dedicated to providing telecommunications services (which comprise the majority of our property, plant and equipment) that have similar physical characteristics, use and expected useful lives are pooled for purposes of depreciation and tracking. The equal life group procedure is used to establish each pool's average remaining useful life. Generally, under the straight-line group method, when an asset is sold or retired in the course of normal business activities, the cost is deducted from property, plant and equipment and charged to accumulated depreciation without recognition of a gain or loss. A gain or loss is recognized in our consolidated statements of operations only if a disposal is unusual. Leasehold improvements are amortized over the shorter of the useful lives of the assets or the expected lease term. Expenditures for maintenance and repairs are expensed as incurred. Interest is capitalized during the construction phase of network and other internal-use capital projects. Employee-related costs for construction of network and other internal use assets are also capitalized during the construction phase. Property, plant and equipment supplies used internally are carried at average cost, except for significant individual items for which cost is based on specific identification.

We perform annual internal reviews to evaluate the reasonableness of the depreciable lives for our property, plant and equipment. Our reviews utilize models that take into account actual usage, physical wear and tear, replacement history, assumptions about technology evolution and, in certain instances, actuarially determined probabilities to estimate the remaining useful life of our asset base. Our remaining useful life assessments assess the possible loss in service value of assets that may precede the physical retirement. Assets shared among many customers may lose service value as those customers reduce their use of the asset. However, the asset is not retired until all customers no longer utilize the asset and we determine there is no alternative use for the asset.

We have asset retirement obligations associated with the legally or contractually required removal of a limited group of property, plant and equipment assets from leased properties and the disposal of certain hazardous materials present in our owned properties. When an asset retirement obligation is identified, usually in association with the acquisition of the asset, we record the fair value of the obligation as a liability. The fair value of the obligation is also capitalized as property, plant and equipment and then amortized over the estimated remaining useful life of the associated asset. Where the removal obligation is not legally binding, the net cost to remove assets is expensed in the period in which the costs are actually incurred.

Capitalized labor associated with employees and contract labor working on capital projects were approximately \$569 million, \$559 million and \$524 million for the years ended December 31, 2018, 2017 and 2016.

We review long-lived tangible assets for impairment whenever facts and circumstances indicate that the carrying amounts of the assets may not be recoverable. For assessment purposes, long-lived assets are grouped with other assets and liabilities at the lowest level for which identifiable cash flows are largely independent of the cash flows of other assets and liabilities, absent a material change in operations. An impairment loss is recognized only if the carrying amount of the asset group is not recoverable and exceeds its estimated fair value. Recoverability of the asset group to be held and used is assessed by comparing the carrying amount of the asset group to the estimated undiscounted future net cash flows expected to be generated by the asset group. If the asset group's carrying value is not recoverable, we recognize an impairment charge for the amount by which the carrying amount of the asset group exceeds its estimated fair value.

Goodwill, Customer Relationships and Other Intangible Assets

Intangible assets arising from business combinations, such as goodwill, customer relationships, capitalized software, trademarks and trade names, are initially recorded at estimated fair value. We amortize customer relationships primarily over an estimated life of 7 to 15 years, using either the sum-of-years-digits or the straight-line methods, depending on the type of customer. We amortize capitalized software using the straight-line method over estimated lives ranging up to 7 years, except for approximately \$237 million of our capitalized software costs, which represents costs to develop an integrated billing and customer care system which is amortized using the straight-line method over a 20 -year period. We amortize our other intangible assets using the sum-of-years-digits or straight-line method over an estimated life of 4 to 20 years. Other intangible assets not arising from business combinations are initially recorded at cost. Where there are no legal, regulatory, contractual or other factors that would reasonably limit the useful life of an intangible asset, we classify the intangible asset as indefinite-lived and such intangible assets are not amortized.

Internally used software, whether purchased or developed by us, is capitalized and amortized using the straight-line method over its estimated useful life. We have capitalized certain costs associated with software such as costs of employees devoting time to the projects and external direct costs for materials and services. Costs associated with software to be used for internal purposes are expensed until the point at which the project has reached the development stage. Subsequent additions, modifications or upgrades to internal-use software are capitalized only to the extent that they allow the software to perform a task it previously did not perform. Software maintenance, data conversion and training costs are expensed in the period in which they are incurred. We review the remaining economic lives of our capitalized software annually. Capitalized software is included in other intangible assets, net, in our consolidated balance sheets.

Our long-lived intangible assets, other than goodwill, with indefinite lives are assessed for impairment annually, or, under certain circumstances, more frequently, such as when events or changes in circumstances indicate there may be an impairment. These assets are carried at the estimated fair value at the time of acquisition and assets not acquired in acquisitions are recorded at historical cost. However, if their estimated fair value is less than the carrying amount, we recognize an impairment charge for the amount by which the carrying amount of these assets exceeds their estimated fair value.

We are required to assess goodwill for impairment at least annually, or more frequently, if an event occurs or circumstances change that would indicate an impairment may have occurred. We are required to write-down the value of goodwill in periods in which the recorded carrying value of equity exceeds the fair value of equity. Our reporting units are not discrete legal entities with discrete full financial statements. Therefore, the equity carrying value and future cash flows is assessed each time a goodwill impairment assessment is performed on a reporting unit. To do so, we assign our assets, liabilities and cash flows to reporting units using reasonable and consistent allocation methodologies, which entail various estimates, judgments and assumptions. We believe these estimates, judgments and assumptions to be reasonable, but changes in any of these can significantly affect each reporting unit's equity carrying value and future cash flows utilized for our goodwill impairment assessment.

We are required to reassign goodwill to reporting units each time we reorganize our internal reporting structure which causes a change in the composition of our reporting units. Goodwill is reassigned to the reporting units using a relative fair value approach. When the fair value of a reporting unit is available, we allocate goodwill based on the relative fair value of the reporting units. When fair value is not available, we utilize the earnings before interest, taxes, depreciation and amortization of each reporting unit as our allocation methodology based on our view that it represents a reasonable proxy for the fair value of the operations being reorganized.

As of October 31, 2018, we assessed goodwill for impairment for our five reporting units, which we determined to be consumer, medium and small business, enterprise, international and global accounts, and wholesale and indirect. We determined that the estimated fair value of our consumer reporting unit's equity was substantially below our carrying value of equity. As a result, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill assigned to our consumer reporting unit during the fourth guarter of 2018.

Subsequent Event

As a result of organizational changes made in January 2019, we are making changes to our revenue reporting categories in 2019 to align them with how the business is managed. Our indirect channel, which primarily targets small to medium-sized enterprises, will move from the wholesale and indirect business unit. We are also moving state and local government customers from the small and medium business unit to the enterprise business unit to gain efficiencies by managing all government customers in a single organization. In addition, the new reporting structure reflects changes made to customer assignments between all five customer-facing business units.

We are making changes to the service-type reporting to make it easier for management and investors to evaluate changes to consumer product revenue. We plan to report consumer revenue in the following categories: broadband; voice; regulatory (includes CAF II and other support funds); and other, which includes retail video and other miscellaneous services. We plan to maintain the structure of our business segment, with minor modifications.

As a result of the organization changes noted above, we will perform a goodwill impairment analysis during the first quarter of 2019. The goodwill impairment analysis has not been completed at the time of this report and any goodwill impairment is not reasonably estimable.

See Note 4—Goodwill, Customer Relationships and Other Intangible Assets for additional information.

Pension and Post-Retirement Benefits

We recognize the funded status of our defined benefit and post-retirement plans as an asset or a liability on our consolidated balance sheet. Each year's actuarial gains or losses are a component of our other comprehensive income (loss), which is then included in our accumulated other comprehensive loss. Pension and post-retirement benefit expenses are recognized over the period in which the employee renders service and becomes eligible to receive benefits. We make significant assumptions (including the discount rate, expected rate of return on plan assets, mortality and health care trend rates) in computing the pension and post-retirement benefits expense and obligations. Note 10—Employee Benefits for additional information.

Foreign Currency

Local currencies of foreign subsidiaries are the functional currencies for financial reporting purposes except for certain foreign subsidiaries, primarily in Latin America. For operations outside the United States that have functional currencies other than the U.S. dollar, assets and liabilities are translated to U.S. dollars at period-end exchange rates, and revenue, expenses and cash flows are translated using average monthly exchange rates. A significant portion of our non-United States subsidiaries have either the British pound, the euro or the Brazilian real as the functional currency, each of which experienced significant fluctuations against the U.S. dollar during the years ended December 31, 2018, 2017 and 2016. Foreign currency translation gains and losses are recognized as a component of accumulated other comprehensive loss in stockholders' equity and in the consolidated statements of comprehensive income (loss) in accordance with accounting guidance for foreign currency translation. We consider the majority of our investments in our foreign subsidiaries to be long-term in nature. Our foreign currency transaction gains (losses), including where transactions with our non-United States subsidiaries are not considered to be long-term in nature, are included within other income, net on the consolidated statements of operations.

Common Stock

At December 31, 2018, we had 4 million unissued shares of CenturyLink, Inc. common stock reserved for acquisitions. In addition, we had 30 million shares authorized for future issuance under our equity incentive plans.

Preferred Stock

Holders of outstanding CenturyLink, Inc. preferred stock are entitled to receive cumulative dividends, receive preferential distributions equal to \$25 per share plus unpaid dividends upon CenturyLink, Inc.'s liquidation and vote as a single class with the holders of common stock.

Section 382 Rights Plan

On February 13, 2019, we adopted a Section 382 Rights Plan to protect our U.S. federal net operating loss carryforwards from certain Internal Revenue Code Section 382 limitations. Under the plan, one preferred stock purchase right was distributed for each share of our outstanding common stock as of the close of business on February 25, 2019, and those rights currently trade in tandem with the common stock until they expire or detach under the plan. This plan was designed to deter trading that would result in a change of control (as defined in Code Section 382), and therefore protect our ability to use our historical federal net operating losses in the future.

Dividends

The declaration and payment of dividends is at the discretion of our Board of Directors.

Recently Adopted Accounting Pronouncements

During 2018, we adopted Accounting Standards Update ("ASU") 2014-09, "Revenue from Contracts with Customers", ASU 2018-02, "Income Statement-Reporting Comprehensive Income: Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income", ASU 2016-16, "Intra-Entity Transfers of Assets Other Than Inventory", ASU 2017-04, "Simplifying the Test for Goodwill Impairment" and ASU 2018-14, "Compensation-Retirement Benefits-Defined Benefit Plans-General: Disclosure Framework-Changes to the Disclosure Requirements for Defined Benefit Plans".

Each of these is described further below.

Revenue Recognition

In May 2014, the FASB issued ASU 2014-09 which replaces virtually all existing generally accepted accounting principles on revenue recognition with a principles-based approach for determining revenue recognition using a new five step model. The core principle of ASU 2014-09 is that an entity should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those goods or services. ASU 2014-09 also includes new accounting principles related to the deferral and amortization of contract acquisition and fulfillment costs.

We adopted the new revenue recognition standard under the modified retrospective transition method. During the year ended December 31, 2018, we recorded a cumulative catch-up adjustment that increased our retained earnings by \$338 million, net of \$119 million of income taxes.

See Note 5—Revenue Recognition for additional information.

Comprehensive Loss

In February 2018, the FASB issued ASU 2018-02 provides an option to reclassify stranded tax effects within accumulated other comprehensive loss to retained earnings in each period in which the effect of the change in the U.S. federal corporate income tax rate in the Tax Cuts and Jobs Act (the "Act") (or portion thereof) is recorded. If an entity elects to reclassify the income tax effects of the Act, the amount of that reclassification shall include the effect of the change in the U.S. federal corporate income tax rate on the gross deferred tax amounts and related valuation allowances, if any, at the date of enactment of the Act related to items remaining in accumulated other comprehensive loss. The effect of the change in the U.S. federal corporate income tax rate on gross valuation allowances that were originally charged to income from continuing operations shall not be included. ASU 2018-02 is effective January 1, 2019, but early adoption is permitted and should be applied either in the period of adoption or retrospectively to each period (or periods) in which the effect of the change in the U.S. federal corporate income tax rate in the Act is recognized. We early adopted and applied ASU 2018-02 in the first quarter of 2018. The adoption of ASU 2018-02 resulted in a \$407 million increase to retained earnings and in accumulated other comprehensive loss. See Note 20—Accumulated Other Comprehensive Loss for additional information.

Income Taxes

In October 2016, the FASB issued ASU 2016-16, "Intra-Entity Transfers of Assets Other Than Inventory" ("ASU 2016-16"). ASU 2016-16 eliminates the current prohibition on the recognition of the income tax effects on the transfer of assets among our subsidiaries. After adoption of ASU 2016-16, the income tax effects associated with these asset transfers, except for the transfer of inventory, will be recognized in the period the asset is transferred versus the current deferral and recognition upon either the sale of the asset to a third party or over the remaining useful life of the asset. We adopted ASU 2016-16 on January 1, 2018. The adoption of ASU 2016-16 did not have a material impact to our consolidated financial statements.

Goodwill Impairment

In January 2017, the FASB issued ASU 2017-04, "Simplifying the Test for Goodwill Impairment" ("ASU 2017-04"). ASU 2017-04 simplifies the impairment testing for goodwill by changing the measurement for goodwill impairment. Under current rules, we are required to compute the fair value of goodwill to measure the impairment amount if the carrying value of a reporting unit exceeds its fair value. Under ASU 2017-04, the goodwill impairment charge will equal the excess of the reporting unit carrying value above its fair value, limited to the amount of goodwill assigned to the reporting unit.

We elected to early adopt the provisions of ASU 2017-04 as of October 1, 2018. We applied ASU 2017-04 to determine the impairment of \$2.7 billion recorded during the fourth quarter of 2018. See Note 4 - Goodwill, Customer Relationships and Other Intangible Assets for additional information.

Retirement Benefits

In August 2018, the FASB issued ASU 2018-14, " Compensation-Retirement Benefits-Defined Benefit Plans-General: Disclosure Framework-Changes to the Disclosure Requirements for Defined Benefit Plans " ("ASU 2018-14"). ASU 2018-14 eliminates requirements for certain disclosures that are not considered cost beneficial, clarifies certain required disclosures and adds additional disclosures under defined benefit pension plans and other postretirement plans. We adopted this guidance during the fourth quarter 2018. The adoption of ASU 2018-14 did not have a material impact to our consolidated financial statements.

Recently Issued Accounting Pronouncements

Financial Instruments

In June 2016, the FASB issued ASU 2016-13, " *Measurement of Credit Losses on Financial Instruments*" ("ASU 2016-13"). The primary impact of ASU 2016-13 for us is a change in the model for the recognition of credit losses related to our financial instruments from an incurred loss model, which recognized credit losses only if it was probable that a loss had been incurred, to an expected loss model, which requires our management team to estimate the total credit losses expected on the portfolio of financial instruments. We are currently reviewing the requirements of the standard and evaluating the impact on our consolidated financial statements.

We are required to adopt the provisions of ASU 2016-13 effective January 1, 2020, but could elect to early adopt the provisions as of January 1, 2019. We expect to adopt ASU 2016-13 on January 1, 2020 and recognize the impacts through a cumulative adjustment to retained earnings as of the date of adoption.

Leases

In February 2016, the FASB issued ASU 2016-02, "Leases" ("ASU 2016-02"), and associated ASUs related to ASU 842, Leases, which require organizations that lease assets to recognize on the balance sheet the assets and liabilities for the rights and obligations created by those leases. In addition, the new guidance will require disclosures to help investors and other financial statement users better understand the amount, timing and uncertainty of cash flows arising from leases. For leases where we are a lessee, the presentation and measurement of the assets and liabilities will depend on each lease's classification as either a finance or operating lease. For leases where we are a lessor, the accounting remains largely unchanged from current U.S. GAAP but does contain some targeted improvements to align with the new revenue recognition guidance issued in 2014 (ASC 606). The new standard is effective for fiscal years, and interim periods within those fiscal years, beginning after December 15, 2018.

We have a cross-functional team in place to evaluate and implement the new guidance and we have substantially completed the implementation of third-party software solutions to facilitate compliance with accounting and reporting requirements. The team continues to review existing lease arrangements and has collected and loaded a significant portion of our lease portfolio into the software. We continue to enhance accounting systems and update business processes and controls related to the new guidance for leases. Collectively, these activities are expected to facilitate our ability to meet the new accounting and disclosure requirements upon adoption in the first quarter of 2019.

ASU 2016-02 requires a modified retrospective transition approach, applying the new standard to all leases existing at the date of initial adoption. An entity may choose to use either (1) the effective date or (2) the beginning of the earliest comparative period presented in the financial statements at the date of initial application. We will apply the transition requirements at the January 1, 2019 effective date by showing a cumulative effect adjustment in the first quarter of 2019, rather than restating any prior periods. In addition, we will elect the package of practical expedients permitted under the transition guidance, which does not require reassessment of prior conclusions related to contracts containing a lease, lease classification and initial direct lease costs. As an accounting policy election, we will exclude short-term leases (term of 12 months or less) from the balance sheet presentation and will account for non-lease and lease components in a contract as a single lease component for most asset classes.

We are in the process of completing our adoption of ASU 2016-02, including reviewing our lease portfolio, completing the implementation and testing of the third-party software solution and exercising internal controls over adoption and implementation of ASU 2016-02. Therefore, the estimated impact on our consolidated balance sheet cannot currently be determined. However, we expect the adoption of ASU 2016-02 will have a material impact on our consolidated balance sheet through the recognition of right of use assets and lease liabilities for our operating leases. The impact to our consolidated statements of operations and consolidated statements of cash flows is not expected to be material. We believe the new standard will have no impact on our debt covenant compliance under our current agreements.

We currently lease real estate, vehicles, dark fiber and a wide variety of equipment. In addition, deferred gains of approximately \$115 million arising from prior period sales-leaseback transactions, which under prior accounting rules would have been recognized on our operating statements over an average period of three years, will be reflected through an adjustment to retained earnings on our balance sheet as of January 1, 2019.

Upon implementing ASU 2016-02, accounting for the failed sale leaseback will no longer be applicable based on our facts and circumstances, and the real estate assets and corresponding financing obligation will be derecognized from our consolidated financial statements. Please see Note 3—Sale of Data Centers and Colocation Business for additional information on the accounting for the failed-sale-leaseback. The elimination of the failed leaseback will result in the following increase (decrease) to our balance sheet at January 1, 2019:

| Balance Sheet | (Dollars in r | nillions) |
|-------------------------------|---------------|-----------|
| Property, plant and equipment | \$ | (409) |
| Deferred rent | | (3) |
| Long-term debt | | (558) |
| Deferred income taxes | | 37 |
| Stockholder's equity | | 115 |
| | | |

(2) Acquisition of Level 3

On November 1, 2017, CenturyLink acquired Level 3 through successive merger transactions, including a merger of Level 3 with and into a merger subsidiary, which survived such merger as our indirect wholly-owned subsidiary under the name of Level 3 Parent, LLC. We entered into this acquisition to, among other things, realize certain strategic benefits, including enhanced financial and operational scale, market diversification and an enhanced combined network. As a result of the acquisition, Level 3 shareholders received \$26.50 per share in cash and 1.4286 shares of CenturyLink common stock, with cash paid in lieu of fractional shares, for each outstanding share of Level 3 common stock they owned at closing, subject to certain limited exceptions. We issued this consideration with respect to all of the outstanding common stock of Level 3, with the exception of shares held by the dissenting common shareholders. Upon closing, CenturyLink shareholders owned approximately 51% and former Level 3 shareholders owned approximately 49% of the combined company.

In addition, each outstanding Level 3 restricted stock unit award granted prior to April 1, 2014 or granted to an outside director of Level 3 was converted into \$26.50 in cash and 1.4286 shares of CenturyLink common stock (and cash in lieu of fractional shares) with respect to each Level 3 share covered by such award (the "Converted RSU Awards"). Each outstanding Level 3 restricted stock unit award granted on or after April 1, 2014 (other than those granted to outside directors of Level 3) was converted into a CenturyLink restricted stock unit award using a conversion ratio of 2.8386 to 1 as determined in accordance with a formula set forth in the merger agreement ("the Continuing RSU Awards").

The aggregate consideration of \$19.6 billion is based on:

- the 517.3 million shares of CenturyLink's common stock (including those issued in connection with the Converted RSU Awards) issued
 to consummate the acquisition and the closing stock price of CenturyLink common stock at October 31, 2017 of \$18.99;
- a combination of (i) the cash consideration of \$26.50 per share on the 362.1 million common shares of Level 3 issued and outstanding as of October 31, 2017, (ii) the cash consideration of \$1 million paid on the Converted RSUs awards; and (iii) the estimated value of \$136 million the Continuing RSU Awards, which represents the pre-combination portion of Level 3's share-based compensation awards replaced by CenturyLink;
- \$60 million for the dissenting common shares issued and outstanding as of October 31, 2017; and
- our assumption at closing of approximately \$10.6 billion of Level 3's long-term debt.

The aggregate cash payments required to be paid on or about the closing date were funded with the proceeds of \$7.945 billion of term loans and \$400 million of funds borrowed under our new revolving credit facility together with other available funds, which included \$1.825 billion borrowed from Level 3 Parent, LLC. For additional information regarding CenturyLink's financing of the Level 3 acquisition see Note 6—Long-Term Debt and Credit Facilities.

We have recognized the assets and liabilities of Level 3 based on the fair value of the acquired tangible and intangible assets and assumed liabilities of Level 3 as of November 1, 2017, the consummation date of the acquisition, with the excess aggregate consideration recorded as goodwill. The estimation of such fair values and the estimation of lives of depreciable tangible assets and amortizable intangible assets required significant judgment. We completed our final fair value determination during the fourth quarter of 2018. Our final fair value determinations were different than those reflected in our consolidated financial statements at December 31, 2017.

In connection with receiving approval from the U.S. Department of Justice to complete the Level 3 acquisition we agreed to divest (i) certain Level 3 network assets in three metropolitan areas and (ii) 24 strands of dark fiber connecting 30 specified city-pairs across the United States. All of the metro network assets were classified as assets held for sale on our consolidated balance sheet as of December 31, 2017.

All of those assets were sold by December 31, 2018. The proceeds from these sales were included in the proceeds from sale of property, plant and equipment in our consolidated statements of cash flows. No gain or loss was recognized with these transactions.

As of October 31, 2018, the aggregate consideration exceeds the aggregate estimated fair value of the acquired assets and assumed liabilities by \$11.2 billion, which we have recognized as goodwill. The goodwill is attributable to strategic benefits, including enhanced financial and operational scale, market diversification and leveraged combined networks that we expect to realize. None of the goodwill associated with this acquisition is deductible for income tax purposes.

The following is our assignment of the aggregate consideration:

| | A | Adjusted November 1, 2017 | | Adjusted November 1, 2017 |
|---|-------------------|------------------------------|-----------------------|---------------------------|
| | Balance as of Pur | | Purchase Price | Balance as of |
| | | December 31, 2017 | Adjustments | October 31, 2018 |
| | | | (Dollars in millions) | |
| Cash, accounts receivable and other current assets (1) | \$ | 3,317 | (26) | 3,291 |
| Property, plant and equipment | | 9,311 | 157 | 9,468 |
| Identifiable intangible assets (2) | | | | |
| Customer relationships | | 8,964 | (533) | 8,431 |
| Other | | 391 | (13) | 378 |
| Other noncurrent assets | | 782 | 216 | 998 |
| Current liabilities, excluding current maturities of long-term debt | | (1,461) | (32) | (1,493) |
| Current maturities of long-term debt | | (7) | _ | (7) |
| Long-term debt | | (10,888) | _ | (10,888) |
| Deferred revenue and other liabilities | | (1,629) | (114) | (1,743) |
| Goodwill | | 10,837 | 340 | 11,177 |
| Total estimated aggregate consideration | \$ | 19,617 | (5) | 19,612 |

⁽¹⁾ Includes accounts receivable, which had a gross contractual value of \$884 million on November 1, 2017 and October 31, 2018.

On the acquisition date, we assumed Level 3's contingencies. For more information on our contingencies, see Note 17—Commitments, Contingencies and Other Items .

Acquisition-Related Expenses

We have incurred acquisition-related expenses related to our acquisition of Level 3. The table below summarizes our acquisition-related expenses, which consist of integration-related expenses, including severance and retention compensation expenses, and transaction-related expenses:

| | Years Ended December 31, | | | |
|------------------------------------|---------------------------|------|--|--|
| | 2018 | 2017 | | |
| | (Dollars in millions) | | | |
| Transaction-related expenses | \$ 2 | 174 | | |
| Integration-related expenses | 391 | 97 | | |
| Total acquisition-related expenses | \$ 393 | 271 | | |

At December 31, 2018, we had incurred cumulative acquisition-related expenses of \$716 million for Level 3. The total amounts of these expenses are included in our selling, general and administrative expenses.

⁽²⁾ The weighted-average amortization period for the acquired intangible assets is approximately 12.0 years .

Level 3 incurred transaction-related expenses of \$47 million on the date of acquisition. This amount is not included in our results of operations.

(3) Sale of Data Centers and Colocation Business

On May 1, 2017, we sold our data centers and colocation business to a consortium led by BC Partners, Inc. and Medina Capital in exchange for cash and a minority stake in the limited partnership that owns the consortium's newly-formed global secure infrastructure company, Cyxtera Technologies ("Cyxtera").

At the closing of this sale, we received pre-tax cash proceeds of \$1.8 billion, and we valued our minority stake at \$150 million, which was based upon the total amount of equity contributions to the limited partnership on the date made. Due to the sale and related restructuring actions we have taken regarding certain subsidiaries involved in the data centers and colocation business, we have estimated a cumulative current tax impact relating to the sale totaling \$65 million, \$18 million of which was accrued in 2016 and \$47 million of which was accrued in 2017.

In connection with our sale of the data centers and colocation business to Cyxtera, we agreed to lease back from Cyxtera a portion of the data center space to provide data hosting services to our customers. Because we have continuing involvement in the business through our minority stake in Cyxtera's parent, we did not meet the requirements for a sale-leaseback transaction as described in ASC 840-40, *Leases - Sale-Leaseback Transactions*. Under the failed-sale-leaseback accounting model, we were deemed under GAAP to still own certain real estate assets sold to Cyxtera, which we continued to reflect on our consolidated balance sheet and depreciate over the assets' remaining useful life. We also treated a certain amount of the pre-tax cash proceeds from the sale of the assets as though it were the result of a financing obligation on our consolidated balance sheet, and our consolidated results of operations included imputed revenue associated with the portion of the real estate assets that we did not lease back and imputed interest expense on the financing obligation. A portion of the rent payments under our leaseback arrangement with Cyxtera were recognized as a reduction of the financing obligation, resulting in lower recognized rent expense than the amounts actually paid each period.

The following table reflects the assets sold to and the liabilities assumed by Cyxtera on May 1, 2017, including the impact of failed-sale-leaseback:

| | Dolla | rs in millions |
|---|-------|----------------|
| Goodwill | \$ | 1,142 |
| Property, plant and equipment | | 1,051 |
| Other intangible assets | | 249 |
| Other assets | | 66 |
| Less assets not removed as a result of the failed-sale-leaseback | | (526) |
| Total net amount of assets derecognized | \$ | 1,982 |
| | | |
| Capital lease obligations | \$ | 294 |
| Other liabilities | | 274 |
| Less imputed financing obligations from the failed-sale-leaseback | | (628) |
| Total net imputed liabilities recognized | \$ | (60) |
| | | |

In addition, the failed-sale-leaseback accounting treatment had the following effects on our consolidated results of operations for the years ended December 31, 2018 and 2017:

| | Positive (Negative) Impact to Net Income December 31, | | |
|--|---|---------------|----------|
| | | | |
| | 2018 2 | | 2017 |
| | | (Dollars in m | illions) |
| Increase in revenue | \$ | 74 | 49 |
| Decrease in cost of sales | | 22 | 15 |
| Increase in loss on sale of business included in selling, general and administrative expense | | _ | (102) |
| Increase in depreciation expense (one-time) | | _ | (44) |
| Increase in depreciation expense (ongoing) | | (69) | (47) |
| Increase in interest expense | | (55) | (39) |
| Decrease in income tax expense | | 7 | 65 |
| Decrease in net income | \$ | (21) | (103) |

After factoring in the costs to sell the data centers and colocation business, excluding the impact from the failed-sale-leaseback accounting treatment, the sale resulted in a \$20 million gain as a result of the aggregate value of the proceeds we received exceeding the carrying value of the assets sold and liabilities assumed. Based on the fair market values of the failed-sale-leaseback assets, the failed-sale-leaseback accounting treatment resulted in a loss of \$102 million as a result of the requirement to treat a certain amount of the pre-tax cash proceeds from the sale of the assets as though it were the result of a financing obligation. The combined net loss of \$82 million was included in selling, general and administrative expenses in our consolidated statement of operations for the year ended December 31, 2017. The sale also resulted in a significant capital loss carryforward, which was entirely offset by a valuation allowance due to our determination that we are not likely to be able to utilize this carryforward prior to its expiration.

We evaluated our minority stake in the limited partnership and determined that we were not the primary beneficiary of the entity. As a result, we classified our \$150 million investment in the limited partnership in other assets on our consolidated balance sheet as of December 31, 2018 and 2017. In addition to our investment, we have a \$3 million payable to Cyxtera, classified in other accrued liabilities on our consolidated balance sheet as of December 31, 2018. We had a receivable for \$10 million and \$49 million, classified in other current assets on our consolidated balance sheet as of December 31, 2018 and 2017, respectively as a result of amounts charged for the lease of data center space. We will continue to have an ongoing obligation to Cyxtera related to our lease of data center space from them. During the year ended December 31, 2018, we paid rent to Cyxtera totaling \$132 million. From May 1, 2017 through December 31, 2017, we paid rent to Cyxtera totaling \$80 million.

Effective November 3, 2016, which is the date we entered into the agreement to sell our data centers and colocation business, we ceased recording depreciation of the property, plant and equipment to be sold and amortization of the business's intangible assets in accordance with applicable accounting rules. Otherwise, we estimate that we would have recorded additional depreciation and amortization expense of \$67 million from January 1, 2017 through May 1, 2017.

Upon adopting ASU 2016-02, accounting for the failed sale leaseback will no longer be applicable based on our facts and circumstances, and the real estate assets and corresponding financing obligation will be derecognized from our consolidated financial statements. Please see "Leases" (ASU 2016-02) in Note 1— Background and Summary of Significant Accounting Policies for additional information on the impact the new lease standard will have on the accounting for the failed-sale-leaseback.

(4) Goodwill, Customer Relationships and Other Intangible Assets

Goodwill, customer relationships and other intangible assets consisted of the following:

| | As of December 31, | | | |
|--|-----------------------|--------|--------|--|
| | 2018 | | 2017 | |
| | (Dollars in millions) | | | |
| Goodwill | \$ | 28,031 | 30,475 | |
| Customer relationships, less accumulated amortization of \$8,492 and \$7,096 | \$ | 8,911 | 10,876 | |
| Indefinite-life intangible assets | \$ | 269 | 269 | |
| Other intangible assets subject to amortization: | | | | |
| Capitalized software, less accumulated amortization of \$2,616 and \$2,294 | \$ | 1,468 | 1,469 | |
| Trade names, less accumulated amortization of \$61 and \$31 | | 131 | 159 | |
| Total other intangible assets, net | \$ | 1,868 | 1,897 | |

Our goodwill was derived from numerous acquisitions where the purchase price exceeded the fair value of the net assets acquired (including the acquisition described in Note 2—Acquisition of Level 3). At December 31, 2018, the net carrying amounts of goodwill, customer relationships and other intangibles assets included \$19.1 billion as a result of our Level 3 acquisition. As of December 31, 2018, the weighted average remaining useful lives of the intangible assets acquired in the acquisition of Level 3 was approximately 11 years in total, approximately 11 years for customer relationships, 3 years for capitalized software and 4 years for trade names.

Total amortization expense for intangible assets for the years ended December 31, 2018, 2017 and 2016 was \$1.8 billion, \$1.2 billion and \$1.2 billion, respectively. As of December 31, 2018, the gross carrying amount of goodwill, customer relationships, indefinite-life and other intangible assets was \$50.0 billion.

We estimate that total amortization expense for intangible assets for the years ending December 31, 2019 through 2023 will be as follows:

| | (Dollars | (Dollars in millions) | |
|------|----------|-----------------------|--|
| 2019 | \$ | 1,691 | |
| 2020 | | 1,589 | |
| 2021 | | 1,156 | |
| 2022 | | 985 | |
| 2023 | | 893 | |

We assess our goodwill and other indefinite-lived intangible assets for impairment annually, or, under certain circumstances, more frequently, such as when events or changes in circumstances indicate there may be impairment. We are required to write down the value of goodwill only when our assessment determines the recorded amount of goodwill exceeds the fair value. Our annual impairment assessment date for goodwill is October 31, at which date we assessed our reporting units, which were consumer, medium and small business, enterprise, international and global accounts, and wholesale and indirect. Our annual impairment assessment date for indefinite-lived intangible assets other than goodwill is December 31.

Our reporting units are not discrete legal entities with discrete full financial statements. Our assets and liabilities are employed in and relate to the operations of multiple reporting units. For each reporting unit, we compare its estimated fair value of equity to its carrying value of equity that we assign to the reporting unit. If the estimated fair value of the reporting unit is greater than the carrying value, we conclude that no impairment exists. If the estimated fair value of the reporting unit is less than the carrying value, we record an impairment equal to the excess amount.

At October 31, 2018, we estimated the fair value of our five above-mentioned reporting units by considering both a market approach and a discounted cash flow method. The market approach method includes the use of comparable multiples of publicly traded companies whose services are comparable to ours. The discounted cash flow method is based on the present value of projected cash flows and a terminal value, which represents the expected normalized cash flows of the reporting units beyond the cash flows from the discrete projection period. We reconciled the estimated fair values of the reporting units to our market capitalization as of October 31, 2018 and concluded that the indicated control premium of approximately 0.1% was reasonable based on recent transactions in the market place. As of October 31, 2018, based on our assessment performed with respect to these reporting units as described above, we concluded that the estimated fair value of our consumer reporting unit was less than our carrying value of equity by approximately \$2.7 billion. As a result, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill assigned to our consumer reporting unit during the fourth quarter of 2018. In addition, based on our assessments performed, we concluded that the goodwill for our four remaining reporting units was not impaired as of October 31, 2018.

As of October 31, 2018, based on our assessment performed with respect to our four reporting units included in our business segment, the estimated fair value of our equity exceeded our carrying value of equity for our medium and small business, enterprise, international and global accounts and wholesale and indirect by 2%, 11%, 30% and 5%, respectively. After the impairment charge described above, the estimated fair value of equity for our consumer reporting unit equals the carrying value of equity for such unit.

As of October 31, 2017, based on our assessments performed, we concluded that our goodwill for our then three reporting units was not impaired as of this date.

We completed our qualitative assessment of our indefinite-lived intangible assets other than goodwill as of December 31, 2018 and 2017 and concluded it is more likely than not that our indefinite-lived intangible assets are not impaired; thus, no impairment charge was recorded in 2018 or 2017.

The following table shows the rollforward of goodwill assigned to our reportable segments from December 31, 2016 through December 31, 2018.

| | Business | | Consumer | Total | |
|--|----------|--------|-----------------------|-------|---------|
| | | | (Dollars in millions) | | |
| As of December 31, 2016 (1) | \$ | 9,372 | 10,278 | | 19,650 |
| Purchase accounting and other adjustments | | 10,825 | _ | | 10,825 |
| As of December 31, 2017 (1) | | 20,197 | 10,278 | | 30,475 |
| Purchase accounting and other adjustments (2)(3) | | 250 | 32 | | 282 |
| Impairment | | _ | (2,726) | | (2,726) |
| As of December 31, 2018 | \$ | 20,447 | 7,584 | \$ | 28,031 |

- (1) Goodwill is net of accumulated impairment losses of \$1.1 billion that related to our former hosting segment now included in our business segment.
- (2) We allocated \$32 million of Level 3 goodwill to consumer as we expect the consumer segment to benefit from synergies resulting from the business combination.
- (3) Includes \$58 million decrease due to effect of foreign currency exchange rate change.

As of December 31, 2018, our goodwill has been allocated to our reporting units as follows:

| | (Dollars | (Dollars in millions) | |
|-----------------------------------|----------|-----------------------|--|
| Medium and small business | \$ | 5,193 | |
| Enterprise | | 5,222 | |
| International and global accounts | | 3,596 | |
| Wholesale and indirect | | 6,436 | |
| Total business segment | | 20,447 | |
| Consumer | | 7,584 | |
| Total goodwill | \$ | 28,031 | |

For additional information on our segments, see Note 15—Segment Information .

(5) Revenue Recognition

Comparative Results

The following tables present our reported results under ASC 606 and a reconciliation to results using the historical accounting method:

| | Year Ended December 31, 2018 | | | |
|--|---|----------------------|---|--|
| | Reported Balances | Impact of ASC 606 | ASC 605 Historical Adjusted Balances | |
| | (Dollars in millions, except per share amour and shares in thousands) | | | |
| Operating revenue | \$ 23,443 | 39 | 23,482 | |
| Cost of services and products (exclusive of depreciation and amortization) | 10,862 | 22 | 10,884 | |
| Selling, general and administrative | 4,165 | 71 | 4,236 | |
| Interest expense | 2,177 | (9) | 2,168 | |
| Income tax expense | 170 | (12) | 158 | |
| Net loss | (1,733) | (33) | (1,766) | |
| | | | | |
| BASIC AND DILUTED LOSS PER COMMON SHARE | | | | |
| BASIC | \$ (1.63) | (0.03) | (1.66) | |
| DILUTED | \$ (1.63) | (0.03) | (1.66) | |
| WEIGHTED-AVERAGE COMMON SHARES OUTSTANDING | | | | |
| BASIC | 1,065,866 | _ | 1,065,866 | |
| DILUTED | 1,065,866 | _ | 1,065,866 | |

The following table presents a reconciliation of certain consolidated balance sheet captions under ASC 606 to the balance sheet results using the historical accounting method:

| | | As of December 31, 2018 | | | | |
|-----------------------------|-----|-------------------------|---------|-----------------------|--|--|
| | | Reported Balances | | Impact of ASC 606 | ASC 605 Historical Adjusted Balances | |
| | | | | (Dollars in millions) | | |
| Other current assets | | \$ | 918 | (172) | 746 | |
| Other long-term assets, net | | | 1,060 | (112) | 948 | |
| Deferred revenue | | | 2,375 | 71 | 2,446 | |
| Deferred income taxes, net | | | 2,395 | (131) | 2,264 | |
| Other long-term liabilities | | | 1,099 | 147 | 1,246 | |
| Accumulated deficit | | | (1,643) | (371) | (2,014) | |
| | 107 | | | | | |

Disaggregated Revenue by Service Offering

The following tables provide disaggregation of revenue from contracts with customers based on service offerings for the year ended December 31, 2018, respectively. It also shows the amount of revenue that is not subject to ASC 606, but is instead governed by other accounting standards.

| | | Yea | 2018 | |
|---|----|---------------|--|---|
| | | Total Revenue | Adjustments for Non-ASC 606 Revenue ⁽⁸⁾ | Total Revenue from Contracts with Customers |
| | | | (Dollars in millions) | |
| Business segment | | | | |
| IP and Data Services (1) | \$ | 6,971 | _ | 6,971 |
| Transport and Infrastructure (2) | | 5,356 | (569) | 4,787 |
| Voice and Collaboration (3) | | 4,401 | _ | 4,401 |
| IT and Managed Services (4) | | 621 | _ | 621 |
| Total business segment revenue | _ | 17,349 | (569) | 16,780 |
| Consumer segment | | | | |
| IP and Data Services (5) | | 308 | (33) | 275 |
| Transport and Infrastructure (6) | | 2,892 | (213) | 2,679 |
| Voice and Collaboration (3) | | 2,171 | _ | 2,171 |
| Total consumer segment revenue | _ | 5,371 | (246) | 5,125 |
| Non-segment revenue | | | | |
| Regulatory Revenue (7) | | 723 | (723) | _ |
| Total non-segment revenue | _ | 723 | (723) | _ |
| Total revenue | \$ | 23,443 | (1,538) | 21,905 |
| | = | | <u> </u> | |
| Timing of Revenue | | | | |
| Goods and services transferred at a point in time | | | | \$ 230 |
| Services performed over time | | | | 21,675 |
| Total revenue from contracts with customers | | | | \$ 21,905 |

- (1) Includes primarily VPN data network, Ethernet, IP, video and ancillary revenue.
- (2) Includes primarily broadband, private line (including business data services), colocation and data centers, wavelength and ancillary revenue.
- (3) Includes local, long-distance and other ancillary revenue.
- (4) Includes IT services and managed services revenue.
- (5) Includes retail video revenue (including our facilities-based video revenue).
- (6) Includes primarily broadband and equipment sales and professional services revenue.
- (7) Includes CAF Phase I, CAF Phase II, federal and state USF support revenue, sublease rental income and failed-sale leaseback income.
- (8) Includes regulatory revenue, lease revenue, sublease rental income, revenue from fiber capacity lease arrangements and failed sale leaseback income, which are not within the scope of ASC 606.

Customer Receivables and Contract Balances

The following table provides balances of customer receivables, contract assets and contract liabilities as of December 31, 2018 and January 1, 2018:

| | Dec | cember 31, 2018 Ja | anuary 1, 2018 | |
|--------------------------|-----|-----------------------|----------------|--|
| | | (Dollars in millions) | | |
| Customer receivables (1) | \$ | 2,346 | 2,504 | |
| Contract liabilities | | 860 | 904 | |
| Contract assets | | 140 | 145 | |

(1) Gross customer receivables of \$2.5 billion and \$2.7 billion, net of allowance for doubtful accounts of \$132 million and \$155 million, at December 31, 2018 and January 1, 2018, respectively.

Contract liabilities are consideration we have received from our customers or billed in advance of providing goods or services promised in the future. We defer recognizing this consideration as revenue until we have satisfied the related performance obligation to the customer. Contract liabilities include recurring services billed one month in advance and installation and maintenance charges that are deferred and recognized over the actual or expected contract term, which ranges from one to seven years depending on the service. Contract liabilities are included within deferred revenue in our consolidated balance sheet.

The following table provides information about revenue recognized for the year ended December 31, 2018:

Revenue recognized in the period from:

Amounts included in contract liability at the beginning of the period (January 1, 2018)

Performance obligations satisfied in previous periods

(Dollars in millions)

\$ 295

Performance Obligations

As of December 31, 2018, our estimated revenue expected to be recognized in the future related to performance obligations associated with customer contracts that are unsatisfied (or partially satisfied) is approximately \$6.9 billion. We expect to recognize approximately 83% of this revenue through 2021, with the balance recognized thereafter.

We do not disclose the value of unsatisfied performance obligations for contracts for which we recognize revenue at the amount to which we have the right to invoice for services performed (for example, uncommitted usage or non-recurring charges associated with professional or technical services to be completed), or contracts that are classified as leasing arrangements that are not subject to ASC 606.

Contract Costs

The following table provides changes in our contract acquisition costs and fulfillment costs:

| | | Year Ended December 31, 2018 | | |
|----------------------------|----|------------------------------|-------------------|--|
| | Ac | quisition Costs | Fulfillment Costs | |
| | | (Dollars in millions) | | |
| eginning of period balance | \$ | 268 | 133 | |
| osts incurred | | 226 | 146 | |
| Amortization | | (172) | (92) | |
| End of period balance | \$ | 322 | 187 | |

Acquisition costs include commission fees paid to employees as a result of obtaining contracts. Fulfillment costs include third party and internal costs associated with the provision, installation and activation of telecommunications services to customers, including labor and materials consumed for these activities.

Deferred acquisition and fulfillment costs are amortized based on the transfer of services on a straight-line basis over the average customer life of 30 months for consumer customers and 12 to 60 months for business customers and amortized fulfillment costs are included in cost of services and products and amortized acquisition costs are included in selling, general and administrative expenses in our consolidated statement of operations. The amount of these deferred costs that are anticipated to be amortized in the next twelve months are included in other current assets on our consolidated balance sheets. The amount of deferred costs expected to be amortized beyond the next twelve months is included in other non-current assets on our consolidated balance sheets. Deferred acquisition and fulfillment costs are assessed for impairment on an annual basis.

(6) Long-Term Debt and Credit Facilities

The following chart reflects the consolidated long-term debt of CenturyLink, Inc. and its subsidiaries, including unamortized discounts and premiums and unamortized debt issuance costs, but excluding intercompany debt:

| | | | As of Decemb | ember 31, |
|---|--------------------|-------------|--------------------|-----------|
| | Interest Rates (1) | Maturities | 2018 | 2017 |
| | | | (Dollars in mi | llions) |
| Senior Secured Debt: (2) | | | | |
| CenturyLink, Inc. | | | | |
| 2017 Revolving Credit Facility (3) | 5.130% - 7.250% | 2022 | \$ 550 | 405 |
| Term Loan A | 5.272% | 2022 | 1,622 | 1,575 |
| Term Loan A-1 | 5.272% | 2022 | 351 | 370 |
| Term Loan B | 5.272% | 2025 | 5,940 | 6,000 |
| Subsidiaries: | | | | |
| Level 3 Financing, Inc. | | | | |
| Tranche B 2024 Term Loan (4) | 4.754% | 2024 | 4,611 | 4,611 |
| Embarq Corporation subsidiaries | | | | |
| First mortgage bonds | 7.125% - 8.375% | 2023 - 2025 | 138 | 151 |
| Senior Notes and Other Debt: | | | | |
| CenturyLink, Inc. | | | | |
| Senior notes | 5.625% - 7.650% | 2019 - 2042 | 8,036 | 8,125 |
| Subsidiaries: | | | | |
| Level 3 Financing, Inc. | | | | |
| Senior notes | 5.125% - 6.125% | 2021 - 2026 | 5,315 | 5,315 |
| Level 3 Parent, LLC | | | | |
| Senior notes | 5.750% | 2022 | 600 | 600 |
| Qwest Corporation | | | | |
| Senior notes | 6.125% - 7.750% | 2021 - 2057 | 5,956 | 7,294 |
| Term loan | 4.530% | 2025 | 100 | 100 |
| Qwest Capital Funding, Inc. | | | | |
| Senior notes | 6.875% - 7.750% | 2021 - 2031 | 697 | 981 |
| Embarq Corporation and subsidiary | | | | |
| Senior note | 7.995% | 2036 | 1,485 | 1,485 |
| Other | 9.000% | 2019 | 150 | 150 |
| Capital lease and other obligations | Various | Various | 801 | 891 |
| Unamortized (discounts) premiums and other, net | | | (8) | 23 |
| Unamortized debt issuance costs | | | (283) | (350) |
| Total long-term debt | | | 36,061 | 37,726 |
| Less current maturities | | | (652) | (443) |
| Long-term debt, excluding current maturities | | | \$ 35,409 | 37,283 |

⁽¹⁾ As of December 31, 2018.

⁽²⁾ See the remainder of this Note for a description of certain parent or subsidiary guarantees and liens securing this debt.

⁽³⁾ The aggregate amount outstanding on our 2017 revolving credit facility at December 31, 2018 was \$550 million with a weighted-average interest rate of 5.322%. These amounts typically change on a regular basis.

⁽⁴⁾ The Tranche B 2024 Term Loan had an interest rate of 4.754% as of December 31, 2018 and 3.557% as of December 31, 2017 .

Debt of CenturyLink, Inc. and its Subsidiaries

At December 31, 2018, most of our outstanding consolidated debt had been incurred by CenturyLink, Inc. or one of the following four other primary borrowers or "borrowing groups," each of which has borrowed funds either on a standalone basis or as part of a separate restricted group with certain of its subsidiaries:

- Qwest Corporation;
- Qwest Capital Funding, Inc. (including its parent guarantor, Qwest Communications International Inc.);
- Embarg Corporation; and
- · Level 3 Parent, LLC (including its finance subsidiary, Level 3 Financing, Inc.).

Each of these borrowers or borrowing groups has entered into one or more credit agreements with certain financial institutions or other institutional lenders, or issued senior notes. Certain of these debt instruments are described further below.

2017 CenturyLink Credit Agreement

In connection with financing its acquisition of Level 3 on November 1, 2017, CenturyLink, Inc. caused its wholly-owned subsidiary, CenturyLink Escrow, LLC, to enter into a credit agreement on June 19, 2017 (the "2017 CenturyLink Credit Agreement") with, among others, Bank of America, N.A., as administrative agent and collateral agent, currently providing for \$10.245 billion in senior secured credit facilities (the "2017 Senior Secured Credit Facilities"). As amended in early 2018, these facilities currently consist of the following:

- a \$2.168 billion revolving credit facility ("2017 Revolving Credit Facility"), with 18 lenders, each with allocations ranging from \$36.4 million to \$167.8 million:
- a \$1.707 billion senior secured Term Loan A credit facility, with 18 lenders, each with commitments ranging from \$28.6 million to \$132.2 million:
- a \$370 million senior secured Term Loan A-1 credit facility with CoBank, ACB; and
- a \$6.0 billion senior secured Term Loan "B" credit facility, the proceeds of which were fully pre-funded to us, net of a discount, into escrow on June 19, 2017 and released to us on November 1, 2017.

Loans under the Term Loan A and A-1 facilities and the 2017 Revolving Credit Facility bear interest at a rate equal to, at our option, the London Interbank Offered Rate ("LIBOR") or the alternative base rate (each as defined in the 2017 CenturyLink Credit Agreement) plus an applicable margin between 2.25% to 3.00% per annum for LIBOR loans and 1.25% to 2.00% per annum for alternative base rate loans, depending on our then current total leverage ratio. Borrowings under the Term Loan B facility bore interest at 1.375% per annum through July 18, 2017 and at 2.75% per annum thereafter through November 1, 2017. Subsequent to November 1, 2017, borrowings under the Term Loan B facility have borne interest at LIBOR plus 2.75% per annum. Loans under each of the term loan facilities require certain specified quarterly amortization payments and certain specified mandatory prepayments in connection with certain asset sales and debt issuances and out of excess cash flow, among other things, subject in each case to certain significant exceptions.

The 2017 Revolving Credit Facility and borrowings under the Term Loan A and A-1 facilities will mature on November 1, 2022. Borrowings under the Term Loan B facility will mature on January 31, 2025.

On November 1, 2017, CenturyLink, Inc. assumed all rights and obligations under the 2017 CenturyLink Credit Agreement, including the right to borrow funds under the 2017 Revolving Credit Facility on the terms and conditions specified in the 2017 CenturyLink Credit Agreement.

All of CenturyLink, Inc.'s obligations under the 2017 Senior Secured Credit Facilities are guaranteed by certain of its subsidiaries. The guarantees by certain of those guarantors are secured by a first priority security interest in substantially all assets (including certain subsidiaries stock) directly owned by them, subject to certain exceptions and limitations.

A portion of the 2017 Revolving Credit Facility in an amount not to exceed \$100 million is available for swingline loans, and a portion in an amount not to exceed \$400 million is available for the issuance of letters of credit.

CenturyLink, Inc. is permitted under the 2017 CenturyLink Credit Agreement to request certain incremental borrowings subject to the satisfaction of various conditions and to certain other limitations. Any incremental borrowings would be subject to the same terms and conditions under the 2017 CenturyLink Credit Agreement.

Term Loans and Certain Other Debt of Subsidiaries

Qwest Corporation

In 2015, Qwest Corporation entered into a term loan in the amount of \$100 million with CoBank, ACB. The outstanding unpaid principal amount of this term loan plus any accrued and unpaid interest is due on February 20, 2025. Interest is paid at least quarterly based upon either the London Interbank Offered Rate ("LIBOR") or the base rate (as defined in the credit agreement) plus an applicable margin between 1.50% to 2.50% per annum for LIBOR loans and 0.50% to 1.50% per annum for base rate loans depending on Qwest Corporation's then current senior unsecured long-term debt rating. At both December 31, 2018 and 2017, the outstanding principal balance on this term loan was \$100 million.

Level 3 Financing, Inc.

At December 31, 2018 and 2017, Level 3 Financing, Inc. owed \$4.611 billion under the Tranche B 2024 Term Loan, which matures on February 22, 2024. The Tranche B 2024 Term Loan carries an interest rate, in the case of base rate borrowings, equal to (i) the greater of the Prime Rate, the Federal Funds Effective Rate plus 50 basis points, or LIBOR plus 100 basis points (with all such terms and calculations as defined or further specified in the applicable credit agreement) plus (ii) 1.25% per annum. Any Eurodollar borrowings under the Tranche B 2024 Term Loan bear interest at LIBOR plus 2.25% per annum.

The Tranche B 2024 Term Loan requires certain specified mandatory prepayments in connection with certain asset sales and other transactions, subject to certain significant exceptions. The obligations of Level 3 Financing, Inc. under the Tranche B 2024 Term Loan are, subject to certain exceptions, secured by certain assets of Level 3 Parent, LLC and certain of its material domestic telecommunication subsidiaries. Also, Level 3 Parent, LLC and certain of its subsidiaries have guaranteed the obligations of Level 3 Financing, Inc. under the Tranche B 2024 Term Loan. Level 3 Communications, LLC and its material domestic subsidiaries guarantee and pledge certain of their assets to secure the obligations of Level 3 Financing, Inc. under the Tranche B 2024 Term Loan.

Embarq Subsidiaries

At December 31, 2018 and 2017, one of our Embarq subsidiaries had outstanding first mortgage bonds. This first mortgage bond is secured by substantially all of the property, plant and equipment of the issuing subsidiary.

Revolving Letters of Credit

We use various financial instruments in the normal course of business. These instruments include letters of credit, which are conditional commitments issued on our behalf in accordance with specified terms and conditions. CenturyLink, Inc. maintains an uncommitted \$225 million revolving letter of credit facility separate from the letter of credit facility included in the 2017 Revolving Credit Facility noted above. Letters of credit issued under this facility are backed by credit enhancements in the form of secured guarantees issued by certain CenturyLink subsidiaries. As of December 31, 2018 and 2017, CenturyLink, Inc.'s outstanding letters of credit under this credit facility totaled \$97 million and \$104 million, respectively.

As of December 31, 2018, Level 3 Parent, LLC had outstanding letters of credit or other similar obligations of approximately \$30 million of which \$24 million is collateralized by cash that is reflected on the consolidated balance sheets in restricted cash and securities.

As of December 31, 2017, Level 3 Parent, LLC had outstanding letters of credit or other similar obligations of approximately \$36 million of which \$30 million is collateralized by cash that is reflected on the consolidated balance sheets in restricted cash and securities.

Senior Notes

CenturyLink, Inc., Level 3 Financing, Inc., Level 3 Parent, LLC, Qwest Corporation, Qwest Capital Funding, Inc. and Embarq Corporation have each issued unsecured senior notes. All of these notes carry fixed interest rates and all principal is due on the notes' respective maturity dates, which rates and maturity dates are summarized in the table above. The senior notes issued by Level 3 Financing, Inc. are guaranteed by its parent, Level 3 Parent, LLC and another of its affiliates. The senior notes issued by Qwest Capital Funding, Inc. are guaranteed by its parent, Qwest Communications International Inc. Except for a limited number of senior notes issued by Qwest Corporation, the issuer generally can redeem the notes, at its option, in whole or in part, (i) pursuant to a fixed schedule of pre-established redemption prices, (ii) pursuant to a "make whole" redemption price or (iii) under certain other specified limited conditions. Under certain circumstances in connection with a "change of control" of CenturyLink, Inc., it will be required to make an offer to repurchase each series of these senior notes (other than two of its older series of notes) at a price of 101% of the principal amount redeemed, plus accrued and unpaid interest. Also, under certain circumstances in connection with a "change of control" of Level 3 Parent, LLC, it, as well as Level 3 Financing, Inc., will be required to make an offer to repurchase each series of its outstanding senior notes at a price of 101% of the principal amount redeemed, plus accrued and unpaid interest.

New Issuances

As described above under "2017 CenturyLink Credit Agreement", on June 19, 2017, CenturyLink, Inc. caused one of its wholly-owned subsidiaries to enter into the 2017 CenturyLink Credit Agreement currently providing for \$10.2 billion of senior secured credit facilities. Upon the execution of the 2017 CenturyLink Credit Agreement, the \$6.0 billion Term Loan B credit facility was fully funded. On November 1, 2017, CenturyLink, Inc. assumed the obligations and borrowed additional sums under such credit agreement.

On April 27, 2017, Qwest Corporation issued \$575 million aggregate principal amount of 6.75% Notes due 2057 and, on May 5, 2017, issued an additional \$85 million aggregate principal amount of such notes pursuant to an over-allotment option in exchange for aggregate net proceeds, after deducting underwriting discounts and other expenses, of \$638 million . All of the 6.75% Notes are senior unsecured obligations and may be redeemed by Qwest Corporation, in whole or in part, on or after June 15, 2022, at a redemption price equal to 100% of the principal amount redeemed plus accrued and unpaid interest to the redemption date.

Repayments

2018

During 2018, CenturyLink and its affiliates redeemed approximately \$1.7 billion in debt securities, which primarily included approximately \$174 million of Qwest Capital Financing 6.5% Notes due 2018, approximately \$164 million of Qwest Corporation 7.5% Notes due 2051, approximately \$925 million of Qwest Corporation 7.0% Notes due 2052 and approximately \$250 million of Qwest 7.25% Notes due 2035.

2017

As described above under "2017 CenturyLink Credit Agreement", on November 1, 2017, CenturyLink, Inc. repaid the outstanding principal amount of \$319 million under its 2012 term loan.

During 2017, subsidiaries of Embarq Corporation paid at maturity the \$72 million principal amount and accrued and unpaid interest due under their 8.77% Notes. CenturyLink, Inc. paid at maturity the \$350 million principal and accrued and unpaid interest due under its 5.15% Notes. Qwest Corporation redeemed \$125 million aggregate principal amount of the remaining \$288 million of its 7.5% Notes due 2051. Qwest Corporation redeemed all \$500 million of its 6.5% Notes due 2017. CenturyLink, Inc. paid at maturity the \$500 million principal and accrued and unpaid interest due under its 6.00% Notes.

Long-Term Debt Maturities

Set forth below is the aggregate principal amount of our long-term debt (excluding unamortized discounts and premiums, net and unamortized debt issuance costs) maturing during the following years:

| | (Dollars | in millions) ⁽¹⁾ |
|----------------------|----------|-----------------------------|
| 2019 | \$ | 652 |
| 2020 | | 1,205 |
| 2021 | | 3,115 |
| 2022 | | 5,283 |
| 2023 | | 2,096 |
| 2024 and thereafter | | 23,503 |
| Total long-term debt | \$ | 35,854 |

⁽¹⁾ In Note 3—Sale of Data Centers and Colocation Business, we describe an imputed financing obligation. The amount outstanding on that imputed financing obligation at December 31, 2018 was \$558 million. The aggregate maturities of long-term debt do not include \$499 million of this obligation, which prior to the end of the lease term on April 30, 2020, will be derecognized along with the remaining net book value of the associated real estate assets.

Interest Expense

Interest expense includes interest on total long-term debt. The following table presents the amount of gross interest expense, net of capitalized interest:

| | Years Ended December 31, | | | |
|------------------------|---------------------------|-------|-------|--|
| | 2018 | 2017 | 2016 | |
| | (Dollars in millions) | | | |
| Interest expense: | | | | |
| Gross interest expense | \$ 2,230 | 1,559 | 1,372 | |
| Capitalized interest | (53) | (78) | (54) | |
| Total interest expense | \$ 2,177 | 1,481 | 1,318 | |

Covenants

CenturyLink, Inc.

With respect to the Term Loan A and A-1 facilities and the 2017 Revolving Credit Facility, the 2017 CenturyLink Credit Agreement requires us to maintain (i) a maximum total leverage ratio of not more than 5.00 to 1.00 until November 1, 2019 and 4.75 to 1.00 thereafter and (ii) a minimum consolidated interest coverage ratio of at least 2.00 to 1.00, with such ratios being determined and calculated in the manner described in the 2017 CenturyLink Credit Agreement.

The 2017 Senior Secured Credit Facilities contain various representations and warranties and extensive affirmative and negative covenants. Such covenants include, among other things and subject to certain significant exceptions, restrictions on our ability to declare or pay dividends, repurchase stock, repay certain other indebtedness, create liens, incur additional indebtedness, make investments, engage in transactions with its affiliates, dispose of assets and merge or consolidate with any other person.

The senior notes of CenturyLink, Inc. were issued under an indenture dated March 31, 1994. This indenture restricts our ability to (i) incur, issue or create liens upon the property of CenturyLink, Inc. and (ii) consolidate with or merge into, or transfer or lease all or substantially all of our assets to any other party. The indenture does not contain any provisions that are impacted by our credit ratings or that restrict the issuance of new securities in the event of a material adverse change to us. However, as indicated above under "Senior Notes", CenturyLink, Inc. will be required to offer to purchase certain of its long-term debt securities issued under this indenture under certain circumstances in connection with a "change of control" of CenturyLink, Inc.

Level 3 Companies

The term loan and senior notes of Level 3 Parent, LLC and Level 3 Financing, Inc. contain various representations and extensive affirmative and negative covenants. Such covenants include, among other things and subject to certain significant exceptions, restrictions on their ability to declare or pay dividends, repay certain other indebtedness, create liens, incur additional indebtedness, make investments, engage in transactions with their affiliates, dispose of assets and merge or consolidate with any other person. Also, as indicated above under "Senior Notes", Level 3 Parent, LLC, as well as Level 3 Financing, Inc., will be required to offer to purchase certain of its long-term debt securities under certain circumstances in connection with a "change of control" of Level 3 Parent, LLC.

Qwest Companies

Under its term loan, Qwest Corporation must maintain a debt to EBITDA (earnings before interest, taxes, depreciation and amortization, as defined in such term loan documentation) ratio of not more than 2.85:1.0, as of the last day of each fiscal quarter for the four quarters then ended. The term loan also contains a negative pledge covenant, which generally requires Qwest Corporation to secure equally and ratably any advances under the term loan if it pledges assets or permit liens on its property for the benefit of other debtholders.

The senior notes of Qwest Corporation were issued under indentures dated April 15, 1990 and October 15, 1999. These indentures contain restrictions on the incurrence of liens and the consummation of certain transactions substantially similar to the above-described covenants in CenturyLink, Inc.'s March 31, 1994 indenture (but contain no mandatory repurchase provisions). The senior notes of Qwest Capital Funding, Inc. were issued under an indenture dated June 29, 1998 containing terms substantially similar to those set forth in Qwest Corporation's indentures.

Embarg

Embarq's senior note was issued pursuant to an indenture dated as of May 17, 2006. While Embarq is generally prohibited from creating liens on its property unless its senior notes are secured equally and ratably, Embarq can create liens on its property without equally and ratably securing its senior notes so long as the sum of all indebtedness so secured does not exceed 15% of Embarq's consolidated net tangible assets. The indenture also contains restrictions on the consummation of certain transactions substantially similar to CenturyLink, Inc.'s above-described covenants (but without mandatory repurchase provision), as well as certain customary covenants to maintain properties and pay all taxes and lawful claims.

Impact of Covenants

The debt covenants applicable to CenturyLink, Inc. and its subsidiaries could materially adversely affect their ability to operate or expand their respective businesses, to pursue strategic transactions, or to otherwise pursue their plans and strategies. The covenants of the Level 3 companies may significantly restrict the ability of CenturyLink, Inc. to receive cash from the Level 3 companies, to distribute cash from the Level 3 companies to other of CenturyLink, Inc.'s affiliated entities, or to enter into other transactions among CenturyLink, Inc.'s wholly-owned entities.

Certain of the debt instruments of CenturyLink, Inc. and its subsidiaries contain cross payment default or cross acceleration provisions. When present, these provisions could have a wider impact on liquidity than might otherwise arise from a default or acceleration of a single debt instrument.

The ability of CenturyLink, Inc. and its subsidiaries to comply with the financial covenants in their respective debt instruments could be adversely impacted by a wide variety of events, including unforeseen contingencies, many of which are beyond their control.

Compliance

At December 31, 2018, CenturyLink, Inc. believes it and its subsidiaries were in compliance with the provisions and financial covenants contained in their respective material debt agreements in all material respects.

Guarantees

CenturyLink, Inc. does not guarantee the debt of any unaffiliated parties, but, as noted above, certain of its largest subsidiaries guarantee (i) its debt and letters of credit outstanding under its 2017 CenturyLink Credit Agreement and its \$225 million revolving letter of credit facility and (ii) the outstanding term loans or senior notes issued by certain other subsidiaries. As further noted above, several of the subsidiaries guaranteeing these obligations have pledged substantially all of their assets to secure their respective guarantees.

Subsequent Event

On February 15, 2019, the Company entered into an aggregate \$2.5 billion in floating-to-fixed interest rate swap agreements with five banks. Under the terms of the agreements, each month, CenturyLink will receive payments equivalent to the 1-month LIBOR from each of the five banks. Also, under the terms of the agreements, CenturyLink will pay to each bank a fixed-rate of 2.47926% monthly, with the first payment due April 30, 2019. The agreements are effective March 31, 2019 and mature on March 31, 2022.

(7) Accounts Receivable

The following table presents details of our accounts receivable balances:

| | As of December 31, | | |
|---------------------------------------|--------------------|---------------|----------|
| | | 2018 | 2017 |
| | | (Dollars in m | illions) |
| Trade and purchased receivables | \$ | 2,094 | 2,245 |
| Earned and unbilled receivables | | 425 | 436 |
| Other | | 21 | 40 |
| Total accounts receivable | | 2,540 | 2,721 |
| Less: allowance for doubtful accounts | | (142) | (164) |
| Accounts receivable, less allowance | \$ | 2,398 | 2,557 |

We are exposed to concentrations of credit risk from residential and business customers within our local service area, business customers outside of our local service area and from other telecommunications service providers. We generally do not require collateral to secure our receivable balances. We have agreements with other telecommunications service providers whereby we agree to bill and collect on their behalf for services rendered by those providers to our customers within our local service area. We purchase accounts receivable from other telecommunications service providers primarily on a recourse basis and include these amounts in our accounts receivable balance. We have not experienced any significant loss associated with these purchased receivables.

The following table presents details of our allowance for doubtful accounts:

| | Beginning | | | Ending |
|------|---------------------------|-----------|------------|---------|
| | Balance | Additions | Deductions | Balance |
| | (Dollars in millions) | | | |
| 2018 | \$ 164 | 153 | (175) | 142 |
| 2017 | 178 | 176 | (190) | 164 |
| 2016 | 152 | 192 | (166) | 178 |

(8) Property, Plant and Equipment

Net property, plant and equipment is composed of the following:

| | Depreciable Lives | | Depreciable | | As of December 31, | | |
|--|----------------------|---------------------|-------------|----------|--------------------|--|--|
| | | | 2018 | 2017 | | | |
| | | (Dollars in million | | illions) | | | |
| Land | N/A | \$ | 871 | 883 | | | |
| Fiber, conduit and other outside plant (1) | 15-45 years | | 23,936 | 22,798 | | | |
| Central office and other network electronics (2) | 3-10 years | | 18,736 | 18,538 | | | |
| Support assets (3) | 3-30 years | | 8,020 | 7,586 | | | |
| Construction in progress (4) | N/A | | 1,704 | 1,399 | | | |
| Gross property, plant and equipment | | | 53,267 | 51,204 | | | |
| Accumulated depreciation | | | (26,859) | (24,352) | | | |
| Net property, plant and equipment | | \$ | 26,408 | 26,852 | | | |

- (1) Fiber, conduit and other outside plant consists of fiber and metallic cable, conduit, poles and other supporting structures.
- (2) Central office and other network electronics consists of circuit and packet switches, routers, transmission electronics and electronics providing service to customers.
- (3) Support assets consist of buildings, cable landing stations, data centers, computers and other administrative and support equipment.
- (4) Construction in progress includes inventory held for construction and property of the aforementioned categories that has not been placed in service as it is still under construction.

We recorded depreciation expense of \$3.3 billion , \$2.7 billion and \$2.7 billion for the years ended December 31, 2018 , 2017 and 2016 , respectively.

Asset Retirement Obligations

At December 31, 2018, our asset retirement obligations balance was primarily related to estimated future costs of removing equipment from leased properties and estimated future costs of properly disposing of asbestos and other hazardous materials upon remodeling or demolishing buildings. Asset retirement obligations are included in other long-term liabilities on our consolidated balance sheets.

As of the Level 3 acquisition date, we recorded liabilities to reflect our fair values of Level 3's asset retirement obligations. Our fair value estimates were determined using discounted cash flow method.

The following table provides asset retirement obligation activity:

| | rears Ended December 31, | | |
|---|--------------------------|----------------------|--------------|
| | 2018 | 2017 | 2016 |
| | | (Dollars in millions |) |
| Balance at beginning of year | \$ 115 | 95 | 91 |
| Accretion expense | 10 | 6 | 6 |
| Liabilities assumed in acquisition of Level 3 (1) | 58 | 45 | _ |
| Liabilities settled | (14) | (3) | (2) |
| Liabilities transferred to Cyxtera | _ | (20) | - |
| Change in estimate | 21 | (8) | |
| Balance at end of year | \$ 190 | 115 | 95 |
| | | | |

Veers Ended December 24

(1) The liabilities assumed during 2018 relate to purchase price adjustments during the year.

During 2018, we revised our estimates for the cost of removal of network equipment, tanks, and asbestos remediation resulting in a \$21 million change in estimate of our asset retirement obligation. The 2018 and 2017 change in estimates are offset to gross property, plant and equipment.

(9) Severance and Leased Real Estate

Periodically, we reduce our workforce and accrue liabilities for the related severance costs. These workforce reductions result primarily from the progression or completion of our post-acquisition integration plans, increased competitive pressures, cost reduction initiatives, process improvements through automation and reduced workload demands due to the loss of customers purchasing certain services.

We report severance liabilities within accrued expenses and other liabilities - salaries and benefits in our consolidated balance sheets and report severance expenses in selling, general and administrative expenses in our consolidated statements of operations. As described in Note 15—Segment Information , we do not allocate these severance expenses to our segments.

We have recognized liabilities to reflect our estimates of the fair values of the existing lease obligations for real estate which we have ceased using, net of estimated sublease rentals. As of the Level 3 acquisition date, we recorded liabilities to reflect the fair values of the existing lease obligations for real estate for which we had ceased using, net of estimated sublease rentals. Our fair value estimates were determined using discounted cash flow methods. We recognize expense to reflect accretion of the discounted liabilities and periodically we adjust the expense when our actual subleasing experience differs from our initial estimates. We report the current portion of liabilities for ceased-use real estate leases in accrued expenses and other liabilities-other and report the noncurrent portion in deferred credits and other liabilities-other in our consolidated balance sheets. We report the related expenses in selling, general and administrative expenses in our consolidated statements of operations. At December 31, 2018, the current and noncurrent portions of our leased real estate accrual were \$24 million and \$86 million, respectively. The remaining lease terms range from less than one year to 12.0 years, with a weighted average of 6.7 years.

Changes in our accrued liabilities for severance expenses and leased real estate were as follows:

| | Severance Real | Estate |
|---|-----------------------|--------|
| | (Dollars in millions) | |
| Balance at December 31, 2016 | \$ 98 | 67 |
| Accrued to expense | 42 | 4 |
| Liabilities assumed in acquisition of Level 3 | 1 | 4 |
| Payments, net | (108) | (13) |
| Reversals and adjustments | _ | 2 |
| Balance at December 31, 2017 | 33 | 64 |
| Accrued to expense | 205 | 70 |
| Payments, net | (151) | (24) |
| Balance at December 31, 2018 | \$ 87 | 110 |

(10) Employee Benefits

Pension, Post-Retirement and Other Post-Employment Benefits

We sponsor various defined benefit pension plans (qualified and non-qualified) which, in the aggregate, cover a substantial portion of our employees including legacy CenturyLink, legacy Qwest Communications International Inc. ("Qwest") and legacy Embarq employees. Pension benefits for participants of the CenturyLink Combined Pension Plan ("Combined Pension Plan") who are represented by a collective bargaining agreement are based on negotiated schedules. All other participants' pension benefits are based on each individual participant's years of service and compensation. We also maintain non-qualified pension plans for certain current and former highly compensated employees. We maintain post-retirement benefit plans that provide health care and life insurance benefits for certain eligible retirees. We also provide other post-employment benefits for certain eligible former employees. We use a December 31 measurement date for all our plans.

Pension Benefits

In connection with the acquisition of Level 3 Communications, Inc. on November 1, 2017, we assumed defined benefit pension plans sponsored by various Level 3 companies for their employees. Based on a valuation analysis, we recognized a \$20 million liability on November 1, 2017 for the unfunded status of the Level 3 pension plans. The net unfunded status recognized on our balance sheets at December 31, 2018 and 2017 was \$11 million and \$20 million, respectively, representing liabilities of \$144 million and \$167 million, and assets of \$133 million and \$147 million, respectively. Due to the insignificant impact of these pension plans on our consolidated financial statements, we have predominantly excluded them from the remaining employee benefit disclosures in this Note.

United States funding laws require a company with a pension shortfall to fund the annual cost of benefits earned in addition to a seven -year amortization of the shortfall. Our funding policy for our Combined Pension Plan is to make contributions with the objective of accumulating ample assets to pay all qualified pension benefits when due under the terms of the plan. The accounting unfunded status of our qualified pension plan was \$1.6 billion and \$2.0 billion as of December 31, 2018 and 2017, respectively.

We made a voluntary cash contribution to our qualified pension plan of \$500 million and \$100 million in 2018 and 2017, respectively, and paid \$5 million of benefits directly to participants of our non-qualified pension plans in both 2018 and 2017, respectively. Based on current laws and circumstances, we are not required to make any contributions to our qualified pension plan in 2019. We currently do not expect to make a voluntary contribution to the trust for our qualified pension plan in 2019. We estimate that in 2019 we will pay \$5 million of benefits directly to participants of our non-qualified pension plans.

As previously mentioned, we sponsor unfunded non-qualified pension plans for certain current and former highly-compensated employees. The net unfunded status of our non-qualified pension plans was \$52 million and \$58 million for the years ended December 31, 2018 and 2017, respectively. Due to the insignificant impact of these pension plans on our consolidated financial statements, we have predominantly excluded them from the remaining employee benefit disclosures in this Note.

Post-Retirement Benefits

In connection with our acquisition of Level 3 Communications, Inc. on November 1, 2017, we assumed post-retirement benefit plans sponsored by Level 3 Communications, L.L.C. and Continental Level 3, Inc. for certain of its current and former employees. Based on a valuation analysis, we recognized less than \$1 million in liability for the unfunded status of Level 3's post-retirement benefit plans. Though largely unfunded, these post-retirement plans, in the aggregate, are immaterial to our consolidated financial statements. Due to the insignificant amount of these post-retirement plans, we have predominantly excluded them from the remaining employee benefit disclosures in this Note.

Our post-retirement benefit plans provide post-retirement benefits to qualified retirees and allow (i) eligible employees retiring before certain dates to receive benefits at no or reduced cost and (ii) eligible employees retiring after certain dates to receive benefits on a shared cost basis. The post-retirement benefits not paid by the trusts are funded by us and we expect to continue funding these post-retirement obligations as benefits are paid. The accounting unfunded status of our qualified post-retirement benefit plan was \$3.0 billion and \$3.4 billion as of December 31, 2018 and 2017, respectively.

Assets in the post-retirement trusts were substantially depleted as of December 31, 2016; however, we continue to pay certain post-retirement benefits through the trusts. No contributions were made to the post-retirement trusts in 2018 and 2017. Benefits not paid from the trusts are expected to be paid directly by us with available cash. In 2018, we paid \$249 million of post-retirement benefits, net of participant contributions and direct subsidies. In 2019, we expect to pay \$270 million of post-retirement benefits, net of participant contributions and direct subsidies. The increase in anticipated post-retirement benefit payments is the result of increased utilization coupled with a continued rise in the cost of care.

We expect our health care cost trend rate to range from 5.0% to 7.0% in 2019, 5.0% to 6.5% in 2020 and grading to 4.50% by 2025. Our post-retirement benefit cost, for certain eligible legacy Qwest retirees and certain eligible legacy CenturyLink retirees, is capped at a set dollar amount. Therefore, those health care benefit obligations are not subject to increasing health care trends after the effective date of the caps.

Expected Cash Flows

The Combined Pension Plan payments, post-retirement health care benefit payments and premiums, and life insurance premium payments are either distributed from plan assets or paid by us. The estimated benefit payments provided below are based on actuarial assumptions using the demographics of the employee and retiree populations and have been reduced by estimated participant contributions.

| | | ed Pension lan | Post-Retirement Benefit Plans | Medicare Part D Subsidy Receipts |
|------------------------------------|-----|-------------------|----------------------------------|-------------------------------------|
| | | | | |
| Estimated future benefit payments: | | | | |
| 2019 | \$ | 966 | 277 | (7) |
| 2020 | | 938 | 269 | (7) |
| 2021 | | 916 | 261 | (7) |
| 2022 | | 891 | 252 | (7) |
| 2023 | | 867 | 243 | (6) |
| 2024 - 2028 | | 3,971 | 1,065 | (26) |
| | 121 | | | |

Net Periodic Benefit Expense

We utilize a full yield curve approach in connection with estimating the service and interest components of net periodic benefit expense by applying the specific spot rates along the yield curve used in the determination of the benefit obligation to the relevant projected cash flow.

The actuarial assumptions used to compute the net periodic benefit expense for our Combined Pension Plan and post-retirement benefit plans are based upon information available as of the beginning of the year, as presented in the following table.

| | Combined Pension Plan | | | Post-Re | etirement Benefit | Plans |
|--|-----------------------|---------------|---------------|---------------|-------------------|---------------|
| · | 2018 | 2017 | 2017 2016 | | 2017 | 2016 |
| Actuarial assumptions at beginning of year: | | | _ | | | |
| Discount rate | 3.14% - 3.69% | 3.25% - 4.14% | 3.34% - 4.46% | 4.26% | 3.90% | 4.15% |
| Rate of compensation increase | 3.25% | 3.25% | 3.25% | N/A | N/A | N/A |
| Expected long-term rate of return on plan assets (1) | 6.50% | 6.50% | 7.00% | 4.00% | 5.00% | 7.00% |
| Initial health care cost trend rate | N/A | N/A | N/A | 7.00% / 5.00% | 7.00% / 5.00% | 5.00% / 5.25% |
| Ultimate health care cost trend rate | N/A | N/A | N/A | 4.50% | 4.50% | 4.50% |
| Year ultimate trend rate is reached | N/A | N/A | N/A | 2025 | 2025 | 2025 |

N/A - Not applicable

Net periodic benefit (income) expense for our combined pension plan includes the following components:

| | Years Ended December 31, | | | | |
|---|--------------------------|--------------------|-------|--|--|
| | 2018 | 2017 | 2016 | | |
| | (Do | llars in millions) | | | |
| Service cost | \$ 66 | 63 | 64 | | |
| Interest cost | 392 | 409 | 425 | | |
| Expected return on plan assets | (685) | (666) | (733) | | |
| Special termination benefits charge | 15 | _ | 13 | | |
| Recognition of prior service credit | (8) | (8) | (8) | | |
| Recognition of actuarial loss | 178 | 204 | 174 | | |
| Net periodic pension benefit (income) expense | \$ (42) | 2 | (65) | | |

Net periodic benefit expense for our post-retirement benefit plans includes the following components:

| | Post-Retirement Plans Years Ended December 31, | | | | |
|--|--|---------------------|------|--|--|
| | 2018 | 2017 | 2016 | | |
| | (De | ollars in millions) | | | |
| Service cost | \$ 18 | 18 | 19 | | |
| Interest cost | 97 | 100 | 111 | | |
| Expected return on plan assets | (1) | (2) | (7) | | |
| Special termination benefits charge | _ | _ | 3 | | |
| Recognition of prior service cost | 20 | 20 | 20 | | |
| Net periodic post-retirement benefit expense | \$ 134 | 136 | 146 | | |

⁽¹⁾ Rates are presented net of projected fees and administrative costs.

We report service costs for our Combined Pension Plan and post-retirement benefit plans in cost of services and products and selling, general and administrative expenses in our consolidated statements of operations for the years ended December 31, 2018, 2017 and 2016. Additionally, a portion of the service cost is also allocated to certain assets under construction, which are capitalized and reflected as part of property, plant and equipment in our consolidated balance sheets. The remaining components of net periodic benefit expense (income) are reported in other income, net in our consolidated statements of operations. As a result of ongoing efforts to reduce our workforce, we recognized a one-time charge in 2018 of \$15 million for special termination benefit enhancements paid to certain eligible employees upon voluntary retirement.

Benefit Obligations

The actuarial assumptions used to compute the funded status for the plans are based upon information available as of December 31, 2018 and 2017 and are as follows:

| | Combined Per | nsion Plan | Post-Retirement Benefit Plans | | | |
|---------------------------------------|--------------|------------|-------------------------------|---------------|--|--|
| | Decembe | er 31, | December 31, | | | |
| | 2018 | 2017 | 2018 | 2017 | | |
| Actuarial assumptions at end of year: | | | | | | |
| Discount rate | 4.29% | 3.57% | 4.26% | 3.53% | | |
| Rate of compensation increase | 3.25% | 3.25% | N/A | N/A | | |
| Initial health care cost trend rate | N/A | N/A | 7.00% / 5.00% | 7.00% / 5.00% | | |
| Ultimate health care cost trend rate | N/A | N/A | 4.50% | 4.50% | | |
| Year ultimate trend rate is reached | N/A | N/A | 2025 | 2025 | | |

N/A - Not applicable

In 2018, 2017 and 2016, we adopted the revised mortality tables and projection scales released by the Society of Actuaries, which decreased the projected benefit obligation of our benefit plans by \$38 million , \$113 million and \$268 million , respectively. The change in the projected benefit obligation of our benefit plans was recognized as part of the net actuarial (gain) loss and is included in accumulated other comprehensive loss, a portion of which is subject to amortization over the remaining estimated life of plan participants, which was approximately 8 to 9 years as of December 31, 2018.

The following tables summarize the change in the benefit obligations for the Combined Pension Plan and post-retirement benefit plans:

| | Combined Pension Plan Years Ended December 31, | | | |
|---|--|---------------------|---------|--|
| | 2018 | 2017 | 2016 | |
| | (Do | ollars in millions) | | |
| Change in benefit obligation | | | | |
| Benefit obligation at beginning of year | \$ 13,064 | 13,244 | 13,287 | |
| Service cost | 66 | 63 | 64 | |
| Interest cost | 392 | 409 | 425 | |
| Plan amendments | _ | _ | 2 | |
| Special termination benefits charge | 15 | _ | 13 | |
| Actuarial (gain) loss | (765) | 586 | 487 | |
| Benefits paid by company | _ | _ | _ | |
| Benefits paid from plan assets | (1,178) | (1,238) | (1,034) | |
| Benefit obligation at end of year | \$ 11,594 | 13,064 | 13,244 | |

Post-Retirement Benefit Plans Years Ended December 31,

Combined Pension Plan

| | 2010 | | 2212 |
|---|-------------|--------------------|-------|
| | 2018 | 2017 | 2016 |
| | (Do | llars in millions) | |
| Change in benefit obligation | | | |
| Benefit obligation at beginning of year | \$ 3,375 | 3,413 | 3,567 |
| Service cost | 18 | 18 | 19 |
| Interest cost | 97 | 100 | 111 |
| Participant contributions | 54 | 54 | 57 |
| Direct subsidy receipts | 8 | 7 | 5 |
| Special termination benefits charge | _ | _ | 3 |
| Plan Amendment | (36) | _ | _ |
| Actuarial (gain) loss | (224) | 112 | (13) |
| Benefits paid by company | (311) | (298) | (191) |
| Benefits paid from plan assets | (4) | (31) | (145) |
| Benefit obligation at end of year | \$ 2,977 | 3,375 | 3,413 |
| | | | |

Our aggregate benefit obligation as of December 31, 2018, 2017 and 2016 was \$14.8 billion, \$16.5 billion and \$16.7 billion, respectively.

Plan Assets

We maintain plan assets for our Combined Pension Plan and certain post-retirement benefit plans. The following tables summarize the change in the fair value of plan assets for the Combined Pension Plan and post-retirement benefit plans:

| | Years Ended December 31, | | | | |
|--|---------------------------|---------|---------|--|--|
| | 2018 | 2017 | 2016 | | |
| | (Dollars in millions) | | | | |
| Change in plan assets | | | | | |
| Fair value of plan assets at beginning of year | \$ 11,060 | 10,892 | 11,072 | | |
| Return on plan assets | (349) | 1,306 | 754 | | |
| Employer contributions | 500 | 100 | 100 | | |
| Benefits paid from plan assets | (1,178) | (1,238) | (1,034) | | |
| Fair value of plan assets at end of year | \$ 10,033 | 11,060 | 10,892 | | |

| | Post-Retirement Benefit Plans Years Ended December 31, | | | | |
|--|---|------|-------|--|--|
| | 2018 | 2017 | 2016 | | |
| | (Dollars in millions) | | | | |
| Change in plan assets | | | | | |
| Fair value of plan assets at beginning of year | \$ 23 | 53 | 193 | | |
| Return on plan assets | (1) | 1 | 5 | | |
| Benefits paid from plan assets | (4) | (31) | (145) | | |
| Fair value of plan assets at end of year | \$ 18 | 23 | 53 | | |

The expected rate of return on plan assets is the long-term rate of return we expect to earn on the plans' assets, net of administrative expenses paid from plan assets. It is determined annually based on the strategic asset allocation and the long-term risk and return forecast for each asset class.

Combined Pension Plan: Our investment objective for the qualified pension plan assets is to achieve an attractive risk-adjusted return over time that will provide for the payment of benefits and minimize the risk of large losses. We employ a liability-aware investment strategy designed to reduce the volatility of pension assets relative to pension liabilities. This strategy is evaluated frequently and is expected to evolve over time with changes in the funded status and other factors. Approximately 31% of plan assets is targeted to long-duration investment grade bonds and interest rate sensitive derivatives and 29% is targeted to assets expected to outperform the liability with moderate funded status risk. Approximately 26% is targeted to public equities and 14% is targeted to private market investments. At the beginning of 2019, our expected annual long-term rate of return on pension assets before consideration of administrative expenses is assumed to be 7.0%. Projected PBGC (Pension Benefit Guaranty Corporation) premium rates reduce the annual long-term expected return net of administrative expenses to 6.5%.

The short term and long term interest crediting rates during 2018 for cash balance components of the Combined Pension Plan 2018 were 3.25% and 4.00%, respectively.

Post-Retirement Benefit Plans: Our investment objective for the post-retirement benefit plans' assets is to achieve an attractive risk-adjusted return and minimize the risk of large losses over the expected life of the assets. At the beginning of 2019, our expected annual long-term rate of return on post-retirement benefit plan assets is assumed to be 4.0%.

Permitted investments: Plan assets are managed consistent with the restrictions set forth by the Employee Retirement Income Security Act of 1974, as amended.

Derivative instruments: Derivative instruments are used to reduce risk as well as provide return. The gross notional exposure of the derivative instruments directly held by the pension benefit plan is shown below. The notional amount of the derivatives corresponds to market exposure but does not represent an actual cash investment. Our post-retirement plans were not invested in derivative instruments for the years ended December 31, 2018 or 2017.

| | | xposure | | | |
|--|----|--|-------|--|--|
| | | Combined Pension Plan Years Ended December 31, | | | |
| | | | | | |
| | | 2018 | | | |
| | | (Dollars in mill | ions) | | |
| <u>Derivative instruments:</u> | | | | | |
| Exchange-traded U.S. equity futures | \$ | 300 | 256 | | |
| Exchange-traded Treasury and other interest rate futures | | 3,901 | 1,830 | | |
| Interest rate swaps | | 83 | 137 | | |
| Credit default swaps | | 66 | 100 | | |
| Equity index swaps | | - | 1 | | |
| Foreign exchange forwards | | 295 | 293 | | |
| Options | | 192 | 259 | | |

Fair Value Measurements: Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between independent and knowledgeable parties who are willing and able to transact for an asset or liability at the measurement date. We use valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs when determining fair value and then we rank the estimated values based on the reliability of the inputs used following the fair value hierarchy set forth by the FASB. For additional information on the fair value hierarchy, see Note 13—Fair Value of Financial Instruments.

At December 31, 2018, we used the following valuation techniques to measure fair value for assets. There were no changes to these methodologies during 2018:

· Level 1—Assets were valued using the closing price reported in the active market in which the individual security was traded.

- Level 2—Assets were valued using quoted prices in markets that are not active, broker dealer quotations, net asset value of shares held by the plans and other methods by which all significant inputs were observable at the measurement date.
- Level 3—Assets were valued using unobservable inputs in which little or no market data exists as reported by the respective institutions at the measurement date.

The tables below present the fair value of plan assets by category and the input levels used to determine those fair values at December 31, 2018. It is important to note that the asset allocations do not include market exposures that are gained with derivatives. Investments include dividend and interest receivables, pending trades and accrued expenses.

| | Fair Value of Combined Pension Plan Assets at December 31, 2018 | | | | | |
|--|---|--------------|--------------|-----------|----|--------|
| | | Level 1 | Level 2 | Level 3 | | Total |
| | | | (Dollars in | millions) | | |
| Investment grade bonds (a) | \$ | 458 | 1,393 | _ | \$ | 1,851 |
| High yield bonds (b) | | _ | 277 | 7 | | 284 |
| Emerging market bonds (c) | | 151 | 181 | _ | | 332 |
| U.S. stocks (e) | | 764 | 2 | 2 | | 768 |
| Non-U.S. stocks (f) | | 601 | _ | _ | | 601 |
| Private debt (i) | | _ | - | 15 | | 15 |
| Multi-asset strategies (I) | | 342 | _ | _ | | 342 |
| Derivatives (m) | | 7 | (2) | _ | | 5 |
| Cash equivalents and short-term investments (n) | | 3 | 907 | _ | | 910 |
| Total investments, excluding investments valued at NAV | \$ | 2,326 | 2,758 | 24 | | 5,108 |
| Investments valued at NAV | | | | | | 4,925 |
| Total pension plan assets | | | | | \$ | 10,033 |

| | Fair Value of Post-Retirement Plan Assets at December 31, 2018 | | | | | |
|--|--|---------|-------------|-----------|----|-------|
| | | Level 1 | Level 2 | Level 3 | | Total |
| | | | (Dollars in | millions) | | |
| Total investments, excluding investments valued at NAV | \$ | _ | _ | _ | | _ |
| Investments valued at NAV | | | | | = | 18 |
| Total post-retirement plan assets | | | | | \$ | 18 |

The tables below present the fair value of plan assets by category and the input levels used to determine those fair values at December 31, 2017. It is important to note that the asset allocations do not include market exposures that are gained with derivatives. Investments include dividend and interest receivable, pending trades and accrued expenses.

| | Fair Value of Combined Pension Plan Assets at December 31, 2017 | | | | | |
|--|---|---------|---------------|-----------|----|--------|
| | | Level 1 | Level 2 | Level 3 | | Total |
| | | | (Dollars in n | nillions) | | |
| Investment grade bonds (a) | \$ | 432 | 1,315 | _ | \$ | 1,747 |
| High yield bonds (b) | | _ | 575 | 7 | | 582 |
| Emerging market bonds (c) | | 217 | 219 | 1 | | 437 |
| U.S. stocks (e) | | 1,030 | 2 | 3 | | 1,035 |
| Non-U.S. stocks (f) | | 706 | _ | _ | | 706 |
| Private debt (i) | | _ | - | 15 | | 15 |
| Multi-asset strategies (I) | | 440 | _ | _ | | 440 |
| Derivatives (m) | | 2 | _ | _ | | 2 |
| Cash equivalents and short-term investments (n) | | _ | 476 | 1 | | 477 |
| Total investments, excluding investments valued at NAV | \$ | 2,827 | 2,587 | 27 | | 5,441 |
| Investments valued at NAV | | | | | | 5,619 |
| Total pension plan assets | | | | | \$ | 11,060 |

Fair Value of Post-Retirement Plan Assets at December 31, 2017

| | at becember 31, 2017 | | | | | |
|--|----------------------|---------|-------------|-----------|----|-------|
| | | Level 1 | Level 2 | Level 3 | ٦ | Γotal |
| | | | (Dollars in | millions) | | |
| U.S. stocks (e) | | 1 | _ | _ | | 1 |
| Total investments, excluding investments valued at NAV | \$ | 1 | _ | _ | | 1 |
| Investments valued at NAV | | | | | | 22 |
| Total post-retirement plan assets | | | | | \$ | 23 |

The table below presents the fair value of plan assets valued at NAV by category for our pension and post-retirement plans at December 31, 2018 and 2017.

Fair Value of Dian Access Valued at NAV

| | Fair Value of Plan Assets Valued at NAV | | | | | | |
|---|---|-------|-------------|---|------|--|--|
| | Combined Pension Plan at December 31, | | | Post-Retirement Benefit Plans at December 31, | | | |
| | | 2018 | 2017 | 2018 | 2017 | | |
| | | | (Dollars in | millions) | | | |
| Investment grade bonds (a) | \$ | 109 | 163 | _ | _ | | |
| High yield bonds (b) | | 388 | 483 | _ | _ | | |
| Emerging market bonds (c) | | _ | 14 | _ | _ | | |
| Diversified strategies (d) | | _ | 538 | _ | _ | | |
| U.S. stocks (e) | | 150 | 73 | _ | _ | | |
| Non-U.S. stocks (f) | | 500 | 627 | _ | _ | | |
| Emerging market stocks (g) | | 75 | 98 | _ | _ | | |
| Private equity (h) | | 347 | 460 | 6 | 10 | | |
| Private debt (i) | | 452 | 374 | 1 | 1 | | |
| Market neutral hedge funds (j) | | 746 | 769 | - | _ | | |
| Directional hedge funds (j) | | 512 | 636 | _ | _ | | |
| Real estate (k) | | 821 | 903 | - | 1 | | |
| Multi-asset strategies (I) | | 763 | 424 | _ | _ | | |
| Cash equivalents and short-term investments (n) | | 62 | 57 | 11 | 10 | | |
| Total investments valued at NAV | \$ | 4,925 | 5,619 | 18 | 22 | | |

The plans' assets are invested in various asset categories utilizing multiple strategies and investment managers. Interests in commingled funds are valued using the net asset value ("NAV") per unit of each fund. The NAV reported by the fund manager is based on the market value of the underlying investments owned by each fund, minus its liabilities, divided by the number of shares outstanding. Commingled funds can be redeemed at NAV, generally within a year of the financial statement date. Investments in private funds, primarily limited partnerships, represent long-term commitments with a fixed maturity date and are also valued at NAV. The plan has unfunded commitments related to certain private fund investments, which in aggregate are not material to the plan. Valuation inputs for these private fund interests are generally based on assumptions and other information not observable in the market. The assumptions and valuation methodologies of the pricing vendors, account managers, fund managers and partnerships are monitored and evaluated for reasonableness. Below is an overview of the asset categories, the underlying strategies and valuation inputs used to value the assets in the preceding tables:

(a) Investment grade bonds represent investments in fixed income securities as well as commingled bond funds comprised of U.S. Treasury securities, agencies, corporate bonds, mortgage-backed securities, asset-backed securities and commercial mortgage-backed securities. Treasury securities are valued at the bid price reported in the active market in which the security is traded and are classified as Level 1. The valuation inputs of other investment grade bonds primarily utilize observable market information and are based on a spread to U.S. Treasury securities and consider yields available on comparable securities of issuers with similar credit ratings. The primary observable inputs include references to the new issue market for similar securities, the secondary trading markets and dealer quotes. Option adjusted spread models are utilized to evaluate securities such as asset backed securities that have early redemption features. These securities are classified as Level 2.

(b) High yield bonds represent investments in below investment grade fixed income securities as well as commingled high yield bond funds. The valuation inputs for the securities primarily utilize observable market information and are based on a spread to U.S. Treasury securities and consider yields available on comparable securities of issuers with similar credit ratings. These securities are primarily classified as Level 2. Securities whose valuation inputs are not based on observable market information are classified as Level 3.

- (c) Emerging market bonds represent investments in securities issued by governments and other entities located in developing countries as well as registered mutual funds and commingled emerging market bond funds. The valuation inputs for the securities utilize observable market information and are primarily based on dealer quotes or a spread relative to the local government bonds. The registered mutual fund is classified as Level 1 while individual securities are primarily classified as Level 2. Securities whose valuation inputs are not based on observable market information are classified as Level 3.
- (d) *Diversified strategies* represent an investment in a commingled fund that primarily has exposures to global government, corporate and inflation linked bonds, global stocks and commodities. This asset category includes investments in a registered mutual fund which is classified at Level 1. The valuation inputs utilize observable market information including published prices for exchange traded securities, bid prices for government bonds, and spreads and yields available for comparable fixed income securities with similar credit ratings.
- (e) *U.S. stocks* represent investments in stocks of U.S. based companies as well as commingled U.S. stock funds. The valuation inputs for U.S. stocks are based on the last published price reported on the major stock market on which the securities are traded and are primarily classified as Level 1. Securities whose valuation inputs are not based on observable market information are classified as Level 3.
- (f) Non-U.S. stocks represent investments in stocks of companies based in developed countries outside the U.S. as well as commingled funds. The valuation inputs for non-U.S. stocks are based on the last published price reported on the major stock market on which the securities are traded and are primarily classified as Level 1.
- (g) Emerging market stocks represent investments in commingled funds comprised of stocks of companies located in developing markets. The commingled funds are valued at NAV based on the market value of the underlying investments using the same valuation inputs described previously for individual stocks.
- (h) *Private equity* represents non-public investments in domestic and foreign buy out and venture capital funds. Private equity funds are primarily structured as limited partnerships and are valued according to the valuation policy of each partnership, subject to prevailing accounting and other regulatory guidelines. The partnerships are valued at NAV using valuation methodologies that consider a range of factors, including but not limited to the price at which investments were acquired, the nature of the investments, market conditions, trading values on comparable public securities, current and projected operating performance, and financing transactions subsequent to the acquisition of the investments. These valuation methodologies involve a significant degree of judgment.
- (i) Private debt represents non-public investments in distressed or mezzanine debt funds and pension group insurance contracts. Pension group insurance contracts are valued based on actuarial assumptions, and are classified as Level 3. Mezzanine debt instruments are debt instruments that are subordinated to other debt issues and may include embedded equity instruments such as warrants. Private debt funds are primarily structured as limited partnerships and are valued at NAV according to the valuation policy of each partnership, subject to prevailing accounting and other regulatory guidelines. The valuation of underlying fund investments are based on factors including the issuer's current and projected credit worthiness, the security's terms, reference to the securities of comparable companies, and other market factors. These valuation methodologies involve a significant degree of judgment.
- (j) Market neutral hedge funds hold investments in a diversified mix of instruments that are intended in combination to exhibit low correlations to market fluctuations. These investments are typically combined with futures to achieve uncorrelated excess returns over various markets. Directional hedge funds —This asset category represents investments that may exhibit somewhat higher correlations to market fluctuations than the market neutral hedge funds. Investments in hedge funds include both direct investments and investments in diversified funds of funds. Hedge funds are valued at NAV based on the market value of the underlying investments which include publicly traded equity and fixed income securities and privately negotiated debt securities. The hedge funds are valued by third party administrators using the same valuation inputs previously described.
- (k) Real estate represents investments in commingled funds and limited partnerships that invest in a diversified portfolio of real estate properties. These investments are valued at NAV according to the valuation policy of each fund or partnership, subject to prevailing accounting and other regulatory guidelines. The valuation inputs of the underlying properties are generally based on third-party appraisals that use comparable sales or a projection of future cash flows to determine fair value.

- (I) *Multi-asset strategies* represent broadly diversified strategies that have the flexibility to tactically adjust exposures to different asset classes through time. This asset category includes investments in a registered mutual fund which is classified as Level 1 and a commingled fund which is valued at NAV based on the market value of the underlying investments.
- (m) *Derivatives* include exchange traded futures contracts which are classified as Level 1, as well as privately negotiated over the counter swaps and options that are valued based on the change in interest rates or a specific market index and are classified as Level 2. The market values represent gains or losses that occur due to fluctuations in interest rates, foreign currency exchange rates, security prices, or other factors.
- (n) Cash equivalents and short-term investments represent investments that are used in conjunction with derivatives positions or are used to provide liquidity for the payment of benefits or other purposes. The valuation inputs of securities are based on a spread to U.S. Treasury Bills, the Federal Funds Rate, or London Interbank Offered Rate and consider yields available on comparable securities of issuers with similar credit ratings and are primarily classified as Level 2. The commingled funds are valued at NAV based on the market value of the underlying investments using the same valuation inputs described above.

Concentrations of Risk: Investments, in general, are exposed to various risks, such as significant world events, interest rate, credit, foreign currency and overall market volatility risk. These risks are managed by broadly diversifying assets across numerous asset classes and strategies with differing expected returns, volatilities and correlations. Risk is also broadly diversified across numerous market sectors and individual companies. Financial instruments that potentially subject the plans to concentrations of counterparty risk consist principally of investment contracts with high quality financial institutions. These investment contracts are typically collateralized obligations and/or are actively managed, limiting the amount of counterparty exposure to any one financial institution. Although the investments are well diversified, the value of plan assets could change materially depending upon the overall market volatility, which could affect the funded status of the plans.

The table below presents a rollforward of the pension plan assets valued using Level 3 inputs:

| Combined Pension Plan Assets | Valued Using Level 3 Inputs |
|------------------------------|-----------------------------|
|------------------------------|-----------------------------|

| | Combined Fension Flan Assets Valued Using Level 5 inputs | | | | | | | |
|------------------------------|--|------------------------|--------------------------|-------------|--------------|------|-------|--|
| | | High Yield Bonds | Emerging Market Bonds | U.S. Stocks | Private Debt | Cash | Total | |
| | (Dollars in millions) | | | | | | | |
| Balance at December 31, 2016 | \$ | 11 | _ | _ | _ | _ | 11 | |
| Net transfers | | (1) | _ | _ | 14 | _ | 13 | |
| Acquisitions | | 2 | 1 | _ | 1 | 1 | 5 | |
| Dispositions | | (5) | _ | 3 | _ | _ | (2) | |
| Balance at December 31, 2017 | | 7 | 1 | 3 | 15 | 1 | 27 | |
| Dispositions | | _ | _ | (2) | - | (1) | (3) | |
| Actual return on plan assets | | _ | (1) | 1 | _ | _ | _ | |
| Balance at December 31, 2018 | \$ | 7 | _ | 2 | 15 | _ | 24 | |

Certain gains and losses are allocated between assets sold during the year and assets still held at year-end based on transactions and changes in valuations that occurred during the year. These allocations also impact our calculation of net acquisitions and dispositions.

For the year ended December 31, 2018, the investment program produced actual losses on qualified pension and post-retirement plan assets of \$350 million as compared to expected returns of \$686 million for a difference of \$1.0 billion. For the year ended December 31, 2017, the investment program produced actual gains on qualified pension and post-retirement plan assets of \$1.3 billion as compared to the expected returns of \$668 million for a difference of \$639 million. The short-term annual returns on plan assets will almost always be different from the expected long-term returns and the plans could experience net gains or losses, due primarily to the volatility occurring in the financial markets during any given year.

Unfunded Status

The following table presents the unfunded status of the Combined Pension Plan and post-retirement benefit plans:

| | Combined Pen | Post-Retirement Benefit Plans | | | |
|--|--------------------|-------------------------------|--------------------------|---------|--|
| | Years Ended De | cember 31, | Years Ended December 31, | | |
| | 2018 | 2017 | 2018 | 2017 | |
| | | (Dollars in n | millions) | | |
| Benefit obligation | \$ (11,594) | (13,064) | (2,977) | (3,375) | |
| Fair value of plan assets | 10,033 | 11,060 | 18 | 23 | |
| Unfunded status | (1,561) | (2,004) | (2,959) | (3,352) | |
| Current portion of unfunded status | _ | _ | (252) | (262) | |
| Non-current portion of unfunded status | \$ (1,561) | (2,004) | (2,707) | (3,090) | |

The current portion of our post-retirement benefit obligations is recorded on our consolidated balance sheets in accrued expenses and other current liabilities-salaries and benefits.

Accumulated Other Comprehensive Loss-Recognition and Deferrals

The following table presents cumulative items not recognized as a component of net periodic benefits expense as of December 31, 2017, items recognized as a component of net periodic benefits expense in 2018, additional items deferred during 2018 and cumulative items not recognized as a component of net periodic benefits expense as of December 31, 2018. The items not recognized as a component of net periodic benefits expense have been recorded on our consolidated balance sheets in accumulated other comprehensive loss:

| | As of and for the Years Ended December 31, | | | | | | |
|--|--|---------|--|--------------------------|-------|---------|--|
| | | 2017 | Recognition of Net Periodic Benefits Expense | Net Change in AOCL | 2018 | | |
| | | | | (Dollars in millions) | | | |
| Accumulated other comprehensive loss: | | | | | | | |
| Pension plans: | | | | | | | |
| Net actuarial (loss) gain | \$ | (2,892) | 179 | (260) | (81) | (2,973) | |
| Prior service benefit (cost) | | 54 | (8) | _ | (8) | 46 | |
| Deferred income tax benefit (expense) (1) | | 1,107 | (418) | 65 | (353) | 754 | |
| Total pension plans | | (1,731) | (247) | (195) | (442) | (2,173) | |
| Post-retirement benefit plans: | | | | | | | |
| Net actuarial (loss) gain | | (250) | _ | 257 | 257 | 7 | |
| Prior service (cost) benefit | | (107) | 20 | _ | 20 | (87) | |
| Deferred income tax benefit (expense) (2) | | 122 | (37) | (63) | (100) | 22 | |
| Total post-retirement benefit plans | | (235) | (17) | 194 | 177 | (58) | |
| Total accumulated other comprehensive loss | \$ | (1,966) | (264) | (1) | (265) | (2,231) | |

⁽¹⁾ Amounts currently recognized in net periodic benefits expense include \$375 million of benefit arising from the adoption of ASU 2018-02. See Note 1— Background and Summary of Significant Accounting Policies for further detail.

(2) Amounts currently recognized in net periodic benefits expense include \$32 million arising from the adoption of ASU 2018-02. See Note 1— Background and Summary of Significant Accounting Policies for further detail.

The following table presents cumulative items not recognized as a component of net periodic benefits expense as of December 31, 2016, items recognized as a component of net periodic benefits expense in 2017, additional items deferred during 2017 and cumulative items not recognized as a component of net periodic benefits expense as of December 31, 2016. The items not recognized as a component of net periodic benefits expense have been recorded on our consolidated balance sheets in accumulated other comprehensive loss:

| | As of and for the Years Ended December 31, | | | | | | | |
|---------------------------------------|--|--|----------------------|--------------------------|---------|--|--|--|
| | 2016 | Recognition of Net Periodic Benefits Expense | Deferrals | Net Change in AOCL | 2017 | | | |
| | | 1) | Dollars in millions) | | | | | |
| Accumulated other comprehensive loss: | | | | | | | | |
| Pension plans: | | | | | | | | |
| Net actuarial (loss) gain | \$ (3,148) | 205 | 51 | 256 | (2,892) | | | |
| Prior service benefit (cost) | 62 | (8) | _ | (8) | 54 | | | |
| Deferred income tax benefit (expense) | 1,191 | (72) | (12) | (84) | 1,107 | | | |
| Total pension plans | (1,895) | 125 | 39 | 164 | (1,731) | | | |
| Post-retirement benefit plans: | | | | | | | | |
| Net actuarial (loss) gain | (137) | _ | (113) | (113) | (250) | | | |
| Prior service (cost) benefit | (127) | 20 | _ | 20 | (107) | | | |
| Deferred income tax benefit (expense) | 102 | (7) | 27 | 20 | 122 | | | |
| Total post-retirement benefit plans | (162) | 13 | (86) | (73) | (235) | | | |
| Total accumulated other comprehensive | \$ (2.057) | 138 | (47) | 91 | (1.966) | | | |

Medicare Prescription Drug, Improvement and Modernization Act of 2003

We sponsor post-retirement health care plans with several benefit options that provide prescription drug benefits that we deem actuarially equivalent to or exceeding Medicare Part D. We recognize the impact of the federal subsidy received under the Medicare Prescription Drug, Improvement and Modernization Act of 2003 in the calculation of our post-retirement benefit obligation and net periodic post-retirement benefit expense.

Other Benefit Plans

Health Care and Life Insurance

We provide health care and life insurance benefits to essentially all of our active employees. We are largely self-funded for the cost of the health care plan. Our health care benefit expense for current employees was \$434 million , \$341 million and \$399 million for the years ended December 31, 2018 , 2017 and 2016 , respectively. Union-represented employee benefits are based on negotiated collective bargaining agreements. Employees contributed \$142 million , \$128 million and \$127 million for the years ended December 31, 2018 , 2017 and 2016 , respectively. Our group basic life insurance plans are fully insured and the premiums are paid by us.

401(k) Plans

We sponsor qualified defined contribution plans covering substantially all of our employees. Under these plans, employees may contribute a percentage of their annual compensation up to certain maximums, as defined by the plans and by the Internal Revenue Service ("IRS"). Currently, we match a percentage of employee contributions in cash. At December 31, 2018 and 2017, the assets of the plans included approximately 12 million shares and 7 million shares, respectively, of our common stock all of which were the result of the combination of previous employer match and participant directed contributions. We recognized expenses related to these plans of \$93 million, \$77 million and \$79 million for the years ended December 31, 2018, 2017 and 2016, respectively.

Upon the November 1, 2017 closing of our acquisition of Level 3, we assumed various defined contribution plans covering substantially all eligible employees of Level 3. On December 31, 2017, we merged the Level 3 Communications, Inc. 401(k) Plan into the CenturyLink Dollar & Sense 401(k) Plan. The resulting plan covers substantially all eligible non-represented employees of the combined company in the US.

Deferred Compensation Plans

We sponsored non-qualified deferred compensation plans for various groups that included certain of our current and former highly compensated employees. The value of liabilities related to these plans was not significant.

(11) Share-based Compensation

We maintain an equity incentive program that allows our Board of Directors (through its Compensation Committee or our Chief Executive Officer as its delegate) to grant incentives to certain employees and outside directors in one or more forms, including: incentive and non-qualified stock options, stock appreciation rights, restricted stock awards, restricted stock units and market and performance shares. Stock options generally expire ten years from the date of grant.

Acquisition of Level 3

Upon the November 1, 2017 acquisition of Level 3, and pursuant to the terms of the merger agreement, we assumed certain of Level 3's share-based compensation awards, which were converted to settle in shares of CenturyLink common stock. Specifically:

- each outstanding Level 3 restricted stock unit award granted prior to April 1, 2014 or granted to an outside director of Level 3 was
 converted into \$26.50 in cash and 1.4286 shares of our common stock (and cash in lieu of fractional shares) with respect to each Level
 3 share covered by such award (the "Converted RSU Awards"); and
- each outstanding Level 3 restricted stock unit award granted on or after April 1, 2014 (other than these granted to outside directors of Level 3) was converted into a CenturyLink restricted stock unit award using a conversion ratio of 2.8386 to 1 as determined in accordance with a formula set forth in the merger agreement (the "Continuing RSU Awards").

The aggregate fair value of the replaced Level 3 awards was \$239 million , of which \$103 million was attributable to service performed prior to the acquisition date and was included in the cost of the acquisition. The fair value of CenturyLink shares was determined based on the \$18.99 closing price of our common stock on November 1, 2017. The remaining \$137 million of the preliminary aggregate fair value of the replaced Level 3 awards was attributable to post-acquisition period and is being recognized as compensation expense, net of estimated forfeitures, over the remaining 1 to 2 year vesting period.

Stock Options

The following table summarizes activity involving stock option awards for the year ended December 31, 2018:

| | Number of Options (in thousands) | Weighted- Average Exercise Price |
|--|----------------------------------|---|
| Outstanding and Exercisable at December 31, 2017 | 1,022 | \$ 27.41 |
| Exercised | (178) | 22.49 |
| Forfeited/Expired | (301) | 30.25 |
| Outstanding and Exercisable at December 31, 2018 | 543 | 27.46 |

The aggregate intrinsic value of our options outstanding and exercisable at December 31, 2018 was less than \$1 million . The weighted-average remaining contractual term for such options was 1.12 years .

During 2018, we received net cash proceeds of \$4 million in connection with our option exercises. The tax benefit realized from these exercises was less than \$1 million. The total intrinsic value of options exercised for the years ended December 31, 2018, 2017 and 2016, was less than \$1 million each year.

Restricted Stock Awards and Restricted Stock Unit Awards

For equity based restricted stock and restricted stock unit awards that contain only service conditions for vesting (time-based awards), we calculate the award fair value based on the closing price of CenturyLink common stock on the accounting grant date. We also grant equity-based awards that contain service conditions as well as additional market or performance conditions. For awards having both service and market conditions, the award fair value is calculated using Monte-Carlo simulations. Awards with service as well as market or performance conditions specify a target number of shares for the award, although each recipient ultimately has the opportunity to receive between 0% and 200% of the target number of shares. For awards with service and market conditions, the percentage received is based on our total shareholder return over the three -year service period versus that of selected peer companies. For awards with service and performance conditions, the percentage received depends upon the attainment of two financial performance targets during the three -year service period.

The following table summarizes activity involving restricted stock and restricted stock unit awards for the year ended December 31, 2018:

| | Number of Shares | Weighted- Average Grant Date Fair Value |
|---------------------------------|---------------------|--|
| | (in thousands) | _ |
| Non-vested at December 31, 2017 | 19,774 | \$ 21.90 |
| Granted (1) | 9,657 | 17.02 |
| Vested | (9,275) | 20.87 |
| Forfeited | (3,097) | 22.12 |
| Non-vested at December 31, 2018 | 17,059 | 19.65 |

⁽¹⁾ Shares granted whose related performance conditions were not finalized at December 31, 2018, were excluded from this figure.

During 2017, we granted 5.2 million shares of restricted stock and restricted stock unit awards at a weighted-average price of \$22.02. During 2016, we granted 3.6 million shares of restricted stock and restricted stock unit awards at a weighted-average price of \$30.83. The total fair value of restricted stock that vested during 2018, 2017 and 2016, was \$169 million, \$60 million and \$47 million, respectively.

Compensation Expense and Tax Benefit

We recognize compensation expense related to our market and performance share-based awards with graded vesting that only have a service condition on a straight-line basis over the requisite service period for the entire award. Total compensation expense for all share-based payment arrangements for the years ended December 31, 2018, 2017 and 2016, was \$185 million, \$111 million and \$80 million, respectively. Our tax benefit recognized in the consolidated statements of operations for our share-based payment arrangements for the years ended December 31, 2018, 2017 and 2016, was \$46 million, \$28 million and \$31 million, respectively. At December 31, 2018, there was \$215 million of total unrecognized compensation expense related to our share-based payment arrangements, which we expect to recognize over a weighted-average period of 1.7 years.

(12) (Loss) Earnings Per Common Share

Basic and diluted (loss) earnings per common share for the years ended December 31, 2018, 2017 and 2016 were calculated as follows:

| | Years Ended December 31, | | | | |
|---|--------------------------|---------------------|-----------------------------------|----------------|--|
| | 2018 | | 2017 | 2016 | |
| | (0 | ollars in millions, | except per share an in thousands) | mounts, shares | |
| Loss income (Numerator): | | | | | |
| Net (loss) income | \$ | (1,733) | 1,389 | 626 | |
| Net (loss) income applicable to common stock for computing basic earnings per common share | | (1,733) | 1,389 | 626 | |
| Net (loss) income as adjusted for purposes of computing diluted earnings per common share | \$ | (1,733) | 1,389 | 626 | |
| Shares (Denominator): | | | | | |
| Weighted average number of shares: | | | | | |
| Outstanding during period | | 1,078,409 | 635,576 | 545,946 | |
| Non-vested restricted stock | | (12,543) | (7,768) | (6,397) | |
| Weighted average shares outstanding for computing basic earnings per common share | | 1,065,866 | 627,808 | 539,549 | |
| Incremental common shares attributable to dilutive securities: | | | | | |
| Shares issuable under convertible securities | | _ | 10 | 10 | |
| Shares issuable under incentive compensation plans | | _ | 875 | 1,120 | |
| Number of shares as adjusted for purposes of computing diluted (loss) earnings per common share | | 1,065,866 | 628,693 | 540,679 | |
| Basic (loss) earnings per common share | \$ | (1.63) | 2.21 | 1.16 | |
| Diluted (loss) earnings per common share (1) | \$ | (1.63) | 2.21 | 1.16 | |

⁽¹⁾ For the year ended December 31, 2018, we excluded from the calculation of diluted loss per share 4.6 million shares potentially issuable under incentive compensation plans or convertible securities, as their effect, if included, would have been anti-dilutive.

Our calculation of diluted (loss) earnings per common share excludes shares of common stock that are issuable upon exercise of stock options when the exercise price is greater than the average market price of our common stock. We also exclude unvested restricted stock awards that are antidilutive as a result of unrecognized compensation cost. Such shares averaged 2.7 million , 4.7 million and 3.3 million for 2018 , 2017 and 2016 , respectively.

(13) Fair Value of Financial Instruments

Our financial instruments consist of cash, cash equivalents and restricted cash, accounts receivable, accounts payable and long-term debt, excluding capital lease and other obligations. Due to their short-term nature, the carrying amounts of our cash, cash equivalents and restricted cash, accounts receivable and accounts payable approximate their fair values.

Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between independent and knowledgeable parties who are willing and able to transact for an asset or liability at the measurement date. We use valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs when determining fair value and then we rank the estimated values based on the reliability of the inputs used following the fair value hierarchy set forth by the FASB.

We determined the fair values of our long-term debt, including the current portion, based on quoted market prices where available or, if not available, based on discounted future cash flows using current market interest rates.

The three input levels in the hierarchy of fair value measurements are defined by the FASB generally as follows:

| Input Level | Description of Input |
|-------------|--|
| Level 1 | Observable inputs such as quoted market prices in active markets. |
| Level 2 | Inputs other than quoted prices in active markets that are either directly or indirectly observable. |
| Level 3 | Unobservable inputs in which little or no market data exists. |

The following table presents the carrying amounts and estimated fair values of our long-term debt, excluding capital lease and other obligations, as well as the input level used to determine the fair values indicated below:

| | | | As of Decem | ber 31, 2018 | As of Decem | nber 31, 2017 | |
|---|----------------|-----------------------|-------------|--------------|--------------------|---------------|--|
| | Input Level | Carrying Amount | | Fair Value | Carrying Amount | Fair Value | |
| | | (Dollars in millions) | | | | | |
| Liabilities-Long-term debt, excluding capital lease and other obligations | 2 | \$ | 35,260 | 32,915 | 36,835 | 36,402 | |

(14) Income Taxes

On December 22, 2017, the Tax Cuts and Jobs Act (the "Act") was signed into law. The Act reduces the U.S. corporate income tax rate from a maximum of 35% to 21% for all corporations, effective January 1, 2018, and makes certain changes to U.S. taxation of income earned by foreign subsidiaries, capital expenditures, interest expense and various other items.

As a result of the reduction in the U.S. corporate income tax rate from 35% to 21%, we re-measured our net deferred tax liabilities at December 31, 2017 and recognized a provisional tax benefit of approximately \$1.1 billion in our consolidated statement of operations for the year ended December 31, 2017. As a result of finalizing our provisional amount recorded in 2017, we recorded a reduction to this amount for purchase price accounting adjustments resulting from the Level 3 acquisition and the tax reform impact on those adjustments of \$92 million in 2018.

The Act also includes certain anti-abuse and base erosion provisions that may impact the amounts of U.S. tax that we pay with respect to income earned by our foreign subsidiaries. We have completed our analysis of the impact of the one-time repatriation tax and concluded that we do not have a tax liability under this provision. We have also completed our analysis of the anti-abuse and base erosion provisions and have recorded a tax expense of \$11 million related to global intangible low-taxed income provisions of the Act and do not have a liability in relation to base erosion and anti-abuse tax provisions of the Act.

| | Years Ended December 31, | | |
|--|--------------------------|-----------------------|------|
| | 2018 | 2017 | 2016 |
| | | (Dollars in millions) |) |
| Income tax expense (benefit) was as follows: | | | |
| Federal | | | |
| Current | \$ (576) | 82 | 335 |
| Deferred | 734 | (988) | 5 |
| State | | | |
| Current | (22) | 21 | 27 |
| Deferred | 52 | 16 | 8 |
| Foreign | | | |
| Current | 36 | 22 | 26 |
| Deferred | (54) | (2) | (7) |
| Total income tax expense (benefit) | \$ 170 | (849) | 394 |

| | Years Ended December 31, | | | | |
|--|--------------------------|--------|-----------------|---------|--|
| | | 2018 | 2017 | 2016 | |
| | | (Dolla | rs in millions) | llions) | |
| Income tax (benefit) expense was allocated as follows: | | | | | |
| Income tax (benefit) expense in the consolidated statements of operations: | | | | | |
| Attributable to income | \$ | 170 | (849) | 394 | |
| Stockholders' equity: | | | | | |
| Compensation expense for tax purposes in excess of amounts recognized for financial reporting purposes | | _ | _ | (2) | |
| Tax effect of the change in accumulated other comprehensive loss | | (2) | 81 | (109) | |

The following is a reconciliation from the statutory federal income tax rate to our effective income tax rate:

| | Yea | Years Ended December 31, | | | |
|--|----------|--------------------------------|--------|--|--|
| | 2018 | 2017 | 2016 | | |
| | (Perc | (Percentage of pre-tax income) | | | |
| Statutory federal income tax rate | 21.0 % | 35.0 % | 35.0 % | | |
| State income taxes, net of federal income tax benefit | (1.5)% | 3.9 % | 2.3 % | | |
| Impairment of goodwill | (36.6)% | — % | — % | | |
| Change in liability for unrecognized tax position | 1.3 % | 1.0 % | 0.2 % | | |
| Tax reform | (5.9)% | (209.8)% | — % | | |
| Net foreign income taxes | 1.8 % | (0.7)% | 0.1 % | | |
| Foreign dividend paid to a domestic parent company | — % | 0.2 % | 1.8 % | | |
| Research and development credits | 0.9 % | (1.4)% | (0.6)% | | |
| Tax impact on sale of data centers and colocation business | <u> </u> | 5.0 % | — % | | |
| Tax benefit of net operating loss carryback | 9.1 % | — % | — % | | |
| Level 3 acquisition transaction costs | — % | 6.0 % | — % | | |
| Other, net | (1.0)% | 3.6 % | (0.2)% | | |
| Effective income tax rate | (10.9)% | (157.2)% | 38.6 % | | |

The effective tax rate for the year ended December 31, 2018 reflects a \$572 million unfavorable impact of the non-deductible goodwill impairment and a \$92 million unfavorable impact due to finalizing the impacts of tax reform. Partially offsetting these amounts is a \$142 million benefit generated by a loss carryback to 2016. The effective tax rate for the year ended December 31, 2017 reflects the benefit of approximately \$1.1 billion from the re-measurement of deferred taxes as noted above, a \$27 million tax expense related to the sale of our colocation business and \$32 million tax impact of non-deductible transaction costs related to the Level 3 acquisition. The effective tax rate for the year ended December 31, 2016 reflects a tax impact of \$18 million from an intercompany dividend payment from one of our foreign subsidiaries to its domestic parent company that was made as part of our corporate restructuring in preparation for the sale of our colocation business.

The tax effects of temporary differences that gave rise to significant portions of the deferred tax assets and deferred tax liabilities were as follows:

| | As of December 31, | |
|--|--------------------|-----------|
| | 2018 | 2017 |
| | (Dollars in m | nillions) |
| Deferred tax assets | | |
| Post-retirement and pension benefit costs | \$ 1,111 | 1,321 |
| Net operating loss carryforwards | 3,445 | 3,951 |
| Other employee benefits | 162 | 112 |
| Other | 553 | 714 |
| Gross deferred tax assets | 5,271 | 6,098 |
| Less valuation allowance | (1,331) | (1,341) |
| Net deferred tax assets | 3,940 | 4,757 |
| Deferred tax liabilities | | |
| Property, plant and equipment, primarily due to depreciation differences | (3,011) | (2,935) |
| Goodwill and other intangible assets | (3,303) | (3,785) |
| Other | (23) | (16) |
| Gross deferred tax liabilities | (6,337) | (6,736) |
| Net deferred tax liability | \$ (2,397) | (1,979) |

Of the \$2.4 billion and \$2.0 billion net deferred tax liability at December 31, 2018 and 2017, respectively, \$2.5 billion and \$2.4 billion is reflected as a long-term liability and \$131 million and \$434 million is reflected as a net noncurrent deferred tax asset at December 31, 2018 and 2017, respectively.

At December 31, 2018, we had federal NOLs of \$7.3 billion, net of limitations of Section 382 of the Internal Revenue Code ("Section 382") and uncertain tax positions, for U.S. federal income tax purposes. If unused, the NOLs will expire between 2022 and 2037. The U.S. federal net operating loss carryforwards expire as follows:

| | Expiring | Amo | unt |
|-------------------------|--------------|-------------|-----------|
| | December 31, | (Dollars in | millions) |
| 2022 | | \$ | 1,043 |
| 2023 | | | 1,440 |
| 2024 | | | 1,402 |
| 2025 | | | 1,042 |
| 2026 | | | 1,525 |
| 2027 | | | 375 |
| 2028 | | | 637 |
| 2029 | | | 645 |
| 2030 | | | 671 |
| 2031 | | | 732 |
| 2032 | | | 348 |
| 2033 | | | 238 |
| 2037 | | | 2,715 |
| NOLs per return | | | 12,813 |
| Uncertain tax positions | | | (5,526) |
| Financial NOLs | | \$ | 7,287 |

At December 31, 2018 we had state net operating loss carryforwards of \$19 billion (net of uncertain tax positions). We also had foreign NOL carryforwards of \$6 billion . At December 31, 2018 , we had federal tax credits of \$52 million . Our acquisitions of Level 3, Qwest and SAVVIS, Inc. caused "ownership changes" within the meaning of Section 382 for the acquired companies. As a result, our ability to use these NOLs and tax credits are subject to annual limits imposed by Section 382. Despite this, we expect to use substantially all of these tax attributes to reduce our future federal tax liabilities, although the timing of that use will depend upon our future earnings and future tax circumstances.

We establish valuation allowances when necessary to reduce the deferred tax assets to amounts we expect to realize. As of December 31, 2018, a valuation allowance of \$1.3 billion was established as it is more likely than not that this amount of net operating loss, capital loss and tax credit carryforwards will not be utilized prior to expiration. Our valuation allowance at December 31, 2018 and 2017 is primarily related to foreign and state NOL carryforwards. This valuation allowance decreased by \$10 million during 2018, primarily due to the impact of foreign exchange rate adjustments and state law changes.

A reconciliation of the change in our gross unrecognized tax benefits (excluding both interest and any related federal benefit) from January 1 to December 31 for 2018 and 2017 is as follows:

| | | 2018 | 2017 |
|---|-----------------------|-------|------|
| | (Dollars in millions) | | |
| Unrecognized tax benefits at beginning of year | \$ | 40 | 16 |
| Assumed in the acquisition of Level 3 | | _ | 18 |
| Tax position of prior periods netted against deferred tax assets | | 1,338 | 2 |
| Increase in tax positions taken in the current year | | 4 | 1 |
| Increase in tax positions taken in the prior year | | 211 | 3 |
| Decrease due to payments/settlements | | (1) | _ |
| Decrease from the lapse of statute of limitations | | (2) | _ |
| Decrease due to the reversal of tax positions taken in a prior year | | (3) | _ |
| Unrecognized tax benefits at end of year | \$ | 1,587 | 40 |

The total amount (including both interest and any related federal benefit) of unrecognized tax benefits that, if recognized, would impact the effective income tax rate was \$256 million and \$66 million at December 31, 2018 and 2017, respectively.

Our policy is to reflect interest expense associated with unrecognized tax benefits in income tax expense. We had accrued interest (presented before related tax benefits) of approximately \$17 million and \$56 million at December 31, 2018 and 2017, respectively.

We, or at least one of our subsidiaries, file income tax returns in the U.S. federal jurisdiction and various states and foreign jurisdictions. With few exceptions, we are no longer subject to U.S. federal, state and local, or non-U.S. income tax examinations by tax authorities for years before 2003. The Internal Revenue Service and state and local taxing authorities reserve the right to audit any period where net operating loss carryforwards are available.

Based on our current assessment of various factors, including (i) the potential outcomes of these ongoing examinations, (ii) the expiration of statute of limitations for specific jurisdictions, (iii) the negotiated settlement of certain disputed issues, and (iv) the administrative practices of applicable taxing jurisdictions, it is reasonably possible that the related unrecognized tax benefits for uncertain tax positions previously taken may decrease by up to \$12 million within the next 12 months. The actual amount of such decrease, if any, will depend on several future developments and events, many of which are outside our control.

(15) Segment Information

At December 31, 2018, we had the following two segments:

- Business Segment. Under our business segment, we provide our products and services to large domestic and global enterprises, small
 and medium businesses, federal, state and local governments and wholesale customers, including other communication providers. Our
 products and services offered to these customers include our IP and Data Services suite of products, which includes VPN and hybrid
 networking, Ethernet and IP services; Transport and Infrastructure, which includes wavelengths and private line, dark fiber, colocation
 and data center services, and professional services; Voice Services, which includes local, long-distance, toll-free and unified
 communications services; and IT and Managed services, all of which are described further under "Products and Services"; and
- Consumer Segment. Under our consumer segment, we provide our products and services to residential customers. Our products and services offered to these customers include our broadband, local and long-distance voice, video and other ancillary services.

The following table summarizes our segment results for 2018, 2017 and 2016 based on the segment categorization we were operating under at December 31, 2018.

| | Years Ended December 31, | | |
|-------------------------------|--------------------------|---------------------|--------|
| | 2018 | 2017 | 2016 |
| | (Do | ollars in millions) | |
| Total segment revenue | \$ 22,720 | 16,924 | 16,766 |
| Total segment expenses | 12,269 | 9,390 | 9,081 |
| Total segment adjusted EBITDA | \$ 10,451 | 7,534 | 7,685 |
| Total margin percentage | 46% | 45% | 46% |
| Business segment: | | | |
| Revenue | \$ 17,349 | 11,220 | 10,704 |
| Expenses | 10,076 | 6,847 | 6,391 |
| Adjusted EBITDA | \$ 7,273 | 4,373 | 4,313 |
| Margin percentage | 42% | 39% | 40% |
| Consumer segment: | | | |
| Revenue | \$ 5,371 | 5,704 | 6,062 |
| Expenses | 2,193 | 2,543 | 2,690 |
| Adjusted EBITDA | \$ 3,178 | 3,161 | 3,372 |
| Margin percentage | 59% | 55% | 56% |

Product and Service Categories

We categorize our products, services and revenue among the following five categories:

- IP and Data Services, which include primarily VPN data networks, Ethernet, IP, video (including our facilities-based video services, CDN services and Vyvx broadcast services) and other ancillary services;
- Transport and Infrastructure, which include broadband, private line (including business data services), data center facilities and services, including cloud, hosting and application management solutions, wavelength, equipment sales and professional services, network security services, dark fiber services and other ancillary services;
- Voice and Collaboration, which includes primarily local and long-distance voice, including wholesale voice, and other ancillary services;
- IT and Managed Services, which include information technology services and managed services, which may be purchased in conjunction with our other network services; and
- Regulatory Revenue, which consist of (i) Universal Service Fund ("USF"), Connect America Fund ("CAF") and other support payments designed to reimburse us for various costs related to certain telecommunications services and (ii) other operating revenue from the leasing and subleasing of space, none of which is included in our segment revenue.

Our operating revenue for our products and services are presented as follows for the years ended December 31, 2018, 2017 and 2016:

Vegre Ended December 21

| | Years Ended December 31, | | | |
|----------------------------------|--------------------------|--------|---------------------|--------|
| | | 2018 | 2017 | 2016 |
| | | (D | ollars in millions) | |
| Business segment | | | | |
| IP and Data Services (1) | \$ | 6,971 | 3,682 | 2,957 |
| Transport and Infrastructure (2) | | 5,356 | 3,569 | 3,807 |
| Voice and Collaboration (3) | | 4,401 | 3,317 | 3,299 |
| IT and Managed Services (4) | | 621 | 652 | 641 |
| Total business segment revenue | | 17,349 | 11,220 | 10,704 |
| | | | | |
| Consumer segment | | | | |
| IP and Data Services (5) | | 308 | 401 | 461 |
| Transport and Infrastructure (6) | | 2,892 | 2,776 | 2,776 |
| Voice and Collaboration (3) | | 2,171 | 2,527 | 2,825 |
| Total consumer segment revenue | | 5,371 | 5,704 | 6,062 |
| | | | | |
| Non-segment revenue | | | | |
| Regulatory Revenue (7) | | 723 | 732 | 704 |
| Total non-segment revenue | | 723 | 732 | 704 |
| | | | | |
| Total revenue | \$ | 23,443 | 17,656 | 17,470 |

- (1) Includes primarily VPN data network, Ethernet, IP and ancillary revenue.
- (2) Includes primarily broadband, private line (including business data services), colocation and data centers, wavelength and ancillary revenue.
- (3) Includes local, long-distance and other ancillary revenue.
- (4) Includes IT services and managed services revenue.
- (5) Includes retail video revenue (including our facilities-based video revenue).
- (6) Includes primarily broadband and equipment sales and professional services revenue.
- (7) Includes CAF Phase I, CAF Phase II, federal and state USF support revenue, sublease rental income and failed-sale leaseback income.

We recognize revenue in our consolidated statements of operations for certain USF surcharges and transaction taxes that we bill to our customers. Our consolidated statements of operations also reflect the offsetting expense for the amounts we remit to the government agencies. The total amount of such surcharges and transaction taxes that we included in revenue aggregated to \$952 million , \$601 million and \$572 million for the years ended December 31, 2018 , 2017 and 2016 , respectively. These USF surcharges, where we record revenue, and transaction taxes are assigned to the product and service categories of each segments based on the underlying revenue. We also act as a collection agent for certain other USF and transaction taxes that we are required by government agencies to bill our customers, for which we do not record any revenue or expense because we only act as a pass-through agent.

Allocations of Revenue and Expenses

Our segment revenue includes all revenue from our business and consumer segments as described in more detail above. Our segment revenue is based upon each customer's classification. We report our segment revenue based upon all services provided to that segment's customers. Our segment expenses include specific expenses incurred as a direct result of providing services and products to segment customers, along with selling, general and administrative expenses that are (i) directly associated with specific segment customers or activities, and (ii) allocated expenses which include network expenses, facilities expenses and other expenses such as fleet and real estate expenses. We do not assign depreciation and amortization expense or impairments to our segments, as the related assets and capital expenditures are centrally managed and are not monitored by or reported to the CODM by segment. Generally speaking, severance expenses, restructuring expenses and certain centrally managed administrative functions (such as finance, information technology, legal and human resources) are not assigned to our segments. Interest expense is also excluded from segment results because we manage our financing on a consolidated basis and have not allocated assets or debt to specific segments. Other income and expense items are not monitored as a part of our segment operations and are therefore excluded from our segment results.

The following table reconciles total segment adjusted EBITDA to net (loss) income for the years ended December 31, 2018, 2017 and 2016:

| | Years Ended December 31, | | | |
|---|--------------------------|---------|-----------------------|---------|
| | 2018 | | 2017 | 2016 |
| | | (D | (Dollars in millions) | |
| Total segment adjusted EBITDA | \$ | 10,451 | 7,534 | 7,685 |
| Regulatory Revenue | | 723 | 732 | 704 |
| Depreciation and amortization | | (5,120) | (3,936) | (3,916) |
| Impairment of goodwill | | (2,726) | _ | _ |
| Other operating expenses | | (2,758) | (2,321) | (2,140) |
| Total other expenses, net | | (2,133) | (1,469) | (1,313) |
| Income (loss) before income tax expense | | (1,563) | 540 | 1,020 |
| Income tax (expense) benefit | | (170) | 849 | (394) |
| Net (loss) income | \$ | (1,733) | 1,389 | 626 |

We do not have any single customer that provides more than 10% of our consolidated total operating revenue.

The assets we hold outside of the U.S. represent less than 10% of our total assets. Revenue from sources outside of the U.S. is responsible for less than 10% of our total operating revenue.

(16) Quarterly Financial Data (Unaudited)

| | First Quarter | Second Quarter | Third Quarter | Fourth Quarter | Total |
|--|------------------|--------------------|--------------------|-------------------|---------|
| | | (Dollars in millio | ns, except per sha | re amounts) | |
| 2018 | | | | | |
| Operating revenue | \$ 5,945 | 5,902 | 5,818 | 5,778 | 23,443 |
| Operating income (loss) | 750 | 767 | 894 | (1,841) | 570 |
| Net income (loss) | 115 | 292 | 272 | (2,412) | (1,733) |
| Basic earnings (loss) per common share | 0.11 | 0.27 | 0.25 | (2.26) | (1.63) |
| Diluted earnings (loss) per common share | 0.11 | 0.27 | 0.25 | (2.26) | (1.63) |
| 2017 | | | | | |
| Operating revenue | \$ 4,209 | 4,090 | 4,034 | 5,323 | 17,656 |
| Operating income | 631 | 367 | 487 | 524 | 2,009 |
| Net income | 163 | 17 | 92 | 1,117 | 1,389 |
| Basic earnings per common share | 0.30 | 0.03 | 0.17 | 1.26 | 2.21 |
| Diluted earnings per common share | 0.30 | 0.03 | 0.17 | 1.26 | 2.21 |

During the fourth quarter of 2018, we recorded a non-cash, non-tax-deductible goodwill impairment charge of \$2.7 billion for goodwill assigned to our consumer segment.

During the first quarter of 2018, we recognized \$71 million of expenses related to our acquisition of Level 3 followed by acquisition-related expenses of \$162 million, \$43 million and \$117 million in the second, third and fourth quarters of 2018, respectively.

During the first quarter of 2017, we recognized \$10 million of expenses related to our acquisition of Level 3 followed by acquisition-related expenses of \$18 million , \$37 million and \$206 million in the second, third and fourth quarters of 2017, respectively. During the first quarter of 2017, depreciation and amortization expense of \$50 million was not recognized on colocation assets held for sale. During the second quarter, we recognized a combined loss of \$119 million resulting from the sale of the colocation business and data centers and the accounting treatment of the failed-sale-leaseback. During the second quarter of 2017, we recognized a one-time depreciation charge of \$44 million related to the failed-sale-leaseback accounting. During the third and fourth quarters of 2017, we recognized \$44 million and \$20 million , respectively, of interest expense related to CenturyLink, Inc.'s \$6 billion secured term loan utilized in the acquisition of Level 3. In the fourth quarter of 2017, we recognized a tax benefit of approximately \$1.1 billion due to the change in the federal corporate tax rate from 35% to 21%.

(17) Commitments, Contingencies and Other Items

We are subject to various claims, legal proceedings and other contingent liabilities, including the matters described below, which individually or in the aggregate could materially affect our financial condition, future results of operations or cash flows. As a matter of course, we are prepared to both litigate these matters to judgment as needed, as well as to evaluate and consider reasonable settlement opportunities.

Irrespective of its merits, litigation may be both lengthy and disruptive to our operations and could cause significant expenditure and diversion of management attention. We review our litigation accrual liabilities on a quarterly basis, but in accordance with applicable accounting guidelines only establish accrual liabilities when losses are deemed probable and reasonably estimable and only revise previously-established accrual liabilities when warranted by changes in circumstances, in each case based on then-available information. As such, as of any given date we could have exposure to losses under proceedings as to which no liability has been accrued or as to which the accrued liability is inadequate. Amounts accrued for our litigation and non-income tax contingencies at December 31, 2018 aggregated to approximately \$123 million and are included in other current liabilities and other liabilities in our consolidated balance sheet as of such date. The establishment of an accrual does not mean that actual funds have been set aside to satisfy a given contingency. Thus, the resolution of a particular contingency for the amount accrued could have no effect on our results of operations but nonetheless could have an adverse effect on our cash flows.

In this Note, when we refer to a class action as "putative" it is because a class has been alleged, but not certified in that matter.

Shareholder Class Action Suits

CenturyLink and certain members of the CenturyLink Board of Directors have been named as defendants in a putative shareholder class action lawsuit filed on January 11, 2017 in the 4th Judicial District Court of the State of Louisiana, Ouachita Parish, captioned Jeffery Tomasulo v. CenturyLink, Inc., et al. The complaint asserts, among other things, that the members of CenturyLink's Board allegedly breached their fiduciary duties to the CenturyLink shareholders in approving the Level 3 merger agreement and, more particularly, that: the consideration that CenturyLink agreed to pay to Level 3 stockholders in the transaction is allegedly unfairly high; the CenturyLink directors allegedly had conflicts of interest in negotiating and approving the transaction; and the disclosures set forth in our preliminary joint proxy statement/prospectus filed in December 2016 are insufficient in that they allegedly fail to contain material information concerning the transaction. The complaint seeks, among other things, a declaration that the members of the CenturyLink Board have breached their fiduciary duties, corrective disclosure, rescissory or other damages and equitable relief, including rescission of the transaction. In February 2017, the parties entered into a memorandum of understanding providing for the settlement of the lawsuit. In January 2019, the court approved the settlement and entered final judgment. An objector filed an appeal, and that appeal is pending. The costs of the settlement are not material to our consolidated financial statements.

CenturyLink and certain CenturyLink board members and officers were named as defendants in a putative shareholder class action lawsuit filed on June 12, 2018 in the Boulder County District Court of the state of Colorado, captioned Houser et al. v. CenturyLink, et al. The complaint asserts claims on behalf of a putative class of former Level 3 shareholders who became CenturyLink shareholders as a result of the transaction. It alleges that the proxy statement provided to the Level 3 shareholders failed to disclose material information of several kinds, including information about strategic revenue, customer loss rates, and customer account issues, among other items. The complaint seeks damages, costs and fees, rescission, rescissory damages, and other equitable relief.

Switched Access Disputes

Subsidiaries of CenturyLink, Inc. are among hundreds of companies involved in an industry-wide dispute, raised in nearly 100 federal lawsuits (filed between 2014 and 2016) that have been consolidated in the United States District Court for the Northern District of Texas for pretrial procedures. The disputes relate to switched access charges that local exchange carriers ("LECs") collect from interexchange carriers ("IXCs") for IXCs' use of LEC's access services. In the lawsuits, IXCs, including Sprint Communications Company L.P. ("Sprint") and various affiliates of Verizon Communications Inc. ("Verizon"), assert that federal and state laws bar LECs from collecting access charges when IXCs exchange certain types of calls between mobile and wireline devices that are routed through an IXC. Some of these IXCs have asserted claims seeking refunds of payments for access charges previously paid and relief from future access charges.

In November 2015, the federal court agreed with the LECs and rejected the IXCs' contention that federal law prohibits these particular access charges, and also allowed the IXCs to refile state-law claims. Since then, many of the LECs and IXCs have filed revised pleadings and additional motions, which remain pending. Separately, some of the defendants, including CenturyLink, Inc.'s LECs, have petitioned the FCC to address these issues on an industry-wide basis.

Our subsidiaries include both IXCs and LECs which respectively pay and assess significant amounts of the charges in question. The outcome of these disputes and lawsuits, as well as any related regulatory proceedings that could ensue, are currently not predictable.

State Tax Suits

Several Missouri municipalities have, beginning in May 2012, asserted claims alleging underpayment of taxes against CenturyLink, Inc. and several of its subsidiaries in a number of proceedings filed in the Circuit Court of St. Louis County, Missouri. These municipalities are seeking, among other things, declaratory relief regarding the application of business license and gross receipts taxes and back taxes from 2007 to the present, plus penalties and interest. In a February 2017 ruling in connection with one of these pending cases, the court entered an order awarding plaintiffs \$4 million and broadening the tax base on a going-forward basis. We have appealed that ruling. In a June 2017 ruling in connection with another one of these pending cases, the court made findings which, if not overturned, will result in a tax liability to us well in excess of the contingent liability we have established. In due course, we plan to appeal that decision. We continue to vigorously defend against these claims.

Billing Practices Suits

In June 2017, a former employee filed an employment lawsuit against us claiming that she was wrongfully terminated for alleging that we charged some of our retail customers for products and services they did not authorize. Starting shortly thereafter and continuing since then, and based in part on the allegations made by the former employee, several legal proceedings have been filed.

In June 2017, McLeod v. CenturyLink, a putative consumer class action, was filed against us in the U.S. District Court for the Central District of California alleging that we charged some of our retail customers for products and services they did not authorize. A number of other complaints asserting similar claims have been filed in other federal and state courts, as well. The lawsuits assert claims including fraud, unfair competition, and unjust enrichment. Also in June 2017, Craig. v. CenturyLink, Inc., et al., a putative securities investor class action, was filed in U.S. District Court for the Southern District of New York, alleging that we failed to disclose material information regarding improper sales practices, and asserting federal securities law claims. A number of other cases asserting similar claims have also been filed.

Beginning June 2017, we also received several shareholder derivative demands addressing related topics. In August 2017, the Board of Directors formed a special litigation committee of outside directors to address the allegations of impropriety contained in the shareholder derivative demands. In April 2018, the special litigation committee concluded its review of the derivative demands and declined to take further action. Since then, derivative cases were filed. Two of these cases, Castagna v. Post and Pinsly v. Post, were filed in Louisiana state court in the Fourth Judicial District Court for the Parish of Ouachita. The remaining derivative cases were filed in federal court in Louisiana and Minnesota. These cases have been brought on behalf of CenturyLink against certain current and former officers and directors of the Company and seek damages for alleged breaches of fiduciary duties.

The consumer putative class actions, the securities investor putative class actions, and the federal derivative actions have been transferred to the U.S. District Court for the District of Minnesota for coordinated and consolidated pretrial proceedings as In Re: CenturyLink Sales Practices and Securities Litigation.

In July 2017, the Minnesota state attorney general filed State of Minnesota v. CenturyTel Broadband Services LLC, et al. in the Anoka County Minnesota District Court, alleging claims of fraud and deceptive trade practices relating to improper consumer sales practices. The suit seeks an order of restitution on behalf of all CenturyLink customers, civil penalties, injunctive relief, and costs and fees. Additionally, we have received and responded to information requests and inquiries from other states.

Peruvian Tax Litigation

In 2005, the Peruvian tax authorities ("SUNAT") issued tax assessments against one of our Peruvian subsidiaries asserting \$26 million of additional income tax withholding and value-added taxes ("VAT"), penalties and interest for calendar years 2001 and 2002 on the basis that the Peruvian subsidiary incorrectly documented its importations. After taking into account the developments described below, as well as the accrued interest and foreign exchange effects, we believe the total amount of exposure is \$11 million at December 31, 2018.

We challenged the assessments via administrative and then judicial review processes. In October 2011, the highest administrative review tribunal (the "Tribunal") decided the central issue underlying the 2002 assessments in SUNAT's favor. We appealed the Tribunal's decision to the first judicial level, which decided the central issue in favor of Level 3. SUNAT and we filed cross-appeals with the court of appeal. In May 2017, the court of appeal issued a decision reversing the first judicial level. In June 2017, we filed an appeal of the decision to the Supreme Court of Justice, the final judicial level. Oral argument was held before the Supreme Court of Justice in October 2018. A decision on this case is pending.

In October 2013, the Tribunal decided the central issue underlying the 2001 assessments in SUNAT's favor. We appealed that decision to the first judicial level in Peru, which decided the central issue in favor of SUNAT. In June 2017, we filed an appeal with the court of appeal. In November 2017, the court of appeals issued a decision affirming the first judicial level and we filed an appeal of the decision to the Supreme Court of Justice. That appeal is pending.

Brazilian Tax Claims

In December 2004, March 2009, April 2009 and July 2014, the São Paulo state tax authorities issued tax assessments against one of our Brazilian subsidiaries for the Tax on Distribution of Goods and Services ("ICMS") with respect to revenue from leasing certain assets (in the case of the December 2004, March 2009 and July 2014 assessments) and revenue from the provision of Internet access services (in the case of the April 2009 and July 2014 assessments), by treating such activities as the provision of communications services, to which the ICMS tax applies. In September 2002, July 2009 and May 2012, the Rio de Janeiro state tax authorities issued tax assessments to the same Brazilian subsidiary on similar issues.

We have filed objections to these assessments, arguing that the lease of assets and the provision of Internet access are not communication services subject to ICMS. The objections to the September 2002, December 2004 and March 2009 assessments were rejected by the respective state administrative courts, and we have appealed those decisions to the judicial courts. In October 2012 and June 2014, we received favorable rulings from the lower court on the December 2004 and March 2009 assessments regarding equipment leasing, but those rulings are subject to appeal by the state. No ruling has been obtained with respect to the September 2002 assessment. The objections to the April and July 2009 and May 2012 assessments are still pending final administrative decisions. The July 2014 assessment was confirmed during the fourth quarter of 2014 at the first administrative level, and we appealed this decision to the second administrative level.

We are vigorously contesting all such assessments in both states and, in particular, view the assessment of ICMS on revenue from equipment leasing to be without merit. These assessments, if upheld, could result in a loss of up to \$37 million at December 31, 2018 in excess of the accruals established for these matters.

Qui Tam Action

Level 3 was notified in late 2017 of a qui tam action pending against Level 3 Communications, Inc. and others in the United States District Court for the Eastern District of Virginia, captioned United States of America ex rel., Stephen Bishop v. Level 3 Communications, Inc. et al. The original qui tam complaint was filed under seal on November 26, 2013, and an amended complaint was filed under seal on June 16, 2014. The court unsealed the complaints on October 26, 2017.

The amended complaint alleges that Level 3, principally through two former employees, submitted false claims and made false statements to the government in connection with two government contracts. The relator seeks damages in this lawsuit of approximately \$50 million, subject to trebling, plus statutory penalties, pre-and-post judgment interest, and attorney's fees. The case is currently stayed.

Level 3 is evaluating its defenses to the claims. At this time, Level 3 does not believe it is probable Level 3 will incur a material loss. If, contrary to its expectations, the plaintiff prevails in this matter and proves damages at or near \$50 million, and is successful in having those damages trebled, the outcome could have a material adverse effect on our results of operations in the period in which a liability is recognized and on our cash flows for the period in which any damages are paid.

Several people, including two former Level 3 employees were indicted in the United States District Court for the Eastern District of Virginia on October 3, 2017, and charged with, among other things, accepting kickbacks from a subcontractor, who was also indicted, for work to be performed under a prime government contract. Of the two former employees, one entered into a plea agreement, and the other is deceased. Level 3 is fully cooperating in the government's investigations in this matter.

Other Proceedings, Disputes and Contingencies

From time to time, we are involved in other proceedings incidental to our business, including patent infringement allegations, administrative hearings of state public utility commissions relating primarily to our rates or services, actions relating to employee claims, various tax issues, environmental law issues, grievance hearings before labor regulatory agencies and miscellaneous third party tort actions.

We are currently defending several patent infringement lawsuits asserted against us by non-practicing entities, many of which are seeking substantial recoveries. These cases have progressed to various stages and one or more may go to trial in the coming 24 months if they are not otherwise resolved. Where applicable, we are seeking full or partial indemnification from our vendors and suppliers. As with all litigation, we are vigorously defending these actions and, as a matter of course, are prepared to litigate these matters to judgment, as well as to evaluate and consider all reasonable settlement opportunities.

We are subject to various foreign, federal, state and local environmental protection and health and safety laws. From time to time, we are subject to judicial and administrative proceedings brought by various governmental authorities under these laws. Several such proceedings are currently pending, but none is reasonably expected to exceed \$100,000 in fines and penalties.

The outcome of these other proceedings described under this heading is not predictable. However, based on current circumstances, we do not believe that the ultimate resolution of these other proceedings, after considering available defenses and any insurance coverage or indemnification rights, will have a material adverse effect on us.

The ultimate outcome of the above-described matters may differ materially from the outcomes anticipated, estimated, projected or implied by us in certain of our statements appearing above in this Note, and proceedings currently viewed as immaterial by us may ultimately materially impact us.

Environmental Contingencies

In connection with our largely historical operations, we have responded to or been notified of potential environmental liability at approximately 200 properties. We are engaged in addressing or have liquidated environmental liabilities at many of those properties. We could potentially be held liable, jointly, or severally, and without regard to fault, for the costs of investigation and remediation of these sites. The discovery of additional environmental liabilities or changes in existing environmental requirements could have a material adverse effect on our business.

Capital Leases

We lease certain facilities and equipment under various capital lease arrangements. Depreciation of assets under capital leases is included in depreciation and amortization expense in our consolidated statements of operations. Payments on capital leases are included in repayments of long-term debt, including current maturities in our consolidated statements of cash flows.

The tables below summarize our capital lease activity:

| | Years Ended December 31, | | | | |
|--|---------------------------|----|----|--|--|
| | 2018 2017 20 | | | | |
| | (Dollars in millions) | | | | |
| Assets acquired through capital leases | \$ 25 | 35 | 45 | | |
| Depreciation expense | 51 | 50 | 70 | | |
| Cash payments towards capital leases | 48 | 48 | 58 | | |
| | | | | | |

| | As of Decen | nber 31, |
|--|---------------|-----------|
| | 2018 | 2017 |
| | (Dollars in n | nillions) |
| Assets included in property, plant and equipment | \$ 427 | 420 |
| Accumulated depreciation | 180 | 154 |

The future annual minimum payments under capital lease arrangements as of December 31, 2018 were as follows:

| | Future Minimum Payments | |
|--|----------------------------|----------------------|
| | | Dollars in millions) |
| Capital lease obligations: | | |
| 2019 | \$ | 51 |
| 2020 | | 36 |
| 2021 | | 23 |
| 2022 | | 21 |
| 2023 | | 20 |
| 2024 and thereafter | | 183 |
| Total minimum payments | | 334 |
| Less: amount representing interest and executory costs | | (100) |
| Present value of minimum payments | | 234 |
| Less: current portion | | (38) |
| Long-term portion | \$ | 196 |

Operating Lease Income

CenturyLink leases various IRUs, office facilities, switching facilities and other network sites to third parties under operating leases. Lease and sublease income is included in operating revenue in the consolidated statements of operations.

For the years ended December 31, 2018, 2017 and 2016, our gross rental income was \$882 million , \$766 million and \$734 million , respectively.

Right-of-Way and Operating Lease Expense

CenturyLink leases various equipment, office facilities, retail outlets, switching facilities and other network sites. These leases, with few exceptions, provide for renewal options and escalations that are either fixed or based on the consumer price index. Any rent abatements, along with rent escalations, are included in the computation of rent expense calculated on a straight-line basis over the lease term. The lease term for most leases includes the initial non-cancelable term plus any term under renewal options that are reasonably assured. For the years ended December 31, 2018, 2017 and 2016, our gross rental expense was \$875 million, \$550 million and \$482 million, respectively. We also received sublease rental income for the years ended December 31, 2018, 2017 and 2016 of \$21 million, \$13 million and \$12 million, respectively.

At December 31, 2018, our future rental commitments for Right-of-Way agreements and operating leases were as follows:

| | Right-of-Way Agreements | Operating Leases | Total |
|-----------------------------------|----------------------------|----------------------|-------|
| | 1) | Dollars in millions) | _ |
| 2019 | 157 | 675 | 832 |
| 2020 | 134 | 443 | 577 |
| 2021 | 112 | 355 | 467 |
| 2022 | 120 | 279 | 399 |
| 2023 | 115 | 241 | 356 |
| 2024 and thereafter | 755 | 969 | 1,724 |
| Total future minimum payments (1) | 1,393 | 2,962 | 4,355 |

⁽¹⁾ Minimum payments have not been reduced by minimum sublease rentals of \$101 million due in the future under non-cancelable subleases.

Purchase Commitments

We have several commitments primarily for marketing activities and support services from a variety of vendors to be used in the ordinary course of business totaling \$921 million at December 31, 2018. Of this amount, we expect to purchase \$322 million in 2019, \$325 million in 2020 through 2021, \$88 million in 2022 through 2023 and \$186 million in 2024 and thereafter. These amounts do not represent our entire anticipated purchases in the future, but represent only those items for which we were contractually committed as of December 31, 2018.

(18) Other Financial Information

Other Current Assets

The following table presents details of other current assets in our consolidated balance sheets:

| | As of December 31, | |
|---|--------------------|----------|
| | 2018 | 2017 |
| | (Dollars in m | illions) |
| Prepaid expenses | \$ 307 | 294 |
| Income tax receivable | 82 | 258 |
| Materials, supplies and inventory | 120 | 128 |
| Contract assets | 134 | _ |
| Contract acquisition costs and deferred activation and installation charges | 167 | 128 |
| Other | 108 | 133 |
| Total other current assets | \$ 918 | 941 |

Selected Current Liabilities

Current liabilities reflected in our consolidated balance sheets include accounts payable and other current liabilities as follows:

| | As of December 31, | | | |
|---------------------------------|--------------------|-------------|-----------|--|
| | | 2018 | 2017 | |
| | | (Dollars in | millions) | |
| Accounts payable | \$ | 1,933 | 1,555 | |
| Other current liabilities: | | | | |
| Accrued rent | \$ | 45 | 34 | |
| Legal contingencies | | 30 | 45 | |
| Other | | 282 | 265 | |
| Total other current liabilities | \$ | 357 | 344 | |

Included in accounts payable at December 31, 2018 and 2017, were (i) \$86 million and \$36 million, respectively, representing book overdrafts and (ii) \$434 million and \$225 million, respectively, associated with capital expenditures.

(19) Labor Union Contracts

As of December 31, 2018, approximately 26% of our employees were members of various bargaining units represented by the Communication Workers of America ("CWA") and the International Brotherhood of Electrical Workers ("IBEW"). We believe that relations with our employees continue to be generally good. Approximately 1% of our employees were subject to collective bargaining agreements that expired prior to December 31, 2018 and are currently being renegotiated. Approximately 2% of our employees are subject to collective bargaining agreements that are scheduled to expire in 2019.

(20) Accumulated Other Comprehensive Loss

Information Relating to 2018

The table below summarizes changes in accumulated other comprehensive loss recorded on our consolidated balance sheet by component for the year ended December 31, 2018 :

| | Pe | ension Plans | Post-Retirement Benefit Plans | Foreign Currency Translation Adjustment and Other | Total |
|---|----|--------------|----------------------------------|--|---------|
| | | | (Dollars in | millions) | |
| Balance at December 31, 2017 | \$ | (1,731) | (235) | (29) | (1,995) |
| Other comprehensive loss before reclassifications | | (195) | 194 | (201) | (202) |
| Amounts reclassified from accumulated other comprehensive loss | | 128 | 15 | _ | 143 |
| Net current-period other comprehensive (loss) income | | (67) | 209 | (201) | (59) |
| Cumulative effect of adoption of ASU 2018-02, Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income | | (375) | (32) | _ | (407) |
| Balance at December 31, 2018 | \$ | (2,173) | (58) | (230) | (2,461) |

The table below presents further information about our reclassifications out of accumulated other comprehensive loss by component for the year ended December 31, 2018 :

| Year Ended December 31, 2018 | | se (Increase) Net Loss | Affected Line Item in Consolidated Statement of Operations |
|---|----------|---------------------------|---|
| | (Dollars | s in millions) | |
| Amortization of pension & post-retirement plans (1) | | | |
| Net actuarial loss | \$ | 178 | Other income, net |
| Prior service cost | | 12 | Other income, net |
| Total before tax | | 190 | |
| Income tax benefit | | (47) | Income tax expense (benefit) |
| Net of tax | \$ | 143 | |
| | | | |

⁽¹⁾ See Note 10—Employee Benefits for additional information on our net periodic benefit (expense) income related to our pension and post-retirement plans.

Information Relating to 2017

The table below summarizes changes in accumulated other comprehensive loss recorded on our consolidated balance sheet by component for the year ended December 31, 2017 :

| | Pei | nsion Plans | Post-Retirement Benefit Plans | Foreign Currency Translation Adjustment and Other | Total |
|--|-----------------------|-------------|----------------------------------|--|---------|
| | (Dollars in millions) | | | | |
| Balance at December 31, 2016 | \$ | (1,895) | (162) | (60) | (2,117) |
| Other comprehensive income (loss) before reclassifications | | 39 | (86) | 31 | (16) |
| Amounts reclassified from accumulated other comprehensive | | | | | |
| loss | | 125 | 13 | _ | 138 |
| Net current-period other comprehensive income (loss) | | 164 | (73) | 31 | 122 |
| Balance at December 31, 2017 | \$ | (1,731) | (235) | (29) | (1,995) |

The table below presents further information about our reclassifications out of accumulated other comprehensive loss by component for the year ended December 31, 2017 :

| Year Ended December 31, 2017 | Decrease (Increase) in Net Income | | Affected Line Item in Consolidated Statement of Operations | | | |
|---|--------------------------------------|-----------------|---|--|--|--|
| | (Dolla | rs in millions) | | | | |
| Amortization of pension & post-retirement plans (1) | | | | | | |
| Net actuarial loss | \$ | 205 | Other income, net | | | |
| Prior service cost | | 12 | Other income, net | | | |
| Total before tax | | 217 | | | | |
| Income tax benefit | | (79) | Income tax expense (benefit) | | | |
| Net of tax | \$ | 138 | | | | |
| | | | | | | |

⁽¹⁾ See Note 10—Employee Benefits for additional information on our net periodic benefit (expense) income related to our pension and post-retirement plans.

(21) Dividends

Our Board of Directors declared the following dividends payable in 2018 and 2017:

| Date Declared | Record Date | Dividend Per Share | | Total Amount | Payment Date |
|-------------------|-------------|-----------------------|------|---------------|--------------|
| | | | | (in millions) | |
| November 14, 2018 | 11/26/2018 | \$ 0.54 | 0 \$ | 586 | 12/7/2018 |
| August 21, 2018 | 8/31/2018 | 0.54 | 0 | 584 | 9/14/2018 |
| May 23, 2018 | 6/4/2018 | 0.54 | 0 | 588 | 6/15/2018 |
| February 21, 2018 | 3/5/2018 | 0.54 | 0 | 586 | 3/16/2018 |
| November 14, 2017 | 11/27/2017 | 0.54 | 0 | 577 | 12/11/2017 |
| August 22, 2017 | 9/5/2017 | 0.54 | 0 | 296 | 9/15/2017 |
| May 24, 2017 | 6/5/2017 | 0.54 | 0 | 297 | 6/16/2017 |
| February 21, 2017 | 3/3/2017 | 0.54 | 0 | 295 | 3/17/2017 |

The declaration of dividends is solely at the discretion of our Board of Directors, which may change or terminate our dividend practice at any time for any reason without prior notice. On March 1, 2019, our Board of Directors declared a quarterly cash dividend of \$0.25 per share.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

None.

ITEM 9A. CONTROLS AND PROCEDURES

Disclosure Controls and Procedures

Disclosure controls and procedures (as defined in Rule 13a-15(e) promulgated under the Securities Exchange Act of 1934 (the "Exchange Act")) are designed to provide reasonable assurance that the information required to be (i) recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and (ii) accumulated and communicated to management, including our Chief Executive Officer and Chief Financial Officer, to allow timely decisions regarding required disclosure. Management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving their objectives, and management necessarily applies its judgment in evaluating the cost benefit relationship of possible controls and procedures.

Based on their most recent evaluation, our Chief Executive Officer and Chief Financial Officer concluded that, as of December 31, 2018, our Company's disclosure controls and procedures were not effective as a result of the material weaknesses in our internal control over financial reporting described below.

Notwithstanding the material weaknesses described below, our management, including our chief executive officer and chief financial officer, believes that the audited consolidated financial statements contained in this Annual Report on Form 10-K fairly present, in all material respects, our financial condition, results of operations and cash flows for the fiscal years presented in conformity with U.S. generally accepted accounting principles. In addition, the material weaknesses described below did not result in the restatements of any of our audited or unaudited consolidated financial statements or disclosures for any previously reported periods.

Changes in Internal Control over Financial Reporting

During the fourth quarter 2018, we identified a deficiency that existed earlier in the year in the design of process controls over manual journal entries. Specifically, we had not designed and implemented controls to ensure all manual journal entries were reviewed. Although this control deficiency did not result in a misstatement in our consolidated financial statements, it created a reasonable possibility that a material misstatement would not have

been prevented or detected on a timely basis. Therefore we concluded the deficiency represented a material weakness in our internal control over financial reporting.

Upon identification of the deficiency, management designed and implemented a new journal entry module into our enterprise resource planning (ERP) system with certain configuration and automated workflow controls that we believe, together with newly-designed and implemented monitoring controls, sufficiently remediated the material weakness prior to December 31, 2018.

Except for the remediation of the manual journal entry material weakness discussed above, and the two additional material weaknesses disclosed below, there were no changes in our internal control over financial reporting during the quarter ended December 31, 2018 that materially affected, or that we believe are reasonably likely to materially affect, our internal control over financial reporting.

Internal Control Over Financial Reporting

Management's Report on Internal Control over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting (as defined in Rule 13a-15(f) of the Exchange Act). Our internal control over financial reporting is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles in the United States. Our system of internal control over financial reporting includes policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of our assets; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with U.S. generally accepted accounting principles and that receipts and expenditures are being made only in accordance with authorizations of our management; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of our assets that could have a material effect on our consolidated financial statements.

Our management, under the oversight of our principal executive officer and principal financial officer, and Audit Committee, assessed the effectiveness of our internal control over financial reporting as of December 31, 2018 using the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission in Internal control - An Integrated Framework ("2013 Framework"). In connection with this assessment, we identified material weaknesses in internal control over financial reporting as of December 31, 2018. A material weakness is defined as a deficiency, or a combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis. As a result of this assessment, management concluded that two material weaknesses existed as described below.

- Ineffective design and operation of process level internal controls over the fair value measurement of certain assets acquired and liabilities assumed from Level 3.
 - These deficiencies arose because (i) we did not conduct an effective risk assessment to identify and assess changes we needed to make to our financial reporting and process level controls, related to fair value measurement of assets acquired and liabilities assumed in the transaction with Level 3, (ii) we did not clearly assign responsibility for the design, implementation, and operation of controls over the fair value measurements and (iii) we did not maintain effective information and communication processes to ensure the right information was available to personnel on a timely basis so they could fulfill their control responsibilities related to the fair value measurements.
- · Ineffective design and operation of certain process level internal controls over the existence and accuracy of revenue transactions.

These deficiencies arose because we did not conduct an effective risk assessment to identify risks of material misstatement related to the existence and accuracy of revenue transactions.

The deficiencies led to material misstatements in the valuation of assets acquired and liabilities assumed in the acquisition of Level 3 that were corrected prior to the issuance of the consolidated financial statements as of and for the year ended December 31, 2018 set out in this annual report on Form 10-K. However, the misstatements did not result in the need to restate any of our audited or unaudited consolidated financial statements or disclosures filed for any previously reported period, but did result in minor adjustments to the consolidated balance sheet as of December 31, 2018 included in our press release in the Form 8-K furnished on February 13, 2019. The deficiencies also led to other immaterial misstatements that were not corrected in the valuation of assets acquired and liabilities assumed in the acquisition and in revenue. These control deficiencies created a reasonable possibility that material misstatements to the consolidated financial statements would not be prevented or detected on a timely basis. Therefore, we concluded they represented material weaknesses and our internal control over financial reporting was not effective as of December 31, 2018.

Our independent registered public accounting firm, KPMG LLP, who audited the consolidated financial statements included in this annual report, has issued an adverse opinion on the effectiveness of our internal control over financial reporting as of December 31, 2018. KPMG LLP's report appears on page 82 of this annual report on Form 10-K.

Remediation Plans

We are taking several steps to remediate the material weaknesses identified above. These steps include the following:

- We plan to implement continuous risk assessment processes that are designed to identify and assess changes that could significantly
 impact our internal control over financial reporting environment and risks of material misstatement related to the revenue recognition
 process.
- We plan to ensure the effective design, implementation and operation of sufficient process level controls over the valuation of assets acquired and liabilities assumed in business combinations by developing written procedures and controls on accounting for business combinations. Additionally, we plan to assign responsibility for financial reporting and internal controls and will endeavor to ensure there are effective communication processes in place so individuals assigned with the overall responsibility to operate these process level controls understand their roles and expectations as it relates to designing, implementing and operating process level controls for future business combinations.
- We plan to design and implement sufficient additional process level control activities over the existence and accuracy of revenue transactions.

We plan to execute our plans to remediate the material weaknesses identified above as soon as feasible. We will test the ongoing effectiveness of the new controls and will consider the material weakness remediated after the new controls operate effectively for a sufficient period of time. There is no assurance, however, that these measures will remediate the material weakness or ensure that our internal controls over financial reporting will be effective in the future.

ITEM 9B. OTHER INFORMATION

None.

PART III

ITEM 10. DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE

The information required by Item 10 is incorporated by reference to the Proxy Statement.

ITEM 11. EXECUTIVE COMPENSATION

The information required by Item 11 is incorporated by reference to the Proxy Statement.

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT AND RELATED STOCKHOLDER MATTERS

Equity Compensation Plan Information

The following table provides information as of December 31, 2018 about our equity compensation plans under which Common Shares are authorized for issuance:

| | Number of securities to be issued upon exercise of outstanding options and rights (a) | Weighted-average exercise price of outstanding options and rights (b) | | Number of securities remaining available for future issuance under plans (excluding securities reflected in column (a)) |
|--|--|---|-----|---|
| Equity compensation plans approved by shareholders | 4,995,518 (1) | \$ | (2) | 33,814,844 |
| Equity compensation plans not approved by shareholders (3) | 6,161,469 | 27.46 | | _ |
| Totals | 11,156,987 (1) | \$ 27.46 | (2) | 33,814,844 |
| | | | _ | |

⁽¹⁾ These amounts include restricted stock units, some of which represent the difference between the number of shares of restricted stock subject to market conditions granted at target and the maximum possible payout for these awards. Depending on performance, the actual share payout of these awards may range between 0-200% of target.

The balance of the information required by Item 12 is incorporated by reference to the Proxy Statement.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS AND DIRECTOR INDEPENDENCE

The information required by Item 13 is incorporated by reference to the Proxy Statement.

ITEM 14. PRINCIPAL ACCOUNTANT FEES AND SERVICES

The information required by Item 14 is incorporated by reference to the Proxy Statement.

⁽²⁾ The amounts in column (a) include restricted stock units, which do not have an exercise price. Consequently, those awards were excluded from the calculation of this exercise price.

⁽³⁾ These amounts represent common shares to be issued upon exercise or vesting of equity awards that were assumed in connection with certain acquisitions.

PART IV

ITEM 15. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

Exhibits identified in parentheses below are on file with the SEC and are incorporated herein by reference. All other exhibits are provided as part of this electronic submission.

| part of this dicetionic submission. | | | |
|-------------------------------------|---|--|--|
| Exhibit | _ | | |
| Number | Desc | ription | |
| 3.1 | <u>Amen</u> | ded and Restated Articles of Incorporation of CenturyLink, Inc., as amended through May 23, 2012 (incorporated by | |
| | refere | nce to Exhibit 3.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and | |
| | Excha | ange Commission on May 30, 2012). | |
| 3.2 | | es of Amendment of Articles of Incorporation setting forth the rights and preferences of the Series CC Junior | |
| | <u>Partic</u> | ipating Preferred Shares, dated as of February 13, 2019 (incorporated by reference to Exhibit 3.1 of CenturyLink, Inc.'s | |
| | Curre | nt Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on February 14, | |
| | <u>2019)</u> | | |
| 3.3 | <u>Bylaw</u> | s of CenturyLink, Inc., as amended and restated through August 24, 2017 (incorporated by reference to Exhibit 3.1 of | |
| | | ryLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on | |
| | Augus | <u>st 24, 2017).</u> | |
| 4.1 | <u>Form</u> | of common stock certificate (incorporated by reference to Exhibit 4.10 of CenturyLink, Inc.'s Registration Statement on | |
| | <u>Form</u> | S-3 filed with the Securities and Exchange Commission on March 2, 2012 (Registration No. 333-179888)). | |
| 4.2 | Section | on 382 Rights Agreement by and between CenturyLink, Inc. and Computershare Trust Company, N.A., dated as of | |
| | <u>Febru</u> | ary 13, 2019 (incorporated by reference to Exhibit 4.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001- | |
| | <u>0778</u> 4 | 1) filed with the Securities and Exchange Commission on February 14, 2019). | |
| 4.3 | Instrument relating to Credit Agreement assumed by CenturyLink, Inc. on November 1, 2017. | | |
| | a. | Credit Agreement, dated as of June 19, 2017, by and among CenturyLink Escrow, LLC and the lenders and agents | |
| | | named therein (incorporated by reference to Exhibit 10.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File | |
| | | No. 001-07784) filed with the Securities and Exchange Commission on June 20, 2017). | |
| | b. | Incremental Assumption Agreement, dated as of January 29, 2018, by and among CenturyLink, Inc. and the lenders | |
| | | and agents named therein (incorporated by reference to Exhibit 4.2(b) of CenturyLink, Inc.'s annual report on Form | |
| | | 10-K for the year ended December 31, 2017 (File No. 001-07784) filed with the Securities and Exchange | |
| | | Commission on March 1, 2018). | |
| 4.4 | Instru | ments relating to CenturyLink, Inc.'s public senior debt. (1) | |
| | а | Indenture, dated as of March 31, 1994, by and between Century Telephone Enterprises, Inc. (currently named | |

- a. Indenture, dated as of March 31, 1994, by and between Century Telephone Enterprises, Inc. (currently named CenturyLink, Inc.) and Regions Bank (successor-in-interest to First American Bank & Trust of Louisiana), as Trustee.
 - (i). Form of 7.2% Senior Notes, Series D, due 2025 (incorporated by reference to Exhibit 4.27 of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 1995 (File No. 001-07784) filed with the Securities and Exchange Commission on March 18, 1996).
 - (ii). Form of 6.875% Debentures, Series G, due 2028, (incorporated by reference to Exhibit 4.9 of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 1997 (File No. 001-07784) filed with the Securities and Exchange Commission on March 16, 1998).
- b. Fifth Supplemental Indenture, dated as of September 21, 2009, by and between CenturyTel, Inc. (currently named CenturyLink, Inc.) and Regions Bank, as Trustee, designating and outlining the terms and conditions of CenturyLink's 7.60% Senior Notes, Series P, due 2039 and 6.15% Senior Notes, Series Q, due 2019 (incorporated by reference to Exhibit 4.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on September 22, 2009).
 - (i). Form of 7.60% Senior Notes, Series P, due 2039 and 6.15% Senior Notes, Series Q, due 2019 (incorporated by reference to Exhibit A to Exhibit 4.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on September 22, 2009).

- c. Sixth Supplemental Indenture, dated as of June 16, 2011, by and between CenturyLink, Inc. and Regions Bank, as Trustee, designating and outlining the terms and conditions of CenturyLink's 6.45% Senior Notes, Series S, due 2021 (incorporated by reference to Exhibit 4.2 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on June 16, 2011).
 - (i). Form of 6.45% Senior Notes, Series S, due 2021 (incorporated by reference to Exhibit A to Exhibit 4.2 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on June 16, 2011).
- d. Seventh Supplemental Indenture, dated as of March 12, 2012, by and between CenturyLink, Inc. and Regions Bank, as Trustee, designating and outlining the terms and conditions of CenturyLink's 5.80% Senior Notes, Series T, due 2022 and 7.65% Senior Notes, Series U, due 2042 (incorporated by reference to Exhibit 4.1 of CenturyLink's Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on March 12, 2012).
 - (i). Form of 5.80% Senior Notes, Series T, due 2022 and 7.65% Senior Notes, Series U, due 2042 (incorporated by reference to Exhibit A to Exhibit 4.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on March 12, 2012).
- e. <u>Eighth Supplemental Indenture, dated as of March 21, 2013, by and between CenturyLink, Inc. and Regions Bank, as Trustee, designating and outlining the terms and conditions of CenturyLink's 5.625% Senior Notes, Series V, due 2020 (incorporated by reference to Exhibit 4.1 of CenturyLink's Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on March 21, 2013).</u>
 - (i). Form of 5.625% Senior Notes, Series V, due 2020 (incorporated by reference to Exhibit A to Exhibit 4.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on March 21, 2013).
- f. Ninth Supplemental Indenture, dated as of November 27, 2013, by and between CenturyLink, Inc. and Regions Bank, as Trustee, designating and outlining the terms and conditions of CenturyLink's 6.75% Senior Notes, Series W, due 2023 (incorporated by reference to Exhibit 4.1 of CenturyLink's Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on November 27, 2013).
 - (i). Form of 6.75% Senior Notes, Series W, due 2023 (incorporated by reference to Exhibit A to Exhibit 4.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on November 27, 2013).
- g. Tenth Supplemental Indenture, dated as of March 19, 2015, by and between CenturyLink, Inc. and Regions Bank, as Trustee, designating and outlining the terms and conditions of CenturyLink's 5.625% Senior Notes, Series X, due 2025 (incorporated by reference to Exhibit 4.2 of CenturyLink's Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on March 19, 2015).
 - (i). Form of 5.625% Senior Notes, Series X, due 2025 (incorporated by reference to Exhibit A to Exhibit 4.2 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on March 19, 2015).
- h. <u>Eleventh Supplemental Indenture, dated as of April 6, 2016, by and between CenturyLink, Inc. and Regions Bank, as Trustee, designating and outlining the terms and conditions of CenturyLink's 7.5% Senior Notes, Series Y, due 2024 (incorporated by reference to Exhibit 4.2 of CenturyLink's Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on April 6, 2016).</u>
 - (i). Form of 7.5% Senior Notes, Series Y, due 2024 (incorporated by reference to Exhibit A to Exhibit 4.2 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on April 6, 2016).
- 4.5 Instruments relating to indebtedness of Qwest Communications International, Inc. and its subsidiaries. (1)

- a. Indenture, dated as of April 15, 1990, by and between The Mountain States Telephone and Telegraph Company (currently named Qwest Corporation) and The First National Bank of Chicago (incorporated by reference to Exhibit 4.2 of Qwest Corporation's annual report on Form 10-K for the year ended December 31, 2002 (File No. 001-03040) filed with the Securities and Exchange Commission on January 13, 2004).
 - (i). First Supplemental Indenture, dated as of April 16, 1991, by and between U S WEST Communications, Inc. (currently named Qwest Corporation) and The First National Bank of Chicago (incorporated by reference to Exhibit 4.3 of Qwest Corporation's annual report on Form 10-K for the year ended December 31, 2002 (File No. 001-03040) filed with the Securities and Exchange Commission on January 13, 2004).
- b. Indenture, dated as of April 15, 1990, by and between Northwestern Bell Telephone Company (predecessor to Qwest Corporation) and The First National Bank of Chicago (incorporated by reference to Exhibit 4.5(b) of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended March 31, 2012 (File No. 001-07784) filed with the Securities and Exchange Commission on May 10, 2012).
 - (i). First Supplemental Indenture, dated as of April 16, 1991, by and between U S WEST Communications, Inc. (currently named Qwest Corporation) and The First National Bank of Chicago (incorporated by reference to Exhibit 4.3 of Qwest Corporation's annual report on Form 10-K for the year ended December 31, 2002 (File No. 001-03040) filed with the Securities and Exchange Commission on January 13, 2004).
- c. Indenture, dated as of June 29, 1998, by and among U S WEST Capital Funding, Inc. (currently named Qwest Capital Funding, Inc.), U S WEST, Inc. (predecessor to Qwest Communications International Inc.) and The First National Bank of Chicago, as trustee (incorporated by reference to Exhibit 4(a) of U S WEST, Inc.'s Current Report on Form 8-K (File No. 001-14087) filed with the Securities and Exchange Commission on November 18, 1998).
 - (i). First Supplemental Indenture, dated as of June 30, 2000, by and among U S WEST Capital Funding, Inc. (currently named Qwest Capital Funding, Inc.), U S WEST, Inc. (predecessor to Qwest Communications International Inc.) and Bank One Trust Company, N.A., as trustee (incorporated by reference to Exhibit 4.10 of Qwest Communications International Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2000 (File No. 001-15577) filed with the Securities and Exchange Commission on August 11, 2000).
- d. Indenture, dated as of October 15, 1999, by and between US West Communications, Inc. (currently named Qwest Corporation) and Bank One Trust Company, N.A., as trustee (incorporated by reference to Exhibit 4(b) of Qwest Corporation's annual report on Form 10-K for the year ended December 31, 1999 (File No. 001-03040) filed with the Securities and Exchange Commission on March 3, 2000).
 - (i). <u>Eighth Supplemental Indenture, dated as of September 21, 2011, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.9 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with the Securities and Exchange Commission on September 20, 2011).</u>
 - (ii). Ninth Supplemental Indenture, dated as of October 4, 2011, by and between Qwest Corporation and U.S.

 Bank National Association (incorporated by reference to Exhibit 4.1 of Qwest Corporation's Current Report on
 Form 8-K (File No. 001-03040) filed with the Securities and Exchange Commission on October 4, 2011).
 - (iii). Tenth Supplemental Indenture, dated as of April 2, 2012, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.11 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with the Securities and Exchange Commission on March 30, 2012).
 - (iv). Eleventh Supplemental Indenture, dated as of June 25, 2012, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.12 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with the Securities and Exchange Commission on June 22, 2012).
 - (v). Twelfth Supplemental Indenture, dated as of May 23, 2013, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.13 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with the Securities and Exchange Commission on May 22, 2013).

- (vi). Thirteenth Supplemental Indenture, dated as of September 29, 2014, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.14 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with the Securities and Exchange Commission on September 26, 2014).
- (vii). Fourteenth Supplemental Indenture, dated as of September 21, 2015, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.15 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with the Securities and Exchange Commission on September 21, 2015).
- (viii). Fifteenth Supplemental Indenture, dated as of January 29, 2016, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.16 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with the Securities and Exchange Commission on January 29, 2016).
- (ix). Sixteenth Supplemental Indenture, dated as of August 22, 2016, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.17 of Qwest Corporation's Form 8-A (File No. 001-03040) filed with Securities and Exchange Commission on August 22, 2016).
- (x). Seventeenth Supplemental Indenture dated as of April 27, 2017, by and between Qwest Corporation and U.S. Bank National Association (incorporated by reference to Exhibit 4.18 of Qwest Corporation's Form 8-A (File No. 03040) filed with the Securities and Exchange Commission on April 27, 2017.
- e. Credit Agreement, dated as of February 20, 2015, by and among Qwest Corporation, the several lenders from time to time parties thereto, and CoBank, ACB, as administrative agent (incorporated by reference to Exhibit 4.5 of Qwest Corporation's Annual Report on Form 10-K for the year ended December 31, 2014 (File No. 001-03040) filed with the Securities and Exchange Commission on February 27, 2015).
- 4.6 Instruments relating to indebtedness of Embarq Corporation. (1)
 - a. Indenture, dated as of May 17, 2006, by and between Embarq Corporation and J.P. Morgan Trust Company,
 National Association, a national banking association, as trustee (incorporated by reference to Exhibit 4.1 of Embarq
 Corporation's Current Report on Form 8-K (File No. 001-32732) filed with the Securities and Exchange Commission
 on May 18, 2006).
 - b. 7.995% Global Note due 2036 of Embarq Corporation (incorporated by reference to Exhibit 4.4 to Embarq Corporation's annual report on Form 10-K for the year ended December 31, 2006 (File No. 001-32372) filed with the Securities and Exchange Commission on March 9, 2007).
- 4.7 Instruments relating to indebtedness of Level 3 Communications, Inc. and its subsidiaries. (1)
 - a. Indenture, dated as of November 14, 2013, by and between Level 3 Communications, Inc., as Guarantor, Level 3 Financing, Inc., as Issuer, and The Bank of New York Mellon Trust Company, N.A., as Trustee, relating to the 6.125% Senior Notes due 2021 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 18, 2013).
 - (i). Supplemental Indenture, dated as of March 14, 2014, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to Level 3 Communications, LLC's unconditioned, unsecured guarantee of the 6.125% Senior Notes due 2021 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on March 17, 2014).
 - (ii). Supplemental Indenture, dated as of March 14, 2014, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the subordination in any bankruptcy, liquidation or winding up proceeding of the guarantee by Level 3 Communications, LLC of the 6.125% Senior Notes due 2021 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.2 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on March 17, 2014).

- (iii). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of November 14, 2013, governing the 6.125% Senior notes due 2021 (incorporated by reference to Exhibit 4.7 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).
- (iv). Supplemental Indenture, dated as of November 1, 2017, among Level 3 Communications, Inc., as guarantor, Level 3 Financing, Inc., as issuer, Level 3 Parent, LLC (f/k/a WWG Merger Sub LLC) and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of November 14, 2013, governing the 6.125% Senior Notes due 2021 (incorporated by referenced to Exhibit 4.7 to the Registrant's Current Report on Form 8-K (file no. 001-35134) dated November 1, 2017).
- b. Indenture, dated as of August 12, 2014, by and between Level 3 Escrow II, Inc., as Issuer, and The Bank of New York Mellon Trust Company, N.A., as Trustee, relating to the 5.375% Senior Notes due 2022 of Level 3 Escrow II, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on August 14, 2014).
 - (i). Securities Assumption Supplemental Indenture, dated as of October 31, 2014, by and between Level 3

 Escrow II, Inc., Level 3 Financing, Inc., Level 3 Communications, Inc. and The Bank of New York Mellon

 Trust Company, N.A., as trustee, relating to Level 3 Financing, Inc.'s unconditional assumption of all of Level
 3 Escrow II, Inc.'s obligations under the 5.375% Senior Notes due 2022 (incorporated by reference to Exhibit
 4.2 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the

 Securities and Exchange Commission on November 5, 2014).
 - (ii). Supplemental Indenture, dated as of October 31, 2013, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the subordination in any bankruptcy, liquidation or winding up proceeding of the guarantee by Level 3 Communications, LLC of the 5.375% Senior Notes due 2022 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.3 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 5, 2014).
 - (iii). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of August 12, 2014, governing the 5.375% Senior Notes due 2022 (incorporated by reference to Exhibit 4.6 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).
 - (iv). Supplemental Indenture, dated as of November 22, 2016, among Level 3 Communications, LLC as guarantor, Level 3 Communications, Inc., as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of August 12, 2014, governing the 5.375% Senior Notes due 2022 (incorporated by reference to Exhibit 4.6 to the Registrant's Current Report on Form 8-K (file no. 001-35134) dated November 28, 2016).
- c. Indenture, dated as of December 1, 2014, by and between Level 3 Communications, Inc. and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the 5.75% Senior Notes due 2022 of Level 3 Communications, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on December 2, 2014).
 - (i). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., and The Bank of New York Mellon Trust Company, N.A., amending the Indenture dated as of December 1, 2014, governing the 5.75% Senior Notes due 2022 (incorporated by reference to Exhibit 4.8 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).

- d. Indenture, dated as of January 29, 2015, by and between Level 3 Communications, Inc., as Guarantor, Level 3 Financing, Inc., as Issuer, and The Bank of New York Mellon Trust Company, N.A., as Trustee, relating to the 5.625% Senior Notes due 2023 of Level Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on January 30, 2015).
 - (i). Supplemental Indenture, dated as of June 3, 2015, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to Level 3 Communications, LLC's unconditioned, unsecured guarantee of the 5.625% Senior Notes due 2023 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on June 4, 2015).
 - (ii). Supplemental Indenture, dated as of June 3, 2015, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the subordination in any bankruptcy, liquidation or winding up proceeding of the guarantee by Level 3 Communications, LLC of the 5.625% Senior Notes due 2023 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.2 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on June 4, 2015).
 - (iii). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of January 29, 2015 governing the 5.625% Senior Notes due 2023 (incorporated by reference to Exhibit 4.5 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).
 - (iv). Supplemental Indenture, dated as of November 1, 2017, among Level 3 Communications, Inc. as guarantor, Level 3 Financing, Inc., as issuer, Level 3 Parent, LLC (f/k/a WWG Merger Sub LLC) and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of January 29, 2015 governing the 5.625% Senior Notes due 2023 (incorporated by reference to Exhibit 4.5 of the Registrant's Current Report on Form 8-K (file no. 001-35134) dated November 1, 2017).
- e. Indenture, dated as of April 28, 2015, by and between Level 3 Communications, Inc., as Guarantor, Level 3
 Financing, Inc., as Issuer, and The Bank of New York Mellon Trust Company, N.A., as Trustee, relating to the
 5.125% Senior Notes due 2023 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3
 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange
 Commission on April 30, 2015).
 - (i). Supplemental Indenture, dated as of September 1, 2015, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to Level 3 Communications, LLC's unconditioned, unsecured guarantee of the 5.125% Senior Notes due 2023 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on September 2, 2015).
 - (ii). Supplemental Indenture, dated as of September 1, 2015, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the subordination in any bankruptcy, liquidation or winding up proceeding of the guarantee by Level 3 Communications, LLC of the 5.125% Senior Notes due 2023 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.3 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on September 2, 2015).

- (iii). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of April 28, 2015 governing the 5.125% Senior Notes due 2023 (incorporated by reference to Exhibit 4.3 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).
- (iv). Supplemental Indenture, dated as of November 1, 2017, among Level 3 Communications, Inc. as guarantor, Level 3 Financing, Inc., as issuer, Level 3 Parent, LLC (f/k/a WWG Merger Sub LLC) and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of April 28, 2015 governing the 5.375% Senior Notes due 2025 (incorporated by reference to Exhibit 4.4 to the Registrant's Current Report on Form 8-K (file no. 001-35134) dated November 1, 2017).
- f. Indenture, dated as of April 28, 2015, by and between Level 3 Communications, Inc., as Guarantor, Level 3

 Financing, Inc., as Issuer, and The Bank of New York Mellon Trust Company, N.A., as Trustee, relating to the
 5.375% Senior Notes due 2025 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.2 of Level 3

 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange
 Commission on April 30, 2015).
 - (i). Supplemental Indenture, dated as of September 1, 2015, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to Level 3 Communications, LLC's unconditioned, unsecured guarantee of the 5.375% Senior Notes due 2025 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.2 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on September 2, 2015).
 - (ii). Supplemental Indenture, dated as of September 1, 2015, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the subordination in any bankruptcy, liquidation or winding up proceeding of the guarantee by Level 3 Communications, LLC of the 5.375% Senior Notes due 2025 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.4 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on September 2, 2015).
 - (iii). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of April 28, 2015 governing the 5.375% Senior Notes due 2025 (incorporated by reference to Exhibit 4.4 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).
 - (iv). Supplemental Indenture, dated as of November 1, 2017, among Level 3 Communications, Inc. as guarantor, Level 3 Financing, Inc., as issuer, Level 3 Parent, LLC (f/k/a WWG Merger Sub LLC) and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of April 28, 2015 governing the 5.125% Senior Notes due 2023 (incorporated by reference to Exhibit 4.3 to the Registrant's Current Report on Form 8-K (file no. 001-35134) dated November 1, 2017).
- g. Indenture, dated as of November 13, 2015, by and between Level 3 Communications, Inc., as Guarantor, Level 3

 Financing, Inc., as Issuer, and The Bank of New York Mellon Trust Company, N.A., as Trustee, relating to the
 5.375% Senior Notes due 2024 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3

 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange
 Commission on November 13, 2015).

- (i). Supplemental Indenture, dated as of February 8, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to Level 3 Communications, LLC's unconditioned, unsecured guarantee of the 5.375% Senior Notes due 2024 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on February 8, 2016).
- (ii). Supplemental Indenture, dated as of February 8, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the subordination in any bankruptcy, liquidation or winding up proceeding of the guarantee by Level 3 Communications, LLC of the 5.375% Senior Notes due 2024 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.2 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on February 8, 2016).
- (iii). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of November 13, 2015 governing the 5.375% Senior Notes due 2024 (incorporated by reference to Exhibit 4.2 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).
- (iv). Supplemental Indenture, dated as of November 1, 2017, among Level 3 Communications, Inc. as guarantor, Level 3 Financing, Inc., as issuer, Level 3 Parent, LLC (f/k/a WWG Merger Sub LLC) and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of November 13, 2015 governing the 5.375% Senior Notes due 2024 (incorporated by reference to Exhibit 4.2 to the Registrant's Current Report on Form 8-K (file no. 001-35134) dated November 1, 2017).
- h. Indenture, dated as of March 22, 2016, by and between Level 3 Communications, Inc., as Guarantor, Level 3

 Financing, Inc., as Issuer, and The Bank of New York Mellon Trust Company, N.A., as Trustee, relating to the
 5.25% Senior Notes due 2026 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 of Level 3

 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange
 Commission on March 22, 2016).
 - (i). Supplemental Indenture, dated as of September 16, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to Level 3 Communications, LLC's unconditioned, unsecured guarantee of the 5.25% Senior Notes due 2026 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.1 to Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on September 16, 2016).
 - (ii). Supplemental Indenture, dated as of September 16, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, relating to the subordination in any bankruptcy, liquidation or winding up proceeding of the guarantee by Level 3 Communications, LLC of the 5.25% Senior Notes due 2026 of Level 3 Financing, Inc. (incorporated by reference to Exhibit 4.2 to Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on September 16, 2016).
 - (iii). Supplemental Indenture, dated as of November 22, 2016, by and between Level 3 Communications, Inc., as guarantor, Level 3 Communications, LLC, as guarantor, Level 3 Financing, Inc., as issuer, and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of March 22, 2016 governing the 5.25% Senior Notes due 2026 (incorporated by reference to Exhibit 4.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on November 28, 2016).

- (iv). Supplemental Indenture, dated as of November 1, 2017, among Level 3 Communications, Inc. as guarantor, Level 3 Financing, Inc., as issuer, Level 3 Parent, LLC (f/k/a WWG Merger Sub LLC) and The Bank of New York Mellon Trust Company, N.A., as trustee, amending the Indenture dated as of March 22, 2016 governing the 5.25% Senior Notes due 2026 (incorporated by reference to Exhibit 4.1 to the Registrant's Current Report on Form 8-K (file no. 001-35134) dated November 1, 2017).
- i. Twelfth Amendment Agreement to the Amended and Restated Credit Agreement, dated as of February 22, 2017, by and between Level 3 Communications, Inc., Level 3 Financing, Inc., the Lenders party thereto and Merrill Lynch Capital Corporation (incorporated by reference to Exhibit 10.1 of Level 3 Communications, Inc.'s Current Report on Form 8-K (File No. 001-35134) filed with the Securities and Exchange Commission on February 27, 2017), including the Amended and Restated Loan Proceeds Notes issued on the same date by Level 3 Communications, LLC to Level 3 Financing, Inc. (incorporated by reference to Exhibit 10.2 and 10.3 of Level 3 Communications, Inc.'s Current Report on Form 8-K (file No. 001-35134) filed with the Securities and Exchange Commission on February 27, 2017).
- 4.8 Certain intercompany debt instruments.
 - a. Revolving Promissory Note, dated as of April 2, 2012 pursuant to which Embarq Corporation may borrow from an affiliate of CenturyLink, Inc. up to \$2.5 billion on a revolving basis (incorporated by reference to Exhibit 4.7(a) of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2012 (File No. 001-07784) filed with the Securities and Exchange Commission on August 9, 2012), as amended and restated by the Amended and Restated Revolving Promissory Note, dated as of September 30, 2017, by and between Embarq Corporation and an affiliate of CenturyLink, Inc.*
 - b. Revolving Promissory Note, dated as of April 18, 2012, pursuant to which Qwest Corporation may borrow from an affiliate of CenturyLink, Inc. up to \$1.0 billion on a revolving basis (incorporated by reference to Exhibit 4.7(b) of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2012 (File No. 001-07784) filed with the Securities and Exchange Commission on August 9, 2012), as amended and restated by the Amended and Restated Revolving Promissory Note, dated as of September 30, 2017, by and between Qwest Corporation and an affiliate of CenturyLink, Inc. *
 - c. Revolving Promissory Note, dated as of September 27, 2012, pursuant to which Qwest Communications International, Inc. may borrow from an affiliate of CenturyLink, Inc. up to \$3.0 billion on a revolving basis (incorporated by reference to Exhibit 4.8 of Qwest Communications International Inc.'s Quarterly Report on Form 10-Q for the period ended September 30, 2012 (File No. 001-15577) filed with the Securities and Exchange Commission on November 13, 2012), as amended and restated by the Amended and Restated Revolving Promissory Note, dated as of September 30, 2017, by and between Qwest Communications International Inc. and an affiliate of CenturyLink, Inc. *
- 10.1+ CenturyLink 2011 Equity Incentive Plan, as amended through May 18, 2016 (incorporated by reference to Appendix A of CenturyLink, Inc.'s Proxy Statement for its 2016 Annual Meeting of Shareholders (File No. 001-07784) filed with the Securities and Exchange Commission on April 5, 2016).
 - (i). Form of Restricted Stock Agreement for non-management directors used since 2011 (incorporated by reference to Exhibit 10.1(a) (ii) of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2011 (File No. 001-07784) filed with the Securities and Exchange Commission on August 9, 2011).
 - (ii). Form of Restricted Stock Agreement for executive officers used for annual recurring grants since May 2013 (incorporated by reference to Exhibit 10.2(i) (iii) of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2013 (File No. 001-07784) filed with the Securities and Exchange Commission on August 8, 2013).
 - (iii). Form of Restricted Stock Agreement for special retention award grants made to certain executive officers on June 1, 2017 (incorporated by reference to Exhibit 10.1(iii) of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2017 (File No. 001-07784) filed with the Securities and Exchange Commission on August 8, 2017).

- (iv). Form of Restricted Stock Agreement for special integration award grants made to certain executive officers on June 1, 2017 (incorporated by reference to Exhibit 10.1(iv) of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2017 (File No. 001-07784) filed with the Securities and Exchange Commission on August 8, 2017).
- (v).* Form of Restricted Stock Agreement for annual time-based equity grants to certain executive officers in 2018 (not including Jeffrey K. Storey).
- (vi)* Form of Restricted Stock Agreement for annual performance-based equity grants to certain executive officers in 2018 (not including Jeffrey K. Storey).
- 10.2+ CenturyLink 2018 Equity Incentive Plan (incorporated by reference to Appendix A of CenturyLink, Inc.'s Proxy Statement for its 2018 Annual Meeting of Shareholders (File No. 001-07784) filed with the SEC on April 12, 2018).
 - (i). Form of Restricted Stock Agreement for annual equity grants to non-management directors in 2018 (incorporated by reference to Exhibit 10.1A to CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2018 (File No. 001-07784) filed with the SEC on August 9, 2018).
 - (ii). Form of RSU Agreement for annual time-based equity grant to Jeffrey K. Storey on May 23, 2018 (incorporated by reference to Exhibit 10.1B to CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2018 (File No. 001-07784) filed with the SEC on August 9, 2018).
 - (iii). Form of RSU Agreement for annual performance-based equity grant to Jeffrey K. Storey on May 23, 2018 (incorporated by reference to Exhibit 10.1C to CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2018 (File No. 001-07784) filed with the SEC on August 9, 2018).
 - (iv). Restricted Stock Agreement for time-based portion of 2018 promotion equity grant to Jeffrey K. Storey on May 23, 2018 (incorporated by reference to Exhibit 10.1D to CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2018 (File No. 001-07784) filed with the SEC on August 9, 2018). Restricted Stock Agreement for performance-based portion of 2018 promotion equity grant to Jeffrey K. Storey on May 23, 2018 (incorporated by reference to Exhibit 10.1E to CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2018 (File No. 001-07784) filed with the SEC on August 9,
 - (v). 2018).

 Restricted Stock Agreement for performance-based retention grant to Sunit S. Patel on June 1, 2018
 (incorporated by reference to Exhibit 10.1F to CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the
 (vi). period ended June 30, 2018 (File No. 001-07784) filed with the SEC on August 9, 2018).
- 10.3+* Supplemental Dollars & Sense Plan, 2014 Restatement, effective January 1, 2014.
- 10.4+ Supplemental Defined Benefit Pension Plan, effective as of January 1, 2012 (incorporated by reference to Exhibit 10.5 of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 2011 (File No. 001-07784) filed with the Securities and Exchange Commission on February 28, 2012).
- 10.5+ 2015 Executive Officer Short-Term Incentive Program (incorporated by reference to Exhibit A of CenturyLink's 2015 Proxy Statement on Form 14A (File No. 001-07784) filed with the Securities and Exchange Commission on April 8, 2015).

Description

- 10.6+ Key Employee Incentive Compensation Plan, dated as of January 1, 1984, as amended and restated as of November 16, 1995 (incorporated by reference to Exhibit 10.1(f) of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 1995 (File No. 001-07784) filed with the Securities and Exchange Commission on March 18, 1996) and amendment thereto dated as of November 21, 1996 (incorporated by reference to Exhibit 10,1(f) of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 1996 (File No. 001-07784) filed with the Securities and Exchange Commission on March 17, 1997), amendment thereto dated as of February 25, 1997 (incorporated by reference to Exhibit 10.2 of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended March 31, 1997 (File No. 001-07784) filed with the Securities and Exchange Commission on May 8, 1997), amendment thereto dated as of April 25, 2001 (incorporated by reference to Exhibit 10.2 of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended March 31, 2001 (File No. 001-07784) filed with the Securities and Exchange Commission on May 15, 2001), amendment thereto dated as of April 17, 2000 (incorporated by reference to Exhibit 10.3(a) of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 2001 (File No. 001-07784) filed with the Securities and Exchange Commission on March 15, 2002) and amendment thereto dated as of February 27, 2007 (incorporated by reference to Exhibit 10.1 of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended June 30, 2007 (File No. 001-07784) filed with the Securities and Exchange Commission on August 8, 2007).
- 10.7+ Form of Indemnification Agreement entered into between CenturyLink, Inc. and each of its directors as of February 24, 2016 (incorporated by reference to Exhibit 10.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on February 29, 2016).
- 10.8+ Form of Indemnification Agreement entered into between CenturyLink, Inc. and each of its officers as of February 24, 2016 (incorporated by reference to Exhibit 10.2 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on February 29, 2016).
- 10.9+* Change of Control Agreement by and between Jeffrey K. Storey and CenturyLink, Inc.
- 10.10+ Change of Control Agreement, effective January 1, 2011, by and between Glen F. Post, III and CenturyLink, Inc. (incorporated by reference to Exhibit 10.11 of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 2010 (File No. 001-07784) filed with the Securities and Exchange Commission on March 1, 2011).
- 10.11+ Form of Change of Control Agreement, effective January 1, 2011 between CenturyLink, Inc. and each of its other executive officers (incorporated by reference to Exhibit 10.12 of CenturyLink, Inc.'s annual report on Form 10-K for the year ended December 31, 2010 (File No. 001-07784) filed with the Securities and Exchange Commission on March 1, 2011).
- 10.12+* Amended and Restated CenturyLink Executive Severance Plan, effective October 10, 2017 (including, as Exhibit E thereto, the Legacy Level 3 Key Executive Severance Plan, as assumed by CenturyLink, Inc. effective November 1, 2017 (covering certain Legacy Level 3 employees through October 31,2019)).
- 10.13+ Amended and Restated CenturyLink, Inc. Bonus Life Insurance Plan for Executive Officers, dated as of April 3, 2008 (incorporated by reference to Exhibit 10.4 of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended March 31, 2008 (File No. 001-07784) filed with the Securities and Exchange Commission on May 7, 2008) and First Amendment thereto (incorporated by reference to Exhibit 10.13 of CenturyLink, Inc.'s Quarterly Report on Form 10-Q for the period ended September 30, 2010 (File No. 001-07784) filed with the Securities and Exchange Commission on November 5, 2010).
- 10.14+ Form of integration award grant letter, dated June 1, 2017, entered into between CenturyLink, Inc. and certain officers (incorporated by reference to Exhibit 10.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on June 1, 2017).
- 10.15+ Offer letter between CenturyLink, Inc. and Jeffrey K. Storey, effective May 23, 2018 (incorporated by reference to Exhibit 10.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784), filed with the SEC on May 25, 2018, which amended, restated and superseded the offer letter between CenturyLink, Inc. and Jeffrey K. Storey, effective April 27, 2017 (incorporated by reference to Exhibit 10.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on November 1, 2017).
- 10.16+ Offer letter between CenturyLink, Inc. and Sunit S. Patel, effective June 1, 2017 (incorporated by reference to Exhibit 10.2 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on November 1, 2017).

| Exhibit Number | Description |
|-------------------|--|
| | • |
| 10.17+ | Offer letter between CenturyLink, Inc. and Indraneel Dev, effective November 6, 2018 (incorporated by reference to Exhibit 10.1 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the SEC on November 7, 2018) |
| 40.40 | |
| 10.18+ | <u>Legacy Qwest Deferred Compensation Plan for Nonemployee Directors, as amended and restated, Amendment to Deferred</u> Compensation Plan for Nonemployee Directors (incorporated by reference to Exhibit 10.2 of Qwest Communications |
| | International Inc.'s Current Report on Form 8-K (File No. 001-15577) filed with the Securities and Exchange Commission on |
| | December 16, 2005 and Exhibit 10.8 to Qwest Communication International Inc.'s Quarterly Report on Form 10-Q for the |
| | period ended September 30, 2008 (File No. 001-15577) filed with the Securities and Exchange Commission on October 29, |
| | 2008) and Amendment No. 2011-1 to Deferred Compensation Plan for Nonemployee Directors (incorporated by reference to |
| | Exhibit 10.15(c) of CenturyLink, Inc.'s annual report for the year ended December 31, 2011 (File No. 001-07784) filed with |
| | the Securities and Exchange Commission on February 28, 2012). |
| 10.19+ | Level 3 Communications, Inc. Stock Incentive Plan, as amended and restated through October 31, 2017 (incorporated by |
| | reference to Exhibit 10.3 of CenturyLink, Inc.'s Current Report on Form 8-K (File No. 001-07784) filed with the Securities and Exchange Commission on November 1, 2017) |
| 10.20 | Shareholder Rights Agreement, dated as of October 31, 2016, by and between CenturyLink, Inc. and STT Crossing Ltd. |
| 10.20 | (incorporated by reference to Exhibit 10.2 of CenturyLink's Current Report on Form 8-K (File No. 001-07784) filed with the |
| | Securities and Exchange Commission on November 3, 2016); as amended by the Assignment and Assumption Agreement, |
| | dated as of February 5, 2018, by and among STT Crossing Ltd., Everitt Investments Pte.Ltd., Aranda Investments Pte.Ltd., |
| | and CenturyLink, Inc. (incorporated by reference to Exhibit 99.3 to Amendment No. 1 to a statement of beneficial ownership |
| | of common shares of CenturyLink, Inc. on Schedule 13D filed with the SEC by Singapore Technologies Telemedia Pte. Ltd. |
| | on February 7, 2018) |
| 21* | Subsidiaries of CenturyLink, Inc. |
| 23* | Independent Registered Public Accounting Firm Consent |
| 31.1* | Certification of the Chief Executive Officer of CenturyLink, Inc. pursuant to Section 302 of the Sarbanes-Oxley Act of 2002. |
| 31.2* | Certification of the Chief Financial Officer of CenturyLink, Inc. pursuant to Section 302 of the Sarbanes-Oxley Act of 2002. |
| 32* | Certification of the Chief Executive Officer and Chief Financial Officer of CenturyLink, Inc. pursuant to Section 906 of the |
| | Sarbanes-Oxley Act of 2002. |
| 101* | Financial statements from the annual report on Form 10-K of CenturyLink, Inc. for the period ended December 31, 2018, |
| | formatted in XBRL: (i) the Consolidated Statements of Operations, (ii) the Consolidated Statements of Comprehensive Income (Loss), (iii) the Consolidated Balance Sheets, (iv) the Consolidated Statements of Cash Flows, (v) the Consolidated |
| | Statements of Stockholders' Equity and (vi) the Notes to Consolidated Financial Statements. |
| | The state of the s |

Exhibit filed herewith.

+ Indicates a management contract or compensatory plan or arrangement.

Certain of the items in Sections 4.4, 4.5, 4.6 and 4.7 (i) omit supplemental indentures or other instruments governing debt that has been retired, or (ii) refer to trustees who may have been replaced, acquired or affected by similar changes. In accordance with Item 601(b) (4) (iii) (A) of Regulation S-K, copies of certain instruments defining the rights of holders of certain of our long-term debt are not filed herewith. Pursuant to this regulation, we hereby agree to furnish a copy of any such instrument to the SEC upon request.

ITEM 16. SUMMARY OF BUSINESS AND FINANCIAL INFORMATION

Not applicable.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this annual report to be signed on its behalf by the undersigned thereunto duly authorized.

| | | - |
|---------|-------|------|
| Century | Link. | Inc. |

Senior Vice President - Controller (Principal Accounting Officer)

Pursuant to the requirements of the Securities Exchange Act of 1934, this annual report has been signed below by the following persons on behalf of the registrant and in the capacities and on the date indicated.

| /s/ Jeff K. Storey | Chief Executive Officer and Director | March 11, 2019 |
|-----------------------|--|----------------|
| Jeff K. Storey | | |
| /s/ Harvey Perry | Chairman of the Board | March 11, 2019 |
| Harvey Perry | | |
| /s/ W. Bruce Hanks | Vice Chairman of the Board | March 11, 2019 |
| W. Bruce Hanks | | |
| /s/ Indraneel Dev | Executive Vice President and Chief Financial Officer | March 11, 2019 |
| Indraneel Dev | | |
| | Senior Vice President - Controller (Principal Accounting | |
| /s/ Eric J. Mortensen | Officer) | March 11, 2019 |
| Eric J. Mortensen | | |
| /s/ Martha H. Bejar | Director | March 11, 2019 |
| Martha H. Bejar | | |
| /s/ Virginia Boulet | Director | March 11, 2019 |
| Virginia Boulet | | |
| /s/ Peter C. Brown | Director | March 11, 2019 |
| Peter C. Brown | | |
| /s/ Kevin P. Chilton | Director | March 11, 2019 |
| Kevin P. Chilton | | |
| /s/ Steven T. Clontz | Director | March 11, 2019 |
| Steven T. Clontz | | |
| /s/ T. Michael Glenn | Director | March 11, 2019 |
| T. Michael Glenn | | |
| /s/ Mary L. Landrieu | Director | March 11, 2019 |
| Mary L. Landrieu | | |
| | | |

| /s/ Glen F. Post, III | Director | March 11, 2019 |
|------------------------|----------|----------------|
| Glen F. Post, III | | |
| /s/ Michael J. Roberts | Director | March 11, 2019 |
| Michael J. Roberts | | |
| /s/ Laurie A. Siegel | Director | March 11, 2019 |
| Laurie A. Siegel | | |

FORM OF RESTRICTED STOCK AGREEMENT UNDER THE

[CENTURYLINK 2011 EQUITY INCENTIVE PLAN *OR* LEGACY LEVEL 3 COMMUNICATIONS, INC. STOCK INCENTIVE PLAN]

(Time-Based Restricted Stock Grants to Certain Officers)

| This RESTRICTED STOCK AGREEMENT (this "Agree | eement") is entered into as of [|] by and |
|---|---|--|
| between CenturyLink, Inc. ("CenturyLink") and [|] ("Award Recipient"). | |
| WHEREAS, CenturyLink maintains the [CenturyLink 2] Inc. Stock Incentive Plan] (the "Plan"), under which the Human subcommittee thereof, (the "Committee") of the Board of Director indirectly grant restricted shares of CenturyLink's common semployees, directors and other service providers of CenturyLink terms, conditions, or restrictions as it may deem appropriate; and | Resources and Compensation Committee, of tors of CenturyLink (the "Board") may, amount tock, \$1.00 par value per share (the "Common or its subsidiaries (collectively, the "Comp | or a duly-authorized ong other things, directly on Stock"), to key |
| WHEREAS, pursuant to the Plan, the Committee has at the terms and conditions specified below. | warded to the Award Recipient restricted sha | ares of Common Stock on |
| NOW, THEREFORE, the parties agree as follows: | | |
| 1. AWARE | O OF SHARES | |
| Upon the terms and conditions of the Plan and this Agre Date") hereby awards to the Award Recipient [] restricted to Sections 2, 3, and 4 hereof, in installments as follows: | | |
| Scheduled Vesting Date | Number of Shares of Restricted Stoo | <u>ek</u> |
| [] | [] | |
| [] | [] | |
| [] | | |
| | 2. | |
| | STRICTIONS ON | |
| RESTRIC | CTED STOCK | |
| 2.1 In addition to the conditions and restrictions provi vote the Restricted Stock, to receive accrued dividends thereon be sold, assigned, | | • |
| {N3767826.1} | | |
| | | |

donated, transferred, exchanged, pledged, hypothecated, or otherwise encumbered prior to vesting, whether voluntarily or involuntarily. All dividends and other distributions relating to the Restricted Stock will accrue when declared and be paid to the Award Recipient only upon the vesting of the related Restricted Stock. Except as otherwise provided in this Section 2.1, the Award Recipient shall be entitled to all rights of a shareholder of CenturyLink with respect to the Restricted Stock, including the right to vote the shares.

2.2 If the shares of Restricted Stock have not already vested or been forfeited under the terms of this Agreement or the Plan, all of the shares of Restricted Stock shall vest and all restrictions set forth in Section 2.1 shall lapse on the date on which the employment of the Award Recipient terminates as a result of (i) death or (ii) disability within the meaning of Section 22(e)(3) of the Internal Revenue Code.

2.3 Termination of Employment Following a Change of Control of CenturyLink:

- (a) If the shares of Restricted Stock have not already vested or been forfeited under the terms of this Agreement or the Plan, and within [___] months following a Change of Control of CenturyLink (as defined in the Plan), the Award Recipient's employment is terminated by the Company or its Affiliates without Cause (as defined below) or by the Award Recipient for Good Reason (as defined below), then all restrictions set forth in Section 2.1 with respect to such vested shares shall lapse, on the 60 th day following the Award Recipient's termination of employment, provided the Release Condition (as defined in Section 2.5) has been satisfied. If the Release Condition is not satisfied, then all unvested Restricted Stock shall automatically terminate and be forfeited as of the 60 th day following termination of employment.
- (b) (i) For purposes of this Section 2.3, "Cause" shall mean the Award Recipient's (A) willful breach of any nondisclosure, noncompetition, nonsolicitation or nondisparagement covenants contained in any agreement between the Company and the Award Recipient; (B) conviction of, or plea of guilty or *nolo contendere* to, a felony or other crime involving dishonesty or moral turpitude; (C) workplace conduct resulting in the payment of civil monetary penalties or the incurrence of civil non-monetary penalties that will materially restrict or prevent the Award Recipient from discharging his obligations to the Company; (D) habitual intoxication during working hours or habitual abuse of or addiction to a controlled substance; (E) material breach of the Company's insider trading, corporate ethics and compliance policies and programs or any other Board-adopted policies applicable to management conduct; (F) participation in the public reporting of any information contained in any report filed by the Company with the Securities and Exchange Commission that was impacted by the Award Recipient's knowing or intentional fraudulent or illegal conduct; or (G) substantial, willful and repeated failure to perform duties as instructed by or on behalf of the Board in writing.
 - (ii) The Award Recipient's employment shall not be deemed terminated for Cause unless the Company shall have delivered to the Award Recipient a termination notice with a copy of a resolution adopted by the affirmative vote of not less than three-quarters of the entire Board at a meeting

called partly or wholly for such purpose (after reasonable notice is provided to the Award Recipient and the Award Recipient has had an opportunity, with counsel, to be heard by the Board) finding that the Award Recipient should be terminated for Cause and specifying in reasonable detail the grounds therefor.

| (iii) | No action or inaction shall be deemed the basis for Cause unless the Award Recipient is terminated | L |
|----------------|--|----|
| therefor prior | to the first anniversary of the date on which such action or omission is first known to [|]. |

- (a) For purpose of this Section 2.3, "Good Reason" shall mean a termination of the Award Recipient's employment under the following circumstances: (1) the Award Recipient has delivered a written notice to the Company, objecting to a "Good Reason Event" (as defined below) within 90 days following the initial existence or occurrence of such event, (2) the Company fails to cure such event or condition within 30 days following receipt of the Award Recipient's written notice (the "30-day Cure Period"), and (3) as a result, the Award Recipient terminates his or her employment no later than [__] months following the expiration of the 30-day Cure Period. A "Good Reason Event" shall mean:
 - (i) Any failure of the Company or its Affiliates to provide the Award Recipient with a position, authority, duties and responsibilities at least commensurate in all material respects with the most significant of those held, exercised and assigned at any time during the 180-day period immediately preceding the Change of Control. The Award Recipient's position, authority, duties and responsibilities after a Change of Control shall not be considered commensurate in all material respects with the Award Recipient's position, authority, duties and responsibilities prior to a Change of Control unless after the Change of Control the Award Recipient holds an equivalent position with, and exercises substantially equivalent authority, duties and responsibilities on behalf of the Post-Transaction Company;
 - (ii) The assignment to the Award Recipient of any duties inconsistent in any material respect with the Award Recipient's position (including status, offices, titles and reporting requirements), authority, duties or responsibilities at the time of the Change of Control, or any other action that results in a diminution in any material respect in such position, authority, duties or responsibilities;
 - (iii) A reduction of the Award Recipient's base salary in effect as of the date of the Change of Control without the Award Recipient's consent, except for across-the-board salary reductions similarly affecting all or substantially all similarly-situated officers of the Company and the Post-Transaction Company;
 - (iv) The Award Recipient is advised of, manifests an awareness of, or becomes aware of facts that would cause a reasonable person to inquire into any failure in any material respect by the Company or its Affiliates to comply with any of the provisions of this Agreement; or

- (v) Any directive requiring the Award Recipient to be based at any office or location more than 50 miles from the location the Award Recipient was based prior to the Change of Control, or requiring the Award Recipient to travel on business to a substantially greater extent than required immediately prior to the Change of Control.
- (b) For purposes of this Section 2.3, "Affiliate" (or variants thereof) shall mean a person that controls, or is controlled by or is under common control with, another specified person, either directly or indirectly.
- 2.4 If the shares of Restricted Stock have not already vested or been forfeited under the terms of this Agreement or the Plan, and the Award Recipient's employment terminates because of retirement on or after attaining the age of 55 with at least ten years of prior service with the Company, then, provided the Committee has specifically approved such action, the shares of Restricted Stock shall not terminate but shall remain outstanding and shall vest, in full or in part (as determined by the Committee), and all restrictions set forth in Section 2.1 with respect to such vested shares shall lapse, on the 60 th day following the Award Recipient's termination of employment, provided the Release Condition (as defined in Section 2.5) has been satisfied. If the Release Condition is not satisfied, then all unvested Restricted Stock shall automatically terminate and be forfeited as of the 60 th day following termination of employment.
- 2.5 For purposes of Section 2.3 and 2.4, "Release Condition" shall mean the Award Recipient's execution, delivery to the Company and non-revocation of a mutual liability release agreement in the form and substance determined by the Company (and the expiration of any revocation period contained in such release agreement) within 60 days following the Award Recipient's termination of employment.

3. TERMINATION OF EMPLOYMENT

Notwithstanding anything in this Agreement to the contrary, all unvested Restricted Stock shall automatically terminate and be forfeited if the employment of the Award Recipient terminates for any reason, unless and to the extent otherwise provided in Section 2.

4. FORFEITURE OF AWARD

4.1 If, at any time during the Award Recipient's employment by the Company or within 18 months after termination of employment, the Award Recipient engages in any activity in competition with any activity of the Company, or inimical, contrary or harmful to the interests of the Company, including but not limited to: (3) conduct relating to the Award Recipient's employment for which either criminal or civil penalties against the Award Recipient may be sought; (3) conduct or activity that results in termination of the Award Recipient's employment for cause; (3) violation of the Company's policies, including, without limitation, the Company's insider trading, ethics and corporate compliance policies and programs; (3) participating in the public reporting of any financial or operating result that was impacted by the participant's

knowing or intentional fraudulent or illegal conduct; (3) accepting employment with, acquiring a 5% or more equity or participation interest in, serving as a consultant, advisor, director or agent of, directly or indirectly soliciting or recruiting any employee of the Company who was employed at any time during the Award Recipient's tenure with the Company, or otherwise assisting in any other capacity or manner any company or enterprise that is directly or indirectly in competition with or acting against the interests of the Company or any of its lines of business (a "competitor"), except for (i) any isolated, sporadic accommodation or assistance provided to a competitor, at its request, by the Award Recipient during the Award Recipient's tenure with the Company, but only if provided in the good faith and reasonable belief that such action would benefit the Company by promoting good business relations with the competitor and would not harm the Company's interests in any substantial manner or (ii) any other service or assistance that is provided at the request or with the written permission of the Company; (3) disclosing or misusing any confidential information or material concerning the Company, except for any disclosures provided in good faith to regulators in response to inquiries or investigations or otherwise made in good faith to any regulator or law enforcement authority; (3) engaging in, promoting, assisting or otherwise participating in a hostile takeover attempt of the Company or any other transaction or proxy contest that could reasonably be expected to result in a Change of Control not approved by the Board; or (3) making any statement or disclosing any information to any customers, suppliers, lessors, lessees, licensors, licensees, employees, or others with whom the Company engages in business that is defamatory or derogatory with respect to the business, operations, technology, management, or other employees of the Company, or taking any other action that could reasonably be expected to injure the Company in its business relationships with any of the foregoing parties or result in any other detrimental effect on the Company, except for any statements or disclosures provided in good faith to regulators in response to inquiries or investigations or otherwise made in good faith to any regulator or law enforcement authority, then the award of Restricted Stock granted hereunder shall automatically terminate and be forfeited effective on the date on which the Award Recipient engages in such activity and (1) all shares of Common Stock acquired by the Award Recipient pursuant to this Agreement (or other securities into which such shares have been converted or exchanged) shall be returned to the Company or, if no longer held by the Award Recipient, the Award Recipient shall pay to the Company, without interest, all cash, securities or other assets received by the Award Recipient upon the sale or transfer of such stock or securities, and (2) all unvested shares of Restricted Stock shall be forfeited.

4.2 If the Award Recipient owes any amount to the Company under Section 4.1 above, the Award Recipient acknowledges that the Company may, to the fullest extent permitted by applicable law, deduct such amount from any amounts the Company owes the Award Recipient from time to time for any reason (including without limitation amounts owed to the Award Recipient as salary, wages, reimbursements or other compensation, fringe benefits, retirement benefits or vacation pay). Whether or not the Company elects to make any such set-off in whole or in part, if the Company does not recover by means of set-off the full amount the Award Recipient owes it, the Award Recipient hereby agrees to pay immediately the unpaid balance to the Company.

4.3 The Award Recipient may be released from the Award Recipient's obligations under Sections 4.1 and 4.2 above only if the Committee or its delegee determines in its sole discretion that such action is in the best interests of the Company.

5. STOCK CERTIFICATES

No stock certificates evidencing the Restricted Stock shall be issued by CenturyLink until the lapse of restrictions under the terms hereof. Instead, ownership of the Restricted Stock shall be evidenced by a book entry with the applicable restrictions reflected. Upon the lapse of restrictions on shares of Restricted Stock, CenturyLink shall issue the vested shares of Restricted Stock (either through book entry issuances or delivery of a stock certificate) in the name of the Award Recipient or his or her nominee, subject to the other terms and conditions hereof, including those governing any withholdings of shares under Section 6 below. Upon receipt of any such vested shares, the Award Recipient is free to hold or dispose of such shares, subject to (3) applicable securities laws, (3) CenturyLink's policy statement on insider trading, and (3) any of CenturyLink's stock ownership guidelines then in effect that are applicable to the Award Recipient.

6. WITHHOLDING TAXES

Notwithstanding any Plan provision to the contrary, at the time that all or any portion of the Restricted Stock vests, CenturyLink will withhold from the shares the Award Recipient otherwise would receive hereunder the number of whole shares of Common Stock, rounding up if necessary, having a value equal to the minimum statutory amount required to be withheld under federal, state and local law (or, if permitted by the Committee and elected by the Award Recipient, such other rate as will not cause adverse accounting consequences and is permitted under applicable IRS withholding rules).

7. ADDITIONAL CONDITIONS

Anything in this Agreement to the contrary notwithstanding, if, at any time prior to the vesting of the Restricted Stock in accordance with Section 1 or 2 hereof, CenturyLink further determines, in its sole discretion, that the listing, registration or qualification (or any updating of any such document) of the shares of Common Stock issuable pursuant hereto is necessary on any securities exchange or under any federal or state securities or blue sky law, or that the consent or approval of any governmental regulatory body is necessary or desirable as a condition of, or in connection with the issuance of shares of Common Stock pursuant thereto, or the removal of any restrictions imposed on such shares, such shares of Common Stock shall not be issued, in whole or in part, or the restrictions thereon removed, unless such listing, registration, qualification, consent or approval shall have been effected or obtained free of any conditions not acceptable to CenturyLink. CenturyLink agrees to use commercially reasonable efforts to issue all shares of Common Stock issuable hereunder on the terms provided herein.

8. NO CONTRACT OF EMPLOYMENT INTENDED

Nothing in this Agreement shall confer upon the Award Recipient any right to continue in the employment of the Company, or to interfere in any way with the right of the Company to terminate the Award Recipient's employment relationship with the Company at any time.

9. BINDING EFFECT

Upon being duly executed and delivered by CenturyLink and the Award Recipient, this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective heirs, executors, administrators, legal representatives and successors. Without limiting the generality of the foregoing, whenever the term "Award Recipient" is used in any provision of this Agreement under circumstances where the provision appropriately applies to the heirs, executors, administrators or legal representatives to whom this award may be transferred by will or by the laws of descent and distribution, the term "Award Recipient" shall be deemed to include such person or persons.

10. EFFECT OF PLAN TERMS AND COMMITTEE ACTIONS

- 10.1 Capitalized terms used but not defined in this Agreement shall have the respective meanings ascribed to them in the Plan.
- 10.2 This Agreement, the rights of the Award Recipient hereunder and the shares of Restricted Stock granted hereby are subject to (i) all of the terms, conditions, restrictions and other provisions of the Plan, as it may be amended from time to time, as fully as if all such provisions were set forth in their entirety in this Agreement and (ii) such rules and regulations as the Committee may adopt for administration of the Plan. It is expressly understood that the Committee is authorized to administer, construe, and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, all of which shall be binding upon the Award Recipient. If any provision of this Agreement conflicts with a provision of the Plan, the Plan provision shall control.
- 10.3 The Plan is discretionary and may be amended, cancelled or terminated by the Company at any time, in its discretion. The grant of Restricted Stock in this Agreement does not create any contractual rights other than as set forth in this Agreement, and does not create a right to receive Restricted Stock or any other Incentives in the future. Future Incentives, if any, will be at the sole discretion of the Company.
- 10.4 The Award Recipient acknowledges receipt from CenturyLink of a copy of the Plan and a prospectus summarizing the Plan and further acknowledges that the Award Recipient was advised to review such materials prior to entering into this Agreement. The Award Recipient waives the right to claim that the provisions of the Plan are not binding upon the Award

Recipient and the Award Recipient's heirs, executors, administrators, legal representatives and successors.

11. ATTORNEYS' FEES AND EXPENSES

Should any party hereto retain counsel for the purpose of enforcing, or preventing the breach of, any provision hereof, including, but not limited to, the institution of any action or proceeding in court to enforce any provision hereof, to enjoin a breach of any provision of this Agreement, to obtain specific performance of any provision of this Agreement, to obtain monetary or liquidated damages for failure to perform any provision of this Agreement, or for a declaration of such parties' rights or obligations hereunder, or for any other judicial remedy, then the prevailing party shall be entitled to be reimbursed by the losing party for all costs and expenses incurred thereby, including, but not limited to, attorneys' fees (including costs of appeal).

12. GOVERNING LAW

This Agreement shall be governed by and construed in accordance with the laws of the State of Louisiana. The Award Recipient and CenturyLink shall submit to the exclusive jurisdiction of, and venue in, the courts in Louisiana in any dispute relating to this Agreement.

13. SEVERABILITY

If any term or provision of this Agreement, or the application thereof to any person or circumstance, shall at any time or to any extent be invalid, illegal or unenforceable in any respect as written, the Award Recipient and CenturyLink intend for any court construing this Agreement to modify or limit such provision so as to render it valid and enforceable to the fullest extent allowed by law. Any such provision that is not susceptible of such reformation shall be ignored so as to not affect any other term or provision hereof, and the remainder of this Agreement, or the application of such term or provision to persons or circumstances other than those as to which it is held invalid, illegal or unenforceable, shall not be affected thereby and each term and provision of this Agreement shall be valid and enforced to the fullest extent permitted by law.

14. OTHER PROVISIONS

- 14.1 It is intended that the payments and benefits provided under this Agreement will comply with the requirements of Section 409A of the Code and the regulations promulgated thereunder ("Section 409A") or an exemption therefrom. The Agreement shall be interpreted, construed, administered, and governed in a manner that effects such intent.
- 14.2 The Plan and this Agreement contain the entire agreement between the parties with respect to the subject matter contained herein. This Agreement may not, without the Award

Recipient's consent, be amended or modified so as to materially adversely affect the Award Recipient's rights under this Agreement, except (i) as provided in the Plan, as it may be amended from time to time in the manner provided therein, or (ii) by a written document signed by each of the parties hereto. Any oral or written agreements, representations, warranties, written inducements, or other communications with respect to the subject matter contained herein made prior to the execution of the Agreement shall be void and ineffective for all purposes.

14.3 Nothing expressed or implied in this Agreement is intended or shall be construed to confer upon or give any person, other than the parties hereto and their successors, assigns, heirs, executors, administrators, or legal representatives, any rights or remedies under, or by reason of, this Agreement.

15. ELECTRONIC DELIVERY AND EXECUTION OF DOCUMENTS

- 15.1 The Company may, in its sole discretion, deliver any documents related to the Award Recipient's current or future participation in the Plan or any other equity compensation plan of the Company by electronic means or request Award Recipient's consent to the terms of an award by electronic means. The plan documents may, but do not necessarily, include: the Plan, any grant notice, this Agreement, the Plan prospectus, and any reports of CenturyLink provided generally to CenturyLink's shareholders. In addition, the Award Recipient may deliver by electronic means any grant notice or award agreement to the Company or to such third party involved in administering the applicable plan as the Company may designate from time to time. Such means of electronic delivery may include the delivery of a link to a Company intranet or the Internet site of a third party involved in administering the applicable plan, the delivery of the document via e-mail or such other means of electronic delivery specified by the Company. By accepting the terms of this Agreement, the Award Recipient also hereby consents to participate in such plans and to execute agreements setting the terms of participation through an on-line or electronic system as described herein.
- 15.2 The Award Recipient acknowledges that the Award Recipient has read Section 15.1 of this Agreement and consents to the electronic delivery and electronic execution of plan documents as described in Section 15.1. The Award Recipient acknowledges that he or she may receive from the Company a paper copy of any documents delivered electronically at no cost to the Award Recipient by contacting the Company by telephone or in writing. The Award Recipient further acknowledges that the Award Recipient will be provided with a paper copy of any documents if the attempted electronic delivery of such documents to the Award Recipient fails. Similarly, the Award Recipient understands that the Award Recipient must provide the Company or any designated third party administrator with a paper copy of any documents if the attempted electronic delivery of such documents by the Award Recipient fails. The Award Recipient may revoke his or her consent to the electronic delivery and execution of documents described in Section 15.1 or may change the electronic mail address to which such documents are to be delivered (if Award Recipient has provided an electronic mail address) at any time by notifying the Company of such revoked consent or revised e-mail address by telephone, postal

service or electronic mail. Finally, the Award Recipient understands that he or she is not required to consent to electronic delivery or execution of documents described in Section 15.1.

16. Data Privacy

As a condition to his or her participation in the Plan, the Award Recipient consents to the collection, use, and transfer of personal data as described in this paragraph. The Award Recipient understands that the Company holds certain personal information about the Award Recipient, including his or her name, home address and telephone number, date of birth, social security number or identification number, salary, nationality, job title, any shares of stock or directorships held in the Company, details of all options or any other entitlement to shares of stock awarded, canceled, exercised, vested, unvested, or outstanding in the Award Recipient's favor, for the purpose of managing and administering the Plan ("Data"). The Award Recipient further understands that CenturyLink or its subsidiaries will transfer Data amongst themselves as necessary for the purpose of implementation, administration, and management of the Award Recipient's participation in the Plan, and that CenturyLink and any of its subsidiaries may each further transfer Data to any third parties assisting the Company in the implementation, administration, and management of the Plan. The Award Recipient understands that these recipients may be located in the United States or elsewhere, and that the recipients' country may have different data privacy laws and protections that the Award Recipient's country. The Award Recipient authorizes them to receive, possess, use, retain, and transfer the Data, in electronic or other form, for the purposes of implementing, administering, and managing the Award Recipient's participation in the Plan, including any requisite transfer to a broker or other third party with whom the Award Recipient may elect to deposit any amounts received pursuant to the Plan and this Agreement, such Data as may be required for the administration of the Plan. The Award Recipient understands that he or she may, at any time, view Data, require any necessary amendments to it or withdraw the consents herein in writing by contacting his or her human resources representative. The Award Recipient further understands that this consent is purely voluntary, and will not affect the Award Recipient's employment or career with the Company, although it may affect the Award Recipient's ability to participate in the Plan.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered on the day and year first above written.

| | CENTURYLINK, INC. | |
|----------------|-------------------|-------------------|
| | By: | [NAME] [TITLE] |
| | | [NAME] |
| N3767826.1} 10 | | |

FORM OF RESTRICTED STOCK AGREEMENT UNDER THE [CENTURYLINK 2011 EQUITY INCENTIVE PLAN *OR* LEGACY LEVEL 3 COMMUNICATIONS, INC. STOCK INCENTIVE PLAN]

(Performance-Based Restricted Stock Grants to Certain Officers)

| This RESTRICTED STOCK AGREEMENT (this "Agreement") is entered into as of [] by and between CenturyLink, Inc. ("CenturyLink") and [] ("Award Recipient"). |
|---|
| WHEREAS, CenturyLink maintains the [CenturyLink 2011 Equity Incentive Plan OR Legacy Level 3 Communications Inc. Stock Incentive Plan] (the "Plan"), under which the Human Resources and Compensation Committee, or a duly-authorized subcommittee thereof, (the "Committee") of the Board of Directors of CenturyLink (the "Board") may, among other things, directly or indirectly grant restricted shares of CenturyLink's common stock, \$1.00 par value per share (the "Common Stock"), to key employees, directors and other service providers of CenturyLink or its subsidiaries (collectively, the "Company"), subject to such terms, conditions, or restrictions as it may deem appropriate; and |
| WHEREAS, the chief executive officer of CenturyLink, acting pursuant to powers delegated to him by the Committee under the Plan, has awarded to the Award Recipient restricted shares of Common Stock on the terms and conditions specified below. |
| NOW, THEREFORE, the parties agree as follows: |
| 1. AWARD OF SHARES |
| 1.1 Upon the terms and conditions of the Plan and this Agreement, CenturyLink as of the date of this Agreement (the "Grant Date") hereby awards to the Award Recipient a total of [] restricted shares of Common Stock (the "Restricted Stock") that vest, subject to Sections 2, 3 and 4 hereof, in installments as described in this Section 1. |
| 1.2 [SPECIFIC VESTING DATES AND PERFORMANCE METRICS OMITTED.] |
| 1.3 The difference between the number of shares of Restricted Stock granted to the Award Recipient under Section 1.1 on the one hand and the maximum number of shares the Award Recipient may earn under Section 1.2 on the other is referred to in this Agreement as the "Additional Shares." Any contingent right of the Award Recipient to receive Additional Shares under Section 1.2 shall be treated as restricted stock units with accompanying dividend equivalent rights under the terms of Section 7 of the Plan. Any Additional Shares earned under Section 1.2 shall be issued on the applicable Vest Date. |
| 2. AWARD RESTRICTIONS ON RESTRICTED STOCK |

- 2.1 In addition to the conditions and restrictions provided in the Plan, neither the shares of Restricted Stock nor the right to vote the Restricted Stock, to receive accrued dividends thereon or to enjoy any other rights or interests thereunder or hereunder may be sold, assigned, donated, transferred, exchanged, pledged, hypothecated, or otherwise encumbered prior to vesting, whether voluntarily or involuntarily. All dividends and other distributions relating to the Restricted Stock and all dividend equivalents on the Additional Shares will accrue when declared and be paid to the Award Recipient only upon the vesting of the related Restricted Stock or Additional Shares. Except as otherwise provided in this Section 2.1, the Award Recipient shall be entitled to all rights of a shareholder of CenturyLink with respect to the Restricted Stock, including the right to vote the shares.
- 2.2 If the shares of Restricted Stock have not already vested or been forfeited under the terms of this Agreement or the Plan, all of the shares of Restricted Stock shall vest and all restrictions set forth in Section 2.1 shall lapse on the date on which the employment of the Award Recipient terminates as a result of (i) death or (ii) disability within the meaning of Section 22(e)(3) of the Internal Revenue Code.

2.3 Termination of Employment Following a Change of Control of CenturyLink:

- (a) If the shares of Restricted Stock have not already vested or been forfeited under the terms of this Agreement or the Plan, and within [___] months following a Change of Control of CenturyLink (as defined in the Plan), the Award Recipient's employment is terminated by the Company or its Affiliates without Cause (as defined below) or by the Award Recipient for Good Reason (as defined below), then the Award Recipient shall retain the rights to all Performance-Vested Shares, provided that the issuance of such shares shall nonetheless remain subject to the terms and conditions of Section 1, including the eligibility to vest in Additional Shares, except that, unless otherwise provided by the Committee, (A) [CALCULATION OF PERFORMANCE METRIC OVER TRUNCATED PERIOD ENDING ON DATE OF CHANGE OF CONTROL], and (B) the payout of vested Performance-Vested Shares and Additional Shares, if any, shall be made as soon as reasonably practicable following such termination of the Award Recipient but not later the 60 th day following the Award Recipient's termination of employment, provided the Release Condition (as defined in Section 2.3(e) has been satisfied. If the Release Condition is not satisfied, then all unvested Restricted Stock shall automatically terminate and be forfeited as of the 60 th day following termination of employment.
- (b) (i) For purposes of this Section 2.3, "Cause" shall mean the Award Recipient's (A) willful breach of any nondisclosure, noncompetition, nonsolicitation or nondisparagement covenants contained in any agreement between the Company and the Award Recipient; (B) conviction of, or plea of guilty or *nolo contendere* to, a felony or other crime involving dishonesty or moral turpitude; (C) workplace conduct resulting in the payment of civil monetary penalties or the incurrence of civil non-monetary penalties that will materially restrict or prevent the Award Recipient from discharging his obligations to the Company; (D) habitual intoxication during working hours or habitual abuse of or addiction to a controlled substance; (E) material breach of the Company's insider trading,

corporate ethics and compliance policies and programs or any other Board-adopted policies applicable to management conduct; (F) participation in the public reporting of any information contained in any report filed by the Company with the Securities and Exchange Commission that was impacted by the Award Recipient's knowing or intentional fraudulent or illegal conduct; or (G) substantial, willful and repeated failure to perform duties as instructed by or on behalf of the Board in writing.

- (ii) The Award Recipient's employment shall not be deemed terminated for Cause unless the Company shall have delivered to the Award Recipient a termination notice with a copy of a resolution adopted by the affirmative vote of not less than three-quarters of the entire Board at a meeting called partly or wholly for such purpose (after reasonable notice is provided to the Award Recipient and the Award Recipient has had an opportunity, with counsel, to be heard by the Board) finding that the Award Recipient should be terminated for Cause and specifying in reasonable detail the grounds therefor.
- (i) No action or inaction shall be deemed the basis for Cause unless the Award Recipient is terminated therefor prior to the first anniversary of the date on which such action or omission is first known to [_____].
- (a) For purpose of this Section 2.3, "Good Reason" shall mean a termination of the Award Recipient's employment under the following circumstances: (1) the Award Recipient has delivered a written notice to the Company, objecting to a "Good Reason Event" (as defined below) within 90 days following the initial existence or occurrence of such event, (2) the Company fails to cure such event or condition within 30 days following receipt of the Award Recipient's written notice (the "30-day Cure Period"), and (3) as a result, the Award Recipient terminates his or her employment no later than [__] months following the expiration of the 30-day Cure Period. A "Good Reason Event" shall mean:
- (i) Any failure of the Company or its Affiliates (as defined below) to provide the Award Recipient with a position, authority, duties and responsibilities at least commensurate in all material respects with the most significant of those held, exercised and assigned at any time during the 180-day period immediately preceding the Change of Control. The Award Recipient's position, authority, duties and responsibilities after a Change of Control shall not be considered commensurate in all material respects with the Award Recipient's position, authority, duties and responsibilities prior to a Change of Control unless after the Change of Control the Award Recipient holds an equivalent position with, and exercises substantially equivalent authority, duties and responsibilities on behalf of, the Post-Transaction Company;
- (ii) The assignment to the Award Recipient of any duties inconsistent in any material respect with the Award Recipient's position (including status, offices, titles and reporting requirements), authority, duties or responsibilities at the time of the Change of Control, or any other action that results in a diminution in any material respect in such position, authority, duties or responsibilities;

- (iii) A reduction of the Award Recipient's base salary in effect as of the date of the Change of Control without the Award Recipient's consent, except for across-the-board salary reductions similarly affecting all or substantially all similarly-situated officers of the Company and the Post-Transaction Company;
- (iv) The Award Recipient is advised of, manifests an awareness of, or becomes aware of facts that would cause a reasonable person to inquire into any failure in any material respect by the Company or its Affiliates to comply with any of the provisions of this Agreement; or
- (v) Any directive requiring the Award Recipient to be based at any office or location more than 50 miles from the location the Award Recipient was based prior to the Change of Control, or requiring the Award Recipient to travel on business to a substantially greater extent than required immediately prior to the Change of Control.
- (b) For purposes of this Section 2.3, "Affiliate" (or variants thereof) shall mean a person that controls, or is controlled by or is under common control with, another specified person, either directly or indirectly.
- (c) For purposes of this Section 2.3 and 2.4, "Release Condition" shall mean the Award Recipient's execution, delivery to the Company and non-revocation of a mutual liability release agreement in the form and substance determined by the Company (and the expiration of any revocation period contained in such release agreement) within 60 days following the Award Recipient's termination of employment.
- 2.4 If the shares of Restricted Stock have not already vested or been forfeited under the terms of this Agreement or the Plan, and the Award Recipient's employment terminates because of retirement on or after attaining the age of 55 with at least ten years of prior service with the Company, then, provided the Committee has specifically approved such action, and the Release Condition (as defined in Section 2.3) has been satisfied, the Award Recipient shall retain the rights to the Performance-Vested Shares, in full or in part (as determined by the Committee), provided that the issuance of such shares shall nonetheless remain subject to the terms and conditions of Section 1, including the eligibility to vest in Additional Shares.

3. TERMINATION OF EMPLOYMENT

Notwithstanding anything in this Agreement to the contrary, all unvested Restricted Stock shall automatically terminate and be forfeited if the employment of the Award Recipient terminates for any reason, unless and to the extent otherwise provided in Section 2.

4. FORFEITURE OF AWARD

4.1 If, at any time during the Award Recipient's employment by the Company or within 18 months after termination of employment, the Award Recipient engages in any activity in

competition with any activity of the Company, or inimical, contrary or harmful to the interests of the Company, including but not limited to: (3) conduct relating to the Award Recipient's employment for which either criminal or civil penalties against the Award Recipient may be sought; (3) conduct or activity that results in termination of the Award Recipient's employment for cause; (3) violation of the Company's policies, including, without limitation, the Company's insider trading, ethics and corporate compliance policies and programs; (3) participating in the public reporting of any financial or operating result that was impacted by the participant's knowing or intentional fraudulent or illegal conduct; (3) accepting employment with, acquiring a 5% or more equity or participation interest in, serving as a consultant, advisor, director or agent of, directly or indirectly soliciting or recruiting any employee of the Company who was employed at any time during the Award Recipient's tenure with the Company, or otherwise assisting in any other capacity or manner any company or enterprise that is directly or indirectly in competition with or acting against the interests of the Company or any of its lines of business (a "competitor"), except for (i) any isolated, sporadic accommodation or assistance provided to a competitor, at its request, by the Award Recipient during the Award Recipient's tenure with the Company, but only if provided in the good faith and reasonable belief that such action would benefit the Company by promoting good business relations with the competitor and would not harm the Company's interests in any substantial manner or (ii) any other service or assistance that is provided at the request or with the written permission of the Company; (3) disclosing or misusing any confidential information or material concerning the Company, except for any disclosures provided in good faith to regulators in response to inquiries or investigations or otherwise made in good faith to any regulator or law enforcement authority; (3) engaging in, promoting, assisting or otherwise participating in a hostile takeover attempt of the Company or any other transaction or proxy contest that could reasonably be expected to result in a Change of Control not approved by the Board; or (3) making any statement or disclosing any information to any customers, suppliers, lessors, lessees, licensors, licensees, employees, or others with whom the Company engages in business that is defamatory or derogatory with respect to the business, operations, technology, management, or other employees of the Company, or taking any other action that could reasonably be expected to injure the Company in its business relationships with any of the foregoing parties or result in any other detrimental effect on the Company, except for any statements or disclosures provided in good faith to regulators in response to inquiries or investigations or otherwise made in good faith to any regulator or law enforcement authority, then the award of Restricted Stock granted hereunder shall automatically terminate and be forfeited effective on the date on which the Award Recipient engages in such activity and (1) all shares of Common Stock acquired by the Award Recipient pursuant to this Agreement (or other securities into which such shares have been converted or exchanged) shall be returned to the Company or, if no longer held by the Award Recipient, the Award Recipient shall pay to the Company, without interest, all cash, securities or other assets received by the Award Recipient upon the sale or transfer of such stock or securities, and (2) all unvested shares of Restricted Stock and contingent rights to receive Additional Shares shall be forfeited.

4.2 If the Award Recipient owes any amount to the Company under Section 4.1 above, the Award Recipient acknowledges that the Company may, to the fullest extent permitted by applicable law, deduct such amount from any amounts the Company owes the Award Recipient from time to time for any reason (including without limitation amounts owed to the Award Recipient as salary, wages, reimbursements or other compensation, fringe benefits, retirement benefits or

vacation pay). Whether or not the Company elects to make any such set-off in whole or in part, if the Company does not recover by means of set-off the full amount the Award Recipient owes it, the Award Recipient hereby agrees to pay immediately the unpaid balance to the Company.

4.3 The Award Recipient may be released from the Award Recipient's obligations under Sections 4.1 and 4.2 above only if the Committee or its delegee determines in its sole discretion that such action is in the best interests of the Company.

5. STOCK CERTIFICATES

No stock certificates evidencing the Restricted Stock shall be issued by CenturyLink until the lapse of restrictions under the terms of this Agreement. Instead, ownership of the Restricted Stock shall be evidenced by a book entry with the applicable restrictions reflected. Upon the lapse of restrictions on shares of Restricted Stock, CenturyLink shall issue the vested shares of Restricted Stock (either through book entry issuances or delivery of a stock certificate) in the name of the Award Recipient or his or her nominee, subject to the other terms and conditions of this Agreement, including those governing any withholdings of shares under Section 6 below. Upon receipt of any such vested shares, the Award Recipient is free to hold or dispose of such shares, subject to (3) applicable securities laws, (3) CenturyLink's policy statement on insider trading, and (3) any of CenturyLink's stock ownership guidelines then in effect that are applicable to the Award Recipient.

6. WITHHOLDING TAXES

Notwithstanding any Plan provision to the contrary, unless the Award Recipient has previously provided the Company with payment of all applicable withholding taxes, at the time that all or any portion of the Restricted Stock vests, CenturyLink shall withhold from the shares the Award Recipient otherwise would receive under this Agreement the number of whole shares of Common Stock having a value equal to the maximum statutory amount required to be withheld under federal, state and local law, rounding up if necessary.

7. ADDITIONAL CONDITIONS

Anything in this Agreement to the contrary notwithstanding, if, at any time prior to the vesting of the Restricted Stock in accordance with Section 1 or 2, CenturyLink further determines, in its sole discretion, that the listing, registration or qualification (or any updating of any such document) of the shares of Common Stock issuable pursuant hereto is necessary on any securities exchange or under any federal or state securities or blue sky law, or that the consent or approval of any governmental regulatory body is necessary or desirable as a condition of, or in connection with the issuance of shares of Common Stock pursuant thereto, or the removal of any restrictions imposed on such shares, such shares of Common Stock shall not be issued, in whole or in part, or the restrictions thereon removed, unless such listing, registration, qualification, consent or approval shall have been effected or obtained free of any conditions not acceptable to CenturyLink.

CenturyLink agrees to use commercially reasonable efforts to issue all shares of Common Stock issuable hereunder on the terms provided herein.

8. NO CONTRACT OF EMPLOYMENT INTENDED

Nothing in this Agreement shall confer upon the Award Recipient any right to continue in the employment of the Company, or to interfere in any way with the right of the Company to terminate the Award Recipient's employment relationship with the Company at any time.

9. BINDING EFFECT

Upon being duly executed and delivered by CenturyLink and the Award Recipient, this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective heirs, executors, administrators, legal representatives and successors. Without limiting the generality of the foregoing, whenever the term "Award Recipient" is used in any provision of this Agreement under circumstances where the provision appropriately applies to the heirs, executors, administrators or legal representatives to whom this award may be transferred by will or by the laws of descent and distribution, the term "Award Recipient" shall be deemed to include such person or persons.

10. EFFECT OF PLAN TERMS AND COMMITTEE ACTIONS

- 10.1 Capitalized terms used but not defined in this Agreement shall have the respective meanings ascribed to them in the Plan.
- 10.2 This Agreement, the rights of the Award Recipient hereunder and the shares of Restricted Stock granted hereby are subject to (i) all of the terms, conditions, restrictions and other provisions of the Plan, as it may be amended from time to time, as fully as if all such provisions were set forth in their entirety in this Agreement and (ii) such rules and regulations as the Committee may adopt for administration of the Plan. It is expressly understood that the Committee is authorized to administer, construe, and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, all of which shall be binding upon the Award Recipient. If any provision of this Agreement conflicts with a provision of the Plan, the Plan provision shall control.
- 10.3 The Plan is discretionary and may be amended, cancelled or terminated by the Company at any time, in its discretion. The grant of Restricted Stock in this Agreement does not create any contractual rights other than as set forth in this Agreement, and does not create a right to receive Restricted Stock or any other Incentives in the future. Future Incentives, if any, will be at the sole discretion of the Company.
- 10.4 The Award Recipient acknowledges receipt from CenturyLink of a copy of the Plan and a prospectus summarizing the Plan and further acknowledges that the Award Recipient was advised to review such materials prior to entering into this Agreement. The Award Recipient waives

the right to claim that the provisions of the Plan are not binding upon the Award Recipient and the Award Recipient's heirs, executors, administrators, legal representatives and successors.

11. ATTORNEYS' FEES AND EXPENSES

Should any party hereto retain counsel for the purpose of enforcing, or preventing the breach of, any provision of this Agreement, including, but not limited to, the institution of any action or proceeding in court to enforce any provision of this Agreement, to enjoin a breach of any provision of this Agreement, to obtain monetary or liquidated damages for failure to perform any provision of this Agreement, or for a declaration of such parties' rights or obligations under this Agreement, or for any other judicial remedy, then the prevailing party shall be entitled to be reimbursed by the losing party for all costs and expenses incurred thereby, including, but not limited to, attorneys' fees (including costs of appeal).

12. GOVERNING LAW

This Agreement shall be governed by and construed in accordance with the laws of the State of Louisiana. The Award Recipient and CenturyLink shall submit to the exclusive jurisdiction of, and venue in, the courts in Louisiana in any dispute relating to this Agreement.

13. SEVERABILITY

If any term or provision of this Agreement, or the application thereof to any person or circumstance, shall at any time or to any extent be invalid, illegal or unenforceable in any respect as written, the Award Recipient and CenturyLink intend for any court construing this Agreement to modify or limit such provision so as to render it valid and enforceable to the fullest extent allowed by law. Any such provision that is not susceptible of such reformation shall be ignored so as to not affect any other term or provision of this Agreement, and the remainder of this Agreement, or the application of such term or provision to persons or circumstances other than those as to which it is held invalid, illegal or unenforceable, shall not be affected thereby and each term and provision of this Agreement shall be valid and enforced to the fullest extent permitted by law.

14. OTHER PROVISIONS

14.1 It is intended that the payments and benefits provided under this Agreement will comply with the requirements of Section 409A of the Code and the regulations promulgated thereunder ("Section 409A") or an exemption therefrom. The Agreement shall be interpreted, construed, administered, and governed in a manner that effects such intent. No acceleration of the vesting of any Additional Shares shall be permitted unless permitted under Section 409A.

- 14.2 The Plan and this Agreement contain the entire agreement between the parties with respect to the subject matter contained herein. This Agreement may not, without the Award Recipient's consent, be amended or modified so as to materially adversely affect the Award Recipient's rights under this Agreement, except (i) as provided in the Plan, as it may be amended from time to time in the manner provided therein, or (ii) by a written document signed by each of the parties hereto. Any oral or written agreements, representations, warranties, written inducements, or other communications with respect to the subject matter contained herein made prior to the execution of the Agreement shall be void and ineffective for all purposes.
- 14.3 Nothing expressed or implied in this Agreement is intended or shall be construed to confer upon or give any person, other than the parties hereto and their successors, assigns, heirs, executors, administrators, or legal representatives, any rights or remedies under, or by reason of, this Agreement.

15. ELECTRONIC DELIVERY AND EXECUTION OF DOCUMENTS

- 15.1 The Company may, in its sole discretion, deliver any documents related to the Award Recipient's current or future participation in the Plan or any other equity compensation plan of the Company by electronic means or request Award Recipient's consent to the terms of an award by electronic means. The plan documents may, but do not necessarily, include: the Plan, any grant notice, this Agreement, the Plan prospectus, and any reports of CenturyLink provided generally to CenturyLink's shareholders. In addition, the Award Recipient may deliver by electronic means any grant notice or award agreement to the Company or to such third party involved in administering the applicable plan as the Company may designate from time to time. Such means of electronic delivery may include the delivery of a link to a Company intranet or the Internet site of a third party involved in administering the applicable plan, the delivery of the document via e-mail or such other means of electronic delivery specified by the Company. By accepting the terms of this Agreement, the Award Recipient also hereby consents to participate in such plans and to execute agreements setting the terms of participation through an on-line or electronic system as described herein.
- 15.2 The Award Recipient acknowledges that the Award Recipient has read Section 15.1 of this Agreement and consents to the electronic delivery and electronic execution of plan documents as described in Section 15.1. The Award Recipient acknowledges that he or she may receive from the Company a paper copy of any documents delivered electronically at no cost to the Award Recipient by contacting the Company by telephone or in writing. The Award Recipient further acknowledges that the Award Recipient will be provided with a paper copy of any documents if the attempted electronic delivery of such documents to the Award Recipient fails. Similarly, the Award Recipient understands that the Award Recipient must provide the Company or any designated third party administrator with a paper copy of any documents if the attempted electronic delivery of such documents by the Award Recipient fails. The Award Recipient may revoke his or her consent to the electronic delivery and execution of documents described in Section 15.1 or may change the electronic mail address to which such documents are to be delivered (if Award Recipient has provided an electronic mail address) at any time by notifying the Company of such revoked consent or revised e-mail address by telephone, postal service or electronic mail. Finally, the Award Recipient

understands that he or she is not required to consent to electronic delivery or execution of documents described in Section 15.1.

16. Data Privacy

As a condition to his or her participation in the Plan, the Award Recipient consents to the collection, use, and transfer of personal data as described in this paragraph. The Award Recipient understands that the Company holds certain personal information about the Award Recipient, including his or her name, home address and telephone number, date of birth, social security number or identification number, salary, nationality, job title, any shares of stock or directorships held in the Company, details of all options or any other entitlement to shares of stock awarded, canceled, exercised, vested, unvested, or outstanding in the Award Recipient's favor, for the purpose of managing and administering the Plan ("Data"). The Award Recipient further understands that CenturyLink or its subsidiaries will transfer Data amongst themselves as necessary for the purpose of implementation, administration, and management of the Award Recipient's participation in the Plan, and that CenturyLink and any of its subsidiaries may each further transfer Data to any third parties assisting the Company in the implementation, administration, and management of the Plan. The Award Recipient understands that these recipients may be located in the United States or elsewhere, and that the recipients' country may have different data privacy laws and protections that the Award Recipient's country. The Award Recipient authorizes them to receive, possess, use, retain, and transfer the Data, in electronic or other form, for the purposes of implementing, administering, and managing the Award Recipient's participation in the Plan, including any requisite transfer to a broker or other third party with whom the Award Recipient may elect to deposit any amounts received pursuant to the Plan and this Agreement, such Data as may be required for the administration of the Plan. The Award Recipient understands that he or she may, at any time, view Data, require any necessary amendments to it or withdraw the consents herein in writing by contacting his or her human resources representative. The Award Recipient further understands that this consent is purely voluntary, and will not affect the Award Recipient's employment or career with the Company, although it may affect the Award Recipient's ability to participate in the Plan.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered on the day and year first above written.

| | CENTURYLINK, INC. | |
|----------------|-------------------|------------------------|
| | By: | [NAME] [TITLE] |
| N3767827.1} 10 | 1 | [NAME] Award Recipient |

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN

2014 RESTATEMENT EFFECTIVE JANUARY 1, 2014

I. PURPOSE OF THE PLAN

1.01 This Supplemental Dollars & Sense Plan was established by CenturyLink, Inc. (the "Company") and its subsidiaries and designated affiliates to provide to certain select management employees the opportunity to defer a portion of their compensation in excess of the deferrals permissible under the terms of the CenturyLink Dollars & Sense 401(k) Plan (the "Dollars & Sense Plan") maintained by the Company and to allow the Company to make matching contributions based on such deferrals in excess of those permissible under such plan. This Plan is not intended to constitute a qualified plan under Section 401(a) of the Internal Revenue Code of 1986, as amended (the "Code"), and is designed to be exempt from the participation, vesting, funding and fiduciary responsibility rules of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"). The Plan is intended to comply with Code §409A. The Plan was restated effective January 1, 2005 and January 1, 2008. This document again restates the Plan to incorporate prior amendments, to reflect revised plan governance procedures, and to make certain other clarifications.

II. DEFINITIONS

As used in this Plan, the following terms shall have the meanings indicated, unless the context otherwise specifies or requires:

- 2.01 **ACCOUNT** shall mean the account established under this Plan in accordance with Section 4.01.
- ACCOUNT BALANCE, as of a given date, shall mean the fair market value of a Participant's Account as determined by the Committee. In 2005, each active Participant was given the right to elect to have his Account Balance transferred to the CenturyLink Retirement Plan ("Retirement Plan") to the extent permitted under the QSERP concept (*i.e.* to the extent possible given discrimination limitations applicable to the Retirement Plan). Each Participant's Account Balance was reduced by the amount that was transferred to the Retirement Plan, if any. In 2005, under the Code §409A transition rules, each Participant was also given the right to elect to take a distribution of the portion of his Account Balance that was not transferred to the Retirement Plan. Each Participant's Account Balance was reduced by the amount distributed to him in 2005, if any.
- 2.03 **BENEFICIARY** shall mean the person or persons designated by the Participant to receive benefits after the death of the Participant.
- 2.04 **BONUS COMPENSATION** shall mean the amount of a Participant's annual cash bonus from which the Participant can no longer make deferral contributions under the Dollars &

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

Sense Plan due to having reached the limit on deferrals under Code Section 402(g) for the Plan Year in question (ignoring any catchup contributions to the Dollars & Sense Plan for those age 50 or higher, and ignoring any subsequent adjustments to contributions under such plan, such as due to nondiscrimination testing refunds, corrective distributions, or other changes in the ultimate amount of deferrals under such plan for the year in question). If a Participant receives a cash bonus and elected to defer a portion of such bonus to the Dollars & Sense Plan, thereby causing the Participant to reach the Code Section 402(g) limit, then Bonus Compensation under this Plan shall only include the portion of the bonus that did not result in additional deferrals to the Dollars & Sense Plan (having reached the Code Section 402(g) limit).

- 2.05 **COMMITTEE** shall mean the CenturyLink Employee Benefits Committee or its designee in Human Resources, Benefits.
 - 2.06 **COMPENSATION COMMITTEE** means the Compensation Committee of the Board of Directors of the Company.
- 2.07 **EFFECTIVE DATE** of this Plan shall mean the first day of the first payroll period commencing on or after January 1, 1995. The effective date of this Restatement shall mean January 1, 2014.
- 2.08 **EMPLOYER** shall mean the Company, any Subsidiary thereof, and any affiliate designated by the Company as a participating employer under this Plan.
- 2.09 **EXCESS SALARY** shall mean the amount of a Participant's regular salary from which the Participant can no longer make deferral contributions under the Dollars & Sense Plan due to having reached the limit on deferrals under Code Section 402(g) for the Plan Year in question (ignoring any catch-up contributions to the Dollars & Sense Plan for those age 50 or higher, and ignoring any subsequent adjustments to contributions under such plan, such as due to nondiscrimination testing refunds, corrective distributions, or other changes in the ultimate amount of deferrals under such plan for the year in question). Salary shall not be deferred under this Plan with respect to any regular payroll payment from which any Dollars & Sense Plan contributions are withheld, but rather, Excess Salary for purposes of this Plan shall relate to salary earned beginning with the following payroll period. As provided in Treasury Regulations §1.409A-2(a)(13), compensation paid after the last day of the Plan Year solely for services performed during the final payroll period described in Code Section 3401(b) shall be treated as compensation for services provided for the subsequent calendar year.
- 2.10 **INCENTIVE COMPENSATION** shall mean any amount awarded to a Participant under the Company's Key Employee Incentive Compensation Plan or other similar executive incentive compensation arrangement maintained by the Company (other than Bonus Compensation defined in Section 2.04), including the amount of any stock award in its cash equivalent at the time of conversion of the award from cash to stock. A Participant's Incentive Compensation shall be determined on an annual basis and shall, for purposes of this Plan, be allocated to the year in which the Participant performed the services with respect to which the Incentive Compensation was awarded.

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

- 2.11 **NOTIONAL** shall mean imaginary, not actual.
- 2.12 **PARTICIPANT** shall mean any officer of the Company, any Subsidiary thereof, and any designated affiliate, who is granted participation in the Plan in accordance with the provisions of Article III.
 - 2.13 **PLAN** shall mean the CenturyLink, Inc. Supplemental Dollars & Sense Plan, as amended and restated herein.
 - 2.14 PLAN YEAR shall mean the calendar year.
- 2.15 **PROFIT SHARING ACCOUNT** shall mean an account first established in 2006 and continuing thereafter under this Plan to which contributions under Section 6.03 were credited. The Profit Sharing Accounts shall be merged into the Accounts for Participants as of the Effective Date and the separate existence of Profit Sharing Accounts shall cease.
- 2.16 **SPECIFIED EMPLOYEE** shall mean a Participant who is a key employee of the Employer under Treasury Regulations §1.409A-1(i) because of final and binding action taken by the Board of Directors of the Company or its Compensation Committee, or by operation of law or such regulation.
- 2.17 **SUBSIDIARY** shall mean any corporation in which the Company owns, directly or indirectly through subsidiaries, at least 50% of the combined voting power of all classes of stock.
 - 2.18 **TRANSFER ACCOUNT** shall mean the account established under this Plan in accordance with Section 4.01.

III. PARTICIPATION

- 3.01 Any employee who is either one of the officers of an Employer in a position to contribute materially to the continued growth and future financial success of an Employer, or one who has made a significant contribution to the Employer's operations, thereby meriting special recognition, shall be eligible to participate provided the following requirements are met:
 - (a) The officer is employed on a full-time basis by an Employer and is compensated by a regular salary; and
- **(b)** The coverage of the officer is duly approved by the Vice President of Compensation and Benefits (the "Approving Officer").
- 3.02 It is intended that participation in this Plan shall be extended only to those officers who are members of a select group of management or highly compensated employees, as determined by the Approving Officer. If a Participant who retired or otherwise terminated employment is rehired, he shall not again become a Participant in the Plan unless the coverage of the officer is again duly approved by the Approving Officer.

IV. ACCOUNTS AND INVESTMENTS

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

- 4.01 An Account shall be established on behalf of each Participant who receives an allocation pursuant to Sections 6.01 and 6.02. Each Participant's Account shall be credited with such allocation, and earnings and gains on his Account Balance, and shall be debited with distributions, losses, and any expenses properly chargeable thereto. A Transfer Account has been established on behalf of each former inactive Participant in the CenturyLink, Inc. Supplemental Defined Contribution Plan ("SDC Plan") who elected no later than December 15, 2005 to have his account balance in that plan transferred to another nonqualified plan of the Company. Such Transfer Account holds the amount transferred from the SDC plan to this Plan for each such inactive Participant. No other amounts shall be accepted as Transfers to a Transfer Account. Such Transfer Account shall be treated as if it were an Account under this Plan, except that in lieu of any other earnings, the balance in each Transfer Account shall be credited with interest at the rate equal to the 6 month Treasury bill rate adjusted each January 1, and the form of payment shall be the form of payment the Participant elected under the SDC Plan and not a lump sum cash payment under Section 9.01. The form of payment the Participant selected under the SDC Plan cannot be changed.
- 4.02 Each Participant shall have the same rights with respect to investment of amounts in his Account hereunder as are available from time to time under the Dollars & Sense Plan, as to permissible investment funds, except as provided below. Investment in securities or other obligations issued by the Employer will not be available under the Plan. The investment rights of each Participant hereunder shall extend to all amounts in his Account, including deferral contributions and matching contributions.
- 4.03 The Account Balances of Participants in the Plan shall be revalued as of the end of each trading day, taking into account the values of the various assets which are Notional investments of the Accounts and taking into account Notional contributions or transfers to each Account during the day and Notional withdrawals or transfers from each Account during the day.

V. PARTICIPANT DEFERRALS

- 5.01 Deferrals shall comply with the following requirements, in addition to other requirements set forth in the Plan:
- (a) <u>Deferral Elections and Sources:</u> Each Participant shall make separate written elections, prior to the first day of each Plan Year (or, as to a Participant who first becomes a Participant in the Plan as of a day other than January 1 and who is not then a participant in any other account balance plan of or agreement with the Employer governed by Code §409A that permits elective deferrals by the Participant, as defined in Treasury Regulations §1.409A-1(c)(2)(i)(A), within 30 days after the date the Participant becomes eligible to Participate in the Plan (but only with respect to compensation paid for services to be performed subsequent to the election)) to defer a portion of his: (i) Excess Salary; (ii) Bonus Compensation; and/or (iii) Incentive Compensation. The amount of allowable deferral pursuant to each of the Participant's elections shall be a whole percentage, not to exceed 50% (25% prior to January 1, 2011).
- **(b)** <u>Compensation Eligible for Deferral</u>: An election to defer Excess Salary shall provide for a deferral to be made from each paycheck, but only as to payroll periods that fall after the payroll period in which Participant deferred sufficient compensation to reach the deferral

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

limit under Code Section 402(g) (as further detailed in Section 2.09). Similarly, an election to defer Bonus Compensation shall provide for a deferral to be made from the bonus payroll check following the participant having reached the Code Section 402(g) limit under the Dollars & Sense Plan (as further detailed in Section 2.04) or alternatively, from the bonus check from which the Participant has made sufficient deferrals to the Dollars & Sense Plan to reach the limit detailed in Section 2.04. Deferrals of Incentive Compensation shall not be contingent on the Participant having reached the Code Section 402(g) limit under the Dollars & Sense Plan.

- (c) <u>Performance-Based Compensation:</u> Notwithstanding the above, with respect to any Bonus or Incentive Compensation that is performance-based, as defined in Treasury Regulations §1.409A(1)(e), each Participant may make a separate written election no later than June 30 of the calendar year performance period. Rehires shall be treated as if they were in their first year of eligibility if they satisfy the 24 month rule in Treasury Regulations §1.409A-2(a)(7)(ii).
- 5.02 Any election made under the terms of Section 5.01 shall be irrevocable until the succeeding January 1. Participants in the Plan shall likewise be unable to change or revoke their deferral elections under the Dollars & Sense Plan for the Plan Year in question (with the exception of after-tax and catch-up contributions under such plan), in accordance with the terms of such plan. However, as permitted by Treasury Regulations §1.409A-3(j)(4)(viii), a Participant may cancel his deferral election under the Plan (and under the Dollars & Sense Plan) due to an unforeseeable emergency or in order to qualify for a hardship distribution pursuant to Treasury Regulations §1.401(k)-1(d)(3).
- 5.03 If a Participant does not make new elections for a succeeding Plan Year under Section 5.01, he will not be able to defer under the Plan for such succeeding Plan Year.

VI. ALLOCATIONS TO PARTICIPANT'S ACCOUNTS

- 6.01 The Employer shall allocate to each Participant's Account the amount of Excess Salary, Bonus Compensation and/or Incentive Compensation deferred by such Participant pursuant to an election made under Section 5.01. The allocation hereunder shall be made as soon as administratively feasible following the date of the payroll check or bonus check to which the deferral by the Participant relates.
 - 6.02 Matching Contributions:
- (a) Excess Salary: The Employer shall allocate a matching contribution to each Participant's Account under this Plan for each applicable pay period during the Plan Year an amount equal to the matching percentage under the Dollars & Sense Plan, applied against the Excess Salary deferral made under this Plan for such pay period.
- **(b)** Bonus Compensation: The Employer may contribute a matching contribution to the Plan based on the matching percentage under the Dollars & Sense Plan, provided the Participant elected to defer a portion of his bonus to this Plan. The amount of match on such bonus deferral is contingent on: (i) the Participant's bonus deferral percentage elected under the Dollars

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

and Sense Plan and (ii) the bonus deferral percentage under this Plan. Matching contributions on Bonus Compensation will be made first to the Dollars and Sense Plan, to the extent of the percentage of the bonus deferred into such Plan, and any excess will be contributed to this Plan. For example, if the matching formula under the Dollars & Sense Plan is a 100% match on the first 1% of bonus deferred, and a 50% match on the next 5% of bonus deferred, and if a Participant elected to defer 6% of his Bonus Compensation to the Dollars and Sense Plan and to this Plan, but he is only able to defer 4% of the bonus into the Dollars & Sense Plan before reaching the limit described in Section 2.04, the Participant would receive a match of 2.5% of the bonus into the Dollars & Sense Plan (1% plus 50% of 3%), and the remaining 1% match (50% of 2%) would be credited to this Plan (since he elected a deferral of at least 2% under this Plan). Note that if a Participant elects to defer a percentage of the bonus to the Dollars and Sense Plan or this Plan

(c) Incentive Compensation: The Employer shall allocate a matching contribution to each Participant's Account under this Plan of an amount equal to the matching percentage under the Dollars & Sense Plan, applied against any Incentive Compensation deferral made under this Plan during a Plan Year.

VII. <u>VESTING OF ACCOUNT</u>

- 7.01 A Participant's Account Balance shall be fully vested at all times.
- 7.02 The Profit Sharing Account of each Participant who was an active employee on November 6, 2006 shall be fully vested and nonforfeitable at all times.

VIII. TIME OF PAYMENT AND BENEFICIARIES,

- 8.01 Except as provided in Section 8.02 and Articles X and XI, a Participant's vested Account Balance is payable immediately upon termination of employment for any reason.
- 8.02 The Account Balance of a deceased Participant shall be paid within 90 days after his death, and shall be made to his Beneficiary designated on a form provided for such purpose by the Committee. If no beneficiary has been properly designated or if no Designated Beneficiary survives the Participant, the death benefits, if any, payable to the beneficiary of the deceased Participant shall be paid to the Participant's surviving spouse. If there is no surviving spouse, the death benefits shall be payable to the Participant's estate (which shall include the Participant's probate estate or living trust).

IX. FORM OF BENEFIT PAYMENT

9.01 Except as provided in Section 4.01, all payments of the Participant's Account Balance to the Participant or to the Participant's Beneficiary shall be in the form of a lump sum cash payment.

X. ACCELERATION OF PAYMENTS.

10.01 If at any time the Plan fails to meet the requirements of Code §409A, an amount equal to the amount required to be included in the Participant's income as a result of the failure to

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

comply with the requirements of Code §409A shall be paid to the Participant in one lump sum on the first day of the month following the Company's determination that the failure has occurred.

- 10.02 If the Plan receives a domestic relations order as defined in Code §414(p)(1)(B) and ERISA §206(d)(3)(B)(ii), the Committee shall accelerate the time of payment to an individual other than the Participant as may be necessary to fulfill such order in an amount not to exceed the Participant's Account Balance, provided that the provisions of ERISA §206(d)(3)(C) through (F) shall apply as if this Plan were governed by part 2 of Title I of ERISA.
- 10.03 The Committee shall accelerate the time or schedule of a payment under the Plan as may be necessary: (1) to comply with an ethics agreement between the Participant and the Federal government, or (2) to comply with applicable Federal, state, local or foreign ethics laws or conflict of interest laws; each as described in Treasury Regulations §1.409A-3(j)(4)(iii).

XI. DELAY OF PAYMENTS

- 11.01 A payment otherwise due hereunder shall be delayed to a date after the designated payment date under the following circumstances:
- (a) Notwithstanding any other provision hereof, any payment that constitutes deferred compensation under Code §409A and the Treasury Regulations thereunder and that is not exempt from coverage by Code §409A and the Treasury Regulations thereunder shall commence upon termination of employment of a Participant who is a Specified Employee on the first day of the seventh month following the date of the Specified Employee's termination of employment, or, if earlier, the date of death of the Specified Employee. Nevertheless, for all other purposes of this Agreement, a payment shall be deemed to have been made on the date it would have been paid had the Employee not been a Specified Employee.
- **(b)** Notwithstanding any other provision hereof, a Participant shall not have separated from service with the Employer on account of termination of employment for reasons other than death if he would not be deemed to have experienced a termination of employment under the default rules of Treasury Regulations §1.409A-1(h).
- (c) If the making of the payment at the date specified under the Plan would jeopardize the ability of the Employer to continue as a going concern (in such case, payment will be made during the first taxable year of the Employer in which the making of the payment would not have such effect).
- (d) Payment where the Employer reasonably anticipates that the making of the payment will violate Federal securities laws or other applicable law, provided that the payment shall be made at the earliest date at which the Employer reasonably anticipates that the making of the payment will not cause such violation. (The making of a payment that would cause inclusion in gross income or the application of any penalty provision or other provision of the Code is not treated as a violation of applicable law).

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

- (e) Payments the deduction for which the Employer reasonably anticipates would be limited by the application of Code §162(m) (in such case, payment will be made at either the earliest date at which the Employer reasonably anticipates that the deduction of the payment will not be so limited or the calendar year in which the Participant separates from service).
- **(f)** Payment may also be delayed upon such other events and conditions as the Commissioner of Internal Revenue may prescribe in generally applicable guidance published in the Internal Revenue Bulletin.

XII. ADDITIONAL RESTRICTIONS ON BENEFIT PAYMENTS

12.01 In no event will there be a duplication of benefits payable under the Plan because of employment by more than one participating Employer.

XIII. ADMINISTRATION AND INTERPRETATION

- 13.01 The Plan shall be administered by the Committee. The Committee shall have full power and authority to interpret and administer the Plan and, subject to the provisions herein set forth, to prescribe, amend and rescind rules and regulations and make all other determinations necessary or desirable for the administration of the Plan.
- 13.02 The decision of the Committee relating to any question concerning or involving the interpretation or administration of the Plan shall be final and conclusive, subject to any appeal rights of Participants set forth in Article XIX.

XIV. NATURE OF THE PLAN

14.01 Benefits under the Plan shall generally be payable by the Employer from its own funds, and such benefits shall not (i) impose any obligation upon the trust(s) of the other employee benefit programs of the Employer; (ii) be paid from such trust(s); nor (iii) have any effect whatsoever upon the amount or payment of benefits under the other employee benefit programs of the Employer. Participants have only an unsecured right to receive benefits under the Plan from the Employer as general creditors of the Employer. The Employer may deposit amounts in a trust established by the Employer for the purpose of funding the Employer's obligations under the Plan. Participants and their beneficiaries, however, have no secured interest or special claim to the assets of such trust, and the assets of the trust shall be subject to the payment of claims of general creditors of the Employer upon the insolvency or bankruptcy of the Employer, as provided in the trust.

XV. EMPLOYMENT RELATIONSHIP

15.01 An employee shall be considered to be in the employment of the Employer as long as he remains an employee of either the Company, any Subsidiary of the Company, any designated affiliate, or any corporation to which substantially all of the assets and business of any of such entities are transferred. Nothing in the adoption of this Plan nor the designation of any Participant shall confer on any employee the right to continued employment by the Employer, or affect in any way the right of the Employer to terminate his employment at any time. Any question as to whether

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

and when there has been a termination of an employee's employment, and the cause, notice or other circumstances of such termination, shall be determined by the Committee, and its determination shall be final.

XVI. AMENDMENT AND TERMINATION OF PLAN

- 16.01 The Company or its designee, the CenturyLink Plan Design Committee, may terminate the Plan and accelerate any payments due (or that may become due) under the Plan:
- (a) Within 12 months of a corporate dissolution of the Company taxed under Code §331, or with the approval of a bankruptcy court pursuant to 11 U.S.C. §503(b)(1)(A), provided that the amounts deferred under the Plan are included in the Participant's gross income in the latest of (i) the calendar year in which the termination occurs, (ii) the calendar year in which the amount is no longer subject to a substantial risk of forfeiture or (iii) the first calendar year in which the payment is administratively practicable.
- **(b)** Within the 30 days preceding or the 12 months following a Change in Control Event (as defined in Treasury Regulations §1.409A-3(i)(5)) provided that Treasury Regulations §1.409A-3(j)(4)(ix)(B) is complied with.
 - (c) In the Company's discretion, provided that Treasury Regulations §1.409A-3(j)(4)(ix)(C) is complied with.
- (d) Due to such other events and conditions as the Commissioner of the IRS may prescribe in generally applicable guidance published in the Internal Revenue Bulletin.
- 16.02 The Company, acting through the Compensation Committee, its Board of Directors, the CenturyLink Plan Design Committee, or any person or entity designated by such entities, may amend this Plan, at any time and from time to time, in its sole discretion. No amendment shall be made to the Plan without the consent of the Company or its designee, the CenturyLink Plan Design Committee. Moreover, no amendments shall divest otherwise vested rights of Participants, their Beneficiaries or Spouses.

XVII. BINDING EFFECT

17.01 This Plan shall be binding on the Company, each Subsidiary and any designated affiliate, the successors and assigns thereof, and any entity to which substantially all of the assets or business of the Company, a Subsidiary, or a designated affiliate are transferred.

XVIII. CONSTRUCTION

18.01 The masculine gender, where appearing in the Plan, shall be deemed to include the feminine gender, and the singular may indicate the plural, unless the context clearly indicates the contrary. The words "hereof", "herein", "hereunder" and other similar compounds of the word "here" shall, unless otherwise specifically stated, mean and refer to the entire Plan, not to any particular provision or Section. Article and Section headings are included for convenience of reference and are not intended to add to, or subtract from, the terms of the Plan.

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

- 18.02 Notwithstanding any other provision of this Plan, it is the intention of the Company that no payment or entitlement pursuant to this Plan will give rise to any adverse tax consequences to any Participant under Code §409A and Treasury Regulations and other interpretive guidance issued thereunder, including that issued after the date hereof (collectively, "Section 409A"). This Plan and any amendments hereto shall be interpreted to that end and (1) to the maximum extent permitted by law, no effect shall be given to any provision herein, any amendment hereto or any action taken hereunder in a manner that reasonably could be expected to give rise to adverse tax consequences under Section 409A and (2) the Company shall take any corrective action reasonably within its control that is necessary to avoid such adverse tax consequences. Any provision of the Plan that would cause a violation of Code §409A if followed shall be disregarded.
- 18.03 Any reference to any section of the Code or the Treasury Regulations shall be deemed to also refer to any successor provisions thereto.

XIX. DEMAND FOR BENEFITS

- 19.01 (a) <u>Filing of Claims for Benefits</u>. Benefits shall ordinarily be paid to a Participant without the need for demand, and to a beneficiary upon receipt of the beneficiary's address and Social Security Number (and evidence of death of the Participant, if needed). Nevertheless, a Participant or a person claiming to be a beneficiary who claims entitlement to a benefit can file a claim for benefits in writing with the Committee.
- (a) <u>Notification to Claimant of Decision</u>. If a claim is wholly or partially denied, a notice of the decision rendered in accordance with the rules set forth below will be furnished to the claimant not later than 90 days after receipt of the claim by the Committee.

If special circumstances require an extension of time for processing the claim, the Committee will give the claimant a written notice of the extension prior to the end of the initial 90 day period. In no event will the extension exceed an additional 90 days. The extension notice will indicate the special circumstances requiring an extension of time and the date by which the Committee expects to render its final decision.

- **(b)** <u>Content of Notice</u>. The Committee will provide to every claimant who is denied a claim for benefits written or electronic notice setting forth in a clear and simple manner:
 - 1. The specific reason or reasons for denial;
 - 2. Specific reference to pertinent plan provisions on which denial is based;
 - 3. A description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such materials or information are necessary; and
 - 4. Appropriate information as to the steps to be taken if the claimant wishes to submit his or her claim for review, including a statement

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

of the claimant's right to bring a civil action under ERISA Section 502(a) following an adverse determination on review.

(c) Review Procedure. After the claimant has received written notification of an adverse benefit determination, the claimant or a duly authorized representative will have 60 days within which to appeal, in writing, such determination. The claimant may submit written comments, documents, records, and any other information relevant to the claim for benefits. The Committee will provide the claimant, upon request and free of charge, reasonable access to and copies of all documents, records, and other information relevant to the claimant's claim for benefits.

The review will take into account all items submitted by the claimant, regardless of whether such information was submitted or considered in the initial benefit determination.

(d) <u>Decision on Review</u>. The decision on review by the Committee will be rendered as promptly as is feasible, but not later than 60 days after the receipt of a request for review, unless the Committee in its sole discretion determines that special circumstances require an extension of time for processing, in which case a decision will be rendered as promptly as is feasible, but not later than 120 days after receipt of a request for review.

If an extension of time for review is required because of special circumstances, written notice of the extension will be furnished to the claimant before termination of the initial 60-day review period and shall indicate the special circumstances requiring an extension of time and the date by which the Committee expects to render the determination on review.

The decision on review will be in written or electronic form. In the event of an adverse benefit determination, the decision shall contain: (1) specific reasons for the adverse determination, written in a clear and simple manner; (2) specific references to the pertinent plan provisions on which the determination is based; (3) a statement that the claimant may request, free of charge, reasonable access to and copies of all documents, records and other information relevant to the claim for benefits; and (4) the claimant's right to bring an action under ERISA Section 502(a).

(e) <u>Failure to Establish and Follow Reasonable Claims Procedure</u>. In the case of the failure of the Committee to establish or follow claims procedures consistent with the requirements of Labor Department Regulations Section 2560.503-1, the claimant shall be deemed to have exhausted the administrative remedies available under the Plan and shall be entitled to pursue any available remedies under section 502(a) of ERISA on the basis that the Plan has failed to provide a reasonable claims procedure that would yield a decision on the merits of the claim.

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

IN WITNESS WHEREOF, CenturyLink, Inc. has executed this Plan this 20th day of December, 2013.

| CENTURYLINK, INC. | |
|--|----------------------------|
| By: <u>/s/</u> <u>Scott A. Trezise</u> | |
| | Scott A. Trezise |
| | Executive Vice President, |
| | Human Resources and Chair, |
| | CenturyLink Plan Design |
| Committee | |
| | |

{N2736618.2}

CENTURYLINK, INC. SUPPLEMENTAL DOLLARS & SENSE PLAN Amended and Restated Effective January 1, 2014

CHANGE OF CONTROL AGREEMENT

CHANGE OF CONTROL AGREEMENT (this "Agreement"), effective as of May 23, 2018 (the "Agreement Date"), between CenturyLink, Inc., a Louisiana corporation (the "Company"), and Jeffrey K. Storey (the "Employee").

WITNESSETH:

WHEREAS, pursuant to an amended and restated offer letter entered into between the Company and the Employee on the Agreement Date (the "Offer Letter"), the Company and the Employee have agreed that the Employee will be provided with certain change of control protections commensurate with his position as President and Chief Executive Officer of the Company;

WHEREAS, the Board of Directors of the Company (the "Board") has determined that it is in the best interests of the Company and its shareholders to enter into this Agreement with the Employee; and

WHEREAS, the Board believes that this Agreement is reasonably designed to retain the services of the Employee and to assure the full dedication of the Employee, free from personal distraction, in the event of an actual or pending change of control of the Company;

NOW, **THEREFORE**, the parties agree as follows:

Article I CERTAIN DEFINITIONS

- 1.1 **Affiliate.** "Affiliate" (and variants thereof) shall mean a Person that controls, or is controlled by, or is under common control with, another specified Person, either directly or indirectly.
- 1.2 **Beneficial Owner.** "Beneficial Owner" (and variants thereof), with respect to a security, shall mean a Person who, directly or indirectly (through any contract, understanding, relationship or otherwise), has or shares (i) the power to vote, or direct the voting of, the security, or (ii) the power to dispose of, or direct the disposition of, the security.
- Business : "Business" shall mean, as of any particular date, the business of (i) providing local or long distance communication services (including CLEC's, ILEC's and international communication services providers with a presence in the United States), network access, Internet access, data, video and IPTV services, web development, hosting, cloud computing, web and managed hosting, content delivery network services, IT services, big data analytics, security; (ii) selling communications products in connection with providing such services or (iii) providing any other material services or selling any other material products then performed or sold by the Company or its Affiliates.

- 1.4 **Cause.** (a) "Cause" shall mean the Employee's (i) willful breach of Section 4.1 or 4.2 of this Agreement; (ii) conviction of, or plea of guilty or *nolo contendere* to, a felony or other crime involving dishonesty or moral turpitude; (iii) workplace conduct resulting in the payment of civil monetary penalties or the incurrence of civil non-monetary penalties that will materially restrict or prevent the Employee from discharging his obligations to the Company; (iv) habitual intoxication during working hours or habitual abuse of or addiction to a controlled substance; (v) material breach of the Company's insider trading, corporate ethics and compliance policies and programs or any other Board-adopted policies applicable to management conduct; (vi) participation in the public reporting of any information contained in any report filed by the Company with the Securities and Exchange Commission that was impacted by the Employee's knowing or intentional fraudulent or illegal conduct; or (vii) substantial, willful and repeated failure to perform duties as instructed by or on behalf of the Board in writing.
- (a) The Employee's employment shall not be deemed terminated for Cause unless the Company shall have delivered to the Employee a termination notice with a copy of a resolution adopted by the affirmative vote of not less than three-quarters of the entire Board at a meeting called partly or wholly for such purpose (after reasonable notice is provided to the Employee and the Employee has had an opportunity, with counsel, to be heard by the Board) finding that the Employee should be terminated for Cause and specifying in reasonable detail the grounds therefor.
- (b) No action or inaction shall be deemed the basis for Cause unless the Employee is terminated therefor prior to the first anniversary of the date on which such action or omission is first known to the Chairman of the Board or the Chair of any standing committee of the Board.

1.5 **Change of Control.** "Change of Control" shall mean:

- (a) the acquisition by any Person of Beneficial Ownership of 30% or more of the outstanding shares of the Company's Common Stock, \$1.00 par value per share (the "Common Stock"), or 30% or more of the combined voting power of the Company's then outstanding securities entitled to vote generally in the election of directors; *provided, however*, that for purposes of this subsection (a), the following acquisitions shall not constitute a Change of Control:
 - (i) any acquisition (other than a Business Combination which constitutes a Change of Control under Section 1.5(c) hereof) of Common Stock directly from the Company,
 - (ii) any acquisition of Common Stock by the Company or its subsidiaries,
 - (iii) any acquisition of Common Stock by any employee benefit plan (or related trust) sponsored or maintained by the Company or any corporation controlled by the Company, or
 - (iv) any acquisition of Common Stock by any entity pursuant to a Business Combination that does not constitute a Change of Control under Section 1.5(c) hereof; or

- (b) individuals who, as of the Agreement Date, constitute the Board (the "Incumbent Board.") cease for any reason to constitute at least a majority of the Board; provided, however, that any individual becoming a director subsequent to the Agreement Date whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds of the directors then comprising the Incumbent Board shall be considered a member of the Incumbent Board, unless such individual's initial assumption of office occurs as a result of an actual or threatened election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Incumbent Board; or
- (c) consummation of a reorganization, share exchange, merger or consolidation (including any such transaction involving any direct or indirect subsidiary of the Company), or sale or other disposition of all or substantially all of the assets of the Company (a "Business Combination"); provided, however, that in no such case shall any such transaction constitute a Change of Control if immediately following such Business Combination,
 - (i) the individuals and entities who were the Beneficial Owners of the Company's outstanding common stock and the Company's voting securities entitled to vote generally in the election of directors immediately prior to such Business Combination have direct or indirect Beneficial Ownership, respectively, of more than 50% of the then outstanding shares of common stock, and more than 50% of the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, of the Post-Transaction Company (as defined in Section 1.13 hereof), and
 - (ii) except to the extent that such ownership existed prior to the Business Combination, no Person (excluding the Post-Transaction Company and any employee benefit plan or related trust of the Company, the Post-Transaction Company or any subsidiary of either corporation) Beneficially Owns, directly or indirectly, 20% or more of the then outstanding shares of common stock of the corporation resulting from such Business Combination or 20% or more of the combined voting power of the then outstanding voting securities of such corporation, and
 - (iii) at least a majority of the members of the board of directors of the Post-Transaction Company were members of the Incumbent Board at the time of the execution of the initial agreement, or of the action of the Board, providing for such Business Combination; or
 - (d) approval by the shareholders of the Company of a complete liquidation or dissolution of the Company.
 - 1.6 Code. "Code" shall mean the Internal Revenue Code of 1986, as amended from time to time.
- 1.7 **Confidential Information**. "Confidential Information" shall mean any information, knowledge or data of any nature and in any form (including information that is electronically transmitted or stored on any form of magnetic or electronic storage media) that directly or indirectly

relates to the past, current or prospective business of the Company and its Affiliates, whether generated by the Company, any of its Affiliates, or any of their respective employees, officers, directors, representatives, consultants, agents or independent contractors, and whether or not marked confidential, including without limitation information relating to operations, products, services, assets, liabilities, franchises, customers, financial condition, results of operations, finances, prospects, strategies, business plans, budgets, projections, pricing information, business acquisitions, joint ventures, processes, research and development ideas, trade secrets, supplier lists, supplier information, distribution and sales data, consultants' reports, marketing strategies, proprietary computer software, and internal notes and memoranda relating to any of the foregoing; *provided*, *however*, that "Confidential Information" shall not include any information that (1) is or becomes generally available to the public other than as a result of a breach of this Agreement, or (1) is or becomes available to the Employee on a non-confidential basis from a source other than the Company, its Affiliates or their respective representatives, provided that such source is not known by the Employee to have violated any confidentiality agreement with the Company in connection with such disclosure.

- 1.8 **Company.** "Company" shall mean CenturyLink, Inc. and shall include any successor to or assignee of (whether direct or indirect, by purchase, share exchange, merger, consolidation or otherwise) all or substantially all of the assets or business of the Company that assumes and agrees to perform this Agreement by operation of law or otherwise.
- 1.9 **Disability.** "Disability" shall mean a condition that would entitle the Employee to receive benefits under the long-term disability insurance policy applicable to the Company's officers at the time because the Employee is totally disabled or partially disabled, as such terms are defined in the policy then in effect. If the Company has no long-term disability plan in effect, "Disability" shall occur if (a) the Employee is rendered incapable because of physical or mental illness of satisfactorily discharging his duties and responsibilities to the Company for a period of 90 consecutive days, (b) a duly qualified physician chosen by the Company and acceptable to the Employee or his legal representatives so certifies in writing, and (c) the Board determines that the Employee has become disabled.
- 1.10 **Employment Term.** "Employment Term" shall mean the period commencing on the date of a Change of Control and ending on the 24-month anniversary of such date.
- 1.11 **Good Reason.** "Good Reason" shall mean any of the following events or conditions described in this Section 1.11, but only if the Employee shall have provided written notice to the Company within 90 days of the initial existence or occurrence of such event or condition and the Company shall have failed to cure such event or condition within 30 days of its receipt of such notice:
- (a) Any failure of the Company or its Affiliates to provide the Employee with a position, authority, duties and responsibilities at least commensurate in all material respects with the most significant of those held, exercised and assigned at any time during the 180-day period immediately preceding the Change of Control. The Employee's position, authority, duties and responsibilities after a Change of Control shall not be considered commensurate in all material respects with the Employee's position, authority, duties and responsibilities prior to a Change of

Control unless after the Change of Control the Employee holds an equivalent position with, and exercises substantially equivalent authority, duties and responsibilities on behalf of, the Post-Transaction Company;

- (b) The assignment to the Employee of any duties inconsistent in any material respect with the Employee's position (including status, offices, titles and reporting requirements), authority, duties or responsibilities as contemplated by Section 3.1(b) of this Agreement, or any other action that results in a diminution in any material respect in such position, authority, duties or responsibilities;
- (c) A reduction of the Employee's base salary in effect as of the date of the Change of Control without the Employee's consent, except for across-the-board salary reductions similarly affecting all or substantially all similarly-situated officers of the Company and the Post-Transaction Company;
- (d) The Employee is advised of, manifests an awareness of, or becomes aware of facts that would cause a reasonable person to inquire into any failure in any material respect by the Company or its Affiliates to comply with any of the provisions of this Agreement; or
- (e) Any directive requiring the Employee to be based at any office or location other than as provided in Section 3.1(b)(ii) hereof or requiring the Employee to travel on business to a substantially greater extent than required immediately prior to the Change of Control.
- 1.12 **Person.** "Person" shall mean a natural person or entity, and shall also mean the group or syndicate created when two or more Persons act as a syndicate or other group (including, without limitation, a partnership or limited partnership) for the purpose of acquiring, holding, or disposing of a security, except that "Person" shall not include an underwriter temporarily holding a security pursuant to an offering of the security.
- 1.13 **Post-Transaction Company.** Unless a Change of Control results from a Business Combination (as defined in Section 1.5(c) hereof), "Post-Transaction Company" shall mean the Company after the Change of Control. If a Change of Control results from a Business Combination, "Post-Transaction Company" shall mean the corporation or other entity resulting from the Business Combination unless, as a result of such Business Combination, an ultimate parent entity controls such resulting entity, the Company or all or substantially all of the Company's assets either directly or indirectly, in which case "Post-Transaction Company" shall mean such ultimate parent entity.
- 1.14 **Specified Employee.** "Specified Employee" shall mean the Employee if the Employee is a key employee under Treasury Regulations Section 1.409A-1(i) because of final and binding action taken by the Board, its Human Resources and Compensation Committee or such committee's delegate, or by operation of law or such regulation.

ARTICLE II STATUS OF CHANGE OF CONTROL AGREEMENTS

As provided in the Offer Letter, the Employee and the Company agree that this Agreement supersedes any and all other prior agreements or arrangements either sponsored by the Company covering the Employee or entered into between the Company and the Employee that provide for severance benefits in the event of a Change of Control of the Company, as defined therein, and is effective as of the Agreement Date for any Change of Control of the Company occurring after such date.

ARTICLE III CHANGE OF CONTROL BENEFITS

- 3.1 **Terms of Employment after Change of Control.** (a) This Agreement shall commence on the Agreement Date and continue in effect through December 31, 2018; *provided, however*, that, commencing on January 1, 2019 and each January 1 thereafter, the term of this Agreement shall automatically be extended for one additional year unless, not later than June 30 of the preceding year, the Company or the Employee shall have given written notice that it does not wish to extend this Agreement; *provided, further*, that, notwithstanding any such non-extension notice by the Company, if a Change of Control of the Company shall have occurred during the original or extended term of this Agreement, this Agreement shall continue in effect through the 24-month anniversary of the Change of Control, subject to any earlier termination of the Employee's status as an employee pursuant to this Agreement; *provided*, *further*, that in no event shall any termination of this Agreement result in any forfeiture of rights that accrued prior to the date of termination.
- (a) During the Employment Term, the Company hereby agrees to continue the Employee in its employ, subject to the terms and conditions of this Agreement. During the Employment Term, (1) the Employee's position (including status, offices, titles and reporting requirements), authority, duties and responsibilities shall be at least commensurate in all material respects with the most significant of those held, exercised and assigned at any time during the 180-day period immediately preceding the Change of Control and (1) the Employee's services shall be performed during normal business hours at the location of the Company's principal executive office at the time of the Change of Control, or the office or location where the Employee was employed immediately preceding the Change of Control or any relocation of any such site to a location that is not more than 50 miles from its location at the time of the Change of Control. The Employee's position, authority, duties and responsibilities after a Change of Control shall not be considered commensurate in all material respects with the Employee's position, authority, duties and responsibilities prior to a Change of Control unless after the Change of Control the Employee holds an equivalent position with, and exercises substantially equivalent authority, duties and responsibilities on behalf of, the Post-Transaction Company.
- 3.2 **Compensation and Benefits.** During the Employment Term, the Employee shall be entitled to the following compensation and benefits:
- (a) <u>Base Salary</u>. The Employee shall receive an annual base salary ("<u>Base Salary</u>"), which shall be paid in at least monthly installments. The Base Salary shall initially be equal to 12 times the highest monthly base salary that was paid or is payable to the Employee, including any base salary which has been earned but deferred by the Employee, by the Company

and its Affiliates with respect to any month in the 12-month period ending with the month that immediately precedes the month in which the Change of Control occurs. During the Employment Term, the Employee's Base Salary shall be reviewed at such time as the Company undertakes a salary review of his peer employees (but at least annually), and, to the extent that salary increases are granted to his peer employees of the Company (or have been granted during the immediately preceding 12-month period to his peer employees of any Affiliate of the Company), the Employee shall be granted a salary increase commensurate with any increase granted to his peer employees of the Company and its Affiliates. Any increase in Base Salary shall not serve to limit or reduce any other obligation to the Employee under this Agreement. Base Salary shall not be reduced during the Employment Term (whether or not any increase in Base Salary occurs) and, if any increase in Base Salary occurs, the term Base Salary as utilized in this Agreement shall refer to Base Salary as so increased from time to time.

- (b) Annual Bonus . In addition to Base Salary, the Employee shall be awarded, for each fiscal year ending during the Employment Term, an annual cash bonus (the "Bonus") in an amount at least equal to the average of the annual bonuses paid to the Employee with respect to the three fiscal years that immediately precede the year in which the Change of Control occurs under the Company's annual bonus plan, or any comparable bonus under a successor plan; provided, however, that if the Company has never paid an annual bonus for a full year to the Employee, the Employee shall be awarded a Bonus in an amount at least equal to the target bonus for which the Employee is eligible for the fiscal year in which the Change of Control occurs, assuming achievement at the target level of the objective performance goals established with respect to such bonus and achievement of 100% of any subjective performance goals or criteria otherwise applicable with respect to such bonus. Each such Bonus shall be paid after the end of the fiscal year and no later than the 15 th day of the third month of the fiscal year next following the fiscal year for which the Bonus is awarded, unless the Employee shall timely elect to defer the receipt of such Bonus pursuant to any deferred compensation plan sponsored by the Company. For purposes of determining the value of any annual bonuses paid to the Employee in any year preceding the year in which the Change of Control occurs, all cash and stock bonuses earned by the Employee shall be valued as of the date of the grant.
- (c) <u>Fringe Benefits</u>. The Employee shall be entitled to fringe benefits commensurate with those provided to his peer employees of the Company and its Affiliates, but in no event shall such fringe benefits be less favorable than the most favorable of those provided by the Company and its Affiliates for the Employee at any time during the one-year period immediately preceding the Change of Control or, if more favorable to the Employee, those provided generally at any time after the Change of Control to his peer employees of the Company and its Affiliates.
- (d) <u>Expenses</u>. The Employee shall be entitled to receive prompt reimbursement for all reasonable expenses incurred by the Employee in accordance with the most favorable agreements, policies, practices and procedures of the Company and its Affiliates in effect for the Employee at any time during the one-year period immediately preceding the Change of Control or, if more favorable to the Employee, as in effect generally at any time thereafter with respect to his peer employees of the Company and its Affiliates.

- (e) <u>Benefit Plans</u>. (i) The Employee shall be entitled to participate in all incentive, savings and retirement plans, practices, policies and programs applicable generally to his peer employees of the Company and its Affiliates, but in no event shall such plans, practices, policies and programs provide the Employee with incentive opportunities (measured with respect to both regular and special incentive opportunities to the extent that any such distinction is applicable), savings opportunities and retirement benefit opportunities, in each case, less favorable than the most favorable of those provided by the Company and its Affiliates for the Employee under any agreements, plans, practices, policies and programs as in effect at any time during the one-year period immediately preceding the Change of Control or, if more favorable to the Employee, those provided generally at any time after the Change of Control to his peer employees of the Company and its Affiliates.
 - (i) The Employee and his family shall be eligible for participation in and shall receive all benefits under welfare benefit plans, practices, policies and programs provided by the Company and its Affiliates (including, without limitation, medical, prescription drug, dental, disability, salary continuance, employee life, group life, accidental death and travel accident insurance plans and programs) to the extent applicable generally to his peer employees of the Company and its Affiliates, but in no event shall such plans, practices, policies and programs provide the Employee and his family with benefits, in each case, less favorable than the most favorable of those agreements, plans, practices, policies and programs in effect for the Employee and his family at any time during the one-year period immediately preceding the Change of Control or, if more favorable to the Employee and his family, those provided generally at any time after the Change of Control to his peer employees of the Company and its Affiliates.
 - (ii) Without limiting the generality of the Company's obligations under this subsection (e), the Company shall comply with all of its obligations under the benefit plans, practices, policies and programs of the Company and its Affiliates that arise in connection with a Change of Control of the Company, including without limitation all obligations that require the Company to (A) fully vest participants under the Company's qualified or non-qualified retirement plans and (B) extend the benefits described in Section 3.5.
- (f) Office and Support Staff. The Employee shall be entitled to an office or offices of a size and with furnishings and other appointments, and to secretarial and other assistance, commensurate with those provided to his peer employees of the Company and its Affiliates.
- (g) <u>Vacation</u>. The Employee shall be entitled to paid vacation in accordance with the most favorable agreements, plans, policies, programs and practices of the Company and its Affiliates as in effect for the Employee at any time during the one-year period immediately preceding the Change of Control or, if more favorable to the Employee, as in effect generally at any time thereafter with respect to his peer employees of the Company and its Affiliates.
- (h) <u>Indemnification</u>. If, in connection with any agreement related to a transaction that will result in a Change of Control of the Company, an undertaking is made to provide the Board with rights to indemnification from the Company (or from any other party to such agreement), the

Employee shall, by virtue of this Agreement, be entitled to the same rights to indemnification as are provided to the Board pursuant to such agreement. Otherwise, the Employee shall be entitled to indemnification rights on terms no less favorable to the Employee than those available under any Company indemnification agreements or the articles of incorporation, bylaws or resolutions of the Company at any time after the Change of Control to his peer employees of the Company. Such indemnification rights shall be with respect to all claims, actions, suits or proceedings to which the Employee is or is threatened to be made a party that arise out of or are connected to his services at any time prior to the termination of his employment, without regard to whether such claims, actions, suits or proceedings are made, asserted or arise during or after the Employment Term.

(i) <u>Directors and Officers Insurance</u>. If, in connection with any agreement related to a transaction that will result in a Change of Control of the Company, an undertaking is made to provide the Board with continued coverage following the Change of Control under one or more directors and officers liability insurance policies, then the Employee shall, by virtue of this Agreement, be entitled to the same rights to continued coverage under such directors and officers liability insurance policies as are provided to the Board, and the Company shall take any steps necessary to give effect to this provision. Otherwise, the Company shall agree to cover the Employee under any directors and officers liability insurance policies as are provided generally at any time after the Change of Control to his peer employees of the Company.

3.3 Obligations upon Termination after a Change of Control.

- (a) Termination by Company for Reasons other than Death, Disability or Cause or by the Employee for Good Reason. If, after a Change of Control and during the Employment Term, the Company or any of its Affiliates terminates the Employee's employment, as defined in Treasury Regulations 1.409A-1(h)(1) ("Separation from Service"), other than for Cause, death or Disability, or the Employee terminates employment for Good Reason, subject to Section 3.3(d) and Section 3.6, if applicable:
 - (i) the Company shall pay to the Employee in a lump sum in cash within five business days of the date of termination an amount equal to three times the sum of (1) the amount of Base Salary in effect pursuant to Section 3.2(a) hereof at the date of termination plus (1) the average of the annual bonuses paid or to be paid to the Employee with respect to the immediately preceding three fiscal years; *provided*, *however*, that, if the Employee has in effect a deferral election with respect to any percentage of the annual bonus which would otherwise become payable with respect to the fiscal year in which termination occurs, such lump sum payment shall be reduced by an amount equal to such percentage times the bonus component of the lump sum payment (which reduction amount shall be deferred in accordance with such election);
 - (ii) the Company shall pay to the Employee in a lump sum in cash, as promptly as practicable but in no case later than the 15 th day of the third month following the end of the fiscal year of the Company in which the termination occurs, a *pro rata* performance bonus, the amount of which shall be determined by multiplying the annual bonus that the Employee would have earned with respect to the entire fiscal year in which the termination occurs, assuming that the Employee had served for the entire fiscal year and

calculated by the Company in good faith to exclude the effects of the Change of Control on the applicable performance metrics used to calculate such bonus (including without limitation excluding the effects of any non-recurring transaction costs or any changes in overhead, interest, tax, intercompany or other expenses arising out of such transaction), by the fraction obtained by dividing the number of days in such year through the date of termination by 365; *provided, however*, that, if the Employee has in effect a deferral election with respect to any percentage of the annual bonus which would otherwise become payable with respect to the fiscal year in which termination occurs, such lump sum payment shall be reduced by an amount equal to such percentage times the lump sum payment (which reduction amount shall be deferred in accordance with such election); and, *further provided*, that if the bonus performance period in effect at the date of termination is less than a year, then the foregoing paragraph will apply with respect to such shortened period and all references to an annual period or 365 days shall mean the applicable shortened period or shortened number of days to the extent the context requires;

- if, at the date of termination, the Company shall not yet have paid to the Employee (or deferred in (iii) accordance with any effective deferral election by the Employee) an annual bonus with respect to a fully completed fiscal year, the Company shall pay to the Employee in a lump sum in cash within five business days of the date of termination but in no case after the 15 th day of the third month following the end of the fiscal year of the Company in which the termination occurs, an amount determined as follows: (1) if the Board (acting directly or indirectly through any committee or subcommittee) shall have already determined the amount of such annual bonus, such amount shall be paid, and (1) if the Board shall not have already determined the amount of such annual bonus, the amount to be paid shall be the greater of the amount provided under Section 3.2(b) hereof or the annual bonus that the Employee would have earned with respect to such completed fiscal year, based solely upon the actual level of achievement of the objective performance goals established with respect to such bonus and assuming the achievement of 100% of any subjective performance goals or criteria otherwise applicable with respect to such bonus; provided, however, that, if the Employee has in effect a deferral election with respect to any percentage of the annual bonus which would otherwise become payable with respect to such completed fiscal year, such lump sum payment shall be reduced by an amount equal to such percentage times the lump sum payment (which reduction amount shall be deferred in accordance with such election); provided, further, that any payment under this subsection (1) (or any payment under any other provision of this Agreement calculated by reference to prior or target bonus amounts) shall be payable notwithstanding any provision to the contrary set forth in any bonus plan or program of the Company;
- (iv) for a period of three years following the date of termination of employment, or such longer period as may be provided by the terms of the appropriate plan, program, practice or policy (the "Continuation Period"), the Company shall at its expense continue on behalf of the Employee and his dependents and beneficiaries the life insurance, disability, medical, dental and hospitalization benefits (including any benefit under any individual benefit arrangement that covers medical, dental or hospitalization expenses not otherwise covered under any general Company plan) provided (x) to the Employee at any

time during the one-year period prior to the Change in Control or at any time thereafter or (y) to other similarly-situated employees who continue in the employ of the Company or its Affiliates during the Continuation Period. If the Employee is a Specified Employee governed by Section 3.3(d), to the extent that any benefits provided to the Employee under this Section 3.3(a)(iv) are taxable to the Employee, then, with the exception of medical insurance benefits, the value of the aggregate amount of such taxable benefits provided to the Employee pursuant to this Section 3.3(a)(iv) during the six-month period following the date of termination shall be limited to the amount specified by Code Section 402(g)(1)(B) for the year in which the termination occurred. The Employee shall pay the cost of any benefits that exceed the amount specified in the previous sentence during the six-month period following the date of termination, and shall be reimbursed in full by the Company during the seventh month after the date of termination. The coverage and benefits (including deductibles and costs) provided in this Section 3.3(a)(iv) during the Continuation Period shall be no less favorable to the Employee and his dependents and beneficiaries than the most favorable of such coverages and benefits during any of the periods referred to in clauses (x) or (y) above; provided, however, in the event of the Disability of the Employee during the Continuation Period, disability benefits shall, to the maximum extent possible, not be paid for the Continuation Period but shall instead commence immediately following the end of the Continuation Period. The Company's obligation hereunder with respect to the foregoing benefits shall be limited to the extent that the Employee obtains any such benefits pursuant to a subsequent employer's benefit plans, in which case the Company may reduce the coverage of any benefits it is required to provide the Employee hereunder as long as the aggregate coverages and benefits of the combined benefit plans is no less favorable to the Employee than the coverages and benefits required to be provided hereunder. At the end of the Continuation Period, the Employee shall have assigned to him, at no cost and with no apportionment of prepaid premiums, any assignable insurance owned by the Company that relates specifically to the Employee unless such assignment is inconsistent with the terms of any split dollar arrangement with the Employee. The Employee will be eligible for coverage under the Consolidated Omnibus Budget Reconciliation Act at the end of the Continuation Period or earlier cessation of the Company's obligation under the foregoing provisions of this Section 3.3(a)(iv) (or, if the Employee shall not be so eligible for any reason, the Company will provide equivalent coverage);

- (v) the Company at its cost shall provide to the Employee outplacement assistance by a reputable firm specializing in such services for the period beginning with the termination of employment and ending one year later; and
- (vi) the Company shall discharge its obligations under all other applicable sections of this Article III, including Sections 3.4, 3.5, 3.6 and 3.7.

To the extent that the amounts payable under Section 3.3(a)(iv), Section 3.3(a)(v), or Section 3.7 are deemed to be reimbursements and other separation payments under Treasury Regulations Section 1.409A-1(b)(9)(v), they shall not be deemed to provide for the deferral of compensation governed by Code Section 409A. If they do constitute deferral of compensation governed by Code Section 409A, they shall be deemed to be reimbursements or in-kind benefits governed by Treasury

Regulations Section 1.409A-3(i)(1)(iv). If the previous sentence applies, (1) the amount of expenses eligible for reimbursement or in-kind benefits provided during the Employee's taxable year shall not affect the expenses eligible for reimbursement or in-kind benefits in any other taxable year, (1) the reimbursement of an eligible expense must be made on or before the last day of the Employee's taxable year following the taxable year in which the expense was incurred and (1) the right to reimbursement or in-kind benefits shall not be subject to liquidation or exchange for another benefit.

The payments and benefits provided in this Section 3.3(a) and under all of the Company's employee benefit and compensation plans shall be without regard to any plan amendment made after any Change of Control that adversely affects in any manner the computation of payments and benefits due the Employee under such plan or the time or manner of payment of such payments and benefits, excluding plan amendments that the Company is required by law to implement. After a Change of Control no discretionary power of the Board or any committee thereof shall be used in a way (and no ambiguity in any such plan shall be construed in a way) which adversely affects in any manner any right or benefit of the Employee under any such plan.

- (b) <u>Death; Disability; Termination for Cause; or Voluntary Termination</u>. If, after a Change of Control and during the Employeen's status as an employee is terminated (1) by reason of the Employee's death or Disability, (1) by the Company for Cause or (1) voluntarily by the Employee other than for Good Reason, this Agreement shall terminate without further obligation to the Employee or the Employee's legal representatives (other than the timely payment or provision of those already accrued to the Employee, imposed by law or imposed pursuant to employee benefit or compensation plans, programs, practices, policies or agreements maintained by the Company or its Affiliates).
- (c) Notice of Termination. Any termination by the Company for Cause or by reason of the Employee's Disability, or by the Employee for Good Reason, shall be communicated by a Notice of Termination to the other party given in accordance with Section 5.2 of this Agreement. For purposes of this Agreement, a "Notice of Termination" means a written notice which (1) indicates the specific termination provision in this Agreement relied upon, (1) to the extent applicable, sets forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of the Employee's employment under the provision so indicated and (1) if the effective date of the termination is other than the date of receipt of such notice, specifies the termination date (which date shall be not more than 30 days after the giving of such notice), provided that the effective date for any termination by reason of the Employee's Disability shall be the 30th day after the giving of such notice, unless prior to such 30th day the Employee shall have resumed the full-time performance of his duties. The failure by the Employee or the Company to set forth in the Notice of Termination any fact or circumstance which contributes to a showing of Cause, Disability or Good Reason shall not waive any right of the Employee or the Company, respectively, hereunder or preclude the Employee or the Company, respectively, from asserting such fact or circumstance in enforcing the Employee's or the Company's rights hereunder.
- (d) <u>Six-Month Delay for Specified Employees</u>. Notwithstanding any other provision hereof, payments hereunder which constitute deferred compensation under Code Section 409A and the Treasury Regulations thereunder and which are not exempt from coverage

by Code Section 409A and the Treasury Regulations thereunder shall commence, if Employee is then a Specified Employee and payment is triggered by his Separation from Service, on the first day of the seventh month following the date of the Specified Employee's Separation from Service, or, if earlier, the date of death of the Specified Employee. On the first day of such seventh month or on the first day of the month following the earlier death of the Specified Employee, the Specified Employee or his estate or spouse, as the case may be, shall be paid in a lump sum the amount that the Specified Employee would have been paid hereunder over the preceding six months (or, if earlier, the months preceding the date of death) but for the fact that he was a Specified Employee. Nevertheless, for all other purposes of this Agreement, the payments shall be deemed to have commenced on the date they would have had the Employee not been a Specified Employee, and payment of any remaining benefits shall be made as otherwise scheduled hereunder.

- 3.4 Accrued Obligations and Other Benefits. It is the intent of this Agreement that upon termination of employment for any reason following a Change of Control the Employee or his legal representatives be entitled to receive promptly, and in addition to any other benefits specifically provided, (a) the Employee's Base Salary through the date of termination to the extent not theretofore paid, (b) any accrued vacation pay, to the extent not theretofore paid, and (c) any other amounts or benefits required to be paid or provided or which the Employee or his legal representatives are entitled to receive under any plan, program, policy, practice or agreement of the Company.
- 3.5 **Stock Options and Other Incentives.** The foregoing benefits provided for in this Article III are intended to be in addition to the value or benefit of any stock options, restricted stock, restricted stock units, performance shares or similar awards, the exercisability, vesting or payment of which is accelerated or otherwise enhanced upon a Change of Control pursuant to the terms of any stock option, incentive or other similar plan or agreement heretofore or hereafter adopted by the Company or the Post-Transaction Company.

3.6 Conditional Payment Reductions.

(a) Notwithstanding any other contrary provisions in any plan, program or policy of the Company, if all or any portion of the benefits payable under this Agreement, either alone or together with other payments and benefits that the Employee receives or is entitled to receive from the Company, would constitute a "parachute payment" within the meaning of Section 280G of the Code, the Company shall reduce the Employee's payments and benefits payable under this Agreement to the extent necessary so that no portion thereof shall be subject to the excise tax imposed by Section 4999 of the Code, but only if, by reason of such reduction, the net after-tax benefit shall exceed the net after-tax benefit if such reduction were not made. "Net after-tax benefit" for these purposes shall mean (i) the sum of the total amount payable to Employee under the Agreement, plus all other payments and benefits which Employee receives or is then entitled to receive from the Company that, alone or in combination with the payments and benefits payable under the Agreement, would constitute a "parachute payment" within the meaning of Section 280G of the Code (each such benefit hereinafter referred to as an "Additional Parachute Payment"), less (ii) the amount of federal income taxes payable with respect to the foregoing calculated at the maximum marginal income tax rate for each year in which the foregoing shall be paid to the

Employee (based upon the rate in effect for such year as set forth in the Code at the time of the payment under the Agreement), less (iii) the amount of excise taxes imposed with respect to the payments and benefits described in (i) above by Section 4999 of the Code. The parachute payments reduced under this section shall be those that the Employee determines provide the Employee the best economic benefit and, to the extent any parachute payments are economically equivalent with each other, each shall be reduced pro rata; *provided*, *however*, that the Employee may elect to have the non-cash payments and benefits due the Employee reduced or eliminated prior to any reduction of the cash payments due under this Agreement.

- (b) All determinations required to be made under this Section 3.6 shall be made by the accounting firm that was the Company's independent auditor prior to the Change of Control or any other third party mutually acceptable to the Employee and the Company (the "Accounting Firm"). The Accounting Firm shall provide detailed supporting calculations both to the Company and the Employee. All fees and expenses of the Accounting Firm shall be borne solely by the Company. Absent manifest error, any determination by the Accounting Firm shall be binding upon the Company and the Employee.
- (c) For purposes of determining whether and the extent to which any payments would constitute a "parachute payment" (i) no portion of any payments or benefits that the Employee shall have waived at such time and in such manner as not to constitute a "payment" within the meaning of section 280G(b) of the Code shall be taken into account, (ii) no portion of the payments shall be taken into account which, in the opinion of tax counsel ("Tax Counsel") reasonably acceptable to the Employee and selected by the Accounting Firm, does not constitute a "parachute payment" within the meaning of section 280G(b)(2) of the Code (including by reason of section 280G(b)(4)(A) of the Code) and, in calculating the excise tax, no portion of such payments shall be taken into account which, in the opinion of Tax Counsel, constitutes reasonable compensation for services actually rendered, within the meaning of section 280G(b)(4)(B) of the Code, in excess of the "base amount" (within the meaning set forth in section 280G(b)(3) of the Code) allocable to such reasonable compensation, and (iii) the value of any non-cash benefit or any deferred payment or benefit included in the payments shall be determined by the Accounting Firm in accordance with the principles of sections 280G(d)(3) and (4) of the Code.
- 3.7 **Legal Fees.** The Company agrees to reimburse the Employee for all legal fees and other expenses which the Employee may reasonably incur as a result of any contest by the Company or the Post-Transaction Company of the validity or enforceability of, or liability under, any provision of this Agreement, but only if, when and to the extent the Employee prevails with respect to such contest.
- 3.8 **Set-Off; Mitigation.** After a Change of Control, the obligations of the Company and its Affiliates to make the payments provided for in this Agreement and otherwise to perform its obligations hereunder shall not be affected by any set-off, counterclaim, recoupment, defense or other claim, right or action which the Company or its Affiliates may have against the Employee or others, other than as expressly provided to the contrary in Section 3.3(a)(iv), Section 3.10 or Section 4.3. It is the intent of this Agreement that in no event shall the Employee be obligated to seek other

employment or take any other action to mitigate the amounts or benefits payable to the Employee under any of the provisions of this Agreement.

- 3.9 Certain Pre-Change-of-Control Terminations. Notwithstanding any other provision of this Agreement, the Employee's employment shall be deemed to have been terminated following a Change of Control by the Company without Cause (and the Employee shall be entitled to receive all payments and benefits associated therewith) if the Employee's employment is terminated by the Company or any of its Affiliates without Cause prior to a Change of Control (whether or not a Change of Control actually occurs) and such termination (1) was at the request or direction of a third party who has taken steps designed to effect a Change of Control or otherwise arose in connection with or in anticipation of a Change of Control or (1) occurred after discussions with a third party regarding a possible Change of Control transaction commenced and such discussions produced (whether before or after such termination) either a preliminary or definitive agreement with respect to such a transaction or a public announcement of the pending transaction (whether or not a Change of Control actually occurs).
- 3.10 **Other Severance Plans.** If the Employee becomes entitled to receive severance benefits under this Article III, the Company shall not be required to pay the Employee any additional severance payment under any other severance or salary continuation policy, plan, agreement or arrangement maintained by the Company or its Affiliates unless such other policy, plan, agreement or arrangement expressly provides to the contrary.

ARTICLE IV NONDISCLOSURE, NONCOMPETITION AND NONSOLICITATION

4.1 Nondisclosure of Confidential Information . The Employee acknowledges and agrees that in the course of the Employee's employment the Employee has been in a position to have access to and develop Confidential Information, and will continue to be in position to receive and develop Confidential Information during the Employee's tenure as an employee of the Company or any of its Affiliates. As long as the Employee is an employee of the Company or any of its Affiliates, the Employee shall hold in a fiduciary capacity for the benefit of the Company all Confidential Information which the Employee obtained during the Employee's employment (whether prior to or after the Agreement Date) and shall use such Confidential Information solely in the good faith performance of his duties for the Company and its Affiliates. If the employment of the Employee is terminated for any reason, then, commencing with the termination date and continuing until the fifth anniversary of such date, the Employee shall (a) not communicate, divulge or make available to any Person (other than the Company and its Affiliates) any such Confidential Information, except with the prior written consent of the Company or as may be required by law or legal process, and (b) deliver promptly to the Company upon its written request any Confidential Information in his possession, including any duplicates thereof and any notes or other records the Employee has prepared with respect thereto, provided that Employee need not deliver to the Company, and may retain, one copy of any personal diaries, calendars, or personal notes of correspondence. If the provisions of any applicable law or the order of any court would require the Employee to disclose or otherwise make available any Confidential Information to a governmental authority or to any other third party, the Employee shall give the Company, unless it is unlawful to

do so, prompt prior written notice of such required disclosure and an opportunity to contest the requirement of such disclosure or apply for a protective order with respect to such Confidential Information by appropriate proceedings.

Notwithstanding the foregoing, and in accordance with 18 U.S.C. § 1833, nothing herein prohibits Employee (1) from disclosing Confidential Information (a) in confidence to a Federal, State, or local government official, either directly or indirectly, or to an attorney, and (b) solely for the purpose of reporting or investigating a suspected violation of law; or (2) from disclosing Confidential Information in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. Disclosures to attorneys, made under seal, or pursuant to court order are also protected in certain circumstances under 18 U.S.C. § 1833.

- 4.2 **Non-competition; Non-solicitation; Non-disparagement** . (a) The Employee agrees that, during the term of this Agreement and for a period following the termination date of two years if the Employee's employment is terminated by the Company or one of its Affiliates for Cause or by the Employee without Good Reason or one year if the Employee's employment is terminated for any other reason, the Employee will not, directly or indirectly, in any capacity whatsoever, either on the Employee's own behalf or on behalf of any other Person with whom the Employee may be employed or otherwise associated:
 - (i) engage or invest in, own, manage, operate, finance, control, acquire an interest in, be employed by, render services to, act as an agent on behalf of, or otherwise in any way participate in, associate with or allow his skill, knowledge, experience or reputation to be used by (whether as a proprietor, partner, stockholder, member, director, officer, employee, joint venturer, investor, consultant, agent, sales representative, broker or other participant) any Person engaged in or planning to become engaged in the Business within the United States; *provided, however*, that the Employee may own passive investments in not more than 1% of the outstanding securities of any Person engaged in such Business (but without otherwise participating in such similar business) if such securities are registered under Section 12 of the Securities Exchange Act of 1934, as amended;
 - (ii) contact any customer of the Company or its Affiliates to solicit, divert or entice away the business of such customer, or otherwise disrupt the relationship between such customer and the Company or its Affiliates;
 - (iii) solicit, induce, influence or attempt to influence any supplier, lessor, lessee, licensor, partner, joint venturer, potential acquiree or any other person who has a business relationship with the Company or its Affiliates, or who on the termination date is engaged in discussions or negotiations to enter into a business relationship with the Company or its Affiliates, to discontinue, reduce or limit the extent of such relationship with the Company or any of its Affiliates;
 - (iv) make contact with any employee of the Company or its Affiliates for the purpose of soliciting such employee for hire, whether as an employee, independent

contractor, consultant or otherwise, or otherwise disrupting such employee's relationship with the Company or its Affiliates; or

- (v) make any statement or disclose any information to any customers, suppliers, lenders, lessors, licensees, other employees of the Company or its Affiliates or others that is defamatory or derogatory with respect to the business, operations, management or other employees of the Company or its Affiliates, or take any other action (excluding making truthful, non-defamatory statements in good faith that do not violate any other provision of this Agreement) that could reasonably be expected to injure the Company in its business relationships with any of the foregoing parties or result in any other detrimental effect on the Company or its Affiliates.
- (b) The Employee agrees that: (i) the covenants and agreements set forth in this Article IV are reasonable both in scope of geographical area and duration, (ii) the Company would not have entered into this Agreement but for such covenants of the Employee, (iii) such covenants have been made as a result of arm-length bargaining in order to induce the Company to enter into this Agreement, and (iv) such covenants and agreements are reasonable and necessary for the protection of the Confidential Information, assets, goodwill and business of the Company. Employee further agrees that the phrase "any Person engaged in or planning to become engaged in the Business within the United States," as used in Section 4.2(a)(i) above, shall include AT&T Inc., Cincinnati Bell Inc., Comcast Corporation, Consolidated Communications Holdings Inc., Crown Castle International Corp., Dish Network Corp., EchoStar Corporation, Frontier Communications Corp., IDT Corp., Liberty Global plc, Mitel Networks Corp., Telephone & Data Systems Inc., TELUS Corp., Viacom, Inc., Viasat, Inc., Zayo Group Holdings, Inc., XO Holdings, Inc., Sprint Nextel Corporation, tw telecom inc., Verizon Communications Inc., Limelight Networks, Inc., Akamai Technologies Inc., Windstream Corporation, Reliance Communications Venture Limited, including in each case their affiliates, successors, and assigns. To the extent permitted by applicable law, the Employee covenants and agrees not to institute, maintain, prosecute or in any way aid in the institution, maintenance or prosecution of any lawsuit, action, claim, arbitration or other proceeding against the Company or any of its Affiliates with respect to the enforceability of the covenants contained in this Article IV and the Employee hereby irrevocably waives all defenses otherwise available to the Employee with respect to the strict enforcement of such covenants and agreements by the Company.
- 4.3 **Injunctive Relief; Forfeiture of Future Payments and Benefits; Other Remedies**. The Employee acknowledges that a breach by the Employee of Sections 4.1 or 4.2 herein would cause immediate and irreparable harm to the Company for which an adequate monetary remedy does not exist; hence, the Employee agrees that, in the event of a breach or threatened breach by the Employee of the provisions of Sections 4.1 or 4.2 herein during or after the effective date of the Employee's termination, the Company shall be entitled to injunctive relief restraining the Employee from such violation without the necessity of proof of actual damage or the posting of any bond, except as required by non-waivable, applicable law. Nothing herein, however, shall be construed as prohibiting the Company from pursuing any other remedy at law or in equity to which the Company may be entitled under applicable law in the event of a breach or threatened breach of this Agreement by the Employee, including without limitation the recovery of damages, costs or expenses, such as reasonable attorneys' fees, incurred by the Company as a result of any such breach or threatened breach. In addition to the foregoing remedies, the Company shall have the right upon

the occurrence of any breach of any nondisclosure, noncompetition or nonsolicitation covenant contained in this Article IV, to cancel any unpaid severance payments, salary, bonus, commissions or reimbursements otherwise outstanding at the termination date, including the suspension, reduction or elimination of payments and benefits under Article III. The Employee acknowledges that any such suspension, reduction or elimination of payments would not constitute, and should not be characterized as, liquidated damages.

4.4 **Governing Law of this Article IV; Consent to Jurisdiction**. Any dispute regarding the reasonableness of the covenants and agreements set forth in this Article IV or the territorial scope or duration thereof or the remedies available to the Company upon any breach of such covenants and agreements, shall be governed by and interpreted in accordance with the laws of the State of Colorado. The parties mutually agree that any such disputes shall be resolved in the United States District Court for the District of Colorado or the state District Court for the 2 nd Judicial District, Denver County, Colorado. The parties agree that it is their mutual intent that the provisions of this Agreement be enforced to the fullest extent permitted under applicable law, whether now or hereafter in effect, and, to the extent permitted by applicable law, the parties waive any provision of applicable law that would render any provision of Article IV invalid or unenforceable.

ARTICLE V MISCELLANEOUS

5.1 Binding Effect; Successors.

- (a) This Agreement shall be binding upon and inure to the benefit of the Company and its successors or assigns, but the Company may assign this Agreement only (i) to an Affiliate or (ii) pursuant to a merger or consolidation in which the Company is not the continuing entity, or the sale or liquidation of all or substantially all of the assets of the Company, provided that the assignee or transferee is the successor to all or substantially all of the assets of the Company and such assignee or transferee assumes the liabilities, obligations and duties of the Company under this Agreement, either contractually on the terms specified below or as a matter of law.
- (b) This Agreement is personal to the Employee and shall not be assignable by the Employee without the consent of the Company (there being no obligation to give such consent) other than such rights or benefits as are transferred by will or the laws of descent and distribution, which shall inure to the benefit of the Employee's legal representatives.
- (c) The Company shall require any successor to or assignee of (whether direct or indirect, by purchase, share exchange, merger, consolidation or otherwise) all or substantially all of the assets or businesses of the Company (1) to assume unconditionally and expressly this Agreement and (1) to agree to perform or to cause to be performed all of the obligations under this Agreement in the same manner and to the same extent as would have been required of the Company had no assignment or succession occurred.
- (d) The Company shall also require all entities that control or that after the transaction will control (directly or indirectly) the Company or any such successor or assignee to agree to cause to be performed all of the obligations under this Agreement.

- (e) The obligations of the Company and the Employee which by their nature may require either partial or total performance after the expiration of the term of the Agreement shall survive such expiration.
- 5.2 **Notices.** All notices hereunder must be in writing and shall be deemed to have been given upon receipt of delivery by: (1) hand (against a receipt therefor), (1) certified or registered mail, postage prepaid, return receipt requested, (1) a nationally recognized overnight courier service or (1) telecopy transmission with confirmation of delivery. All such notices must be addressed as follows:

If to the Company, to:

CenturyLink, Inc. 100 CenturyLink Drive Monroe, Louisiana 71203 Attn: General Counsel

If to the Employee, to:

Jeffrey K. Storey c/o CenturyLink, Inc. 1025 Eldorado Boulevard Building 2000 Broomfield, Colorado 80021

or such other address as to which any party hereto may have notified the other in writing.

- 5.3 **Governing Law.** This Agreement shall be construed and enforced in accordance with and governed by the internal laws of the State of Colorado without regard to principles of conflict of laws.
- 5.4 **Withholding.** The Employee agrees that the Company has the right to withhold, from the amounts payable pursuant to this Agreement, all amounts required to be withheld under applicable income or employment tax laws, or as otherwise stated in documents granting rights that are affected by this Agreement.
- 5.5 Amendment and Compliance with Law. No provision of this Agreement may be modified or amended except by an instrument in writing signed by both parties. Notwithstanding any other provision of this Agreement, it is the intention of the parties to this Agreement that no payment or entitlement pursuant to this Agreement will give rise to any adverse tax consequences to the Employee under Code Section 409A and Treasury Regulations and other interpretive guidance issued thereunder, including those issued after the date hereof (collectively, "Section 409A"). This Agreement and any amendments hereto shall be interpreted and administered to that end and (i) to the maximum extent permitted by law, no effect shall be given to any provision herein, any

amendment hereto or any action taken hereunder in a manner that reasonably could be expected to give rise to adverse tax consequences under Section 409A and (i) the parties shall take any corrective action reasonably within their control that are necessary to avoid such adverse tax consequences.

- 5.6 **Severability.** If any term or provision of this Agreement, or the application thereof to any person or circumstance, shall at any time or to any extent be invalid, illegal or unenforceable in any respect as written, the Employee and the Company intend for any court construing this Agreement to modify or limit such provision so as to render it valid and enforceable to the fullest extent allowed by law. Any such provision that is not susceptible of such reformation shall be ignored so as to not affect any other term or provision hereof, and the remainder of this Agreement, or the application of such term or provision to persons or circumstances other than those as to which it is held invalid, illegal or unenforceable, shall not be affected thereby and shall be valid and enforced to the fullest extent permitted by law.
- 5.7 **Waiver of Breach.** Except as expressly provided herein to the contrary, the failure by any party to enforce any of its rights hereunder shall not be deemed to be a waiver of such rights, unless such waiver is an express written waiver. The waiver by either party of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach thereof.
- 5.8 **Remedies Not Exclusive.** No remedy specified herein shall be deemed to be such party's exclusive remedy, and accordingly, in addition to all of the rights and remedies provided for in this Agreement, the parties shall have all other rights and remedies provided to them by applicable law, rule or regulation, including without limitation the right to claim interest with respect to any payment not timely made hereunder.
- 5.9 **Company's Reservation of Rights.** The Employee acknowledges and understands that (1) the Employee is employed at will by either the Company or one of its Affiliates (the "Employer"), (1) the Employee serves at the pleasure of the board of directors of the Employer, and (1) the Employer has the right at any time to terminate the Employee's status as an employee, or to change or diminish his status during the Employment Term, subject to the rights of the Employee to claim the benefits conferred by this Agreement. Notwithstanding any other provisions of this Agreement to the contrary, this Agreement shall not entitle the Employee or his legal representatives to any severance or other benefits of any kind prior to a Change of Control or to any such benefits if Employee is not employed by the Company or one of its Affiliates on the date of a Change of Control, except in each case for those rights afforded under Section 3.9.
- 5.10 **Non-exclusivity of Rights.** Subject to Section 5.9, nothing in this Agreement shall prevent or limit the Employee's continuing or future participation in any plan, program, policy or practice provided by the Company or any of its Affiliates and for which the Employee may qualify, nor shall anything herein limit or otherwise restrict such rights as the Employee may have under any contract or agreement with the Company or any of its Affiliates.
- 5.11 **Demand for Benefits.** Unless otherwise provided herein, the payment or payments due hereunder shall be paid to the Employee without the need for demand, and to a beneficiary upon the receipt of the beneficiary's address and social security number. In all such cases, the

Employee or beneficiary shall provide all required tax withholding information or forms upon the Company's request. Nevertheless, the Employee or a Person claiming to be a beneficiary who claims entitlement to a benefit can file a claim for benefits hereunder with the Company. Unless otherwise provided herein, the Company shall accept or reject the claim within ten business days of its receipt. If the claim is denied, the Company shall give the reason for denial in a written notice that refers to the provision of this Agreement that forms the basis of the denial. If any additional information or material is necessary to perfect the claim, the Company will identify these items in writing and explain why such additional information is necessary.

- 5.12 **Authority.** The Company represents and warrants that (1) its execution and delivery of this Agreement has been duly authorized by the Board and (1) no other corporate proceedings are necessary to authorize the Company's execution, delivery and performance of this Agreement.
- 5.13 **Counterparts.** This Agreement may be executed in one or more counterparts, each of which shall be deemed to be an original but all of which together shall constitute one and the same instrument.
- 5.14 **Interpretation.** Any reference to any section of the Code or the Treasury Regulations shall be deemed to also refer to any successor provisions thereto.
- 5.15 **The Employee's Acknowledgment**. The Employee represents to the Company that he has read and understands, and agrees to be bound by, each of the terms of this Agreement, including Article IV.

IN WITNESS WHEREOF, the Company and the Employee have caused this Change of Control Agreement to be executed as of the Agreement Date.

CENTURYLINK, INC.

By: /s/ Stacey W. Goff
Stacey W. Goff

Executive Vice President and General Counsel

EMPLOYEE:

/s/ Jeffrey K. Storey
Jeffrey K. Storey

{N3633038.2}

CENTURYLINK EXECUTIVE SEVERANCE PLAN

AS UPDATED, AMENDED AND RESTATED EFFECTIVE OCTOBER 10, 2017

CENTURYLINK EXECUTIVE SEVERANCE PLAN As Amended and Restated Effective August 20, 2012

TABLE OF CONTENTS

Page

| ARTICLE I | | PURPOSE OF PLAN | 1 |
|------------|------------------------------------|-------------------------|----------|
| | Section 1.01 | Purpose of the Plan | 1 |
| | Section 1.02 | Application of the Plan | 1 |
| | Section 1.03 | Gender and Number | 1 |
| ARTICLE II | | DEFINITIONS | 1 |
| | Base Salary | | 1 |
| | Board | | 1 |
| | Cause | | 2 |
| | CEO | | 2 |
| | Change in Control | | 2 |
| | CIC Good Reason Resign | nation | 3 |
| | CIC Termination | | 4 |
| | COBRA | | 4 |
| | Code | | 4 |
| | Company | | 4 |
| | Comparable Position | | 4 |
| | Competitive Employment | t | 5 |
| | Competitor | | 5 |
| | Effective Date | | 5 |
| | Eligible Employee | | 6 |
| | Employee | | 7 |
| | ERISA | | 7 |
| | Executive Officer | | 7 |
| | Involuntary Termination | | 7 |
| | Non CIC Good Reason R | esignation | 7 |
| | Non-CIC Termination | | 8 |
| | Non-Comparable Position | n | 8 |
| | Non-Compete Period | | 9 |
| | Participant | | 9 |
| | Participating Company | | 9 |
| | Permanent Disability Plan | | 9 |
| | Plan Administrator | | 9 |
| | Release | | 9 |
| | Service | | 10 10 |
| | Service Severance Benefit | | 10 |
| | Severance Benefit Severance Period | | 10 |
| | Specified Employee | | 10 |
| | Subsidiary | | 10 |
| | Termination or Terminate | ad. | 11 |

TABLE OF CONTENTS

(continued)

Page

| | Termination Date | | 11 |
|--------------|------------------------|---|-----|
| | Total Targeted Cash Co | ompensation | 11 |
| | Voluntary Resignation | | 11 |
| | Year of Service | | 11 |
| ARTICLE III | | ELIGIBILITY FOR BENEFITS | 11 |
| | Section 3.01 | Conditions | 11 |
| ARTICLE IV | | DETERMINATION OF SEVERANCE BENEFITS | 12 |
| | Section 4.01 | Severance Benefits - Non-CIC Termination | 12 |
| | Section 4.01A | Lower Performer Severance Benefit | 12 |
| | Section 4.02 | Severance Benefits - CIC Termination | 13 |
| | | Severance Benefits - NON-CIC Terminations or CIC | |
| | Section 4.03 | Terminations | 13 |
| | g .: 101 | Voluntary Resignation; Termination for Death or Permanent | 1.4 |
| | Section 4.04 | Disability | 14 |
| | Section 4.05 | Termination for Cause | 15 |
| | Section 4.06 | Deductions from Severance Benefits | 15 |
| | Section 4.07 | Repayment of Severance Benefits upon subsequent rehire | 15 |
| ARTICLE V | | TIMING AND METHOD OF PAYMENT AND LIMITATION ON BENEFITS | 15 |
| | Section 5.01 | Timing and Method of Payment | 15 |
| | Section 5.02 | Limitation on benefits | 18 |
| ARTICLE VI | | RESTRICTIVE COVENANTS | 19 |
| | Section 6.01 | Code of Conduct | 19 |
| | Section 6.02 | Proprietary Information | 19 |
| | Section 6.03 | Non-Competition | 19 |
| | Section 6.04 | Inducement of Employees, Customers and Others | 20 |
| | Section 6.05 | No Adverse Actions | 20 |
| | Section 6.06 | Return of Property | 20 |
| | Section 6.07 | Non-Disparagement | 20 |
| | Section 6.08 | Assistance with Claims | 20 |
| | Section 6.09 | Reasonableness | 21 |
| | Section 6.10 | Equitable Relief | 21 |
| | Section 6.11 | Survival of Provisions | 21 |
| ARTICLE VII | | PLAN ADMINISTRATOR | 22 |
| | Section 7.01 | Authority and Duties | 22 |
| | Section 7.02 | Compensation of the Plan Administrator | 22 |
| | Section 7.03 | Records, Reporting and Disclosure | 22 |
| | Section 7.04 | Discretion | 22 |
| | Section 7.05 | Implementation of the Plan is Non-Fiduciary | 23 |
| ARTICLE VIII | | AMENDMENT, SUSPENSION AND TEMERINATION | 24 |
| | Section 8.01 | Amendment, Suspension and Termination | 24 |
| | Section 8.02 | Continuation of Plan | 24 |

TABLE OF CONTENTS

(continued)

Page

| EXHIBIT E | | LEVEL 3 COMMUNICAITONS, INC. KEY EMPLOYEE SEVERANCE PLAN (KESP) | |
|-------------|---------------|---|----|
| EXHIBIT D | | PERIOD OF SUBSIDIZED COBRA COVERAGE FOR DIRECTORS/VICE PRESIDENTS | |
| EXHIBIT C | | LIST OF PARTICIPATING COMPANIES | |
| EXHIBIT B | | ARBITRATION PROVISION | |
| EXHIBIT A | | FORM OF RELEASE AGREEMENT SUBJECT TO CHANGE BY THE PLAN ADMINISTRATOR FROM TIME TO TIME | |
| | Section 10.12 | Clerical Error | 29 |
| | Section 10.11 | Controlling Law | 29 |
| | Section 10.10 | Lost Payees | 28 |
| | Section 10.09 | Payments to Incompetent Persons | 28 |
| | Section 10.08 | Unfunded Plan | 28 |
| | Section 10.07 | Severability of Provisions | 28 |
| | Section 10.06 | No Contract of Employment | 28 |
| | Section 10.05 | No Mitigation | 27 |
| | Section 10.04 | Notices | 28 |
| | Section 10.03 | Non-Alienation of benefits | 27 |
| | Section 10.02 | Forum Selection | 27 |
| | Section 10.01 | Waiver of Jury Trial | 26 |
| ARTICLE X | | MISCELLANEOUS | 26 |
| | Section 9.04 | Deadline to Bring a Plan Claim, Arbitration of Civil Suit | 26 |
| | Section 9.03 | Appeals of Denied Administrative Claims | 25 |
| | Section 9.02 | Initial Claim | 25 |
| THETTELL IT | Section 9.01 | Claims | 24 |
| ARTICLE IX | | CLAIMS PROCEDURES | 24 |

Article I PURPOSE OF PLAN

Section 1.01 Purpose of the Plan . The Plan is not intended to be an "employee pension benefit plan" or "pension plan" within the meaning of section 3(2) of ERISA. Rather, the Plan is intended to be a "welfare benefit plan" within the meaning of section 3(1) of ERISA and to meet the descriptive requirements of a plan constituting a "severance pay plan" within the meaning of regulations published by the Secretary of Labor at 29 CFR § 2510.3-2(b). No employee contributions are required or permitted. Accordingly, no employee has or shall have a vested right to such benefits.

Application of the Plan . The Plan shall generally be applicable to an Involuntary Termination, a Non-CIC Good Reason Resignation or a CIC Good Reason Resignation, as defined herein and except as described in Exhibits C and E. This Plan supersedes any other plan, policy, program, or arrangement providing an Eligible Employee with severance or related benefits, with the exception of (i) any vested benefits under another severance or change of control plan, policy, or arrangement which became effective prior to the Effective Date, and (ii) any contract between CenturyLink (including its predecessors) and an Eligible Employee concerning the provision of Severance Benefits following a termination of employment. The Plan applies to all Eligible Employees who are terminated due to one of the reasons listed in Article IV of this Plan. The Effective Date for the implementation of this Restatement is October 10, 2017, and it applies based on the date an Eligible Employee is notified of termination, an offer for a Non-Comparable Position or a Non-CIC Good Reason Trigger Event, and not the actual date of termination from employment with the Company (e.g., an employee notified on March 27, 2016 with a last day worked of April 15, 2016, is eligible for Benefits under the formula stated in the July 1, 2014 Plan Restatement rather than the formula stated in this April 1, 2016 Restatement); however, some provisions within Exhibit C and Exhibit E are expressly contingent upon the closing of the LVLT Corporate Transaction (as that term is defined below), and, in the event the LVLT Corporate Transaction does not occur, such provisions are null and void. Notwithstanding any other provision hereof, there is intended to be no duplication of Severance Benefits to an Eligible Employee. As such, if an Eligible Employee has any vested rights under another severance or change of control plan, policy, arrangement or contract for severance benefits following termination of employment, this Plan shall not apply to such Eligible Employee with respect to such termination.

Section 1.02 Gender and Number . Whenever used herein, the masculine pronoun shall include the feminine and the singular shall encompass the plural, and vice versa.

Article II DEFINITIONS

The following definitions apply except as expressly provided in Exhibits C or E.

"Base Salary" means the Eligible Employee's annual rate of base pay, as reflected in the Company's payroll or personnel files, in effect on a specified date.

- "Board" means the Board of Directors of CenturyLink, Inc., or any successor thereto, or its delegate, including but not limited to, a committee thereof specifically designated for purposes of making determinations hereunder.
- "Cause" means an Eligible Employee's (i) failure to substantially comply with supervisor directives, including directives related to job performance (ii) misconduct, (iii) conduct that violates the Company's ethics and compliance program, including its Code of Conduct, (iv) conduct that is injurious to the Company's reputation, customer relationships, employees or finances, (v) act(s) that, if proven in a court of law, would constitute a felony crime, (vi) act(s) of dishonesty, fraud or moral turpitude, or (vii) violation of any of the restrictive covenants found in Article VI. The Plan Administrator, in its sole discretion, shall determine whether the elements of Cause have been met, including whether the facts and circumstances indicate that the Employee's actions constitute a violation of the Company's ethics and compliance program and/or are injurious to the Company.
- "CEO" means the Executive Officer who is designated by the Board as the Chief Executive Officer of the Company.
- "Change in Control" means any of the following events:
- (a) the acquisition, directly or indirectly, by any "person" or "group" (as those terms are defined in sections 3(a)(9), 13(d), and 14(d) of the Exchange Act and the rules thereunder, including Rule 13d-5(b)) of "beneficial ownership" (as determined pursuant to Rule 13d-3 under the Exchange Act) of securities entitled to vote generally in the election of directors ('voting securities") of the Company that represent 30% or more of the combined voting power of the Company's then outstanding voting securities, other than:
- (i) an acquisition by a trustee or other fiduciary holding securities under any employee benefit plan (or related trust) sponsored or maintained by the Company or any person controlled by the Company or by any employee benefit plan (or related trust) sponsored or maintained by the Company or any person controlled by the Company, or
- (ii) an acquisition of voting securities by the Company or a corporation owned, directly or indirectly, by the stockholders of the Company of at least 50% of the voting power of the Company's then outstanding securities in substantially the same proportions as their ownership of stock of the Company, or
- (iii) an acquisition of voting securities pursuant to a transaction described in subsection (c) below that would not be a Change in Control under subsection (c) below;
- (b) a change in the composition of the Board that causes less than a majority of the directors of the Company to be directors that meet one or more of the following descriptions:
- (i) a director who has been a director of the Company for a continuous period of at last 24 months (or, if less, since the date the shares of Company common stock were listed on the New York Stock Exchange) or,

- (ii) a director whose election or nomination as a director was approved by a vote of at least two-thirds of the then directors described in subsections 2.04(b)(i), (ii) or (iii) by prior nomination or election, but excluding, for the purpose of this subsection (ii), any director whose initial assumption to office occurred as a result of an actual or threatened (y) election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a person or group other than the Board or (z) tender offer, merger, sale of substantially all of the Company's assets, consolidation, reorganization or business combination that would be a Change in Control under subsection (c) below on consummation thereof, or
- (iii) who were serving on the Board as result of the consummation of a transaction described in subsection (c) below that would not be a Change in Control under subsection (c) below;
- (c) the consummation by the Company (whether directly involving the Company or indirectly involving the Company through one or more intermediaries) of
 - (i) a consolidation, merger, reorganization or business combination or
 - (ii) a sale or disposition of all or substantially all of the Company's assets or
 - (iii) the acquisition of assets or stock of another entity,

in each case, other than in a transaction, (x) that results in the Company's voting securities outstanding immediately before the transaction continuing to represent (either by remaining outstanding or by being converted into voting securities of the Company or the person that, as a result of the transaction, controls, directly or indirectly, the Company or owns, directly or indirectly, all or substantially all of the Company's assets or otherwise succeeds to the business of the Company (the Company or such person, the "Successor Entity")) directly or indirectly, at least 50% of the combined voting power of the Successor Entity's outstanding voting securities immediately after the transaction and (y) after which more than 50% of the members of the Board of the Successor Entity were members of the Board at the time of the Board's approval of the transaction or other action of the Board approving the transaction (or whose election or nomination was approved by a vote of at least two-thirds of the members who were members of the Board at that time), and (z) after which no person or group beneficially owns voting securities representing 30% or more of the combined voting power of the Successor Entity; provided, however, no person or group shall be treated for purposes of this subsection (z) as beneficially owning 30% or more of combined voting power of the Successor Entity solely as a result of the voting power held in the Company before the consummation of the transaction; or

- (d) a liquidation or dissolution of the Company other than in connection with a transaction described in subsection (c) above that would not be a Change in Control thereunder.
- "CIC Good Reason Resignation" means an Eligible Employee's written resignation that has been delivered to the Company within 30 calendar days of the notification to the Employee of any of the following events (each a "Trigger Event") related to a Change in Control:

- (a) a reduction of more than 10% in the Eligible Employee's Base Salary in effect immediately prior to the Change in Control.
- (b) a demotion that results in both (i) the movement of an Eligible Employee from one job level to a lower job level (that is, Senior Vice-President to Vice-President or below, Vice-President to Director or below, or Director to Manager or below) and (ii) a material reduction or diminution of authority, duties, responsibilities, and/or budget or fiscal authority. "Demotion" excludes any move from one job to another initiated by the Employee.
- (c) A relocation of the Eligible Employee's principal work location such that (i) the driving distance from the work location of the new position to the current position held immediately prior to the change in job location is more than 50 miles and (ii) the increase in commuting distance from the employee's current home location to his/her work location (comparing current position location against the new position location) is more than 50 miles.
 - ♦If an Employee's current position or new position is classified as "work at home" (i.e., with no assigned office at a CenturyLink facility), his/her home address is or, as applicable, will be deemed his/her work location with respect to such work at home position.

In order to constitute a CIC Good Reason Resignation, one of the above three Trigger Events must occur within one year following a Change in Control.

Notwithstanding the foregoing, the written resignation shall not constitute a "CIC Good Reason Resignation" if the Company fully corrects the circumstances which gave rise to Eligible Employee's claim that there has been a CIC Good Reason Resignation within 30 days following receipt of written notice from the Eligible Employee (the "30-day Cure Period"). If the Company does not fully correct the circumstances that gave rise to the CIC Good Reason Resignation within the 30-calendar day Cure Period, the Eligible Employee's employment shall be deemed to be a CIC Good Reason Resignation and to have Terminated no later than two years following the expiration of the 30-calendar day Cure Period or the Company's notice that it does not intend to cure, whichever is earlier.

- "CIC Termination" means an Eligible Employee's Involuntary Termination within one year after the date of a Change in Control or his CiC-Good Reason Resignation.
- "COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, as codified in section 4980B of the Code, and Part 6 of Title I of ERISA and as amended by subsequent legislation.
- "Code" means the Internal Revenue Code of 1986, as amended, and regulations thereunder.
- "Company" means CenturyLink, Inc. and, solely CenturyLink, Inc's U.S.-based affiliates and subsidiaries, as well as any predecessor (specifically including but not limited to U.S.-based affiliates and subsidiaries of EMBARQ Corporation, Qwest Communications International, Inc., etc.) that are Participating Companies as defined in the Plan and on Exhibit C; non-participating Companies are also separately identified on Exhibit C.

- "Comparable Position" means a position offered by the Company or by another entity (e.g., one providing services to or purchasing assets from the Company) in connection with a reduction in headcount, an outsourcing to a vendor, a reorganization impacting two or more distinct business groups, a sale of distinct business groups, or a merger or acquisition other than a Change in Control, <u>and</u> that
- (i) involve a relocation of 50 miles or less such that either (i) the driving distance from the work location of the new position to the current position held immediately prior to the change in job location is 50 miles or less or (ii) the increase in commuting distance from the employee's current home location to his/her work location (comparing current position location against the new position location) is 50 miles or less.
 - If an Employee's current position or new position is classified as "work at home" (i.e., with no assigned office at a CenturyLink facility), his/her home address is or, as applicable, will be deemed his/her work location with respect to such work at home position.

and

- (ii) has a Total Targeted Compensation (Base Salary plus target short term incentive or sales incentive opportunity) that is at least 90% of the Participant's current Total Targeted Compensation. A job at job grades P6 and below (or their equivalent) is not a "Comparable Position."
- "Competitive Employment" means the direct or indirect performance of duties or responsibilities (whether paid or unpaid and whether as a consultant, employee or otherwise) for a Competitor, including, without limitation, the ownership of any interest in, the provision of any financing, management or advisory services to, any connection with or being a principal, partner or agent of, any Competitor; provided that the Eligible Employee may passively own less than 1% of the outstanding shares of any Competitor.
- "Competitor" means any person, firm, company, corporation or other entity, that markets, sells, resells or otherwise engages in the business of selling to any business or government customer any product or service that is competitive, in any territory or location where the Company does business, with any of the following services, which may be changed from time to time and over time: local exchange telephone service (including services provided by wireless and cable providers), E-911 service, long distance services, dial-up and high-speed Internet access, security monitoring and related services, video delivery, web development, web hosting, high speed data services, wireless communications, IPTV and other video services (including satellite or cable television services), data hosting, cloud computing, any other product or service provided to customers of the Company which the Employee sold or supported, and any other products and services ancillary or related thereto. The burden is on the Eligible Employee to demonstrate that such person is not a Competitor.
- "Effective Date" means October 10, 2017, when this Restatement becomes effective. The Plan applies to all Eligible Employees of a Participating Company who are terminated due to one of the reasons listed in Article IV of this Plan; however, some provisions within Exhibit C are expressly contingent upon the LVLT Corporate Transaction Closing, and, in the event the LVLT Corporate Transaction does not occur, such provisions are null and void. The Effective Date applies based

on the date an Eligible Employee is notified of termination, the offer of a Non-Comparable Position or the Non-CIC Good Reason Resignation, and not the actual date of termination from employment with the Company (*e.g.*, an employee notified on September 20, 2017 with a last day worked of October 15, 2017, is eligible for Benefits under the formula stated in the April 1, 2016 Plan Restatement rather than the formula stated in this September 28, 2017 Restatement, however all other terms and conditions of this Plan apply).

- "Eligible Employee" means an Employee who holds, as of his Termination Date, a Director or Vice-President level job in grades P7 or above or E1 or above, or equivalent grades if a different job grading system is used, as reflected in the Company's personnel records. "Director or Vice-President level" jobs include those job titles which were classified as Director (M6 or M7) or Vice-President (E1 or above) jobs as of January 1, 2016. A Legacy Level 3 Employee will become an Eligible Employee upon the closing the LVLT Corporate Transaction. A Legacy Level 3 SVP will become an Eligible Employee (and eligible for benefits under Sections 4.01, 4.01A, 4.02 and 4.03) immediately after he or she ceases to participate in the Level 3 Communications Key Employee Severance Plan (KESP), provided he or she continues in employment with a Participating Company. An Employee who is terminated for job performance is an Eligible Employee, but solely for limited Severance Benefits as set for in Section 4.01A. An otherwise "Eligible Employee" is not eligible for: (a) benefits under this Plan relating to a Non-CiC Termination, including Sections 4.01 and 4.03 ("Non-CiC Severance Benefits"), if he has signed and not revoked a binding contract with CenturyLink (including its predecessors) concerning the provision of benefits similar to Non-CiC Severance Benefits to that Employee following a termination of employment and (b) benefits under this Plan relating to a CiC Termination, including Sections 4.02 and 4.03 ("CiC Severance Benefits "), if he has signed and not revoked a binding contract with CenturyLink (including its predecessors) concerning the provision of benefits similar to CiC Severance Benefits to that Employee following a termination of employment. If there is any question as to whether an Employee is an Eligible Employee for purposes of the Plan, the Administrator shall make the determination. The term "Eligible Employee" also excludes an Employee who signed a waiver of his rights under this Plan subsequent to the Effective Date. Additionally, an Employee is not an "Eligible Employee" if:
- (1) he is covered by the provisions of a collective bargaining agreement which address the same issues as this Plan, in which case the collective bargaining agreement will prevail and this Plan shall not apply;
- (2) he is covered by the provisions of the CenturyLink Reduction In Force Policy or an individual employee agreement which address the same issues as this Plan, in which case the CenturyLink Reduction In Force Plan or such individual agreement will prevail and this Plan shall not apply;
- (3) the provisions of any other reduction in force or severance plan or policy apply to any group of employees of the Company (such as employees who join the Company via merger or acquisition who are covered by a reduction in force or severance plan or policy that remains in effect after the acquisition; employees of a particular subsidiary or division; or employees at a particular location), in which case the terms of such plan or policy shall apply and this Plan shall not apply;

- (4) he is a Temporary Employee, which term is defined to mean all employees of a Participating Company who are hired to perform an assignment which is not intended to be ongoing and which is intended to have a specified end date (by reference to a calendar end date or a project end date) and/or who are classified on the payroll and/or HR systems as "temporary;"
- (5) his employment with the Company is Terminated voluntarily, such as by resignation (whether or not the employee claims a constructive discharge or forced resignation) or retirement (whether or not such voluntary Termination may have been influenced by pending or threatened layoffs), or Terminated involuntarily due to attendance (other than absences that are protected by applicable law) or misconduct, including but not limited to violation of the Code of Conduct or other CenturyLink policies;
- (6) he is offered a Comparable Position or he is offered and accepts a position with CenturyLink, a vendor to which CenturyLink outsourced the employee's work, a buyer to which CenturyLink has sold the business, or any successor to CenturyLink, even if such position is a Non-Comparable Position;
- (7) he is not classified in CenturyLink's payroll and/or HR systems as an "employee," regardless of whether such person is determined by any court, governmental agency or other entity to be an employee under any federal, state or local law, regulation or rule;
- (8) he is employed by a non-U.S. subsidiary or affiliate, or he/she has been transferred from a non-U.S. subsidiary or affiliate to U.S. subsidiary or affiliate for a finite period of time pursuant to a letter of assignment; or is employed by a non-Participating Company identified on Exhibit C; or
- (9) he is not employed by a Participating Company, or the participation of the Participating Company is limited by the Plan such that the employee is not eligible, as described in Exhibit C; or
- (10) he is considered classified on the payroll and/or HR systems as a CenturyLink Government Services Modified Benefits employee or is categorized on the payroll and / or HRIS systems as a Project Based Employee.
- "Employee" means an individual employed by a Participating Company within the United States as an employee, and excludes any person working for the Company through a temporary service or on a leased basis or who is hired by the Company as an independent contractor, consultant, or otherwise, and excludes a person who is not an employee, or not treated as such, for purposes of withholding federal employment taxes, as evidenced by payroll records or a written agreement with the individual, regardless of any contrary governmental or judicial determination or holding relating to such status or tax withholding.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and regulations thereunder.

- **"Executive Officer"** means an Eligible Employee who is (i) a Vice-President, including Senior Vice-President and Executive Vice-President, reporting directly to the CEO, (ii) an Officer, other than the CEO, designated on an annual or other periodic basis as a Section 16 Officer of the Company, or (iii) an Officer, other than the CEO, designated by the Compensation Committee of the Board as a Senior Officer (as such term is defined in the Compensation Committee's charter).
- "Involuntary Termination" means a Termination of the Eligible Employee's employment, initiated by the Company for any reason other than Cause, Permanent Disability or death.
- "Legacy Level 3 Employee" means any person who (i) is employed by Level 3 Communications, Inc., or any affiliate of it at the Vice-President level, other than any person covered by the Level 3 Key Executive Severance Plan (the "KESP" at Exhibit E), immediately prior to the closing of the LVLT Corporate Transaction and (ii) remains employed by Level 3 Communications, Inc., or any affiliate of it immediately after the closing of the LVLT Corporate Transaction.
- "Legacy Level 3 SVP" means person who (i) is, immediately prior to the closing of the LVLT Corporate Transaction, employed by Level 3 Communications, Inc., or any affiliate of it at the Senior Vice-President level or above and is covered by the Level 3 Key Executive Severance Plan (the "KESP" at Exhibit E) and (ii) remains employed by Level 3 Communications, Inc., or any affiliate of it immediately following the closing of the LVLT Corporate Transaction.
- **"LVLT Corporate Transaction"** means the corporate combination transaction described in the Agreement and Plan of Merger, dated as of October 31, 2016, by and among CenturyLink, Inc., Wildcat Merger Sub 1 LLC, WWG Merger Sub LLC and Level 3 Communications. Inc.
- "Non-CiC Good Reason Resignation" means an Eligible Employee's written notice to the Company, delivered within 7 calendar days of the offer to the Employee of a Non-Comparable Position, declining such offer and resigning employment. The written resignation shall not constitute a "Non-CiC Good Reason Resignation" if the Company fully corrects the circumstances giving rise to the Non-CiC Good Reason Resignation within 30 days following receipt of written notice from the Eligible Employee (the "30-day Cure Period"). If the Company does not fully correct the circumstances that gave rise to the Non-CiC Good Reason Resignation within the 30-day Cure Period, the Eligible Employee's employment shall be deemed to be a Non-CiC Good Reason Resignation and to have Terminated no later than two years following the expiration of the 30-day Cure Period or the Company's notice that it does not intend to cure, whichever is earlier.
- "Non-CIC Termination" means an Eligible Employee's Involuntary Termination of employment prior to a Change in Control or his Non-CiC Good Reason Resignation.
- "Non-Comparable Position" means a position offered by the Company or by another entity (e.g., one providing services to or purchasing assets from the Company) in connection with a reduction in headcount, an outsourcing to a vendor, a reorganization impacting two or more distinct business groups, a sale of distinct business groups, or a merger or acquisition other than a Change in Control, <u>and</u> that

- (i) involves a relocation of more than 50 miles in driving distance such that both (i) the driving distance from the work location of the new position to the current position held immediately prior to the change in job location is more than 50 miles and (ii) the increase in commuting distance from the employee's current home location to his/her work location (comparing current position location against the new position location) is more than 50 miles.
 - If an Employee's current position or new position is classified as "work at home" (i.e., with no assigned office at a CenturyLink facility), his/her home address is or, as applicable, will be deemed his/her work location with respect to such work at home position.

and/or

- (ii) has a Total Targeted Compensation (Base Salary plus target short term incentive or sales incentive opportunity) that is less than 90% of the Participant's current Total Targeted Compensation. A job at job grades P6 and below (or their equivalent) is a "Non-Comparable Position."
- " **Non-Compete Period** " means the one-year (12 consecutive months, not necessarily a calendar year or month) time period immediately following Termination Date.
- "Participant" means an Eligible Employee entitled to Severance Benefits.
- "Participating Company" means any divisions, subsidiaries or affiliates of the Company that participate in the Plan on or after the Effective Date of this restatement of the Plan and as listed, or as may be restricted, on Exhibit C. The CenturyLink Plan Design Committee or the Vice President of Human Resources with responsibility for benefits, or their respective designee, have the authority to amend Exhibit C at any time and from time to time, including but not limited to, adding or restricting participation of an entity as a Participating Company as such participation is approved by the CenturyLink Plan Design Committee.
- "Permanent Disability" means that an Eligible Employee has a permanent and total incapacity from engaging in any employment for the Company for physical or mental reasons. A "Permanent Disability" shall be deemed to exist if the Eligible Employee is judged to satisfy the requirements for disability benefits under the Company's long-term disability plan.
- "Plan" means the CenturyLink Executive Severance Plan, as set forth herein and as it may be amended from time to time.
- "Plan Administrator" is the CenturyLink Employee Benefits Committee as delegated by the Board of Directors to administer the terms of the Plan. In the event any member of the Employee Benefits Committee is entitled to Severance Benefits under the Plan or makes a claim for benefits under the Plan, the remaining members of the Employee Benefits Committee shall act as the Plan Administrator for purposes of administering the terms of the Plan with respect to such employee. The Plan Administrator may delegate all or any portion of its authority under the Plan to any other person(s).

- "Release" means the Confidential Separation Agreement and Release in Full of All Claims, substantially in the form attached hereto as Exhibit A, as the same may be amended from time to time in the sole discretion of the Company.
- "Service" means the total number of completed years the Eligible Employee was an Employee of the Company, as reflected in the Company's payroll and human resources records pursuant to the provisions of the CenturyLink Service Bridging Policy. Any period of employment with the Company, a Subsidiary, or a predecessor Company for which an Eligible Employee previously received Severance Benefits from the Company, shall be excluded from Service and, thus, not included in calculating any Severance Benefit, except to the extent the Eligible Employee repaid CenturyLink the Severance Benefits relating to such Service pursuant to Section 4.07.
- "Severance Benefit" means the severance pay amounts and benefits that a Participant is eligible to receive pursuant to Article IV of the Plan based on the Eligible Employee's Service and job performance.
- "Severance Period" means the number of weeks for which an Eligible Employee is entitled to receive Severance Benefits pursuant to Article IV of the Plan based on the Eligible Employee's Service.
- "Specified Employee" means, subject to section 409A(2)(B)(i) of the Code and Treasury Regulations issued thereunder, (i) an Employee who is an officer of the Company or its Subsidiaries and having annual compensation (within the meaning of section 414(q)(4) of the Code) greater than \$170,000 (for 2016, as it may be adjusted for inflation as described in section 416(i) of the Code), provided the number of such officers who are considered Specified Employees shall be limited to 50 employees as described in section 416(i) of the Code, (ii) a five-percent owner of the Company and its Subsidiaries, or (iii) a one-percent owner of the Company and its Subsidiaries greater than \$150,000. For purposes of determining who are Specified Employees, all Employees of the Company and its Subsidiaries, including employees who are nonresident aliens, shall be considered. The Plan Administrator, or its duly-appointed designee, shall determine the Specified Employees each year in accordance with section 416(i) of the Code, the "specified employee" requirements of section 409A of the Code, and applicable regulations. Specified Employees shall be identified as of December 31 of each year with respect to the 12-month period beginning on the next following April 1.
- "Subsidiary" means (a) a U.S.-based subsidiary of the Company (wherever incorporated, including but not limited to any subsidiary of EMBARQ Corporation, Qwest Communications International, Inc.), (b) any separately organized business unit, whether or not incorporated, of the Company, (c) any U.S.-based Company that is required to be aggregated with the Company pursuant to section 414 of the Code and regulations issued thereunder, and (iv) notwithstanding anything to the contrary herein, in order for employees a Subsidiary to be considered as eligible for benefits under the Plan, such Subsidiary must be listed as a Participating Company on Exhibit C and participation in this Plan is restricted as indicated on Exhibit C.
- "**Termination**" or "**Terminated**" means a separation from service with the Company as defined in Treasury Regulation § 1.409A-1(h).

- "Termination Date" means the date on which the active employment of the Eligible Employee by the Participating Company is severed, whether by reason of an Involuntary Termination, Voluntary Resignation, CIC Good Reason Resignation, Non-CiC Good Reason Resignation or Termination for Cause.
- "Total Targeted Cash Compensation" means the Eligible Employee's annual rate of Base Salary in effect on a specified date plus the targeted annual incentive opportunity under either the short-term incentive compensation program or the sales incentive compensation program, as applicable, on that same date.
- "Voluntary Resignation" means any retirement or Termination that is not initiated by the Company other than a CIC Good Reason Resignation.
- "Year of Service" means each completed full year of Service.

Article III ELIGIBILITY FOR BENEFITS

Section 3.01 Conditions

- (a) Except as expressly provided in Exhibit E, eligibility for and receipt of any Severance Benefits under the Plan are expressly conditioned on the Eligible Employee's (i) execution of a Release in connection with his termination of employment with the Company; (ii) compliance with all the terms and conditions of such Release; (iii) compliance with all the terms and conditions of the restrictive covenants set forth in Article VI; (iv) execution of a written agreement that authorizes the deduction of amounts owed to the Company prior to the payment of any Severance Benefit (or in accordance with any other schedule as the Plan Administrator may determine to be appropriate); provided that, to the extent applicable, any such deduction shall be made in compliance with section 409A of the Code; and (v) acknowledgement that all decisions and determinations of the Plan Administrator shall be final and binding on the Eligible Employee, his beneficiaries and any other person having or claiming an interest under the Plan on his behalf. Subject to and contingent upon the consummation of the LVLT Corporate Transaction, Severance Benefits for Legacy Level 3 Employees during the one-year period immediately following the closing of the LVLT Corporate Transaction are described in Exhibit C, subject to all applicable Plan terms.
- (b) If the Plan Administrator determines that the Eligible Employee has not fully complied with any of the terms of the Plan or the Release, the Plan Administrator, acting on behalf of the Company, may deny Severance Benefits not yet paid or discontinue the payment of the Eligible Employee's Severance Benefits and may require the Eligible Employee to repay any portion of any Severance Benefits already received under the Plan, by providing written notice of such repayment obligation to the Eligible Employee. If the Plan Administrator notifies a Participant that repayment of all or any portion of the Severance Benefits received under the Plan is required, such amounts shall be repaid within 30 calendar days of the date the written notice is sent. Any remedy under this subsection (b) shall be in addition to, and not in place of, any other remedy, including injunctive relief, that the Company may have.

Article IV DETERMINATION OF SEVERANCE BENEFITS

Section 4.01 Severance Benefits – Non-CIC Termination . Except as noted on Exhibit C, each Participant who incurs an Involuntary Termination, other than a CIC Termination or a Lower Performer termination, or who declines an offer of a Non-Comparable Position or incurs a Non-CIC Good Reason Resignation, shall, subject to the conditions of the Plan, receive the Severance Benefits as provided in this Section. The Severance Benefits described in this Section shall be paid in accordance with Article V.

| Tier Title | Severance Period |
|--|--|
| Directors at P7 or P8 | 2 weeks of Severance Benefits per year of Service, with a minimum of 12 weeks of Severance Benefits and a maximum of 34 weeks of Severance Benefits; Severance Benefits are calculated using Base Salary |
| Vice Presidents at E1 and E2 | 2 weeks of Severance Benefits per year of Service, with a minimum of 18 weeks of Severance Benefits and a maximum of 52 weeks of Severance Benefits; Severance Benefits are calculated using Base Salary |
| Vice-President at E3 or above, other than Executive Officers | 6 weeks of Severance Benefits plus an additional 2 weeks of Severance Benefits for each Year of Service up to a maximum of 52 weeks, but in no event less than 39 weeks; Severance Benefits are calculated using Total Targeted Cash Compensation. |
| Executive Officers | 52 weeks of Severance Benefits, calculated using Total Targeted Cash Compensation |
| СЕО | 104 weeks of Severance Benefits, calculated using Total Targeted Cash Compensation |

Eligible Employees who are in positions classified as "Regular Part-Time" status will receive severance benefits prorated based on the Eligible Employee's weekly base salary or wages as of the employee's last day of work (excluding overtime, bonus or incentive compensation) as noted in the HR payroll system. Any Eligible Employee who is in a position that is classified as "Regular Full Time" status but working less than a 40 hour work week will receive severance benefits prorated based on the Eligible Employee's weekly base salary or wages as of the employee's last day of work (excluding overtime, bonus or incentive compensation) as noted in the HR payroll system.

Section 4.01A Lower Performer Severance Benefit: Eligible Employees below the E3 job level who have at least one year of service are eligible for this Lower Performer Severance Benefit if they are terminated for poor job performance, such as missing sales targets or deficient leadership of their teams. An Employee who is terminated for poor attendance (other than absences that are protected by applicable law), misconduct, including but not limited to violation of the CenturyLink Code of Conduct, or violating other CenturyLink policies or supervisor directives, remains ineligible

for Severance Benefits, including this Lower Performer Severance Benefit. As with other Severance Benefits under the Plan, the Lower Performer Severance Benefit is contingent upon the Participant signing an acceptable Release and complying with the restrictive covenants in Article VI of this Plan. The Severance Benefit under this subsection is:

| Tier Title | Severance Period |
|------------------------------------|---|
| Teresidents at E.I. and E.Z. Willi | 1 week of Severance Benefits per year of Service, with a minimum of 2 weeks of Severance Benefits and a maximum of 8 weeks of Severance Benefits; Severance Benefits are calculated using Base Salary; the benefit is prorated, as applicable, for Part-Time status or Full-Time status working less than 40 hours in a work week |

Section 4.02 Severance Benefits – CIC Termination . Except as noted on Exhibit C, each Participant who incurs a CIC Termination shall, subject to the conditions of the Plan, receive the Severance Benefits subject to the conditions of the Plan, as provided in this Section, which benefits shall be paid in accordance with Article V:

| Tier Title | Severance Period |
|--|---|
| | 6 weeks of Severance Benefits, calculated using Total Targeted Cash Compensation plus an additional 2 weeks of Severance Benefits for each Year of Service up to a maximum of 52 weeks, but in no event less than 26 weeks; the benefit is prorated, as applicable, for Part-Time status or Full-Time status working less than 40 hours in a work week. |
| Vice-President at E1 or above, other than Executive Officers and the CEO | 52 weeks of Severance Benefits, calculated using Total Targeted Cash Compensation; the benefit is prorated, as applicable, for Part-Time status or Full-Time status working less than 40 hours in a work week |

Section 4.03 Severance Benefits – Non-CIC Terminations or CIC Terminations . Except as expressly noted on Exhibit C, the following additional provisions apply to Participants who incur either a CIC Termination or a Non-CIC Termination:

- (a) <u>Employee Benefits</u>. Medical (including prescription drug), vision and dental benefits for the Participant and his/her dependents, if enrolled in the CenturyLink Health Care Plan at the time of Termination, will continue through the last day of the month in which the Termination occurs. Eligibility for life insurance coverage and short-term disability coverage and benefits end on the date of termination from employment; however, in accordance with applicable laws and the respective plan documents, notice of any rights to continue, convert, or obtain replacement coverage will be provided to Participants.
 - (i) <u>COBRA</u>. Participants, and if applicable, their dependents, who participate in the CenturyLink Health Care Plan may elect under COBRA to continue coverage for a monthly charge for a maximum of 18 months beyond the date coverage would otherwise terminate,

subject to the provisions of the applicable plans and COBRA. The Company will provide COBRA enrollment information to employees within 45 days following the Termination Date. Employees who wish to continue coverage must return the COBRA enrollment forms within 60 days from receipt of the COBRA notification.

Participants whose job levels are below E3 (or its equivalent) who elect COBRA coverage will be eligible to receive a subsidy to their COBRA payments according to the schedule in the chart attached at Exhibit D. Participants whose job levels are at E3 and above who elect COBRA coverage will be eligible to receive a subsidy to their COBRA payments according to the applicable schedule in Exhibit D. For all Participants, the subsidy means that the Participant will pay the same premium contribution rate for medical/prescription, vision and dental coverage during the period of the subsidy as comparable active employees pay. Following the expiration of the subsidy, the employee will pay the COBRA rate, which is 102% of the cost of coverage, as long as coverage remains in place.

- (b) The COBRA subsidy will terminate upon the earlier of (i) the date the Participant discontinues COBRA coverage or otherwise becomes ineligible for COBRA coverage, or (ii) the date the Participant is eligible to enroll (regardless of whether he actually enrolls) in medical coverage sponsored by a subsequent employer of the Participant.
- (c) <u>Short Term Incentive or Sales Incentive Payment for Period Prior to Termination Date</u>. Short-term incentive or sales incentive compensation for the period prior to the Termination Date shall be calculated and paid in accordance with the terms of the applicable short-term incentive compensation plan.
- (d) <u>Retirement Plans</u>. The provisions of any applicable qualified and/or nonqualified defined contribution or defined benefit plan maintained by the Company pursuant to which an Eligible Employee is eligible to participate, shall control with respect to any recognition of service during the Severance Period and the eligibility for benefits before, during and after the Severance Period.
- (e) <u>Equity</u>. The provisions of any applicable equity incentive plan or equity compensation plan maintained by the Company pursuant to which an Eligible Employee has received an equity grant and the Eligible Employee's relevant grant agreement shall control with respect to the treatment of the Eligible Employee's equity grants upon the Eligible Employee's Termination.
- (f) <u>Outplacement Assistance</u>. The Company at its cost shall provide to the Participant outplacement assistance by a reputable firm specializing in such services. Such assistance shall comply with the requirements of the duration of Treasury Regulation $\S 1.409A-1(b)(9)(v)(A)$.
- (g) Any failure to execute a Release within the time period set forth in Section 5.01 will result in the forfeiture of the compensation (if any) contingent upon execution of such Release.
- **Section 4.04 Voluntary Resignation; Termination for Death or Permanent Disability** . If the Eligible Employee's employment terminates on account of the Eligible Employee's (a) Voluntary

Resignation, (b) death, or (c) Permanent Disability, then the Eligible Employee shall not be entitled to receive Severance Benefits under this Plan and shall be entitled only to those benefits (if any) as may be available under the Company's applicable equity incentive plans, equity compensation plans, or equity grant agreements; other separation or severance plans, policies, or arrangements; or any other employee health, welfare, retirement or benefit plan.

Section 4.05 Termination for Cause . If the Company terminates the Eligible Employee's employment for Cause, the Eligible Employee shall not be entitled to receive Severance Benefits under this Plan. Notwithstanding any other provision of the Plan to the contrary, if the Plan Administrator determines that an Eligible Employee engaged in conduct that constitutes Cause at any time prior to the Eligible Employee's Termination Date, any Severance Benefits payable to the Eligible Employee under Article IV shall immediately cease, and the Eligible Employee shall be required to return any Severance Benefits paid to the Eligible Employee prior to such determination. The Company may withhold paying Severance Benefits under the Plan pending resolution of an inquiry that could lead to a determination that Cause exists.

Section 4.06 Deductions from Severance Benefits . The Plan Administrator reserves the right to make deductions in accordance with applicable law for any monies owed to the Company by the Eligible Employee or the value of Company property that the Eligible Employee has retained in his possession, *e.g.*, if an employee (a) was overpaid commissions/sales incentive, (b) had a negative PTO balance, (c) retained a company laptop or other property, or (d) used his corporate card for unauthorized personal purchases. To the extent applicable, any such deduction from Severance Benefits shall be made in compliance with section 409A of the Code. To the maximum extent allowed by applicable law, Eligible Employees are deemed, by their participation in the Plan and execution of a Release (which is a pre-condition to receipt of severance benefits hereunder), to have consented in writing to such deductions.

Section 4.07 Repayment of Severance Benefits upon subsequent rehire . Employees rehired by CenturyLink before the expiration of the period covered by their Severance Benefit will be required to repay all or part of the Severance Benefit upon rehire. The repayment percentage is obtained by dividing the number of days remaining in the layoff period (*i.e.* , the total number of days included in the Severance Benefit calculation minus the number of prior days in layoff status) by the total number of days included in the Severance Benefit calculation. The amount to be repaid is equal to the total Severance Benefit multiplied by the repayment percentage.

Article V TIMING AND METHOD OF PAYMENT AND LIMITATION ON BENEFITS

Section 5.01 Timing and Method of Payment

(a) <u>Non-CiC Severance Benefits</u>. Subject to Section 5.02 below, Non-CiC Severance Benefits (benefits under Sections 4.01 and 4.01A) shall be paid periodically, in prorated installments on a bi-weekly basis in accordance with the Company's normal payroll cycle, less withholding for all applicable federal, state and local taxes and other applicable withholdings and deductions, over the Participant's Severance Period. The first installment (including any retroactive installments)

shall begin no later than 60 days after the Eligible Employee's Termination Date, subject to the Eligible Employee's execution of a Release and, if applicable, the expiration of any revocation period for such Release within such 60-day period.

- (b) <u>CiC Severance Benefits</u>. Subject to Section 5.02 below, CiC Severance Benefits (benefits under Section 4.02) shall be paid in a single lump sum cash payment, less withholding for all applicable federal, state and local taxes and other applicable withholdings and deductions, no later than 60 days after the Eligible Employee's Termination Date, subject to the Eligible Employee's execution of a Release and, if applicable, the expiration of any revocation period for such Release within such 60-day period.
- (c) <u>General Rules</u>. In the event of the Participant's death after he becomes entitled to Severance Benefits under the Plan, but prior to full payment of all Severance Benefits due to such Participant, any remaining Severance Benefits due to the Participant under Section 4.01, 4.01A or 4.02 above shall be paid to the Participant's estate in a lump sum payment within 60 days following written notification of the Participant's death. Interest will not be credited on any unpaid Severance Benefit due to a Participant. Payment(s) shall be made by direct deposit or by mailing to the last address provided by the Participant to the Company or such other reasonable method as determined by the Plan Administrator.
 - (d) <u>Timing of Payment Rules Applicable to Specified Employees</u>.
- (i) If the Participant is a Specified Employee on the date of the Participant's Termination and there are benefits provided to the Employee hereunder that are taxable to the Participant, then the value of the aggregate amount of such taxable benefits provided to the Participant during the six month period following the Termination Date shall be limited to (A) the amount specified by section 402(g)(1)(B) of the Code for the year in which the Termination occurred, (B) medical benefits that are allowed to be provided during such time pursuant to Treasury Regulation § 1.409A-1(b)(9)(v)(B), (C) any amounts subject to the short-term deferral exception to section 409A of the Code, (D) any amount payable under the involuntary separation from service exception to section 409A of the Code, and (E) other payments to the extent they are covered by an exception to such 6-month delay applicable to Specified Employees, under section 409A of the Code, applicable Treasury Regulations and IRS guidance.
- (ii) To the extent there has been delayed commencement of any portion of the Severance Benefits to which the Participant is entitled under this Plan in order to avoid a prohibited distribution under section 409A(a)(2)(B)(i) of the Code (any such delayed commencement, a "Payment Delay"), such portion of the Participant's Severance Benefit payments shall not be provided to Participant prior to the earlier of (A) the expiration of the six-month period measured from the date of Participant's Termination or (2) the date of the Participant's death. Upon the earlier of such dates, all payments deferred pursuant to the Payment Delay shall be paid in a cash lump sum payment to the Participant, and any remaining payments due under the Plan shall be paid as otherwise provided herein.
- (e) <u>Application of Section 409A of the Code</u>. To the extent that the Company determines that any compensation or benefit payable under the Plan constitutes nonqualified deferred

compensation within the meaning of section 409A of the Code, the Plan, with respect to such compensation or benefit, will be construed and interpreted to comply with the requirements of section 409A of the Code or one or more applicable exceptions thereto, in accordance with the statutory provisions of section 409A of the Code and any Treasury Regulations and other interpretive guidance issued thereunder. Notwithstanding any provision of the Plan to the contrary:

- (i) <u>No Guaranty of Tax Liability</u>. In no event does the Company guarantee any particular tax consequences, outcome or tax liability to Eligible Employees. No provision of the Plan shall be interpreted or construed to transfer any liability for failure to comply with the requirements of section 409A of the Code from an Eligible Employee or any other individual to the Company.
- (ii) <u>Additional Actions Permitted</u>. In the event the Company determines that any compensation or benefit payable hereunder may be subject to section 409A of the Code, the Company (without any obligation to do so or obligation to indemnify any Eligible Employee for any failure to do so) may adopt, without the consent of any Eligible Employee, such amendments to the Plan or take any other actions that the Company in its sole discretion determines are necessary or appropriate for such compensation or benefit to either (a) be exempt from section 409A of the Code or (b) comply with the requirements of section 409A of the Code.
- (iii) <u>Actual Payment Date</u>. Whenever a payment under the Plan specifies a payment period, the actual date of payment within such specified period shall be within the sole discretion of the Company, and no Participant shall have any right (directly or indirectly) to determine the year in which such payment is made.
- (iv) <u>Installments</u>. Each installment payment payable hereunder shall be deemed to be a separate payment for purposes of section 409A of the Code. With respect to compensation or benefits subject to Code Section 409A that are paid pursuant to the Company's payroll schedule, in the event of any change in that payroll schedule after such compensation or benefits are no longer subject to a substantial risk of forfeiture, each installment or payment to be made to an Eligible Employee under this Plan shall be made according to such new payroll schedule in effect at the time of such payment, but in any event within thirty (30) days of the payroll date that would have applied pursuant to the payroll schedule in effect on the date that the compensation or benefits to such Employee were no longer subject to a substantial risk of forfeiture.
- (a) <u>Reimbursements</u>. To the extent that any amounts payable hereunder are deemed to be reimbursements and other separation payments under Treasury Regulation §1.409A-1(b)(9)(v), they shall not be deemed to provide for the deferral of compensation governed by section 409A of the Code. If they do constitute deferral of compensation governed by Code Section 409A, they shall be deemed to be reimbursements or in-kind benefits governed by Treasury Regulation § 1.409A-3(i)(1)(iv). If the previous sentence applies, (i) the amount of expenses eligible for reimbursement or in-kind benefits provided during the Employee's taxable year shall not affect the expenses eligible for reimbursement or in-kind benefits in any other taxable year, provided that the foregoing shall not be violated by any lifetime or annual limits contained in the Company's group health plans; (ii) the reimbursement of an eligible expense must be made on or before the last day

of the Eligible Employee's taxable year following the taxable year in which the expense was incurred and (iii) the right to reimbursement or in-kind benefits shall not be subject to liquidation or exchange for another benefit.

(vi) Application of the exceptions to section 409A of the Code are additive, that is, the exceptions may be "stacked" and used in combination with other exceptions, including but not limited to the short-term deferral exception and the separation pay exception.

Section 5.02 Limitation on Benefits

- (a) Notwithstanding anything set forth in the Plan to the contrary, if any payment or benefit, including the Severance Benefits, an Eligible Employee would receive from the Company ("Payment") would (i) constitute a "parachute payment" within the meaning of section 280G of the Code and (ii) but for this sentence, be subject to the excise tax imposed by section 4999 of the Code (the "Excise Tax"), then such Payment shall be reduced to the Reduced Amount. The "Reduced Amount" shall be either (i) the largest portion of the Payment that would result in no portion of the Payment being subject to the Excise Tax or (ii) the largest portion, up to and including the total, of the Payment, whichever amount, after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in the Eligible Employee's receipt, on an after-tax basis, of the greater amount of the Payment notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. If a reduction in payments or benefits (or a cancellation of the acceleration of vesting of stock options or equity awards) constituting "parachute payments" is necessary so that the Payment equals the Reduced Amount, such reduction and/or cancellation of acceleration shall occur in the order that provides the maximum economic benefit to the Eligible Employee. In the event that acceleration of vesting of a stock option or equity award is to be reduced, such acceleration of vesting also shall be canceled in the order that provides the maximum economic benefit to the Eligible Employee.
- (b) The Company shall appoint a nationally-recognized accounting firm with appropriate subject matter expertise to make the determinations required under this Section 5.03.
- (c) The Company shall bear all expenses with respect to the making of the determinations by such accounting firm required to be made under this Section 5.03. The accounting firm engaged to make the determinations under this Section 5.03 shall provide its calculations, together with detailed supporting documentation, to the Company and the Eligible Employee as soon as practicable after the date on which the Eligible Employee's right to a Payment is triggered (if requested at that time by the Company or the Eligible Employee) or such other time as requested by the Company or the Eligible Employee. If the accounting firm determines that no Excise Tax is payable with respect to a Payment, either before or after the application of the Reduced Amount, it shall furnish the Company with an opinion reasonably acceptable to the Eligible Employee that no Excise Tax will be imposed with respect to such Payment. Any good faith determinations of the accounting firm made under this Section 5.03 shall be final, binding, and conclusive upon the Company and the Eligible Employee.

Article VI RESTRICTIVE COVENANTS

Section 6.01 Code of Conduct . Each Eligible Employee shall adhere in all respects to the Company's ethics and compliance program, Code of Conduct, Employee Handbook and/or other applicable policies, as they may from time to time be established, interpreted, amended or terminated.

Section 6.02 Proprietary Information

- (a) Each Eligible Employee shall acknowledge that, during the course of his employment, the Eligible Employee has learned or will learn or develop Proprietary Information. Each Eligible Employee shall further acknowledge that unauthorized disclosure or use of such Proprietary Information, other than in discharge of the Eligible Employee's duties, will cause the Company irreparable harm. Except in the course of his employment with the Company, in pursuit of the business of the Company, or as otherwise required in employment with the Company or by applicable law, each Eligible Employee shall not, during the course of his employment or at any time following termination of his employment, directly or indirectly, disclose, publish, communicate, or use on his behalf or another's behalf, any Proprietary Information. If during or after his employment, the Eligible Employee has any questions about whether particular information is Proprietary Information, the Eligible Employee shall consult with the Company's General Counsel or other representative designated by the Company.
- (b) Each Eligible Employee shall also agree to promptly disclose to the Company any information, ideas, or inventions made or conceived by him or her that results from or are suggested by services performed by the Eligible Employee for the Company, and to assign to the Company all rights pertaining to such information, ideas, or inventions. Knowledge or information of any kind disclosed by the Eligible Employee to the Company shall be deemed to have been disclosed without obligation on the part of the Company to hold the same in confidence, and the Company shall have the full right to use and disclose such knowledge and information without compensation to the Eligible Employee.

Section 6.03 Non-Competition

- (a) During the Eligible Employee's employment with the Company and during the Non-Compete Period, each Eligible Employee shall agree, to the maximum extent allowed under applicable law, that he shall not engage in Competitive Employment.
- (b) If an Eligible Employee ceases to be employed by the Company because of the sale, spin-off, divestiture, or other disposition by the Company of a Subsidiary, division, or other divested unit employing the Eligible Employee, this provision shall continue to apply during the Non-Compete Period, except that the Eligible Employee's continued employment for the Subsidiary, division, or other divested unit disposed of by the Company shall not be deemed a violation of this provision.

Section 6.04 Inducement of Employees, Customers and Others . During an Eligible Employee's employment with the Company and during the Non-Compete Period, each Eligible

Employee shall agree, to the maximum extent allowed under applicable law, that he will not, directly or indirectly, solicit, induce, or encourage any employee, consultant, agent or customer of the Company or its Subsidiaries or vendors or other parties doing business with the Company or its Subsidiaries, to terminate their employment, agency, or other relationship with the Company or its Subsidiaries or to render services for or transfer business to any Competitor, and each Eligible Employee shall not initiate discussion with any such person for any such purpose or authorize or knowingly cooperate with the taking of any such actions by any other individual or entity on behalf of the Competitor.

Section 6.05 No Adverse Actions . During the Non-Compete Period, each Eligible Employee shall not, without the prior written consent of the Company, in any manner, solicit, request, advise, or assist any other person to (a) undertake any action that would be reasonably likely, or is intended, to result in a reorganization, merger, share exchange, transfer of ownership of 50% or more of the outstanding voting securities of the Company, consolidation, or sale or disposition of all or substantially all of the assets of the Company, or (b) seek to control or change the composition of the Board in any material manner.

Section 6.06 Return of Property . Each Eligible Employee shall, upon the Eligible Employee's Termination Date, return to the Company all property of the Company in the Eligible Employee's possession, including all notes, reports, sketches, plans, published memoranda, or other documents, whether in hard copy or in electronic form, created, developed, generated, received, or held by the Eligible Employee during the Eligible Employee's employment, concerning or related to the Company's business, whether or not containing or relating to Proprietary Information. Each Eligible Employee shall not remove, by e-mail, by removal of computer discs or hard drives, or by other means, any of the above property containing Proprietary Information, or reproductions or copies thereof, or any apparatus from the Company's premises without the Company's written consent.

Section 6.07 Non-Disparagement . Each Eligible Employee shall agree to refrain from making any statements about the Company, its Subsidiaries or their officers or directors that would disparage, or reflect unfavorably upon the image or reputation of the Company, its Subsidiaries or any such officer or director.

Section 6.08 Assistance with Claims

- (a) Each Eligible Employee shall agree, that, during and after the Eligible Employee's employment by the Company, the Eligible Employee shall assist the Company, on a reasonable basis, in the defense of any claims or potential claims that may be made or threatened to be made against it in any action, suit, or proceeding, whether civil, criminal, administrative, or investigative ("Proceeding.") and shall assist the Company in the prosecution of any claims that may be made by the Company in any Proceeding, to the extent that such claims may relate to the Eligible Employee's services.
- (b) Each Eligible Employee shall agree, unless precluded by law, to promptly inform the Company if the Eligible Employee is asked to participate (or otherwise become involved) in any Proceeding involving such claims or potential claims.

(c) Each Eligible Employee shall also agree, unless precluded by law, to promptly inform the Company if the Eligible Employee is asked to assist in any investigation (whether governmental or private) of the Company or its Subsidiaries (or its actions), regardless of whether a lawsuit has then been filed against the Company or its Subsidiaries with respect to such investigation.

Section 6.09 Reasonableness . In the event that any of the provisions of this Article VI should ever be adjudicated to exceed the time, geographic, service, or other limitations permitted by applicable law in any jurisdiction, then such provisions shall be deemed reformed in such jurisdiction to the maximum time, geographic, service, or other limitations permitted by applicable law.

Section 6.10 Equitable Relief

- (a) Each Eligible Employee shall acknowledge that the restrictions contained in this Article VI are reasonable and necessary to protect the legitimate interests of the Company, its Subsidiaries and its affiliates, that the Company would not have established this Plan in the absence of such restrictions, and that any violation of any provision of this Article VI will result in irreparable injury to the Company. Each Eligible Employee shall represent that his experience and capabilities are such that the restrictions contained in this Article VI will not prevent the Eligible Employee from obtaining employment or otherwise earning a living at the same general level of economic benefit as is currently the case. Each Eligible Employee shall further represent and acknowledge that (i) he has been advised by the Company to consult his own legal counsel in respect of this Plan, and (ii) that he has had full opportunity, prior to agreeing to participate in this Plan, to review thoroughly this Plan with his counsel.
- (b) Each Eligible Employee shall agree that the Company shall be entitled to preliminary and permanent injunctive relief, without the necessity of proving actual damages and without posting a bond or other security, as well as an equitable accounting of all earnings, profits and other benefits arising from any violation of this Article VI, which rights shall be cumulative and in addition to any other rights or remedies to which the Company may be entitled.
- (c) Each Eligible Employee shall also irrevocably and unconditionally consent to the service of any process, pleadings, notices or other papers in a manner permitted by the notice provisions of Section 10.04.
- **Section 6.11 Survival of Provisions** . The obligations contained in this Article VI shall survive the termination of each Eligible Employee's employment with the Company and shall be fully enforceable, to the maximum extent allowed under applicable law, thereafter.

Article VII PLAN ADMINISTRATOR

Section 7.01 Authority and Duties . The Company has established this Plan on behalf of Company. The Company is the "Plan Sponsor" and the CenturyLink Employee Benefits Committee (the "EBC") is the "Plan Administrator" (unless and to the extent that Plan Administrator duties

have been otherwise delegated), as those terms are defined under ERISA. In the absence of an EBC, the Company is the Plan Administrator. As Plan Administrator, the EBC shall have primary responsibility for the operation and administration of the Plan, and this responsibility shall be delegated as designated in the Plan or as determined by the EBC. To the extent of such delegation, the delegate shall be a "named fiduciary" of the Plan, as that term is defined in ERISA, and shall have all of the authority, discretion and powers of the Plan Administrator. It shall be the duty of the Plan Administrator to properly administer the Plan. The Plan Administrator shall have the full power, authority and discretion to construe, interpret and administer the Plan, to make factual determinations, to correct deficiencies therein, and to supply omissions. The Plan Administrator may adopt such rules and regulations and may make such decisions as it deems necessary or desirable for the proper administration of the Plan.

As set forth in this Section, "administration" of this Plan does not include any decisions to amend or terminate all or any portion of the Plan, which functions are reserved to the Company in its capacity as Plan Sponsor and not in any fiduciary capacity.

Section 7.02 Compensation of the Plan Administrator . The Plan Administrator shall receive no compensation for services as such. However, all reasonable expenses of the Plan Administrator shall be paid or reimbursed by the Company upon proper documentation. The Plan Administrator shall be indemnified by the Company against personal liability for actions taken in good faith in the discharge of the Plan Administrator's duties.

Section 7.03 Records, Reporting and Disclosure . The Plan Administrator shall keep a copy of all records relating to the payment of Severance Benefits to Participants and former Participants and all other records necessary for the proper operation of the Plan. All Plan records shall be made available to the Company and to each Eligible Employee for examination during business hours except that an Eligible Employee shall examine only such records as pertain exclusively to the examining Eligible Employee and to the Plan. The Plan Administrator shall prepare and shall file as required by law or regulation all reports, forms, documents and other items required by ERISA, the Code, and every other relevant statute, each as amended, and all regulations thereunder (except that the Company, as payor of the Severance Benefits, shall prepare and distribute to the proper recipients all forms relating to withholding of income or wage taxes, Social Security taxes, and other amounts that may be similarly reportable).

Section 7.04 Discretion . Any decisions, actions or interpretations to be made under the Plan by the Plan Administrator shall be made in its sole and absolute discretion, subject to the terms of the Plan and applicable law, and need not be uniformly applied and such decisions, actions or interpretations shall be final, binding and conclusive upon all parties, with respect to denied claims for Severance Benefits. Not in limitation, but in amplification of the foregoing and of the authority conferred upon the Plan Administrator, the Plan Sponsor specifically intends that the Plan Administrator and its duly authorized delegates have the greatest permissible discretion to construe the terms of the Plan and to determine all questions concerning eligibility, participation, and benefits. The decisions by the Plan Administrator or any delegates shall be conclusive and binding, and any interpretation, determination, or other action by them is intended to be subject to the most deferential standard of review. Such standard of review is not to be affected by any real or alleged conflict of

interest on the part of the Plan Administrator or its delegates. The EBC delegated its authority and discretion to review and grant or deny initial claims under the Plan to Severance Analyst, Legal & Corporate Administration, and this delegation has been accepted. In addition to the duties and powers described hereunder and elsewhere in this Plan, the Plan Administrator or its delegate is specifically given the discretionary authority and such powers as are necessary for the proper administration of the Plan, including, but not limited to, the following:

- to resolve ambiguities or inconsistencies,
- to supply omissions and the like,
- to make determinations, grants, or denials of the amount, manner, and time of payment of any Benefits under the terms of the Plan.
- to authorize its agents or delegates to execute or deliver any instrument or make payments on the Plan Administrator's behalf or with respect to the Plan;
- to select and retain counsel, service providers, vendors, employ agents, and provide for such clerical, accounting, actuarial, legal, consulting and/or claims processing services as it deems necessary or desirable to assist the Plan Administrator in the administration of the Plan;
- to prepare and distribute, in such manner as the Plan Administrator determines to be appropriate, summary plan descriptions and other information explaining the Plan;
- to receive from each Participating Company and from Eligible Employees such information as shall be necessary for the proper administration of the Plan, and to require such information as a condition to receiving benefits under the Plan;
- to furnish the Company, upon request, such annual reports with respect to the administration of the Plan as the Plan Administrator deems reasonable and appropriate;
- to receive, review and keep on file, as the Plan Administrator deems necessary or appropriate, reports of Plan payments and reports of disbursements for expenses; and
- in general to decide and /or settle questions and disputes,

and all such authorizations, interpretations, determinations, decisions and settlements shall be final and binding for purposes of the Plan.

Section 7.05 Implementation of the Plan is Non-Fiduciary . The decision to implement the Plan and as to which employees, if any, will be included in an Involuntary Termination or terminated due to job performance, as well as all other Management Team decisions, including but not limited to, offering ongoing employment, shall be determined by the Management Team as Plan sponsor functions and not in any fiduciary capacity. Management shall have unlimited discretion under the Plan in making Involuntary Termination decisions.

Article VIII AMENDMENT, SUSPENSION AND TERMINATION

Section 8.01 Amendment, Suspension and Termination

- (a) <u>In General</u>. Except as otherwise provided in this Article VIII, the Plan may be amended, suspended or terminated, without prior notice, in whole or in part at any time, with future or with retroactive effect, by a writing approved by Board of Directors and signed on behalf of the Company by an authorized officer. The Plan may also be amended in writing by the CenturyLink Plan Design Committee. Amendment or termination of the Plan with respect to any Subsidiary of the Company shall not affect the rights of any Eligible Employee to any Plan benefit to which such Eligible Employee may have become irrevocably entitled under the Plan prior to the date such amendment or termination is adopted. On termination of the Plan, all rights to benefits end unless an Eligible Employee already has become entitled to such benefits. The CenturyLink Plan Design Committee or the Vice President Legal with responsibility for Severance Benefits, or their respective designee, have the authority to amend Exhibit C, "List of Participating Companies," at any time and from time to time, including but not limited to, adding an entity as a Participating Company as such participation is approved by the CenturyLink Plan Design Committee.
- (b) <u>Prospective Operation of Amendments</u>. Any amendment, suspension or termination shall operate prospectively only and shall not impair an Eligible Employee's right to retain any amount paid to him/her prior to the date of the amendment, suspension or termination (except as expressly provided herein) or cause the cessation of Severance Benefits after an Eligible Employee has executed a Release as required under Article III.
- **Section 8.02** Continuation of Plan . Subject to Section 8.01, the Plan shall continue and shall be binding on the Company and its successors until the Company and/or its successors have fully performed all of the Company's obligations under the Plan with respect to all Eligible Employees covered under the Plan as of the Effective Date.

Article IX CLAIMS PROCEDURES

Section 9.01 Claims . Except as expressly provided in Exhibit E, the Plan Administrator, in its sole discretion, shall determine eligibility for Severance Benefits and the amount of Severance Benefits, as well as the administration of Severance Benefits, in accordance with the terms of this Plan. An Eligible Employee or his beneficiary, as applicable (the "claimant") may contest such determinations by completing and filing with the Plan Administrator a written request for review in the manner specified by the Plan Administrator within 90 days following the Eligible Employee's Termination Date. A Participant must send a written claim to the Plan Administrator as follows:

The CenturyLink Employee Benefits Committee c/o CenturyLink, Inc. 214 E. 24 th Street Vancouver, WA 98663.

Each such application must be supported by such information as the Plan Administrator deems relevant and appropriate. The claimant may not bring an action for any alleged wrongful denial of Plan benefits in a court of law unless the claims and appeals procedures described in this Article IX are exhausted and a final determination has been made by the Plan Administrator or its delegate. If the claimant challenges a decision by the Plan Administrator or its delegate, any review by a court of law will be limited to the facts, evidence and issues presented to the Plan Administrator during the claims procedure set forth in this Article IX. Facts and evidence that become known to the claimant after such individual has exhausted the claims procedure must be brought to the attention of the Plan Administrator for reconsideration. Issues not raised with the Plan Administrator will be deemed waived.

Section 9.02 Initial Claim . In the event that any claim under Section 9.01 is denied in whole or in part, the claimant whose claim has been so denied shall be notified of such denial in writing by the Plan Administrator, or its delegate, within ninety (90) days after the receipt of the claim for benefits. This period may be extended an additional ninety (90) days if the Plan Administrator, or its delegate, determines such extension is necessary and the Plan Administrator, or its delegate, provides notice of the extension to the claimant prior to the end of the initial ninety (90) day period. The notice advising of the denial shall: (i) specify the reason or reasons for denial, (ii) make specific reference to the Plan provisions on which the determination was based, (iii) describe any additional material or information necessary for the claimant to perfect the claim (explaining why such material or information is needed), and (iv) describe the Plan's review procedures and the time limits applicable to such procedures, including a statement of the claimant's right to bring a civil action under section 502(a) of ERISA following an adverse benefit determination on review.

Section 9.03 Appeals of Denied Administrative Claims . All appeals of denied claims shall be made by the following procedure:

- (a) A claimant whose claim has been denied shall file with the Plan Administrator a written notice of appeal of the denial. Such notice shall be filed within sixty (60) calendar days of receipt of notification by the Plan Administrator of the denial of a claim, shall be made in writing, and shall set forth all of the facts upon which the appeal is based. Appeals not timely filed shall be barred.
- (b) The Plan Administrator shall consider the merits of the claimant's written presentations, the merits of any facts or evidence in support of the denial of benefits, and such other facts and circumstances as the Plan Administrator shall deem relevant.
- (c) The Plan Administrator shall render a determination upon the appealed claim which determination shall be accompanied by a written statement as to the reasons therefore. The determination shall be made to the claimant within sixty (60) days of the claimant's request for review, unless the Plan Administrator determines that special circumstances require an extension of time for processing the claim. In such case, the Plan Administrator shall notify the claimant of the need for an extension of time to render its decision prior to the end of the initial sixty (60) day period, and the Plan Administrator shall have an additional sixty (60) day period to make its determination. The determination so rendered shall be binding upon all parties. If the determination is adverse to the claimant, the notice shall: (i) provide the reason or reasons for denial, (ii) make

specific reference to the Plan provisions on which the determination was based, (iii) state that the claimant is entitled to receive, upon request and free of charge, reasonable access to, and copies of, all documents, records and other information relevant to a the claimant's claim for benefits, and (iv) state that the claimant has the right to bring a civil action under section 502(a) of ERISA.

Section 9.04 Deadline to Bring a Plan Claim, Arbitration or Civil Suit

- (a) Claims regarding any Benefit under the Plan must be submitted in writing within six (6) months of the earlier of:
 - (i) the date a claimant has reason to believe that the Plan has not been properly administered or
 - (ii) the date that a claimant did not receive a Plan benefit to which he believes he was entitled.
- (b) Plan claims and appeals procedures as explained, above must be completely and timely exhausted by a claimant/Participant. A civil lawsuit, arbitration or other proceeding must be filed not later than one (1) year after the exhaustion of these internal Plan remedies as described in this Article IX.

Article X MISCELLANEOUS

Section 10.01 Waiver of Jury Trial

- (a) The Company waives and each Employee upon becoming an Eligible Employee in the Plan shall waive his, her or its right to a jury trial in any court action arising under the Plan or otherwise and whether made by claim, counter-claim, third-party claim or otherwise.
- (b) If for any reason the jury waiver is held to be unenforceable, but only in that event, the Eligible Employee and the Company agree to binding arbitration for any dispute arising out of this Plan or any claim arising under any federal, state or local statutes, laws or regulations, pursuant to the arbitration terms set forth on attached Exhibit B.
- (c) The agreement of the Eligible Employee to waive his right to a jury trial will be binding on his beneficiaries or assigns and will survive the termination of this Plan.
- Section 10.2 Forum Selection . Except as the laws of the United States may otherwise require, any and all disputes relating to or arising under this Policy to enforce or interpret this Policy must be brought by civil action. Any court proceeding brought by you or the Company to enforce rights or obligations related to or arising under any provision of this Plan (a "Claim") must be brought in the United States District Court in Denver, Colorado (the "Court") or, if and only if another proceeding is pending between you and the Company in another jurisdiction and a Claim would be related to such pending proceeding, in such other jurisdiction. Each party agrees to

personal jurisdiction in either venue. Each Participant, by virtue of his or her participation in this Policy, irrevocably consents to the jurisdiction and venue in the Court and that any and all disputes shall be adjudicated solely by the Court, and further irrevocably waives any defense based on lack of venue, personal jurisdiction, *forum non conveniens*, transfer, priority doctrines, and any defense(s) of similar type or import except that, as noted in Section 10.01, if the Court refuses to enforce the jury waiver in Section 10.01, the case shall proceed in arbitration.

Section 10.3 Non-Alienation of Benefits . None of the payments, benefits or rights of any Participant shall be subject to any claim of any creditor of any Participant, and, in particular, to the fullest extent permitted by law, all such payments, benefits and rights shall be free from attachment, garnishment (if permitted under applicable law), trustee's process, or any other legal or equitable process available to any creditor of such Participant. No Eligible Employee shall have the right to alienate, anticipate, commute, plead, encumber or assign any of the benefits or payments that he may expect to receive, continently or otherwise, under this Plan, except for the designation of a beneficiary.

Section 10.04 Notices . All notices and other communications required hereunder shall be in writing and shall be delivered personally or mailed by registered or certified mail, return receipt requested, or by overnight express courier service. In the case of the Eligible Employee, mailed notices shall be addressed to him or her at the home address which he most recently communicated to the Company in writing. In the case of the Company, mailed notices shall be addressed to the Plan Administrator.

Section 10.05 No Mitigation . Eligible Employees shall not be required to mitigate the amount of any Severance Benefit provided for in this Plan by seeking other employment or otherwise, nor shall the amount of any Severance Benefit provided for herein be reduced by any compensation earned by other employment or otherwise, except if the Eligible Employee is re-employed by the Company, in which case Severance Benefits shall cease.

Section 10.06 No Contract of Employment . Neither the establishment of the Plan, nor any modification thereof, nor the creation of any fund, trust or account, nor the payment of any benefits shall be construed as giving any Eligible Employee or any person whosoever, the right to be retained in the service of the Company, and all Eligible Employees shall remain at-will employees, subject to discharge to the same extent as if the Plan had never been adopted.

Section 10.07 Severability of Provisions . Except to the extent provided in Section 6.09, if any provision of this Plan shall be held invalid, illegal or unenforceable by a court of competent jurisdiction, as to such jurisdiction that provision shall be limited ("blue penciled") to the minimum extent necessary so that this Plan shall otherwise remain enforceable in full force and effect. To the extent such provision cannot be so modified, the offending provision shall, as to such jurisdiction, be deemed severable from the remainder of this Agreement, and the remaining provisions contained in this Agreement shall be construed to preserve to the maximum permissible extent the intent and purposes of this Agreement.

Section 10.08 Unfunded Plan . The Plan shall not be funded. No Participant shall have any right to, or interest in, any assets of the Company that may be applied by the Company to the payment of Severance Benefits. Payments of Severance Benefits under the Plan shall be paid from the Company's general assets, in accordance with the terms of the Plan.

Section 10.09 Payments to Incompetent Persons . Any Severance Benefit payable to or for the benefit of a minor, an incompetent person or other person incapable of receipting therefore shall be deemed paid when paid to such person's guardian or to the party providing or reasonably appearing to provide for the care of such person, and such payment shall fully discharge the Company, the Plan Administrator and all other parties with respect thereto.

Section 10.10 Lost Payees . A Severance Benefit shall be deemed forfeited if the Committee is unable to locate an Eligible Employee to whom a Severance Benefit is due. Such Severance Benefit shall be reinstated if application is made by the Eligible Employee for the forfeited Severance Benefit while this Plan is in operation.

Section 10.11 Controlling Law . This Plan shall be construed and enforced according to the laws of the State of Colorado to the extent not superseded by Federal law. Any provision of this Plan that is determined by a court to be in conflict with any applicable Federal or State laws shall be deemed amended by this paragraph to conform to the minimum requirements of such laws, except to the extent they are preempted by ERISA.

Section 10.12 Clerical Error. By participation in this Plan, each Eligible Employee agrees that any amount paid in error will be returned promptly to Company and a payment error does not create a legally binding right to payment under this Plan. If a clerical error or other mistake occurs, however occurring, including but not limited to, by the Plan Administrator, its delegates, members of the Employee Benefits Group of the Company's Human Resources Organization, Vendors, a Participant, such clerical error or mistake does not and shall not create a right to a Benefit under the Plan. When an error is found, it will be corrected or adjusted appropriately as soon as practicable. Interest shall not be payable by the Plan, the Plan Administrator or any delegate with respect to a Benefit corrected or adjusted.

Executed this 13th day of October, 2017

CENTURYLINK, INC.

By: /s/ Scott A. Trezise
Scott A. Trezise, Chair
CenturyLink Plan Design Committee

Title: Executive Vice President, Human Resources

EXHIBIT A

Form of Release Agreement subject to change by the Plan Administrator from time to time

CONFIDENTIAL SEPARATION AGREEMENT AND RELEASE IN FULL OF ALL CLAIMS

For Employees Age 40 or Over

| Work City, State: |
|--|
| Notice to Employee: You should discuss this Confidential Separation Agreement with an attorney prior to signing it. In any event |
| you should thoroughly review and understand the effect of this document before acting upon it. Therefore, please take this |
| Confidential Congretion Agreement home and corofully consider it before you decide whether to sign it. You have 45 calendar days |

Confidential Separation Agreement home and carefully consider it before you decide whether to sign it. You have 45 calendar days from the date of receipt of this Confidential Separation Agreement in which to decide whether to sign and return it.

Capitalized words and phrases used in this document but not defined in this document will be as defined in the CenturyLink Executive Severance Plan (the "Plan").

This Confidential Separation Agreement and Release in Full of All Claims ("Agreement") is made by and between

<Name> (hereinafter referred to as "Employee")

AND

Employee Name:

Employee ID:

CenturyLink, including for the purposes of this Agreement, its parent, affiliates (including, without limitation, EMBARQ Corporation and its subsidiary and affiliate companies), subsidiaries, predecessors, successors, assigns, management companies or any related or parent organizations or entities of which it is or becomes a part, as well as the foregoing entities' respective shareholders, officers, directors, employees, agents, employee benefit plans, successors and assigns (collectively, "Employer").

- Intent to Resolve all Claims. In connection with Employee's separation and termination of employment with Employer at the close of business on <LDW>, Employer and Employee desire to settle and compromise fully and finally all differences between them, including, but not limited to, all claims Employee has or might have asserted against Employer arising out of the employment with Employer or termination of that employment.
- Severance Pay . Employee is entitled to Severance Benefits Payment in accordance with the terms of the Plan, provided all the following have occurred:

- (1) Employee's employment has Terminated, (2) Employee has signed this Agreement on or after his/her last day of work and has returned it to Employer at the address provided in the notification letter, (3) the revocation period described in paragraph 20 has expired and Employee has not revoked the Agreement, and (4) Employee has complied in full with the terms of this Agreement, including but not limited to paragraphs 9 and 10. Employer agrees to pay Employee the gross sum of \$<Amount> less withholdings, as severance pay, 60 days after the Employees separation from employment, in accordance with the terms of the Plan. In addition, Employee is entitled to outplacement assistance in accordance with Section 4.03 of the Plan.
- 3. <u>Payment in Full of All Compensation and Benefits</u>. Employee hereby expressly acknowledges and agrees that (a) Employee has been paid for all work performed and (b) Employee is not entitled to any additional payment of wages (including overtime), compensation, leave, or benefit of any kind from Employer as of the date of separation, except for accrued and unpaid PTO, applicable prorated bonus program payment, if eligible, and additional severance pay payable under the terms of this Agreement. Payments for accrued and unpaid PTO, applicable prorated bonus and severance payments will be made per the provisions described in the separation package that was provided to Employee.
- 4. <u>Termination of Group Health Insurance/Continuation Coverage</u>. Subject to applicable laws, Employee and dependent group health (medical, prescription, dental and vision) insurance coverage, if any, will end on <Benefit End Date>. Employer will provide, in accordance with applicable laws and plan documents, notice of any rights to continue, convert, or obtain (a) replacement health coverage at Employee's own expense, and (b) replacement coverage for any disability and life insurance coverage in effect prior to Employee's separation from employment. Employer will provide Employee, if eligible, the COBRA subsidy under Section 4.03 of the Plan.
- 5. <u>Full and Complete Waiver and Release</u>. Employee hereby accepts Employer's payment of Severance Benefits in accordance with the requirements of Article V of the Plan and promises in this Agreement in full settlement and satisfaction of all grievances, claims, actions and lawsuits of every nature and kind whatsoever, known or unknown, suspected or unsuspected, past, present or future, in any way related to or arising from the employment relationship between Employer and Employee or the termination of that relationship, which Employee has or might have asserted against Employer in the future.

Employee releases, acquits, and forever discharges Employer, and each of its individual directors, officers, employees, agents and insurers, and their successors, and all other persons who might be claimed to be liable, to the fullest extent allowed by law, of and from any and all grievances, claims, actions and lawsuits arising out of or related to Employee's employment with Employer or the termination of that employment, other than as expressly excepted in paragraph 6. Although (as noted in paragraph 6) this Agreement does not limit Employee's right to file a charge with an administrative agency or participate in an agency investigation, Employee waives the right to recover money in connection with any charge or investigation by any agency, regardless of whether Employee or someone else

initiated that charge or investigation. Employee hereby assigns to Employer all rights to such compensation, if any, in consideration of the payments received under this Agreement.

This release includes to the fullest extent allowed by law, but is not limited to, any and all claims, grievances, actions or lawsuits in any forum, which have been, might have been, or in the future might be asserted by Employee and/or on behalf of Employee, under local, state and federal laws, administrative regulations, Executive Orders, and wage payment or equal employment opportunity legislation such as Title VII of the Civil Rights Act of 1964; the Sarbanes-Oxley Act of 2002 (15 USC § 78d-3); the Civil Rights Acts of 1866, 1870, 1871, and 1991; 42 U.S.C. § 1981, the Age Discrimination in Employment Act; the Family and Medical Leave Act; the Older Workers Benefit Protection Act; the Americans with Disabilities Act; the Labor Management Relations Act; the National Labor Relations Act; the Employee Retirement Income Security Act, the Worker Adjustment and Retraining Notification (WARN) Act, 23 U.S.C. § 2101, including WARN Act claims that may arise after the date on which this Agreement became effective, and Executive Order 11246, all as amended. This release also includes, to the fullest extent allowed by law, any and all state or common-law claims, including whistleblower, retaliation, tort and wrongful discharge claims; contract claims, including express or implied contract and breach of the covenant of good faith and fair dealing; and claims for attorneys' fees.

- 6. Exceptions to the Release. Notwithstanding any other provisions of this Agreement, nothing in this Agreement will waive, release, modify, or otherwise affect any of Employee's accrued and vested rights under existing pension and profit-sharing plans, such as claims for vested benefits, or under state workers' compensation and unemployment laws. This Agreement is not intended to change or modify any provision of any benefit plan governed by ERISA. With the exception of claims under the WARN Act, this Agreement does not govern claims or rights based on events that may occur after the date of this Agreement. Also excluded from this Agreement is your right to file a charge with an administrative agency, participate in an agency investigation and to file an unfair labor practice charge. You are, however, waiving all rights to recover money in connection with any such charge, investigation or related lawsuit.
- 7. <u>No Pending Claims</u>. Employee represents that no complaints, grievances, or claims related to Employee's employment with Employer or termination of that employment that are pending or filed against Employer, including any such grievances, claims, or charges filed or pending with any local, state, or federal agency or court as of the date this Agreement was signed
- 8. <u>No Admission of Liability</u>. Nothing in this Agreement shall be interpreted as an admission of liability as to any of the complaints, claims, or lawsuits that it releases. Employer and each of its individual directors, officers, employees, agents and insurers, and their successors, individually and collectively, expressly deny any such liability.
- 9. <u>Agreement to be Bound by Restrictive Covenants in Executive Severance Plan</u>. Employee agrees to abide by all Restrictive Covenants and promises set forth in Sections 6.01 (Code of Conduct), 6.02 (Proprietary Information), 6.03 (Non-Competition),

6.04 (Inducement of Employees, Customers and Others), 6.05 (No Adverse Actions), 6.06 (Return of Property), 6.07 (Non-Disparagement) and 6.08 (Assistance with Claims) of the Plan, the terms of which are incorporated herein by reference. Employee acknowledges that he has received a copy of the Plan and has read and understands Sections 6.01 through 6.08 of the Plan. Employee understands that if Employee breaches the provisions of Sections 6.01 through 6.08, Employer is entitled to recover the relief specified in Section 6.10 of the Plan, including any and all damages caused by such breach.

- 10. <u>Confidentiality of Agreement</u>. With the exception of necessary communications to taxing authorities, accountants, and attorneys; communications to immediate family members who reside with Employee; and any communications required by law, Employee agrees and covenants that the terms, amounts, and fact of this Agreement shall be kept strictly confidential. Employee further agrees not in any way (directly or indirectly) to communicate or disclose, or participate in the communication or disclosure, of any of that information to others (except as may be necessary to enforce the rights contained in this Agreement in an appropriate legal proceeding); including any of Employer's past, present, or future employees or customers, and further agrees and covenants that the same restrictions shall apply with respect to representatives of the media.
- 11. <u>Property of Employee and Employer</u>. Employee has possession of all of Employee's personal property that was on Employer's premises.
- 12. <u>Repayment Requirements</u> Employees rehired by CenturyLink before the expiration of the period covered by their separation allowance will be required to repay all or part of the allowance. The repayment amount will be determined by subtracting the number of days in layoff status from the separation allowance applicable to the number of days included in the separation allowance calculation. The difference is the amount to be repaid.
- 13. <u>Choice of Venue, Jury Trial and Class Action Damage Waiver</u>. Venue for litigating any claims under this Agreement is proper in the court for the Federal District Court for Colorado, and Employee expressly consents to the jurisdiction of such court. Employee expressly waives and relinquishes the right to a trial before a jury in any action, brought in any court, concerning this Agreement or any other claim against Employer. Employee also expressly waives the right to collect money damages in any class or collective action against Employer.
- 14. <u>Severability</u>. To the extent that any provisions in this Agreement are deemed illegal and/or unenforceable, the parties agree that this Agreement shall be interpreted so that all other obligations and protections established by it are enforceable.
- 15. <u>Controlling Law</u>. Except as governed by federal law, this Agreement will be governed by the laws of [Colorado <u>or</u> the state where Employee last worked for Employer].

- 16. <u>Final, Binding and Entire Agreement</u>. This Agreement contains the entire agreement and understanding between the parties related to its subject matter, and supersedes and replaces all prior negotiations and understandings, written or oral, express or implied, concerning that subject matter. This Agreement may be amended only by a written document signed by the parties which specifically states that it was intended as an amendment.
- 17. Acknowledgements. You acknowledge and agree that the severance pay described in paragraph 2 of this Agreement is in addition to whatever you already are entitled to receive apart from this Agreement. You also acknowledge that Employer provided to you, as Exhibit B to the "Waiver of Rights or Claims" letter, a list setting forth: the class, unit or group of individuals covered by this separation or exit incentive program; any eligibility factors for such program; the job titles and ages of all individuals in the same decisional unit eligible or selected for the program; and the ages of all individuals in the same job classification or decisional unit who were not eligible or selected for the program.
- 18. <u>WARN</u>. If Employer paid any payments under WARN, Employee agrees and acknowledges that the method used to determine the rate of pay and benefits is reasonable and fairly compensates Employee for the sixty-day WARN Act period. In addition, Employee agrees and acknowledges that the separation pay is a voluntary and unconditional payment from Employer, which Employer is not legally obliged to provide unless you sign this Agreement, and such pay is sufficient and adequate consideration for your release of any WARN Act claims that you may have either now or in the future as a result of your termination of employment.

19. Consideration and Revocation Periods. Employee hereby acknowledges having been allowed at least forty-five (45) days to consider this Agreement and that if the Agreement is signed sooner, such decision is entirely voluntary. Employee has been advised in writing by Employer to consult an attorney prior to signing this Agreement and to ask the attorney to review and explain this Agreement. Employee has thoroughly reviewed this Agreement and understands this Agreement and the effect of it. Employee understands that Employee may revoke this Agreement for a period of seven (7) days after signing it, and that the Agreement will not become effective or enforceable until the seven-day period has expired without such revocation.

| LEGAL NAME (For Employee Use Only) | For ORGANIZATION NAME (For HR Use Only) |
|---------------------------------------|---|
| Employee Signature | Human Resource Signature |
| Employee Name (Printed) | Human Resource Title |
| | Date of Signature |

EXHIBIT B Arbitration Provision

Any arbitration will be held in the Monroe, Louisiana area (or such other location as may be mutually agreed upon by the Company and the Eligible Employee) and be subject to the Governing Law provision of this Plan and in accordance with the National Rules for the Resolution of Employment Disputes then in effect of the American Arbitration Association, 9 U.S.C. § 1, et seq., before a single arbitrator mutually selected by the Company and the Eligible Employee. The arbitrator should be an attorney, admitted to the practice of law and in good standing in the jurisdiction in which the arbitration is held. Employee expressly waives the right to collect money damages in any class or collective action in arbitration against Employer. Discovery in the arbitration will be governed by the Local Rules applicable in the United States District Court for the Western District of Louisiana. Any award entered by the arbitrator shall be final, binding and non-appealable and judgment may be entered thereon by either party in accordance with applicable law in any court of competent jurisdiction. This arbitration provision shall be specifically enforceable. The arbitrators shall have no authority to modify any provision of this Plan or to award a remedy for a dispute involving this Plan other than a benefit specifically provided under or by virtue of the Plan. Injunctive relief is not available as a remedy. The award must be a reasoned award, providing an analysis of the basis for the decision and not merely a conclusion. Each party shall be responsible for its own expenses relating to the conduct of the arbitration (including reasonable attorneys' fees and expenses) and shall share the fees of the American Arbitration Association and the arbitrators.

EXHIBIT C Lists of Participating and Non-Participating Companies

Participating Companies: Subject to the exceptions described below, CenturyLink, Inc. and its Subsidiaries and affiliates as of December 31, 2016 (including subsidiaries and affiliates of Embarq Corporation and Qwest Communications International, Inc.).

Participating Companies, with limited participation: Level 3 Communications, Inc. and its affiliated or subsidiary companies shall become Participating Companies under the Plan, *only if* the LVLT Corporate Transaction closes, and only as follows:

- **L** Eligibility: A Legacy Level 3 Employee shall become an Eligible Employee upon the closing of the LVLT Corporation Transaction. A Legacy Level 3 Employee shall become eligible for Severance Benefits in accordance with Section 4.01A upon the closing of that transaction, and shall become eligible for Severance Benefits in accordance with Sections 4.01, 4.02 and 4.03 of this Plan upon the first (1 st) anniversary of the closing of that transaction. Between the closing of that Transaction and the first (1 st) anniversary of such closing, a Legacy Level 3 Employee shall be eligible for severance benefits in accordance with this Exhibit C due to:
- An involuntary termination without Cause with a Last Day Worked (LDW) prior to the first (1 st) anniversary of the closing of the LVLT Corporate Transaction.
- A reduction of total targeted compensation (base pay plus short-term incentive opportunity or applicable sales incentive opportunity) of more than 10% that is scheduled to occur at any time, even after the first (1 st) anniversary of Closing Date but which is communicated at any time prior to the first anniversary of Closing Date.
- Relocation provided that (1) the relocation effective date is between Closing Date and the first (1 st) anniversary of Closing Date, and (2) requires a driving distance of more than 50 miles between the work location of the new position and the current position held immediately prior to Closing Date. If an Employee's current position or new position is classified as "work at home" (i.e., with no assigned office at a LVLT facility), his/her home address is or, as applicable, will be deemed his/her work location with respect to such work at home position.
- **II. Benefits:** The Severance Benefits for an eligible Legacy Level 3 Employee are summarized as follows:

Severance - Vice Presidents (who are not KESP Participants)

26 weeks of Base Pay

COBRA Subsidy or Lump Sum COBRA Payment

Not eligible

Bonus or STI (RIFd and not eligible for retirement)

If RIF'd with a last day work in Q4, a pro-rated bonus based on number of days from Jan 1 to term date; adjusted for changes in pay and bonus target; also adjusted for LOAs. To the extent applicable, the Individual Performance Percent will be deemed to be 90% and payout will be adjusted by the Region Multiplier. Paid in March or April following bonus payouts for active employees. Any such pro-rated bonus shall be reduced by the amount, if any, of any pro-rated bonus paid immediately prior to the Effective Time pursuant to the terms of the Merger Agreement, to the extent that the two pro-rated bonuses cover the same period of service.

Bonus or STI (RIFd and eligible for retirement)

A pro-rated bonus based on number of days from Jan 1 to term date; adjusted for changes in pay and bonus target; also adjusted for LOAs. To the extent applicable, the Individual Performance Percent will be deemed to be 100% and payout will be adjusted by the Region Multiplier. Paid in March or April following bonus payouts for active employees. Any such pro-rated bonus shall be reduced by the amount, if any, of any pro-rated bonus paid immediately prior to the Effective Time pursuant to the terms of the Merger Agreement, to the extent that the two pro-rated bonuses cover the same period of service.

Sales Compensation

Paid in accordance with sales comp plan.

Outplacement

6 months (\$2,950 value)

Excise Tax Provision

In the event of a change in control of the Company, if the Company reasonably believes that an employee would receive "excess parachute payments" and be subject to US Federal excise tax pursuant to Sections 280G and 4999 of the Internal Revenue Code of 1986, as amended, the Company may in its sole discretion reduce any payment, benefit, or vesting to such employee by the smallest amount necessary for the employee to avoid the excise tax. Such reduction shall only occur if and to the extent that, on an after-tax basis, the employee would be better with the reduction than the employee would be by paying the excise tax.

III. Benefits: The Severance Benefits for an eligible Legacy Level 3 SVP are as follows:

A Legacy Level 3 SVP who is eligible under the Level 3 Communications, Inc. Key Employee Severance Plan, effective March 19, 2012 (the "KESP") shall remain eligible under the KESP, which is attached as Exhibit E to this Plan and shall become a component of it, until the earlier of (i) the second (2 nd) anniversary of the Closing of the LVLT Corporate Transaction or (ii) the date that he or she waives participation in the KESP. Upon the date he or she no longer participates in the KESP, he or she shall no longer participate in Exhibit E to the Plan, but will participate in all other provisions of this Plan, including the terms of Article IV of this Plan.

Non-Participating Companies: The Companies noted below are not participating in the Plan:

► CenturyLink Marketing Solutions, LLC.

The CenturyLink Plan Design Committee or the Vice President of Legal with responsibility for Severance Benefits, or their respective designee, have the authority to amend Exhibit C at any time and from time to time, including but not

| limited to, add Committee. | ding an e | entity | as a | Participating | Company | as such | participation | is approved | by the | CenturyLink | Plan | Design |
|-------------------------------|-----------|--------|------|---------------|---------|---------|---------------------------------|-------------|--------|-------------|------|--------|
| | | | | | | | | | | | | |
| | | | | | | | | | | | | |
| | | | | | | | EVERANCE PI ffective October | | | | | |
| | | | | | | C-2 | | | | | | |

EXHIBIT D

<u>Period of Subsidized COBRA Coverage for</u> Directors and Vice Presidents Under Section 4.01

| | | Severance (week | as) | | | | |
|---------------------|-----------|-----------------|--------------------|-----------|-------------|--------------------|--|
| Years of Service | Directors | VPs (E1-E2) | VPs (E3 and above) | Directors | VPs (E1-E2) | VPs (E3 and above) | |
| Up to 1 | 12 | 18 | 39 | 3 | 5 | 9 | |
| 2 | 12 | 18 | 39 | 3 | 5 | 9 | |
| 3 | 12 | 18 | 39 | 3 | 5 | 9 | |
| 4 | 12 | 18 | 39 | 3 | 5 | 9 | |
| 5 | 12 | 18 | 39 | 3 | 5 | 9 | |
| 6 | 12 | 18 | 39 | 3 | 5 | 9 | |
| 7 | 14 | 18 | 39 | 4 | 5 | 9 | |
| 8 | 16 | 18 | 39 | 4 | 5 | 9 | |
| 9 | 18 | 18 | 39 | 5 | 5 | 9 | |
| 10 | 20 | 20 | 39 | 5 | 5 | 9 | |
| 11 | 22 | 22 | 39 | 6 | 6 | 9 | |
| 12 | 24 | 24 | 39 | 6 | 6 | 9 | |
| 13 | 26 | 26 | 39 | 6 | 6 | 9 | |
| 14 | 28 | 28 | 39 | 7 | 7 | 9 | |
| 15 | 30 | 30 | 39 | 7 | 7 | 9 | |
| 16 | 32 | 32 | 39 | 8 | 8 | 9 | |
| 17 | 34 | 34 | 40 | 8 | 8 | 9 | |
| 18 | 34 | 36 | 42 | 9 | 9 | 9 | |
| 19 | 34 | 38 | 44 | 9 | 9 | 9 | |
| 20 | 34 | 40 | 46 | 9 | 10 | 9 | |
| 21 | 34 | 42 | 48 | 9 | 10 | 9 | |
| 22 | 34 | 44 | 50 | 9 | 11 | 9 | |
| 23 | 34 | 46 | 52 | 9 | 11 | 12 | |
| 24 | 34 | 48 | 52 | 9 | 12 | 12 | |
| 25 | 34 | 50 | 52 | 9 | 12 | 12 | |
| 26 | 34 | 52 | 52 | 9 | 12 | 12 | |
| 27+ | 34 | 52 | 52 | 9 | 12 | 12 | |

Executive Officers and the CEO are eligible for a COBRA subsidy of up to one year in the event they are eligible for severance under Section 4.01, regardless of their years of service.

Period of Subsidized COBRA coverage for Lower Performance Severance Under Section 4.01A

| | Severan | ice (weeks) | COBRA (months) | | |
|--------------------------|-----------|-------------|----------------|-------------|--|
| Years of Services | Directors | VPs (E1-E2) | Directors | VPs (E1-E2) | |
| 1 | 2 | 2 | 1 | 1 | |
| 2 | 2 | 2 | 1 | 1 | |
| 3 | 3 | 3 | 1 | 1 | |
| 4 | 4 | 4 | 1 | 1 | |
| 5 | 5 | 5 | 2 | 2 | |
| 6 | 6 | 6 | 2 | 2 | |
| 7 | 7 | 7 | 2 | 2 | |
| 8 or more | 8 | 8 | 2 | 2 | |

Period of Subsidized COBRA Coverage for Directors and Vice Presidents at E1 or above Under Section 4.02

| Directors | | | | |
|--------------------|------------------------------------|--|--|--|
| Weeks of Severance | Maximum Months of COBRA Subsidy | | | |
| 26-38 | 6 | | | |
| 39-51 | 9 | | | |
| 52 | 12 | | | |

Vice-Presidents at E1 or above, excluding Executive Officers and the CEO, are eligible for a COBRA subsidy of up to one year in the event they are eligible for severance under Section 4.02, regardless of their years of service.

Exhibit E

LEVEL 3 COMMUNICATIONS, INC. KEY EXECUTIVE SEVERANCE PLAN

Article I - Purpose

The purpose of the Level 3 Communications, Inc. Key Executive Severance Plan (the "Plan") is to provide eligible executives of Level 3 Communications, Inc., a Delaware corporation (the "Company"), who are involuntarily terminated from employment in certain limited circumstances, with severance and welfare benefits as set forth in the Plan.

Article II - Definitions

As used in the Plan, the following terms shall have the respective meanings set forth below, and, when the meaning is intended, such terms are capitalized:

"Accrued Obligations" means, with respect to a Participant, the sum of (A) the Participant's Base Salary through the Date of Termination to the extent not theretofore paid; (B) the Participant's business expenses that are reimbursable in accordance with the Company's policies and for which the Participant submits for reimbursement within thirty (30) days following the Date of Termination, but have not been reimbursed by the Company as of the Date of Termination and (C) to the extent not yet paid as of the Date of Termination, the Participant's annual bonus, if any, for the fiscal year immediately preceding the fiscal year in which the Date of Termination occurs. Notwithstanding the foregoing, if the Participant has made an irrevocable election under any nonqualified deferred compensation plan subject to Section 409A of the Code to defer any portion of the Base Salary or annual bonus described in clause (A) or (B) above, then for all purposes of this Section, such deferral election, and the terms of the applicable arrangement, shall apply to the same portion of the amount described in such clause (A) or (B), and such portion shall not be considered as part of the Accrued Obligations but shall instead be paid in accordance with Section 4.7.

- "Adverse Amendment" has the meaning ascribed to that term in Section 8.5.
- "Affiliate" means any direct or indirect parent entity of the Company that has majority control over the Company, and each direct or indirect subsidiary thereof.
- "Applicable COBRA Premium" means, with respect to a Participant, the total monthly cost of continuation coverage under COBRA, as of the Participant's Date of Termination, in respect of the maximum level of coverage in effect for the Participant and the Participant's spouse and dependents at the Date of Termination.

- "Base Salary" means the Participant's annual rate of base salary in effect immediately prior to the occurrence of the facts, circumstance or reasons giving rise to the Participant's Termination of Employment, or, if greater, the Participant's annual rate of base salary in effect immediately prior to a Change in Control.
- "Board" means the Board of Directors of the Company and, after a Change in Control, the board of directors of the ultimate parent entity that controls either the Company or the company into which the Company was merged or consolidated.
- "Cash Severance Benefit" means the cash severance benefit applicable to a Participant as set forth in such Participant's Participation Notice.
- "Cause" means (i) the willful and continued failure by a Participant to substantially perform his or her duties with the Company and its Affiliates (other than any such failure resulting from his or her incapacity due to physical or mental impairment, or any such actual or anticipated failure after the issuance of a notice of termination by him or her for Good Reason) after a written demand for substantial performance is delivered to the Participant by the Company, which demand specifically identifies the manner in which he or she has not substantially performed his or her duties; (ii) the willful engagement by a Participant in conduct that is demonstrably and materially injurious to the Company or any of its Affiliates, monetarily or otherwise; (iii) a Participant's indictment for, conviction of or plea of guilty or no contest to, any felony or (iv) a Participant's violation of the terms of such Participant's Restrictive Covenant Agreement; provided, however, that to the extent that any act or failure to act otherwise constituting Cause hereunder is curable, such Participant shall be given not less than ten (10) days' written notice by the Company's Chief Executive Officer (or in the case of a Participant who is (or was at any time while a Participant) a "named executive officer" (within the meaning of Item 402 of Regulation S-K issued under the Exchange Act) of the Company, the Chief Executive Officer or the Board) of the Company's intention to terminate him or her with Cause. Such notice of a termination with Cause shall state in detail the grounds on which the proposed termination with Cause is based, and a termination with Cause shall be effective at the expiration of such ten (10) day notice period unless the Participant has fully cured during such period such act or failure to act that gives rise to Cause.

For purposes of this definition, no act, or failure to act, on a Participant's part shall be deemed "willful" unless done, or omitted to be done, by the Participant not in good faith and without reasonable belief that his or her action or omission was in the best interests of the Company and its Affiliates. Any act, or failure to act, based upon (A) the lawful instruction or direction of the Board, (B) the lawful instruction of the Chief Executive Officer of the Company or the Participant's direct supervisor or (C) the advice of counsel for the Company shall be conclusively presumed to be done, or omitted to be done, by the Participant in good faith and in the best interests of the Company. Following a Change in Control, a Participant shall not be deemed to have terminated employment for Cause unless and until there shall have been delivered to the Participant a copy of a resolution duly adopted by the affirmative vote of not fewer than three quarters of the entire membership of the Board (excluding the Participant, if the Participant is a member of the Board) at a meeting of the Board called and held for such

purpose (after reasonable notice is provided to the Participant and the Participant is given an opportunity, together with counsel for the Participant, to be heard before the Board), finding that, in the good faith opinion of the Board, the Participant is guilty of the conduct constituting Cause, and specifying the particulars thereof in detail.

"Change in Control" means the occurrence of any of the following events:

- (a) a change in ownership or control of the Company effected through a transaction or series of related transactions (other than an offering of common stock of the Company to the general public through a registration statement filed with the Securities and Exchange Commission) whereby any "person" (as defined in Section 3(a)(9) of the Exchange Act) or any two or more persons deemed to be one "person" (as used in Sections 13(d)(3) and 14(d)(2) of the Exchange Act), other than the Company or any of its Affiliates, or an employee benefit plan maintained by the Company or any of its Affiliates, directly or indirectly acquire, other than pursuant to an acquisition from the Company, "beneficial ownership" (within the meaning of Rule 13d-3 under the Exchange Act) of securities of the Company possessing more than fifty percent (50%) of the total combined voting power of the Company's securities outstanding immediately after such acquisition; or
- (b) the date upon which individuals who, as of the Effective Date, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board; provided, however, that any individual becoming a director subsequent to the Effective Date whose election, or nomination for election by the Company's stockholders, was approved by a vote of at least a majority of the directors then constituting the Incumbent Board shall be considered as though such individual were a member of the Incumbent Board since the Effective Date, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of an actual or threatened election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board or
- (c) the sale or disposition, in one or a series of related transactions, of all or substantially all of the assets of the Company to any "person" (as defined in Section 3(a)(9) of the Exchange Act) or to any two or more persons deemed to be one "person" (as used in Sections 13(d)(3) and 14(d)(2) of the Exchange Act) other than the Company's Affiliates.

Notwithstanding the foregoing, for all purposes of the Plan other than Section 4.5, a Change in Control shall also include "change in control event" as described in Treasury Regulation Section 1.409A-3(i)(5) with respect to any Participant that is employed at an Affiliate of the Company at such time that such Affiliate incurs such change in control event.

"COBRA" means the continuation coverage requirements under Section 4980B of the Code and Part 6 of Title I of ERISA or any similar provision under other applicable law.

"Code" means the Internal Revenue Code of 1986, as amended.

- "Committee" means the Compensation Committee of the Board or a committee or committees of the Board appointed by the Board to administer the Plan.
- "<u>Date of Termination</u>" means the date of the Participant's Termination of Employment with the Company and its Affiliates as determined under Section 4.1 of the Plan.
 - "Effective Date" means March 19, 2012.
 - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
 - "Exchange Act" means the Securities Exchange Act of 1934, as amended.
 - "Good Reason" means, without the Participant's express written consent, the occurrence of any of the following events:
 - (i) a material diminution in the Participant's authority, duties, responsibilities or reporting requirements, and following a Change in Control, the assignment to the Participant of any duties inconsistent in any respect with the Participant's position (including status, offices, titles and reporting requirements), or any action by the Board, the Company or any of its Affiliates that results in a diminution in such position, authority, duties or responsibilities, excluding for this purpose an isolated, insubstantial, inadvertent and immaterial action not taken in bad faith and that is remedied promptly after receipt of notice thereof given by the Participant;
 - (ii) a reduction in the Participant's Base Salary or target short-term incentive award opportunity, except, prior to a Change in Control, an across-the-board reduction that similarly affects all similarly situated employees of the Company and its Affiliates;
 - (iii) the relocation of the Participant's principal place of business to a location that is outside the 50-mile radius from the Participant's then-current principal place of business or, following a Change in Control, a requirement imposed by the Company or its Affiliate that the Participant be based at any office or location other than the office or location where the Participant was employed immediately preceding the Change in Control;
 - (iv) following a Change in Control, the failure to continue to provide the Participant with employee benefits substantially similar to those enjoyed by him or her under the pension, life insurance, medical, health, accident and disability plans, or any retirement or fringe benefit material to the Participant for which he or she was eligible at the time of the Change in Control;
 - (v) the failure to obtain a satisfactory agreement from any successor to the Company or any acquiror of any Affiliate or division of the Company to assume and agree to perform the Plan pursuant to Section 7.1 herein;

- (vi) the adoption of an Adverse Amendment except as permitted under Section 8.5 herein; and
- (vii) any purported termination of the Participant's employment that is not effected pursuant to a notice of termination satisfying the requirements of Section 4.1 herein.

A Participant may terminate his or her employment with Good Reason by providing the Company thirty (30) days' written notice setting forth in reasonable specificity the event that constitutes Good Reason, which written notice, to be effective, must be provided to the Company within ninety (90) days following the initial occurrence of such event. During such thirty (30) day notice period, the Company shall have a cure right (if curable), and if not cured within such period, the Participant's termination will be effective upon the expiration of such cure period. A Participant's right to terminate his or her employment for Good Reason shall not be affected by his or her incapacity due to physical or mental impairment. A Participant's continued employment shall not constitute consent to, or a waiver of rights with respect to, any circumstance constituting Good Reason hereunder.

- "Participant" means each employee of the Company or its Affiliates, other than the Chief Executive Officer of the Company, who is designated as a Participant by the Chief Executive Officer of the Company and delivered a Participation Notice and who acknowledges his or her participation in the Plan by executing a Restrictive Covenant Agreement. A person shall cease to be a Participant upon his or her removal as a Participant from the Plan (but any such determination made in respect of a Participant shall be considered an Adverse Amendment with respect to the affected Participant and is subject to the provisions of Section 8.5).
- "Participation Notice" means the notice provided to an employee of the Company or its Affiliates that designates such individual as a Participant in the Plan and the terms and conditions of such individual's participation in the Plan (subject to the Participant's acknowledgement by executing a Restrictive Covenant Agreement), which notice shall be substantially in the form set forth on Exhibit A.
- "Potential Change in Control" shall be deemed to have occurred if (i) the Company enters into an agreement, the consummation of which would result in the occurrence of a Change in Control, or (ii) the Committee adopts a resolution to the effect that, for purposes of the Plan, a Potential Change in Control of the Company has occurred.
- "Prorated Bonus" means, with respect to a Participant, an amount equal to the product of (A) the most recent Target Bonus and (B) a fraction, the numerator of which is the number of days in the current fiscal year through the Date of Termination and the denominator of which is 365.
- "Qualifying Termination" means a Participant's Termination of Employment (i) by the Company other than for Cause or (ii) by the Participant for Good Reason. A Termination of

Employment for any other reason, including by reason of death, disability or retirement, shall not be treated as a Qualifying Termination.

- "Restrictive Covenant Agreement" means an agreement substantially in the form attached hereto as Exhibit B.
- <u>"Severance Multiplier"</u> means the severance multiplier applicable to a Participant as set forth in such Participant's Participation Notice.
- "Specified Employee" means any Participant who is a "Key Employee" (as defined in Code Section 416(i) without regard to paragraph (5) thereof), as determined by the Company in accordance with its uniform policy with respect to all arrangements subject to Section 409A of the Code, based upon the twelve (12) month period ending on each December 31st (such twelve (12) month period is referred to below as the "identification period"). All Participants who are determined to be key employees under Section 416(i) of the Code (without regard to paragraph (5) thereof) during the identification period shall be treated as Specified Employees for purposes of the Plan during the twelve (12) month period that begins on the first day of the fourth (4th) month following the close of such identification period.
- <u>"Target Bonus"</u> means a Participant's target annual bonus for the fiscal year in which the Change in Control occurs, if applicable, or in which a Participant's Date of Termination occurs, whichever is greater; provided that if a target annual bonus has not been established for the applicable fiscal year, then the term <u>"Target Bonus"</u> shall refer to the Participant's target annual bonus established for the immediately preceding fiscal year.
- <u>"Termination of Employment"</u> means the event where the Participant has a "separation from service," as defined under Section 409A of the Code, with the Company and its Affiliates.

Article III - Effectiveness of the Plan

This Plan shall be effective as of the Effective Date. Nothing in the Plan shall be deemed to entitle any Participant to continued employment with the Company or any Affiliate of the Company.

Article IV - Payments Upon a Qualifying Termination; Change in Control Benefits 4.1. <u>Termination of Employment</u>.

Any purported termination of a Participant's employment by the Company or its Affiliates or by a Participant (including notice required for purposes of a Participant to terminate his or her employment with Good Reason) shall be communicated by written notice of termination to the other party in accordance with this Section 4.1 and Section 8.1 (regarding notices). For

purposes of the Plan, a "notice of termination" shall mean a notice indicating the specific termination provision in the Plan relied upon and setting forth in reasonable detail the

- (a) facts and circumstances providing a basis for the Participant's Termination of Employment under the provision so indicated (including such information required herein for purposes of a Participant to terminate his or her employment with Good Reason). The failure by the Participant or the Company to set forth in such notice of termination any fact or circumstance that contributes to a showing of Good Reason or Cause shall not waive any right of the Participant or the Company hereunder or preclude the Participant or the Company from asserting such fact or circumstance in enforcing the Participant's or the Company's rights hereunder.
- (b) If a Participant has a Qualifying Termination, the Date of Termination shall be the date specified in the notice of termination (which, in the case of a termination other than for Cause or a termination for Good Reason, shall not be fewer than five (5) nor more than twenty (20) business days from the date such notice is given). If a Participant's Termination of Employment is for Cause, the Date of Termination shall be as set forth in the definition of "Cause" in Article II hereunder.
- **4.2.** <u>Severance Payments</u>. If the Participant has a Qualifying Termination, then, subject to Section 4.3, the Company shall or shall cause an Affiliate, to provide the Participant with the following:
 - (a) Accrued Obligations the
 - (b) Prorated Bonus, payable either (x) with respect to a Qualifying Termination occurring prior to a Change in Control, in two installments—fifty percent (50%) on the sixtieth (60th) day following such Qualifying Termination and fifty percent (50%) on the first business day of the seventh calendar month following the month in which the Date of Termination occurred, or (y) with respect to a Qualifying Termination occurring on or after a Change in Control, in a single lump sum on the sixtieth (60th) day following such Qualifying Termination;
 - (c) a lump sum payment in an amount equal to the Cash Severance Benefit;
- (c) continued coverage for such Participant and his or her spouse and dependents in all medical and dental insurance plans maintained by the Company and its Affiliates for the most senior officers of any of the Company and its Affiliates (collectively, the "Continued Benefit Plans") for a period of months equal to the greater of (x) the Participant's Severance Multiplier multiplied by twelve (12) (such period, the "Contractual Coverage Period") and (y) the length of period during which the Participant and his or her spouse or domestic partner and dependents are eligible to elect continuation coverage under the Continued Benefit Plan pursuant to COBRA; provided that such coverage shall run concurrently with the period provided under COBRA, and the cost of such Continued Benefits shall be paid by the Participant on an after-tax basis; and provided further that such coverage shall cease earlier than the expiration of continued coverage period upon any date that the Participant becomes eligible to receive medical and dental insurance through a new employer;

- (d) a lump sum cash payment equal to (x) the Applicable COBRA Premium multiplied by the total number of months in the Contractual Coverage Period (without regard to any early termination of such period pursuant to subsection (d) above) minus (y) the monthly employee contribution rate that is paid by Company employees generally for the same or similar coverage, as in effect on the Date of Termination (and which amount shall in no event be greater than the employee contribution rate for the applicable level of coverage as in effect immediately prior to the Date of Termination);
- (e) reimbursement for the cost of outplacement services (the scope and provider of which shall be selected by the Participant in the Participant's sole discretion) incurred by the Participant prior to the last day of the second calendar year following the year in which the Date of Termination occurred; provided that the cost of such outplacement shall not exceed \$10,000 and the Participant shall be reimbursed for the cost of outplacement services promptly upon submission of reasonable substantiation of such costs but not later than the third calendar year following the calendar year in which the Date of Termination occurred;
 - (f) vesting of outstanding equity-based awards as described below:
 - (x) vesting of that portion of such Participant's equity-based awards (other than performance-based restricted stock units ("PRSUs"), which are covered in paragraph (y) immediately following) (whether granted under the Level 3 Communications, Inc. Stock Plan or otherwise) that would have vested, but for such Termination of Employment, based on the Participant's continued employment with the Company or any of its Affiliates within the twelve (12) months following such Termination of Employment (without regard to any subsequent acceleration events), and (y) vesting of a prorated portion of all outstanding PRSUs through the Termination of Employment (measured in each instance to the last day of the calendar month in which the Termination of Employment occurs) based on the percentage of time that the Participant was employed during the "Performance Period" applicable to each such outstanding PRSU (the "Pro Rata Portion"). The actual number of PRSUs that will vest pursuant to this paragraph will be the product of (i) the Pro Rata Portion, (ii) the number of PRSUs subject to the Award and (iii) the applicable performance percentage applicable to such PRSU (as determined pursuant to the terms of the applicable Award Letter) based on the actual achievement of the applicable Performance Objective during the entire Performance Period.

Notwithstanding the foregoing, with respect to a Termination of Employment occurring on or after the date on which such Participant becomes retirement-eligible in accordance with the Company's retirement plan or policy then in effect, all of such Participant's equity-based awards granted prior to April 1, 2014 (whether granted under the Level 3 Communications, Inc. Stock Plan

or otherwise) that vest based on the Participant's continued employment with the Company or any of its Affiliates shall vest on the date of Termination of Employment. Any outperform stock

appreciation rights held by the Participant ("OSOs") that become vested pursuant to this paragraph (g) shall be settled on the Settlement Date (as defined in and set forth in the applicable OSO award agreement). Any PRSUs for which additional vesting occurs on account of this paragraph (g) shall be settled as soon as possible after the determination of the satisfaction of the Performance Objective set forth in each applicable award letter. This paragraph (g) shall apply to a given award only to the extent that it provides better treatment (from the Participant's perspective) than provided by the applicable plan, award letter or agreement or other agreement between the Company and the Participant that governs such award. For the avoidance of doubt, the additional vesting set forth in this paragraph (g) shall only apply to the vesting conditions that are based on the Participant's continued employment with the Company or any of its Affiliates through a specified date and shall not apply to any performance-based vesting conditions, which shall continue to apply in accordance with the terms of the applicable award agreement; and

- (h) except as otherwise set forth below, to the extent not theretofore paid or provided, any benefits under any employee benefit plan, agreement, arrangement or policy in accordance with the terms of the underlying plan, agreement, arrangement or policy, other than plans, agreements, arrangements or policies providing for severance benefits.
- **4.3.** Conditions to Receipt of Severance Benefits. As a condition to receipt of any payment or benefits under Section 4.2, the Participant must execute, deliver to the Company and not revoke a Release Agreement substantially in the form attached hereto as Exhibit C (and any revocation period contained in such Release Agreement shall have expired) within sixty (60) days following the Participant's Date of Termination. In addition, such payments and benefits shall immediately terminate, and the Company and its Affiliates shall have no further obligations to a Participant with respect thereto, in the event that such Participant breaches any provision of his or her Restrictive Covenant Agreement.
- **4.4.** <u>Timing of Payment</u>. Subject to satisfaction of the conditions set forth in Section 4.3 and Article VI, the amounts payable pursuant to Section 4.2(a), 4.2(c) and 4.2(e) with respect to a Participant shall be made on the sixtieth (60th) day following the Participant's Date of Termination or, if such day does not fall on a business day, the first business day thereafter.
- **4.5.** Change in Control Benefits . Except as otherwise provided under an award agreement governing an award outstanding as of the Effective Date, upon a Change in Control, regardless of whether a Participant experiences a Termination of Employment, all equity-based awards granted prior to April 1, 2014 (whether granted under the Level 3 Communications, Inc. Stock Plan or otherwise) and held by each Participant shall vest; provided, however, that any OSOs that become vested pursuant to this Section 4.5 shall be settled on the Settlement Date (as defined in and set forth in the applicable OSO award agreement), and awards that vest based on the satisfaction of performance criteria (which, for the avoidance of doubt, shall not include OSOs) shall vest pursuant to this Section 4.5 only to the extent that such performance criteria are satisfied based on pro-forma

performance over the entire applicable performance period extrapolated from the first day of the applicable performance period through the date of the Change in Control. Nothing in this Section

4.5 shall affect the vesting provisions contained in agreements and/or award letters applicable to any equity based awards granted to Participant after April 1, 2014.

Notwithstanding the foregoing, this Section 4.5 shall apply to a given equity-based award only to the extent that it provides better treatment than provided by the applicable plan, award agreement or other agreement between the Company and the Participant that governs such award.

- **4.6.** No Duplication of Benefits . Except as otherwise expressly provided pursuant to the Plan, the Plan shall be construed and administered in a manner that avoids duplication of compensation and benefits (including payments made in lieu of notice of termination under law or applicable contract) that may be provided under any other plan, program, policy or other arrangement or individual contract with or provided by the Company or any Affiliates. In the event that a Participant is covered by any other plan, program, policy or other arrangement or individual contract in effect as of his or her Date of Termination, that may duplicate the payments and benefits provided for in Section 4.2(a) through (g), the Company may reduce or eliminate the duplicative benefits or compensation provided for under the Plan or under the contract or law that provides such benefit (specifically, but without limiting the foregoing, any payments made in lieu of notice shall be deducted from the Cash Severance Benefit), but solely to the extent that such reduction or elimination does not cause the Participant to be subject to penalty taxes under Section 409A of the Code.
- **4.7.** No Effect on Other Benefits. This Plan does not abrogate any of the usual entitlements which a Participant has or will have, first, while a regular employee, and subsequently, after termination, and a Participant shall be entitled to receive all benefits payable to him or her under each and every "employee benefit plan" (as defined under Section 3(2) of ERISA, whether or not subject to ERISA) and deriving from his or her employment with the Company and its Affiliates, but solely in accordance with the terms and provisions thereof.
- 4.8 Modified Cutback . If any payment, benefit, or distribution of any type to or for the benefit of any Participant, whether paid or payable, provided or to be provided, or distributed or distributable pursuant to the terms of this Plan or otherwise (collectively, the "Parachute Payments") would subject the Participant to the excise tax (the "Excise Tax") imposed under Section 4999 of the Internal Revenue Code of 1986, as amended (the "Code"), the Parachute Payments shall be reduced so that the maximum amount of the Parachute Payments (after reduction) shall be one dollar (\$1.00) less than the amount which would cause the Parachute Payments to be subject to the Excise Tax; provided, that the Parachute Payments shall only be reduced to the extent the after-tax value of amounts received by the Participant after application of the above reduction would exceed the after-tax value of the amounts received without application of such reduction. For this purpose, the after-tax value of an amount shall be determined taking into account all federal, state, and local income, employment and excise taxes applicable to such amount. If a

reduction in the payments to a Participant is required, then the reduction shall occur in the following order: (i) first by reducing or eliminating any cash

CENTURYLINK EXECUTIVE SEVERANCE PLAN As Updated, Amended and Restated Effective October 10, 2017

E-10

severance payments (with the payments to be made furthest in the future being reduced first), (ii) then by reducing or eliminating any accelerated vesting of restricted stock or similar awards (including any restricted stock units), (iii) then by reducing or eliminating any other remaining Parachute Payments; *provided*, that no such reduction or elimination shall apply to any nonqualified deferred compensation amounts (within the meaning of Section 409A (" *Section 409A*") of the Code) to the extent such reduction or elimination would accelerate or defer the timing of such payment in manner that does not comply with Section 409A; and, *further*, *provided*, that to the extent permitted by Sections 409A and Sections 280G and 4999 of the Code, if a different reduction procedure would be permitted without violating Section 409A or losing the benefit of the reduction under Sections 280G and 4999 of the Code, the Participant may designate a different order of reduction.

- (a) An initial determination as to whether (x) any of the Parachute Payments received by a Participant in connection with the occurrence of Change in Control shall be subject to the Excise Tax, and (y) the amount of any reduction, if any, that may be required pursuant to this Section 4.8, shall be made by an independent accounting, valuation or compensation advisory firm selected by the Company (the "280G Firm") prior to the consummation of such change in the ownership or effective control of the Company or in the ownership of a substantial portion of the assets of the Company. Promptly upon request, each Participant shall be furnished with notice of all determinations made as to the Excise Tax payable with respect to such Participant's Parachute Payments, together with the related calculations of the 280G Firm.
 - (b) For purposes of this Section 4.8–
- (i) no portion of the Parachute Payments, the receipt or enjoyment of which a Participant shall have effectively waived in writing prior to the date of payment of the Parachute Payments, shall be taken into account;
- (ii) no portion of the Parachute Payments shall be taken into account which in the opinion of the 280G Firm does not constitute a "parachute payment" within the meaning of Section 280G(b)(2) of the Code;
- (iii) the Parachute Payments shall be reduced only to the extent necessary so that the Parachute Payments (other than those referred to in the immediately preceding clause (i) or (ii)) in their entirety constitute reasonable compensation for services actually rendered within the meaning of Section 280G(b)(4) of the Code or are otherwise not subject to disallowance as deductions, in the opinion of the 280G Firm or tax counsel referred to in such clause (ii); and (iv) the value of any non-cash benefit or any deferred payment or benefit included in the Parachute Payments shall be determined by the 280G Firm based on Sections 280G and 4999 of the Code, or on substantial authority within the meaning of Section 6662 of the Code.

Article V - Withholding Taxes

The Company and its Affiliates may withhold from all payments due to the Participant (or his beneficiary or estate) hereunder all taxes which, by applicable federal, state, local or other law, are required to be withheld.

Article VI - Certain Additional Agreements under Section 409A

6.1. Delay of Payment . In the event that a payment to be made pursuant to

Section 4.2 or any other amount under the Plan that constitutes nonqualified deferred compensation subject to Section 409A of the Code is to be made to a Specified Employee, such payment will be delayed for six (6) months after the Date of Termination and paid in a single lump sum on the first business day of the month following the end of such six (6) month period. If a Participant who is a Specified Employee dies within six (6) months following such Date of Termination, any such delayed payments shall not be further delayed, and shall be immediately payable within thirty (30) days to his or her estate in accordance with the applicable provisions of the Plan.

6.2. Section 409A. The Plan is intended to comply with the requirements of Section 409A of the Code or an exemption or exclusion therefrom and, with respect to amounts that are subject to Section 409A of the Code, shall in all respects be administered in accordance with Section 409A of the Code. Each payment under the Plan shall be treated as a separate payment for purposes of Section 409A of the Code. In no event may any Participant, directly or

indirectly, designate the calendar year of any payment to be made under the Plan. Notwithstanding any other provision of the Plan to the contrary, with respect to any Participant's equity-based award that constitutes nonqualified deferred compensation subject to Section 409A of the Code ("NQDC Award"), to the extent the Plan provides that upon a Change in Control the Participant is to become vested in, and such NQDC Award to paid or settled (to the extent vested), as applicable, the Participant shall become so vested in the NQDC Award upon such Change in Control but payment or settlement, as applicable, of the NQDC Award shall not be paid or be settled (and shall instead be paid or settled as of the event or date otherwise provided by its terms) unless the Change in Control constitutes a "change in control event" as described in Treasury Regulation Section 1.409A-3(i)(5). All reimbursements and in-kind benefits provided under the Plan that constitute nonqualified deferred compensation subject to Section 409A of the Code shall be made or provided in accordance with the requirements of Section 409A of the Code, including, without limitation, that (i) in no event shall reimbursements by the Company under the Plan be made later than the end of the calendar year next following the calendar year in which the applicable fees and expenses were incurred; provided, that the Participant shall have submitted an invoice for such fees and expenses at least ten (10) days before the end of the calendar year next following the calendar year in which such fees and expenses were incurred; (ii) the amount of in-kind benefits that the Company is obligated to pay or provide in any given

calendar year (other than medical reimbursements described in Treas. Reg. § 1.409A-3(i)(1)(iv)(B)) shall not affect the in-kind benefits that the Company is obligated to pay or provide in any other calendar year; (iii) the Participant's right to have the Company pay or provide such reimbursements and in-kind benefits may not be liquidated or exchanged for any other benefit and (iv) in no event shall the Company's obligations to make such reimbursements or to provide such in-kind benefits apply later than the Participant's remaining lifetime (or if longer, through the twentieth (20th) anniversary of the Date of Termination). While the payments and benefits provided hereunder are intended to be structured in a manner to avoid the implication of any penalty taxes under Section 409A of the Code, in no event whatsoever shall the Company or any of its Affiliates be liable for any additional tax, interest or penalties that may be imposed on a Participant as a result of Section 409A of the Code or any damages for failing to comply with Section 409A of the Code (other than for withholding obligations or other obligations applicable to employers, if any, under Section 409A of the Code).

6.3. No Adverse Action . Neither the Company nor any Affiliate of the Company shall take any action that would expose any payment or benefit to a Participant under the Plan to the additional tax imposed under Section 409A of the Code unless (i) the Company is obligated to take the action under an agreement, plan or arrangement, (ii) a Participant requests the action, (iii) the Company advises such Participant in writing that the action may result in the imposition of the additional tax and (iv) such Participant subsequently requests the action in a writing that acknowledges that he or she will be responsible for any effect of the action under Section 409A.

Article VII - Successors; Binding Agreement

7.1. The Company will require any successor (whether direct or indirect, by purchase, merger, consolidation, by operation of law or otherwise) to all or substantially all of the business or assets of the Company to unconditionally assume all of the obligations of the Company hereunder. Further, the Company will require each Affiliate or the acquiror of each Affiliate or division that employs a Participant and that ceases to be an Affiliate or division of the Company to establish and maintain a plan that is identical, in all material respects, to the Plan, and designate the Participant as a participant in such plan until the second anniversary of the divestiture of such Affiliate or division. Failure of the Company to obtain such assumption prior to the effectiveness of any such succession shall constitute Good Reason hereunder and shall entitle the Participants to compensation and other benefits in the same amount and on the same terms as the Participants would be entitled hereunder if they had a Qualifying Termination, except that for purposes of implementing the foregoing, the date on which any succession becomes effective shall be deemed the Date of Termination.

7.2. The benefits provided under the Plan shall inure to the benefit of and be enforceable by the Participant's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees. If the Participant shall die while any amounts would be payable

| n |
|---|
| |
| |
| _ |
| |
| |
| _ |

the Plan to such person or persons appointed in writing by the Participant to receive such amounts or, if no person is so appointed, to the Participant's estate.

Article VIII - Miscellaneous

8.1. <u>Notices</u>. All notices and other communications hereunder shall be in writing and shall be given by hand delivery to the other party, by courier, by email to the Participant's email address at the Company (if the Participant is employed by the Company at the time of delivery of the notice), or by mail, postage prepaid, addressed as follows:

if to the Participant:

At the most recent address on file at the Company.

if to the Company:

Level 3 Communications, Inc. 1025 Eldorado Blvd.
Broomfield, CO 80021

Attention: Chief Legal Officer

or to such other address as either party shall have furnished to the other in writing in accordance herewith. Notices and communications shall be effective when actually received by the addressee.

Notice to the Participant shall be deemed given when mailed or given by hand delivery. 8.2. No Setoff; No Mitigation;

Resolution of Disputes and Costs.

(a) The Company's obligation to make the payments provided for in the Plan and otherwise to perform its obligations hereunder shall not be affected by any set-off, counterclaim, recoupment, defense or other claim, right or action that the Company may have against the Participant or others. In no event shall a Participant be obligated to seek other employment or take other action by way of mitigation of the amounts payable to the Participant under any of the provisions of the Plan and such amounts shall not be reduced whether or not the Participant obtains other employment.

Participants may submit claims for benefits by giving notice to the Company pursuant to Section 8.1. If a Participant believes that he or she has not received coverage or benefits to which he or she is entitled under the Plan, the Participant may notify the Company in writing of a claim for coverage or benefits. If the claim for coverage or benefits is denied in whole

- or in part, the Company shall notify the applicant in writing of such denial within thirty (30) days of such claim (which may be extended to sixty (60) days under special circumstances), with such notice setting forth (i) the specific reasons for the denial; (ii) the Plan provisions upon which the denial is based; (iii) any additional material or information necessary for the applicant to perfect his or her claim and (iv) the procedures for requesting a review of the denial. Upon a denial of a claim by the Company, the Participant may (i) request a review of the denial by the Committee or, where review authority has been so delegated, by such other person or entity as may be designated by the Committee for this purpose; (ii) review any Plan documents relevant to his or her claim and (iii) submit issues and comments to the Committee or its delegate that are relevant to the review. Any request for review must be made in writing and received by the Committee or its delegate within sixty (60) days of the date the applicant received notice of the initial denial, unless special circumstances require an extension of time for processing. The Committee or its delegate will make a written ruling on the applicant's request for review setting forth the reasons for the decision and the Plan provisions upon which the denial, if appropriate, is based. This written ruling shall be made within thirty (30) days of the date the Committee or its delegate receives the applicant's request for review unless special circumstances require an extension of time for processing, in which case a decision will be rendered as soon as possible, but not later than sixty (60) days after receipt of the request for review. All extensions of time permitted by this Section 8.2 will be permitted at the sole discretion of the Committee or its delegate. If the Committee does not provide the Participant with written notice of the denial of his or her appeal, the Participant's claim shall be deemed denied. Notwithstanding anything in the Plan to the contrary, any court, tribunal or arbitration panel that adjudicates any dispute, controversy or claim arising between a Participant and the Company, or any of their delegates or successors, in respect of a Participant's Qualifying Termination occurring after a Potential Change in Control, will apply a <u>de novo</u> standard of review to any determinations
- (c) If any contest or dispute shall arise under the Plan involving a Participant's Termination of Employment or involving the failure or refusal of the Company to perform fully in accordance with the terms hereof, in either case following a Change in Control, the Company shall or shall cause an Affiliate to reimburse the Participant on a current basis for all reasonable legal fees and related expenses, if any, incurred by the Participant at any time from the Date of Termination through the Participant's remaining lifetime (or, if longer, through the 20th anniversary of the Change in Control) in connection with such contest or dispute (regardless of the result thereof), together with interest at the rate provided in Section 1274(b)(2)(B) of the Code, such interest to accrue thirty (30) days from the date the Company receives the Participant's statement for such fees and expenses through the date of payment thereof, regardless of whether or not the Participant's claim is upheld by a court of competent jurisdiction or an arbitration panel; provided, however, that the Participant shall be required to repay immediately any such amounts to the Company to the extent that a court or an arbitration panel issues a final and non-appealable order setting forth the determination that the position taken by the Participant was frivolous or advanced by the

made by the Company, Committee or any other person. Such de novo standard shall apply notwithstanding the grant of full discretion hereunder to any person or characterization of any such decision by such person as final, binding or conclusive on any

party.

Participant in bad faith. The amount of such legal fees and expenses that the Company is obligated to pay in any given calendar year shall not affect the legal fees and expenses that the Company is obligated to pay in any other calendar year, and the Participant's right to have the Company pay such legal fees and expenses may not be liquidated or exchanged for any other benefit.

- **8.3.** <u>Survival</u>. The respective obligations and benefits afforded to the Company, its Affiliates and the Participant shall survive the termination of the Plan.
- **8.4.** Governing Law; Validity. To the extent not preempted by federal law, the Plan, and all benefits and agreements hereunder, and any and all disputes in connection therewith, shall be governed by and construed in accordance with the substantive laws of the State of Delaware, without regard to conflict or choice of law principles which might otherwise refer the construction, interpretation or enforceability of the Plan to the substantive law of another jurisdiction.
- **8.5.** Amendment and Termination. The Board may amend (and, by amendment, terminate) the Plan at any time; provided, however, that (i) no amendment that reduces or eliminates any benefit or other entitlement of any Participant or that is otherwise adverse to the interests of a Participant (an "Adverse Amendment") may take effect for a period of at least twelve (12) months, and any such amendment shall be void and of no effect unless the Participant was notified of such amendment; (ii) no Adverse Amendment may be adopted during the period of time beginning on a Potential Change in Control and ending on the earlier of (a) the termination of the agreement that constituted the Potential Change in Control and (b) the second anniversary of the resulting Change in Control, without the Participant's written consent and (iii) no Adverse Amendment may be adopted during the period commencing on a Change in Control and ending on the second anniversary of the Change in Control without the Participant's written consent. For the avoidance of doubt, the Board is expressly permitted, without the consent of any Participant, to amend the list of "Restricted Entities" set forth on Exhibit D hereto in its reasonable discretion in order to reflect changes in the businesses and industries in which the Company or any of its Affiliates operates or other changes in business circumstances, and no such amendment shall be deemed an Adverse Amendment hereunder. The Company shall notify all Participants of each amendment to Exhibit D in accordance with Section 8.1 above. The Participant's or the Company's failure to insist upon strict compliance with any provision of this Agreement or the failure to assert any right the Participant or the Company may have hereunder, including, without limitation, the right of the Participant to terminate employment for Good Reason, shall not be deemed to be a waiver of such provision or right or any other provision or right of this Agreement; provided, however, that in all events a Participant shall not have Good Reason to terminate his or her employment based upon the occurrence or non-occurrence of a particular event more than ninety (90) days following the initial occurrence or non-occurrence of such event.

8.6. <u>Interpretation and Administration</u>. The Plan shall be administered by the Committee. Unless otherwise provided in the Plan, actions of the Committee shall be taken by a

majority vote of its members. Subject to Section 8.2 of the Plan, the Committee shall have the authority (i) to exercise all of the powers granted to it under the Plan, (ii) to construe, interpret and

implement the Plan, (iii) to prescribe, amend and rescind rules and regulations relating to the Plan, (iv) to make all determinations necessary or advisable in administration of the Plan and (v) to correct any defect, supply any omission and reconcile any inconsistency in the Plan. The invalidity or unenforceability of any provision of the Plan shall not affect the validity or enforceability of any other provision of the Plan.

8.7. <u>Liability</u>. No member of the Board shall be personally liable by reason of any contract or other instrument executed by such member or on his or her behalf in his or her capacity as a member of the Board or Committee or for any mistake of judgment made in good faith or upon the advice of counsel, and the Company shall indemnify and hold harmless each member of the Committee, and each other employee, officer or director of the Company to whom any duty or power relating to the administration or interpretation of the Plan may be allocated or delegated, against all costs and expenses (including counsel fees) and liabilities (including sums paid in settlement of a claim) arising out of any act or omission to act in

connection with the Plan unless arising out of such person's own fraud or willful misconduct; provided, however, that approval of the Board shall be required for the payment of any amount in settlement of a claim against any such person. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such persons may be entitled under the Company's certificate or articles of incorporation or by-laws, each as may be amended from time to time, as a matter of law or otherwise, or any power that the Company may have to indemnify them or hold them harmless.

- **8.8.** Type of Plan. This Plan is intended to be, and shall be interpreted (a) as an unfunded employee welfare plan under Section 3(1) of ERISA and Section 2520.104-24 of the Department of Labor Regulations, maintained primarily for the purpose of providing employee welfare benefits, to the extent that it provides welfare benefits, and under Sections 201, 301 and 401 of ERISA, as a plan that is unfunded and maintained primarily for the purpose of providing deferred compensation, to the extent that it provides such compensation, in each case for a select group of management or highly compensated employees and (b) to comply with the requirements of Section 409A of the Code.
- **8.9.** <u>Funding</u>. The Company and the Affiliate of the Company employing the Participant shall be joint and severally responsible for any payments and benefits hereunder. The Plan shall be unfunded and all payments hereunder and expenses incurred in connection with the Plan shall be paid from the general assets of the Company.
- **8.10.** <u>Headings</u>. The Section headings contained herein are for convenience of reference only, and shall not be construed as defining or limiting the matter contained thereunder.

| 8.11. Entire Agreement. In the event of any inconsistency or conflict between the terms of the Plan and the terms of any other plan in which a Participant participates or any agreement to which a Participant is a party, the terms of this Plan shall control. |
|---|
| 8.12. Nonassignability. Benefits under the Plan may not be sold, assigned, transferred, pledged, anticipated, mortgaged or otherwise encumbered, transferred, hypothecated or conveyed in advance of actual receipt of the amounts, if any, payable hereunder or any part thereof, by the Participant. |
| |
| |
| |
| |
| |
| |
| |
| CENTURYLINK EXECUTIVE SEVERANCE PLAN As Updated, Amended and Restated Effective October 10, 2017 |
| E-17 |
| |
| |
| |
| |
| |
| |

EXHIBIT A

Participation Notice

Personal & Confidential

 $[\bullet], 2012$

[NAME]

[ADDRESS]

Dear [FIRST NAME]:

I am pleased to inform you that you have been selected to participate in the Level 3 Communications, Inc. Key Executive Severance Plan (the "Plan"), which has been established to provide eligible executives of the Company, who are involuntarily terminated from employment in certain limited circumstances, with severance and welfare benefits. The terms and conditions of your participation are set forth in and governed by the terms of the Plan and this participation notice (this "Participation Notice").

For purposes of your participation in the Plan, the following terms shall have the respective meanings set forth herein:

<u>"Cash Severance Benefit"</u> means an amount equal to the product of the Severance Multiplier and [the sum of] your Base Salary [and your Target Bonus on the Date of Termination].

"Severance Multiplier" means [0.75] [1] [2].

Other capitalized terms used but not otherwise defined herein shall have the same meaning as set forth in the Plan.

It is important that the terms and conditions of your participation in the Plan as set forth in this Participation Notice be kept confidential, as they pertain only to you. Please acknowledge your acceptance of these terms and your participation in the Plan by executing a Restrictive Covenant Agreement as set forth in the Plan. If you have any questions regarding this Participation Notice or the Plan, please direct those questions to [•].

Sincerely,

EXHIBIT B

RESTRICTIVE COVENANT AGREEMENT

As a condition of my becoming a Participant in the Level 3 Communications, Inc. Key Executive Severance Plan (the "Plan") and receiving any payments or benefits thereunder, and in consideration of my continued employment with the Company and its Affiliates (collectively, the "Company Group"), I acknowledge the terms and conditions of my participation in the Plan as set forth in my Participation Notice and agree to the following:

Section 1. Definitions.

Capitalized terms used, but not defined, herein shall have the meaning given to them in the Plan.

Section 2. Participation in the Plan.

By executing this Restrictive Covenant Agreement (the "Agreement"), I acknowledge that I have been designated as a Participant in the Plan, and that my becoming a Participant is conditioned upon my execution of this Agreement. I acknowledge further that I have been provided a copy of the Plan and, as an express condition to my participation in, eligibility for, or receipt of, any payments or benefits under the Plan, I agree to be bound by the terms of both the Plan and this Agreement. I acknowledge further that a breach of this Agreement will constitute Cause under the Plan, pursuant to which my employment may be terminated without any right to receive any payments or benefits thereunder, and in addition to any other rights or remedies that the Company or any of its Affiliates may have as a result of any breach of this Agreement, I acknowledge and agree that any payments or benefits to which I might otherwise be entitled under the Plan will immediately terminate, and neither the Company nor any of its Affiliates will have any further obligations to me under the Plan, following any breach of this Agreement.

Section 3. Confidential Information and Intellectual Property.

- (a) I represent that I am a party to an Employee Confidentiality and Intellectual Property Agreement with the Company, dated [•], and agree that such Employee Confidentiality and Intellectual Property Agreement shall survive and not be superseded by this Agreement and shall survive the termination of my employment.
- (b) Company Policies. I acknowledge and agree that in addition to any covenants or restrictions set forth herein, I will at all times continue to be bound by the Company's intellectual property and confidential information policies as in effect from time to time.

Section 4. Returning Company Group Documents and Property.

I agree that, at the time of any termination of my employment with the Company Group for any reason, I will deliver to the Company (and will not keep in my possession, recreate, or

deliver to anyone else) any and all other property belonging to the Company or any other member of the Company Group, including but not limited to all Confidential Information and all other documents (including any copies thereof) in any form belonging to the Company,

materials, information, and property developed by me pursuant to my employment or otherwise belonging to the Company or any member of the Company Group, cell phone, smart phone, iPad, computer (including any laptop or desktop computer, and peripheral devices), beeper, keys, card access to the building and office floors, employee handbook, phone card, computer user name and password, disks, and voicemail code. I agree further that any property situated on the Company's premises and owned by the Company (or any other member of the Company Group), including disks and other storage media, filing cabinets, and other work areas, is subject to inspection by personnel of any member of the Company Group at any time with or without notice.

Section 5. Disclosure of Agreement.

As long as it remains in effect, I will disclose the existence of this Agreement to any prospective employer, partner, coventurer, investor, or lender prior to entering into an employment, partnership, or other business relationship with such person or entity.

Section 6. Noncompetition; Nonsolicitation; Nondisparagement.

(a) Noncompetition and Nonsolicitation. During the period of my employment with any of the members of the Company Group and the Restricted Period (as defined below), I will not (i) directly or indirectly encourage, solicit, or induce, or in any manner attempt to encourage, solicit, or induce, any person employed by, or providing consulting services to, any member of the Company Group to terminate such person's employment or services (or in the case of a consultant, to materially reduce such services) with the Company Group; (ii) hire any individual who was employed by the Company Group within the six (6) month period prior to the date of such hiring and with whom I had contact during my employment with the Company Group within the six (6) month period prior to the date of such hiring; (iii) induce or attempt to induce any customer, supplier, licensee, or other business relation of the Company Group to cease doing business with or reduce the amount of its business with the Company Group or interfere with the relationship between any such customer, supplier, licensee, or business relation and the

Company Group; or (iv) directly or indirectly engage in, own, invest in, manage, be employed by, consult, advise, assist, loan money to, or promote business for any person(s) or entity who or which is engaged in the same business as the Company Group, offers for sale the same products or services as the Company Group, or otherwise is a competitor of Company Group. Clause (iv) of the preceding sentence shall include but shall not be limited to the companies (and types of companies) set forth on Exhibit D to the Plan, which, for the avoidance of doubt, may be updated or amended by the Company from time to time without my consent in accordance with the terms of the Plan. For purposes of this Agreement, the term "Restricted Period" means the period commencing on my Date of Termination and continuing thereafter for a number of months equal to the product of (I) twelve and (II) the Severance Multiplier. Notwithstanding the foregoing, the

restrictions in clause (iv) of this Section 6 shall apply to the Restricted Period only in the event of a Qualifying Termination and may be waived during the Restricted Period by the Chief Executive Officer and the Chief Administrative Officer, in their sole and absolute discretion, in a written waiver signed by both officers.

(b) Nondisparagement. I agree that, during the period of my employment with any of the members of the Company Group and at all times thereafter, I will not make any disparaging or defamatory comments regarding any member of the Company Group or its respective current or former directors, officers, or employees in any respect or make any comments concerning any aspect of my relationship with any member of the Company Group or any conduct or events which precipitated any termination of my employment from any member of the Company Group. However, my obligations under this subsection (b) shall not apply to disclosures required by applicable law, regulation, or order of a court or governmental agency.

Section 7. Reasonableness of Restrictions.

I acknowledge and recognize the highly competitive nature of the business of the members of the Company Group, that access to Confidential Information renders me special and unique within the industry of the members of the Company Group, and that I will have the opportunity to develop substantial relationships with existing and prospective clients, accounts, customers, consultants, contractors, investors, and strategic partners of the members of the Company Group during the course of and as a result of my employment with any of the members of the Company Group. In light of the foregoing, I recognize and acknowledge that the restrictions and limitations set forth in this Agreement are reasonable and valid in geographical and temporal scope and in all other respects and are essential to protect the value of the business and assets of the Company Group. I acknowledge further that the restrictions and limitations set forth in this Agreement will not materially interfere with my ability to earn a living following the termination of my employment with the Company Group and that my ability to earn a livelihood without violating such restrictions is a material condition to my employment with the Company Group.

Section 8. Independence; Severability; Blue Pencil.

Each of the rights enumerated in this Agreement shall be independent of the others and shall be in addition to and not in lieu of any other rights and remedies available to the members of the Company Group at law or in equity. If any of the provisions of this Agreement or any part of any of them is hereafter construed or adjudicated to be invalid or unenforceable, the same shall not affect the remainder of this Agreement, which shall be given full effect without regard to the invalid portions. If any of the covenants contained herein are held to be invalid or unenforceable because of the duration of such provisions or the area or scope covered thereby, I agree that the court making such determination shall have the power to reduce the duration, scope, and area of such provision to the maximum and/or broadest duration, scope, and area permissible by law, and in its reduced form said provision shall then be enforceable.

Section 9. Injunctive Relief.

I expressly acknowledge that any breach or threatened breach of any of the terms or conditions set forth in this Agreement may result in substantial, continuing, and irreparable injury to one or more of the members of the Company Group. Therefore, I hereby agree that, in addition to any other remedy that may be available to any member of the Company Group, any member of the Company Group, on its own behalf or on behalf of any other member or members of the Company Group, shall be entitled to seek injunctive relief, specific performance, or other equitable relief by a court of appropriate jurisdiction in the event of any breach or threatened breach of the terms of this Agreement without the necessity of proving irreparable harm or injury as a result of such breach or threatened breach. Notwithstanding any other provision to the contrary, I acknowledge and agree that the Restricted Period shall be tolled during any period of violation of any of the covenants in Section 6 hereof and during any other period required for litigation during which the Company or any other member of the Company Group seeks to enforce such covenants against me if it is ultimately determined that I was in breach of such covenants.

Section 10. Cooperation.

I agree that, following any termination of my employment, I will continue to provide reasonable cooperation to the Company and any other member of the Company Group and its or their respective counsel in connection with any investigation, administrative proceeding, or litigation relating to any matter that occurred during my employment in which I was involved or of which I have knowledge. As a condition of such cooperation, the Company shall reimburse me for reasonable out-of-pocket expenses incurred at the request of the Company with respect to my compliance with this paragraph. I also agree that, in the event that I am subpoenaed by any person or entity (including, but not limited to, any government agency) to give testimony or provide documents (in a deposition, court proceeding, or otherwise) that in any way relates to my employment by the Company Group, I will give prompt notice of such request to the Company and will make no disclosure until the Company or the other applicable member of the Company Group has had a reasonable opportunity to contest the right of the requesting person or entity to such disclosure.

Section 11. General Provisions.

(a) Governing Law and Jurisdiction. EXCEPT WHERE PREEMPTED BY FEDERAL LAW, THE VALIDITY, INTERPRETATION, CONSTRUCTION, AND PERFORMANCE OF THIS AGREEMENT IS GOVERNED BY AND IS TO BE CONSTRUED UNDER THE LAWS OF THE STATE OF DELAWARE APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED IN THAT STATE, WITHOUT REGARD TO CONFLICT OF LAWS RULES. FURTHER, I HEREBY WAIVE ANY RIGHT TO TRIAL BY JURY IN CONNECTION WITH ANY SUIT, ACTION, OR PROCEEDING UNDER OR IN CONNECTION WITH THIS AGREEMENT.

| (b) Entire Agreement. This Agreement, together with the Plan and any other agreements executed by me in |
|--|
| connection with my participation in the Plan sets forth the entire agreement and understanding between the Company and me |
| relating to the subject matter herein |
| and merges all prior discussions between us. No modification or amendment to this Agreement or any waiver of any rights under this |
| Agreement will be effective unless in writing signed by the party to be charged. Any subsequent change or changes in my duties |
| obligations, rights, or compensation will not affect the validity or scope of this Agreement. |
| |
| |

- (c) No Right of Continued Employment. I acknowledge and agree that nothing contained herein or in the Plan shall be construed as granting me any right to continued employment by the Company Group, and the right of my employer to terminate my employment at any time and for any reason, with or without cause, is specifically reserved.
- (d) Successors and Assigns. This Agreement will be binding upon my heirs, executors, administrators, and other legal representatives and will be for the benefit of the Company, its successors, and its assigns. I expressly acknowledge and agree that this Agreement may be assigned by the Company without my consent to any other member of the Company Group as well as any purchaser of all or substantially all of the assets or stock of the Company or of any business or division of the Company Group for which I provide services, whether by purchase, merger, or other similar corporate transaction.
- (e) Survival. The provisions of this Agreement shall survive the termination of my employment with the Company Group or the assignment of this Agreement by the Company to any successor in interest or other assignee.

* * *

 I, \quad , have executed this Agreement on the respective date set forth below:

Date:

(Signature)

(Type/Print Name)

EXHIBIT C

RELEASE AGREEMENT

This Release Agreement (this "Agreement") is executed and agreed to by [**Participant**] pursuant to the terms and conditions of the Level 3 Communications, Inc. Key Executive Severance Plan (the "Plan"). Capitalized terms used, but not defined, herein shall have the meaning given to them in the Plan.

In consideration of the promises set forth herein, I acknowledge and agree to the following:

Section 1. Opportunity for Review; Acceptance.

This Agreement may be executed at any time during the period commencing on [date of receipt of the agreement] and ending on [add 45 days if RIF, 21 if non-RIF] (the "Review Period"), and during the Review Period, the terms and conditions set forth herein should be carefully reviewed and considered. To accept this Agreement, and the terms and conditions contained herein, prior to the expiration of the Review Period, this Agreement must be executed and dated where indicated below and returned to the Company in accordance with the notice provisions set forth in the Plan. Notwithstanding anything contained herein to the contrary, this Agreement will not become effective or enforceable for a period of seven (7) calendar days following the date of its execution (the "Revocation Period"), during which time such execution may be revoked by notifying the Company in accordance with the notice provisions set forth in the Plan no later than 5:00 p.m. on the seventh (7th) calendar day following its execution. Provided that the Agreement is timely executed and not timely revoked, the eighth (8th) day following the date on which this Agreement is executed shall be its effective date. In the event that the Agreement is not timely executed and delivered to the Company, or if the Agreement is otherwise timely revoked during the Revocation Period, this Agreement will be null and void and of no effect, and the Company will have no obligations to provide any severance benefits under the Plan that are conditioned upon the execution and non-revocation of this Agreement.

Section 2. Employment Status and Separation Payments.

(a) Employment Status. I acknowledge and agree that my employment with the Company and its Affiliates (the "Company Group") terminated on **[DATE]** (the "Date of Termination"), and that after the Date of Termination I will not represent myself as being an employee, officer, agent, or representative of the Company or any other member of the Company Group.

Accrued Benefits. I understand that the Date of Termination shall be the termination date of my employment for purposes of participation in and coverage under all benefit plans and programs sponsored by or through the Company and any other member of the Company Group, except as otherwise provided herein. I will be paid or provided the Accrued

Obligations in accordance with the terms of the Plan regardless of whether this Agreement becomes effective.

Severance Payments and Benefits. In consideration of my release and waiver of claims set forth in Section 3 below, and conditioned upon my execution and non-revocation of this Agreement, and subject to all other terms and conditions of the Plan, including without limitation my continued compliance with my Restrictive Covenant Agreement, the Company will provide me with the severance payments and benefits set forth in Section 4.2 of the Plan in accordance with the terms of the Plan.

(c) All Wages Paid, No Further Benefits. I hereby acknowledge and agree that, except as provided in this Agreement, I have been paid and/or have received all other compensation and/or benefits to which I am entitled and that no other compensation or benefits are due to me other than my vested account under the Level 3 Communication, Inc. 401(k) Plan and any rights to continuation coverage that I have under COBRA. I further acknowledge and agree that the Company owes no other obligations to me, including any obligations arising out of any alleged written or oral employment agreement, policy, plan or procedure of the Company, or alleged understanding or arrangement between me and the Company or any other member of the Company Group. I, therefore, acknowledge and agree that the benefits described above in this Section 2 exceed any claim that I might have to any other benefits or compensation of any kind.

Section 3. Release and Waiver of Claims.

- (a) Definitions. As used in this Agreement, the term "claims" will include all claims, covenants, warranties, promises, undertakings, actions, suits, causes of action, obligations, debts, accounts, attorneys' fees, judgments, losses, and liabilities, of whatsoever kind or nature, in law, equity, or otherwise.
- (b) Release. For and in consideration of the payments and benefits described in Section 2 above, and other good and valuable consideration, I, for and on behalf of myself and my heirs, administrators, executors, and assigns, effective the date hereof, do fully and forever release, remise, and discharge each member of the Company Group and their successors and assigns, together with their respective officers, directors, partners, shareholders, employees, and agents (collectively, and with the Company, the "Company Parties") from any and all claims whatsoever up to the date hereof that I had, may have had, or now have against the Company Parties, whether known or unknown, for or by reason of any matter, cause, or thing whatsoever, including any claim arising out of or attributable to my employment or the termination of my employment with the Company Group, whether for tort, breach of express or implied employment contract, intentional infliction of emotional distress, wrongful termination, unjust dismissal, defamation, libel, or slander, or under any federal, state, or local law dealing with discrimination based on age, race, sex, national origin, handicap, religion, disability, or sexual orientation. This release of claims includes, but is not limited to, all claims arising under the Age Discrimination in Employment Act, Title VII of the Civil Rights Act, the Americans with Disabilities Act, the Civil Rights Act of 1991, the Family Medical Leave

Act, and the Equal Pay Act, each as may be amended from time to time, and all other federal, state, and local laws, the common law, and any other purported restriction on an employer's right to terminate the employment of employees. I intend that the release contained herein shall constitute a general release of any and all claims that I may have against the Company Parties to the fullest extent permissible by law.

- (c) No Claims. I acknowledge and agree that as of the date I execute this Agreement, I have no knowledge of any facts or circumstances that give rise or could give rise to any claims under any of the laws listed in the preceding paragraph.
- (d) ADEA Release. By executing this Agreement, I understand that I am specifically releasing all claims relating to my employment and its termination under the Age Discrimination in Employment Act (ADEA), a United States federal statute that, among other things, prohibits discrimination on the basis of age in employment and employee benefit plans.
- (e) Preservation of Rights. Notwithstanding the foregoing, nothing in this Agreement shall be a waiver of (i) my rights with respect to payment of amounts under this Agreement, (ii) any claims that cannot be waived by law including, without limitation any claims filed with the Equal Employment Opportunity Commission, the U.S. Department of Labor, and claims under the ADEA that arise after the date of this Agreement, or (iii) if applicable, my rights to indemnification as provided by, and in accordance with the terms of, the Company's by-laws or a Company insurance policy providing such coverage, as any of such may be amended from time to time.
- (f) Government Investigations. Notwithstanding any other provision of this Agreement to the contrary, nothing in this Agreement, or in any other agreement between the Company Group and me shall be construed as releasing, waiving or in any way limiting my ability, in good faith, to initiate or participate in any federal, state, or local government investigation of the Company Group or to obtain financial compensation or other pecuniary or nonpecuniary award associated with any such investigation.
- (g) Acknowledgement of Full and Final Release. I acknowledge and agree that by virtue of the foregoing, I have waived any relief available to me (including without limitation, monetary damages, equitable relief, and reinstatement) under any of the claims or causes of action waived in this Section 3. I agree, therefore, that I will not accept any award or settlement from any source or proceeding (including but not limited to any proceeding brought by any other person or by any government agency) with respect to any claim or right waived in this Agreement.

Section 4. Knowing and Voluntary Waiver.

I expressly acknowledge and agree that I—

(a) Am able to read the language, and understand the meaning and effect, of this Agreement;

- (b) Have no physical or mental impairment of any kind that has interfered with my ability to read and understand the meaning of this Agreement or its terms, and that I am not acting under the influence of any medication, drug, or chemical of any type in entering into this Agreement;
- (c) Am specifically agreeing to the terms of the release contained in this Agreement because the Company has agreed to provide me with the severance payments and benefits provided by the Plan, which the Company has agreed to provide because of my agreement to

accept it in full settlement of all possible claims that I might have or ever have had, and because of my execution of this Agreement;

- (d) Acknowledge that but for my execution of this Agreement, I would not be entitled to the severance payments and benefits provided by the Plan;
- (e) Understand that, by entering into this Agreement, I do not waive rights or claims under the ADEA that may arise after the date on which I execute this Agreement;
- (f) Had or could have had the entire Review Period in which to review and consider this Agreement, and that if I execute this Agreement prior to the expiration of the Review Period, I have voluntarily and knowingly waived the remainder of the Review Period;
 - (g) Was advised to consult with my attorney regarding the terms and effect of this Agreement; and
 - (h) Have signed this Agreement knowingly and voluntarily.

Section 5. No Re-Employment.

I hereby agree to waive any and all claims to re-employment with the Company or any other member of the Company Group. I affirmatively agree not to seek further employment with the Company or any other member of the Company Group.

Section 6. Successors and Assigns.

The provisions hereof shall inure to the benefit of my heirs, executors, administrators, legal personal representatives, and assigns and shall be binding upon my heirs, executors, administrators, legal personal representatives, and assigns.

Section 7. Severability.

If any provision of this Agreement shall be held by any court of competent jurisdiction to be illegal, void, or unenforceable, that provision shall be of no force and effect. The illegality or unenforceability of any provision, however, shall have no effect upon and shall not impair the enforceability of any other provision of this Agreement.

Section 8. Confidentiality.

The terms and conditions of this Agreement are and shall be deemed to be confidential, and shall not be disclosed by me to any person or entity without the prior written consent of the Company, except if required by law, and to, attorneys, and immediate family, provided that, to the maximum extent permitted by applicable law, rule, code, or regulation, they agree to maintain the confidentiality of the Agreement.

Section 9. Non-Admission.

Nothing contained in this Agreement will be deemed or construed as an admission of wrongdoing or liability on the part of me or any member of the Company Group.

Section 10. Entire Agreement.

This Agreement, together with the Plan, my Participation Notice and my Restrictive Covenant Agreement, constitutes the entire understanding and agreement between me and each member of the Company Group regarding the termination of my employment. This Agreement supersedes all prior negotiations, discussions, correspondence, communications, understandings, and agreements between me and any member of the Company Group relating to the subject matter of this Agreement.

Section 11. Governing Law; Jurisdiction.

EXCEPT WHERE PREEMPTED BY FEDERAL LAW, THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH FEDERAL LAW AND THE LAWS OF THE STATE OF DELAWARE, APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED IN THAT STATE. EACH PARTY TO THIS AGREEMENT HEREBY WAIVES ANY RIGHT TO TRIAL BY JURY IN CONNECTION WITH ANY SUIT, ACTION OR PROCEEDING UNDER OR IN CONNECTION WITH THIS AGREEMENT.

IN WITNESS WHEREOF, I have executed this Agreement as of the date set forth below.

[Participant]

Dated:

EXHIBIT D

Restricted Entities

For purposes of the Plan and each Participant's obligations under his or her Restrictive Covenant Agreement, the following entities shall be deemed "Restricted Entities": (i) companies that include within their corporate structure competitive local exchange carrier(s) or incumbent local exchange carrier(s), which with affiliates have, for their most recent fiscal year, annual consolidated total communications revenue equal to or greater than \$1 Billion; (ii) providers of content delivery network services which with affiliates have, for their most recent fiscal year, consolidated total content delivery network revenues greater than \$50 million; (iii) international communication services providers which with affiliates have a presence in the United States and, with affiliates, have, for their most recent fiscal year, annual consolidated total revenue equal to or greater than \$1 Billion; or (iv) XO Holdings, Inc., CenturyLink, Inc., AT&T Inc., Sprint Nextel Corporation, tw telecom inc., Verizon Communications Inc., Limelight Networks, Inc., Akamai Technologies Inc., Windstream Corporation, Reliance Communications Venture Limited, including in each case their affiliates, successors, and assigns.

CENTURYLINK, INC. SUBSIDIARIES OF THE REGISTRANT AS OF DECEMBER 31, 2018

State of incorporation

| Subsidiary | incorporation or formation |
|---|-------------------------------|
| Actel, LLC | Delaware |
| Century Cellunet International, Inc. | Louisiana |
| Cellunet of India Limited | Mauritius |
| Century Telephone of West Virginia, Inc. | West Virginia |
| CenturyLink Communications, LLC | Delaware |
| Cognilytics, Inc. | California |
| Cognilytics Software and Consulting Private Limited | India |
| ElasticBox, Inc. | Delaware |
| netAura, LLC | Delaware |
| Qwest International Services Corporation | Delaware |
| CenturyLink do Brasil Participações Ltda. | Brazil |
| CenturyLink Comunicações Ltda. | Brazil |
| SAVVIS do Brasil Ltda. | Brazil |
| Savvis Telecomunicações Ltda. | Brazil |
| Qwest N Limited Partnership | Delaware |
| Qwest Transoceanic, Inc. | Delaware |
| Qwest Communications International Ltd. | United Kingdom |
| CenturyLink Limited | United Kingdom |
| CenturyLink Austria GmbH | Austria |
| Qwest Holdings, BV | Netherlands |
| CenturyLink Belgium Sprl | Belgium |
| Qwest France SAS | France |
| Qwest Germany GmbH | Germany |
| Qwest Netherlands BV | Netherlands |
| Qwest Peru S.R.L. | Peru |
| Qwest Telecommunications Asia, Limited | Hong Kong |
| CenturyLink Technology Australia Pty Limited | Australia |
| CenturyLink Corporation Japan | Japan |
| Qwest Communications Korea, Limited | Korea |
| Qwest Hong Kong Telecommunications, Limited | Hong Kong |
| CenturyLink Technology Singapore Pte Ltd. | Singapore |
| Qwest Taiwan Telecommunications, Limited | Taiwan |
| SEAL Consulting, Inc. | New Jersey |
| AppFog, Inc. | Delaware |
| CenturyLink Europe B.V. | Netherlands |
| CenturyLink Europe B.V., The Netherlands, filial Sweden | Sweden |
| CenturyLink Europe B.V., Sucursal en España | Spain |
| CenturyLink France S.A.S. | France |
| CenturyLink Italia S.r.I. | Italy |
| CenturyLink New Zealand Limited | New Zealand |
| | |

| CenturyLink Philippines DecenturyLink Poland Sp Zo.0. Poland CenturyLink Switzerland A.G. Switzerland CenturyLink Taiwan Limited Taiwan CenturyLink Taiwan Limited Hong Kong CenturyLink Technology Hong Kong Limited Hong Kong CenturyLink Tr Consulting (Shanghai) Co., Ltd. China Digital Savvis HK Holding 1 Limited British VI Digital Savvis HK Holding 1 Limited British VI Digital Savvis HK Holding 1 Limited British VI Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. Argentina SAVVIS Argentina S.A. Argentina SAVVIS Communications Chile, S.A. Argentina SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington Washington WanniNet Japan K.K. Japan CenturyTel of Adamsville, Inc. Tennessee CenturyTel of Adamsville, Inc. Wisconsin CenturyTel of Adamsville, Inc. Wisconsin CenturyTel Fleview of Wisconsin, Inc. Wisconsin CenturyTel Fleview of Wisconsin, Inc. Delaware CenturyTel Fleview of Wisconsin, Inc. Delaware CenturyTel Fleview of Wisconsin, Inc. Louisiana CenturyTel of Central Indiana, Inc. Louisiana CenturyTel of Central Arkansas, LLC CenturyTel of Olaborne, Inc. Arkansas CenturyTel of Olaborne, Inc. Arkansas CenturyTel of Olaborne, Inc. Louisiana CenturyTel of Olaborne, Inc. Louisiana CenturyTel of Olaborne, I | Subsidiary | incorporation or formation |
|--|---|-------------------------------|
| CenturyLink Switzerland A.G. CenturyLink Talwan Limited CenturyLink Talwan Limited CenturyLink Troonsulting (Shanghai) Co., Ltd. China Digital Savvis HK JV Limited Digital Savvis HK JV Limited Digital Savvis HK Holding 1 Limited Digital Savvis Investment Management HK Limited Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink Trechnology Malaysia Sdn. Bhd. Malaysia CenturyLink (Thailand) Limited Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink (Thailand) Limited Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. Argentina SAVVIS Argentina S.A. SAVVIS Gommunications Chile, S.A. SAVVIS Gommunications Chile, S.A. SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited India SAVVIS Homyary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WamiNet Japan K.K. CenturyTel of Adamsville, Inc. CenturyTel of Adamsville, Inc. CenturyTel Broadband Services, LLC CenturyTel Broadband Services, LLC CenturyTel Televideo, Inc. CenturyTel Televideo, Inc. CenturyTel Televideo Misconsin, Inc. CenturyTel Televideo, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Central Arkansas, LLC CenturyTel Of Central Indiana, Inc. CenturyTel Of Central Indiana, Inc. CenturyTel of Central Arkansas, LLC CenturyTel Of Northwest Arkansas, LLC CenturyTel Of Missouri, LLC CenturyTel Of Missouri, LLC CenturyTel Of Missouri, LLC Centu | CenturyLink Philippines, Inc. | Philippines |
| CenturyLink Taiwan Limited Hong Kong CenturyLink Technology Hong Kong Limited Hong Kong CenturyLink Technology Hong Kong Limited British VI Digital Savvis HK JV Limited British VI Digital Savvis HK Holding 1 Limited British VI Digital Savvis Investment Management HK Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink Technology Malaysia Sdn. Bhd. Malaysia CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. Chile SAVVIS Federal Systems, Inc. Delaware SAVVIS Federal Systems, Inc. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications Private Limited Korea Mexico SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary Telecommunications KFT Washington WarniNet Japan K.K. Washington WarniNet Japan K.K. Washington WarniNet Japan K.K. Louisiana CenturyTel of Adamsville, Inc. Tennessee CenturyTel of Arkansas, Inc. Arkansas CenturyTel Fordadband Services, LLC Louisiana CenturyTel Televideo, Inc. Wisconsin CenturyTel Televideo (Inc. Wisconsin Delaware CenturyTel Televideo Misconsin, Inc. Delaware CenturyTel Fordadband Services, Inc. Delaware CenturyTel Fordadband Wireless, LLC Louisiana Indiana Inc. Indiana Inc. Indiana CenturyTel Of Central Indiana, Inc. Louisiana CenturyTel Of Central Indiana, Inc. Louisiana CenturyTel Fordarband Kireless, LLC Louisiana CenturyTel Of Central Arkansas, LLC Louisiana CenturyTel Of Central Arkansas, LLC Louisiana CenturyTel Holdings, Inc. Louisiana CenturyTel Of Central Arkansas, LLC Louisiana CenturyTel Holdings Alabama, Inc. Louisiana CenturyTel Holdings Mabama, Inc. Louisiana CenturyTel Of Missouri, Inc. Missouri, Inc. L | CenturyLink Poland Sp Zo.o. | Poland |
| CenturyLink Technology Hong Kong Limited CenturyLink IT Consulting (Shanghai) Co., Ltd. China Digital Savvis HK JV Limited British VI Digital Savvis HK Holding 1 Limited British VI Digital Savvis Investment Management HK Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink Technology Malaysia Sdn. Bhd. Malaysia CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. SAVVIS Argentina S.A. SAVVIS Communications Chile, S.A. SAVVIS Communications Chile, S.A. SAVVIS Communications International, Inc. Delaware SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WarmINet Japan K.K. CenturyTel of Adamsville, Inc. CenturyTel of Adamsville, Inc. CenturyTel of Adamsville, Inc. CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. CenturyTel TeleVideo, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chatser, Inc. CenturyTel of Chatser, Inc. CenturyTel of Chatser, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chatser, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Contral Arkansas, LLC CenturyTel of Oldings Miabama, Inc. CenturyTel of Missouri, Inc. Ce | CenturyLink Switzerland A.G. | Switzerland |
| CenturyLink IT Consulting (Shanghai) Co., Ltd. Digital Savvis HK, JV Limited British VI Digital Savvis IN HK Holding 1 Limited Digital Savvis Investment Management HK Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink Technology Malaysia Sdn. Bhd. CenturyLink (Thailand) Limited DataGardens, Inc. SAVVIS Argentina S.A. SAVVIS Argentina S.A. SAVVIS Gommunications Chile, S.A. SAVVIS Federal Systems, Inc. CenturyLink Korea Limited SAVVIS Communications International, Inc. CenturyLink Korea Limited Korea SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited Korea SAVVIS Mexico, S.A. de C.V. Mexico SAVVIS Mexico, S.A. de C.V. Mexico CenturyLink of Louisiana, LLC CenturyLink of Louisiana, LLC CenturyTel of Arkansas, Inc. CenturyTel of Arkansas, Inc. CenturyTel TeleVideo, Inc. CenturyTel Foodband Services, LLC CenturyTel Foodband Services, Inc. CenturyTel Foodband Services, Inc. CenturyTel Of Central Indiana, Inc. CenturyTel of Indiana, Inc. CenturyTel of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Mabama, Inc. CenturyTel of Missouri, Inc. Missouri CenturyTel of Missouri, Inc. CenturyTel of | CenturyLink Taiwan Limited | Taiwan |
| Digital Savvis HK JV Limited British VI Digital Savvis HK Holding 1 Limited British VI Digital Savvis HK Holding 1 Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink Technology Malaysia Sdn. Bhd. Malaysia CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. Argentina SAVVIS Communications Chile, S.A. Chile SAVVIS Federal Systems, Inc. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited India SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mashington Wam!Net Japan K.K. Japan CenturyLink of Louisiana, LLC CenturyTel of Arkansas, Inc. Arkansas CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel TeleVideo, Inc. Wisconsin CenturyTel TeleVideo, Inc. Delaware CenturyTel Foodband Services, Inc. Delaware CenturyTel Foodband Services, Inc. Delaware CenturyTel Of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Central Arkansas, LLC CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, Inc. | CenturyLink Technology Hong Kong Limited | Hong Kong |
| Digital Savvis HK Holding 1 Limited Hong Kong Digital Savvis Investment Management HK Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink Technology Malaysia Sdn. Bhd. Malaysia CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. Argentina SAVVIS Communications Chile, S.A. Argentina SAVVIS Communications Chile, S.A. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. Delaware CenturyLink Korea Limited Korea SAVVIS Communications Private Limited India SAVVIS Communications Private Limited India SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WamlNet Japan K.K. Japan CenturyLel of Adamsville, Inc. Tennessee CenturyTel of Adamsville, Inc. Tennessee CenturyTel of Arkansas, Inc. Arkansas CenturyTel Televideo, Inc. Wisconsin CenturyTel Televideo, Inc. Wisconsin Qwest Broadband Services, Inc. Delaware CenturyTel Teleview of Wisconsin, Inc. CenturyTel Teleview of Wisconsin, Inc. CenturyTel Teleview of Wisconsin, Inc. CenturyTel Foodband Wireless, LLC CenturyTel of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Claiborne, Inc. CenturyTel of Claiborne, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Orthral Arkansas, LLC CenturyTel Holdings, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, Inc. CenturyTel Holdings Missouri, Inc. | CenturyLink IT Consulting (Shanghai) Co., Ltd. | China |
| Digital Savvis Investment Management HK Limited Hong Kong Digital Savvis Management Subsidiary Limited Hong Kong CenturyLink Technology Malaysia Sdn. Bhd. CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. SAVVIS Communications Chile, S.A. SAVVIS Federal Systems, Inc. Delaware SAVVIS Communications International, Inc. Delaware CenturyLink Korea Limited Korea SAVVIS Communications Private Limited Korea SAVVIS Communications Private Limited India SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary SAVVIS Hungary Televion Mexico Hungary SAVVIS Hungary Mexico Hungary SAVVIS Hungary Mexico Hungary SAVVIS Hungary Hungary | Digital Savvis HK JV Limited | British VI |
| Digital Savvis Management Subsidiary Limited CenturyLink Technology Malaysia Sdn. Bhd. CenturyLink (Thailand) Limited Thailand DataGardens, Inc. Canada SAVVIS Argentina S.A. Argentina SAVVIS Communications Chile, S.A. Chile SAVVIS Federal Systems, Inc. Delaware SAVVIS Communications International, Inc. CenturyLink Korea Limited Korea SAVVIS Communications Private Limited Norea SAVVIS Communications Private Limited Norea SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WamlNet Japan K.K. Japan CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. Tennessee CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel Flevideo, Inc. Wisconsin CenturyTel Televideo, Inc. Wisconsin CenturyTel of Central Indiana, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel Of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Missouri, Inc. Missouri CenturyTel of Missouri, Inc. CenturyTel of Missouri, LLC | Digital Savvis HK Holding 1 Limited | British VI |
| CenturyLink Technology Malaysia Sdn. Bhd. CenturyLink (Thailand) Limited DataGardens, Inc. Canada SAVVIS Argentina S.A. SAVVIS Argentina S.A. Chile SAVVIS Federal Systems, Inc. Delaware SAVVIS Communications Chile, S.A. Chile SAVVIS Communications International, Inc. Delaware CenturyLink Korea Limited Korea SAVVIS Communications Private Limited SAVVIS Communications Private Limited SAVVIS Communications Private Limited SAVVIS Communications Private Limited SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington Wamfilet Japan K.K. Japan CenturyTel of Adamsville, Inc. CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Goldbiorne, Inc. CenturyTel Goldbiorne, Inc. CenturyTel Goldbiorne, Inc. CenturyTel Goldbiorne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Alabama, Inc. CenturyTel of Alabama, Inc. CenturyTel of Northwest Arkansas, LLC CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. Missouri CenturyTel Holdings Missouri, Inc. Missouri CenturyTel Holdings Missouri, Inc. | Digital Savvis Investment Management HK Limited | Hong Kong |
| CenturyLink (Thailand) Limited DataCardens, Inc. Canada SAVVIS Argentina S.A. Argentina SAVVIS Communications Chile, S.A. Chile SAVVIS Communications Chile, S.A. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. CenturyLink Korea Limited SAVVIS Communications Private Limited SAVVIS Communications Private Limited SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary Telecommunications KFT SAVVIS Hungary SAVVIS Communications KFT SAVVIS Communications KFT SAVVIS Hungary SAVVIS Communications KFT SAVVIS Hungary SAVVIS Communications KFT SAVVIS Mexico SAVVIS Communications SAVIS SAV | Digital Savvis Management Subsidiary Limited | Hong Kong |
| CenturyLink (Thailand) Limited DataCardens, Inc. Canada SAVVIS Argentina S.A. Argentina SAVVIS Communications Chile, S.A. Chile SAVVIS Communications Chile, S.A. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. CenturyLink Korea Limited SAVVIS Communications Private Limited SAVVIS Communications Private Limited SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Hungary Telecommunications KFT SAVVIS Hungary SAVVIS Communications KFT SAVVIS Communications KFT SAVVIS Hungary SAVVIS Communications KFT SAVVIS Hungary SAVVIS Communications KFT SAVVIS Mexico SAVVIS Communications SAVIS SAV | CenturyLink Technology Malaysia Sdn. Bhd. | Malaysia |
| SAVVIS Argentina S.A. Chile SAVVIS Communications Chile, S.A. Chile SAVVIS Federal Systems, Inc. Delaware SAVVIS Communications International, Inc. Delaware SAVVIS Communications International, Inc. Delaware CenturyLink Korea Limited Korea SAVVIS Communications Private Limited India SAVVIS Communications Private Limited India SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WamlNet Japan K.K. Japan CenturyLink of Louisiana, LLC Louisiana CenturyTel of Adamsville, Inc. Tennessee CenturyTel of Arkansas, Inc. Arkansas CenturyTel of Arkansas, Inc. Wisconsin CenturyTel Broadband Services, LLC Louisiana CenturyTel Elevideo, Inc. Wisconsin CenturyTel Televideo, Inc. Wisconsin Qwest Broadband Services, Inc. Delaware CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. Indiana CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Chester, Inc. CenturyTel of Chester, Inc. CenturyTel of Chester, Inc. CenturyTel of Central Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. | | Thailand |
| SAVVIS Communications Chile, S.A. SAVVIS Federal Systems, Inc. Delaware SAVVIS Communications International, Inc. Delaware CenturyLink Korea Limited SAVVIS Communications Private Limited SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Tier 3, Inc. Washington Washington WamINet Japan K.K. Japan CenturyLink of Louisiana, LLC CenturyTel of Arkansas, Inc. CenturyTel of Arkansas, Inc. CenturyTel Flevideo, Inc. Wisconsin CenturyTel TeleVideo, Inc. Wisconsin CenturyTel Jelevideo Wisconsin, Inc. Wisconsin Qwest Broadband Services, LLC CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Of Northwest Arkansas, LLC CenturyTel Of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, Inc. CenturyTel of Missouri, Inc. Missouri CenturyTel of Missouri, LLC CenturyTel of Missouri, Inc. | DataGardens, Inc. | Canada |
| SAVVIS Federal Systems, Inc. SAVVIS Communications International, Inc. CenturyLink Korea Limited SAVVIS Communications Private Limited SAVVIS Communications Private Limited SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WamINet Japan K.K. CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel Televideo, Inc. CenturyTel Televideo, Inc. CenturyTel Teleview of Wisconsin, Inc. Qwest Broadband Services, ILC CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Chester, Inc. CenturyTel of Chester, Inc. CenturyTel of Chester, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC | SAVVIS Argentina S.A. | Argentina |
| SAVVIS Communications International, Inc. CenturyLink Korea Limited SAVVIS Communications Private Limited India SAVVIS Hungary Telecommunications KFT SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WamlNet Japan K.K. CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel Televideo, Inc. CenturyTel Teleview of Wisconsin, Inc. Qwest Broadband Services, Inc. CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Claiborne, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Ocentral Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Of Missouri, LLC | SAVVIS Communications Chile, S.A. | Chile |
| CenturyLink Korea Limited Korea SAVVIS Communications Private Limited India SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington WamiNet Japan K.K. Japan CenturyLink of Louisiana, LLC Louisiana CenturyTel of Adamsville, Inc. Tennessee CenturyTel Broadband Services, LLC Louisiana CenturyTel Televideo, Inc. Wisconsin CenturyTel Preleview of Wisconsin, Inc. Wisconsin CenturyTel Broadband Services, LLC Louisiana CenturyTel Televideo, Inc. Indiana CenturyTel Televideo, Inc. Delaware CenturyTel Foodband Wireless, LLC Louisiana CenturyTel Foodband Wireless, LLC Louisiana CenturyTel Of Central Indiana, Inc. Indiana CenturyTel of Central Indiana, Inc. Louisiana CenturyTel of Chester, Inc. Louisiana CenturyTel Foodband Wireless, LLC Louisiana CenturyTel of Chester, Inc. Louisiana CenturyTel of Chester, Inc. Louisiana CenturyTel of Chester, Inc. Louisiana CenturyTel of Central Arkansas, LLC CenturyTel Arkansas Holdings, Inc. Arkansas CenturyTel of Northwest Arkansas, LLC CenturyTel of Oldings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC CenturyTel of Missouri, LLC | SAVVIS Federal Systems, Inc. | Delaware |
| SAVVIS Communications Private Limited SAVVIS Hungary Telecommunications KFT Hungary SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington Wam!Net Japan K.K. Japan CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. Tennessee CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel Televideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin Qwest Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Claiborne, Inc. CenturyTel of Claiborne, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC | SAVVIS Communications International, Inc. | Delaware |
| SAVVIS Hungary Telecommunications KFT SAVVIS Mexico, S.A. de C.V. Mexico Tier 3, Inc. Washington Wam!Net Japan K.K. Japan CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. Tennessee CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin CenturyTel Broadband Services, Inc. CenturyTel Broadband Services, Inc. CenturyTel Of Central Indiana, Inc. CenturyTel Of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Of Inchissouri, LLC CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Of Missouri, LLC | CenturyLink Korea Limited | Korea |
| SAVVIS Mexico, S.A. de C.V. Tier 3, Inc. Washington Wam!Net Japan K.K. Japan CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin CenturyTel Broadband Services, Inc. CenturyTel Broadband Services, Inc. CenturyTel Of Central Indiana, Inc. CenturyTel Broadband Wireless, LLC CenturyTel Of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Of Central Arkansas, LLC CenturyTel of Contral Arkansas, LLC CenturyTel Of Contral Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Missouri, LLC CenturyTel Missouri, LLC CenturyTel Holdings Missouri, Inc. CenturyTel Missouri, LLC | SAVVIS Communications Private Limited | India |
| SAVVIS Mexico, S.A. de C.V. Tier 3, Inc. Washington Wam!Net Japan K.K. Japan CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin CenturyTel Broadband Services, Inc. CenturyTel Broadband Services, Inc. CenturyTel Of Central Indiana, Inc. CenturyTel Broadband Wireless, LLC CenturyTel Of Central Indiana, Inc. CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Of Central Arkansas, LLC CenturyTel of Contral Arkansas, LLC CenturyTel Of Contral Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Missouri, LLC CenturyTel Missouri, LLC CenturyTel Holdings Missouri, Inc. CenturyTel Missouri, LLC | SAVVIS Hungary Telecommunications KFT | Hungary |
| Wam!Net Japan K.K. CenturyLink of Louisiana, LLC CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin Qwest Broadband Services, Inc. Delaware CenturyTel Broadband Wireless, LLC CenturyTel Foethral Indiana, Inc. Indiana CenturyTel of Central Indiana, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Onorthwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana CenturyTel Holdings Missouri, Inc. Missouri CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana | SAVVIS Mexico, S.A. de C.V. | Mexico |
| CenturyTel of Adamsville, Inc. CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin CenturyTel Broadband Services, Inc. CenturyTel Broadband Services, Inc. CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. Indiana CenturyTel of Chester, Inc. Iowa CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC | Tier 3, Inc. | Washington |
| CenturyTel of Adamsville, Inc. CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. Wisconsin CenturyTel/Teleview of Wisconsin, Inc. Wisconsin Qwest Broadband Services, Inc. Delaware CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. Indiana CenturyTel of Claiborne, Inc. Iowa CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana | Wam!Net Japan K.K. | Japan |
| CenturyTel of Arkansas, Inc. CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. CenturyTel/Teleview of Wisconsin, Inc. Wisconsin Qwest Broadband Services, Inc. CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. Indiana CenturyTel of Claiborne, Inc. Tennessee CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, Inc. Missouri CenturyTel of Missouri, LLC Louisiana | CenturyLink of Louisiana, LLC | Louisiana |
| CenturyTel Broadband Services, LLC CenturyTel TeleVideo, Inc. CenturyTel/Teleview of Wisconsin, Inc. Wisconsin Qwest Broadband Services, Inc. CenturyTel Broadband Wireless, LLC CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. Indiana CenturyTel of Claiborne, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana CenturyTel Holdings Missouri, Inc. CenturyTel Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana CenturyTel Holdings Missouri, Inc. Missouri CenturyTel of Missouri, LLC Louisiana | CenturyTel of Adamsville, Inc. | Tennessee |
| CenturyTel TeleVideo, Inc. CenturyTel/Teleview of Wisconsin, Inc. Qwest Broadband Services, Inc. CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel Arkansas, LLC CenturyTel Of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana CenturyTel Holdings Missouri, Inc. Missouri CenturyTel of Missouri, LLC | CenturyTel of Arkansas, Inc. | Arkansas |
| CenturyTel/Teleview of Wisconsin, Inc. Qwest Broadband Services, Inc. CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. Indiana CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Central Arkansas, LLC CenturyTel of Alabama, Inc. CenturyTel Holdings Alabama, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana | CenturyTel Broadband Services, LLC | Louisiana |
| Qwest Broadband Services, Inc. CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. Indiana CenturyTel of Chester, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Alabama, LLC CenturyTel of Missouri, Inc. CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana | CenturyTel TeleVideo, Inc. | Wisconsin |
| CenturyTel Broadband Wireless, LLC CenturyTel of Central Indiana, Inc. Indiana CenturyTel of Chester, Inc. Iowa CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. CenturyTel Holdings, Inc. Century Marketing Solutions, LLC CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. Arkansas CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. Alabama CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. Missouri CenturyTel of Missouri, LLC Louisiana | CenturyTel/Teleview of Wisconsin, Inc. | Wisconsin |
| CenturyTel of Central Indiana, Inc. CenturyTel of Chester, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. Century Marketing Solutions, LLC CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Alabama, LLC CenturyTel of Missouri, Inc. CenturyTel of Missouri, LLC Louisiana | Qwest Broadband Services, Inc. | Delaware |
| CenturyTel of Chester, Inc. CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. Century Marketing Solutions, LLC CenturyTel Arkansas Holdings, Inc. CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana CenturyTel of Missouri, LLC Louisiana | CenturyTel Broadband Wireless, LLC | Louisiana |
| CenturyTel of Claiborne, Inc. CenturyTel Holdings, Inc. Century Marketing Solutions, LLC CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Alabama, LLC CenturyTel of Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana | CenturyTel of Central Indiana, Inc. | Indiana |
| CenturyTel Holdings, Inc. Century Marketing Solutions, LLC CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Alabama, LLC CenturyTel of Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC CenturyTel of Missouri, LLC Louisiana | CenturyTel of Chester, Inc. | lowa |
| Century Marketing Solutions, LLC CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC CenturyTel of Missouri, LLC Louisiana | CenturyTel of Claiborne, Inc. | Tennessee |
| CenturyTel Arkansas Holdings, Inc. CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. CenturyTel Holdings Missouri, LLC CenturyTel of Missouri, LLC Louisiana | CenturyTel Holdings, Inc. | Louisiana |
| CenturyTel of Central Arkansas, LLC CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. Missouri CenturyTel of Missouri, LLC Louisiana | Century Marketing Solutions, LLC | Louisiana |
| CenturyTel of Northwest Arkansas, LLC CenturyTel Holdings Alabama, Inc. CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC CenturyTel of Missouri, LLC Louisiana | CenturyTel Arkansas Holdings, Inc. | Arkansas |
| `CenturyTel Holdings Alabama, Inc.AlabamaCenturyTel of Alabama, LLCLouisianaCenturyTel Holdings Missouri, Inc.MissouriCenturyTel of Missouri, LLCLouisiana | CenturyTel of Central Arkansas, LLC | Louisiana |
| CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana Louisiana | CenturyTel of Northwest Arkansas, LLC | Louisiana |
| CenturyTel of Alabama, LLC CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana Louisiana | `CenturyTel Holdings Alabama, Inc. | Alabama |
| CenturyTel Holdings Missouri, Inc. CenturyTel of Missouri, LLC Louisiana | | Louisiana |
| CenturyTel of Missouri, LLC Louisiana | • | Missouri |
| | | Louisiana |
| | CenturyTel Investments of Texas, Inc. | Delaware |

| Subsidiary | incorporation or formation |
|---|-------------------------------|
| CenturyTel of the Northwest, Inc. | Washington |
| Brown Equipment Corp. | Nevada |
| Carter Company, Inc. | Hawaii |
| Cascade Autovon Company | Washington |
| CenturyTel/Cable Layers, Inc. | Wisconsin |
| CenturyTel of Central Wisconsin, LLC | Delaware |
| CenturyTel of Colorado, Inc. | Colorado |
| CenturyTel of Eagle, Inc. | Colorado |
| CenturyTel of Eastern Oregon, Inc. | Oregon |
| CenturyTel Entertainment, Inc. | Washington |
| CenturyTel of Fairwater-Brandon-Alto, LLC | Delaware |
| CenturyTel of Forestville, LLC | Delaware |
| CenturyTel of the Gem State, Inc. | ldaho |
| CenturyTel of Inter Island, Inc. | Washington |
| CenturyTel of Larsen-Readfield, LLC | Delaware |
| CenturyTel of the Midwest-Kendall, LLC | Delaware |
| CenturyTel of the Midwest-Wisconsin, LLC | Delaware |
| CenturyTel of Minnesota, Inc. | Minnesota |
| CenturyTel of Monroe County, LLC | Delaware |
| CenturyTel of Montana, Inc. | Oregon |
| CenturyTel of Northern Wisconsin, LLC | Delaware |
| CenturyTel of Northwest Wisconsin, LLC | Delaware |
| CenturyTel of Oregon, Inc. | Oregon |
| CenturyTel of Paradise, Inc. | Washington |
| CenturyTel of Cowiche, Inc. | Washington |
| CenturyTel of Postville, Inc. | lowa |
| CenturyTel of Southern Wisconsin, LLC | Delaware |
| CenturyTel of the Southwest, Inc. | New Mexico |
| CenturyTel Telecom Service, Inc. | Washington |
| CenturyTel Telephone Utilities, Inc. | Washington |
| CenturyTel of Upper Michigan, Inc. | Michigan |
| CenturyTel of Washington, Inc. | Washington |
| CenturyTel/WORLDVOX, Inc. | Oregon |
| CenturyTel of Wyoming, Inc. | Wyoming |
| Eagle Valley Communications Corporation | Colorado |
| International Communications Holdings, Inc. | Delaware |
| PTI Communications of Ketchikan, Inc. | Alaska |
| PTI Communications of Minnesota, Inc. | Minnesota |
| PTI Transponders, Inc. | Oregon |
| Universal Manufacturing Corp. | Wisconsin |
| CenturyTel of Idaho, Inc. | Delaware |
| CenturyTel Internet Holdings, Inc. | Louisiana |
| centurytel.com, LLC | Louisiana |
| CenturyTel Investments, LLC | Louisiana |
| | |

| CenturyTel of Michigan, Inc. CenturyTel Mobile Communications, Inc. CenturyTel Mobile Communications, Inc. CenturyTel of Mountain Home, Inc. CenturyTel of North Mississippi, Inc. CenturyTel of North Mississippi, Inc. CenturyTel of Northern Michigan, Inc. CenturyTel of Odon, Inc. CenturyTel of Odon, Inc. CenturyTel of Olio, Inc. CenturyTel of Oolio, Inc. CenturyTel of Ooliewah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel of Redfield, Inc. CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos, Inc. CenturyTel Security Systems, Inc. CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel South Arkansas, Inc. CenturyTel South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Company CenturyTel of Company CenturyTel or Company Centel Corporation Centel Capital Corporation Centel Capital Corporation Centel Capital Corporation Centel Texas Center Capital Corporation Center Texas Center T | oration mation |
|--|-------------------|
| CenturyTel Mobile Communications, Inc. CenturyTel of Mountain Home, Inc. CenturyTel of Mountain Home, Inc. CenturyTel of North Mississippi, Inc. CenturyTel of Northern Michigan, Inc. CenturyTel of Odon, Inc. CenturyTel of Odon, Inc. CenturyTel of Odolin, Inc. CenturyTel of Odolicwah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel SM Telecorp, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Service Group, LLC CenturyTel Service Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Mae Dallas, Inc. CenturyTel Meb Solutions, LLC CenturyTel of Lake Dallas, Inc. CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Woodsnin, LLC CenturyTel Of South Arkansas CenturyTel Of South Arkansas CenturyTel Of South Arkansas CenturyTel Of Solutions, LLC CenturyTel of Comporation CenturyTel of Solutions, LLC CenturyTel of Comporation CenturyTel of Solutions, LLC CenturyTel of Solutions, LLC CenturyTel of Solutions, LLC CenturyTel of Comporation CenturyTel Of Solutions, LLC CenturyTel of Comporation Centel Corporation Centel Corporation Delaware Centel Corporation Centel Corporation Delaware Centel Capital Corporation Centel Capital Corporation Delaware Centel Texas Central Telephone Company of Texas EQ Central Telephone Company of Virginia | |
| CenturyTel of Mountain Home, Inc. CenturyTel of North Mississippi, Inc. CenturyTel of Northern Michigan, Inc. CenturyTel of Northern Michigan, Inc. CenturyTel of Odon, Inc. CenturyTel of Ohio, Inc. CenturyTel of Ohio, Inc. CenturyTel of Oblitwah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel of Redfield, Inc. CenturyTel Sm Telecorp, Inc. CenturyTel Telecommunications, Inc. CenturyTel Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel of South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel Tele-Max, Inc. CenturyTel Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Of Lake Dallas, Inc. CenturyTel Of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Corporation Carolina Telephone and Telegraph Company LLC North Carolina Sc One Company Centel Capital Corporation Centel Capital Corporation Delaware Centel Capital Corporation Centel Capital Corporation Delaware Centel Capital Corporation Centel Capital Corporation Delaware Centel Capital Corporation Centel Capital Corporation Centel Capital Corporation Delaware Centel Texas Central Telephone Company of Texas Central Telephone Company of Virginia Virginia | |
| CenturyTel of North Mississippi, Inc. CenturyTel of Northern Michigan, Inc. CenturyTel of Odon, Inc. CenturyTel of Odio, Inc. CenturyTel of Ohio, Inc. CenturyTel of Ootlewah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel of Redfield, Inc. CenturyTel Sm Telecorp, Inc. CenturyTel Telecommunications, Inc. CenturyTel Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Security Systems, Inc. CenturyTel Security Systems, Inc. CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Of Wisconsin, LLC CenturyTel Of Wisconsin, LLC CenturyTel Service Group, Inc. CenturyTel Of Wisconsin, LLC CenturyTel Of South Aransas CenturyTel Of Wisconsin, LLC CenturyTel Telephone and Telegraph Company LLC North Carolin NCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Centel-Texas, Inc. Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Century Telephone Company of Texas Texas EQ Central Telephone Company of Virginia | |
| CenturyTel of Northern Michigan, Inc. CenturyTel of Odon, Inc. CenturyTel of Ohio, Inc. CenturyTel of Ohio, Inc. CenturyTel of Ooltewah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel Telecommunications, Inc. CenturyTel Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Service Group, LLC CenturyTel Service Holden South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Visconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel Oropration Carolina Telephone and Telegraph Company LLC NoCUTS, Inc. Sc One Company Kansas Centel Capital Corporation Centel Capital Corporation Delaware Centel Capital Corporation Centel Texas Center Texas Center Texas Center Texas Central Telephone Company of Texas Fe Central Telephone Company of Virginia Virginia | |
| CenturyTel of Odon, Inc. CenturyTel of Ohio, Inc. CenturyTel of Olicwah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel Telecommunications, Inc. CenturyTel Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Service Group, LLC CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Max Inc. CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel Orporation Carolina Telephone and Telegraph Company LLC North Carolin NOCUTS, Inc. Sender Corporation Centel Corporation Centel Corporation Centel Corporation Centel Corporation Centel Capital Corporation Centel Capital Corporation Centel Capital Corporation Centel Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Central Telephone Company of Virginia Virginia | |
| CenturyTel of Odon, Inc. CenturyTel of Ohio, Inc. CenturyTel of Olicwah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel Telecommunications, Inc. CenturyTel Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Service Group, LLC CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Max Inc. CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel Orporation Carolina Telephone and Telegraph Company LLC North Carolin NOCUTS, Inc. Sender Corporation Centel Corporation Centel Corporation Centel Corporation Centel Corporation Centel Capital Corporation Centel Capital Corporation Centel Capital Corporation Centel Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Central Telephone Company of Virginia Virginia | |
| CenturyTel of Ooltewah-Collegedale, Inc. CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel SM Telecorp, Inc. CenturyTel SM Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel South Arkansas, Inc. CenturyTel South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel Veb Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC Carolina Telephone and Telegraph Company LLC North Carolina NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Central Telephone Company of Virginia Virginia | |
| CenturyTel of Port Aransas, Inc. CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel SM Telecommunications, Inc. CenturyTel Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Service Group, LLC CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel/Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of South Aransas Cerolina Telephone and Telegraph Company LLC North Carolina Telephone Solutions SC One Company Kansas Centel Corporation Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Oelaware Central Telephone Company of Virginia | |
| CenturyTel of Redfield, Inc. CenturyTel SM Telecorp, Inc. CenturyTel Telecommunications, Inc. CenturyTel Telecommunications, Inc. CenturyTel San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Service Group, LLC CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel Jele-Max, Inc. CenturyTel Meb Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wi | |
| CenturyTel SM Telecorp, Inc. CenturyTel Telecommunications, Inc. CenturyTel of San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel Service Group, LLC CenturyTel South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Supply Group, Inc. CenturyTel Jele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Delaware Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Centel-Texas, Inc. Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel Telecommunications, Inc. CenturyTel of San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel of South Arkansas, Inc. CenturyTel of South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel-Max, Inc. CenturyTel-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC Carolina Telephone and Telegraph Company LLC North Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Telephone Company of Virginia | |
| CenturyTel of San Marcos, Inc. CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel of South Arkansas, Inc. CenturyTel of South Arkansas, Inc. CenturyTel Sepply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel/Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC Central Telephone Company of Texas Central Telephone Company of Virginia Virginia | |
| CenturyTel San Marcos Investments, LLC CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel of South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisc | |
| CenturyTel Security Systems, Inc. CenturyTel Service Group, LLC CenturyTel of South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel-Max, Inc. CenturyTel Jake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC Carolina Telephone and Telegraph Company LLC North Carolin NOCUTS, Inc. SC One Company Centel Carpital Corporation Centel Capital Corporation Delaware Centel Capital Corporation Centel-Texas, Inc. Central Telephone Company Central Telephone Company Central Telephone Company Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel Service Group, LLC CenturyTel of South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel/Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC Carolina Telephone and Telegraph Company LLC North Carolina Telephone and Telegraph Company LLC NoCUTS, Inc. Sc One Company Kansas Centel Corporation Kansas Centel Capital Corporation Delaware Central Telephone Company of Texas EQ Central Telephone Company Central Telephone Company Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel of South Arkansas, Inc. CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC CenturyTel of Wisconsin, LLC Carolina Telephone and Telegraph Company LLC North Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Telephone Company Central Telephone Company Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel Supply Group, Inc. CenturyTel/Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC Louisiana Embarq Corporation Delaware Carolina Telephone and Telegraph Company LLC North Carolin NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel/Tele-Max, Inc. CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC Louisiana CenturyTel of Wisconsin, LLC Embarq Corporation Delaware Carolina Telephone and Telegraph Company LLC North Carolin NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel of Lake Dallas, Inc. CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC Louisiana Embarq Corporation Delaware Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Kansas Centel Capital Corporation Delaware Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel Web Solutions, LLC CenturyTel of Wisconsin, LLC Embarq Corporation Delaware Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Central Telephone Company of Virginia Virginia | |
| CenturyTel of Wisconsin, LLC Embarq Corporation Delaware Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Central Telephone Company of Virginia Virginia | |
| Embarq Corporation Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Centel Capital Corporation Centel-Texas, Inc. Central Telephone Company of Texas EQ Central Texas Equipment LLC Central Telephone Company Central Telephone Company of Virginia | |
| Carolina Telephone and Telegraph Company LLC NOCUTS, Inc. Pennsylvania SC One Company Kansas Centel Corporation Kansas Centel Capital Corporation Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company | |
| NOCUTS, Inc. SC One Company Kansas Centel Corporation Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Central Telephone Company of Virginia | |
| SC One Company Centel Corporation Kansas Centel Capital Corporation Delaware Centel-Texas, Inc. Texas Central Telephone Company of Texas EQ Central Texas Equipment LLC Texas Central Telephone Company Delaware Central Telephone Company Virginia | a . |
| Centel CorporationKansasCentel Capital CorporationDelawareCentel-Texas, Inc.TexasCentral Telephone Company of TexasTexasEQ Central Texas Equipment LLCTexasCentral Telephone CompanyDelawareCentral Telephone Company of VirginiaVirginia | |
| Centel Capital CorporationDelawareCentel-Texas, Inc.TexasCentral Telephone Company of TexasTexasEQ Central Texas Equipment LLCTexasCentral Telephone CompanyDelawareCentral Telephone Company of VirginiaVirginia | |
| Centel-Texas, Inc.TexasCentral Telephone Company of TexasTexasEQ Central Texas Equipment LLCTexasCentral Telephone CompanyDelawareCentral Telephone Company of VirginiaVirginia | |
| Central Telephone Company of TexasTexasEQ Central Texas Equipment LLCTexasCentral Telephone CompanyDelawareCentral Telephone Company of VirginiaVirginia | |
| EQ Central Texas Equipment LLC Texas Central Telephone Company Delaware Central Telephone Company of Virginia Virginia | |
| Central Telephone Company Delaware Central Telephone Company of Virginia Virginia | |
| Central Telephone Company of Virginia Virginia | |
| | |
| Embarq Florida, Inc. Florida | |
| | |
| The Winter Park Telephone Company Florida | |
| CenturyLink Intellectual Property LLC Delaware | |
| Embarq Directory Trademark Company, LLC Delaware | |
| CenturyLink Public Communications, Inc. Florida | |
| CenturyLink Sales Solutions, Inc. Delaware | |
| Embarq, Inc. Kansas | |
| Embarq Capital Corporation Delaware | |
| SC Seven Company Kansas | |
| Embarq Interactive Holdings LLC Delaware | |

| Subsidiary | incorporation or formation |
|---|----------------------------|
| Embarq Interactive Markets LLC | Delaware |
| Embarq Management Company | Delaware |
| EQ Management Equipment LP | Nevada |
| Embarq Mid-Atlantic Management Services Company | North Carolina |
| Embarq Minnesota, Inc. | Minnesota |
| Embarq Missouri, Inc. | Missouri |
| SC Eight Company | Kansas |
| Embarq Network Company LLC | Delaware |
| EQ Equipment Leasing, Inc. | Delaware |
| United Telephone Company of the Carolinas LLC | South Carolina |
| SC Two Company | Kansas |
| United Telephone Company of Eastern Kansas | Delaware |
| United Telephone Company of Florida | Florida |
| United Telephone Company of Indiana, Inc. | Indiana |
| SC Four Company | Kansas |
| United Telephone Company of Kansas | Kansas |
| Embarq Midwest Management Services Company | Kansas |
| United Teleservices, Inc. | Kansas |
| United Telephone Company of New Jersey, Inc. | New Jersey |
| United Telephone Company of the Northwest | Oregon |
| United Telephone Company of Ohio | Ohio |
| SC Five Company | Kansas |
| United Telephone Company of Pennsylvania LLC, The | Pennsylvania |
| SC Six Company | Kansas |
| United Telephone Company of Southcentral Kansas | Arkansas |
| United Telephone Company of Texas, Inc. | Texas |
| EQ United Texas Equipment LLC | Texas |
| United Telephone Company of the West | Delaware |
| United Telephone Southeast LLC | Virginia |
| SC Three Company | Kansas |
| Madison River Communications Corp. | Delaware |
| Gallatin River Holdings L.L.C. | Delaware |
| Gallatin River Communications L.L.C. | Delaware |
| Madison River Finance Corp. | Delaware |
| Madison River Holdings LLC | Delaware |
| Madison River LTD Funding LLC | Delaware |
| Coastal Utilities, Inc. | Georgia |
| Gulf Coast Services, LLC | Alabama |
| Gulf Telephone Company, LLC | Alabama |
| Madison River Management LLC | Delaware |
| Mebtel, Inc. | North Carolina |
| Pacific Telecom Cellular of Alaska RSA #1, Inc. | Alaska |
| Qwest Communications International, Inc. | Delaware |
| EUnet International Limited | United Kingdom |
| | |

| Subsidiary | incorporation or formation |
|--|----------------------------|
| EUnet International B.V. | Netherlands |
| Qwest B.V. | Netherlands |
| Qwest Capital Funding, Inc. | Colorado |
| Qwest Europe LLC | Delaware |
| Qwest Services Corporation | Colorado |
| CenturyLink Investment Management Company | Colorado |
| Qwest Corporation | Colorado |
| 1200 Landmark Center Condominium Association, Inc. | Nebraska |
| Block 142 Parking Garage Association | Colorado |
| Qwest Database Services, Inc. | Colorado |
| Qwest India Holdings, LLC | Delaware |
| CenturyLink Technologies India Private Limited | India |
| Seal Infotech Private Limited | India |
| The El Paso County Telephone Company | Colorado |
| Qwest Dex Holdings, Inc. | Delaware |
| Qwest Government Services, Inc. | Colorado |
| Qwest Wireless, L.L.C. | Delaware |
| SkyComm Technologies Corporation | Delaware |
| Spectra Communications Group, LLC | Delaware |
| TelUSA Holdings, LLC | Delaware |
| Telephone USA of Wisconsin, LLC | Delaware |
| Western Re, Inc. | Louisiana |
| Wildcat Holdco LLC | Delaware |
| Level 3 Parent, LLC | Delaware |
| ACME Grating Ventures, L.L.C. | Delaware |
| AmSoft Information Services Limited | Mauritius |
| Broadwing Communications, LLC | Delaware |
| Broadwing, LLC | Delaware |
| BTE Equipment, LLC | Delaware |
| Central Host, Inc. | California |
| Continental Holdings Inc. | Wyoming |
| Continental Level 3, Inc. | Delaware |
| Continental Mineral Sales, Inc. | Delaware |
| Corvis Canada Inc. | Quebec |
| Corvis Gratings Company | Nova Scotia, Canada |
| Eldorado Acquisition Two, Inc. | Delaware |
| Fibernet UK Limited | United Kingdom |
| Front Range Insurance Company, Inc. | Hawaii |
| CCC Canada Holding, Inc. | Delaware |
| OOO "CenturyLink Communications" | Russia |
| FTV Communications, LLC | Delaware |
| CenturyLink Impsat Holdings I Limited | United Kingdom |
| CenturyLink Impsat Holdings II Limited | United Kingdom |
| GC Pan European Crossing UK Limited | United Kingdom |
| | |

| SAC Argentina Uruguay Branch Global Crossing Americas Solutions, LLC Global Crossing Americas Solutions, Inc. Puerto Rico Global Crossing International Networks Ltd. Bermuda Global Crossing International, Ltd. Bermuda Global Crossing North America, Inc. Michigan Global Crossing North America, Inc. Michigan Global Crossing North American Holdings, Inc. Delaware Global Crossing Telecommunications, Inc. Michigan Global Crossing Telecommunications, Inc. Michigan Global Crossing Telemanagement VA, LLC Virginia Impsat Fiber Networks, LLC Delaware IP Networks, Inc. Delaware Poelaware Poelaware Delaware Uruguay Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware CenturyLink Argentina, S.A. CenturyLink Colombia S.A. CenturyLink Colombia S.A. CenturyLink Colombia S.A. CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Ireland Level 3 Communications A.B. Sweden CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Buy Ltd Level 3 Communications Sulgaria EOOD Bulgaria CenturyLink Communications Euglaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A. Spain CenturyLink Communications España S.A. Socursal Emportugal CenturyLink Communications España S.A. Soc | Subsidiary | incorporation or formation |
|--|--|---|
| Global Crossing International Networks Ltd. Global Crossing International Networks Ltd. Global Crossing International, Ltd. Global Crossing International, Ltd. Global Crossing Local Services, Inc. Michigan Global Crossing North America, Inc. Michigan Global Crossing North America, Inc. Michigan Global Crossing Telecommunications, Inc. Global Crossing Telecommunications, Inc. Global Crossing Telecommunications, Inc. Global Crossing Telemanagement VA, LLC Virginia Impast Fiber Networks, LLC Delaware IP Networks, Inc. Delaware IP Networks, Inc. Delaware Legend Circle Holdings, Inc. Delaware Legend Circle Holdings, Inc. Delaware Level 3 Asia, Inc. Delaware Level 3 Asia, Inc. Delaware Level 3 Asia, Inc. Delaware CenturyLink Argentina, S.A. Chille CenturyLink Communications (Inc. Delaware CenturyLink Communications (Inc. Delaware CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Level 3 Communications A.B. Sweden CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Australia Level 3 Communications NL BV CenturyLink Communications NL BV CenturyLink Communications Shaps Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications NL BV (Greek Branch) Greece Level 3 Communications Shaps A. Level 3 Communications Shaps A. Level 3 Communications Shaps A. Level 3 Communications Erspaha S.A. Spain Level 3 Communications Erspaha S.A. Spa | SAC Argentina Uruguay Branch | Uruguay |
| Global Crossing International, Ltd. Global Crossing International, Ltd. Global Crossing International, Ltd. Global Crossing North America, Inc. Global Crossing North America, Inc. Global Crossing North American Holdings, Inc. Global Crossing Telecommunications, Inc. Uriginia Impast Fiber Networks, LLC Delaware Inc. Delaware Inc. Delaware Inc. Legend Circle Holdings, Inc. Delaware Legend Circle Holdings, Inc. Delaware Level 3 Asia, Inc. Level 3 Asia, Inc. Level 3 Asia, Inc. Level 3 Asia, Inc. Delaware Level 3 Conthemational, Inc. Delaware CenturyLink Argentina, S.A. Chile CenturyLink Colombia S.A. Chile CenturyLink Colombia S.A. Colombia Level 3 Communications (Asia Pacific) Limited Hong Kong CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Ireland Level 3 Communications A.B. Sweden CenturyLink Communications A.B. Sweden CenturyLink Communications A.B. Level 3 Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A. | Global Crossing Americas Solutions, LLC | Delaware |
| Global Crossing International, Ltd. Global Crossing International, Ltd. Global Crossing International, Ltd. Global Crossing North America, Inc. Global Crossing North America, Inc. Global Crossing North American Holdings, Inc. Global Crossing Telecommunications, Inc. Uriginia Impast Fiber Networks, LLC Delaware Inc. Delaware Inc. Delaware Inc. Legend Circle Holdings, Inc. Delaware Legend Circle Holdings, Inc. Delaware Level 3 Asia, Inc. Level 3 Asia, Inc. Level 3 Asia, Inc. Level 3 Asia, Inc. Delaware Level 3 Conthemational, Inc. Delaware CenturyLink Argentina, S.A. Chile CenturyLink Colombia S.A. Chile CenturyLink Colombia S.A. Colombia Level 3 Communications (Asia Pacific) Limited Hong Kong CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Ireland Level 3 Communications A.B. Sweden CenturyLink Communications A.B. Sweden CenturyLink Communications A.B. Level 3 Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A. | Global Crossing Americas Solutions, Inc. | Puerto Rico |
| Global Crossing North America, Inc. Global Crossing North America, Inc. Global Crossing North America, Inc. Global Crossing Telecommunications, Inc. Michigan Global Crossing Telecommunications, Inc. Michigan Global Crossing Telemanagement VA, LLC Wirginia Impast Fiber Networks, LLC P Networks, Inc. Delaware IP Networks, Inc. Delaware Will Continental Lignite, Inc. Delaware Legend Circle Holdings, Inc. Delaware Legend Circle Holdings, Inc. Delaware Level 3 Asia, Inc. Delaware Level 3 Asia, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware CenturyLink Colombia S.A. Chile CenturyLink Colombia S.A. Colombia Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (MPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Ireland Level 3 Communications AB. Sweden CenturyLink Communications AB. Sweden CenturyLink Communications AUSTIA Grobby Level 3 Communications Austrialia Pty Ltd Level 3 Communications Austria GmbH Level 3 Communications Austria GmbH Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Sulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Hong Kong CenturyLink Loeland ehf. CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Stalia Level 3 Communications Hong Kong Limited Level 3 | | Bermuda |
| Global Crossing North America, Inc. Global Crossing North America, Inc. Global Crossing North America, Inc. Global Crossing Telecommunications, Inc. Michigan Global Crossing Telecommunications, Inc. Michigan Global Crossing Telemanagement VA, LLC Wirginia Impast Fiber Networks, LLC P Networks, Inc. Delaware IP Networks, Inc. Delaware Will Continental Lignite, Inc. Delaware Legend Circle Holdings, Inc. Delaware Legend Circle Holdings, Inc. Delaware Level 3 Asia, Inc. Delaware Level 3 Asia, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware CenturyLink Colombia S.A. Chile CenturyLink Colombia S.A. Colombia Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (MPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Ireland Level 3 Communications AB. Sweden CenturyLink Communications AB. Sweden CenturyLink Communications AUSTIA Grobby Level 3 Communications Austrialia Pty Ltd Level 3 Communications Austria GmbH Level 3 Communications Austria GmbH Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Sulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Hong Kong CenturyLink Loeland ehf. CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Stalia Level 3 Communications Hong Kong Limited Level 3 | Global Crossing International, Ltd. | Bermuda |
| Global Crossing North America, Inc. Global Crossing North American Holdings, Inc. Global Crossing Telecommunications, Inc. Global Crossing Telecommunications, Inc. Global Crossing Telemanagement VA, LLC Impsat Fiber Networks, LLC IP Networks, Inc. KMI Continental Lignite, Inc. Legend Circle Holdings, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Level 3 Asia, Inc. Level 3 CDN International, Inc. CenturyLink 3 Chile S.A. CenturyLink Colombia S.A. CenturyLink Colombia S.A. Level 3 Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications A.B. CenturyLink Communications B.P Level 3 Communications A.B. CenturyLink Communications B.P Level 3 Communications A.B. CenturyLink Communications B.P Level 3 Communications B.D Level 3 Communications B.D Level 3 Communications B.D Level 3 Communications B.D Level 3 Communications S.D Level 3 Communications S.D Level 3 Communications Communications B.D Level 3 Communications C | - | Michigan |
| Global Crossing North American Holdings, Inc. Global Crossing Telecommunications, Inc. Global Crossing Telemanagement VA, LLC Impsat Fiber Networks, LLC P Networks, Inc. Delaware IP Networks, Inc. Delaware KMI Continental Lignite, Inc. Legend Circle Holdings, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware Level 3 Communications (Asia Pacific) Limited CenturyLink Argentina, S.A. Colombia Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (Asia Pacific) Limited Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV Netherlands CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Saugraia EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications Communications Communic | | - |
| Global Crossing Telecommunications, Inc. Global Crossing Telemanagement VA, LLC Impast Fiber Networks, LLC Delaware IP Networks, Inc. Delaware KMI Continental Lignite, Inc. Delaware KMI Continental Lignite, Inc. Delaware CenturyLink Argentina, S.A. Legend Circle Holdings, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware Level 3 CDN International, Inc. Delaware CenturyLink Colombia S.A. ConturyLink Colombia S.A. Level 3 CON International, Inc. CenturyLink Colombia S.A. Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (MPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications A.B. Sweden CenturyLink Communications A.B. Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV Netherlands CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications Europe Limited United Kingdom CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.a.r.I. France CenturyLink Communications France s.a.r.I. France CenturyLink Communications France s.a.r.I. Level 3 Communications Lalia Level 3 Communications Lalia Level 3 Communications Sapan KK Japan CenturyLink Communications Sapan KK Japan CenturyLink Communications CDN Ireland Limited Limited Kingdom | | Delaware |
| Global Crossing Telemanagement VA, LLC Impsaf Fiber Networks, LLC IP Networks, Inc. Delaware IP Networks, Inc. Delaware KMI Continental Lignite, Inc. Delaware Legend Circle Holdings, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware Level 3 CDN International, Inc. CenturyLink 3 Chile S.A. Colombia CenturyLink Colombia S.A. CenturyLink Colombia S.A. CenturyLink Colombia S.A. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications A.B. CenturyLink Communications B.BV CenturyLink Communications B.BV CenturyLink Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications España S.A. Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications España S.A. CenturyLink Communications Coloma Co | | Michigan |
| Impsat Fiber Networks, Inc. IP Networks, Inc. Delaware KMI Continental Lignite, Inc. Delaware CenturyLink Argentina, S.A. Level 3 Asia, Inc. Delaware Level 3 CDN International, Inc. CenturyLink 3 Chile S.A. Chile CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Paps Denmark Level 3 Communications A.B. Sweden CenturyLink Communications Aps Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Australia Level 3 Communications NL BV (Greek Branch) Greece Level 3 Communications NL BV (Greek Branch) Greece Level 3 Communications Sepaña S.A. Spain CenturyLink Communications España S.A. Level 3 Communica | - | |
| IP Networks, Inc. KMI Continental Lignite, Inc. Legend Circle Holdings, Inc. CenturyLink Argentina, S.A. Level 3 Asia, Inc. Level 3 CDN International, Inc. CenturyLink A Schile S.A. Chile CenturyLink 3 Chile S.A. CenturyLink Colombia S.A. Level 3 Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications Ireland Limited Level 3 Communications A.B. CenturyLink Communications Aps Level 3 Communications Aps Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV CenturyLink Communications NL BV (Greek Branch) CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A. Spain CenturyLink Communications España S.A. Level 3 Communications España S.A. Spain CenturyLink Communications España S.A. Spain CenturyLink Communications España S.A. Level 3 Communications España S.A. Spain CenturyLink Communications España S.A. Spain CenturyLink Communications España S.A. Level 3 Communications | | - |
| KMI Continental Lignite, Inc. Legend Circle Holdings, Inc. CenturyLink Argentina, S.A. Level 3 Asia, Inc. Level 3 CDN International, Inc. CenturyLink Colombia S.A. Colile CenturyLink Colombia S.A. Colombia Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications Pleand Limited Level 3 Communications A.B. CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications DL BV (Greek Branch) Greece Level 3 Communications Sulagria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Estonia OÜ CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited CenturyLink Ledand ehf. CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications | • | |
| Legend Circle Holdings, Inc. CenturyLink Argentina, S.A. Level 3 Asia, Inc. Level 3 CDN International, Inc. CenturyLink 3 Chile S.A. CenturyLink 3 Chile S.A. CenturyLink Colombia S.A. Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications Ireland Limited Level 3 Communications AB. CenturyLink Communications AB. CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications SU BOY CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications Estonia OÚ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Selalia Level 3 Communications Selalia Level 3 Communications Selalia Level 3 Communications CDN Ireland Limited Level 3 Communications Limited | | |
| CenturyLink Argentina, S.A. Level 3 Asia, Inc. Level 3 CDN International, Inc. CenturyLink 3 Chile S.A. CenturyLink Colombia S.A. CenturyLink Communications (Asia Pacific) Limited CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications Ireland Limited Level 3 Communications Ireland Limited Level 3 Communications A.B. CenturyLink Communications A.B. CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Sulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications Litalia Level 3 Communications Litalia Level 3 Communications Litalia Level 3 Commun | - | |
| Level 3 Asia, Inc. Level 3 CDN International, Inc. CenturyLink 3 Chile S.A. CenturyLink Colombia S.A. Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Italia Level 3 Communications Italia Level 3 Communications A.B. CenturyLink Communications ApS CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Erope Limited United Kingdom CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited CenturyLink Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | = |
| Level 3 CDN International, Inc. CenturyLink 3 Chile S.A. CenturyLink Colombia S.A. Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications Ireland Limited Level 3 Communications A.B. CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Level 3 Communications Australia Pty Ltd Level 3 Communications Austria GmbH Level 3 Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A. Level 3 Communications España S.A. Spain Level 3 Communications España S.A. Level 3 Communications Europe Limited United Kingdom CenturyLink Communications Hong Kong Limited Hong Kong CenturyLink Communications Italia Level 3 Communications Japan KK Level 3 Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited | | |
| CenturyLink 3 Chile S.A. Colombia S.A. Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Level 3 Communications Ireland Limited Level 3 Communications ApS CenturyLink Communications ApS Denmark Level 3 Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Level 3 Communications Austria GmbH Austria Level 3 Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.a.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Communications Italia Level 3 Communications Japan KK Level 3 Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | Delaware |
| CenturyLink Colombia S.A. Level 3 Communications (Asia Pacific) Limited Level 3 Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Level 3 Communications Ireland Limited Level 3 Communications A.B. Sweden CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Level 3 Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Communications Italia Level 3 Communications Japan KK Level 3 Communications Japan KK CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | |
| Level 3 Communications (Asia Pacific) Limited CenturyLink Communications (IMPSAT) Nederland B.V. Netherlands CenturyLink Communications Ireland Limited Level 3 Communications A.B. Sweden CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Austria GmbH Austria Level 3 Communications NL BV Retherlands CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications NL BV (Greek Branch) Greece Level 3 Communications Bulgaria EOOD Bulgaria CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Erope Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Communications Italia Level 3 Communications Japan KK Hungary CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | Colombia |
| CenturyLink Communications (IMPSAT) Nederland B.V. CenturyLink Communications Ireland Limited Level 3 Communications A.B. CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Austria Level 3 Communications Austria GmbH Level 3 Communications NL BV Netherlands CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications NL BV (Greek Branch) Greece Level 3 Communications Sulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain CenturyLink Communications España S.A. Spain Level 3 Communications España S.A. Spain Level 3 Communications Espana S.A Sucursal Em Portugal Level 3 Communications Espana S.A Sucursal Em Portugal Level 3 Communications Europe Limited CenturyLink Communications Europe Limited CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Communications Hong Kong Limited Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | • | Hong Kong |
| CenturyLink Communications Ireland Limited Level 3 Communications A.B. CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Australia Pty Ltd Austria Level 3 Communications Austria GmbH Austria Level 3 Communications NL BV Netherlands CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Italy Level 3 Communications Japan KK Hungary CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited United Kingdom United Kingdom | | <u> </u> |
| Level 3 Communications A.B. CenturyLink Communications ApS Denmark Level 3 Communications Australia Pty Ltd Australia Level 3 Communications Austria GmbH Level 3 Communications NL BV Netherlands CenturyLink Communications NL BV (Greek Branch) CenturyLink Communications NL BV (Greek Branch) CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Level 3 Communications Hong Kong Limited CenturyLink Iceland ehf. CenturyLink Communications Italia Level 3 Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom Level 3 Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | Ireland |
| Level 3 Communications Australia Pty Ltd Level 3 Communications Austria GmbH Level 3 Communications NL BV Netherlands CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. CenturyLink Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | Sweden |
| Level 3 Communications Australia Pty Ltd Level 3 Communications Austria GmbH Level 3 Communications NL BV Netherlands CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. CenturyLink Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | CenturyLink Communications ApS | Denmark |
| Level 3 Communications NL BV CenturyLink Communications NL BV (Greek Branch) Greece Level 3 Communications Canada Co. Nova Scotia CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Spain Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | Australia |
| CenturyLink Communications NL BV (Greek Branch) Level 3 Communications Canada Co. CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | Level 3 Communications Austria GmbH | Austria |
| Level 3 Communications Canada Co. CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | Level 3 Communications NL BV | Netherlands |
| Level 3 Communications Canada Co. CenturyLink Communications Bulgaria EOOD Bulgaria CenturyLink Communications España S.A. Level 3 Communications España S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | CenturyLink Communications NL BV (Greek Branch) | Greece |
| CenturyLink Communications España S.A. Spain Level 3 Communications Espana S.A Sucursal Em Portugal Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Italy Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Ireland Level 3 Communications Limited United Kingdom | | Nova Scotia |
| CenturyLink Communications España S.A. Sucursal Em Portugal Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Italy Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Ireland Level 3 Communications Limited United Kingdom | CenturyLink Communications Bulgaria EOOD | Bulgaria |
| Level 3 Communications Espana S.A Sucursal Em Portugal Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Italy Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited United Kingdom United Kingdom | CenturyLink Communications España S.A. | |
| Level 3 Communications Estonia OÜ Estonia CenturyLink Communications Europe Limited United Kingdom CenturyLink Communications France s.à.r.l. France CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Italy Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Ireland Level 3 Communications Limited United Kingdom | Level 3 Communications Espana S.A Sucursal Em Portugal | • |
| CenturyLink Communications France s.à.r.l. CenturyLink Communications Germany GmbH Level 3 Communications Hong Kong Limited CenturyLink Iceland ehf. CenturyLink Communications Italia Level 3 Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | Estonia |
| CenturyLink Communications France s.à.r.l. CenturyLink Communications Germany GmbH Level 3 Communications Hong Kong Limited CenturyLink Iceland ehf. CenturyLink Communications Italia Level 3 Communications Italia Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | CenturyLink Communications Europe Limited | United Kingdom |
| CenturyLink Communications Germany GmbH Germany Level 3 Communications Hong Kong Limited Hong Kong CenturyLink Iceland ehf. Iceland CenturyLink Communications Italia Italy Level 3 Communications Japan KK Japan CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Ireland Level 3 Communications Limited United Kingdom | CenturyLink Communications France s.à.r.l. | |
| CenturyLink Iceland ehf. CenturyLink Communications Italia Level 3 Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | CenturyLink Communications Germany GmbH | Germany |
| CenturyLink Iceland ehf. CenturyLink Communications Italia Level 3 Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | - |
| Level 3 Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | | |
| Level 3 Communications Japan KK CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Level 3 Communications Limited United Kingdom | CenturyLink Communications Italia | Italy |
| CenturyLink Hungary Communications Kft Hungary CenturyLink Communications CDN Ireland Limited Ireland Level 3 Communications Limited United Kingdom | • | • |
| CenturyLink Communications CDN Ireland Limited Ireland Level 3 Communications Limited United Kingdom | · | • |
| Level 3 Communications Limited United Kingdom | | - · |
| - | | United Kingdom |
| | Level 3 Communications of Virginia, Inc. | Virginia |

| Subsidiary | incorporation or formation |
|--|----------------------------|
| CenturyLink Communications Finland Oy | Finland |
| CenturyLink Communications PEC Ireland Limited | Ireland |
| CenturyLink Communications PEC Luxembourg I S.á.r.l. | Luxembourg |
| CenturyLink Communications PEC Luxembourg II S.á.r.l. | Luxembourg |
| CenturyLink Communications PEC Services Europe Limited | Ireland |
| CenturyLink Communications PEC Services Ireland Limited | Ireland |
| CenturyLink Communications Telekomünikasyon Hizmetleri Limited | Turkey |
| CenturyLink Communications RS d.o.o. Beograd-Stari Grad | Serbia |
| Level 3 Communications S.A. | Belgium |
| CenturyLink Communications Romania S.R.L. | Romania |
| CenturyLink Communications CZ s.r.o. | Czech Republic |
| Level 3 Communications Singapore Pte. Ltd. | Singapore |
| Level 3 Communications South Africa (Pty) Limited | South Africa |
| Level 3 Communications Sp. z o.o. | Poland |
| Level 3 Communications spol. s.r.o. | Slovakia |
| Level 3 Communications St. Croix, Inc. | Virgin Islands (US) |
| CenturyLink Communications Switzerland AG | Switzerland |
| CenturyLink Communications UK Limited | United Kingdom |
| Level 3 Communications, LLC | Delaware |
| CenturyLink Comunicações do Brasil Ltda. | Brazil |
| CenturyLink Ecuador S.A. | Ecuador |
| CenturyLink EMEA Holdings Limited | United Kingdom |
| Level 3 Enhanced Services, LLC | Delaware |
| Level 3 EON, LLC | Delaware |
| Level 3 Europe B.V. | Netherlands |
| Level 3 Financing, Inc. | Delaware |
| Level 3 GC Limited | Bermuda |
| Level 3 Holdings, B.V. | Netherlands |
| Level 3 Holdings, Inc. | Delaware |
| Level 3 International Services, Inc. | Delaware |
| Level 3 International, Inc. | Delaware |
| Level 3 Komunikacijske Usluge d.o.o. | Croatia |
| CenturyLink Latin American Solutions, LLC | Delaware |
| CTL Mexico II, S. de R.L. de C.V. | Mexico |
| CTL Mexico Landing S. de R.L. | Mexico |
| CTL Mexico Servicios, S. de R. L. de C. V. | Mexico |
| CenturyLink Panama, Inc. | Panama |
| CenturyLink Participaçóes e Comercial Ltda. | Brazil |
| CenturyLink Communications Norge AS | Norway |
| CenturyLink Peru S.A. | Peru |
| Level 3 Telecom Data Services, LLC | Delaware |
| Level 3 Telecom Holdings II, LLC | Delaware |
| Level 3 Telecom Holdings, LLC | Delaware |
| Level 3 Telecom Management Co., LLC | Delaware |
| | |

| Subsidiary | incorporation or formation |
|--|----------------------------|
| Level 3 Telecom of Alabama, LLC | Delaware |
| Level 3 Telecom of Arizona, LLC | Delaware |
| Level 3 Telecom of Arkansas, LLC | Delaware |
| Level 3 Telecom of California, LP | Delaware |
| Level 3 Telecom of Colorado, LLC | Delaware |
| Level 3 Telecom of DC, LLC | Delaware |
| Level 3 Telecom of Florida, LP | Delaware |
| Level 3 Telecom of Georgia, LP | Delaware |
| Level 3 Telecom of Hawaii, LP | Delaware |
| Level 3 Telecom of Idaho, LLC | Delaware |
| Level 3 Telecom of Illinois, LLC | Delaware |
| Level 3 Telecom of Indiana, LP | Delaware |
| Level 3 Telecom of Iowa, LLC | Delaware |
| Level 3 Telecom of Kansas City, LLC | Delaware |
| Level 3 Telecom of Kentucky, LLC | Delaware |
| Level 3 Telecom of Louisiana, LLC | Delaware |
| Level 3 Telecom of Maryland, LLC | Delaware |
| Level 3 Telecom of Minnesota, LLC | Delaware |
| Level 3 Telecom of Mississippi, LLC | Delaware |
| Level 3 Telecom of Nevada, LLC | Delaware |
| Level 3 Telecom of New Jersey, LP | Delaware |
| Level 3 Telecom of New Mexico, LLC | Delaware |
| Level 3 Telecom of New York, LP | Delaware |
| Level 3 Telecom of North Carolina, LP | Delaware |
| Level 3 Telecom of Ohio, LLC | Delaware |
| Level 3 Telecom of Oklahoma, LLC | Delaware |
| Level 3 Telecom of Oregon, LLC | Delaware |
| Level 3 Telecom of South Carolina, LLC | Delaware |
| Level 3 Telecom of Tennessee, LLC | Delaware |
| Level 3 Telecom of Texas, LLC | Delaware |
| Level 3 Telecom of Utah, LLC | Delaware |
| Level 3 Telecom of Virginia, LLC | Virginia |
| Level 3 Telecom of Washington, LLC | Delaware |
| Level 3 Telecom of Wisconsin, LP | Delaware |
| Level 3 Telecom, LLC | Delaware |
| Level 3 Telecom, LP | Delaware |
| Level 3 telekomunikacijski storitve d.o.o. | Slovenia |
| CenturyLink Telecomunicaciones S.A. | Venezuela |
| CenturyLink Costa Rica, S.R.L. | Costa Rica |
| Level Three Communications Israel Ltd. | Israel |
| Level Three Telecommunications Kenya | Kenya |
| CenuryLink Communications Luxembourg S.à r.l | Luxembourg |
| CenturyLink Brasil Comunicações e Serviços Ltda. | Brazil |
| CenturyLink Holding Brasil Ltda | Brazil |
| | |

| Subsidiary | or formation |
|--|-----------------|
| SAC Peru S.R.L. | Peru |
| TelCove of Pennsylvania, LLC | Delaware |
| TelCove Operations, LLC | Delaware |
| The IRC Company, Inc. | Delaware |
| Vyvx, LLC | Delaware |
| Williams Communicaciones Chile Limitada | Chile |
| WilTel Communications (Cayman) Limited | Cayman Islands |
| WilTel Communications Network, Inc. | New Brunswick |
| WilTel Communications Pty Limited | New South Wales |
| WilTel Communications, LLC | Delaware |
| WilTel International Telecom (Chile) Limited | Cayman Islands |
| XCOM Technologies of New York, Inc. | New York |
| Xspedius Management Co. International, LLC | Delaware |

Consent of Independent Registered Public Accounting Firm

The Board of Directors CenturyLink, Inc.:

We consent to the incorporation by reference in the registration statements (No. 333-227251, No. 333-202411,

No. 333-212575, No. 333-187366, and No. 333-179888) on Form S-3, the registration statements (No. 333-225154, No. 33-60061, No. 333-160391, No. 333-37148, No. 333-60806, No. 333-150157,

No. 333-124854, No. 333-150188, No. 333-174571, and No. 333-221267) on Form S-8, and the registration statements (No. 33-48956, No. 333-17015, No. 333-167339, No. 333-174291, No. 333-206725 and

No. 333-215121) on Form S-4 of CenturyLink, Inc. of our reports dated March 11, 2019, with respect to the consolidated balance sheets of CenturyLink, Inc. and subsidiaries as of December 31, 2018 and 2017, the related consolidated statements of operations, comprehensive (loss) income, cash flows, and stockholders' equity for each of the years in the three-year period ended December 31, 2018, and the related notes, and the effectiveness of internal control over financial reporting as of December 31, 2018, which reports appear in the December 31, 2018 annual report on Form 10-K of CenturyLink, Inc.

Our report dated March 11, 2019, on the effectiveness of internal control over financial reporting as of December 31, 2018, expresses our opinion that CenturyLink, Inc. did not maintain effective internal control over financial reporting as of December 31, 2018 because of the effect of material weaknesses on the achievement of the objectives of the control criteria and contains an explanatory paragraph that states material weaknesses have been identified related to (i) ineffective design and operation of process level controls over the fair value measurement of certain assets acquired and liabilities assumed in a business combination, which arose because the Company did not conduct an effective risk assessment to identify and assess changes needed to process level controls resulting from the business combination, did not clearly assign responsibility for controls over the fair value measurements, and did not maintain effective information and communication processes to ensure the necessary information was available to personnel on a timely basis so they could fulfill their control responsibilities related to the fair value measurements; and (ii) ineffective design and operation of process level controls over the existence and accuracy of revenue transactions, which arose because the Company did not conduct an effective risk assessment to identify risks of material misstatement related to revenue transactions, and included in management's assessment.

/s/ KPMG LLP

Shreveport, Louisiana March 11, 2019

CERTIFICATION

I, Jeff K. Storey, Chief Executive Officer, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of CenturyLink, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15 (e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's Board of Directors:
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

| Date: March 11, 2019 | /s/ Jeff K. Storey |
|----------------------|-------------------------|
| | Jeff K. Storey |
| | Chief Executive Officer |

CERTIFICATION

- I, Indraneel Dev, Executive Vice President and Chief Financial Officer, certify that:
 - 1. I have reviewed this Annual Report on Form 10-K of CenturyLink, Inc.;
 - 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
 - 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15 (e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's Board of Directors:
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

| Date: March 11, 2019 | /s/ Indraneel Dev |
|----------------------|------------------------------------|
| | Indraneel Dev |
| | Executive Vice President and Chief |
| | Financial Officer |

Chief Executive Officer and Chief Financial Officer Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

Each of the undersigned, acting in his capacity as the Chief Executive Officer or Chief Financial Officer of CenturyLink, Inc. ("CenturyLink"), certifies that, to his knowledge, the Annual Report on Form 10-K for the year ended December 31, 2018 of CenturyLink fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934 and that the information contained in the Form 10-K fairly presents, in all material respects, the financial condition and results of operations of CenturyLink as of the dates and for the periods covered by such report.

A signed original of this statement has been provided to CenturyLink and will be retained by CenturyLink and furnished to the Securities and Exchange Commission or its staff upon request.

| Date: March 11, 2019 | |
|-------------------------|--|
| /s/ Jeff K. Storey | /s/ Indraneel Dev |
| Jeff K. Storey | Indraneel Dev |
| Chief Executive Officer | Executive Vice President and Chief Financial Officer |